



arin County is known for its creativity, innovation, and high quality of life. This lifestyle depends a great deal on the beautiful natural setting and excellence of residential neighborhoods in Marin, as well as many other factors that affect how people learn, work, obtain goods and services, and play. The Socioeconomic Element of the Countywide Plan focuses on the people of Marin County and seeks to reinforce the complex connections between individual well-being, economic prosperity, community involvement, cultural richness, and the environment.

Because a truly healthy community embraces and cares for its least fortunate members, the Socioeconomic Element emphasizes the need for universal health care, abundant child care, community policing, full civic participation, open access to information, education and the arts, proper nutrition and physical fitness. The Element seeks to enhance quality of life for everyone in Marin. Its policies and programs seek to improve conditions for disadvantaged and underrepresented groups, bolster a strong and diverse economy, and engender fair and just social relationships within the Marin community.

Below are the topics covered in this portion of the Countywide Plan:

- ♦ Economy
- ◆ Child Care
- Public Safety
- Community Participation
- Diversity
- ♦ Education
- Environmental Justice
- Public Health
- Arts and Culture
- Historical and Archaeological Resources
- Parks and Recreation

Topics related to environmental hazards, housing, transportation, noise, and community facilities are located in the Built Environment Element.



4.1-2

# 4.2 Key Trends and Issues

## Economy

Marin County has enjoyed relative prosperity and economic diversity during the past decade. Many businesses have endured and prospered in Marin. However, some companies have grown to the extent that they have had difficulty finding adequate space and workers who can afford to live in Marin, and in some cases have moved out of the county. Increasing labor costs, traffic congestion, and a shortage of affordable housing have impacted local business viability. Agricultural operations generally have not benefited from trends that have buoyed other sectors of the economy.

## Child Care

While the availability of licensed child care is increasing, demand continues to outpace supply for both infant and school-age children. Estimates from 2001 indicate that local providers could serve fewer than 10,000 of the more than 24,000 children countywide who needed care, and that this gap is most severe for infants (ages 0–2) and for after-school care. The need for child care is expected to grow as the cost of living in Marin remains high, the local employment base expands to include more women, and implementation of welfare reform continues. Projected job growth in lower-paying service and retail trade sectors is likely to increase the need for subsidized and affordable child care even further.

### **Public Safety**

The area of public safety covers a variety of factors: law enforcement, fire protection, the criminal justice system, and emergency preparedness. Marin's crime rate has been consistently lower than the state's for many years. In addition, overall crime in the county has decreased slightly, from 7,533 reported instances in 1996 to 6,724 in 2005. However, although the number of domestic violence calls per capita is significantly lower than the state average, the proportion of physical abuse cases in Marin has been higher than the number of cases statewide since 1998.

### **Community Participation**

The percentages of voter registration and election turnout are higher in Marin County than for the state as a whole. According to a 2001 survey, the percentage of county residents indicating that they spend time volunteering for charitable, political, and community service activities is higher than the national average. The same survey shows that charitable giving is well above the national average for organizations related to human services, the environment, and the arts and humanities; equivalent to the national average for educational organizations; and below the national average for religious and health organizations. However, not all segments of the population participate in civic activities such as voting, and minority groups are underrepresented in appointed community and advisory groups. In particular, there is not enough ethnically diverse participation in community decision making.

### Diversity

The ethnic diversity of Marin's population is low but is increasing. In 1990, 88.7% of the population was white and 11.3% was African-American, Asian, Pacific Islander, or other races. People of Hispanic

origin (who may be of any racial group) made up 7.8% of the population. According to the 2000 Census, the nonwhite population increased to 16% and the Hispanic population to 11.1%. The white population was 84%.

#### Education

The public education system stands ready to serve all children even though the resources are limited. While Marin County has one of the lowest dropout rates in the state, as long as one student drops out, there is room for improvement. Not all children have access to early education programs and to quality education. Educational inequities exist based on income, geography, and race. Graduation and dropout rates need to be monitored to determine how best to improve educational opportunities for traditionally underserved populations.

#### **Environmental Justice**

Environmental injustice is indicated by the disproportionate level of toxins and other health hazards affecting lower income communities. People in these areas are less able to afford pesticide-free food, and children are more likely to be exposed to lead-based paint and pollutants in the air, soil, and water. Some of the hazardous materials present in Marin (such as fuel and batteries) are produced and



"At all levels and in all realms, people must have a say in the decisions that affect their lives."

- The Environmental Justice and Climate Change Initiative

disposed of elsewhere, thereby creating additional health impacts in other (often less affluent) communities.

#### **Public Health**

Despite the general good health of county residents, some disturbing concerns face the Marin population. The breast cancer rate in Marin is among the highest in the United States, and other cancer rates are high, including that for prostate cancer. Obesity is prevalent among both adults and children. Targeted health issues in the county also include hepatitis C, heart disease, asthma, and environmental illnesses. Recent changes in Medicare may impact insurance coverage for the increasingly aging county population, who, along with other special needs groups, face a shortage of accessible and affordable care services.

#### Arts and Culture

Marin is a culturally rich community, and the arts industry is a strong contributor to the local economy and quality of life. In 1999, the arts industry in Marin employed roughly 2,200 persons. In 2000, approximately 22% of Marin households gave to the arts and humanities, compared with 11% nationally. Although wages in the arts industry remain relatively low, they increased 13% between 1998 and 1999 from an annual average wage of \$23,000 to \$26,000.

## **Historical and Archaeological Resources**

Marin County contains 630 recorded archaeological sites. These sites include settlements and villages, hunting camps, quarries, rock art, and trails associated with Native American habitation of the area. The distribution of known archaeological sites in the county is concentrated in urban areas and on the Point Reyes Peninsula.

## **Parks and Recreation**

Parks and recreational amenities are critical to the quality of life, and therefore the economy. Marin County residents and visitors are fortunate to have access to nearly half of the land in the county as parks and open space, including approximately 500 miles of trails through much of this land. City, county, state and national parks offer a wide variety of recreational opportunities, from hiking and sightseeing to soccer, golf, and baseball. Visitors to Marin support a \$500 million per year tourism industry.

# 

MARIN COUNTYWIDE PLAN

# 4.3 Framework

# The Vision

The 21st century in Marin will include a diverse and vital economy that is increasingly less dependent on fossil fuels, hazardous chemicals, and manufactured substances that accumulate in nature. Community members will enjoy rich cultural diversity and participate in community activities, recreation, and the arts. Affordable choices for housing and child and elder care in the workplace and in the community will complement high-quality education and services available to people of all ages, cultures, and income levels. Support systems will be in place to help those in need, and families will live, work, and play in a safe and healthy environment.

Topics in the Socioeconomic Element include:

**Economy (see Section 4.4):** Actions required to ensure economic vitality in Marin are described in this Section of the Countywide Plan. A diverse mix of commercial uses provides jobs, stimulates capital investment, and supports public services. Maintaining a strong economy will rely on retaining existing businesses and creating limited opportunities for them to expand, as well as appropriately attracting new commercial enterprises to the county. It also will require finding ways to provide goods and services needed by residents and visitors that currently are in short supply. (See the Housing and Transportation sections of the Built Environment Element for additional programs pertaining to the economy.)

**Child Care (see Section 4.5):** Increasing the number of child care facilities countywide will be encouraged through zoning and permitting revisions and incentives for developers and employers. Identifying appropriate sites for child care would also assist in the establishment of new facilities. Enhanced child care options will be encouraged further through school programs, training for providers and parents, and financial support.

**Public Safety (see Section 4.6):** Community involvement in public safety issues helps to keep Marin's neighborhoods safe. Participation will be encouraged in improved design and management of public facilities, fire protection, emergency preparedness, and crime control. Residents will be encouraged to be proactive in the identification of issues in their neighborhoods that may diminish safety, such as overcrowding, interpersonal disputes, and unreported crimes. Community educational, restorative justice, and neighborhood awareness programs will complement traditional law enforcement roles. Community involvement in crime control and improved design and management of public facilities will be encouraged to keep Marin's neighborhoods safe. Traditional law enforcement will be complemented by therapeutic and restorative concepts in the criminal justice system. Mandatory counseling for perpetrators and safe havens for victims of domestic violence can aid in post-abuse recovery, and community policing efforts may help prevent crime. Expansion of recreational and substance control programs, including those for youth, also may help reduce the threat of crime and foster community pride.

**Community Participation (see Section 4.7):** Groups that historically have not been as involved in the community need to be given opportunities for increased participation. Public meetings should be made

accessible to all interested citizens through outreach, appropriate locations, translation, and child care. Balanced ethnic representation on County committees and commissions will also be needed.

**Diversity (see Section 4.8):** Diversity should be celebrated at community events, workplaces, and schools. Reaching out to, recognizing, and encouraging leadership among minority communities will go a long way toward including currently underrepresented groups in important decision making.

**Education (see Section 4.9):** Adequate educational opportunities will be sought through after-school and summer programs, and through adult education such as parenting and English as a Second Language (ESL) classes. Library programs may be enhanced through marketing and teen and adult volunteer involvement.

**Environmental Justice (see Section 4.10):** A healthy environment will be sought for all Marin residents through reduction of toxins, particularly those concentrated in lower income residential areas.

**Public Health (see Section 4.11):** Preventive treatment and universal access to care will be promoted by working with local health care agencies. Healthy lifestyles and living and work environments will be a primary focus of these programs.

Arts and Culture (see Section 4.12): Efforts will be undertaken to increase access to arts and culture in the county, heighten awareness of existing cultural resources, and expand opportunities for local artists and performers.

Historical and Archaeological Resources (see Section 4.13): Preservation of cultural and archaeological sites will be enhanced through requirements for surveying and protecting resources, and collaboration with other agencies.

**Parks and Recreation (see Section 4.14):** County parks will need to continue to provide opportunities for active recreation, including playing fields, swimming pools, golf courses, tennis courts, picnic areas, and children's playgrounds.

# 4.4 Economy

## Background

Commercial activity creates income that allows people to pursue the lifestyles they value, and it generates revenue that pays for the services needed to maintain a safe and healthy environment. Therefore, economic vitality in Marin is integrally related to issues addressed throughout the Countywide Plan. For example, economic activity responds to, and creates demand for, changes in the land use pattern, while transportation infrastructure affects the ability of workers to travel to and from jobs, and the ability of businesses to receive and deliver goods and services.

A robust economy relies on a range of commercial activities broad enough to compensate for adversities in any one industry and to weather larger economic cycles. This Section of the Countywide Plan seeks to attract and retain businesses that provide goods and services needed locally in an environmentally aware manner, and that offer stable, living wage employment in interesting, pleasant, and healthy work environments close to employee residences or transit.

The policies, programs, and data presented in this Section derive from an Economic Element prepared in conjunction with the Marin Economic Commission (see the reference to *Marin Countywide Plan Supporting Documents* in the "Technical Background Reports and Other Supporting Documents" section of the Introduction to the Countywide Plan). Because only about 10% of Marin's economic activity and 2% of its jobs<sup>1</sup> are located in the unincorporated county, the policies and programs in this Section will prove more effective if also adopted by local towns and cities.



#### Four Principles for Economic Sustainability

- 1. Plug the leaks. Where possible, stop the outflow of local dollars by producing goods locally that Marin residents consume, or using the ones we do import more efficiently.
- 2. Support existing businesses.
- 3. Encourage new local enterprise for example, by adding value before exporting, and facilitating lending (through special microenterprise banks or other arrangements).
- 4. Recruit new businesses that are compatible with existing businesses. In this context, "compatible" means that a new business develops underutilized resources, meets needs unfulfilled by existing businesses, complements existing economic activities, and is consistent with community social and environmental values.
- Source: Rocky Mountain Institute's Economic Renewal Program: An Introduction, by Michael J. Kinsley.

<sup>&</sup>lt;sup>1</sup>Based on Association of Bay Area Governments (ABAG) 2002 projections. Because these projections assign employment and population to sub-county areas based on planning boundaries rather than city limit lines, estimates and projections for Corte Madera, Fairfax, Larkspur-Kentfield, Mill Valley, Novato, Ross, San Anselmo, San Rafael, Sausalito, Marin City, and Tiburon may also cover some of the surrounding unincorporated county.

# **Key Trends and Issues**

### Is there enough commercial space?

Between 1989 and 2002, more than 6.5 million square feet of office, retail, and industrial space was built in the county. However, during that same period, more than 40 companies left Marin, vacating about 2.5 million square feet of primarily office space (7.2% of the total countywide). Since then, county commercial vacancy rates have steadily decreased. In 2006, commercial vacancy rates for office, retail, and industrial spaces stood at 16.4% and 2.8%, respectively.<sup>1</sup>

### Why is it difficult for businesses to locate and stay in Marin?

Commercial construction is becoming increasingly expensive, and the high cost of housing locally requires companies in Marin to pay higher wages than they might elsewhere. Reasons for businesses leaving include high rents, difficulty in recruiting and retaining employees due to the high cost of housing and long commute times, and increased cost of transporting goods along the often-congested City-Centered Corridor.

# What kind of job base does Marin have?

The county job base — estimated at more than 120,000 workers in 2000 — continues to grow by more than 1% annually. Almost half of the employees in the county work in the service and retail sectors, and jobs are becoming more concentrated there. The ratio of jobs per household (more than 1.2 in 2000) is expected to increase, largely because housing costs require more than one income to support a household.

### What special attributes do Marin businesses have?

Marin companies tend to be highly productive, and the average Marin worker produces 5% more revenue than the average U.S. worker. A high percentage of Marin County businesses have fewer than ten employees, and nearly one-quarter are home based. The Marin economy is also highly successful at creating jobs, with a 20-year job growth rate nearly 10% higher than the national and California averages. In addition, Marin continues to be a center of creativity and innovation, including examples ranging from the rise of the mountain bike industry to a high concentration of multimedia enterprises. Marin is also home to a higher-than-average concentration of artists, designers, small and home-based firms, and managerial and professional workers.

### What kinds of businesses does the county need?

A Targeted Industries Study completed in January 2004 for the Marin Economic Commission and the Community Development Agency concluded that the County must be proactive in ensuring that both local-serving and broader-based businesses thrive, and that neither sector comes to dominate the local economy. The study determined that the industries listed in Figure 4-2 present growth opportunities that can help address the key economic development issues facing Marin: the need to more closely link jobs with housing, traffic congestion, land use constraints, and social inequity.

<sup>1</sup>Source: Keegan & Coppin, Inc.

#### Figure 4-1 Business Building Blocks



# What Are the Desired Outcomes?

# **GOAL EC-I**

**A Vibrant Economy.** Establish and maintain a diverse and sustainable local economy.

## Policies

**EC-1.1** Attract and Retain Businesses. Support businesses that contribute to a robust, viable, and sustainable economy, and are consistent with the goals and policies of the Countywide Plan.



**EC-1.2 Provide Land for Commercial Enterprise.** Ensure that adequate and appropriate sites suitable for commercial uses are available.



*"Wasting resources costs the earth and lowers your competitive edge."* 

 Sustainable Business Team, Government Office for the South West, UK, 2000 **EC-1.3 Promote Green Business.** Support businesses that utilize environmentally sound practices.

**EC-1.4 Implement the Recommendations of the Targeted Industries Study.** Continue to refine the County's overall economic agenda and identify specific action steps for updating and achieving the recommendations of the Targeted Industries Study.

**EC-1.5 Consider the Impacts of Climate Change.** Identify strategies to protect the economy from the impacts of sea level rise, natural disasters, and disease outbreaks.

#### Why is this important?

Retaining and attracting a diversity of businesses that are a good match with the goals and needs of the local community yields more local benefits.

**Environment:** The average piece of produce in California in 2001 traveled 1,494 miles to reach the consumer. Purchase and use of locally produced goods and services reduces transportation costs and impacts, and thereby reduces greenhouse gas emissions and our ecological footprint. Green business practices by local companies further reduce the environmental impacts of economic activity.

**Economy:** Purchase of locally produced goods and utilization of locally provided services recirculate dollars within the community, improving the health of the local economy and generating the greatest local benefit from each dollar spent.

**Equity:** Recirculation of local dollars increases the opportunity for local employment by demand for additional local services and products.

Existing Clusters	Targeted Businesses
Real Estate and Construction	Green Building
Business Services	Boutique Consulting Environmental Technology
Multimedia	Digital Imaging (Motion Pictures) Interactive Media and Game Development Engineering and Design Software

Figure 4-2	Existing	and	Targeted	<b>Businesses</b>
I Igui C H-L	LAISCING	anu	i ai getteu	Dusinesses



Existing Clusters	Targeted Businesses
Finance and Insurance	Integrated Wealth Management Services Online Financial Services Personal Financial Advising
Restaurants and Tourism	Agri-Tourism Outdoor Recreation and Equipment Arts and Crafts
Health Services	Alternative Healing and Meditation Alternative Medicine Biotech Emergent Care
Agriculture	Organic Value-Added (Niche) Agricultural Products Food Product Manufacturing

Source: Marin Economic Commission, Targeted Industries Study, 2004.



#### How will results be achieved?

# **Implementing Programs**

EC-1.a

**Retain and Attract Appropriate Companies.** Work with local cities and towns, chambers of commerce, and other business groups to ensure that businesses' retention and attraction efforts are directed toward companies (see Figure 4-3) that will

- supply goods and services needed locally (especially if currently unavailable);
- create few or no adverse environmental impacts and participate in recycling and similar green business and sustainable partner programs;



- contribute to social equity by providing for employee health insurance, child care, and other needs;
- provide jobs that reduce the need for residents to commute to work outside the county;
- offer employees options such as carpooling, transit subsidies, flexible hours, and home-based work that help ease vehicle dependency and traffic congestion;
- encourage and support affordable housing efforts;
- be locally owned businesses.

	Screening Criteria
Economic	Above-average wages
	Locally owned businesses
	Emphasis on value-added activities
	Primary vs. secondary engine of growth <sup>1</sup>
	High productivity
	Occupational diversity and upward mobility
	Industry diversity
Environment	Average firm size <sup>2</sup>
	Reduce dependence on inputs from other regions
	Employs local residents
	Telecommuting or transit friendly
	Allows flextime
	Potential sustainable partner
Equity	Creative and innovative
	Links to aging population
	Consistent with County goals and principles

#### Figure 4-3 Targeted Industries Screening Criteria

Source: Marin Economic Commission, Targeted Industries Study, 2004.

**EC-1.b** *Streamline Minor Project Review.* Amend the Development Code to streamline review for minor projects with minimal environmental impact, such as interior tenant improvements, that enhance development for businesses targeted in program EC-1.a.

<sup>1</sup>Businesses targeted for the county should include primary engines of growth (attracting wealth and investment to the region) rather than only secondary, local-serving activities.

<sup>2</sup>The average size of firms targeted for the county should be no bigger than the largest firms currently operating in Marin.

#### Figure 4-4 Marin Target Industries

Boutique Consulting	Personal Financial Advising
Information Technology	Online Financial Services
Engineering	Interactive Media and Game Development
Management	Engineering and Design Software
Telecommunications	Organic Value-Added Agriculture Products
Other Professional Fields	Food Product Manufacturing
Environmental Technology	Agri-Tourism
Green Building	Outdoor Recreation and Equipment
Architecture Services	Arts and Crafts
Research and Development	Alternate Healing and Meditation
Construction	Alternative Medicine
Links to Environmental Technology	Biotech
Integrated Wealth Management	Emergency Care Services

EC-1.c	<b>Facilitate Digital Infrastructure.</b> Amend
	the County Development Code as
	necessary to facilitate installation of
	digital communications infrastructure for
	businesses.

- EC-1.d *Involve the Economic Commission.* Support the work of the Marin Economic Commission to inform decision makers regarding economic policy.
- EC-1.e Solicit Input from the Workforce Investment Board. Seek assistance from the Workforce Investment Board regarding the needs and recommendations for training and retraining the workforce.
- EC-1.f Inventory Available Space. Work with local cities and towns, chambers of commerce, and real estate representatives to inventory existing business space, as well as vacant and underutilized commercial sites.
- EC-1.g *Intensify Uses.* Encourage the Redevelopment Agency to pursue

"In [our company], there is a direct link between the energy we consume and the emissions we release into the environment. Cutting energy consumption means reducing emissions — it's a simple equation. And cutting energy consumption also makes good business sense. . . . [O]ne of the ways we become more competitive is by driving down energy costs by becoming more energy efficient."

 Dan Paszkowski, Vice President, Economic Affairs, Mining Association of Canada (MAC)

# ĸ

intensification and reuse of underutilized sites that further the goals and policies of the Countywide Plan.

- **EC-1.h** *Encourage Transit-Oriented Development.* Work with local cities and towns to encourage patterns of commercial development that support use of public transit, including modifying development regulations to facilitate commercial and/or mixed use projects at sites near transit stops.
- **EC-1.i Buy Green and Low-Packaging Products.** Purchase products from local green businesses (certified by appropriate authorities) and those that have minimal or no packaging and high recycled-material content; use renewable energy and environmentally friendly printing resources whenever possible.

**EC-1.j** *Promote Green Purchasing.* Encourage public agencies and private institutions to establish sustainable procurement programs through educational forums and access to information.

Sustainable Partner Business Standards include:

- Reduce environmental impacts by using closed-loop systems.
- Inventory greenhouse gas emissions and demonstrate a reduction of fossil fuel consumption.
- Enact a strong environmental purchasing policy.
- Provide on-site or subsidized nearby child care.
- Voluntarily comply with the Marin County "living wage" ordinance.
- Conduct education about environmental and sustainability issues.
- Commit to corporate philanthropy.

**EC-1.k** *Provide Assistance with Green Practices.* Expand the Green Business, Sustainable Partner, and Building Energy Efficient Structures (BEST) programs to provide technical and permitting assistance to businesses seeking to comply with environmental regulations (such as non-point pollution source water quality controls).

**EC-1.1** *Study Targeted Businesses.* Assess targeted businesses to determine the extent to which the industries are integrated into the fabric of Marin's economy and the feasibility of further growth in Marin (see Figure 4-1).

**EC-1.m** *Partner with the Private Sector.* Initiate a collaborative process with the business community and others such as the College of Marin and the Buck Institute to implement the recommendations of the 2004 Targeted Industries Study, including business mentoring, telecommuting incubation, workforce preparation, jobs/housing initiatives, and cohesive County marketing strategies and programs.

**EC-1.n** *Facilitate Review for Targeted Uses.* Amend the discretionary review process as necessary to facilitate project review for desired uses targeted in program EC-1.a, and utilize the preapplication review to help proponents understand and navigate the review process.

- **EC-1.0** *Incorporate Economic Impacts of Climate Change into Planning.* Consider integrating economic disaster planning into disaster preparedness and mitigation plans, and analyze impacts to the economy from climate change.
- **EC-1.p** *Implement Economic Programs.* Consider retaining an Economic Sustainability Specialist to implement economic programs.

# What Are the Desired Outcomes?

# **GOAL EC-2**

**Social Equity in the Workplace.** Ensure that all persons have access to meaningful employment with fair compensation, adequate benefits, and a decent work environment.

# **Policies**

EC-2.1	<b>Expand Job Opportunities.</b> Improve employment prospects for county residents.
EC-2.2	<b>Lead by Example.</b> Implement socially responsible business practices that increase the health of the
	community and the productivity of County operations



### Why is this important?

Studies show that workers who are compensated fairly report a greatly enhanced sense of recognition for work. This is linked to increased job commitment, reduced turnover, and increased productivity. Productive members of the workforce are an essential component of a healthy economy and healthy community.

**Economy:** A study commissioned by the Economic Policy Institute in Washington, D.C. and carried out by researchers at the Johns Hopkins University determined that providing jobs with a living wage increases productivity and reduces job turnover significantly. High productivity and low job turnover contribute to a strong economy.

**Equity:** The poverty rate for residents of Marin County went from 6.4% in 1993 to 7.0% in 1997. The average Marin County wage in 2003 was \$47,013, while the median price of a home stood at \$657,500. Job training and work opportunities with benefits and good working conditions will improve the quality of life for many Marin residents.

# <u>₹ЛЛХ ЛХ Х М</u> MARIN COUNTYWIDE PLAN

## How will results be achieved?

# Implementing Programs

- **EC-2.a** *Promote Job Training.* Update publications listing available vocational and technical skills programs (including those in languages other than English spoken by a significant percentage of county residents), and work with the Marin Employment Connection to place unemployed residents (including youth) in appropriate skill enhancement programs.
- **EC-2.b** *Encourage Employment.* Work with local public and private employment advocacy groups to encourage businesses, especially the County of Marin, to provide jobs for youth, senior citizens, people with disabilities, the homeless, and other traditionally underemployed groups.
- **EC-2.c** *Consider Employment Services for Day Laborers.* Work with the Marin Employment Connection and other interested organizations to explore the creation of a community hiring hall or other employment services for day laborers.
- **EC-2.d** *Provide Employee Support Services.* Strongly encourage the provision of employee support services, including child care, in conjunction with County approval of large mixed-use and commercial projects.
- **EC-2.e** *Offer a Range of Jobs.* Offer part-time, entry level, intern, and job-sharing positions at the County.
- **EC-2.f** *Pay Living Wages.* Provide fair compensation in accordance with the County living wage ordinance.
- **EC-2.g** *Offer Workplace Flexibility for Parents.* The County of Marin should consider becoming a model employer by allowing working parents to share jobs and telecommute, and by providing on-site child care and/or child care subsidies.

熊谷

# Figure 4-5 Relationships of Goals to Guiding Principles

This figure illustrates the relationships of each goal in this Section to the Guiding Principles.

Goals Goals	1. Link equity, economy, and the environment locally, regionally, and globally.	2. Minimize the use of finite resources, and use all resources efficiently and effectively.	3. Reduce the use and minimize the release of hazardous materials.	4. Reduce greenhouse gas emissions that contribute to global warming.	5. Preserve our natural assets.	6. Protect our agricultural assets.	7. Provide efficient and effective transportation.	8. Supply housing affordable to the full range of our workforce and diverse community.	9. Foster businesses that create economic, environmental, and social benefits.	10. Educate and prepare our workforce and residents.	11. Cultivate ethnic, cultural, and socioeconomic diversity.	12. Support public health, public safety, and social justice.
EC-1 A Vibrant Economy	•	•	•	•		•	•	•	•	•	•	•
EC-2 Social Equity in the Workplace	•	•		•				•	•	•	•	•

#### How Will Success Be Measured?

# **Indicator Monitoring**

Nonbinding indicators, benchmarks, and targets<sup>1</sup> will help to measure and evaluate progress. This process will also provide a context in which to consider the need for new or revised implementation measures.

Indicators	Benchmarks	Targets <sup>1</sup>
Gross county production in major sectors: agriculture, retail, manufacturing, services, etc.	See Marin Profile dataset <sup>2</sup> for 2000.	Increase 10% by 2020.
Rate of unemployment.	1.7% in 2000.	Remain in the lowest 10% of California counties through 2020.
Number of individuals below the poverty level.	15,601 individuals in 1999.	Track number of individuals below poverty level through 2020.
Number of certified green businesses.	0 in 2000.	Increase to 250 by 2010, and 400 by 2015.
Number of "Sustainable Partner" certified businesses.	0 in 2000.	Increase to 50 by 2015, and 100 by 2020.
County bond rating.	See dataset <sup>3</sup> for 2000.	No decrease in bond rating through 2020.

<sup>1</sup>Many factors beyond Marin County government control, including adequate funding and staff resources, may affect the estimated time frames for achieving targets and program implementation.

<sup>2</sup>Available through the Marin County Community Development Agency.

<sup>3</sup>Available through the Marin County Tax Assessor's office.

# **Program Implementation**

The following table summarizes responsibilities, potential funding priorities, and estimated time frames for proposed implementation programs. Program implementation within the estimated time frame<sup>1</sup> will be dependent upon the availability of adequate funding and staff resources.

Program	Responsibility	Potential Funding	Priority	Time Frame
EC-1.a – Retain and Attract Appropriate Companies.	CDA	Existing budget and may require additional grants or revenue. <sup>2</sup>	High	Short term
EC-1.b – Streamline Minor Project Review.	MEC, CDA	Existing budget	High	Short term
EC-1.c – Facilitate Digital Infrastructure.	CDA, Marin Telecommunications Authority (MTA)	Existing budget	Medium	Med. term
EC-1.d – Involve the Economic Commission.	MEC, CDA	Existing budget	High	Short term
EC-1.e – Solicit Input from the Workforce Investment Board.	MEC, CDA	Existing budget	High	Short term
EC-1.f – Inventory Available Space.	MEC, CDA	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Short term
EC-1.g – Intensify Uses.	CDA	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Ongoing
EC-1.h – Encourage Transit-Oriented Development.	CDA, CWPA, TAM, Marin Cities and Towns	Existing budget	High	Ongoing
EC-1.i – Buy Green and Low-Packaging Products.	DPW, Other County Departments	Will require additional grants or other revenue <sup>2</sup>	Medium	Long term
EC-1.j – Promote Green Purchasing.	DPW, CDA	Will require additional grants or other revenue <sup>2</sup>	Medium	Long term

#### Figure 4-6 Economy Program Implementation

<sup>1</sup>Time frames include: Immediate (0-1 years); Short term (1-4 years); Med. term (4-7 years); Long term (over 7 years); and Ongoing (existing programs already in progress whose implementation is expected to continue into the foreseeable future). <sup>2</sup>Completion of this task is dependent on acquiring additional funding. Consequently, funding availability could lengthen or shorten the time frame and ultimate implementation of this program.



Program	Responsibility	Potential Funding	Priority	Time Frame
EC-1.k – Provide Assistance with Green Practices.	CDA	Existing budget	High	Immediate
EC-1.1 – Study Targeted Businesses.	MEC, CDA	Existing budget	High	Immediate
EC-1.m – Partner with the Private Sector.	MEC	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Med. term
EC-1.n – Facilitate Review for Targeted Uses.	CDA, MEC	Existing budget	Medium	Short term
EC-1.0 – Incorporate Economic Impacts of Climate Change into Planning.	CDA, OES, and Disaster Council	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Med. term
EC-1.p – Implement Economic Programs.	CDA, OES, and Disaster Council	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Med. term
EC-2.a – Promote Job Training.	Workforce Investment Board (WIB), Marin Employment Connection, MEC	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Med. term
EC-2.b – Encourage Employment.	WIB, MEC, HR	Existing budget	Medium	Long term
EC-2.c – Consider Employment Services for Day Laborers.	Marin Employment Connection and CBO's	Will require additional grants or revenue <sup>2</sup>	Medium	Long term
EC-2.d – Provide Employee Support Services.	CDA, H&HS	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Short term
EC-2.e – Offer a Range of Jobs.	HR, All County Departments	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Ongoing
EC-2.f - Pay Living Wages.	All County Departments	Existing budget	High	Ongoing
EC-2.g – Offer Workplace Flexibility for Parents.	BOS, CAO, HR, H&HS	Will require additional grants or revenue <sup>2</sup>	Medium	Long term



©MarinCARES

# 4.5 Child Care

# Background

Adequate child care is an essential component of the Marin social and economic fabric, and demand significantly exceeds the capacity of licensed providers (see Figure 4-7). Policies and programs intended to ease this shortage through employer provision of child care are contained both in this Section and in Section 4.4.



Figure 4-7 Licensed Child Care Supply and Demand in Marin County, 2002

Changes to County regulations can increase the availability of sites for child care provision, which is critical because existing locations are being lost. Subsidies also are needed to ensure that existing child care facilities can continue to operate and that new ones will have the opportunity to locate in Marin. New funding sources may be required to ensure quality child care for those who need it.

# **Key Trends and Issues**

#### What are the economics of child care in Marin?

Child care is essential for labor-force participation and local economic development. Although licensed child care is a \$57.5 million industry in Marin County, parent fees alone cannot cover the full cost of care. Currently parent fees account for approximately 85% of the cost of licensed child care in Marin. Government subsidies account for less than 10% of Marin's licensed child care cost, and less than 1% is provided by the corporate sector.

#### Why aren't there more child care providers?

School facilities once used for child care are reverting to classroom use, and holding on to commercial and residential space for child care is proving difficult in the face of competition with other uses. Child care providers tend not to have experience with the development review process, which can impede establishment of new and expanded facilities. The low wages traditionally paid to child care workers coupled with the high cost of housing make it difficult to retain qualified staff.

# What Are the Desired Outcomes?

# GOAL CH-I

Additional Child Care Facilities. Increase the number of child care facilities countywide.

#### **Policies**

CH-1.1	<b>Encourage Development of Affordable Child Care</b> <b>Facilities.</b> Facilitate establishment of affordable child care facilities in all residential zones and at appropriate community facilities.
	community facilities.



- CH-1.2 Establish Child Care Requirements for Development. Require on-site child care or inlieu fees for child care in new public and private mixed-use and nonresidential development.
- CH-1.3 Streamline Permitting for Child Care Facilities. Evaluate and determine how to expedite the permitting process for large family day care homes and child care facilities.
- CH-1.4 Provide Regulatory Incentives for Child Care Facilities. Modify County regulations to explicitly encourage development of child care facilities.

	Small Family	Large Family	Child		
Standards	Child Care Homes	Child Care Homes	Care Centers		
	Allowed within any single-	Allowed within any single-	Allowed in the zoning		
Zoning	family residence located in	family residence located in	districts determined by		
	an agricultural or residential	an agricultural or residential	Article II (Zoning Districts		
	zoning district	zoning district	and Allowable Land Uses)		
		Requires approval of a	Requires approval of a		
Permit	No land use permit	Large Family Day-Care Use	Child Day-Care Center Use		
Requirements	required	Permit by the Zoning	Permit by the Zoning		
		Administrator	Administrator		
Number of Children in Care	Eight or fewer children	Nine to 14 children	Fifteen or more children		

#### Figure 4-8 Child Day-Care Chart

#### Why is this important?

In Marin County, an estimated 25,232 children under 14 live in working families (that is, families with two working parents or a working single parent) that are likely to need care for their children. The demand for child care in Marin consistently exceeds the licensed supply.

**Economy:** The combination of low wages paid to child care workers and the high cost of living in Marin County makes finding and retaining qualified child care staff a challenge. Encouraging child care facilities can help reduce employee turnover, which impacts both the quantity and quality of available child care. Added facilities also would boost the local economy by adding revenue and employment stability, and by contributing to employee satisfaction and productivity.

**Equity:** Low income families experience the brunt of the lack of child care. A weakened economy can precipitate cuts in subsidized child care for those who need it most. Increasing the availability of child care facilities will benefit all families, including lower income families.

#### How will results be achieved?

# Implementing Programs

CH-1.a Allow Child Care at Community Facilities. Amend the Development Code to allow child care as a permitted use at places of worship, schools, and other appropriate community facilities. CH-1.b Establish a Child Care Nexus. Conduct a study to quantify the impact of new nonresidential development on child care demand. CH-1.c Require Child Care Through Development Review. Based on the outcome of the child care study (Program CH-1.b), adopt an ordinance requiring on-site child care or in-lieu fees for child care for new or redeveloped public and private nonresidential or mixed-use development. CH-1.d *Expedite Application Review.* Amend the Development Code to reduce application requirements and review time for child care uses in residential zones. CH-1.e **Designate a Review Guide.** Charge a planner in the Community Development Agency with coordinating child care facility applications and shepherding them through the project review process. CH-1.f Map Appropriate Sites. Create a map that identifies appropriate locations for future child care facilities to help guide project proponents. CH-1.g Consider Fee and Permit Waivers. Research and consider adopting a whole or partial fee waiver for child care facilities and determine whether large family day-care projects that meet specified standards should be exempt from use-permit requirements. CH-1.h Facilitate On-Site Child Care. Amend the Development Code to exclude floor area ratio requirements for child care facilities in conjunction with a range of development projects. CH-1.i **Reduce Parking Requirements.** Review the Development Code to reduce parking requirements for new projects or uses that include adequate child care facilities where

appropriate.

**CH-1.j** *Provide Amnesty.* Establish an "amnesty" program for large family day-care providers who do not yet have a use permit.

# What Are the Desired Outcomes?

# GOAL CH-2

**Enhanced Child Care Options.** Expand the range of available child care options.

# **Policies**

CH-2.1	<b>Pursue Expanded Options.</b> Help coordinate efforts to broaden child care options in the county.
CH-2.2	<b>Encourage Subsidized Child Care.</b> Encourage financial support to supplement child care operations.



#### Why is this important?

In Marin County there were 26,352 children competing for 9,195 child care slots in 2000. The rate of available child care slots remained steady in Marin between 1996 and 2001, at 2.7 children per available slot. Affordable, high-quality child care is a component of our social infrastructure that is a necessity for working parents.

**Economy:** The need for child care is expected to grow as the local labor-force base expands to include more women. The need for affordable and subsidized child care is expected to grow as the local labor-force base expands in the lower-paying service and retail trade sectors. Projected job growth in the lower-paying service and retail trade sectors will increase the need for subsidized and affordable child care.

**Equity:** More than 500 children are eligible for state-subsidized child care, but have not received it and have been placed on the County's waiting list because state funding is insufficient to cover their needs. Subsidizing child care will directly provide lower income households with better access to care. Broadening the options for child care, especially through the workplace, will allow children from a wider range of socioeconomic groups to receive adequate care.

### How will results be achieved?

# Implementing Programs

- **CH-2.a** *Expand School Programs.* Work with the schools to enhance the availability of extended-day child care programs.
- CH-2.b *Encourage Child Care at a Range of Facilities.* Work with local child care advocacy groups to promote child care at businesses, shopping centers, schools, colleges, places of worship, hospitals, and other appropriate locations.



- **CH-2.c** *Train Providers and Parents.* Work with the Marin Child Care Council and other community-based organizations to offer training and support for child care providers and parents.
- CH-2.d *Offer Child Care to County Employees.* Explore and pursue the possible provision of child care at the Civic Center or other appropriate locations for County employees.
- **CH-2.e** *Explore Funding Options.* Actively seek new funding sources for child care operations and to pay for additional affordable child care placements for low income persons.
- **CH-2.f** *Review Incentive Options.* Consider financial incentives, such as tax credits, for employers that provide adequate child care.

熊於

# Figure 4-9 Relationships of Goals to Guiding Principles

This figure illustrates the relationships of each goal in this Section to the Guiding Principles.

Guiding Principles	1. Link equity, economy, and the environment locally, regionally, and globally.	2. Minimize the use of finite resources, and use all resources efficiently and effectively.	3. Reduce the use and minimize the release of hazardous materials.	4. Reduce greenhouse gas emissions that contribute to global warming.	5. Preserve our natural assets.	6. Protect our agricultural assets.	7. Provide efficient and effective transportation.	8. Supply housing affordable to the full range of our workforce and diverse community.	9. Foster businesses that create economic, environmental, and social benefits.	10. Educate and prepare our workforce and residents.	11. Cultivate ethnic, cultural, and socioeconomic diversity.	12. Support public health, public safety, and social justice.
CH-1 Additional Child Care Facilities								•	•	•		•
CH-2 Enhanced Child Care Options				•				•	•	•		•

#### How Will Success Be Measured?

# **Indicator Monitoring**

Nonbinding indicators, benchmarks, and targets<sup>1</sup> will help to measure and evaluate progress. This process will also provide a context in which to consider the need for new or revised implementation measures.

Indicator	Benchmark	Target		
Child care supply and demand by age.	See Healthy Marin Partnership dataset for 2000.	Supply increases until it is within 10% of child care demand for all ages and income categories by 2015.		

<sup>1</sup>Many factors beyond Marin County government control, including adequate funding and staff resources, may affect the estimated time frames for achieving targets and program implementation.

# **Program Implementation**

The following table summarizes responsibilities, potential funding priorities, and estimated time frames for proposed implementation programs. Program implementation within the estimated time frame.<sup>1</sup> will be dependent upon the availability of adequate funding and staff resources.

Program	Responsibility	Potential Funding	Priority	Time Frame
CH-1.a – Allow Child Care at Community Facilities.	CDA	Existing budget	Medium	Short term
CH-1.b – Establish a Child Care Nexus.	Child Care Commission, CDA	Will require additional grants or revenue. <sup>2</sup>	High	Short term
CH-1.c – Require Child Care Through Development Review.	CDA	Existing budget	Medium	Med. term
CH-1.d – Expedite Application Review.	CDA	Existing budget	High	Short term
CH-1.e – Designate a Review Guide.	CDA	Existing budget	High	Short term
CH-1.f - Map Appropriate Sites.	CDA	Will require additional grants or revenue <sup>2</sup>	Medium	Short term
CH-1.g – Consider Fee and Permit Waivers.	CDA	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Short term
CH-1.h – Facilitate On-Site Child Care.	CDA	Existing budget	Medium	Short term
CH-1.i - Reduce Parking Requirements.	CDA, DPW	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Short term
CH-1.j – Provide Amnesty.	CDA	Existing budget and may require additional grants or revenue <sup>2</sup>	Low	Long term
CH-2.a – Expand School Programs.	Child Care Commission	Existing budget	Medium	Ongoing

#### Figure 4-10 Child Care Program Implementation

<sup>1</sup>Time frames include: Immediate (0-1 years); Short term (1-4 years); Med. term (4-7 years); Long term (over 7 years); and Ongoing (existing programs already in progress whose implementation is expected to continue into the foreseeable future). <sup>2</sup>Completion of this task is dependent on acquiring additional funding. Consequently, funding availability could lengthen or shorten the time frame and ultimate implementation of this program.



Program	Responsibility	Potential Funding	Priority	Time Frame
CH-2.b – Encourage Child Care at a Range of Facilities.	Child Care Commission	Existing budget	High	Ongoing
CH-2.c – Train Providers and Parents.	Child Care Commission & MCCC	Existing budget	High	Ongoing
CH-2.d – Offer Child Care to County Employees.	H&HS, HR	Will require additional grants or revenue <sup>2</sup>	Medium	Med. term
CH-2.e – Explore Funding Options.	Child Care Commission	Existing budget	High	Ongoing
CH-2.f - Review Incentive Options.	Child Care Commission	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Med. term



# 4.6 Public Safety

# Background

Marin residents can directly influence public safety. For example, residents can participate in community policing and restorative programs in cooperation with law enforcement agencies. Community policing involves residents in preventing crime by forming neighborhood watch groups and other cooperative ventures with local law enforcement agencies.

Restorative justice programs seek to repair harm, reduce risk, and build community by helping the offender to understand the harm caused to the victim

# <u>₩ΛΙΛΥ ΜΛ ΛΛΝ</u> MARIN COUNTYWIDE PLAN

and the community. Through various forms of mediation, the offender is better able to understand the depth and nature of his/her wrongdoing and to accept responsibility to the victim and community for repairing that harm. In return, the community is responsible for helping the wrongdoer develop competency to become a law-abiding, contributing member of the community. Examples of such programs are victim-offender mediation programs, neighborhood accountability boards, and problem-solving courts (domestic violence, mental health, drug courts, and teen courts).

Traditional criminal justice agencies (law enforcement, courts, prosecution and defense, and probation) continue to play an essential role in maintaining and promoting public safety for those who commit more serious or violent crimes, and for those for whom alternative efforts have failed. Maintenance of efforts to assist those who cycle through the criminal justice system and the programs that attempt to prevent this recycling are essential to community safety as well as to individual development of life skills. In Marin, resources should be focused on persons with mental and emotional issues and substance abuse problems that lead to domestic violence, child abuse, and other related crimes directly attributable to these problems. The use of incarceration at state and local levels should be used as a last-resort option and for serious and violent criminal behavior.

Effective emergency response requires sophisticated and coordinated efforts by local and State agencies, but it also depends on awareness and prompt action by citizens. The County maintains an Emergency Operations Plan that is intended to provide adequate preparation and agency response to natural or human-caused disasters that threaten the health or property of residents and businesses. The plan describes how emergency management will be coordinated; identifies personnel responsibilities and actions necessary to protect health and safety, property, and the environment; and details procedures before, during, and after a major event. However, the plan recognizes that during the first 72 hours following a major event, community members must be self-sufficient. Effective and timely public communication and awareness are therefore primary components of both the Emergency Operations Plan and this Section of the Countywide Plan.

Policies and programs addressing fire safety, emergency medical services, and hazardous materials storage and transport are contained in the Environmental Hazards Section of the Built Environment Element.

# **Key Trends and Issues**

### What crimes are most problematic in Marin?

Despite the relatively low and dropping rates for almost all types of crime in Marin, crime prevention in the county needs to be more widespread, especially with regard to child abuse and neglect, elder abuse, and domestic violence. While domestic violence occurs across all segments of the community, concerns have been raised regarding the correlation between domestic violence and substance abuse. The incidence of alcohol- and drug-related school offenses in Marin is high (4.2 per 1,000 enrolled) compared with the state average (3.7). Drunk-driving rates are very high as well.

## What kinds of emergencies and after-effects do residents need to prepare for?

Threats to life, property, and the environment in Marin County are increasing in variety and frequency. For example, the chance of a major earthquake (6.7 on the Richter Scale) hitting the Bay Area before 2030 is estimated at approximately 70%. Local governments are increasingly required to develop plans and procedures that address disaster-related issues, including sheltering special needs populations, complying with the Americans with Disabilities Act, mitigating economic losses including tourism, and addressing the mental and emotional needs of victims and responders.

# What Are the Desired Outcomes?

# GOAL PS-I

**Safe Neighborhoods.** Ensure that county neighborhoods remain safe places to live.

# Policies

PS-1.1	<ul> <li>Encourage Community Involvement in Crime Control.</li> <li>Promote community policing and restorative justice</li> <li>programs, such as the County Adolescent and Adult Drug</li> <li>Courts; other problem-solving courts, such as domestic</li> <li>violence, mental health, and teen courts; the Victim-</li> <li>Offender Reconciliation Program (VORP); Neighborhood Accountability Boards; and</li> <li>other restorative programs. Support and encourage reporting of child and adult abuse</li> <li>and neglect.</li> </ul>
PS-1.2	<b>Improve Infrastructure to Discourage Crime.</b> Remedy any public facilities with problems that might encourage criminal activity, such as low lighting and blind spots that result from landscape features or fences.

**PS-1.3** Analyze Implications of Sea Level Rise for Neighborhood Safety. Analyze potential safety implications from sea level rise and prepare contingency plans in consultation with the Marin Disaster Council.

#### Why is this important?

The 2001 Marin Community Health survey found that 20% of young adults age 18-24 reported experiencing some type of physical violence or threat of violence within the past year.

**Economy:** Identifying and fixing infrastructure that invites crime can encourage residents to take pride in their neighborhoods and reduce vandalism, which historically has targeted natural and cultural resources as well as public and private property. Businesses can also benefit from crime-deterring infrastructure, including safety measures such as adequate street lighting.

**Equity:** Marin has had a higher percentage of physical abuse cases than the state average since 1998. Involving the community in efforts to prevent crime and dealing with its effects on victims can help

Economy

PS-I

Equity

\* 熊 🋠

break down some of the socioeconomic barriers in the community that may contribute to criminal activity. Safe neighborhoods contribute to a high quality of life for all residents.

#### How will results be achieved?

# **Implementing Programs**

PS-1.a	<i>Educate the Public about Community Enforcement Techniques.</i> Work with criminal justice agencies and community groups to support programs that offer information about community policing, restorative justice, reporting of child and adult abuse and neglect, and other crime prevention techniques. Information should be provided in multiple languages understood by most of the affected community members.
PS-1.b	<i>Involve Businesses in Neighborhood Watch.</i> Support efforts to strengthen and expand neighborhood watch programs, and ensuring that businesses participate in these programs.
PS-1.c	<i>Ensure Adequate Lighting.</i> Upgrade street lighting in urbanized areas where specified by community plans to fulfill neighborhood safety needs, provided that unnecessary light and glare are avoided.
PS-1.d	<i>Clean Up Neighborhoods.</i> Work with law enforcement agencies and community groups to promote cleanup, graffiti removal, and other neighborhood beautification efforts.
PS-1.e	<i>Review Structure Designs.</i> Involve law enforcement agencies in review of the design of new and rehabilitated buildings, including lighting and landscaping, to identify ways to increase resident safety.
PS-1.f	<i>Prepare Contingency Plans.</i> Work with the Bay Conservation and Development Commission (BCDC) and the Marin Disaster Council to analyze implications of sea level rise and increased violent storm events and flooding on neighborhood safety, and prepare contingency plans.
#### What Are the Desired Outcomes?

#### **GOAL PS-2**

**Decreased Crime.** Reduce rates for all types of crime, including child abuse and neglect, elder and dependent abuse and neglect, domestic violence, juvenile offenses, physical and financial abuse of seniors, and hate indicator crimes.

#### Policies

**PS-2.1 Counteract Domestic Violence and Juvenile Crime.** Decrease the incidence of domestic violence, including child abuse and neglect, elder and dependent adult abuse and neglect, and crimes by or against youth.



**PS-2.2** Support Services for Mentally Ill Criminal Offenders. Reduce the incidence of crimes by the mentally ill by continuing to support the Support and Treatment After Release (STAR) and mental health court programs.

#### Why is this important?

Since 1996, crime rates have steadily decreased in Marin County as well as in California. Some crimes are more predominant in Marin, however. Physical abuse cases are higher in Marin than the state average and hate crimes increased 6.3% between 1996 and 2001. Drug and alcohol treatment reduces drug use by 40% to 60%, reduces crime by 40% to 60%, and increases employment prospects by 40%.

**Economy:** For every \$1 spent on drug use prevention, communities save between \$4 to \$5 in costs for drug abuse treatment and counseling.

**Equity:** Thirty-seven percent of 11<sup>th</sup> graders in Marin report driving after drinking, higher than the California average of 23%. Seventy-nine percent of adults 18 and over in Marin currently drink alcohol, and 23% report binge drinking. In three out of four cases of domestic violence, the victim reported that alcohol or drugs had been a factor. Parental alcohol and drug abuse cases contribute to seven out of ten child abuse or neglect incidents as well as 75% of all foster care placements.

#### How will results be achieved?

#### **Implementing Programs**

- **PS-2.a** *Continue Counseling.* Work with law enforcement agencies to establish mandatory counseling for all perpetrators of domestic violence.
- **PS-2.b** *Enhance School Programs.* Work with schools to support and expand after-school recreation, youth mentoring, conflict resolution, team building, self-awareness, and community involvement programs.



- **PS-2.c** *Seek Funding for Youth Programs.* Continue and expand support through the Department of Health and Human Services, and other resources, of nonprofit organizations that administer youth mentoring programs.
- **PS-2.d** *Support Youth Outreach Efforts.* Work with community and faith-based organizations to reach out to troubled youth.
- **PS-2.e** *Provide Counseling to Troubled Youth.* Continue to support the use of mental health staff at juvenile hall and high schools to provide counseling.
- **PS-2.f** Seek Funding for and Create Community Restorative Justice Programs. Educate community leaders and seek financial support from public and private agencies to establish and maintain programs that use restorative justice concepts.
- **PS-2.g** *Continue Mental Health Training for Law Enforcement Agencies.* Continue the collaboration between County Health and Human Services staff and law enforcement personnel to provide training and education on methods for addressing mental health patients in the criminal justice system.
- **PS-2.h** *Promote Alternatives to Jail for Mental Health Offenders.* Support efforts to strengthen and expand programs that provide assistance to people with mental health problems who enter the criminal justice system, such as the mental health court and Support and Treatment After Release (STAR) programs.

#### What Are the Desired Outcomes?

## **GOAL PS-3**



**Effective Emergency and Disaster Preparedness.** Provide proper emergency and disaster preparedness services through effective and coordinated emergency management plans and procedures.

#### **Policies**

**PS-3.1** Plan Thoroughly for Emergencies. Ensure that the County, its citizens, businesses, and services are prepared for effective response and recovery in the event of emergencies or disasters.

**PS-**3.2

**Safe Public Structures.** Protect public health and safety through appropriate siting and rehabilitation of public facilities.

#### Why is this important?

The probability for a major earthquake (6.7) occurring in the Bay Area before 2030 is estimated at approximately 70%. Effective response to emergencies, including earthquakes, fires, and weather-related events, has many benefits for the Marin community.

**Environment:** Fire has become a greater hazard in Marin as fuel loads have increased due to suppression of natural fires and as residential development has occurred adjacent to wild lands. The well-being of the natural environment may depend heavily on response to major emergencies, such as hazardous materials release or wildfires.

**Economy:** The global cost of natural disasters is anticipated to top \$300 billion annually by 2025. The San Francisco earthquake of 1906 caused direct losses of about \$24 million and fire losses of about \$500 million, which is the 2004 equivalent of \$10 billion combined. Economic capacity to function depends on the ability to recover quickly and effectively from any major emergency or disaster.

**Equity:** The City-Centered Corridor, where people are most concentrated, is a high-risk area for emergencies. Most fires reported between 1990 and 1997 originated in the City-Centered Corridor. Effective emergency preparedness and relief are essential to ensure the long-term safety and health of people in our communities.

#### How will results be achieved?

#### **Implementing Programs**

PS-3.a	<i>Maintain Effective Communication Systems.</i> Work with public safety and health and human services agencies countywide to maintain interagency and public communications systems that will provide mutual aid and be reliable during and following an emergency (including completing the Marin Emergency Radio Authority project). <sup>1</sup>
PS-3.b	<i>Maintain Adequate Response Resources.</i> Identify the need for and maintain adequate staffing levels, equipment, and resources, and undertake disaster preparedness training as necessary to provide essential emergency public services.
PS-3.c	<i>Distribute Public Information.</i> Offer written materials that inform residents, businesses, and local groups about the current nature and extent of local safety hazards and emergency plans, including evacuation plans and procedures to accommodate special needs populations. Information should be provided in languages understood by most of the affected community members.
PS-3.d	<b>Conduct Disaster-Awareness Efforts.</b> Establish an annual Emergency Preparedness Awareness Week, during which seminars are offered in schools and other civic and neighborhood locations to teach citizens how to prepare for potential emergencies. Encourage residents to go through the Community Emergency Response Training program so that they can serve as civilian volunteers during an emergency.
PS-3.e	<b>Promote Community Involvement.</b> Work with neighborhood groups and other civic organizations to establish councils that will conduct a variety of disaster-preparedness functions, including emergency response training and removal of vegetation around

<sup>1</sup>The Marin Emergency Radio Authority was established as a Joint Powers Authority to unify local communications systems.

buildings in areas prone to wildland fire (also see Program EH-4.h in the Environmental Hazards Section).

- **PS-3.f** *Promote Structural and Nonstructural Safety.* Provide and inform the public of the available educational guides promoting structural and nonstructural earthquake safety. Encourage natural gas safety, water heater bracing, and installation of automatic natural gas shut-off valves in buildings. Encourage retrofit of older buildings and securing nonstructural elements of a building to prevent the falling or throwing of objects.
- **PS-3.g** Locate Emergency Services Facilities Appropriately. Locate and design emergency buildings and vital utilities, communication systems, and other public facilities so that they can remain operational during and after an emergency or disaster. Encourage that these structures and facilities are designed to be earthquake proof to ensure continuous operation even during extreme seismic ground shaking.
- **PS-3.h** *Promote Agency Emergency Planning.* Encourage jurisdictions and institutions to create and adopt emergency response plans.
- **PS-3.i** *Site Public Structures Safely.* Locate facilities necessary for the protection of public safety and/or the provision of emergency services away from areas subject to inundations, subsidence, or slope or ground failure in a seismic event as feasible. Prohibit placement of critical public facilities such as schools, hospitals, streets, communication systems, utility and public safety structures, and reservoirs in designated fault zones.
- **PS-3.j** *Develop Evacuation Plans.* Work with agencies that provide emergency preparedness, response, and recovery services to formulate definitive plans and procedures for evacuation of hazard-prone areas.
- PS-3.k *Ensure Seismic Design Regulations.* Continue to implement County ordinances to ensure that new construction meets California Building Code seismic design requirements, provides seismic shut-off devices, and anchors liquid petroleum gas tanks. Require geological assessment (for example, Soils Investigation and Geologic/Geotechnical reports) for grading permits to determine the effects of seismic ground shaking on proposed grading and associated new construction.

#### What Are the Desired Outcomes?

#### **GOAL PS-4**

**Decreased Exposure to Hazardous Materials.** Reduce the risks to human and environmental health from hazardous materials.

#### Policy

**PS-4.1** 

**Regulate and Reduce Hazardous Material Use.** Control the use and storage of hazardous materials to minimize their presence in, and potential dangers to, the community and environment.



#### Why is this important?

Materials that pose dangers to human health and the environment are used for many purposes. Storage, transport, and disposal of hazardous materials increase exposure to these materials. In 2005, 35,000 incidents were reported in the United States where hazardous materials in storage or transit were released into the environment.

**Environment:** Annually, more than 240 million gallons, or nearly 60% of the used motor oil generated by "do it yourself" oil changes in the United States, are disposed of in ways that damage our environment. One quart of motor oil released into a storm drain can contaminate over 250,000 gallons of water. Toxic spills and the release of hazardous materials cause devastating impacts on the natural systems that support life.

**Economy:** More than 500 Marin County businesses are regulated users of hazardous materials. The economic risk of using such materials is reduced by safely storing, transporting, and disposing of hazardous materials, and by developing and using alternative, less harmful products and processes.

**Equity:** The City-Centered Corridor has the greatest concentration of people and businesses in Marin. Schools, hospitals, high occupancy buildings, and nursing homes are especially vulnerable to hazardous material exposure. Lower income residents are often disproportionately impacted by exposure to hazardous materials.

#### How will results be achieved?

#### Implementing Programs

**PS-4.a** *Regulate Development Near Waste Sites.* Adopt specific regulations for development of land on or adjacent to a known solid or hazardous waste site.

**PS-4.b** *Regulate Hazardous Material Use.* Identify businesses that use, store, dispose of, or transport hazardous materials, and require them to follow measures that protect public health and safety.

- **PS-4.c** *Restrict Transport.* Work with federal and State agencies to require all transport of hazardous materials to follow approved routes.
- **PS-4.d** *Prepare for Hazardous Materials Incidents.* Plan for response to an emergency involving a major release of hazardous materials (see Policy PS-3.1; also see the Environmental Justice Section of the Socioeconomic Element).
- **PS-4.e** *Precautionary Principle.* Continue to implement the precautionary principle in County purchases and actions, which calls for a careful analysis and selection of the available alternatives presenting the least potential threat to human health and natural systems.
- **PS-4.f** *Reduce Hazardous Materials on County Property.* Develop and implement a policy to reduce the use of hazardous materials in County buildings, on County property, and in County operations.
- **PS-4.g** *Promote Ecologically Friendly Products.* Continue to evaluate and provide incentives for use of ecologically friendly products. This includes County procurement policies that give price preferences to recycled or post-consumer products, use of integrated pest management products that are nontoxic, and the promotion of green businesses that incorporate ecologically friendly products into their business operation.
- **PS-4.h** *Hazardous Materials Education.* Continue to educate the public about the safe use, transport, and disposal of hazardous materials and encourage (for example, through incentive programs) the use of less-toxic substances in residential and County operations.
- **PS-4.i** *Hazardous Materials Disposal.* Promote, educate, and encourage the public and businesses to properly dispose of any hazardous materials or waste at Marin County's permanent household hazardous waste collection facility.

熊於

#### Figure 4-11 Relationships of Goals to Guiding Principles

This figure illustrates the relationships of each goal in this Section to the Guiding Principles.

Guiding Principles	1. Link equity, economy, and the environment locally, regionally, and globally.	2. Minimize the use of finite resources, and use all resources efficiently and effectively.	3. Reduce the use and minimize the release of hazardous materials.	4. Reduce greenhouse gas emissions that contribute to global warming.	5. Preserve our natural assets.	6. Protect our agricultural assets.	7. Provide efficient and effective transportation.	8. Supply housing affordable to the full range of our workforce and diverse community.	9. Foster businesses that create economic, environmental, and social benefits.	10. Educate and prepare our workforce and residents.	11. Cultivate ethnic, cultural, and socioeconomic diversity.	12. Support public health, public safety, and social justice.
PS-1 Safe Neighborhoods	•				•		•	•	•	•	•	•
PS-2 Decreased Crime	•				•					•	•	•
PS-3 Effective Emergency and Disaster Preparedness	•		•	•	•		•			•	•	•
PS-4 Decreased Exposure to Hazardous Materials	•		•						•	•	•	•

#### How Will Success Be Measured?

#### **Indicator Monitoring**

Nonbinding indicators, benchmarks, and targets<sup>1</sup> will help to measure and evaluate progress. This process will also provide a context in which to consider the need for new or revised implementation measures.

Indicators	Benchmarks	Targets
Survey of public perception of safety in unincorporated areas.	89% in 2000 and 88% in 2004 reported feeling safe in their community.	No decrease through 2020.
Recidivism rates in unincorporated county for child endangerment bookings.	61% recidivism rate.	Decrease recidivism rate through 2020.
Recidivism rates for violent crime (number of perpetrators of violent crime, excluding simple assaults, who have been previously booked in Marin county for any offense).	68% recidivism rate.	No increase through 2020.
Juvenile crime rate.	58.9 in 2000.	No increase through 2020.

<sup>1</sup>Many factors beyond Marin County government control, including adequate funding and staff resources, may affect the estimated time frames for achieving targets and program implementation.

#### **Program Implementation**

The following table summarizes responsibilities, potential funding priorities, and estimated time frames for proposed implementation programs. Program implementation within the estimated time frame <sup>1</sup> will be dependent upon the availability of adequate funding and staff resources.

Program	Responsibility	Potential Funding	Priority	Time Frame		
PS-1.a – Educate the Public About Community Enforcement Techniques.	Sheriff	Existing budget	Medium	Ongoing		
PS-1.b – Involve Businesses in Neighborhood Watch.	Sheriff	Existing budget	Medium	Ongoing		
PS-1.c – Ensure Adequate Lighting.	DPW, Community Service Districts (CSDs)	Will require additional grants or other revenue. <sup>2</sup>	Medium	Long term		
PS-1.d - Clean Up Neighborhoods.	Sheriff	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing		
PS-1.e – Review Structure Designs.	CDA, Sheriff	Existing budget, grants	High	Ongoing		
PS-1.f – Prepare Contingency Plans.	Marin Disaster Council, OES, BCDC, and CDA	Existing budget, grants	High	Med. term		
PS-2.a – Continue Counseling.	District Attorney's Office	Existing budget, grants	High	Ongoing		
PS-2.b – Enhance School Programs.	Sheriff, H&HS	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium- high	Ongoing		
PS-2.c – Seek Funding for Youth Programs.	Sheriff, H&HS, Probation	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium- high	Ongoing		

#### Figure 4-12 Public Safety Program Implementation

<sup>&</sup>lt;sup>1</sup>Time frames include: Immediate (0-1 years); Short term (1-4 years); Med. term (4-7 years); Long term (over 7 years); and Ongoing (existing programs already in progress whose implementation is expected to continue into the foreseeable future). <sup>2</sup>Completion of this task is dependent on acquiring additional funding. Consequently, funding availability could lengthen or shorten the time frame and ultimate implementation of this program.



Program	Responsibility	Potential Funding	Priority	Time Frame
PS-2.d – Support Youth Outreach Efforts.	Sheriff, H&HS, Probation	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium- high	Ongoing
PS-2.e – Provide Counseling to Troubled Youth.	Sheriff, H&HS, Probation	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
PS-2.f – Seek Funding for and Create Community Restorative Justice Programs	Probation, Public Defender	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Long term
PS-2.g – Continue Mental Health Training for Law Enforcement Agencies.	H&HS	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
PS-2.h – Promote Alternatives to Jail for Mental Health Offenders.	H&HS, Sheriff	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium- high	Ongoing
PS-3.a – Maintain Effective Communication Systems.	DPW, MERA	Existing budget, MERA bond	High	Immediate
PS-3.b - Maintain Adequate Response Resources.	All Departments	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Ongoing
PS-3.c – Distribute Public Information.	OES	Existing budget and may require additional grants or revenue <sup>2</sup>	Low	Ongoing
PS-3.d - Conduct Disaster- Awareness Efforts.	OES	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Ongoing
PS-3.e – Promote Community Involvement.	OES	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
PS-3.f – Promote Structural and Nonstructural Safety.	CDA	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Ongoing
PS-3.g – Locate Emergency Services Facilities Appropriately.	CAO	Capital Projects Fund	High	Immediate

# 

## SOCIOECONOMIC ELEMENT

Program	Responsibility	Potential Funding	Priority	Time Frame
PS-3.h – Promote Agency Emergency Planning.	OES	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
PS-3.i – Site Public Structures Safely.	DPW, BOS	Will require additional grants or other revenue <sup>2</sup>	High	Long term
PS-3.j – Develop Evacuation Plans.	OES, Local jurisdictions	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Immediate
PS-3.k – Ensure Seismic Design Regulations.	CDA, DPW	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Short
PS-4.a – Regulate Development Near Waste Sites.	CDA	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Med. term
PS-4.b - Regulate Hazardous Material Use.	DPW	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
PS-4.c - Restrict Transport.	Federal Department of Transportation, California Highway Patrol, CalTrans	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
PS-4.d – Prepare for Hazardous Materials Incidents.	HazMat, JPA, DPW, OES	Existing budget, JPA contributions and may require additional grants or revenue <sup>2</sup>	High	Ongoing
PS-4.e – Precautionary Principle.	All County Departments	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
PS-4.f – Reduce Hazardous Materials on County Property.	DPW	Will require additional grants or other revenue <sup>2</sup>	Medium	Long term
PS-4.g – Promote Ecologically Friendly Products.	BOS, CDA, DPW, other applicable departments	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
PS-4.h – Hazardous Materials Education.	State, county and local jurisdictions	Will require additional grants or other revenue <sup>2</sup>	Medium	Med. term



Program	Responsibility	Potential Funding	Priority	Time Frame
PS-4.i - Hazardous Materials Disposal.	JPA, Novato Sanitary District	Will require additional grants or other revenue <sup>2</sup>	Medium	Med. term

# <image>

#### 4.7 Community Participation

#### Background

Community participation in public decision making is essential to good governance. Broad and informed participation creates healthy and just local government and community atmosphere. State and federal environmental justice laws and policies increasingly emphasize the need for full public participation in the decision-making process (see Section 4.10).

Ensuring public access to clear, timely, and appropriate information is an important first step in encouraging residents to participate in decisions that affect them. Meetings must be accessible to all interested persons, and the composition



"Everybody can be great. Because anybody can serve. You don't have to have a college degree to serve. You don't have to make your subject and your verb agree to serve. You don't have to know the second theory of thermodynamics to serve. You only need a heart full of grace. A soul generated by love."

- Martin Luther King

of each local decision-making body should be diverse enough to truly represent the community it serves.

#### **Key Trends and Issues**

#### Who is underrepresented in community dialogue?

Public forums for decision making have not typically offered information in languages other than English, nor have they always been held in places most convenient for people interested in a particular issue. Ethnic minorities, including recent immigrants, have not been represented on local advisory committees or decision-making bodies in proportion to their percentage of the overall population. Voices of frail older adults living alone or in institutions such as skilled-nursing facilities are often not heard at these forums, particularly seniors who are no longer able to drive.

#### What Are the Desired Outcomes?

#### **GOAL CP-I**



**Broad and Diverse Participation.** Incorporate all segments of the community in County planning efforts and local decision making.

#### Policies

**CP-1.1 Provide Access for All.** Take all feasible steps to ensure that everyone interested in participating in community forums has the materials necessary to contribute to informed decisions, as well as convenient access to meeting venues.

**CP-1.2 Encourage Full Representation on Committees and Commissions.** Encourage broad and diverse membership on County committees and commissions, and seek to include the underrepresented.

#### Why is this important?

While 72% of Marin residents reported voting in a local or statewide election during the past year, well above the national and state averages, older (91%) and longer-term residents (82%) are much more likely to vote than those ages 18 to 35 (52%), and newer residents (58%). Because community needs are best defined by the people who live and work there, increased participation is needed from sectors that are not as well represented.

**Environment:** In Marin, plans and policies developed without widespread community involvement are more likely to result in environmental injustices and solutions that are not holistic.

**Economy:** Encouraging community stakeholder collaboration early in the process can facilitate decision making and lead to significant cost savings resulting from timely resolution of planning issues.

**Equity:** County actions will be more equitable and more likely to succeed if all segments of the community are included in the decision-making process. Widespread access to meeting venues and information in understandable formats can increase the community's stake in and support for local decisions.

#### How will results be achieved?

#### Implementing Programs

CP-1.a	<b>Inform and Engage the Community.</b> Employ a vapromote two-way communication with a broad ar about County decision-making meetings, including radio, and television.	riety of techniques to inform and ad diverse range of the community ag print, information technology, post,
CP-1.b	<i>Ensure Convenience of Meetings.</i> Hold community meetings at locations and times convenient for community members desiring to provide input.	You make a living by what
CP-1.c	<i>Offer Translation Services.</i> Provide translation services upon request at Board of Supervisors and Planning Commission meetings, as well as at other appropriate government- sponsored community forums.	you get, but you make a life by what you give." – Winston Churchill
CP-1.d	<i>Distribute Multilingual Materials.</i> Provide meetin multiple languages, and publicize meetings in nor	g notices and meeting materials in 1-English media.
CP-1.e	<i>Consider Providing Child Care at Meetings.</i> Evaluate care upon request for members of the public wish and Planning Commission meetings, as well as at sponsored community forums.	uate the feasibility of providing child ning to attend Board of Supervisors other appropriate government-
CP-1.f	<i>Encourage Attendance by All Community Memb</i> youth, seniors, and persons with disabilities (inclu auditorily impaired individuals) at Board of Supe other County-sponsored meetings.	<b>bers.</b> Accommodate participation by Iding information for visually and rvisors, Planning Commission, and

# 券

"At all levels and in all realms, people must have a say in the decisions that affect their lives."

 The Environmental Justice and Climate Change Initiative **CP-1.g** *Represent All Constituencies.* Reach out and recruit a wide range of community members for County committees and commissions, and work to achieve broad and diverse participation in rough proportion to the ethnic and gender makeup of Marin County.

**CP-1.h** *Encourage Turnover.* Modify County rules to encourage turnover of committee and commission members to increase representation of the communities they serve.

\* 熊 🏌

Ì

#### Figure 4-13 Relationships of Goals to Guiding Principles

This figure illustrates the relationships of this Section's goal to the Guiding Principles.

Goal Goal	1. Link equity, economy, and the environment locally, regionally, and globally.	2. Minimize the use of finite resources, and use all resources efficiently and effectively.	3. Reduce the use and minimize the release of hazardous materials.	4. Reduce greenhouse gas emissions that contribute to global warming.	5. Preserve our natural assets.	6. Protect our agricultural assets.	7. Provide efficient and effective transportation.	8. Supply housing affordable to the full range of our workforce and diverse community.	9. Foster businesses that create economic, environmental, and social benefits.	10. Educate and prepare our workforce and residents.	11. Cultivate ethnic, cultural, and socioeconomic diversity.	12. Support public health, public safety, and social justice.
CP-1 Broad and Diverse Participation	•									•	•	•

#### How Will Success Be Measured?

#### **Indicator Monitoring**

Nonbinding indicators, benchmarks, and targets <sup>1</sup> will help to measure and evaluate progress. This process will also provide a context in which to consider the need for new or revised implementation measures.

Indicator	Benchmark	Target
Voter turnout in general elections.	84.6% in 2000.	No decrease through 2020.

<sup>1</sup>Many factors beyond Marin County government control, including adequate funding and staff resources, may affect the estimated time frames for achieving targets and program implementation.

#### **Program Implementation**

The following table summarizes responsibilities, potential funding priorities, and estimated time frames for proposed implementation programs. Program implementation within the estimated time frame.<sup>1</sup> will be dependent upon the availability of adequate funding and staff resources.

Program	Responsibility	Potential Funding	Priority	Time Frame
CP-1.a – Inform and Engage the Community.	All Applicable Departments	Existing budget and may require additional grants or revenue. <sup>2</sup>	High	Ongoing
CP-1.b – Ensure Convenience of Meetings.	All Applicable Departments	Existing budget	High	Ongoing
CP-1.c – Offer Translation Services.	All Applicable Departments	Will require additional grants or revenue <sup>2</sup>	High	Ongoing
CP-1.d – Distribute Multilingual Materials.	All Applicable Departments	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
CP-1.e – Consider Providing Child Care at Meetings.	All Applicable Departments	Will require additional grants or revenue <sup>2</sup>	Medium	Ongoing
CP-1.f – Encourage Attendance by All Community Members.	All Applicable Departments	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
CP-1.g – Represent All Constituencies.	All Applicable Departments	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
CP-1.h - Encourage Turnover.	All Applicable Departments	Existing budget	Medium	Ongoing

Figure 4-14 Community Participation Program Implementation

<sup>&</sup>lt;sup>1</sup>Time frames include: Immediate (0-1 years); Short term (1-4 years); Med. term (4-7 years); Long term (over 7 years); and Ongoing (existing programs already in progress whose implementation is expected to continue into the foreseeable future). <sup>2</sup>Completion of this task is dependent on acquiring additional funding. Consequently, funding availability could lengthen or shorten the time frame and ultimate implementation of this program.

# 

MARIN COUNTYWIDE PLAN

# 林谷桃

#### SOCIOECONOMIC ELEMENT



## 4.8 Diversity

#### Background

Encouraging and celebrating diversity and cultivating a rich community fabric require supporting populations that have traditionally been underrepresented (Figure 4-15). Increased ethnic and cultural awareness, and understanding by the staffs of public agencies and nonprofit organizations, can do much to promote self-sufficiency and allow for leadership among minority groups.

Cultural and economic vitality needs to be supported in neighborhoods with high proportions of minority residents. Specific needs of the recent-immigrant

community, such as English as a Second Language classes and documentation assistance, should be met. This Section of the Countywide Plan seeks to ensure that people of all cultures, ethnicities, ages, genders, and sexual orientations feel welcome and integrated in Marin.



Figure 4-15 Racial Minority Populations in Marin

#### **Key Trends and Issues**

#### Is Marin really an integrated community?

Marin is growing in diversity, but community integration is not keeping pace. For example, parts of Novato, San Rafael, and Marin City house diverse communities, including a large proportion of the county's Latino and African-American population, while many other communities are not nearly as heterogeneous. Some residents have expressed the belief that racial, ethnic, and cultural diversity are not adequately supported in Marin.

#### What Are the Desired Outcomes?

#### **GOAL DIV-I**



Celebrating Diversity. Support and enhance ethnic and cultural diversity.

#### Policies

**DIV-1.1 Publicize Diversity.** Celebrate and call attention to diversity through community dialogue, events, festivals, and public art installations.

- **DIV-1.2** Enhance Multilingual/Multicultural Capacity. Provide County information in languages other than English, and offer services that are responsive to persons of diverse cultural and ethnic backgrounds.
- **DIV-1.3 Promote Diversity in the Workforce.** Provide paid and volunteer job opportunities for people of diverse backgrounds, including economically, physically, and socially disadvantaged people.
- **DIV-1.4 Promote Leadership in Minority Communities.** Encourage individuals from underrepresented minority groups to represent their communities, including by joining County staff, commissions, and boards.

#### Why is this important?

In 1990, 89% of the population was white, and 11% was African-American, Asian, Pacific Islander, or of other races. People of Latino origin (who may be of any racial group) composed 8% of the population. In 2000, the nonwhite population increased to 16% and the Latino population to 11%, while the white population decreased to 84%. Hate crimes decreased between 1998 and 2000, but were concentrated geographically in Novato (59 offenses), San Rafael (13), and the unincorporated areas of Marin (11).

**Economy:** Diversity in the workplace can extend economic benefits to the community by fostering a broad range of ideas and styles that resonate throughout the community.

**Equity:** Eighteen percent of housing complaints logged in Marin between 1999 and 2000 were based on race or ethnicity. A "Race Audit" conducted in 2000 concluded that an African-American person could encounter discrimination or difficulties when finding housing 47% of the time. Recognition and celebration of diversity as well as increased participation in decision making can increase equal opportunities in housing, education, and employment while improving quality of life.

#### How will results be achieved?

#### Implementing Programs

- DIV-1.a Identify Spaces to Celebrate Diversity. Locate public and private spaces, such as civic plazas and shopping malls, that may be used for cultural awareness activities, including festivals, art exhibitions, and poetry presentations.
  DIV-1.b Encourage Diversity-Oriented Organizations. Support organizations that create art, organize events, and foster community dialogue about or promoting diversity.
  DIV-1.c Promote Diversity at Community Events. Support community festivals that promote
- **DIV-1.c** *Promote Diversity at Community Events.* Support community festivals that promote diversity, such as international day at the Marin County Fair.



- DIV-1.d *Encourage Advocacy Efforts by Local Groups.* Support the work of the Marin Human Rights Commission and other local diversity advocacy groups that focus on cultural diversity issues.
- **DIV-1.e** *Promote Understanding in Schools.* Work with schools, colleges, and community groups and organizations, such as the Southern Poverty Law Center, to promote education about, and understanding of, various cultures and ethnic diversity.

"We must learn to live together as brothers, or perish together as fools."

– Peter Schwartz

**DIV-1.f** *Publicize Diversity on the Internet.* Create a bulletin board on the County website where the community can advertise and discuss multicultural events.

**DIV-1.g** *Value Multilingual Skills.* Evaluate wages for multilingual County employees to ensure that they receive appropriate compensation.

**DIV-1.h** *Train Employees.* Expand efforts to make County services more culturally appropriate through the Cultural Competence Committee, including by expanding educational programs and training for County employees and officials.

**DIV-1.i** *Assist Recent Immigrants.* Provide access to programs, information, meeting space, and volunteer opportunities needed by recent immigrants, including English language instruction and work opportunities.

**DIV-1.j** *Ensure Equal Employment.* Support the efforts of the Marin County Affirmative Action Committee to achieve the goal of equal employment opportunity.



- **DIV-1.k** *Practice Fair Hiring.* Adhere to the Equal Employment Opportunity Policy/Affirmative Action Policy developed and adopted by the County.
- **DIV-1.1** *Require Nondiscrimination by Contractors.* Ensure that agencies contracting with the County implement nondiscrimination policies and practices, and comply with the living wage ordinance.
- **DIV-1.m** *Provide Volunteer Opportunities.* Offer volunteer positions to diverse populations through special programs in collaboration with community groups and the County Office of Education.

**熊 茶** 

j, j

#### Figure 4-16 Relationships of Goal to Guiding Principles

This figure illustrates the relationships of this Section's goal to the Guiding Principles.

Goal Goal	1. Link equity, economy, and the environment locally, regionally, and globally.	2. Minimize the use of finite resources, and use all resources efficiently and effectively.	3. Reduce the use and minimize the release of hazardous materials.	4. Reduce greenhouse gas emissions that contribute to global warming.	5. Preserve our natural assets.	6. Protect our agricultural assets.	7. Provide efficient and effective transportation.	8. Supply housing affordable to the full range of our workforce and diverse community.	9. Foster businesses that create economic, environmental, and social benefits.	10. Educate and prepare our workforce and residents.	11. Cultivate ethnic, cultural, and socioeconomic diversity.	12. Support public health, public safety, and social justice.
DIV-1 Celebrating Diversity								•	٠	٠	٠	•



#### How Will Success Be Measured?

#### **Indicator Monitoring**

Nonbinding indicators, benchmarks, and targets<sup>1</sup> will help to measure and evaluate progress. This process will also provide a context in which to consider the need for new or revised implementation measures.

Indicator	Benchmark	Target
Diversity (age, gender, race, economic level) of County employees.	See Marin County Human Resources Department data.	Representation of employees from diverse backgrounds equals workforce availability as identified in the Equal Employment Opportunity Plan dated 2005- 2010.

<sup>1</sup>Many factors beyond Marin County government control, including adequate funding and staff resources, may affect the estimated time frames for achieving targets and program implementation.

# 

#### **Program Implementation**

The following table summarizes responsibilities, potential funding priorities, and estimated time frames for proposed implementation programs. Program implementation within the estimated time frame.<sup>1</sup> will be dependent upon the availability of adequate funding and staff resources.

Program	Responsibility	Potential Funding	Priority	Time Frame
DIV-1.a – Identify Spaces to Celebrate Diversity.	Marin Center	Existing budget, Private Donations, and may require additional grants or revenue. <sup>2</sup>	High	Ongoing
DIV-1.b – Encourage Diversity Oriented Organizations.	Marin Center	Existing budget, Private Donations, and may require additional grants or revenue <sup>2</sup>	High	Ongoing
DIV-1.c – Promote Diversity at Community Events.	Marin Center	Existing budget Donations	High	Ongoing
DIV-1.d – Encourage Advocacy Efforts by Local Groups.	BOS, HR	Existing budget	High	Ongoing
DIV-1.e – Promote Understanding in Schools.	County Office of Education, School District and CBO's	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
DIV-1.f – Publicize Diversity on the Internet.	Marin Center and CBO's	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
DIV-1.g – Value Multilingual Skills.	HR	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Ongoing
DIV-1.h - Train Employees.	HR	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing

#### Figure 4-17 Diversity Program Implementation

<sup>1</sup>Time frames include: Immediate (0-1 years); Short term (1-4 years); Med. term (4-7 years); Long term (over 7 years); and Ongoing (existing programs already in progress whose implementation is expected to continue into the foreseeable future). <sup>2</sup>Completion of this task is dependent on acquiring additional funding. Consequently, funding availability could lengthen or shorten the time frame and ultimate implementation of this program.

# 

## SOCIOECONOMIC ELEMENT

Program	Responsibility	Potential Funding	Priority	Time Frame
DIV-1.i – Assist Recent Immigrants.	Civic Center Volunteers	Existing budget	Medium	Ongoing
DIV-1.j – Ensure Equal Employment.	Human Resources, Equal Employment Division	Existing budget	High	Ongoing
DIV-1.k – Practice Fair Hiring.	Human Resources	Existing budget	High	Ongoing
DIV-1.1 – Require Nondiscrimination by Contractors.	County Administrator's Office	Existing budget	High	Ongoing
DIV-1.m – Provide Volunteer Opportunities.	Civic Center Volunteers	Existing budget	Medium	Ongoing

# 秋 \* # \*\*

MARIN COUNTYWIDE PLAN



#### 4.9 Education

#### Background

Because educational institutions and programs are generally very successful in Marin, the Countywide Plan directs attention to policies and programs dealing with identified scholastic weaknesses in the county. In particular, this includes addressing educational inequities based on income, geography, and race. The need exists to expand after-school, adult, and other community-based educational programs, and to ensure that all students have access to nutritious meals at school.

The responsibility to provide quality education involves a range of entities. Developers of new subdivisions have an obligation to dedicate land or pay fees for school purposes; school districts must



"If you plan for a year, plant rice. If you plan for 10 years, plant trees. If you plan for 100 years, educate your children."

- Chinese proverb

provide sufficient classroom space and teaching personnel to meet State-mandated requirements; and the County, parents, and students can help shape school curriculum by participating in decision making at the school-district and State levels.

#### **Key Trends and Issues**

# Which students are not receiving the full benefit of the education system?

The high-quality education for which Marin is known is generally less accessible to students in rural areas, from lower income households, or in ethnic minority groups. Early education and extracurricular programs, counseling and other support services, and free or reduced-cost

school meals need to be made more widely available. Educational opportunities for recent immigrants in particular need improvement.

#### What Are the Desired Outcomes?

#### **GOAL EDU-I**

Adequate School Facilities. Ensure that adequate school<sup>1</sup> facilities are available to meet the needs of current and future Marin County residents.



#### Policy

**EDU-1.1** Assist with School Planning. Coordinate with the school districts to determine appropriate locations and layouts for future facilities.

#### Why is this important?

Adequate school facilities are essential for providing a quality education for Marin students. Schools that are adequate and well designed promote a better learning environment.

**Environment:** While 70% of parents walked or biked to school, only 17% of their children do so today. Well-planned and properly located school facilities will ensure that education is readily accessible where students live, thus reducing gas emissions produced by school commutes.

**Economy:** College graduates are statistically more likely to have healthy children and to be involved with their children's education both inside and outside of school. More than 90% of preschoolers

<sup>1</sup>School refers to K-12 programs and institutions of higher education.

whose mothers are college graduates are read to at least three times a week, compared with 76% of their counterparts whose mothers are high school graduates. A well-educated and healthy population will contribute to better-informed decision making, and will have productive and meaningful employment in the future.

**Equity:** In the San Rafael City Elementary and Sausalito Elementary schools, at least half of the students received free or reduced-cost meals, 50.0% and 56.1%, respectively, during the 2000–2001 school year. Providing sufficient classroom space, teaching equipment, and nutritious meals in enough locations to serve students equitably will help eliminate social inequities in schools and beyond, as students graduate, seek further education, and enter the workplace.

#### How will results be achieved?

#### **Implementing Programs**

- **EDU-1.a** *Share Data.* Provide demographic data that schools and colleges can use in projecting facility needs.
- **EDU-1.b** *Preserve Future Facilities Options.* Encourage school districts and colleges to lease facilities not currently needed for teaching to reserve those sites for future school needs; interim uses might include child care centers, recreation centers, community meeting places, private schools, offices, and art studios.
- **EDU-1.c** *Consider Less-Populated Areas.* Work with school districts and colleges to ensure that quality and conveniently accessible education is available in all geographic areas of the county.
- **EDU-1.d** *Monitor School Meal Programs.* Work with the schools to ensure that all students have affordable access to healthy meals, proper nutritional guidance, and alternatives to unhealthy and less nutritious food.

#### What Are the Desired Outcomes?

## **GOAL EDU-2**

**Ample Educational Opportunities.** Ensure that all students in Marin have the best educational opportunities available.

#### **Policies**

EDU-2.1 Supplement Classroom Education. Enhance preschool, school, and after-school educational programs.
 EDU-2.2 Expand Adult Education. Promote and enhance adult education, particularly for non-English speakers and for lower income individuals.



- **EDU-2.3** Strengthen Library Programs. Support the ability of local libraries to provide expanded services to all persons.
- **EDU-2.4 Promote Climate Change Education.** Assist in building understanding of sustainability and climate change issues in schools.

#### Why is this important?

"Knowledge is the only kind of wealth that multiplies as you give it away."

ki

- Peter Schwartz

Traditional classroom education alone cannot provide all of the educational skills and experience that students of all ages need and desire. Children who consistently participate in extracurricular activities are 70% more likely to attend college, 66% more likely to vote as young adults, and 50% more likely to volunteer.

**Environment:** Expanded educational opportunities may include training in environmental awareness addressing conservation and sustainable work and living practices, and other subjects that result in support for the environment.

**Economy:** Participation in extracurricular programs results in a broader range of knowledge and skills in students, which helps them to contribute more fully to the community workforce. A national study of 500 6th through 12th graders found that students who spent time in extracurricular activities were six times less likely to drop out of school, two times less likely to be arrested, and about 75% less likely to become involved in drug use. A young person who becomes a lifetime criminal costs society between \$1.4 million and \$1.7 million.

**Equity:** The Marin County Free Library system is expecting that visits to the library system will increase as the demographics of the county evolve. In particular, the number of visits from migrant and elderly populations is expected to grow. Additional educational programs can bring people of diverse backgrounds together in settings where mutual recognition and coordinated group efforts are often keys to individual success.

#### How will results be achieved?

#### **Implementing Programs**

EDU-2.a	<b>Promote After-School Programs.</b> Work with school districts, community centers, and libraries to provide interesting and well-attended after-school educational activities.
EDU-2.b	<i>Enhance Literacy and English-Language Programs.</i> Continue and expand the Marin Literacy program, and increase the number and capacity of English as a Second Language classes in the county.
EDU-2.c	<i>Offer Tutoring.</i> Work with school districts to provide after-school tutoring opportunities, especially for children of lower income households.

- EDU-2.d *Include Sustainability and Diversity in School Curricula.* Encourage expansion of school and college curricula to include education about sustainability and diversity.
- **EDU-2.e** *Support Access to Technology.* Promote placement and use of computers and other advanced educational tools in classrooms, libraries, and after-school settings, including for students who could not otherwise afford or gain access to such equipment.
- EDU-2.f *Provide Summer Programs.* Support the expansion of summer camp opportunities and summer library programs, especially for children of lower income households.
- EDU-2.g *Include Youth Input.* Work with the Marin County Youth Commission to review and make recommendations on scholastic policies that affect youth.
- EDU-2.h *Encourage Financial Training.* Support the provision of free financial management education, especially for families with limited incomes.
- EDU-2.i *Encourage Parent Education.* Support education for parents of youth through the Marin County Parent University.

"Education is for improving the lives of others and for leaving your community and world better than you found it."

- Marian Wright Edelman, 2000

- **EDU-2.j** *Promote Volunteerism.* Engage older adults and youth as volunteers through Civic Center Volunteers in partnership with nonprofit organizations.
- **EDU-2.k** *Prioritize Educational Opportunities for the Homeless.* Pursue scholarships and related programs to support homeless people in attending school, college, and training programs.
- **EDU-2.1** *Support Library Services.* Promote, market, preserve, and expand library services and programs throughout Marin County in partnership with library support groups.
- **EDU-2.m** *Improve Library Facilities.* Support the modernization and expansion of public library facilities as needed.
- **EDU-2.n** *Enlist Volunteers to Supplement Programs.* Expand existing library programs, including after-school tutoring and/or homework help, by involving teen and adult volunteers from the Civic Center Volunteers Program.



- **EDU-2.0** *Reach Out to Persons with Limited Access.* Provide appropriate library services, including homebound delivery, that make materials accessible to disabled individuals and seniors.
- **EDU-2.p** *Encourage Climate Change Curricula.* Encourage nonprofits and school districts to develop curricula for increased understanding of sustainability and climate change issues by students.
\* 熊 徐

i i

#### Figure 4-18 Relationships of Goals to Guiding Principles

This figure illustrates the relationships of each goal in this Section to the Guiding Principles.

Goals	<b>Guiding Principles</b>	1. Link equity, economy, and the environment locally, regionally, and globally.	2. Minimize the use of finite resources, and use all resources efficiently and effectively.	3. Reduce the use and minimize the release of hazardous materials.	4. Reduce greenhouse gas emissions that contribute to global warming.	5. Preserve our natural assets.	6. Protect our agricultural assets.	7. Provide efficient and effective transportation.	8. Supply housing affordable to the full range of our workforce and diverse community.	9. Foster businesses that create economic, environmental, and social benefits.	10. Educate and prepare our workforce and residents.	11. Cultivate ethnic, cultural, and socioeconomic diversity.	12. Support public health, public safety, and social justice.
EDU-1 Adequate School Facilities		•			•			•			•	•	•
EDU-2 Ample Educational Opportunities		•									•	•	•

#### How Will Success Be Measured?

#### **Indicator Monitoring**

Nonbinding indicators, benchmarks, and targets <sup>1</sup> will help to measure and evaluate progress. This process will also provide a context in which to consider the need for new or revised implementation measures.

Indicators	Benchmarks	Targets
Level of educational attainment	See Census dataset. <sup>2</sup>	Level of educational attainment
by ethnicity.		by ethnicity does not vary more
		than 30% by 2015, and 25% by
		2020.
High school dropout rate by	3% or less variation by ethnicity	Dropout rate does not vary more
ethnicity.	in 2000.	than 2% by 2015.
Number of items circulated	8.70 items in circulation per	11 items per capita in 2010 and
annually by the 11 County-	capita in 2000.	11.5 items per capita in 2015.
operated libraries per capita.		
Number of sessions on public-	177,578 sessions in 2003.	340,000 or more sessions in
use computers at the 11 County-		2015, and 360,000 or more
operated libraries.		sessions in 2020.

<sup>1</sup>Many factors beyond Marin County government control, including adequate funding and staff resources, may affect the estimated time frames for achieving targets and program implementation.

 $^2\!\mathrm{Available}$  through United States Census.

#### **Program Implementation**

The following table summarizes responsibilities, potential funding priorities, and estimated time frames for proposed implementation programs. Program implementation within the estimated time frame.<sup>1</sup> will be dependent upon the availability of adequate funding and staff resources.

Program	Responsibility	Potential Funding	Priority	Time Frame
EDU-1.a - Share Data.	CDA, County Office of Education, School Districts	Existing budget	Medium	Ongoing
EDU-1.b – Preserve Future Facilities Options.	County Office of Education, School Districts	Existing budget and may require additional grants or revenue. <sup>2</sup>	Medium	Ongoing
EDU-1.c - Consider Less- Populated Areas.	County Office of Education	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Ongoing
EDU-1.d – Monitor School Meal Programs.	H&HS, County Office of Education	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
EDU-2.a – Promote After- School Programs.	County Office of Education, School Districts, CBO's	Will require additional grants or revenue <sup>2</sup>	Medium	Ongoing
EDU-2.b – Enhance Literacy and English- Language Programs.	Library & CBO's	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
EDU-2.c - Offer Tutoring.	Civic Center Volunteers	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Short term
EDU-2.d – Include Sustainability and Diversity in School Curricula.	Environmental Education Council of Marin (EECOM), County Office of Education	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Ongoing

#### Figure 4-19 Education Program Implementation

<sup>1</sup>Time frames include: Immediate (0-1 years); Short term (1-4 years); Med. term (4-7 years); Long term (over 7 years); and Ongoing (existing programs already in progress whose implementation is expected to continue into the foreseeable future). <sup>2</sup>Completion of this task is dependent on acquiring additional funding. Consequently, funding availability could lengthen or shorten the time frame and ultimate implementation of this program.



Program	Responsibility	Potential Funding	Priority	Time Frame
EDU-2.e – Support Access to Technology.	Office of Education	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
EDU-2.f – Provide Summer Programs.	Municipal Park & Rec. Depts.	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Ongoing
EDU-2.g – Include Youth Input.	County Office of Education	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Ongoing
EDU-2.h - Encourage Financial Training.	CBO's (Marin Family Action)	Grants, other funding	High	Ongoing
EDU-2.i – Encourage Parent Education.	H&HS	Existing budget	Medium	Ongoing
EDU-2.j – Promote Volunteerism.	Civic Center Volunteers with CBO's (Volunteer Center)	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Ongoing
EDU-2.k – Prioritize Educational Opportunities for the Homeless.	H&HS	Will require additional grants or revenue <sup>2</sup>	Medium	Long term
EDU-2.1 – Support Library Services.	Library	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
EDU-2.m – Improve Library Facilities.	Library	Will require additional grants or revenue <sup>2</sup>	High	Long term
EDU-2.n – Enlist Volunteers to Supplement Programs.	Civic Center Volunteers	Existing budget	High	Ongoing
EDU-2.0 – Reach Out to Persons with Limited Access.	Library	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
EDU-2.p – Encourage Climate Change Curricula.	Marin Office of Education, CDA, and nonprofits	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Ongoing

#### 4.10 Environmental Justice

#### Background

*Environmental justice* is the fair treatment of people of all races, cultures, and incomes with respect to the development, adoption, implementation, and enforcement of criminal and environmental laws, regulations, and policies (California Government Code Section 65040.12). The Governor's Office of Planning and Passangh as and policies and policies (California Government Limitian)

Planning and Research coordinates environmental justice programs statewide, and the California Environmental Protection Agency establishes State environmental justice policies and standards.

The intent of environmental justice is to ensure that all persons are able to live in a safe and healthy environment. Identifying areas countywide where incompatible, intensive nonresidential uses are negatively impacting the quality of life is a critical step in combating environmental injustice. This Section of the Countywide Plan seeks to identify and eliminate environmental injustice in Marin. The Public Health Section of this Element contains policies and programs intended to reduce the overall volume and impact of toxic contaminants. The Public Safety Section of this Element contains policies and programs intended to foster safer communities.

"Environmental Justice means the fair treatment of people of all races, tribes and economic groups in the implementation and enforcement of environmental protection laws."

- Redefining Progress, 2000

#### **Key Trends and Issues**

#### Are there areas in Marin that suffer from environmental injustice?

The 2001 Marin Community Health Survey results indicate that Marin residents experience sensitivities to environmental exposures. Allergies or sensitivities to chemicals were reported by 17% of adults, while 7% reported electromagnetic and radio frequency sensitivities. Women more often than men reported sensitivities to these exposures. Environmental sensitivities varied by annual household income, with 27% of lower income<sup>1</sup> individuals reporting chemical sensitivities versus 14% of higher income.<sup>2</sup> individuals. Electromagnetic and radio frequency sensitivities were reported by 13% of lower income individuals reported sensitivities. Although local data is not currently available linking these sensitivities and adverse health outcomes, many County and community programs strive to ensure a safe and healthy environment for all individuals.

<sup>1</sup>Under 200% of Federal Poverty Level. <sup>2</sup>Over 300% of Federal Poverty Level.



"The United States emits one quarter of the world's gases that cause global warming."

 U.S. Energy Information Administration, 2001

#### Is Marin County exporting environmental hazards?

Marin County has a relatively small number of businesses using or emitting hazardous material in large amounts. However, while Marin County is not home to businesses such as oil refineries or plastic manufacturers, most county residents and business owners use gasoline and plastics on a daily basis. These and other products used locally can impose health impacts on the communities where they are manufactured and where they are ultimately disposed of. This can also result in the export of environmental hazards to communities that may already be disproportionately impacted by toxins. In nearby Contra Costa County, for example, where

there are a number of oil-processing companies, 67,000 pounds of known carcinogens, 93,000 pounds of known developmental toxins, and 19,800 pounds of known reproductive toxins are released into the air annually.

#### What Are the Desired Outcomes?

### GOAL EJ-I



**Environmental Justice.** Ensure that all persons in Marin live in a safe and healthy environment.

#### **Policies**

**EJ-1.1 Identify and Target Impacted Areas.** Use available measurement data to map locations with known toxins and other health-threatening pollutants.

**EJ-1.2 Reduce the Effects of Toxins.** Decrease the presence and impact of toxins, particularly in disproportionately impacted communities.

- **EJ-1.3** Avoid New Toxin Sources. Stringently evaluate the siting of facilities that might significantly increase pollution, especially near already disproportionately impacted communities.
- **EJ-1.4 Encourage County Participation in Decision Making.** Significantly increase the role and influence in land use and environmental decisions of residents from disproportionately impacted communities.

#### Why is this important?

In Bayview Hunters Point (Southeast San Francisco), 91% of the population is people of color and 30% earn less than \$15,000 per year. There are four times as many hazardous waste sites in Bayview Hunters Point as in any other part of San Francisco. Bayview Hunters Point is also home to a PG&E power plant, which generates some of the power used by Marin residents. Although Marin residents do not face any direct health impacts from the power plant, residents of Bayview Hunters Point may, as San Francisco Health Department figures indicate twice as many cervical cancer cases, twice as many breast cancer cases for African-American women under 50 years, and the highest rates of asthma hospitalization in California.

**Environment:** The United States is home to 4% of the world's population but emits 20% of the carbon dioxide ( $CO_2$ ) emissions that contribute to global climate change. In 2002, people in Marin consumed 133.9 million gallons of fuel, emitting 1.37 million tons of  $CO_2$  into the air. The impact of Marin's consumption is felt in places where fuel and other products are derived and processed and where Marin's waste is exported.

**Economy:** US IKEA, a furniture producer with a store in the Bay Area, reduced energy demand through conservation and retrofitting, saving \$514,321 annually and avoiding an estimated 4.5 million pounds of CO<sub>2</sub> emissions. The hotel chain Scandic is saving \$1.4 million annually since 1996 by reducing energy and water costs while reducing annual CO<sub>2</sub> emissions by more than 10%. Businesses can benefit from relying on products and implementing business practices that do not expose workers and surrounding communities to toxins and the related negative health impacts.

**Equity:** In Marin, asthma rates for Latino children (13%) are higher than for white children (9.6%). Asthma rates for African-American adults (21.4%) are higher than for white adults (9.1%). Low income communities are less able to afford pesticide-free food, and children in low income families are more likely to be exposed to lead-based paint as well as toxins in the air, soil, and water.

#### How will results be achieved?

#### Implementing Programs

EJ-1.a Investigate a Possible Nexus. Compare locations with high levels of toxins and sites of businesses with Hazardous Waste Permits to census tract data on income and ethnicity to determine where any correlations may exist between toxins and disproportionately impacted communities.
EJ-1.b Create a Brownfield Map. Work with the East of the Permit of

**EJ-1.b Create a Brownheld Map.** Work with the Environmental Protection Agency and local jurisdictions to identify and create a map of brownfield sites in Marin County.

"People of color, indigenous peoples and workers bear a disproportionate health, social, and economic burden of a society addicted to a fossil fuel economy."

– Dr. Robert Bullard

- **EJ-1.c** *Conduct Brownfield Education.* Provide education to elected officials and agency staff regarding the brownfield cleanup and development process and associated funding sources.
- **EJ-1.d** *Support State Efforts.* Work with the State of California Environmental Protection Agency to establish appropriate mechanisms to identify and address environmental justice gaps in land use decisions.
- **EJ-1.e** *Abate Toxins.* Where correlations are shown to exist between businesses with Hazardous Waste Permits and disproportionately impacted communities (Program EJ-1.a), take steps to abate the release of toxins into the environment by those businesses, including the following:
  - create buffer zones around significant sources of risk; and
  - relocate sources away from residential areas or sites with sensitive receptors, if feasible.
- **EJ-1.f** *Coordinate Efforts to Reduce Exposure.* Encourage State and regional agencies and County staff to work with businesses, neighborhood groups, and schools to reduce toxic exposure in disproportionately impacted communities.
- **EJ-1.g** *Deny Pollution-Source Proposals.* Amend the Development Code to incorporate the authority to withhold permits for new toxin sources.

Other CWP Goals Suppor	ting Environmental Justice
AIR - 1, 2, 3, 4	<b>TR -</b> 2, 3
<b>AG - 1</b> , 2, 3	<b>PFS -</b> 2, 3, 5
<b>CD - 1</b> , 2, 3	<b>PS -</b> 4
<b>DES -</b> 2	CP - 1
EN - 1, 2, 3	<b>PH - 1</b> , 3
MIN - 1	PK - 1

**EJ-1.h** *Require Pollution Analysis.* Amend the Development Code to require applications for new or modified facilities that may produce toxins to incorporate a pollution prevention analysis that includes:

- opportunities for replacing hazardous materials with substitute materials;
- prioritized selection of alternative materials (i.e., nontoxics are considered first, then the next-least-toxic material, etc.);
- clear justification for any proposal to use a material other than the least toxic available, based on a thorough analysis of all available alternatives;
- opportunities for using any hazardous material in a closed loop cycle.

**EJ-1.i** *Engage the Local Community.* Reach out to community members, leaders, and organizations, and environmental justice groups when considering land use actions that could affect local environmental and personal health.

\* 熊 🏌

#### Figure 4-20 Relationships of Goals to Guiding Principles

This figure illustrates the relationships of this Section's goal to the Guiding Principles.

Goal	1. Link equity, economy, and the environment locally, regionally, and globally.	2. Minimize the use of finite resources, and use all resources efficiently and effectively.	3. Reduce the use and minimize the release of hazardous materials.	4. Reduce greenhouse gas emissions that contribute to global warming.	5. Preserve our natural assets.	6. Protect our agricultural assets.	7. Provide efficient and effective transportation.	8. Supply housing affordable to the full range of our workforce and diverse community.	9. Foster businesses that create economic, environmental, and social benefits.	10. Educate and prepare our workforce and residents.	11. Cultivate ethnic, cultural, and socioeconomic diversity.	12. Support public health, public safety, and social justice.
EJ-1 Environmental Justice	•		•	٠	•			•	•		•	•

#### How Will Success Be Measured?

#### **Indicator Monitoring**

Nonbinding indicators, benchmarks, and targets <sup>1</sup> will help to measure and evaluate progress. This process will also provide a context in which to consider the need for new or revised implementation measures.

Indicator	Benchmark	Target
Amount of solid waste exported	34,594 tons in 2000.	No increase through 2020.
from Marin County annually.		

<sup>1</sup>Many factors beyond Marin County government control, including adequate funding and staff resources, may affect the estimated time frames for achieving targets and program implementation.

#### **Program Implementation**

The following table summarizes responsibilities, potential funding priorities, and estimated time frames for proposed implementation programs. Program implementation within the estimated time frame.<sup>1</sup> will be dependent upon the availability of adequate funding and staff resources.

Program	Responsibility	Potential Funding	Priority	Time Frame
EJ-1.a – Investigate a Possible Nexus.	H&HS, CDA, CBO's	Existing budget, will require additional grants or revenue. <sup>2</sup>	Medium	Med. term
EJ-1.b – Create a Brownfield Map.	California DHS, CDA, Hazardous and Solid Waste JPA	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Ongoing
EJ-1.c – Conduct Brownfield Education.	CDA	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
EJ-1.d – Support State Efforts.	CDA	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Ongoing
EJ-1.e – Abate Toxins.	California DHS, BAAQMD, CDA	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
EJ-1.f – Coordinate Efforts to Reduce Exposure.	California DHS, BAAQMD, CDA, H&HS	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Ongoing
EJ-1.g – Deny Pollution- Source Proposals.	CDA	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Med. term
EJ-1.h – Require Pollution Analysis.	CDA	Existing budget	Medium	Med. term
EJ-1.i – Engage the Local Community.	CDA	Existing budget	High	Ongoing

	Figure 4-21	
Environmental	Justice Program	Implementation

<sup>1</sup>Time frames include: Immediate (0-1 years); Short term (1-4 years); Med. term (4-7 years); Long term (over 7 years); and Ongoing (existing programs already in progress whose implementation is expected to continue into the foreseeable future). <sup>2</sup>Completion of this task is dependent on acquiring additional funding. Consequently, funding availability could lengthen or shorten the time frame and ultimate implementation of this program.

# 秋 \* \* \*

MARIN COUNTYWIDE PLAN



Healthy Marin Partnership Prevention Pavilion, Marin County Fair

### 4.11 Public Health

#### Background

Health is influenced by many factors outside the health care system. These factors, known as determinants of health, include social and physical environments, lifestyle, behavior, and genetics (see Figure 4-22). Successful efforts to address the determinants of health require a public health approach.

Public health focuses on the well-being of populations and communities, and on the principle that everyone is entitled to protection from the world's hazards and from unnecessary death and disability.<sup>1</sup>

<sup>1</sup>D. Beauchamp, *Public Health as Social Justice.* 

At its core, public health is about prevention and creating social, economic, and environmental conditions that support health. Promoting and protecting the health, safety, self-sufficiency, and well-being of a community requires a comprehensive, well-coordinated effort that employs multiple strategies and enlists many partners from the community. It is this combination of science, skills, and commitment directed to the maintenance and improvement of the health of all people through collective action that makes public health practices successful.



Policies, legislation, and organizational practices are powerful influences in shaping an individual's attitude. Therefore, strengthening someone's skills and knowledge alone may not be sufficient to prevent unhealthy, disease-causing behavior. Public health activities that focus exclusively on individual behavioral change isolated from broader community factors will have limited success. Any effort to improve the health of a community must be part of comprehensive efforts that include policies, programs, and organizational practices.

#### Figure 4-23 Components for a Comprehensive Community Health Effort

- Influencing policy and legislation
- Mobilizing neighborhoods and communities
- Changing organizational practices
- Fostering coalitions and networks
- Educating providers
- Promoting community education
- Strengthening individual knowledge and skills

Since tobacco use, poor diet, lack of exercise, and alcohol consumption underlie the top 10 leading causes of death in Marin (see Figures 4-24 and 4-25), adopting healthy lifestyles that avoid such behaviors can prevent or reduce the devastating effects of chronic diseases such as heart disease, cancer, stroke, and diabetes.



Local governments and their community partners are in a unique position to improve the public's health because of their capacity to establish public policies and educate the community, factors that are critical in influencing the conditions and norms that play a key role in chronic disease and death. For example, land use policies that restrict alcohol and tobacco outlet density have been associated with decreases in underage alcohol and tobacco use and reduced crime, including violence and **DUI** arrests.

Promoting and protecting the health, safety, self-sufficiency, and well-being of the Marin community requires controlling the availability, accessibility, acceptability, marketing, and promotion of tobacco, alcohol, food and nutrition options, and physical activity, while ensuring that affordable, appropriate, and quality services are accessible for all residents. Prevention efforts must be addressed in a comprehensive and coordinated manner, utilizing multiple strategies in multiple arenas and across populations, with a particular emphasis on low income and minority populations who face a higher prevalence of chronic conditions and increased health risk.

#### MARIN COUNTYWIDE PLAN



#### Figure 4-25 Factors Underlying the Leading Causes of Death in the United States, 20001

<sup>1</sup>Mokdad, A.H., Marks J.S., et al. Actual Causes of Death in the United States 2000. JAMA 2004; 291(10): 1238–1246. Note: Doesn't add up to 100% because list not exhaustive.



This Section of the Countywide Plan focuses on addressing the key determinants of health, namely the behavioral patterns that underlie the leading causes of death in Marin – alcohol consumption, tobacco use, poor diet, and lack of physical activity. Successfully influencing these behaviors requires a comprehensive approach that results in community norms and conditions that support individuals to make healthier choices. This Section will also address other determinants of health, including access to health care, and environmental factors, such as access to housing.

#### **Key Trends and Issues**

#### Are eating habits in Marin leading to obesity and other health problems.<sup>9</sup>

**Community Health Survey results indicated that Marin shares in the national obesity epidemic.** The county is a long way from achieving the U.S. Healthy People 2010 goal of no more than 5% of children and adolescents being overweight and 15% of adults being obese (see Figure 4-27). Chronic diseases associated with obesity are also increasing. Factors contributing to obesity include poor eating habits, lack of physical activity, and school and community environments that make it difficult to access healthy foods and physical activity. For the senior population, concerns with obesity and unhealthy lifestyles are also alarming. The Marin Community Health Survey (2001) found that 50.7% of Marin adults over the age of 60 are overweight or obese, and only 34.7% of seniors eat five servings of fruits and vegetables daily. The survey also indicates that 17.2% of seniors 60 and older never get moderate physical activity.



#### Is alcohol consumption in Marin a major health issue?

The high rates of alcohol consumption among Marin teens, adults, and seniors merit serious concern. The California Healthy Kids Survey revealed that teenagers drink alcohol at levels that exceed both state and national averages for 30-day use and "binge drinking." Alcohol use is associated with the top leading causes of death and injury among teens, including homicide, suicide, automobile crashes, fights, and alcohol-related sexual assault. In addition, pro-alcohol marketing and promotion influences, ease of youth access to alcohol, and high levels of alcohol consumption among adults all contribute to an environment saturated with permissive social norms around alcohol use and over-consumption. This issue is of particular concern among Marin's senior population, as the California Health Interview Survey revealed that 72% of senior citizens reported drinking alcohol regularly and 6% reported drinking more than five drinks on any one occasion, exceeding state and national averages.



Are young people in Marin still smoking?

Despite the success of the local and statewide tobacco control and prevention efforts, some Marin residents are still smoking. Local surveys, reports from school nurses and counselors, and statewide studies indicate that there is still progress to be made in the area of youth smoking. Aggressive tobacco industry marketing strategies have led to an increase in smoking among 18- to 24-year-olds (a 5% increase occurred in California between 1994 and 2000). Professional focus groups in Marin County in 2002 revealed increases in tobacco-marijuana blended products and marijuana smoking among youth.

The 15.3% of Marin adults who still smoke create secondhand smoke dangers for all residents, especially children. Children are most susceptible to problems associated with smoke, such as ear



infections, eye irritation, congestion, asthma, throat irritation, pneumonia, bronchitis, and cancer. Exposure to secondhand smoke continues in cars, homes, building entrances, waiting lines, parks, bus stops, and other outdoor venues.

#### Do Marin residents have access to affordable, quality health care?

Affordability and availability are major issues. Persons from lower income families, especially children, are most likely to obtain care at emergency rooms and may not get needed preventive or ongoing health care, such as immunizations and vision and dental checks. Language and cultural barriers may also limit access to quality care. Even for families fortunate enough to have health insurance, lack of provider capacity can impede timely access to health services, particularly specialty and dental care. Twenty-three percent of Hispanic/Latino adults have no health care coverage, compared with 6.2% of Non-Hispanic White adults. Hispanics were significantly less likely to have health care coverage than any other ethnicity.

Approximately 3,300 Marin children currently lack health insurance and may have difficulty accessing health care, according to local data, including the Marin Community Health Survey, reports from School Nurses and School Linked Service caseworkers, data from the Child Health and Disability Program, as well as data compiled by Certified Applicant Assistors in the community. Of this number, approximately two-thirds are eligible for public insurance programs. This indicates a need for integrated and proactive outreach, enrollment, and retention efforts. The parents of these children often do not qualify for public health programs and cannot afford private health insurance. Also, according to the 2000 Marin Community Health Survey, 12.1% of Hispanic/Latino children have no health care coverage compared to 1.8% of Non-Hispanic White children, and 15.4% of children with household incomes under the Federal Poverty Level (FPL) had no health care coverage. Only 0.9% of children with household incomes 300% of the FPL had no health care coverage.

The quality of health care depends largely on health insurance. Coverage in Marin varies by age, income, and ethnicity (as reported in the 2001 Marin Community Health Survey). While more than 90% of Marin adults have health insurance, that number is less than 80% for persons 18 to 24. Only 64% of adults in low income households have health coverage. Only 76% of adults of Hispanic origin have health insurance. Although Medicare, and in some cases Medi-Cal, provide health care coverage for seniors, the Marin Community Health Survey found that 58% of Marin older adults 65 and older have no coverage for dental services, 38.8% have no coverage for mental health services, and 28.2% have no coverage for eye exams.

# Are residential care housing and home care services for Marin's older adults affordable and available?

Home care for seniors needs to be more affordable and available. Growing older may mean increasing physical and mental impairments that can rob people of their independence. The Buck Institute for Age Research reported in its 1995 study *Health and Functioning in Marin* that one out of every four persons over the age of 85 had trouble bathing, while one in three had difficulty walking, and one in four over 75 needed assistance with at least one activity of daily living, such as shopping, housekeeping, and meal preparation. The aging of the population is placing pressure on the health care system and the provision of emergency medical services. Many older adults need treatment for discomfort or pain, and

about one-third of county seniors experience some kind of disability. As demand and costs rise for inhome services and assisted living, spouses and families of seniors increasingly must serve as caregivers.

There is a shortage of residential care housing for low income seniors and people with special needs. Marin has only approximately 1,000 low income independent housing units with long wait lists for occupancy, while almost one in two county residents over 75 (about 8,000 persons) meets the very low income criteria set by HUD (less than \$40,000/year). While 79% of persons age 60 and older are homeowners in the county, some are struggling to maintain their homes and stay in them as long as possible. Many of the approximately 13,000 seniors who rent would qualify for affordable senior housing if more units were available.

#### What Are the Desired Outcomes?

#### **GOAL PH-I**



Reduced Rates of Obesity, Eating Disorders, and Chronic Disease Such as Heart Disease and Breast Cancer. Improve individual and community health through prevention, screening, education, and treatment strategies regarding nutrition and physical activity related health issues.

#### **Policies**

**PH-1.1 Promote Nutrition Education and Access to Healthy Foods.** Provide affordable healthy foods, and fresh, locally grown fruits and vegetables in schools and other public places.

- **PH-1.2 Promote Physical Activity.** Increase opportunities for and interest in safe and pleasant physical activity.
- **PH-1.3 Promote Healthy Environments.** Provide school and community environments and policies that foster healthy lifestyles and behavior.
- PH-1.4 Develop Disease Prevention and Management Programs. Create collaborative community-based chronic disease screening, prevention, and management programs.

#### Why is this important?

Fifty-eight percent of adult men, 49% of seniors, 38% of boys, and 30% of girls between the ages of 2 and 17 are overweight or obese. Poor nutritional habits and lack of physical activity can lead to obesity, beginning the causal pathway of many chronic disease disorders such as diabetes, osteoporosis, heart disease, and hypertension; lack of cognitive development in children; and, in extreme cases, eating disorders and death.

**Environment:** Land use policies that protect and foster sustainable agricultural practices, diverse local food production, and local marketing support a physically healthy community.



Healthy Marin Partnership Prevention Pavilion, Marin County Fair

**Economy:** Nationally, the public pays about \$39 billion a year, the equivalent of \$175 per person, for Medicare and Medicaid expenses that are obesity related (i.e., Type II diabetes, cardiovascular disease, and some types of cancer). Chronic disease disorders burden the economy with very high health care costs, are associated with both mental and physical failure, and result in a less productive workforce.

**Equity:** Inactivity and poor diet are beginning to approach tobacco use as the leading causes of death for Americans. Media and marketing practices aimed at promoting unhealthy foods and beverages, while idealizing certain body types, promote disordered eating behaviors. Across the country, 200 school districts have signed exclusive contracts with soft-drink companies to sell their beverages in schools. Policies that promote healthy eating and physical activity are therefore vital to health and social wellbeing. Adequate access to healthy foods and physical activity promotes healthfulness in children and families, and better academic performance in schools, and helps to prevent sickness and chronic disease for all residents.

#### How will results be achieved?

#### Implementing Programs

**PH-1.a** *Implement Policies That Promote Healthy Eating and Physical Activity.* Advocate and support policies that do the following: encourage sound nutrition, physical activity, and education programs in all schools, senior centers, and community-based organizations; work with local vendors and agricultural forums to develop access to and availability of affordable, quality, locally grown foods for schools and the community, especially for individuals with limited income or at risk of disease; promote organizational policies around providing healthy food options at meetings, in vending machines, and food concessions, and providing opportunities to engage in physical activity; support land

use policies, zoning, and conditional use permit regulations to control the location and density of food and physical activity choices, including sidewalks, safe routes to schools, parks, gardens, etc., to promote healthier communities; advocate and support policies that restrict the availability, accessibility, placement, and promotion of low-nutrient-dense foods.

- PH-1.b Increase Access to Healthy Foods/Beverages. Support neighborhood-oriented, specific sources of healthful foods, such as farmers' markets and local outlets. Support food banks, pantries, and other sources that help provide federal food assistance to low income residents so that all families, seniors, schools, and community-based organizations are able to access, purchase, and increase intake of fresh fruits, vegetables, and other nutritious foods.
- **PH-1.c** *Collaborate on Breast Cancer Research.* Continue communications between the Buck Institute, Health and Human Services, and other interested organizations on the epidemiology of breast cancer and relevant research on the biology of breast cancer.
- PH-1.d *Improve Access to Early Prevention Screening and Treatment.* Promote reimbursement for and mobilize physicians, registered dietitians, and other qualified health care workers to screen and treat behaviors that may lead to obesity and/or eating disorders, breast cancer, and other preventable diseases. Make public education, social marketing, and media campaigns about the prevention and treatment of obesity and disordered eating a countywide priority. Continue public education regarding the risk factors for and early identification of breast cancer.
- PH-1.e Support School and Community Physical Fitness and Nutrition Education Programs. Develop and sustain the Marin Nutrition Wellness and Physical Activity Collaborative, comprising schools, community groups, and organizations that can help implement and support school- and community-based nutrition and physical fitness programs and policies. Develop a strategic plan around funding and support for sustainable programs.
- **PH-1.f** *Provide Reliable Information, Training, and Technical Assistance.* Support schools, senior centers, and community organizations in efforts to develop and implement nutrition and physical education standards and policies.

#### What Are the Desired Outcomes?

### **GOAL PH-2**

**Communities Free of Tobacco and Alcohol Dependency, and Other Drug Abuse Problems.** Reduce alcohol, tobacco, and other drug use by creating a social and physical environment that supports healthy choices, ensuring access to treatment services and enforcing existing regulations.

Adopt and Enforce Tobacco Control Laws. Continue

#### **Policies**

**PH-2.1** 



updating and enforcing tobacco control laws that pertain to location and retailing practices of tobacco stores (that sell to underage youth), smoking restrictions, and smoke-free home and workplace laws.

- **PH-2.2 Reduce Youth Access to Alcohol.** Make alcohol less available, accessible, and acceptable to youth, and restrict promotion and placement of alcohol targeted at youth through updating and enforcing underage drinking laws, especially those that pertain to retail establishments and social settings.
- **PH-2.3 Raise Awareness of Alcohol and Other Drug Issues.** Reduce positive associations and increase perceptions of harm among youth by changing social norms through supporting a comprehensive, evidence-based prevention approach that includes media, enforcement, education, and policy.
- **PH-2.4 Expand Evidence-Based Prevention and Treatment Programs.** Ensure that alcohol and substance abuse prevention and treatment programs utilize best available practices.
- **PH-2.5** Increase Tobacco Cessation Services. Ensure that smoking cessation services are available to Marin residents.
- **PH-2.6** Improve Access to Treatment Services. Ensure that Marin residents have access to affordable, high-quality, culturally relevant, linguistically, geographically, and age-appropriate treatment services that reflect the needs of the population and drug abuse trends.



#### Why is this important?

In 1998, the cost of drug abuse in the United States was \$143.4 billion. Comprehensive and effective prevention and treatment services can help ensure significant cost savings to society. Residents of Marin County are impacted by the health and social problems associated with alcohol and other drugs.



"Heightened smog levels trigger asthma attacks and pose health threats to children and the elderly in particular."

> –Natural Resources Defense Council, 2004

**Economy:** In California, currently the estimated cost of alcohol and other drug abuse to society is \$35 billion annually. This estimate only includes costs associated with loss of productivity, losses due to crime, health care costs, prevention and treatment services, and criminal justice costs. Reducing the health care burden from tobacco, alcohol, and substance use will increase the overall health of the community.

**Equity:** In 2002, the estimated number of persons ages 12 or older needing treatment for an illicit drug problem was 7.7 million in the United States. Locally, the disproportionately high rates of substance use and the general social acceptability of alcohol manifest themselves

in juvenile arrests, hospital discharges, and death rates above the state average. Smoking often leads to chronic diseases including lung, breast, and other forms of cancer, as well as to cardiovascular diseases such as stroke and heart attacks. Secondhand smoke contributes to pediatric diseases and hospitalizations, asthma, sudden infant death syndrome, and adult respiratory and cardiovascular diseases.

#### How will results be achieved?

#### Implementing Programs

PH-2.a	<i>Support Alcohol Policy Development.</i> Promote policies that restrict the availability, accessibility, promotion, and placement of alcohol in proximity to youth. Collaborate with local law enforcement agencies, community coalitions, and Health and Human Services to consider adopting land use and alcohol policies, such as Social Host ordinances, conditional use permit ordinances, or related approved ordinances.
PH-2.b	<i>Encourage Enforcement of Alcohol Laws.</i> Coordinate and collaborate with local law enforcement to conduct merchant education and compliance checks of establishments licensed with Alcohol Beverage Control.
PH-2.c	<b>Conduct Media Activities.</b> Provide regular submissions to local radio, print, and televised media highlighting significant issues and trends related to alcohol and other drug use, as well as available prevention and treatment services.
PH-2.d	<i>Implement Responsible Beverage Service Programs.</i> Form a local coalition of youth, parents, health advocates, the business community, law enforcement, and interested

community members to establish and advocate for Responsible Beverage Service programs and policies, and train event hosts and establishments licensed with Alcohol Beverage Control in responsible sales and service practices.

- **PH-2.e** *Provide Training and Technical Assistance.* Increase the capacity of service providers and other community partners by providing regular and ongoing training and technical assistance in areas including, but not limited to, the following: best practices for alcohol and other drug prevention and treatment, problem identification and referral, dual diagnosis, working with diverse populations, media advocacy, program evaluation, trends in alcohol and other drug use, and policy advocacy.
- **PH-2.f** *Develop and Maintain Community Partnerships.* Participate in and support collaborative relationships that address relevant and related public health issues and are inclusive of nontraditional partners.
- **PH-2.g** *Provide Tobacco Education.* Support public education programs regarding the hazards of tobacco use and secondhand smoke through media and educational events, and by encouraging incorporation of cessation and prevention programs into the protocols of community-based organizations.
- PH-2.h *Enhance Youth Tobacco Prevention Programs.* Increase school- and college-based prevention programs to discourage tobacco addiction among youth and 18- to 24-year-olds, the fastest-growing group of smokers in California.



**PH-2.i** *Provide Smoking Cessation Programs.* Support efforts to provide residents with access to smoking treatment programs at hospitals and clinics, and through substance abuse treatment providers' community-based organizations, schools, colleges, and employers.



- PH-2.j *Enforce Tobacco Control Laws.* Maintain and enhance the partnership between local law enforcement and the County Environmental Health and Health and Human Services Tobacco Education Program to enforce regulations pertaining to smoking and selling tobacco to minors.
- **PH-2.k** *Seek Funding for Substance Abuse Prevention.* Maintain a continuum of care and expand services using available funding sources to address prevention and treatment gaps.
- PH-2.1 *Conduct Regular and Ongoing Evaluations.* Perform program monitoring and evaluation to assess substance abuse prevention program performance, effectiveness, and alignment with desired outcomes.

#### What Are the Desired Outcomes?

#### GOAL PH-3



Adequate Access to Quality Health Care. Ensure that all community members have affordable and convenient access to a full range of primary, preventive, and specialty health care, including mental health, vision, and dental care.

#### **Policies**

**PH-3.1** Increase Provider Capacity. Ensure that additional health care services are provided in a way that is culturally sensitive and linguistically appropriate for the diverse communities of Marin.

- **PH-3.2** Increase Health Insurance Options. Enhance funding for health insurance products for children and adults not eligible for publicly funded health programs, including support for state and national single payer systems.
- **PH-3.3** Enhance Access to Public Benefit Programs. Create integrated access to all health and public benefit programs in the community.
- **PH-3.4** Increase Awareness of Preventive Health Care. Promote and support cross-cultural education and awareness of the importance of a regular health care provider and preventive health care.
- **PH-3.5** Enhance Access to Mental Health Programs. Ensure a full range of mental health services from outpatient to acute care for the mentally ill.

#### Why is this important?

In 2001, there were 8,752 medical emergency illnesses in Marin. When routine preventive health care is inaccessible due to either lack of insurance or lack of provider capacity, care is given in the highest-

cost setting. Adequate health care and insurance are essential to the overall vitality of the community because the high social and economic costs to society of incomplete care and coverage are tremendous.

**Economy:** Health insurance coverage reduces the probability that workers will change jobs by 26% to 31%. Providing full coverage can decrease health related absenteeism of employees by an average of 9%. Adequate access to health care reduces costs to the community by avoiding unnecessary hospitalizations and emergency room visits, and minimizes use of high-cost services such as skilled nursing home and in-home support services.

**Equity:** The poverty rate is steadily increasing in Marin despite a decrease in California and the nation. When residents cannot pay for health insurance coverage, they must choose between receiving no care or paying a higher cost during a medical emergency. Access to quality health care promotes good health in children from birth, supports learning in school-age children, and prevents unnecessary sickness, disability, and death for all residents.

#### How will results be achieved?

#### Implementing Programs

PH-3.a	<i>Foster Collaboration.</i> Facilitate collaborative partnerships between County Health and Human Services, medical professionals, community-based agencies, and service providers, schools, and local funding agencies.
<b>PH-</b> 3.b	<i>Support Health Educational Opportunities.</i> Support and encourage training and educational opportunities in the health professions to meet the needs of an increasingly diverse population.
PH-3.c	<i>Improve Access to Health Care.</i> Improve access to health care for underserved populations by expanding the children's health initiative; transitioning that initiative as feasible to include other underserved populations; developing a one-stop electronic application process that can be used to enroll clients in health insurance and other public benefit programs; and supporting State and national proposals for a single payer system.
PH-3.d	<i>Improve Service Delivery and Utilization.</i> Develop proactive outreach and enrollment programs for insurance benefits; integrate case management services with primary medical care; support electronic medical record portability to improve utilization and quality of services; promote preventive care; and ensure insurance retention.
РН-3.е	<i>Support Disease Prevention Education Programs.</i> Promote chronic disease self- management education, and support prevention programs, including prenatal services, tailored to the social and cultural perspectives of a diverse population.
<b>PH-</b> 3.f	<i>Employ a Precautionary Approach.</i> To guide disease prevention efforts in the county, utilize the Precautionary Principle, which calls for taking precautionary measures when an activity may threaten human health.



**PH-3.g** Support Mental Health Programs. Encourage adequate funding to provide a full range of mental health treatment programs, support services, and psychiatric beds for the mentally ill, particularly for the severely and emotionally disturbed.

#### What Are the Desired Outcomes?

#### **GOAL PH-4**



Increase Access to Care and Services for Seniors Who Reside in Their Own Homes and Residents of Affordable Housing. Improve access to health care for individuals and groups that traditionally have difficulty obtaining adequate care and appropriate housing.

#### **Policies**

**PH-4.1 Enhance Low income Independent Housing for Seniors.** Continue to develop and expand senior housing services, such as the Division of Aging's demonstration program, Healthy Mackey Terrace, at the Environmental Association for Housing's (EAH) senior apartment complex in Novato.

- PH-4.2 Improve Linkage Between Housing and Services. Advocate for improved linkages between housing providers and the County Health and Human Services Department.
- PH-4.3 Expand Access to Private Home Care. Ensure that seniors and persons with special housing needs obtain adequate, affordable, in-home health care services.
- PH-4.4 Promote Care for Seniors and Lower income Households. Encourage expansion of inhome care, assisted-living opportunities, and services to Healthy Families and Medi-Cal patients.
- **PH-4.5 Provide Opportunities to "Age in Place."** Provide increased opportunities for seniors and persons with disabilities to remain in their homes and communities.
- PH-4.6 Plan for Climate Change. Plan for the public health implications of climate change, including disease and temperature effects.

#### Why is this important?

Adequate and affordable health care for everyone is essential to the functioning and health of the entire community. Increasingly, older adults are forced to leave the county to find affordable housing that includes appropriate care, and people with lower incomes or special needs encounter difficulty obtaining and paying for medical services and finding appropriate housing. Quality home care can make the difference between whether a person stays in his or her own home or whether he or she must move to a more restrictive housing environment, such as an assisted-living facility or a nursing home.

Without adequate and affordable home care, increasing numbers of older persons will have to be institutionalized at much greater societal and personal cost.

**Economy:** The lack of affordable assisted living in Marin leaves older persons living in affordable senior housing complexes without adequate support and with no choice but nursing home care as they age and need more services. Rent for self-sufficient seniors at one of Marin County's retirement residences costs an average of \$3,207 per month. Assisted-living facilities that provide protective oversight cost an average of \$3,689 per month or \$44,268 per year. Convalescent or nursing homes that provide full medical support can cost an average of \$54,000 a year. In contrast, home and community care with minimal supervision is more cost effective at only \$5,820 per year.

**Equity:** There is little to no affordable assisted living in Marin County. Prices for retirement residences and assisted-living facilities on average cost over \$3,000 a month. For the over 1,000 low income older persons living in senior housing in Marin, nursing home care may be the only choice when they need

more assistance. Providing low-cost preventive services in senior housing complexes is more cost effective. It also supports an individual's right to live as independently as possible.

#### How Will Results be Achieved?

#### Implementing Programs

PH-4.a	Co-locate Adult Day Health Care with
	Senior Housing. Encourage efforts by
	EAH and Senior Access or others to
	develop new senior housing with on-site
	health care services.

- PH-4.b *Improve Service Linkages.* Explore the financial feasibility of providing all residents in low income housing with access to a Service Coordinator.
- PH-4.c *Expand Senior Services.* Amend the Development Code to allow senior day care services as a permitted use in residential zones, and use inclusionary housing requirements to develop a fund for services in existing low income housing.



- Entrances to homes without steps
- Hallways and doors that comfortably accommodate strollers and wheelchairs
- Lever door handles and doors of the appropriate weight
- Electrical outlets that can be accessed without having to move furniture
- Rocker-action light switches to aid people with a loss of finger dexterity
- Showers that can accommodate a wheelchair, and that have adjustable shower heads to accommodate people of different heights
- Kitchens with varying counter heights



- **PH-4.d** *Advocate for Affordable Assisted Living.* Advocate for level of care reimbursement methodologies to increase affordable assisted living, including for persons with mental illness.
- PH-4.e *Improve Balance of Community and Institutional Care.* Seek a Home and Community-Based Services waiver to redress any inappropriate balance of community versus institutional care.
- PH-4.f *Expand Affordable Housing for Seniors and People with Special Needs.* Advocate for the development of shared housing and other strategies and encourage universal design as an operating principle for housing providers, CDBG, and the Marin Community Foundation.
- **PH-4.g** *Develop Models for Shared In-Home Supportive Services.* Explore options for inhome support services such as clustering provision times within a senior or disabled residence, and simplify approval procedures for shared services consumers.
- PH-4.h *Expand Home Care Registry.* Maintain and expand the home care registry in Marin.
- PH-4.i *Expand Home Care Services.* Pursue funding to subsidized home care for low income seniors, support wage increases, and offer training opportunities for home care workers, create an On-Call Emergency Home Care Back-Up System, and develop an assisted-management home care model.
- PH-4.j *Lobby for Increased Medi-Cal Options.* Advocate for waivers to allow Medi-Cal payments for home- and community-based services.
- PH-4.k *Provide Prescription Drug Assistance.* Continue Adult Social Service referrals for seniors to access resources for low-cost prescriptions via the intake information referral unit.
- PH-4.1 *Promote Universal Building Designs.* Encourage universal building design techniques that enable seniors and persons with disabilities to remain in their homes by implementing the recommendations within the Marin County Single Family Residential Design Guidelines and considering additional requirements when updating the Development Code.
- PH-4.m *Identify Potential Responses to Climate Change.* Work with the Intergovernmental Panel on Climate Change (IPCC) and other leading health organizations to identify critical public health issues and identify potential responses necessary related to climate change.



#### Figure 4-29 Countywide Plan Policies and Programs Related to the Senior Population

Section	Policy/Program Number	Description
2.6	Agriculture and Food	
	Program AG-2.n	Increase Access to Local, Quality Food for All
2.9	Trails	
	Program TRL-1.b	Develop Designation Criteria
	Policy TRL-2.5	Provide Access for Seniors and Persons with Special Needs
	Program TRL-2.i	Evaluate Existing Trails for Seniors and Disabled Access
	Program TRL-2.j	Distribute Information about Seniors and Disabled Access
3.4	<b>Community Development</b>	
	Policy CD-2.1	Provide a Mix of Housing
	Program CD-2.a	Increase the Affordable Housing Supply
	Program CD-2.b	Provide a Variety of Housing Types and Prices
	Program CD-2.c	Enact Zoning Changes
	Program CD-2.d	Identify and Plan Mixed-Use Sites
3.5	Community Design	
	Program DES-2.b	Universal Design
3.7	Environmental Hazards	
	Policy EH-2.3	Ensure Safety of High-Occupancy Structures
3.8	Housing	
	Program HS-2.d	Assist in Maximizing Use of Rehabilitation Programs
	Policy HS-3.10	Encourage Home Sharing
	Program HS-3.e	Modify Multi-Family Sites Zoning
	Program HS-3.f	Seek Increased Multi-Family Housing Opportunities
	Program HS-3.h	Encourage Co-housing, Cooperatives, and Similar Collaborative Housing
	Policy HS-3.28	Legalize Existing Second Units
	Policy HS-4.1	Cater to Special Needs Groups
	Program HS-4.c	Coordinate Efforts in the Effective Use of Available Rental Assistance
		Programs
	Policy HS-4.2	Provide Affordable Housing for Special Needs Households
	Policy HS-4.3	Offer Density Bonuses for Special Needs Households
	Policy HS-4.8	Link Health and Human Services Programs
	Policy HS-4.9	Provide Emergency Housing Assistance
3.9	Transportation	
	Program TR-2.b	Countdown Crossing
3.13	St. Vincent's/Silveira	
	Policy SV-2.3	Land Use Designations/Desired Outcomes
4.4	Economy	
	Program EC-2.b	Encourage Employment
4.7	Community Participation	
	Program CP-1.g	Encourage Attendance by All Community Members



Section	Policy/Program Number	Description
4.9	Education	
	Program EDU-2.m	Reach Out to Persons with Limited Access
4.11	Public Health	
	Program PH-1.a	Implement Policies That Promote Healthy Eating and Physical Activity
	Program PH-1.c	Collaborate on Breast Cancer Research
	Program PH-1.d	Improve Access to Early Prevention-Screening and Treatment
	Program PH-1.e	Support School and Community Physical Fitness and Nutrition Programs
	Program PH-1-f	Provide Reliable Information, Training, and Technical Assistance
	Program PH-3.c	Streamline the Application Process
	Program PH-3.d	Improve Service Delivery and Utilization
	Policy PH-4.1	Enhance Low income Independent Housing for Seniors
	Policy PH-4.3	Expand Access to Private Home Care
	Program PH-4.a	Co-locate Adult Day Health Care with Senior Housing
	Program PH-4.b	Improve Service Linkages
	Program PH-4.c	Expand Senior Services
	Program PH-4.f	Expand Affordable Housing for Seniors and People with Special Needs
	Program PH-4.g	Develop Models for Shared In-Home Supportive Services
	Program PH-4.h	Expand Home Care Registry
	Program PH-4.i	Expand Home Care Services
4.12	Arts and Culture	
	Program ART-2.d	Encourage Arts Programs for Special Needs Groups
4.14	Parks and Recreation	
	Program PK-1.j	Meet Special Group Needs

熊茶

#### Figure 4-30 Relationships of Goals to Guiding Principles

This figure illustrates the relationships of each goal in this Section to the Guiding Principles.

Goals Goals	1. Link equity, economy, and the environment locally, regionally, and globally.	2. Minimize the use of finite resources, and use all resources efficiently and effectively.	3. Reduce the use and minimize the release of hazardous materials.	4. Reduce greenhouse gas emissions that contribute to global warming.	5. Preserve our natural assets.	6. Protect our agricultural assets.	7. Provide efficient and effective transportation.	8. Supply housing affordable to the full range of our workforce and diverse community.	9. Foster businesses that create economic, environmental, and social benefits.	10. Educate and prepare our workforce and residents.	11. Cultivate ethnic, cultural, and socioeconomic diversity.	12. Support public health, public safety, and social justice.
PH-1 Reduced Rates of												
Obesity, Eating												
Disorders, and Chronic												
Disease Such as Heart												•
Disease and Breast												
Cancer												
PH-2 Communities												
Free of Tobacco and												
Alcohol Dependency,												•
and Other Drug Abuse												
Problems												
PH-3 Adequate Access												
to Quality Health Care												-
PH-4 Increase Access												
to Care and Services												
for Seniors Who							•	•				
Reside in Their Own							-	-				•
Homes and Residents												
of Affordable Housing												

#### How Will Success Be Measured?

#### **Indicator Monitoring**

Nonbinding indicators, benchmarks, and targets <sup>1</sup> will help to measure and evaluate progress. This process will also provide a context in which to consider the need for new or revised implementation measures.

Indicators	Benchmarks	Targets			
Percent of insured county residents.	92% in 2001.	No decrease by 2020.			
Percent of insured County employees.	100% in 2000.	No decrease through 2020.			
Percent of Medi-Cal users.	56% in 2000.	Achieve 60% users (out of eligible population) in 2015 and 65% in 2020.			
Numbers of children/youth insured annually through Medi- Cal and Healthy Families.	5,235 in 2000.	Increase the number of children insured through 2020.			
Number of servings of fruits and vegetables consumed daily by children.	53% of children ate five or more servings of fruit and vegetables in the day prior to the survey.	Percentage of children eating 5 or more servings of fruit and vegetables per day increases 10% by 2020.			
Amount of time children in grade 11 spend engaged in physical activity.	53% of children exercised at least 20 minutes on at least 3 days per week in 2002.	Amount of time spent in physical activity by children-grade 11 increases by 10% in 2020.			
Percent of population overweight and obese by age and gender.	See H&HS dataset.	Obesity decreases 10% by 2020.			

<sup>1</sup>Many factors beyond Marin County government control, including adequate funding and staff resources, may affect the estimated time frames for achieving targets and program implementation.

#### **Program Implementation**

The following table summarizes responsibilities, potential funding priorities, and estimated time frames for proposed implementation programs. Program implementation within the estimated time frame <sup>1</sup> will be dependent upon the availability of adequate funding and staff resources.

Program	Responsibility	Potential Funding	Priority	Time Frame	
PH-1.a – Implement Policies That Promote Healthy Eating and Physical Activity.	H&HS, Schools; Community Partners	Existing budget and may require additional grants or revenue. <sup>2</sup>	High	Ongoing	
PH-1.b – Increase Access to Healthy Foods/Beverages.	H&HS, Agriculture Commissioner, Farm Advisor	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing	
PH-1.c – Collaborate on Breast Cancer Research.	H&HS, CBO's	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing	
PH-1.d – Improve Access to Early Prevention– Screening and Treatment.	H&HS, Community Partners	Existing budget	High	Ongoing	
PH-1.e – Support School and Community Physical Fitness and Nutrition Education Programs.	H&HS, Community Partners	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing	
PH-1.f – Provide Reliable Information, Training, and Technical Assistance.	H&HS	Existing budget	High	Ongoing	
PH-2.a – Support Alcohol Policy Development.	H&HS, CDA, Local Policymakers, Community Partners	Will require additional grants or revenue <sup>2</sup>	High	Ongoing	
PH-2.b – Encourage Enforcement of Alcohol Laws.	Local Law Enforcement; H&HS CDA	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing	

#### Figure 4-3 I Public Health Program Implementation

<sup>&</sup>lt;sup>1</sup>Time frames include: Immediate (0-1 years); Short term (1-4 years); Med. term (4-7 years); Long term (over 7 years); and Ongoing (existing programs already in progress whose implementation is expected to continue into the foreseeable future). <sup>2</sup>Completion of this task is dependent on acquiring additional funding. Consequently, funding availability could lengthen or shorten the time frame and ultimate implementation of this program.



Program	Responsibility	Potential Funding	Priority	Time Frame	
PH-2.c – Conduct Media Activities.	H&HS, Local Media	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing	
PH-2.d – Implement Responsible Beverage Service Programs.	H&HS	Existing budget	High	Ongoing	
PH-2.e – Provide Training and Technical Assistance.	H&HS	Existing budget	High	Ongoing	
PH-2.f – Develop and Maintain Community Partnerships.	H&HS Community Partners	Existing budget	High	Ongoing	
PH-2.g – Provide Tobacco Education.	H&HS Community Partners	Existing budget	High	Ongoing	
PH-2.h – Enhance Youth Tobacco Prevention Programs.	H&HS Education; Community Partners	Existing budget	High	Ongoing	
PH-2.i – Provide Smoking Cessation Programs.	H&HS Community Based Organizations; CBO's	Existing budget	High	Ongoing	
PH-2.j - Enforce Tobacco Control Laws.	H&HS Local Law Enforcement; CDA (EHS)	Existing budget	High	Ongoing	
PH-2.k – Seek Funding for Substance Abuse Prevention.	H&HS	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing	
PH-2.1 – Conduct Regular and Ongoing Evaluations.	H&HS	Existing budget	High	Ongoing	
PH-3.a – Foster Collaboration.	H&HS Community Partners	Existing budget	Medium	Ongoing	
PH-3.b – Support Health Educational Opportunities.	H&HS Community Partners	Existing budget	High	Ongoing	
PH-3.c – Improve Access to Health Care.	H&HS	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing	
PH-3.d – Improve Service Delivery and Utilization.	H&HS Community Partners	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing	
# 

# SOCIOECONOMIC ELEMENT

Program	Responsibility	Potential Funding	Priority	Time Frame
PH-3.e – Support Disease Prevention Education Programs.	H&HS	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
PH-3.f – Employ a Precautionary Approach.	H&HS	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
PH-3.g – Support Mental Health Programs.	H&HS	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
PH-4.a – Co-locate Adult Day Health Care with Senior Housing.	H&HS	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Ongoing
PH-4.b – Improve Service Linkages.	H&HS	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Ongoing
PH-4.c – Expand Senior Services.	CDA	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Med. term
PH-4.d – Advocate for Affordable Assisted Living.	H&HS	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Ongoing
PH-4.e – Improve Balance of Community and Institutional Care.	H&HS	Existing budget	High	Ongoing
PH-4.f - Expand Affordable Housing for Seniors and People with Special Needs.	H&HS	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Ongoing
PH-4.g – Develop Models for Shared In-Home Supportive Services.	H&HS	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
PH-4.h – Expand Home Care Registry.	H&HS	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
PH-4.i – Expand Home Care Services.	H&HS	Existing budget	High	Ongoing



Program	Responsibility	Potential Funding	Priority	Time Frame
PH-4.j – Lobby for Increased Medi-Cal Options.	H&HS	Will require additional grants or revenue <sup>2</sup>	Medium	Immediate
PH-4.k – Provide Prescription Drug Assistance.	H&HS	Existing budget	High	Ongoing
PH-4.1 – Promote Universal Building Designs.	CDA	Existing budget	High	Ongoing
PH-4.m – Identify Potential Responses to Climate Change.	H&HS	Existing budget, grants	Medium	Med. term



## 4.12 Arts and Culture

### Background

Arts and culture are key components of a healthy and interesting place. Arts and cultural activities have the potential to connect people within a community and tap into individual creativity. The intent of this Section is to help make the experience and practice of the arts and cultural opportunities accessible and available to everyone. Toward that end, professional artists need to be supported in their work, and art should be viewed as an integral part of a sustainable community, as well as a necessary part of educational and economic development.

**PR** 

Marin is ripe with opportunities to participate in myriad arts and cultural experiences, including performance, dance, literature, poetry, music, cinema, painting, sculpture, photography, crafts, and much more, as detailed in Fig. 4-32a. The wide variety of arts-based and cultural organizations should be supported and encouraged to cultivate partnerships that promote involvement, education, and diversity through arts and cultural events.

#### Figure 4-32a A Listing of Some Arts-Based and Cultural Organizations in Marin

Cultural/Community Facilities and Programming					
142 Throckmorton Theatre	Headlands Center for the Arts				
Bay Area Discovery Museum	Marin Art and Garden Center				
Bolinas Museum	Marin Center Veterans Auditorium				
California Film Institute	Marin History Museum				
Center Stage at Marin JCC	Marin Museum of the American Indian				
Dance Palace, Point Reyes	O'Hanlon Center for the Arts				
Downtown Art Center/Artworks Downtown	San Anselmo Playhouse				
Falkirk Cultural Center	San Geronimo Valley Cultural Center				
Gallery Route One	Tamalpa Institute				
Hamilton Art Center					
Performing Arts					
Ali Akbar College of Music	Marin Shakespeare Company				
Antenna Theatre	Marin Symphony Association				
Belrose Café	Marin Theatre Company				
Chaucer Theatre	Mountain Play Association				
Marin Ballet	Ross Valley Players				
Marin Center Presents	SingersMarin				
Literary Arts					
Book Passage	Marin Poetry Center				
Marin County Free Library	Marin Writers Center				
Arts Education					
College of Marin	Stapleton School of Performing Arts				
Dominican College	Young Imaginations				
Marin California Poets in the Schools	Youth in Arts				
Marin Dance Theatre					
Major Events					
Italian Street Painting Festival	Marin Symphony				
Marin Art Festival	Mill Valley Fall Arts Festival				
Marin County Fair	Mill Valley Film Festival				
Marin County Open Studios	The Mountain Play				
Marin Shakespeare Festival	Sausalito Art Festival				

#### **Arts Services**

Asian American Alliance of Marin Baulines Craft Guild Bread and Roses DrawBridge Frank Lloyd Wright Civic Center Conservancy Friends of Marin Center Marin Arts Council Marin Center Renaissance Council Marin County Cultural Services Commission Marin Historical Society Marin Society of Artists

### **Key Trends and Issues**

#### Is Marin a center of cultural creativity?

Marin is a home base for much of the music and cultural creativity that spread across the country in the 1960s. This included musicians such as Jerry Garcia, Phil Lesh, and Bob Weir of the Grateful Dead, and Janis Joplin, Carlos Santana, Maria Muldaur, and Bonnie Raitt, as well as cultural icons such as Donald McCoy. Today, the county remains home to many musicians, writers, artists, designers, and cultural icons. The creative legacy of George Lucas, Gary Fisher, Anne Lamott, Isabel Allende, Sim Van Der Ryn, Paul Hawken, Stewart Brand, and many others fosters a longstanding culture of iconoclasm, activism, cutting-edge thinking, and difference from the norm.



Marin County is a center of world-class art and creativity where innovation is highly valued. In a ranking of all counties in the United States, Marin ranks number one in terms of individuals engaged in "creative" work (see Figure 4-32b). In terms of innovation, Marin County produced a higher percentage of patents per capita in 1999 than the state of California or the United States (see Figure 4-32c).

						Share of Total Employment			
County	State	Total Population	Total Employed	Rank Arts Share	Rank Management and Professional Share	Arts, Design, Entertainment, Sports, and Media	Managerial and Professional Except Arts		
Marin	CA	247,289	128,855	1	4	5.73%	46.79%		
Boulder	СО	291,288	162,428	2	5	3.53%	46.65%		
Santa Cruz	CA	255,602	129,380	3	33	3.10%	37.16%		
Santa Barbara	CA	399,347	180,716	4	79	2.88%	32.48%		
Leon	FL	239,452	122,840	5	12	2.88%	42.38%		
Washtenaw	MI	322,895	172,373	6	6	2.72%	45.59%		
Dane	WI	426,526	246,064	7	16	2.69%	40.93%		
Alachua	FL	217,955	105,293	8	15	2.68%	41.32%		
Sarasota	FL	325,957	135,419	9	113	2.63%	29.06%		
Larimer	СО	251,494	136,903	10	36	2.61%	37.03%		
U.S. Average						1.92%	31.73%		

Figure 4-32b Top 10 County Share of Workforce in Creative Occupations, 2000

Source: Economic Competitiveness Group. Don Grimes, University of Michigan, based on U.S. Census occupational data. The arts employment is based upon individuals (residents) identifying themselves as working in "arts, design, entertainment, sports, or media" occupations.

The Marin County Fair also illustrates the high caliber of arts and cultural creativity being fostered in Marin. The Fair has won awards for many years and is considered to be on the cutting edge on many fronts. The Fair has included features ranging from a broad array of art exhibitors, a poetry slam, a world-class multimedia exhibition, and local student exhibitors.

#### How do art and culture in Marin contribute to the economy?

Creativity and innovation are precisely where much of the energy driving the economy comes from (Richard Florida, *The Rise of the Creative Class*, New York: Basic Books, 2002). In Marin County, the mutually reinforcing cycle among creativity, innovation, and wealth creation is well known. The tremendous synergies between the region's values of philanthropy, open space preservation, support for the arts, and creativity result in an astounding degree of wealth creation, which in turn refuels the region's ability to fund charitable causes, preserve open space, and fund the arts (see Figure 4-33). These traits are, and will continue to be, tremendous economic assets for the region. Where Silicon Valley converted technological expertise into marketable products that drive the world's technological infrastructure, Marin County has translated artistic creativity into a world-renowned cluster of arts-based

County	Patents 1999	Population 1999	Ratio
County	Tatentis, 1777		Nacio
Santa Clara	5,664	1,658,000	0.0034
San Mateo	1,153	698,300	0.0017
Santa Cruz	245	251,600	0.0010
Alameda	1,186	1,412,400	0.0008
Westchester, NY	614	937,279	0.0007
Marin	154	243,800	0.0006
San Francisco	393	762,400	0.0005
CA TOTAL	16,776	33,140,000	0.0005
US TOTAL	83,905	267,801,951	0.0003
Los Angeles	23,480	9,330,100	0.0003
Sarasota, FL	85	3,396,255	0.0003
Monterey	61	390,500	0.0002

Figure 4-32c Ratio of U.S. Patents per Person by County, 1999

Source: Economic Competitiveness Group. U.S. Patent & Trademark Office, U.S. Utility Patents Granted 1999, A Technology Assessment & Forecast Report, April 2000 Population data from Global Insight, 2003. This data is the most recent available.

innovation, from digital imaging to mountain bikes to Birkenstock to small, craft-based specialty products. In addition, arts and entertainment are a significant portion of the local economy: A 1997 Marin Arts Council survey showed that the combined annual operating budgets of 34 Marin arts organizations exceeded \$11 million, including more than \$5 million for employment.



#### Is there enough support for and promotion of the arts and culture in Marin?

Support for the arts in Marin is very strong: A survey in 2000 by the Marin Community Foundation found that 22% of county households contributed financially to the arts and humanities, double the rate nationally. A 1997 Marin Arts Council survey found that more than 1,200 performances of over 700 productions were staged annually. However, many artists trained in the county move on to work elsewhere around the country, an indication that local funding efforts could be enhanced. Public funding for the arts has decreased, in part due to a decline in support from the California Arts Council. Marin Community Foundation arts funding has remained relatively steady as a total over the last 10 years but has constituted a smaller percentage of their total discretionary giving over time.

### What Are the Desired Outcomes?

### **GOAL ART-I**



Art and Culture as a Community Resource. Support the arts, local artists, and local arts and cultural organizations.

#### Policies

**ART-1.1** Encourage the Arts and Cultural Events. Promote arts and culture through events and marketing.

**ART-1.2** Support Local Artists and Cultural Creativity. Contribute to a community atmosphere and economic framework that appreciates, supports, and fosters the work of local artists and performers.

**ART-1.3 Expand Arts and Culture Revenue.** Expand and improve revenue streams for performing arts, libraries, and other revenue-generating art and culture venues.

**ART-1.4** Install Public Art. Increase the presence of art in public spaces.

#### Why is this important?

Art enriches the community in many ways. Studies have shown that the presence of cultural organizations in a neighborhood stimulates residents' involvement in other civic activities, thus strengthening the fabric of the community.

**Environment:** Protection of the natural environment provides inspiration for many forms of art and has historically provided the basis for significant cultural artistic expression.

**Economy:** As of 1999, there were more than 2,200 people employed in arts-related industries in Marin. Nationally, of all travelers who included a cultural activity while on a trip in 1998, one-third of those added extra time to their trip to accommodate more cultural activity. This translates to 26.8 million adults adding some 14 million additional travel nights. Revenues from arts and culture will increase with further promotion and support.

**Equity:** Studies of low income neighborhoods show that those with high cultural participation are much more likely to have significantly low delinquency and truancy rates. Arts and cultural activities strengthen communities while providing a highly tangible means of expressing both individuality and pride in the community, for artists and art appreciators alike.

#### How will results be achieved?

#### Implementing Programs

- **ART-1.a** *Promote Arts and Culture.* Work with a wide range of organizations and other interested parties to support partnerships, publicize the arts, provide educational opportunities, and evaluate and eliminate potential barriers to attracting and retaining artists and arts and cultural groups.
- **ART-1.b** *Support Community Efforts.* Work with neighborhood associations and other community organizations to encourage community arts and cultural programs and services.



- **ART-1.c** *Create a Website to Promote Arts and Cultural Programming.* Include a comprehensive, countywide calendar of events, a facility inventory, and links to county arts and cultural resources.
- ART-1.d *Maximize Use of County Facilities for Arts and Cultural Events.* Promote diverse, high-quality arts-related programs and services, including literature, poetry readings, and performing arts in parks, libraries, and other County facilities.
- **ART-1.e** *Survey and Publicize Arts Landmarks.* Inventory significant arts landmarks and reference them in County publications.

- **ART-1.f** *Host Events.* Hold performances and exhibitions by accomplished local and visiting artists, and provide professionally curated exhibition and gallery spaces for high-caliber artists in County facilities.
- **ART-1.g** *Involve Artists in Design.* Include an artist on design teams assembled for planning public projects.
- **ART-1.h** *Foster a Poet Laureate Program.* Work with literary organizations to establish a Marin County Poet Laureate program.
- **ART-1.i** *Find Sites for Public Art.* Encourage placement of art in public spaces, including formulating an inventory of potential sites.
- **ART-1.j** *Support Artist Housing.* Encourage the creation of mixed-use and live/work units for artists in Marin.
- **ART-1.k** *Establish a Public Art Nexus.* Conduct a study to evaluate Public Art Nexus requirements.
- **ART-1.1** *Require Public Art.* Amend the Development Code to require public art (or in-lieu fees) as a component of new development projects.
- **ART-1.m** *Promote Artistic Streetscapes and Signs.* Amend the Development Code to encourage artistic signs and streetscape features in both public and private development projects.



**ART-1.n** *Enhance the Marin Center.* Allocate funds to operate, preserve, and expand access to the arts and cultural services at the Marin Center and its public grounds.

**ART-1.0** *Consider Improved "Open Studios" Standards.* Evaluate potential standards that would support continued "Open Studio" events while minimizing their impact on residential neighborhoods.

#### What Are the Desired Outcomes?

### **GOAL ART-2**

**Participation in the Arts and Cultural Activities.** Encourage and support participation in arts and cultural activities by all populations in the county.

#### Policy

ART-2.1

**Encourage Participation in the Arts and Cultural Events.** Ensure that everyone in Marin has ample opportunity to participate in arts and cultural events.

#### Why is this important?

Among neighborhoods that were considered "blighted" during the 1990s, those with an active cultural scene were more likely to see their poverty rate drop. Cultural creativity has long been a key ingredient to the Marin lifestyle, and opportunities for broad participation in artistic activities can provide added benefits throughout the county.

**Economy:** Arts and cultural activities play a significant role in Marin's economy. Approximately 22% of Marin households contribute to the arts and humanities, compared with 11% nationally. Greater participation in and support of arts and cultural activities can contribute to a stronger local economy.

**Equity:** Many art and cultural activities represent a low-cost opportunity for all community members to participate. Differences in income, cultural background, and even language should not pose a barrier for anyone seeking access to and enjoyment of arts and culture.



Economy

ART-2

Environment

Equity

## <u>₹ЛЛХ ЯЛ Л° №</u> MARIN COUNTYWIDE PLAN

#### How will results be achieved?

#### Implementing Programs

- ART-2.a *Provide Discounts.* Make arts and cultural programming available to all ages, populations, and cultural and socioeconomic backgrounds. Utilize a sliding scale to allow all persons to access County-owned or County-operated facilities, and work with Friends of Marin Center to expand the subsidized voucher program that offers reduced-cost tickets to events for the artists, performers, and elderly, youth, low income, and disabled populations.
- **ART-2.b** *Promote Arts as Recreation.* Work with the Marin County Office of Education to include art classes in community recreation programs.
- **ART-2.c** *Encourage Learning Opportunities in the Arts.* As space is expanded at Marin Center and appropriate facilities, include learning opportunities for arts and cultural development.
- ART-2.d *Encourage Arts Programs for Special Needs Groups.* Encourage integration of arts into programs at senior and assisted-living centers, and facilities for youth, disabled, and lower income persons.
- **ART-2.e** *Expand Arts Education.* Work with schools to attract students to art, music, and cultural classes and programs (also see policies and programs under Goal EDU-2 in the Education Section of this Element).
- **ART-2.f** *Support Cultural Events.* Promote in-depth awareness of the diverse cultures of Marin County. Work with a variety of groups to organize programs that celebrate and promote cultural heritage, customs, and awareness (also see the goals, policies, and programs in the Diversity Section of this Element).
- **ART-2.g** *Provide Public Venues.* Continue to sponsor arts and cultural events at County facilities, and identify strategic public spaces for the promotion of the arts and culture, including public art and community gathering.
- **ART-2.h** *Support Public Art.* Explore stable, long-term funding sources to continue the installation and maintenance of art in public spaces, such as endowments and partnerships with private organizations.
- ART-2.i *Consider Funding for the Arts.* Investigate the feasibility of establishing an arts fund. Make this fund available to support arts and cultural facilities, improvements, and programming.

### What Are the Desired Outcomes?

# **GOAL ART-3**

**Enhance Marin Center.** Encourage and support the Renaissance Partnership in redesigning and rebuilding Marin Center to serve as a cultural focal point in Marin.

### **Policies**

ART-3.1 Improve County Campus. Complete studies and consulting work that lead the County of Marin toward implementing elements of the Master Plan and a renovated campus.
 ART-3.2 Lead Marin Center Renaissance. Lead and support strategic planning efforts of Marin Center Renaissance Partnership.



#### Why is this important?

The Marin Center is a valuable community asset that can be further developed to house a variety of cultural and artistic activities.

**Economy:** In the United States, the arts industry generates \$134 billion annually in economic activity, supports 4.9 million jobs, and returns more than \$24 billion in total government revenue. In addition, 65% of U.S. travelers include cultural events on their trips. A renovated Marin Center campus could draw people from all over the Bay Area for theater, music, and art. Some will dine in Marin restaurants and stay in local hotels. In addition, building renovation and construction will provide jobs for local residents.

**Equity:** Since the county owns and controls Marin Center, activities should be available and accessible to the whole Marin County community.

#### How will results be achieved?

#### **Implementing Programs**

- ART-3.a *Involve Community in the Renaissance Project.* Continue meetings of the Renaissance Partnership Steering Committee to review and champion efforts to move the planning process forward.
- ART-3.b *Keep Supervisors Informed.* Present the new Marin Center Master Plan to the Board of Supervisors for regular updates, as program and design elements are developed in upcoming phases.
- **ART-3.c** *Use an Integrated Approach at Marin Center.* Include an integrated approach to the Marin Center Master Plan, inclusive of a range of "cultural center" facilities, from the library to innovative performance space to high-quality gallery spaces.

<u>\* 熊 佟 熊</u>

#### Figure 4-34 Relationships of Goals to Guiding Principles

This figure illustrates the relationships of each goal in this Section to the Guiding Principles.

Guiding Principles	Link equity, economy, and the environment locally, regionally, and globally.	Minimize the use of finite resources, and use all resources efficiently and effectively.	Reduce the use and minimize the release of hazardous materials.	Reduce greenhouse gas emissions that contribute to global warming.	Preserve our natural assets.	Protect our agricultural assets.	Provide efficient and effective transportation.	Supply housing affordable to the full range of our workforce and diverse community.	Foster businesses that create economic, environmental, and social benefits.	<ol> <li>Educate and prepare our workforce and residents.</li> </ol>	1. Cultivate ethnic, cultural, and socioeconomic diversity.	2. Support public health, public safety, and social justice.
Goals	-	01	0.0	7	1,	Ψ		$\sim$	0,	-	-	_
ART-1 Art and Culture												
as a Community	•								•	•	•	
Resource												
ART-2 Participation in												
the Arts and Cultural									•	•	•	
AK1-3 Enhance Marin Center	•								•		•	

#### How Will Success Be Measured?

#### **Indicator Monitoring**

Nonbinding indicators, benchmarks, and targets<sup>1</sup> will help to measure and evaluate progress. This process will also provide a context in which to consider the need for new or revised implementation measures.

Indicators	Benchmarks	Targets
Number of art exhibits at	12 in 2000.	Increase to 24 by 2010 and 29 in
County facilities.		2020.
Number of artists participating	1,210 artists participated in 2000.	Increase 20% by 2015 and 30%
in the fine arts exhibit at the		by 2020.
Marin County Fair.		

<sup>1</sup>Many factors beyond Marin County government control, including adequate funding and staff resources, may affect the estimated time frames for achieving targets and program implementation.



### **Program Implementation**

The following table summarizes responsibilities, potential funding priorities, and estimated time frames for proposed implementation programs. Program implementation within the estimated time frame.<sup>1</sup> will be dependent upon the availability of adequate funding and staff resources.

Program	Responsibility	Potential Funding	Priority	Time Frame	
ART-1.a – Promote Arts and Culture.	Cultural Services Commission (CSC), MC & CBO's	Existing budget, Private contributions and may require additional grants or revenue. <sup>2</sup>	High	Ongoing	
ART-1.b – Support Community Efforts.	CSC, MC & CBO's	Existing budget, Private contributions and may require additional grants or revenue <sup>2</sup>	High	Ongoing	
ART-1.c – Create a Website to Promote Arts and Cultural Programming.	CSC, MC & CBO's	Private contributions	High	Short term	
ART-1.d – Maximize Use of County Facilities for Arts and Cultural Events.	CSC & MC	Fees	High	Ongoing	
ART-1.e – Survey and Publicize Arts Landmarks.	CSC, MC & CBO's	Volunteers/donated time	Low	Med. term	
ART-1.f - Host Events.	CSC, MC & CBO's	Existing budget, Private contributions and may require additional grants or revenue <sup>2</sup>	High	Ongoing	
ART-1.g – Involve Artists in Design.	1. CAO, CDA 2. DPW	<ol> <li>Existing budget, Private</li> <li>Contributions, and may require</li> <li>additional grants or revenue<sup>2</sup></li> <li>Will require</li> <li>additional grants or revenue<sup>2</sup></li> </ol>	1. Medium 2. Low	1. Ongoing 2. Long term	

#### Figure 4-35 Arts and Culture Program Implementation

<sup>1</sup>Time frames include: Immediate (0-1 years); Short term (1-4 years); Med. term (4-7 years); Long term (over 7 years); and Ongoing (existing programs already in progress whose implementation is expected to continue into the foreseeable future). <sup>2</sup>Completion of this task is dependent on acquiring additional funding. Consequently, funding availability could lengthen or shorten the time frame and ultimate implementation of this program.

# 

# SOCIOECONOMIC ELEMENT

Program	Responsibility	Potential Funding	Priority	Time Frame
ART-1.h – Foster a Poet Laureate Program.	CSC, CBO's	Existing budget, Private contributions and may require additional grants or revenue <sup>2</sup>	Medium	Immediate
ART-1.i – Find Sites for Public Art.	CSC, CDA MC	Existing budget, fees, and may require additional grants or revenue <sup>2</sup>	Medium	Long term
ART-1.j – Support Artist Housing.	CSC, CDA, MC, MCF	Will require additional grants or revenue <sup>2</sup>	Medium	Long term
ART-1.k – Establish a Public Art Nexus.	CSC, CDA	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Med. term
ART-1.1 – Require Public Art.	CDA	Existing budget	Medium	Long-term
ART-1.m – Promote Artistic Streetscapes and Signs.	CDA	Existing budget	Low	Long-term
ART-1.n – Enhance the Marin Center.	МС	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Short term
ART-1.0 – Consider Improved "Open Studio" Standards.	CDA	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Med. term
ART-2.a – Provide Discounts.	Friends of Marin Center with CBO's	Existing budget, Private contributions, and may require additional grants or revenue <sup>2</sup>	High	Ongoing
ART-2.b – Promote Arts as Recreation.	CSC, Parks & Recreation, CBO's	Existing budget	Medium	Ongoing
ART-2.c – Encourage Learning Opportunities in the Arts.	MC, CSC, CBO's	Will require additional grants or revenue <sup>2</sup>	Low	Long term
ART-2.d – Encourage Arts Programs for Special Needs Groups.	CSC, H&HS, CBO's	Existing budget, Private contributions, and may require additional grants or revenue <sup>2</sup>	High	Ongoing





Program	Responsibility	Potential Funding	Priority	Time Frame
ART-2.e – Expand Arts Education.	CSC	Will require grants, Private contributions	Medium	Ongoing
ART-2.f – Support Cultural Events.	CSC, CBO's	Will require grants, Private contributions, fees & sponsorships	High	Ongoing
ART-2.g – Provide Public Venues.	CSC, MC	Existing budget, Private contributions, sponsorships, and may require additional grants or revenue <sup>2</sup>	High	Ongoing
ART-2.h – Support Public Art.	CSC	Will require additional grants or revenue <sup>2</sup>	Low	Long-term
ART-2.i – Consider Funding for the Arts.	CSC	Will require grants, Private contributions, fees & sponsorships	Low	Long-term
ART-3.a – Involve Community in the Renaissance Project.	CAO, MC	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
ART-3.b – Keep Supervisors Informed.	МС	Existing budget	High	Ongoing
ART-3.c – Use an Integrated Approach at Marin Center.	МС	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing



## 4.13 Historical and Archaeological Resources

### Background

Historic preservation is crucial to maintaining Marin's identity, character, and links to the past. For thousands of years, Marin County was home to the Coastal Miwok Indians, who left behind a rich legacy in various archaeological sites throughout the county. This civilization came to an abrupt end when European settlers arrived in the early 1800s. However, the Coastal Miwok still exist as the Federated Indians of Graton Rancheria (FIGR), including the Coast Miwok and Southern Pomo, and they continue to have a rich cultural heritage that includes, among other things, basket making, dances and ceremonies, and a complex and intricate language.

Rural areas in West Marin are defined by historic ranches and small towns. Historic residential and commercial architectural styles in the county are typically Greek Revival, Queen Anne, Italianate commercial, and Bungalow. Inland, a string of small towns along the bay were first developed as vacation spots and later as homes for commuters who worked in San Francisco. Many of these towns, including Fairfax, Larkspur, and Sausalito, have done an excellent job of maintaining their historic character and historic downtowns. Typical historic styles in these area are Shingle Style, Arts and Crafts, Mission Revival, Italianate, and Modern (see Figure 4-36). Well-known architects who built in Marin include Julia Morgan, Bernard Maybeck, Willis Polk, Frank Lloyd Wright, and Joseph Eichler, among others.

Strong importance is placed upon preserving Marin's many historical and archaeological resources (see Map 4-1). These resources deserve effective protection, including thorough requirements for conducting site-specific investigations prior to new development. Historical and archaeological resources also need to be mapped accurately. This section of the Countywide Plan calls for enhancing collaboration among a variety of interested organizations and identifying potential funding opportunities that will help preserve historical and archaeological resources in the county.

<b>Greek Revival:</b> Dates from the 1850s and consists of simple shapes, sharp lines, and doors and windows at regular intervals. Example: Presbyterian Church in Tomales.
<b>Queen Anne:</b> Mid- to late 1800s; it is marked by lots of ornamentation and detailing. Examples: many of the residences in Point Reyes Station.
<b>Italianate:</b> From 1840 to 1880, this style was used primarily for commercial structures on main streets; it typically has a false front with brackets beneath the cornice line. Example: Inverness Post Office.
<b>Shingle Style:</b> Arrived in California in the late 1800s and was characterized by the use of unpainted wooden shingles. Examples: Sausalito Woman's Club, Mill Valley Outdoor Art Club.
<b>Mission Revival:</b> From the early to mid-20th century; this style is defined by wide arches, low-lying roofs, and stucco façades. Examples: Sand Castle Foresters' Hall and the Grandi Building in Point Reyes Station.
<b>Western Stick:</b> Typical in the Bay Area from the late 1890s until the 1920s. Known for its wood detailing, wood shingles, porches, and larger windows, which are necessarily the same size. Examples: residences in Mill Valley, Larkspur, Sausalito, and Fairfax.
<b>California Bungalow:</b> Popular in the 1920s and marked by an open floor plan, front porches, a raised foundation, use of natural materials and attention to detail. Examples: Historic residences in Mill Valley, Larkspur, Sausalito, and Fairfax.
<b>Modern:</b> Originated in the late 1940s to 1950s and used simple lines to truly express the use of materials. Examples: Eichler homes, the Civic Center.

#### Figure 4-36 Historic Architectural Styles of Marin County

#### List of properties in Marin on the National or California Register

- 1 Tomales Presbyterian Church and Cemetery
- 2 Pierce Ranch
- 3 Oldest House North of San Francisco Bay
- 4 Station KPH Marconi Wireless Telegraph Company of America
- 5 Sweetser Mansion
- 6 Stephen Porcella House and Fashion Shop
- 7 Brock Schreiber Boathouse and Beach
- 8 Hamilton Army Air Field Discontiguous Historic District
- 9 Pioneer Paper Mill
- 10 Point Reyes National Seashore Dairy District
- 11 St. Vincent's School for Boys
- 12 Dixie Schoolhouse
- 13 Point Reyes Light Station
- 14 China Camp
- 15 Point Reyes Lifeboat Rescue Station
- 16 Marin County Civic Center
- 17 Olema Lime Kilns
- 18 Home of Lord Charles Snowden Fairfax
- 19 Erskine B. McNear House
- 20 Robert Dollar House
- 21 Bradford House
- 22 Robert Dollar Estate
- 23 Boyd House
- 24 Victor Talking Machine Exhibit
- 25 San Rafael Improvement Club
- 26 Mission San Rafael Arcangel
- 27 Sir Francis Drake Bridge
- 28 Shady Lane Bridge
- 29 Norwood Avenue Bridge
- 30 Glenwood Avenue Bridge
- 31 Lagunitas Bridge
- 32 Green Brae Brick Yard
- 33 Site of the Lighter Wharf at Bolinas
- 34 Larkspur Downtown Historic District
- 35 Dolliver House
- 36 Alexander Acacia Bridge
- 37 Mill Valley Air Force Station Historic District
- 38 Outdoor Art Club
- 39 First Sawmill In Marin County
- 40 Benjamin and Hilarita Lyford House
- 41 Lyford's Stone Tower
- 42 San Francisco and North Pacific Railroad Station House-Depot
- 43 Valentine Rey House
- 44 Angel Island, U.S. Immigration Station
- 45 William G. Barrett House, Casa Madrona Hotel
- 46 Sausalito Central Business Historic District
- 47 Sausalito Woman's Club
- 48 Griswold House
- 49 Forts Baker, Barry and Cronkhite
- 50 Point Bonita Light Station

25 SOURCE: State Office of Historic Preservation



# MAP 4-1 HISTORIC RESOURCES

#### Legend

1			7			f
l		_	_		_	
-	-	_		-		
L						I

County Boundary City Boundary Highways and Major Roads

#### Streams

Perennial
-----------

Intermittent

#### Ephemeral

#### Water Bodies

Lakes
Lagoons

#### Historic Resources

Landm	ark Type	National Register or State Historical Landmark
æ	Church	National
æ	Church	State
h	Federal	National
h	State	State
×	House	National
×	House	State
å	School	National
å	School	State
i	Private	National
		Miles
0 1	2	4 6 8
THIS MAP W THE COUN USE OF TH	VAS DEVELOPED TY OF MARIN IS I IS MAP BEYOND	FOR GENERAL PLAN PURPOSES. IOT RESPONSIBLE OR LIABLE FOR ITS INTENDED PURPOSE.

File: Historic Resources 4-1.mxd

Date: April 16, 2007

### **Key Trends and Issues**

# Are historical and archaeological resources in jeopardy in Marin?

In recent years, Marin County has seen a loss of its historical resources due to demolition and substantial alterations to historic structures. In rural areas this has occurred where large private family compounds replaced historic ranches. In the City-Centered Corridor this has occurred in historic subdivisions where smaller historic homes have been replaced or were substantially altered.

The majority of archaeological sites in the county exist in the rural areas and inland along the bay. However, these resources remain at risk. Marin County contains many historic vistas that are endangered by future development, including rock outcroppings, groves of historic trees, and views along ridgelines. Increased tourism in Marin may also impact historic and archaeological resources if they are not carefully managed.

# X

#### State and Federal Historical Organizations and Agencies

**California Historical Society:** The society provides historic resources, educational programs, and exhibitions to the general public about the history of California.

#### State Historic Preservation Office:

This is a resource providing technical and training assistance both to the County and to owners of historic properties.

**National Historic Register:** This is a federally recognized list of historically significant resources and properties throughout the country.

#### Figure 4-37 Marin Historical and Archaeological Organizations and Repositories

#### Marin Historical Organizations:

- Marin History Museum
- ♦ Sausalito Historical Society
- ♦ Fairfax Historical Society
- Mill Valley Historical Society
- San Anselmo Historical Commission
- Tomales Regional History Center
- Ross Historical Society
- Novato Historical Guild
- Angel Island: Immigrant Journeys of Chinese
   Americans
- ♦ FIGR: Sacred Sites Protection Committee
- Northwest Archaeology Center
- Anne T. Kent California Room
- Tiburon Heritage and Arts Commission

Note: This is a partial listing.



### What Are the Desired Outcomes?

## **GOAL HAR-I**



**Historical Resource Protection.** Identify and protect archaeological and historical resources as major contributors to quality of life and community vitality in Marin.

#### **Policies**

HAR-1.1 Preserve Historical and Archaeological Resources. Identify archaeological and historical resource sites.

**HAR-1.2 Document Historical Information.** Provide documents, photographs, and other historical information whenever possible to be catalogued in the Anne T. Kent California Room in the Marin County Free Library.

- **HAR-1.3** Avoid Impacts to Historical and Archaeological Resources. Ensure that human activity avoids damaging cultural resources, where feasible.
- **HAR-1.4 Participate in Historical Preservation Efforts.** Work with federal, State, and local agencies, and interested individuals, groups, and educational organizations to obtain funding and employ other methods to preserve archaeological and historical sites.
- **HAR-1.5 Regulate Alteration of Historical Buildings.** Limit the ability to modify historical structures, and require development to respect the heritage, context, design, and scale of older structures and neighborhoods.

#### Why is this important?



In addition to honoring those who came before us, State law requires protection of archaeological and historic resources. Preservation is important to maintaining community identity.

**Environment:** Historic building restoration and archaeological site preservation efforts enhance overall environmental quality. When downtowns are revitalized, historic neighborhoods are restored, and buildings are rehabilitated, there is less need to get lumber from forests for new homes, and less pressure to pave over farmland.

**Economy:** Maintaining a community's historical character contributes to economic vitality by making the community more attractive to visitors and residents. Also, adaptive reuse of historical properties for businesses encourages investment. Preserved historic buildings are assets to a community over time, not only because of the services provided within, but because of the unique contribution of their architecture to the look and feel of a community.

**Equity:** In California, owners of historic homes can save between 40% and 60% annually in property taxes (ref: California State Office of Historic Preservation). Owners of historic homes in lower income areas can receive financial assistance to safeguard the historical integrity of these resources.

#### How will results be achieved?

#### **Implementing Programs**

- **HAR-1.a** *Map Resource Areas.* Update the County sensitivity map (not site specific) that identifies potential locations of archaeological resources, and survey and evaluate existing archaeology resources every three years (while maintaining confidentiality regarding the location of archaeological sites). Consult with FIGR as appropriate in map updates.
- **HAR-1.b** *Inventory Historical Resources.* Prepare a comprehensive survey of historic buildings and buildings of architectural significance in compliance with federal and State standards for designating historical resources and buildings.
- **HAR-1.c** *Nullify Outdated Regulations.* Repeal ordinance 1589 (adopted 1967), which has since been superseded by State and federal environmental regulations and County procedures to protect archaeological deposits.
- HAR-1.d *Require Archaeological Surveys for New Development.* Require archaeological surveys conducted on site by a State-qualified archaeologist for new development proposed in areas identified as potential resource locations on the County sensitivity map (see Program HAR-1.a).
- **HAR-1.e** *Require Permanent Protection.* Require development at an archaeological site to, where feasible, avoid the resource and provide permanently deeded open space that incorporates the resource.
- HAR-1.f Involve Appropriate Authorities. Refer development applications that could potentially affect cultural resources to the California Archaeological Inventory, the Northwest Regional Office of the California Historical Resources Information System, and/or Native American representatives, as appropriate.
- **HAR-1.g** *Create a County Historical Commission.* Establish a Historical Preservation Commission (or expand an existing commission) to prepare a cultural resource preservation plan in partnership with the County Historical Society and to review



projects related to historical resources. Include a representative from the FIGR on the Historical Commission.

HAR-1.h Seek Certified Local Government Status. Once a survey of historical and archaeological resources is conducted pursuant to federal standards (Programs HAR-1.a and HAR-1.b) and a County Historical Commission is established (Program HAR-1.g), apply to the State Department of Historical Preservation to become a Certified Local Government that can participate directly in federal and State historical preservation programs.



Native American rock carving

- **HAR-1.i** *Seek Funding to Protect Resource Sites.* Apply to federal, State, and local sources for funds to acquire historic resource sites for parks or other public purposes and to preserve artifacts.
- HAR-1.j *Facilitate Community Development Block Grant Funding.* Assist low income owners of historical homes with obtaining low-interest loans for renovation through the Federal Community Development Block Grant program.

HAR-1.k	<b>Promote Incentives for Restoration.</b> Inform owners of eligible properties about, and encourage them to apply for, local, State, and federal incentives for preservation and restoration of historical and cultural resources, such as the following:
	<ul> <li>County Redevelopment Agency acquisition or bond financing;</li> <li>conservation ("facade") easements or preservation contracts with tax abatement benefits;</li> <li>federal tax credits for restoration work on income producing properties; and</li> <li>State Heritage Fund grants to cities and towns, counties, districts, local agencies, nonprofit organizations, and Native American tribes.</li> </ul>
HAR-1.l	<i>Adopt Preservation Guidelines.</i> Adopt guidelines for preservation of structures of local historical or architectural interest, and historical trees and other landscape elements.
HAR-1.m	<b>Require Design Compatibility.</b> Require projects on sites with or adjacent to cultural resources to complement the appearance of those resources and provide adequate buffers to protect them from potential adverse impacts.
HAR-1.n	<i>Allow Flexibility in Standards for Restoration.</i> Amend the Development Code to allow flexibility in on-site parking and setback provisions to facilitate restoration of historical structures (provided any variance from standards does not conflict with ensuring public health and safety).
HAR-1.o	<b>Promote Adaptive Reuse.</b> Amend the Development Code to allow an appropriate range of reuse options for older buildings, including mixed-use redevelopment.
HAR 1.p	Consultation Regarding Confidentiality of Important Sites. If land designated or

**HAR 1.p** *Consultation Regarding Confidentiality of Important Sites.* If land designated or proposed to be designated as open space contains a historical resource (as defined in Public Resources Code section 21084.1), with cultural significance, the County shall conduct consultations with FIGR. The purpose of the process is to determine the level of confidentiality required to protect the cultural resource and to provide an appropriate level of dignity in any management plan.

### What Are the Desired Outcomes?

# **GOAL HAR-2**

**Community Involvement in Historical Preservation.** Increase public awareness of local history and historical sites, and the need to protect these resources.

### Policy

**HAR-2.1 Encourage Recognition of Significant Sites.** Support efforts by community members, including owners of property with historical significance, to learn about and seek preservation and protection of these resources.



#### Why is this important?

Community involvement is key to maintaining historical resources and properties. Historic preservation has been shown to stimulate community involvement and foster community spirit, while making communities economically viable.

**Environment:** According to a national survey conducted by the American Institute of Architects, more than 20% of Americans rank historic preservation as a priority of utmost importance. Development proposals that could adversely affect archaeological and historical resources should require a higher level of scrutiny.

**Economy:** Forty-six percent of the almost 200 million total U.S. travelers in 1998 included a cultural, arts, heritage, or historical activity while on a trip. Improving and promoting Marin's historical resources and listing these significant properties on the federal or State Historical Register will attract visitors to the area and bring business to bolster the economy during the following year.

**Equity:** Partnering with private sector groups and individuals strengthens the bonds that make Marin a robust community. The pride that community members take in preserving resources contributes to overall quality of life. People benefit from aesthetic improvement to streetscapes and key historical sites.

#### How will results be achieved?

#### Implementing Programs

Nominate Historical Resources. Work with local historical societies to nominate
significant historical resource sites for listing in the federal or State Historical Register,
including buildings more than 50 years old that

- exemplify, embody, or reflect key elements of the county's cultural, social, economic, political, aesthetic, engineering, architectural, or natural history; and/or
- are identified with persons or events significant in local, state, or national history.
- **HAR-2.b** *Partner with Owners of Historical Buildings.* Work with private owners of landmark structures to support rehabilitation of historical buildings.
- **HAR-2.c** *Install Markers and Plaques.* Place historical markers on County roadways and plaques at significant structures to attract and inform the public about important historical sites and events.
- HAR-2.d *Promote Native American Awareness and School Enrichment.* Work with tribal members and the Marin Museum of the American Indian to promote educational programs about Native American history and culture for children, families, and adults, as well as school enrichment and summer camp programs.
- **HAR-2.e** *Support Development of Educational Materials.* Work with local historical societies and other resource agencies to develop educational programs and to prepare and

distribute materials describing local history and specific sites (except as restricted by State guidelines).

- HAR-2.f Support Local History Education and Preservation. Maintain the Anne T. Kent California Room at the Marin Civic Center as a historical information resource, and work with local historical societies to maintain reference libraries of restoration techniques, trades resources, and successful preservation projects.
- HAR-2-g *Preserve Cultural History.* Expand the existing Carla Ehat Oral History Program for the Anne T. Kent California Room to document and create new oral histories from Marin residents.
- **HAR-2.h** *Promote Educational Events.* Participate in preservation-related activities such as National Historical Preservation Week in May.
- HAR-2.i *Implement SB 18 Tribal Consultation Requirements.* In accordance with the new State Law SB 18, require tribal consultation prior to adopting or amending any general plan, community plan, or specific plan.
  - a. Send proposal information to the Native American Heritage Commission (NAHC) and request contact information for tribes with traditional lands or places located within the geographic areas affected by the proposed changes.
  - b. Contact each tribe identified by NAHC in writing and provide them the opportunity to consult about the proposed project.
  - c. Organize a consultation with tribes that respond to the written notice within 90 days.
  - d. Refer proposals to adopt or amend the Countywide Plan, community plan, or specific plans to each tribe included on the NAHC list at least 45 days prior to the proposed action.
  - e. Provide notice of a public hearing at least 10 days in advance to tribes and any other persons who have requested that such notice be provided.

新祭桃

#### Figure 4-38 Relationships of Goals to Guiding Principles

This figure illustrates the relationships of each goal in this Section to the Guiding Principles.

Goals Goals	1. Link equity, economy, and the environment locally, regionally, and globally.	2. Minimize the use of finite resources, and use all resources efficiently and effectively.	3. Reduce the use and minimize the release of hazardous materials.	4. Reduce greenhouse gas emissions that contribute to global warming.	5. Preserve our natural assets.	6. Protect our agricultural assets.	7. Provide efficient and effective transportation.	8. Supply housing affordable to the full range of our workforce and diverse community.	9. Foster businesses that create economic, environmental, and social benefits.	10. Educate and prepare our workforce and residents.	11. Cultivate ethnic, cultural, and socioeconomic diversity.	12. Support public health, public safety, and social justice.
HAR-1 Historical Resource Protection		•									•	
HAR-2 Community Involvement in Historical Preservation		•								•	•	

#### How Will Success Be Measured?

#### **Indicator Monitoring**

Nonbinding indicators, benchmarks, and targets<sup>1</sup> will help to measure and evaluate progress. This process will also provide a context in which to consider the need for new or revised implementation measures.

Indicator	Benchmark	Target
Number of historic properties listed in the State & Federal Register.	41 federal and 13 State sites were listed in 2004.	Continue to increase through 2020.

<sup>1</sup>Many factors beyond Marin County government control, including adequate funding and staff resources, may affect the estimated time frames for achieving targets and program implementation.

### **Program Implementation**

The following table summarizes responsibilities, potential funding priorities, and estimated time frames for proposed implementation programs. Program implementation within the estimated time frame.<sup>1</sup> will be dependent upon the availability of adequate funding and staff resources.

Program	Responsibility	Potential Funding	Priority	Time Frame
HAR-1.a - Map Resource Areas.	CDA, FIGR	Existing budget and may require additional grants or revenue. <sup>2</sup>	Medium	Med. term
HAR-1.b – Inventory Historical Resources.	CDA, FIGR	Existing budget, and may require additional grants or revenue <sup>2</sup>	Low	Long term
HAR-1.c – Nullify Outdated Regulations.	CDA	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Short term
HAR-1.d – Require Archaeological Surveys for New Development.	CDA	Existing budget	High	Immediate
HAR-1.e – Require Permanent Protection.	CDA	Existing budget	High	Immediate
HAR-1.f – Involve Appropriate Authorities.	CDA	Existing budget	High	Ongoing
HAR-1.g - Create a County Historical Commission.	Marin County Historical Society (MCHS), FIGR, other CBO's, BOS	Will require additional grants or revenue <sup>2</sup>	Medium	Long term
HAR-1.h – Seek Certified Local Government Status.	CDA, CBO's, MCHS	Existing budget	Medium	Med. term
HAR-1.i – Seek Funding to Protect Resource Sites.	MCHS, other CBO's	Will require additional grants or revenue <sup>2</sup>	Medium	Long term
HAR-1.j – Facilitate Community Development Block Grant Funding.	CDA	Existing budget	High	Ongoing

Figure 4-39
Historical and Archaeological Resources Program Implementation

<sup>1</sup>Time frames include: Immediate (0-1 years); Short term (1-4 years); Med. term (4-7 years); Long term (over 7 years); and Ongoing (existing programs already in progress whose implementation is expected to continue into the foreseeable future). <sup>2</sup>Completion of this task is dependent on acquiring additional funding. Consequently, funding availability could lengthen or shorten the time frame and ultimate implementation of this program.

4.13-12

Historical and Archaeological Resources

# 

# SOCIOECONOMIC ELEMENT

Program	Responsibility	Potential Funding	Priority	Time Frame
HAR-1.k – Promote Incentives for Restoration.	CDA	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Med. term
HAR-1.1 – Adopt Preservation Guidelines.	CDA	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Long term
HAR-1.m – Require Design Compatibility.	CDA	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Med. term
HAR-1.n – Allow Flexibility in Standards for Restoration.	CDA, DPW	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Long term
HAR-1.0 – Promote Adaptive Reuse.	CDA	Existing budget	Medium	Med. term
HAR-1.p – Consultation Regarding Confidentiality of Important Sites.	Department of Parks and Open Space, CDA	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
HAR-2.a – Nominate Historical Resources.	CBO's, FIGR	Will require Private Donations, Grants and other revenue <sup>2</sup>	Medium	Ongoing
HAR-2.b – Partner with Owners of Historical Buildings.	MCHS, other CBO's, CDA	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Long term
HAR-2.c – Install Markers and Plaques.	MCHS, other CBO's, FIGR	Private Donations and Grants	Medium	Long term
HAR-2.d – Promote Native American Awareness and School Enrichment.	Marin Museum of the American Indian, Schools, Marin Cities and Towns, FIGR	Grants, Fees and Private Donations	High	Ongoing
HAR-2.e – Support Development of Educational Materials.	MCHS, FIGR, other CBO's	Private Donations and Grants	Medium	Ongoing
HAR-2.f – Support Local History Education and Preservation.	Library, California Room	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
HAR-2.g – Preserve Cultural History.	Libraries, CBO's	Private Donations and Grants	Medium	Ongoing



Program	Responsibility	Potential Funding	Priority	Time Frame
HAR-2.h – Promote Educational Events.	MCHS, FIGR, other CBO's	Private Donations and Grants	Medium	Med. term
HAR-2.i – Implement SB 18 Tribal Consultation Requirements.	CDA	Existing budget	High	Ongoing



## 4.14 Parks and Recreation

#### **Background**

Marin County has multiple roles and responsibilities concerning parks and recreation. It is a direct service provider at County-owned parks and coordinates with cities and towns, special districts, and other park agencies to identify community park needs and ways to enhance park and recreation services and facilities. Through the development review process, the County must also ensure that new projects provide adequate recreational opportunities for their residents.

This section of the Countywide Plan focuses on augmenting and improving active recreation facilities in Marin. (See the Open Space Section in the Natural Systems and Agriculture Element for programs regarding lands managed primarily for habitat and scenic values, and lower-impact, passive recreation. Also see the Natural Systems and Agriculture Element, Map 2-17, Marin County Open Space and Parks.) County and city parks in Marin already provide a variety of active recreation amenities, including playing fields, pools, golf courses, tennis and volleyball courts, skate parks, and children's playgrounds. County Service Areas and special districts manage additional park and recreation facilities, as do some school districts.



"In my view, wholesome pleasure, sport, and recreation are as vital to this nation as productive work and should have a large share in the national budget."

- Walt Disney

### **Key Trends and Issues**

# Does Marin have enough developed recreation facilities?

State law allows cities and counties to acquire parkland through dedication or payment of in-lieu fees during subdivision review (Government Code Section 66477, known as the Quimby Act), and to purchase surplus school sites for recreation if public lands in the vicinity are not adequate to meet community outdoor recreation needs (Education Code Sections 17485–17500). The Novato, Las Gallinas, Upper and Lower Ross Valley, and West Marin Planning Areas fall short of the Quimby standard of 3 to 5 acres of parkland per 1,000 residents

(see Figure 4-41). The County is even further from reaching the National Park Association requirement of 10 acres per 1,000 residents. The demand for a wide range of developed facilities for active recreation continues to grow.

	•		
Recreation Facilities	Recommended Number of Units per Population	Existing Number of Units per Marin County Population <sup>1</sup>	Recommended Service Radius
Swimming Pools	1 per 20,000	1 per 17,650	15 to 30 minutes travel time
Golf	1 per 25,000	1 per 49,500	<sup>1</sup> / <sub>2</sub> to 1 hour travel time
Baseball	1 per 5,000	1 per 7,500	<sup>1</sup> / <sub>4</sub> to <sup>1</sup> / <sub>2</sub> mile
Soccer	1 per 20,000	1 per 31,000	1 to 2 miles
Football	1 per 20,000	1 per 31,000	15 to 30 minutes travel time
Basketball	1 per 5,000	1 per 12,350	<sup>1</sup> / <sub>4</sub> to <sup>1</sup> / <sub>2</sub> mile
Tennis	1 court per 2,000	1 per 2,100	<sup>1</sup> / <sub>4</sub> to <sup>1</sup> / <sub>2</sub> mile
Running Track (1/4 mile)	1 per 20.000	1 per 31 000	30 to 60 minutes travel time

#### Figure 4-40 Recreation Standards and Guidelines

Includes facilities open to the public only.

Source: National Recreation and Park Association, 2001.



#### Figure 4-41 Park Acreage by Planning Area (Excluding Schools) Compared with Quimby Act and National Park Association Requirements

Planning Area	Developed <sup>1</sup> Park Acreage	Quimby Act Requirements (5 acres per 1,000 people)	Quimby Act Surplus or Deficit	National Requirements (10 acres per 1,000 people)	National Requirements Surplus or Deficit
1. Novato	153	273	(120)	545	(392)
2. Las Gallinas	60	143	(82)	286	(225)
3. San Rafael Basin	211	200	11	400	(189)
4. Upper Ross Valley	124	128	(4)	256	(132)
5. Lower Ross Valley	74	170	(95)	340	(265)
6. Richardson Bay	262	260	2	521	(285)
7. West Marin	45	62	(17)	123	(78)
Total in Marin County	932	1,236	(304)	2,473	(1,541)

<sup>1</sup>Developed for the purpose of active recreation. Includes city-owned parks.

Source: 2003 Marin County Community Development Agency and 2000 United States Census.

### What Are the Desired Outcomes?

## **GOAL PK-I**

**A High-Quality Parks and Recreation System.** Provide park and recreation facilities and programs to meet the various needs of all county residents.

#### **Policies**

**PK-1.1 Conduct and Coordinate Park Planning.** Develop park and recreation facilities and programs to provide for active recreation, passive enjoyment, and protection of natural resources as a complement to local, state, and national parks and open space in Marin.



- **PK-1.2 Consider User Needs, Impacts, and Costs.** Plan and develop any needed new park and recreation facilities and programs to meet the desires of the community and protect environmental resources.
- **PK-1.3 Protect Park Resources from Impacts of Climate Change.** Identify strategies to protect park resources from the effects of climate change, such as violent weather, plant loss or change due to moisture and temperature changes, and sea level rise.
#### Why is this important?

Several planning areas in Marin County fall short of the Quimby Recreation Act standard of 3 to 5 acres of parkland per 1,000 residents. The County is even further from reaching the National Park Association requirements of 10 acres per 1,000 residents. Parks provide places for active recreation near where people live at little or no cost to park users.

**Environment:** According to a 2002 study conducted by the Responsive Management research firm, 64% of Americans consider it very important to conserve and protect open space. Providing park facilities for active recreation within the community has the potential to reduce demand and conflict in open space and other areas managed for natural resource purposes. In addition, use of sustainable design principles and recycled materials in parks reduces impacts on natural resources.

**Economy:** Recreation throughout the United States is a multi-billion dollar industry. In the United States, sales of outdoor gear, footwear, and other accessories total more than \$18 billion annually. Over the last 15 years, consumer spending on recreation and entertainment has increased from 6.5% to 10.5%. Marin County contributes through a multitude of local sports shops that sell supplies for soccer, baseball, golf, tennis, fishing, and other recreational pursuits. Parks throughout the county are critical in providing access to these activities.

**Equity:** Parks represent a low-cost opportunity for recreational endeavors, allowing all people an equal opportunity to participate. The development of a master plan for parks can address the dynamics of a changing population and recommend appropriate recreational opportunities.

#### How will results be achieved?

#### Implementing Programs

PK-1.a

**Update the Parks Master Plan.** Update the County Parks Master Plan to assess current facilities, determine appropriate locations for new facilities, and explore funding options.

## ĸ

"You cannot save the land apart from the people or the people apart from the land."

– Wendell Berry

**PK-1.b** *Assess User Needs.* Monitor use of County parks, periodically assess user needs for all socioeconomic segments of the population, and contact other local park and recreation providers regarding their user needs.

**PK-1.c** *Support Local Agencies.* Support park and recreation planning efforts by cities and towns, special districts, and other public agencies. Pursue assistance in formulating funding packages and obtaining funding; negotiate with school districts, developers, or other potential recreation providers for recreational opportunities.

SOCIOECONOMIC ELEMENT

- **PK-1.d** *Explore Options with Local Agencies.* Work with cities and towns and schools to determine how their facilities contribute to meeting park and recreation needs in Marin, and to determine which school fields and other recreation facilities may remain available to county residents for use during non-school hours.
- **PK-1.e** *Replace Closed Facilities.* Encourage and join efforts to replace closed facilities that were operated by other local jurisdictions or the private sector.
- **PK-1.f** *Prepare an Acquisition Plan.* For each proposed new park site, complete an acquisition plan that addresses user needs, access modes, alternative sites, environmental impacts, and financing options.
- **PK-1.g** *Prepare an Impact Assessment.* Complete an analysis for each proposed facility that identifies significant resource values and provides a plan for management of vegetation, wildlife, and water resources (if applicable).
- **PK-1.h**Develop Master Plans for Individual<br/>Parks. Prepare a master plan and a<br/>capital improvement plan for each<br/>acquired park area that includes a site<br/>development plan, phasing for<br/>development, estimated cost for each<br/>phase and long-term operation and<br/>maintenance, estimated revenue<br/>generation, and funding sources for<br/>development.
- **PK-1.i** *Conduct a Facilities Inventory.* Conduct a detailed facilities inventory (which would augment the information in Figure 4-41).

"Parks are at the center of a community's character; they reflect and strengthen the sense of place and identity that makes cities fit places for people to live."

ki

- Conservation Foundation, 1972

#### PK-1.j Meet Special Group Needs. Ensure that parks are designed to meet the needs of youths, seniors, and people with disabilities, and annually review special programs targeting those groups, revising them as appropriate.

- **PK-1.k** *Consider Group Camping and Picnic Needs.* Identify areas appropriate for overnight camping by groups with permits from the County, and provide group picnic areas as needed.
- **PK-1.1** *Consider Garden Sites.* Study the feasibility of allowing community gardens in some park areas, and create a garden pilot program.
- **PK-1.m** *Minimize Toxins.* Protect the health of park visitors by utilizing the least toxic means available of reducing weeds and other pests to acceptable levels where appropriate.

- **PK-1.n** *Renovate Fields.* Continue periodic renovations of existing recreational fields as needed.
- **PK-1.0** *Prepare Naming and Sponsorship Guidelines.* Work with interested parties to evaluate the potential benefits and liabilities of accepting funding for naming rights and sponsorship arrangements regarding parks and recreational facilities. Consider prohibiting corporate naming rights that would commercialize public lands and facilities, although continued honorary and memorial naming should be considered as appropriate.
- **PK-1.p** *Explore Use of Artificial Turf.* Explore the use of artificial turf for McInnis Park fields and other high-use fields.
- **PK-1.q** *Renovate Boat Launches.* Renovate the two existing boat launches maintained by the County.
- **PK-1.rConsider Mooring Opportunities.** Evaluate the demand for and feasibility of providing<br/>additional boat mooring locations in areas subject to recreational use, including<br/>Richardson and Tomales bays, and the future reclamation of the San Rafael Rock<br/>Quarry.
- **PK-1.s** *Improve Americans with Disabilities Act (ADA) Accessibility at Parks.* Upgrade bathrooms, stairways, and parking areas to improve the ADA accessibility of park facilities.
- **PK-1.t** *Continue Ongoing Park Maintenance Programs.* Continue ongoing management and maintenance programs to ensure the long-term protection of existing park resources and park infrastructure. Explore opportunities for cost savings and innovation that meet the objectives of protecting Marin County parks.
- **PK-1.u** *Protect Environmentally Sensitive Park Areas.* Protect undeveloped natural park areas such as Tiburon Uplands and Gallinas Creek at McInnis Park, and reassess existing park areas to determine whether they should be re-designated as open space. Where necessary, work with local fire management agencies to reduce fuel loads in an environmentally sensitive manner.
- **PK-1.v** *Prepare Contingency Plans.* Analyze risks to park resources from violent weather, plant and aquatic changes, and sea level rise, and prepare appropriate contingency plans.



#### SOCIOECONOMIC ELEMENT

Planning Area	Facility Name	Facility Type	Total Acres	Restrooms (incl. portables)	Turf Area (developed turf)	Developed Picnic Areas	Trails	Open Space/Natural Areas	Parking (and/or street parking)	Historical Feature	Dogs Off Leash	Playground	Vista Points	Pool
1. Novato	Stafford Lake	R	139.0	х	х	х	х	х	0			х		
	Black Point Boat Launch	BL	2.00	х		х			0					
	Novato Multi-Use Path	Р	5.60				x	x						
2. Las Gallinas	Pueblo Park <sup>1</sup>	М	2.00						✓			x		
	Candy's Park <sup>1</sup>	М	0.1			X			✓			x		
	Castro Park <sup>1</sup>	Ν	1.49		х	X			✓			x		
	Adrian Rosall Park <sup>1</sup>	М	0.70		х	х			✓					
	McInnis Park	R	75.00	x	х	X	х	х	0				x	
	Lagoon Park	С	10.00	x	х	X	х		0	x		x		
	Mission Pass Multi-Use Path	Р	0.46				х	х	✓					
3. San Rafael Basin	McNear's Beach Park	R	55.00	х	х	х	х		0	х			x	х
4. Upper Ross Valley	Deer Park	С	30.00	х		X	х	х	0					
5. Lower Ross Valley	Creekside Park	С	25.65	х	х	х	х	х	✓			х		
	Creekside Multi-Use Path	Р	2.75	x			х	х	✓					
6. Richardson Bay	Mill Valley/Sausalito Path	Р	20.66				х	х	✓					
	Paradise Beach Park	R	19.00	x	х	X	х	х	0	x			x	
	Tiburon Uplands	NP	24.00				х	х	✓			х	x	
7. West Marin	Whitehouse Pool	F	22.00	х			х	х	0				x	
	Chicken Ranch Beach	В	3.00	х			х	х	0				x	
	Agate Beach	В	7.00	х			х	х	0				x	
	Bolinas Park	Ν	1.00	х					✓					
	Forest Knolls Park	Ν	0.60			х			✓			х		
	Village Green <sup>2</sup>	С	2.00		x	х	х		✓					
	Miller Park Boat Launch	BL	6.00	х		х			0				x	
	Upton Beach	В	4.00						✓					
Total Acres			459.02											
Kev: $\checkmark$ = Street <sup>1</sup> F	unded by CSA #18 (Las Gallinas Valley).	$\mathbf{R} = \mathbf{I}$	Regional I	Park:	M =	Min	i Par	k• N	= Ne	ighh	orho	od P	ark•	

#### Figure 4-42 County-Operated Park and Recreation Facility Types

o = Lot

<sup>1</sup> Funded by CSA #18 (Las Gallinas Valley). R = Regional Park; M = Mini Park; N = Neighborhood Park; <sup>2</sup> Funded by CSA #33 (Stinson Beach). C = Community Park; BL = Boat Launch; P = Multi-Use Path; NP = Nature Preserve; F= Fishing Access; B = Beach

新祭桃

#### Figure 4-43 Relationships of Goal to Guiding Principles

This figure illustrates the relationships of this Section's goal to the Guiding Principles.

Goal Guiding Principles	1. Link equity, economy, and the environment locally, regionally, and globally.	2. Minimize the use of finite resources, and use all resources efficiently and effectively.	3. Reduce the use and minimize the release of hazardous materials.	4. Reduce greenhouse gas emissions that contribute to global warming.	5. Preserve our natural assets.	6. Protect our agricultural assets.	7. Provide efficient and effective transportation.	8. Supply housing affordable to the full range of our workforce and diverse community.	9. Foster businesses that create economic, environmental, and social benefits.	10. Educate and prepare our workforce and residents.	11. Cultivate ethnic, cultural, and socioeconomic diversity.	12. Support public health, public safety, and social justice.
PK-1 A High-Quality Parks and Recreation System	•		٠		•							•

# SOCIOECONOMIC ELEMENT

#### How Will Success Be Measured?

#### **Indicator Monitoring**

Nonbinding indicators, benchmarks, and targets<sup>1</sup> will help to measure and evaluate progress. This process will also provide a context in which to consider the need for new or revised implementation measures.

Indicators	Benchmarks	Targets
Number of neighborhood,	459 acres in 2000.	Acquire 40 acres by 2010 and
community, and regional parks		develop 10 acres; acquire an
in County government		additional 40 acres by 2015 and
jurisdiction in acres per capita.		develop 20 acres.

<sup>1</sup>Many factors beyond Marin County government control, including adequate funding and staff resources, may affect the estimated time frames for achieving targets and program implementation.

#### **Program Implementation**

The following table summarizes responsibilities, potential funding priorities, and estimated time frames for proposed implementation programs. Program implementation within the estimated time frame.<sup>1</sup> will be dependent upon the availability of adequate funding and staff resources.

Program	Responsibility	Potential Funding	Priority	Time Frame
PK-1.a - Update the Parks Master Plan.	County Parks	Will require additional grants or revenue. <sup>2</sup>	High	Short term
PK-1.b - Assess User Needs.	County Parks	Will require additional grants or revenue <sup>2</sup>	High	Short term
PK-1.c – Support Local Agencies.	County Parks	Existing budget	Low	Med. term
PK-1.d – Explore Options with Local Agencies.	County Parks	Existing budget	Medium	Med. term
PK-1.e – Replace Closed Facilities.	County Parks	Will require additional grants or revenue <sup>2</sup>	Low	Long term
PK-1.f – Prepare an Acquisition Plan.	County Parks	Existing budget	Medium	Med. term
PK-1.g – Prepare an Impact Assessment.	County Parks	Existing budget	Medium	Med. term
PK-1.h – Develop Master Plans for Individual Parks.	County Parks	Will require additional grants or revenue <sup>2</sup>	High	Med. term
PK-1.i – Conduct a Facilities Inventory.	County Parks	Will require additional grants or revenue <sup>2</sup>	High	Short term
PK-1.j - Meet Special Group Needs.	County Parks	Existing budget and may require additional grants or revenue <sup>2</sup>	Medium	Short term
PK-1.k – Consider Group Camping and Picnic Needs.	County Parks	Will require additional grants or revenue <sup>2</sup>	Medium	Med. term

#### Figure 4-44 Parks and Recreation Program Implementation

<sup>&</sup>lt;sup>1</sup>Time frames include: Immediate (0-1 years); Short term (1-4 years); Med. term (4-7 years); Long term (over 7 years); and Ongoing (existing programs already in progress whose implementation is expected to continue into the foreseeable future). <sup>2</sup>Completion of this task is dependent on acquiring additional funding. Consequently, funding availability could lengthen or shorten the time frame and ultimate implementation of this program.

# 

## SOCIOECONOMIC ELEMENT

Program	Responsibility	Potential Funding	Priority	Time Frame
PK-1.1 – Consider Garden Sites.	County Parks	Existing budget	Medium	Med. term
PD-1.m – Minimize Toxins.	County Parks	Existing budget	High	Ongoing
PK-1.n - Renovate Fields.	County Parks	Existing budget	High	Ongoing
PK-1.0 –Prepare Naming and Sponsorship Guidelines.	County Parks	Existing budget	Medium	Med. term
<b>PK-1.p – Explore the Use of Artificial Turf.</b>	County Parks	Will require private donations, Bonds, Grants	High	Ongoing
PK-1.q – Renovate Boat Launches.	County Parks	Will require additional grants or revenue <sup>2</sup>	High	Med. term
PK-1.r - Consider Mooring Opportunities.	RBRA & State of California	Existing budget and may require additional grants or revenue <sup>2</sup>	Low	Long term
PK-1.s – Improve ADA Accessibility.	County Parks	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
PK-1.t – Continue Ongoing Park Maintenance Programs.	Department of Parks and Open Space	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Ongoing
PK-1.u – Protect Environmentally Sensitive Park Areas.	Department of Parks and Open Space	Existing budget and may require additional grants or revenue <sup>2</sup>	High	Med. term
PK-1.v – Prepare Contingency Plans.	Department of Parks and Open Space	Will require additional grants or revenue <sup>2</sup>	Medium	Long term



#### **TABLE OF FIGURES**

Figure 1-1	Global Oil Production 1930-2050: Maximum, Minimum, and Mean-Case Scenarios	1.3-5
Figure 1-2	Countywide Greenhouse Gas Emissions, 2005	1.3-6
Figure 1-3	Humanity's Ecological Footprint	1.3-7
Figure 1-4	Ecological Footprint Comparison	1.3-8
Figure 1-5	Ecological Footprint of Marin County, 2004	1.3-8
Figure 1-6	Footprint Land Requirements	
Figure 1-7	Number of Earths Required If the World Population Footprint Equaled a Bay Area County	1.3-10
Figure 1-8	Marin Agricultural Land Trust Easements	1.5-1

Figure 2-1	Typical Cross-Sections of Wetland Conservation Areas	2.4-18
Figure 2-2	Typical Cross-Section of a Stream Conservation Zone	2.4-25
Figure 2-3	Relationships of Goals to Guiding Principles	2.4-40
Figure 2-4	Biological Resources Program Implementation	2.4-42
Figure 2-5	Relationships of Goals to Guiding Principles	2.5-12
Figure 2-6	Water Resources Program Implementation	2.5-14
Figure 2-7	Relationships of Goals to Guiding Principles	2.6-16
Figure 2-8	Environmental Hazards Program Implementation	2.6-18
Figure 2-9	Summary of Measured Air Quality Exceedances	2.7-3
Figure 2-10	0 Global Temperature	2.7-4
Figure 2-1	1 Countywide Emissions Analysis	2.7-5
Figure 2-12	2 Cost to Society of Insurable, Weather-Related Damages from 1950 through 1	19992.7-6
Figure 2-1a	<sup>3</sup> California and National Ambient Air Quality Standards	2.7-8
Figure 2-14	4 Projects with Potentially Significant Emissions	2.7-9
Figure 2-13	5 Relationships of Goals to Guiding Principles	2.7-19
Figure 2-16	6 Atmosphere and Climate Program Implementation	2.7-21
Figure 2-17	7 Mission Statements of Key Public Land Managers and Land Conservation	
	Organizations in Marin	2.8-3
Figure 2-18	8 Relationships of Goals to Guiding Principles	2.8-14
Figure 2-19	9 Open Space Program Implementation	2.8-16
Figure 2-20	0 Miles of Trails in Marin County by Managing Agency	2.9-2



Figure 2-21	Relationships of Goals to Guiding Principles	
Figure 2-22	Trails Program Implementation	
Figure 2-23	Marin County Agricultural Value by Commodity Category, 1942-2002.	
Figure 2-24	Status of Lands in Agricultural Use in Marin County	
Figure 2-25	Milk Production in Hundreds of Pounds, 1942 through 2002	
Figure 2-26	Vegetable Acreages 1935–1967	
Figure 2-27	Fruit, Nut, and Vegetable Acreages 1974-2003	
Figure 2-28	Acreage and Value of Aquaculture Products, 1990–2003	
Figure 2-29	Relationships of Goals to Guiding Principles	
Figure 2-30	Agriculture and Food Program Implementation	

Figure 3-1a Land Use and Demographic Data for Marin County	3.4-4
Figure 3-1b Marin County Demographics	3.4-5
Figure 3-2a Typical Cross Section Through Marin County Showing Environmental Transect	3.4-9
Figure 3-2b Transect Showing Ranges of Residential Density for Countywide Plan Land Use	
Categories	3.4-31
Figure 3-3 HOD Unit Allocations by Traffic Impact Areas	3.4-11
Figure 3-4a General Commercial/Mixed Use	3.4-36
Figure 3-4b Office Commercial/Mixed Use	3.4-37
Figure 3-4c Neighborhood Commercial/Mixed Use	3.4-38
Figure 3-4d Recreational Commercial	3.4-38
Figure 3-4e Industrial	3.4-39
Figure 3-5 Relationships of Goals to Guiding Principles	3.4-43
Figure 3-6 Community Development Program Implementation	3.4-45
Figure 3-7 Strawberry Shopping Center Photo Simulation	3.5-8
Figure 3-8 Marinwood Shopping Center Photo Simulation	3.5-16
Figure 3-9 Relationships of Goals to Guiding Principles	3.5-17
Figure 3-10 Design Program Implementation	3.5-19
Figure 3-11 PG&E Electricity Sources, 2005	3.6-2
Figure 3-12 Marin Electricity Use, 2000	
Figure 3-13 Marin Natural Gas Use, 2000	
Figure 3-14 Marin Residential Electricity Consumption, 1994–2001	3.6-5
Figure 3-15 Marin Nonresidential Electricity Consumption, 1994-2001	3.6-5
Figure 3-16 Renewable Energy	3.6-12
Figure 3-17 LEED Rating System	3.6-16
Figure 3-18 Relationships of Goals to Guiding Principles	3.6-18



Figure 3-19	Energy Program Implementation
Figure 3-20	Relationships of Goals to Guiding Principles
Figure 3-21	Mineral Resources Program Implementation
Figure 3-22	Figure deleted
Figure 3-23	Figure deleted
Figure 3-24	Figure deleted
Figure 3-25	Figure deleted
Figure 3-26	Figure deleted
Figure 3-27	Figure deleted
Figure 3-28	Figure deleted
Figure 3-29	Total Average Daily Trips per Household in Marin
Figure 3-30	Total Daily Weekday Trips Generated in Marin County
Figure 3-31	Destination of Work Trips for Marin County Residents
Figure 3-32	Composition of All Vehicle Trips on the Marin Roadway Network (Traffic–A.M.
<b>E</b> :	Feak flow)       0.9-7         Intersection Lowal of Service Definitional Simulized Intersections       2.0.8
Figure 3-33	Intersection Level of Service Definitions: Signalized Intersections
Figure 3-34	Beedway Service Definitions: Stop Sign Controlled
Figure 3-33	Koadway Segment Level of Service Definitions
Figure 3-30	Monitored Roadway Locations for Weekday, P.M. Feak Level of Service
Figure 3-37	Relationships of Goals to Guiding Principles
Figure $3-38$	Transportation Program Implementation
Figure 3-39	Relationship Between Noise Levels and Annoyance (Schultz Curve)
Figure $3-40$	Typical Noise Levels
Figure $3-41$	Acceptable Noise Levels
Figure 3-42	Roadway Noise Comparison, 1987 and 2001
Figure 3-43	Benchmarks for Allowable Noise Exposure from Stationary Noise Sources
Figure 3-44	Relationships of Goals to Guiding Principles
Figure 3-45	Noise Program Implementation
Figure 3-46	Urban Service Area Concept
Figure 3-47	Solid Waste Diversion
Figure 3-48	Residential Disposal (Pounds per Resident per Day)
Figure 3-49	Relationships of Goals to Guiding Principles
Figure 3-50	Public Facilities and Services Program Implementation
Figure 3-51	Land Use and Demographic Data for the Novato Planning Area (PA #1)
Figure 3-52	Land Use and Demographic Data for the Las Gallinas Planning Area (PA #2)
Figure 3-53	Land Use and Demographic Data for the San Rafael Basin Planning Area
	(PA #3)



Figure 3-54	Land Use and Demographic Data for the Upper Ross Valley Planning Area (PA #4)	3.12-17
Figure 3-55	Land Use and Demographic Data for the Lower Ross Valley Planning Area (PA #5).	3.12-19
Figure 3-56	Land Use and Demographic Data for the Richardson Bay Planning Area (PA #6)	3.12-22
Figure 3-57	Land Use and Demographic Data for the West Marin Planning Area (PA #7).	3.12-25
Figure 3-58	Relationships of Goals to Guiding Principles	

Figure 4-1	Business Building Blocks	4.4-3
Figure 4-2	2 Existing and Targeted Businesses	4.4-4
Figure 4-3	3 Targeted Industries Screening Criteria	4.4-6
Figure 4-4	Marin Target Industries	4.4-7
Figure 4-5	5 Relationships of Goals to Guiding Principles	4.4-11
Figure 4-6	Economy Program Implementation	4.4-13
Figure 4-7	<sup>7</sup> Licensed Child Care Supply and Demand in Marin County, 2002	4.5-2
Figure 4-8	Child Day-Care Chart	4.5-3
Figure 4-9	Relationships of Goals to Guiding Principles	4.5-7
Figure 4-1	0 Child Care Program Implementation	4.5-9
Figure 4-1	1 Relationships of Goals to Guiding Principles	4.6-11
Figure 4-1	2 Public Safety Program Implementation	4.6-13
Figure 4-1	3 Relationships of Goals to Guiding Principles	
Figure 4-1	4 Community Participation Program Implementation	4.7-7
Figure 4-1	5 Racial Minority Populations in Marin	4.8-2
Figure 4-1	6 Relationships of Goal to Guiding Principles	4.8-6
Figure 4-1	7 Diversity Program Implementation	4.8-8
Figure 4-1	8 Relationships of Goals to Guiding Principles	4.9-7
Figure 4-1	9 Education Program Implementation	4.9-9
Figure 4-2	20 Relationships of Goals to Guiding Principles	4.10-5
Figure 4-2	21 Environmental Justice Program Implementation	4.10-7
Figure 4-2	22 Determinants of Health, 2002 (for U.S. Population)	4.11-2
Figure 4-2	23 Components for a Comprehensive Community Health Effort	4.11-2
Figure 4-2	24 Leading Causes of Death in Marin County, 2000	<b>4.11-</b> 3
Figure 4-2	25 Factors Underlying the Leading Causes of Death in the United States, 2000	4.11-4
Figure 4-2	26 Breast Cancer Incidence Trends in Marin County	4.11-4
Figure 4-2	27 Overweight Population in Marin	4.11-5



Figure 4-28 Alcohol – Use, Marin Teens, Adults, and Seniors	4.11-6
Figure 4-29 Countywide Plan Policies and Programs Related to the Senior Population	4.11-19
Figure 4-30 Relationships of Goals to Guiding Principles	4.11-21
Figure 4-31 Public Health Program Implementation	<b>4.11-2</b> 3
Figure 4-32a A Listing of Some Arts-Based and Cultural Organizations in Marin	4.12-2
Figure 4-32b Top 10 County Share of Workforce in Creative Occupations, 2000	4.12-4
Figure 4-32c Ratio of U.S. Patents per Person by County, 1999	4.12-5
Figure 4-33 Virtuous Cycle	4.12-5
Figure 4-34 Relationships of Goals to Guiding Principles	4.12-12
Figure 4-35 Arts and Culture Program Implementation	4.12-14
Figure 4-36 Historic Architectural Styles of Marin County	4.13-2
Figure 4-37 Marin Historical and Archaeological Organizations and Repositories	<b>4.13-</b> 3
Figure 4-38 Relationships of Goals to Guiding Principles	4.13-10
Figure 4-39 Historical and Archaeological Resources Program Implementation	4.13-12
Figure 4-40 Recreation Standards and Guidelines	4.14-2
Figure 4-41 Park Acreage by Planning Area (Excluding Schools) Compared with Quimby A and National Park Association Requirements	ct 4.14–3
Figure 4-42 County-Operated Park and Recreation Facility Types	4.14-7
Figure 4-43 Relationships of Goal to Guiding Principles	4.14-8
Figure 4-44 Parks and Recreation Program Implementation	4.14-10

Figure 5-1	Special-Status Animal Species Known From Marin County	5-13
Figure 5-2	Special-Status Plant Species Known From Marin County	5-16

## \* \* \* \* \*



#### TABLE OF MAPS

Map 1-1	County of Marinfollowing page	1.1-2
<b>M</b> ap 1-2	Environmental Corridors	1.1-2
M 0.1	No. 10 CH - 1	04.0
Map 2-1	Vegetation	2.4-2
Map 2-2	Special-Status Species and Sensitive Natural Communities <i>following page</i>	2.4-2
Map 2-3	Wetlands/Streams	2.4-4
Map 2-4	Watersheds with Streams and Observed Steelhead Trout and Coho Salmon following page	2.4-4
Man 2-5a	Baylands Corridor following page	2.4-6
Map 2-5h	Baylands Corridor following page	2.4-6
Map 2-6	Sudden Oak Death following page	24-6
Map 2-7	Major Watersheds following page	2.5-2
Map 2-8	Parcels with Buildings and Septic Systems	2.5-2
Map 2-9	Seismic Shaking Amplification Hazards	2.6-4
Map 2-10	Fault Hazards	2.6-4
Map 2-11	Liquefaction Susceptibility Hazards	2.6-4
Map 2-12	Flooding	2.6-4
Map 2-13	Urban-Wildland Interface Zone	2.6-4
<b>M</b> ap 2-14	State Responsibility Areas (SRA's) For Fire Protectionfollowing page	2.6-4
<b>M</b> ap 2-15	Fire Risk	2.6-4
<b>M</b> ap 2-16	Sensitive Receptor Sites in Unincorporated Marin Countyfollowing page	2.7-10
Map 2-17	Open Space and Parks	2.8-2
Map 2-18	Coastal, Ridge and Bay Trails	2.9-2
Map 2-19a	a Marin Countywide Trails Planfollowing page	2.9-2
Map 2-19	D Marin Countywide Trails Planfollowing page	2.9-2
Map 2-196	Marin Countywide Trails Plan	2.9-2
Map 2-196	d Marin Countywide Trails Planfollowing page	2.9-2
Map 2-196	e Marin Countywide Trails Planfollowing page	2.9-2
Map 2-19f	Marin Countywide Trails Plan	2.9-2
Map 2-19§	g Marin Countywide Trails Planfollowing page	2.9-2
Map 2-191	n Marin Countywide Trails Planfollowing page	2.9-2
<b>M</b> ap 2-19i	Marin Countywide Trails Planfollowing page	2.9-2
Map 2-19j	Marin Countywide Trails Planfollowing page	2.9-2

\* \* \* \* \*

Map 2-20	Protected Agricultural	Lands	following page	2.10-6
----------	------------------------	-------	----------------	--------

<b>M</b> ap 3-1a	Environmental Features Focusing Development within	
	the City-Centered Corridor following page	3.4-2
<b>M</b> ap 3-1b	Environmental Features Focusing Development within	94.0
<b>M</b> 0.0	the City-Centered Corridor	3.4-2
Map 3-2a	Housing Overlay Designation	3.4-14
Map 3-2b	Housing Overlay Designation	3.4-14
<b>M</b> ap 3-2c	Traffic Impact Areas as Determined by Screenlines following page	3.4-14
<b>M</b> ap 3 <b>-</b> 3	Community Plan Areas	3.4-20
<b>M</b> ap 3 <b>-</b> 4	Ridge and Upland Greenbelt Areas following page	3.5-14
<b>M</b> ap 3 <b>-</b> 5	Location of Mineral Resource Preservation Sites	3.7-2
<b>M</b> ap 3-6a	Proposed Transportation Improvements	3.9-4
<b>M</b> ap 3-6b	Proposed Transportation Improvements	3.9-4
<b>M</b> ap 3 <b>-</b> 7	Monitored Roadway Locations For Level Of Service	3.9-8
<b>M</b> ap 3 <b>-</b> 8	Roadway Network of Marin County following page	3.9-12
<b>M</b> ap 3-9a	Bikeways of Marin County	3.9-20
<b>M</b> ap 3-9b	Bikeways of Marin County	3.9-20
<b>M</b> ap 3-9c	Bikeways of Marin County	3.9-20
<b>M</b> ap 3-10a	a Transit Corridors of Marin County	3.9-22
Map 3-10	o Transit Corridors of Marin County	3.9-22
Map 3-10	e Transit Corridors of Marin County	3.9-22
Map 3-11	Location of Long Term Noise Measurements	3.10-4
Map 3-12	Existing and Proposed Noise Contours	3.10-4
Map 3-13	Existing Noise Contours for the Airport at Gnoss Field	3.10-7
Map 3-14	Projected Future Noise Contours for the Airport at Gnoss Field	3.10-8
Map 3-15	Existing Noise Contours for the Richardson Bay Heliport	3.10-9
Map 3-16	San Rafael Airport Noise Contours on pg	3.10-10
Map 3-17	Index to Community Facilities and Sphere of Influence Maps	3.11-2
Map 3-18a	a Coastal Community Facilities (Northern)	3.11-2
Map 3-18	Coastal Community Facilities (Southern)	3.11-2
Map 3-19	Inland Rural Corridor Community Facilities following page	311-2
Map 3-90	Novato Community Facilities and Sphere of Influence following page	3 11_9
Map 3-91	San Anselmo Community Facilities and Sphere of Influence following page	3.11_9
Map 2 99	San Parseino Community Facilities and Sphere of Influence	311 9
Map 2.99	Fairfay Community Facilities and Sphere of Influence	2 11 9
Map 9.94	Pass Community Facilities and Sphere of Influence	2 11 0
wap 3-24	Noss Community Facilities and Sphere of Innuence	0.11 - 2



Map 3-25 Larkspur	r Community Facilities and Sphere of Influencefollowing page	3.11-2
Map 3-26 Corte Ma	adera Community Facilities and Sphere of Influence <i>following page</i>	3.11-2
Map 3-27 Mill Vall	ey Community Facilities and Sphere of Influencefollowing page	3.11-2
Map 3-28 Tiburon	Community Facilities and Sphere of Influencefollowing page	3.11-2
Map 3-29 Belveder	re Community Facilities and Sphere of Influencefollowing page	3.11-2
Map 3-30 Sausalito	Community Facilities and Sphere of Influence	3.11-2
Map 3-31 Marin Co	ounty Fire Agenciesfollowing page	3.11-2
Map 3-32 Marin Co	ounty School Districtsfollowing page	3.11-2
Map 3-33 Planning	Areas	3.12-2
Map 3-34 Land Us	e Policy Map Setfollowing page	3.12-30
<b>M</b> ap 0.0	Marin County Land Use Policy Map Indexfollowing page	3.12-30
<b>M</b> ap 1.0	Planning Area 1 (Novato) Land Use Map Indexfollowing page	3.12-30
<b>M</b> ap 1.1a	North Novato Land Use Policy Mapfollowing page	3.12-30
<b>M</b> ap 1.1b	North Novato Land Use Policy Mapfollowing page	3.12-30
<b>M</b> ap 1.2	West Novato Land Use Policy Mapfollowing page	3.12-30
<b>M</b> ap 1.3	Indian Valley Land Use Policy Mapfollowing page	3.12-30
$\operatorname{Map} 1.4$	Southwest Novato Land Use Policy Mapfollowing page	3.12-30
$\mathbf{Map}\ 1.5$	Black Point Land Use Policy Mapfollowing page	3.12-30
<b>M</b> ap 1.6	Bel Marin Keys Land Use Policy Mapfollowing page	3.12-30
<b>M</b> ap 1.7	Loma Verde Land Use Policy Mapfollowing page	3.12-30
<b>M</b> ap 2.0	Planning Area 2 (Las Gallinas Valley) Land Use	
	Map Index	3.12-30
<b>M</b> ap 2.1	Lucas Valley Environs Land Use Policy Mapfollowing page	3.12-30
Map 2.2	Lucas Valley Land Use Policy Mapfollowing page	3.12-30
<b>M</b> ap 2.3	Marinwood Land Use Policy Mapfollowing page	3.12-30
Map 2.4	St. Vincent's and Silveira Land Use Policy Mapfollowing page	3.12-30
<b>M</b> ap 2.5.0	Santa Venetia Land Use Policy Map Indexfollowing page	3.12-30
Map 2.5.1	Santa Venetia Land Use Policy Map (Map 1 of 2)following page	3.12-30
Map 2.5.2	Santa Venetia Land Use Policy Map (Map 2 of 2)following page	3.12-30
<b>M</b> ap 2.6	Los Ranchitos Land Use Policy Mapfollowing page	3.12-30
<b>M</b> ap 3.0	Planning Area 3 (San Rafael Basin) Land Use Map Index	3.12-30
<b>M</b> ap 3.1	Upper Sun Valley Land Use Policy Mapfollowing page	3.12-30
<b>M</b> ap 3.2	Point San Pedro Land Policy Mapfollowing page	3.12-30
<b>M</b> ap 3.3	Bayside Acres and Country Club Land Use Policy Mapfollowing page	3.12-30
<b>M</b> ap 3.4	California Park Land Use Policy Mapfollowing page	3.12-30



<b>M</b> ap 4.0	Planning Area 4 (Upper Ross Valley) Land Use	
	Map Index	
Map 4.1	Sleepy Hollow Land Use Policy Map following page 3.12-30	
Map 4.2	West Fairfax Land Use Policy Map following page 3.12-30	
<b>M</b> ap 4.3	Southwest of Fairfax Land Use Policy Map following page 3.12-30	
<b>M</b> ap 5.0	Planning Area 5 (Lower Ross Valley)	
<b>M</b> ap 5.1.0	Kentfield Land Use Policy Map Key	
Map 5.1.1	Kentfield Land Use Policy Map (Map 1 of 2)following page 3.12-30	
Map 5.1.2	Kentfield Land Use Policy Map (Map 2 of 2)following page 3.12-30	
Map 5.2	Lucky Drive Area/Greenbrae Boardwalk Land Use Policy	
	Map Key	
<b>M</b> ap 5.3	San Quentin Land Use Policy Map following page 3.12-30	
<b>M</b> ap 6.0	Planning Area 6 (Richardson Bay) Land Use Map Index following page 3.12-30	
<b>M</b> ap 6.1.0	Tamalpais Area Land Use Policy Map Index following page 3.12-30	
Map 6.1.1	Tamalpais Area Land Use Policy Map, Muir Woods Park (Map 1 of 5) following page 3 12-30	
Map 6.1.2	Tamalnais Area Land Use Policy Map. Homestead Valley	
101up 0.1.2	(Map 2 of 5)	
Map 6.1.3a	Tamalpais Area Land Use Policy Map (Map 3 of 5) following page 3.12-30	
Map 6.1.3b	Tamalpais Area Land Use Policy Map (Map 4 of 5) following page 3.12-30	
Map 6.1.4	Tamalpais Area Land Use Policy Map, Almonte	
	(Map 5 of 5) following page 3.12-30	
Map 6.2	Marin City Land Use Policy Mapfollowing page 3.12-30	
<b>M</b> ap 6.3.0	Strawberry Land Use Map Index following page 3.12-30	
Map 6.3.1	North Strawberry & Alto Land Use Policy Map	
	(Map 1 of 2) <i>following page</i> 3.12-30	
<b>M</b> ap 6.3.2	South Strawberry Land Use Policy Map (Map 2 of 2) following page 3.12-30	
<b>M</b> ap 6.4	Waldo Point Land Use Policy Map following page 3.12-30	
<b>M</b> ap 6.5	Tiburon Peninsula Land Use Policy Map following page 3.12-30	
<b>M</b> ap 7.0	Planning Area 7 (West Marin) Land Use Policy Map Indexfollowing page 3.12-30	
Map 7.1	Dillon Beach Land Use Policy Map	
<b>M</b> ap 7.2	Tomales Land Use Policy Map	
<b>M</b> ap 7.3.0	East Shore Land Use Policy Map Key following page 3.12-30	
Map 7.3.1	East Shore Land Use Policy Map (Map 1 of 2) following page 3.12-30	
Map 7.3.2	East Shore Land Use Policy Map (Map 2 of 2) following page 3.12-30	
Map 7.4.1	Northwest Marin County Land Use Policy Map	
	(Map 1 of 2) following page 3.12-30	



Ν	Лар 7.4.2	Northwest Marin County Land Use Policy Map		
		(Map 2 of 2)	following page.	3.12-30
Ν	Aap 7.5	Point Reyes Station Land Use Policy Map	following page.	3.12-30
Ν	Лар 7.6	Inverness Land Use Policy Map	following page.	3.12-30
Ν	Лар 7.7	Olema Land Use Policy Map	following page	3.12-30
Ν	Лар 7.8	Southwest Marin County Land Use Policy Map	following page.	3.12-30
Ν	<b>A</b> ap 7.9	Nicasio Land Use Policy Map	following page.	3.12-30
Ν	<b>A</b> ap 7.10.0	San Geronimo Valley Land Use Policy Map (Map 1 of 5)	following page.	3.12-30
Ν	Map 7.10.1	Woodacre Land Use Policy Map (Map 2 of 5)	following page.	3.12-30
Ν	<b>A</b> ap 7.10.2	San Geronimo Land Use Policy Map (Map 3 of 5)	.following page	3.12-30
Ν	<b>A</b> ap 7.10.3	Forest Knolls Land Use Policy Map (Map 4 of 5)	following page.	3.12-30
Ν	Map 7.10.4	Lagunitas Land Use Policy Map (Map 5 of 5)	.following page	3.12-30
Ν	<b>A</b> ap 7.11	Bolinas Land Use Policy Map	following page	3.12-30
Ν	<b>A</b> ap 7.12	Stinson Beach Land Use Policy Map	following page.	3.12-30
Ν	<b>A</b> ap 7.13	Muir Beach Land Use Policy Map	.following page	3.12-30
Map 4-1	Historic 1	Resources	following page	4.13-2

## \* \* \* \* \*



#### Figure 5-1 Special-Status Animal Species Known or Suspected from Marin County

Common Name (Scientific Name)	Status Federal/State	Habitat
Amphibians/Reptiles		
Green sea turtle ( <i>Chelonia mydas</i> )	FΓ / -	Open ocean.
Loggerhead sea turtle ( <i>Caretta caretta</i> )	FΓ / -	Open ocean.
Leatherback sea turtle ( <i>Dermochelys</i>	FE / -	Open ocean.
coriacea)		
Northwestern pond turtle ( <i>Clemmys</i>	SC / CSC	Streams/ponds/lakes.
marmorata marmorata)	EE/020	
California tiger salamander (Ambystoma	F1/CSC	Breeds in pools and adults occupy surrounding
<i>califormense</i> )		grasslands/open woodlands.
California red-legged frog ( <i>Rana aurora</i> draytonii)	F1 / CSC	Forests/woodlands/grasslands and streamsides.
Foothill yellow-legged frog (Rana boylii)	SC / CSC	Streams with rocky substrate.
Birds		
Tricolored blackbird ( <i>Agelaius tricolor</i> ) (nesting colony)	SC / CSC	Freshwater marsh and surrounding fields.
Great egret (Ardea alba) (rookery)	- / -	Colonial nester in large trees.
Great blue heron (Ardea herodias) (rookery)	- / -	Colonial nester in trees, cliffsides, marshes.
Burrowing owl (Athene cunicularia) (burrow	- / CSC	Open grasslands/scrub.
sites)		
Western snowy plover (Charadrius	FT / CSC	Nesting along sandy beaches and shorelines.
alexandrinus nivosus) (nesting)	. ~ ~ ~	
Northern harrier ( <i>Circus cyaneus</i> ) (nesting)	- / CSC	Nesting in marsh and low shrubs.
Black swift ( <i>Cypseloides niger</i> ) (nesting)	SC / CSC	Nesting on cliffs and behind falls.
Yellow warbler ( <i>Dendroica petechia</i> brewsteri) (pesting)	rbler ( <i>Dendroica petechia</i> SC / CSC Nesting in willows and riparian cover.	
Snowy egret ( <i>Egretta thula</i> ) (rookery)	- / -	Colonial nester in trees, cliffsides, near marshland.
White-tailed kite ( <i>Elanus leucurus</i> ) (nesting)	SC / FP	Nesting in grassland/marshland with trees.
Tufted puffin (Fratercula cirrhata)	- / CSC	Colonial nester on offshore islands/cliffs.
Salt marsh common yellowthroat ( <i>Geothlypis</i>	SC / CSC	Salt and brackish-water marsh.
trichas sinuosa)		
California black rail ( <i>Laterallus jamaicensis</i>	- / ST; FP	Coastal salt marsh.
coturniculus)		
Black-crowned night heron (Nycticorax	- / -	Colonial nester in trees/shrubs near marshland.
<i>nycticorax</i> ) (rookery)		
Ashy storm-petrel ( <i>Oceanodroma</i>	SC / CSC	Colonial nester on offshore islands.
<i>homochroa</i> ) (rookery)		
Osprey ( <i>Pandion haliaetus</i> ) (nesting)	- / CSC	Nesting in trees associated with water bodies.
California brown pelican ( <i>Pelecanus</i>	FE/SE; FP	Coastal/bay shorelines and open water.
occidentalis californicus)		
California clapper rail ( <i>Kallus longirostris</i>	FE/SE	Salt and brackish marsh.

5-13



Common Name (Scientific Name)	Status Federal/State	Habitat
obsoletus	I cucia, Suite	
California least tern ( <i>Sterna antillarum</i>	FE/SE: FP	Coastal/bay shorelines and open water.
browni)	12, 52, 11	
Northern spotted owl ( <i>Strix occidentalis</i>	FT / -	Forest and woodland.
caurina)	/	
Fish		
Tidewater goby ( <i>Eucyclogobius newberry</i> )	FE/ CSC	Brackish water, marsh/bays,
Tomales roach ( <i>Lavinia symmetricus ssp.</i>	- / CSC	Tributaries of Tomales Bay.
symmetricus)	,	
Coho salmon ( <i>Oncorhynchus kisutch</i> )	FT / SE	Spawns in freshwater streams.
Chinook salmon (Oncorhynchus tshawytscha)	FT/	Spawns in freshwater streams.
Steelhead trout (Oncorhynchus mykiss)	FT/CSC	Spawns in freshwater streams.
Invertebrates	,	
Tomales isopod ( <i>Caecidotea tomalensis</i> )	- / -	Freshwater marsh/ponds.
Monarch butterfly ( <i>Danaus plexippus</i> )	-/-	Overwinters in blue gum eucalyptus.
(colonies)	,	0 51
Black abalone ( <i>Haliotis cracherodii</i> )	C/-	Rocky intertidal zone and ocean waters.
White abalone ( <i>Haliotis sorenseni</i> )	FE/-	Rocky intertidal zone and ocean waters.
Williams' bronze shoulderband	- / -	Known only from Hogg Island.
(Helminthoglypta arrosa williamsi)		
Peninsula coast range shoulderband snail	- / -	Known only from Point Reyes headland.
(Helminthoglypta nickliniana awania)		
Ricksecker's water scavenger beetle	- / -	Aquatic habitat/pools and ponds.
(Hydrochara rickseckeri)		
Mission blue butterfly ( <i>Icaricia icarioides</i>	FE / -	Shrubs/grasslands with lupine host.
missionensis)		
San Bruno elfin ( <i>Incisalia mossii bayensis</i> )	FE/-	Coastal scrub with stonecrop host plant.
Bumblebee scarab beetle ( <i>Lichnanthe ursina</i> )	- / -	Coastal dunes.
Tiburon micro-blind harvestman (Microcina	- / -	Serpentine outcrops near spring/seeps.
tiburona)		
Myrtle's silverspot (Speyeria zerene myrtleae)	FE / -	Scrub/grassland with larval host.
California freshwater shrimp (Syncaris	FE/SE	Freshwater streams with undercut banks.
pacifica)		
Mammals		
Pallid bat (Antrozous pallidus)	- / CSC	Roosts in protected locations.
Point Reyes mountain beaver (Aplodontia	- / CSC	Springs/ seeps with dense cover.
rufa phaea)		
Guadalupe fur seal (Arctocephalus townsendi)	FT/ST; FP	Open ocean, beaches.
Sei whale ( <i>Balaenoptera borealis</i> )	FE / -	Open ocean.
Blue whale ( <i>Balaenoptera musculus</i> )	FE / -	Open ocean.
Finback whale ( <i>Balaenoptera physalus</i> )	FE / -	Open ocean.
Townsend's western big-eared bat	- / CSC	Roosts in protected locations.
(Corynorhinus townsendii townsendii)		
Grey whale (Eschrichtius robustus)	FE / -	Open ocean.
Right whale (Eubalaena glacialis)	FE/-	Open ocean.
Stellar sea lion ( <i>Eumetopias jubatus</i> )	FT / -	Open ocean, beaches.



Common Name ( <i>Scientific Name</i> )	Status Federal/State	Habitat
Greater western mastiff-bat (Eumops perotis	SC/CSC	Roots in protected locations.
californicus)		
Southern sea otter ( <i>Enhydra lutris nereis</i> )	FT / FP	Near shore marsh habitat.
Humpback whale ( <i>Megaptera novaeangliae</i> )	FE/-	Open ocean.
Long-eared myotis bat ( <i>Myotis evotis</i> )	SC/-	Roosts in protected locations.
Fringed myotis bat ( <i>Myotis thysanodes</i> )	SC/	Roots in protected locations.
Long-legged myotis bat ( <i>Myotis volans</i> )	SC/-	Roots in protected locations.
Yuma myotis bat ( <i>Myotis yumanensis</i> )	SC/-	Roots in protected locations.
Sperm whale ( <i>Physeter catodon</i> )	FE/-	Open ocean.
Salt marsh harvest mouse ( <i>Reithrodontomys</i>	FE / SE; FP	Coastal salt marsh.
raviventris)		
Angel Island mole ( <i>Scapanus latimanus</i>	- / CSC	Coastal scrub/prairie on Angel Island.
insularis)		
Point Reyes jumping mouse (Zapus trinotatus	- / CSC	Coastal scrub/grassland from Point Reyes.
orarius)		





#### Figure 5-2 Special-Status Plant Species Known or Suspected from Marin County

Common Name (Scientific Name)	Status Federal/ State/CNPS	Habitat
Pink sand-verbena (Abronia umbellata ssp. breviflora)	SC / - / 1B	Coastal dunes/stand.
Blasdale's bent grass ( <i>Agrostis blasdalet</i> )	SC / - / 1B	Coastal dunes/scrub/prairie.
Point Reyes bent grass ( <i>Agrostis clivicola var punta-reyesensis</i> )	SC / - / -	Coastal scrub/prairie/coniferous forest.
Sonoma alopecurus (Alopecurus aequalis var sonomensis)	FE / - / 1B	Freshwater marsh/riparian scrub.
Napa false indigo (Amorpha californica var napensis)	- / - / 1B	Forest/chaparral/woodland.
Bent-flowered fiddleneck (Amsinckia hunaris)	- / - / 1B	Coastal bluff scrub/woodland/ grassland.
Mount Tamalpais manzanita (A <i>rctostaphylos hookeri ssp. montana</i> )	SC / - / 1B	Chaparral/grassland.
Marin manzanita (Arctostaphylos virgata)	- / - / 1B	Coniferous forest/chaparral.
Coastal marsh milk-vetch (Astragalus pycnostachyus var p.)	- / - / 1B	Dunes/marshes/swamps.
Point Reyes blennosperma ( <i>Blennosperma nanum var. robustum</i> )	SC / SR / 1B	Coastal prairie/scrub.
Small groundcone ( <i>Boschniakia hookeri</i> )	-/-/2	Coniferous forests.
Thurber's reed grass ( <i>Calamagrostis crassiglumis</i> )	SC / - / 2	Coastal scrub/freshwater marsh.
Tiburon mariposa lily ( <i>Calochortus tiburonensis</i> )	FT / ST / 1B	Serpentine grassland.
Coastal bluff morning glory ( <i>Calystegia purpurata ssp. saxicola</i> )	- / - / 1B	Dunes/coastal scrub.
Swamp harebell ( <i>Campanula californica</i> )	SC / - / 1B	Bogs/ferns/marshes in coniferous forest.
Flaccid sedge ( <i>Carex leptalea</i> )	-/-/2	Bogs/fens/meadows/seeps.
Lyngbye's sedge ( <i>Carex lyngbyei</i> )	-/-/2	Marshes/swamps.
Tiburon Indian paintbrush ( <i>Castilleja affinis ssp. neglecta</i> )	FE / ST / 1B	Serpentine grassland.
Humbolt <b>B</b> ay owl's clover ( <i>Castilleja ambigua ssp. humboldtiensis</i> )	SC / - / 1B	Coastal salt marsh.
Mount Vision ceanothus ( <i>Ceanothus gloriosus var. porrectus</i> )	SC / - / 1B	Coniferous forest/coastal scrub/prairie.
Mason's ceanothus ( <i>Ceanothus masonii</i> )	SC/SR/1B	Chaparral/serpentine.
San Francisco Bay spineflower ( <i>Chorizanthe cuspidata var. cuspidata</i> )	SC / - / 1B	Coastal scrub/prairie/dunes.
Woolly headed spineflower ( <i>Chorizanthe cuspidata var. villosa</i> )	- / - / 1B	Coastal scrub/prairie/dunes.
Robust spineflower (Chorizanthe robusta var. robusta)	FE/-/1B	Woodlands, coastal dunes/scrub.
Sonoma spineflower (Chorizanthe valida)	FE/SE/1B	Coastal prairie.
Franciscan thistle ( <i>Cirsium andrewsii</i> )	- / - / 1B	Forest/coastal bluff scrub/prairie/ coastal scrub.
Mount Tamalpais thistle (Cirsium hydrophilum var. vaseyi)	SC / - / 1B	Forest/chaparral.
Raiche's red ribbons ( <i>Clarkia concinna ssp. raichei</i> )	SC / - / 1B	Coastal bluff scrub.



Common Name ( <i>Scientific Name</i> )	Status Federal/	Habitat
	State/CNPS	
Round-headed Chinese houses (Collinsia corymbosa)	- / - / 1B	Coastal dunes.
Point Reyes bird's beak (Cordylanthus maritimus ssp.	SC / - / 1B	Coastal salt marsh/dunes.
palustris)		
Soft bird's beak (Cordylanthus mollis ssp. mollis)	FE / SR / 1B	Coastal salt marsh.
Baker's larkspur ( <i>Delphinium bakeri</i> )	FE / SR / 1B	Coastal scrub.
Yellow larkspur ( <i>Delphinium luteum</i> )	FE / SR / 1B	Chaparral/coastal scrub/prairie.
Western leatherwood ( <i>Dirca occidentalis</i> )	- / - / 1B	Forest/chaparral/woodland.
Supple daisy ( <i>Erigeron supplex</i> )	- / - / 1B	Coastal bluff scrub/prairie.
Minute pocket-moss ( <i>Fissidens pauperculus</i> )	- / - / 1B	Forest floor along coast.
Fragrant fritillary ( <i>Fritillaria liliacea</i> )	SC / - / 1B	Coastal scrub/prairie/grassland.
Dune gilia ( <i>Gilia capitata ssp. chamissonis</i> )	- / - / 1B	Dunes/coastal scrub.
Wooly-headed gilia (Gilia capitata ssp. tomentosa)	- / - / 1B	Coastal bluff scrub/outcrops.
Dark-eyed gilia ( <i>Gilia millefoliata</i> )	- / - / 1B	Coastal dunes.
San Francisco gumplant ( <i>Grindelia hirsutula var. maritima</i> )	-/-/1B	Coastal bluff scrub/coastal scrub/ grassland.
Diablo helianthella ( <i>Helianthella castanea</i> )	- / - / 1B	Forest/chaparral/woodland/coastal scrub/grassland.
Short-leaved evax (Hesperevax sparsiflora var. brevifolia)	-/-/2	Coastal bluff scrub/dunes.
Marin western flax (Hesperolinon congestum)	FT/ST/1B	Chaparral/grassland.
Santa Cruz tarplant ( <i>Holocarpha macradenia</i> )	FT / FE / 1B	Coastal prairie/coastal scrub/ grassland.
Kellogg's horkelia ( <i>Horkelia cuneata ssp. sericea</i> )	SC / - / 1B	Coniferous forest/coastal scrub/ chaparral.
Point Reyes horkelia ( <i>Horkelia marinensis</i> )	SC / - / 1B	Coastal scrub/prairie/dunes.
Thin-lobed horkelia ( <i>Horkelia tenuiloba</i> )	- / - / 1B	Coastal scrub/chaparral.
Baker's goldfields ( <i>Lasthenia macrantha ssp. baken</i> )	- / - / 1B	Coniferous forest/coastal scrub.
Perennial goldfields ( <i>Lasthenia macrantha ssp. macrantha</i> )	- / - / 1B	Coastal bluff scrub/dunes/coastal scrub.
Beach layia ( <i>Layia carnosa</i> )	FE/SE/1B	Coastal dunes.
Tamalpais lessingia ( <i>Lessingia micradenia var. micradenia</i> )	SC / - / 1B	Chaparral/grassland in serpentine.
Mason's lilaeopsis ( <i>Lilaeopsis masonii</i> )	SC/SR/1B	Fresh and brackish marsh.
Coast lily ( <i>Lilium maritimum</i> )	- / - / 1B	Forest/prairie/coastal scrub/marshes/swamps.
Point Reyes meadowfoam ( <i>Limnanthes douglasii ssp. sulphurea</i> )	SC / SE / 1B	Freshwater marsh/prairie/seeps.
Large-flowered linanthus (Linanthus grandiflorus)	SC / - / 4	Coastal bluff scrub.
Tidestrom's lupine (Lupinus tidestromii)	FE/SE/1B	Coastal dunes.
Marsh microseris ( <i>Microseris paludosa</i> )	- / - / 1B	Forest/woodland/coastal scrub/ grassland.
Baker's navarretia ( <i>Navarretia leucocephala ssp. bakeri</i> )	- / - / 1B	Woodland/seeps/pools/grassland/ forest.



Common Name (Scientific Name)	Status Federal/ State/CNPS	Habitat
Marin County navarretia ( <i>Navarretia rosulata</i> )	- / - / 1B	Coniferous forest/chaparral.
White-rayed pentachaeta (Pentachaeta bellidiflora)	FE / SE / 1B	Grassland on serpentine.
North Coast phacelia (Phacelia insularis var. continentis)	-/-/1B	Coastal bluff scrub/dunes.
Point Reyes rein orchid (Piperia elegans ssp. decurtata)	-/-/1B	Coastal bluff scrub only from Point Reyes National Seashore.
Hairless popcorn flower ( <i>Plagiobothrys glaber</i> )	/ / 1A	Meadows/seeps/marshes/swamps.
North Coast semaphore grass (Pleuropogon hooverianus)	SC / ST / 1B	Forest/steeps.
Marin knotweed ( <i>Polygonum marinense</i> )	SC / - / 3	Marshes/swamps.
Tamalpais oak ( <i>Quercus parvula var. tamalpaisensis</i> )	-/-/1B	Coniferous forest only on Mount Tamalpais.
California beaked-rush ( <i>Rhynchospora californica</i> )	SC / - / 1B	Bogs/marshes/seeps/coniferous forest.
Point Reyes checkerbloom ( <i>Sidalcea calycosa ssp. rhizomata</i> )	- / - / 1B	Marshes/swamps.
Marin checkerbloom ( <i>Sidalcea hickmanii ssp. viridis</i> )	SC / - / 1B	Chaparral.
Purple-stemmed checkerbloom ( <i>Sidalcea malviflora ssp. purpurea</i> )	-/-/1B	Forest/prairie.
Santa Cruz microseris (Stebbinsoseris decipiens)	SC/-/1B	Forest/chaparral/coastal scrub/prairie.
Tamalpais jewel-flower (Streptanthus batrachopus)	SC / - / 1B	Coniferous forest/chaparral.
Mount Tamalpais jewel-flower ( <i>Streptanthus glandulosus ssp. pulchellus</i> )	-/-/1B	Chaparral/grassland.
Tiburon jewel-flower (Streptanthus niger)	FE / SE / 1B	Grassland on serpentine.
Showy Indian clover ( <i>Trifolium amoenum</i> )	FE / - / 1B	Grassland/coastal bluff scrub.
San Francisco owl's clover ( <i>Triphysaria floribunda</i> )	SC / - / 1B	Coastal prairie/grassland.

STATUS DESIGNATIONS

Federal:

- FE = Listed as "endangered" under the federal Endangered Species Act.
- FT = Listed as "threatened" under the federal Endangered Species Act.
- PE = Proposed for federal listing as "endangered."
- PT = Proposed for federal listing as "threatened."
- C = A candidate species under review for federal listing. Candidates include taxa for which the USFWS has sufficient biological information to support a proposal to list as endangered or threatened.
- SC = Species of concern; formerly considered a candidate species for listing by the USFWS.

State:

- SE = Listed as "endangered" under the California Endangered Species Act.
- SR = Listed as "rare" under the California Endangered Species Act.
- ST = Listed as "threatened" under the California Endangered Species Act.
- CP = California fully protected species; individual may not be possessed or taken at any time.
- CSC = Considered a species of special concern by the CDFG; taxa have no formal legal protection, but nest sites and communal roosts are generally recognized as significant biotic features.

CNPS:

1A = Plants of highest priority; plants presumed extinct in California.



- 1B = Plants of highest priority; plants rare and endangered in California and elsewhere.
- 3 = Plants requiring additional information; a review list.
- 4 = Plants of limited distribution; a watch list.

## **CWP** Glossary

Accessible Housing. Housing units accessible and adaptable to the needs of the physically disabled.

**Accessory Structure.** A structure that is physically detached from, secondary and incidental to, and commonly associated with the primary structure or use.

Accessory Use. A subordinate use that is incidental to the principal use on the same lot or building site.

Acres (Gross). The entire acreage of a site, generally excluding perimeter roadways but including interior roadways and easements and areas below the high-tide line.

Acres (Net). The developable portion of a site, after excluding public or private road rights-of-way, public open space, and primary floodways.

Adaptive Reuse. The conversion of obsolescent or historic buildings to provide the opportunity for new uses within a community, such as the conversion of a former hospital or school to residential or mixed use, or the conversion of a bank building to a store or office.

Adequate Sites. Pursuant to State law, having sufficient land zoned to meet regional "fair share" housing unit allocations. In their housing elements, cities and counties need to identify an inventory of land suitable for residential development, including vacant sites and underutilized sites.

Adjacent. Having a common border.

Adverse Impact. A negative consequence for the physical, social, or economic environment resulting from an action or project.

Affordability. The generally accepted banking/governmental standard for determining whether a person can afford housing means spending no more than 30% of one's gross monthly household income on housing costs, which for owner housing would include principal, interest, utilities, and insurance. For example, a beginning schoolteacher earning \$26,750 per year can afford to pay up to \$668 per month for housing. A police officer or firefighter earning \$42,800 per year can afford up to \$1,070 per month. See "Income Levels" and "Income Limit (Housing)."

**Affordable Housing.** Dwelling units that are rented or sold at rates that are affordable to households of moderate, low, or very low income. Housing is considered affordable when a household pays less than 30% of its gross monthly income (GMI) for housing, including utilities.

**Agency.** A governmental entity, department, office, or administrative unit responsible for carrying out regulations.



GLOSSARY

Agricultural Preserve. Land designated for agriculture or conservation. See "Williamson Act."

Agricultural Production. The commercial production of agricultural crops as defined above.

**Agricultural Production and Stewardship Plan.** A plan that identifies existing and proposed agricultural uses and resources for a property. The intent of these plans is to demonstrate the following: (1) the long-term agricultural use of the property will be preserved; (2) agricultural infrastructure has been established or will be enhanced; (3) agricultural uses proposed in connection with the residence are appropriate to the site; (4) sound land stewardship has been implemented or will be enacted; and (5) at least 90% of the usable land of the property will be engaged in agricultural production.

Agricultural Viability. The collective success and/or ongoing effectiveness of agricultural operations and enterprises.

**Agricultural Worker Housing.** Any attached or detached dwelling unit used to house agricultural workers and their family members, including temporary mobile homes. For the purposes of calculating density, no more than one food-preparation area shall be provided for each agricultural worker housing unit.

**Agriculture.** The breeding, raising, pasturing, and grazing of livestock for the production of food and fiber; the breeding and raising of bees, fish, poultry, and other fowl; and the planting, raising, harvesting, and producing of agricultural, aquacultural, horticultural, and forestry crops.

**Agri-tourism.** A business conducted by farmers or ranchers on their working agricultural operation for the enjoyment and education of visitors. It is intended to promote farm products and to generate additional or supplemental farm income.

**Air Pollution.** Concentrations of substances found in the atmosphere that exceed naturally occurring quantities and are undesirable or harmful.

Air Quality. Federal and state standards for emissions of locally generated pollutants: carbon monoxide, ozone, nitrogen dioxide, sulfur dioxide, and total suspended particulates.

**Air Rights.** The right to the use of air space over property owned by another. Air rights are granted for space above an existing right-of-way, parking lot, or other type of property. Air rights can be sold or leased and a platform built over the existing use, where new development can be constructed.

**Alluvium.** A general term for the sediments laid down in riverbeds, floodplains, lakes, fans at the foot of a mountain slope, and estuaries during relatively recent geologic times.



#### GLOSSARY

Alquist-Priolo Earthquake Fault Zone. A regulatory zone, delineated by the State Geologist, within which site-specific geologic studies are required to identify and avoid fault rupture hazards prior to subdivision of land and/or construction of most structures for human occupancy.

**Ambient.** Surrounding on all sides; used to describe measurements of existing conditions with respect to traffic, noise, air, and other environments.

**Amenity.** A feature that increases attractiveness or value. Examples of amenities include public open space adjacent to or near a building or group of buildings, aesthetically pleasing architecture, good schools, bicycle lanes, and storage spaces.

Anadromous Fish. Species of fish that mature in the ocean and migrate into streams to spawn.

**Aquaculture.** The raising and harvesting of aquatic organisms, including shellfish, mollusks, crustaceans, kelp, and algae.

**Aquifer.** An underground, water-bearing layer of earth, porous rock, sand, or gravel, through which water can seep or be held in natural storage. Aquifers generally hold sufficient water to be used as a water supply.

Archaeological. Relating to the material remains of past human life, culture, or activities.

**Arterial.** A major street carrying the traffic of local and collector streets to and from freeways and other major streets, with controlled intersections and generally providing direct access to properties.

**Assisted Housing.** Generally multi-family rental housing, but sometimes single-family ownership units, whose construction, financing, sales prices, or rents have been subsidized by federal, State, or local housing programs.

Assisted Housing Developments. Multi-family rental housing that receives governmental assistance under federal programs, State and local multi-family revenue bond programs, local redevelopment programs, the federal Community Development Block Grant Program, or local in-lieu fees. The term also includes multi-family rental units that were developed pursuant to a local inclusionary housing program or used to qualify for a density bonus.

Bankfull. Indicates the height (or stage) of a stream that just fills the stream channel.

Barrier-Free Design. Design that provides access for persons with physical disabilities. See "Universal Design."

**Base Flood.** In any given year, a 100-year flood that has a 1% likelihood of occurring and is recognized as a standard for acceptable risk.



**Baylands.** Lands within or adjacent to a bay and touched by tidal action, as well as lands that the tides would encompass in the absence of any levees or other constructed structures. Baylands include tidal flats, tidal marsh, lagoons, and diked lands.

**Below-Market-Rate (BMR) Housing.** Housing that is sold or rented at a price that is below the prevailing rate for equivalent housing units within the same community.

**Bicycle Lane (Class II facility).** A corridor expressly reserved for bicycles on a street or roadway in addition to any lanes for use by motorized vehicles.

**Bicycle Path (Class I facility).** A paved route not on a street or roadway and expressly reserved for bicycles traversing an otherwise unpaved area. Bicycle paths may parallel roads but typically are separated from them by landscaping.

**Bicycle Route (Class III facility).** A facility shared with motorists and identified only by signs. A bicycle route has no bicycle pavement markings or lane stripes.

Bikeway. A term that encompasses bicycle lanes, bicycle paths, and bicycle routes.

**Biodiesel.** A domestically producible, nontoxic, biodegradable, renewable fuel made from vegetable oil, methanol, and a catalyst (usually sodium hydroxide or potassium hydroxide). The vegetable oils primarily used in the United States are soy, canola, and rapeseed. Biodiesel contains no petroleum, but it can be blended with petroleum diesel in any ratio or used without petroleum.

**Biogas.** Biogas energy is recovered methane from landfills or agricultural operations used to power an engine or a turbine.

**Biological Diversity, Biodiversity.** The number and abundance of species found within a common environment. This includes the variety of genes, species, and ecosystems, and the ecological processes that connect everything in a common environment.

**Biotic Community.** A group of living organisms characterized by a distinctive combination of both animal and plant species in a particular habitat.

**Brownfield.** A piece of industrial or commercial property that has been abandoned, is idle, or is underused, and often is environmentally contaminated, especially one considered a potential site for redevelopment.

**Buffer.** A land area that is designed to block or absorb unwanted impacts to the area beyond the buffer. Buffers generally allow for some flexibility of use within the designated area, with restrictions increasing closer to the protected resource.

**Building Height.** The greatest vertical distance from finished grade to the roof, excluding architectural features such as chimneys.



## **Buildout.** Development of land to its full potential or theoretical capacity as permitted under current or proposed planning or zoning designations.

**Business Incubator.** A center that supports new and growing businesses under one roof, offering a wide range of business training, support programs, flexible leases, and shared equipment in a professional working environment.

**California Environmental Quality Act (CEQA).** A State law, originally enacted in 1970, that requires public agencies to document and consider the environmental effects of a proposed action before a decision is issued. See California Public Resources Code Sections 21000, et seq.

**California Housing Finance Agency (CHFA).** A State agency, established by the Housing and Home Finance Act of 1975, that is authorized to sell revenue bonds and generate funds for the development, rehabilitation, and conservation of low and moderate income housing.

**Canopy.** (1) The part of any stand of trees represented by the tree crowns. Typically refers to the uppermost layer of foliage but can be used to describe lower layers in a multistoried forest. (2) The projection of a roof structure beyond the exterior of a building, such as an eave or porch roof.

**Capital Improvements Program (CIP).** A program, administered by a city or county government, that schedules permanent improvements, usually for a minimum of five years in the future, to fit the projected fiscal capability of the local jurisdiction. The program generally is reviewed annually for conformance to and consistency with the General Plan.

Carbon Dioxide. A colorless, odorless, nonpoisonous gas that is a normal part of the atmosphere.

**Carbon Monoxide.** A colorless, odorless, highly poisonous gas produced by automobiles and other machines with internal combustion engines that imperfectly burn fossil fuels such as oil and gas.

**Census (U.S.).** The official decennial enumeration of the population conducted by the federal government.

**Channelization.** (1) The straightening and/or deepening of a watercourse for purposes of storm runoff control or ease of navigation. Channelization often includes lining of stream banks with a retaining material such as concrete. (2) At the intersection of roadways, the directional separation of traffic lanes through the use of curbs or raised islands, which limit the paths that vehicles may take through the intersection.

**Charrette.** Any collaborative session in which a group of designers and interested parties drafts a solution to a design problem. In urban planning, the charrette has become a technique for consulting with all stakeholders while integrating the aptitudes and interests of a diverse group of people.

Circulation. The movement of people and goods within a region.

**City.** *City* with a capital *C* generally refers to the government or administration of a city. *City* with a lowercase *c* may mean any city or may refer to the geographical area of a city (e.g., the city's bikeway system).

GLOSSARY

**Climate Change.** Any long-term significant change in measures of climate such as temperature, precipitation, or wind. Possible causes include changes in natural factors, natural processes, and/or human activities.

**Closed-Loop Systems.** Reusing materials to achieve zero waste. Such systems incorporate any used material into new products to reduce the need for raw material and eliminate all waste being disposed of or transmitted into the environment.

**Clustered Development.** Development in which a number of dwelling units are placed in closer proximity than usual or attached with the purpose of retaining an open space area.

**Coastal County.** A county, or city and county, that lies, in whole or in part, within the coastal zone. *See "Coastal Zone."* 

**Coastal-Dependent Development or Use.** Any development or use that requires a site on or adjacent to the sea to be able to function at all.

**Coastal Development Permit.** A permit for any development within the coastal zone that is required pursuant to the California Coastal Act.

**Coastal Plan.** The California Coastal Zone Conservation Plan prepared and adopted by the California Coastal Zone Conservation Commission and submitted to the governor and the Legislature on December 1, 1975, pursuant to the California Coastal Zone Conservation Act of 1972.

Coastal-Related Development. Any use that is dependent on a coastal-dependent development or use.

**Coastal Zone.** The land and water area of the State of California from the Oregon border to the border of the Republic of Mexico, extending seaward to the State's outer limit of jurisdiction, including all offshore islands, and extending inland generally 1,000 yards from the mean high tide line of the sea. In significant coastal estuarine, habitat, and recreational areas it extends inland to the first major ridgeline paralleling the sea, or five miles from the mean high tide line of the sea, whichever is less, and in developed urban areas the zone generally extends inland less than 1,000 yards.

**Cohousing.** A type of shared housing arrangement. Cohousing developments have individual units with kitchens combined with a common kitchen and meeting room. They may also include such common features as child care facilities, artist studios, darkrooms, and woodworking shops. Cohousing developments are normally organized as condominiums, although they can also be organized as cooperatives.

**Collector.** A street for traffic moving between arterial and local streets, generally providing direct access to properties.



**Colluvium.** Rock, organic debris, and soil accumulated at the foot of a slope.

**Commercial.** A land use classification that permits facilities for the buying and selling of commodities and services.

**Community Development Block Grant (CDBG).** A grant program administered by the U.S. Department of Housing and Urban Development (HUD) on a formula basis for entitlement communities and by the State Department of Housing and Community Development (HCD) for non-entitled jurisdictions. This grant allots money to cities and counties for housing acquisition and rehabilitation and community development, including public facilities and economic development.

**Community Noise Equivalent Level (CNEL).** A 24-hour energy equivalent level derived from a variety of single-noise events with weighting factors of 5 and 10 dBA (*see "dBA"*) applied to the evening (7 p.m. to 10 p.m.) and nighttime (10 p.m. to 7 a.m.) periods, respectively, to allow for the greater sensitivity to noise during these hours.

**Community Plan.** A planning document that sets forth goals, objectives, policies and programs to address specific issues related to a particular unincorporated community. Community plans are considered part of the Marin Countywide Plan.

**Community Park.** A park that serves a population of 10,000 to 30,000 within a three-mile radius and usually contains specialized facilities such as swimming pools, tennis courts, community centers, and sports field complexes.

**Community Redevelopment Agency.** A local agency created under California redevelopment law or a local legislative body that has elected to exercise the powers granted to such an agency for the purpose of planning, developing, re-planning, redesigning, clearing, reconstructing, and/or rehabilitating all or part of a specified area with residential, commercial, industrial, and/or public (including recreational) structures and facilities. The redevelopment agency's plans must be compatible with adopted community general plans.

**Community Separator.** Landforms such as hills, ridgelines, watercourses, floodplains, or other environmentally sensitive areas that have served to physically separate communities. *See "Ridge and Upland Greenbelt."* 

**Community Service Area.** A geographic sub-area of a city or unincorporated area used for the planning and delivery of parks, recreation, and other human services based on an assessment of the service needs of the population in that sub-area.

**Community Services District.** A special district with taxing authority and an elected board of directors used to provide public services.

**Condominium.** As defined by Civil Code Section 1351(f), a development where undivided interest in common in a portion of real property is coupled with a separate interest in space called a unit, the boundaries of which are described on a recorded final map, parcel map, or condominium plan. The



area within the boundaries may be filled with air, earth, or water, or any combination, and need not be physically attached to any land except by easements for access and, if necessary, support.

**Congestion Management Program (CMP).** A mechanism employing growth management techniques, including traffic level of service standards, development mitigation programs, transportation systems management, and capital improvement programming, for the purpose of controlling and/or reducing the cumulative regional traffic impacts of development. All cities and counties with urbanized areas are required to adopt and update a Congestion Management Program.

**Congregate Housing.** Long-term supportive housing in a group setting, which includes independent living and sleeping accommodations in conjunction with shared dining and recreational facilities. Residents of congregate care facilities occupy individual apartments, most of which include kitchens, although these may be minimal.

**Connectivity.** The degree to which similar habitats are linked.

Conservation. The management of natural resources to prevent waste, destruction, or neglect.

Conservation Contract. See "Williamson Act."

Conservation Easement. See "Easement, Conservation."

**Contract-Restricted Land.** Land with development potential restricted by contract, such as the Williamson Act for agricultural lands, or transfer of development rights from designated open space.

**County.** *County* with a capital *C* is the County of Marin. *County* with a lowercase *c* refers to the geographical area of Marin County.

**Countywide Planning Agency (CWPA).** A panel composed of representation from the Marin County Board of Supervisors and each of the 11 cities and towns. The CWPA reviews land use policy recommendations of a countywide nature.

**Cover.** Any feature that conceals wildlife or fish. Cover may be dead or live vegetation, boulders, or undercut stream banks. Animals use cover to escape from predators and other threats, to rest, to feed, and to breed.

Creek. See "Stream."

**Critical Facility.** A facility that either (1) provides emergency or essential services or (2) houses or serves many people who would be injured or killed in case of disaster damage to the facility. Examples include hospitals, fire stations, police and emergency services facilities, utility facilities, and communications facilities.

**Critical Habitat.** An area designated for the survival and recovery of federally listed threatened or endangered species.



**Cumulative Impact.** As used in CEQA, refers to two or more individual effects that, when considered together, are considerable or that compound or increase other environmental impacts.

GLOSSARY

**Dam Inundation Area.** An area of potential flooding from dam rupture in the event of an earthquake or major storm. The California Dam Safety Act requires that counties plan for such a failure.

**dB.** Decibel, a unit used to express the relative intensity of sound as it is heard by the human ear.

**dBA.** The "A-weighted" scale for measuring sound in decibels that weighs or reduces the effects of low and high frequencies in order to simulate human hearing. Each increase of 10 dBA intensifies the noise tenfold and doubles the perceived loudness.

**Dedication.** A donation by an owner or developer of private land for public use and the acceptance of land for such use by the governmental agency having jurisdiction over the public function for which it will be used. A city or a county often makes dedication of land for roads, parks, school sites, or other public uses a condition of approval of a development.

**Deed Restricted.** A Deed of Trust recorded against the title of a property, usually required as a condition of approval or purchase, specifying the terms of use, level of affordability, or other agreed conditions of a property. Affordable housing is often established through a deed restriction that stipulates the income eligibility of buyers/renters, the duration of affordability, and the level of equity increase allowed at sale.

**Defensible Space.** (1) In firefighting and prevention, a 30-foot area of noncombustible surfaces separating urban and wildland areas. (2) In urban areas, open spaces, entry points, and pathways configured to provide maximum opportunities to rightful users and/or residents to defend themselves against intruders and criminal activity.

**Density Bonus.** The allocation of development rights that allow a parcel to accommodate additional square footage or additional residential units beyond the maximum for which the parcel is zoned, usually in exchange for the provision or preservation of an amenity at the same site or at another location. Under California law a housing development may be eligible to receive a density bonus of 20% to 35% for very low or lower income households and senior housing. Additional benefits for reduced parking, reductions in site development, incentives, or concessions may be required. Units must remain affordable for no less than 30 years.

**Density, Control of.** A limitation on the occupancy of land. Density can be controlled through zoning in the following ways: use restrictions, minimum lot-size requirements, floor area ratios, land use intensity ratios, setback and yard requirements, minimum house size requirements, ratios comparing number and types of housing units to land area, limits on units per acre, and other means. Allowable density often serves as the major distinction between residential districts.


**Density (Residential).** The number of dwellings per acre of lot area, unless otherwise stated, for residential uses. Densities specified in the general plan may be expressed in units per gross acre or per net acre. *See "Acres, Gross" and "Acres, Net."* 

**Density Transfer.** A way of retaining open space by concentrating densities, usually in compact areas adjacent to existing urbanization and utilities, while leaving unchanged historic, sensitive, or hazardous areas. *See "Transfer of Development Rights."* 

**Design Review.** Regulations and a public review process intended to preserve and enhance the natural beauties of the land and the built environment, maintain and improve the relationship between development and the amenities and attractiveness of an area or neighborhood or an area, and stimulate creative design and innovative use of materials.

**Developable Acres, Net.** The portion of a site that can be used for density calculations. Public or private road rights-of-way are not included in the net developable acreage of a site. *See "Acres, Net."* 

**Developable Land.** Land that is suitable as a location for structures and that can be developed free of hazards to and without disruption of or significant impact on natural resource areas.

**Development.** Any activities occurring on land and in or under water including placement or construction of any solid material or structure; construction of roadways and other infrastructure; discharge or disposal of dredged material or any other waste materials; grading, dredging, or mining; subdivision or change in the density or intensity of use of land; change in the intensity of use or water; construction, reconstruction, demolition, or alteration to the size of any structure, public or private; and the removal or harvesting of vegetation for other than agricultural purposes but excluding routine repair and maintenance activities.

**Development Application.** A request for approval of a project requiring discretionary action, including but not limited to Master Plans, Precise Development Plans, Tentative Maps and Subdivisions, Design Review, Use Permits, Variances, and Tidelands Permits. Development applications do not include ministerial permits and actions such as building permits.

**Development Code.** The Marin County Development Code, Title 22 of the Marin County Code, consisting of the County's zoning and subdivision regulations.

#### Development Fee. See "Impact Fee."

**Development Permit.** Any entitlement, as described in the Development Code, including, but not limited to Design Review, Floating Home Adjustment Permits, Use Permits, Temporary Use Permits, Tidelands Permits, Variances, Master Plans, or Precise Development Plans.

**Development Rights.** The right to develop land by a landowner who maintains fee-simple ownership over the land, or by a party other than the owner who has obtained the rights to develop. Such rights are usually expressed in terms of density allowed under existing zoning. For example, one development right may equal one unit of housing or may equal a specific number of square feet of gross floor area in



one or more specified zoning districts. See "Interest, Fee," "Interest, Less-Than-Fee," and "Transfer of Development Rights."

#### Development Rights, Transfer of (TDR). See "Transfer of Development Rights."

**Differential Settlement.** Uneven settlement that occurs with time or during earthquake shaking in poorly consolidated granular soils adjacent to bedrock. Loss of strength or the loss of water and sand through liquefaction often does not occur evenly over broad areas. Thus the ground settles different amounts in adjacent spots. Differential settlement can be very destructive to buildings.

**Diked Bay Marshlands.** Lands originally subject to tidal action or within the historic limits of a bay that are now separated from the bay by a dike, levee, or other protective structure.

Disabled. See "Persons with Disabilities."

**Discretionary Decision.** As used in CEQA, an action taken by a governmental agency that calls for the exercise of judgment in deciding whether to approve and/or how to carry out a discretionary project.

**Discretionary Project or Permit.** A project or permit that the decision-making body may approve, approve with conditions, or deny. *See "Development Application."* 

Diversion. The direction of water in a stream away from its natural course.

**Duet (Residential).** A detached building divided into two units sharing a common wall, which is designed for occupation as the residence of two families living independently of each other. Similar to a duplex except the connected units are on separate lots.

**Duplex.** A detached building under single ownership that is designed for occupation as the residence of two families living independently of each other.

**Dwelling or Dwelling Unit.** A room or group of internally connected rooms that have sleeping, cooking, eating, and sanitation facilities, but not more than one kitchen, which constitute an independent housekeeping unit, occupied by or intended for one household on a long-term basis. Types of dwellings include single-family dwellings, two-family dwellings, multifamily dwellings, mobile homes, condominiums and townhouses, and floating homes.

**Easement.** The right to use property owned by another for specific purposes or to gain access to another property. For example, utility companies often have an easement on private property allowing the company to install and maintain utility facilities.

**Easement, Conservation.** A tool for acquiring open space with less than full-fee purchase, whereby a public agency buys only certain specific rights from the landowner. These may be positive rights (providing the public with the opportunity to hunt, fish, hike, or ride over the land), or they may be restrictive rights (limiting the use to which the landowner may devote the land in the future).

**Easement, Scenic.** A tool that allows a public agency to use an owner's land for scenic enhancement, such as roadside landscaping or vista preservation.

GLOSSARY

**Ecological Footprint.** A measurement of the use of natural resources expressed as the number of acres of biologically productive area used to support one person.

**Ecology.** The interrelationships of living things to one another and to their environment, or the study of these interrelationships.

**Ecosystem.** An arrangement of living and non-living things, and the forces that influence them. Living things include plants and animals. Non-living parts of ecosystems may be rocks and minerals, as well as manmade features such as structures and roadways. Forces affecting ecosystems include weather, fire, disease, and human-induced changes, such as habitat removal to accommodate agricultural and urban development.

**Ecotone.** The transition zone between two biotic communities, such as between oak woodlands and grasslands.

EIR/EIS. See "Environmental Impact Report" and "Environmental Impact Statement."

**Elderly Housing.** Typically one- and two-bedroom apartments or condominiums designed to meet the needs of persons 62 years of age and older, or, if more than 150 units, persons 55 years of age and older, and restricted to occupancy by them.

**Emergency Shelter.** A facility that provides immediate and short-term housing and supplemental services for the homeless. Shelters come in many sizes, but an optimum size is considered to be 20 to 40 beds. Supplemental services may include food, counseling, and access to other social programs.

**Eminent Domain.** The right of a public entity to acquire private property for public use by condemnation and the payment of just compensation.

**Emission Standard.** The maximum amount of a pollutant legally permitted to be discharged from a single source, either mobile or stationary.

**Employee Housing.** An accessory residential dwelling unit or dwelling units located in, or adjacent to, a commercial building on a parcel having a primary commercial land use and occupied by an employee or employees of the commercial use(s) on the same property or a family member who is actively engaged in such commercial use.

**Endangered Species.** A species of animal or plant is considered to be endangered when its prospects for survival and reproduction are in immediate jeopardy from one or more causes.

**Energy Conservation Versus Energy Efficiency.** Energy conservation connotes "doing without" in order to save energy, rather than using less energy to do the same thing. For example, turning off lights, turning down the air conditioner, and making fewer vehicle trips are all conservation measures. Energy



efficiency means using less energy/electricity to perform the same function, such as installing lighting that uses less electricity, installing additional insulation, and switching to a vehicle with better gas mileage.

**Environment.** The physical conditions that exist within an area that will be affected by a proposed project. The conditions include land, air, water, minerals, flora, fauna, noise, and objects of historical or aesthetic significance. (CEQA definition.)

**Environmental Assessment.** A preliminary evaluation of site resources, conditions, and policy considerations including a composite development constraints and capabilities analysis prepared prior to filing of a development application for undeveloped, agricultural, or redevelopment lands and adjacent water areas located within the Baylands Corridor and Bayfront Conservation Zone, pursuant to the Development Code and Marin County Environmental Impact Review Guidelines. The environmental assessment is intended to guide preparation of development plans and assist in public agency review of such plans. The assessment may be later used in conjunction with describing the environmental setting section of an environmental review document, but is not a document prepared pursuant to the California Environmental Quality Act (CEQA).

**Environmental Impact Report (EIR).** A report required by the California Environmental Quality Act (CEQA) that assesses all the environmental characteristics of an area and determines what effects or impacts will result if the area is altered or disturbed by a proposed action. *See "California Environmental Quality Act."* 

**Environmental Impact Statement (EIS).** A document required of federal agencies by the National Environmental Policy Act for major projects or legislative proposals significantly affecting the environment. A tool for decision making, it describes the positive and negative effects of the undertaking and cites alternative actions.

**Environmental Justice.** The fair treatment of people of all races, cultures, and incomes with respect to the development, adoption, implementation, and enforcement of environmental laws, regulations, and policies.

**Environmentally Sensitive Area.** Any area in which plant or animal life or habitats are either rare or especially valuable because of their special nature or role in an ecosystem, and which could be easily disturbed or degraded by human activities and developments.

**Erosion.** The process by which soil and rock are detached and moved by running water, wind, ice, and gravity.

**Estuary.** The lower course of a river or stream where tidal influence is noticeable; the mixing zone of fresh and salt waters near the mouth of a river or stream.

**Exaction.** A contribution or payment required as an authorized precondition for receiving a development permit; usually refers to mandatory dedication (or fee in lieu of dedication) requirements found in many subdivision regulations.



Expansive Soils. Soils that swell when they absorb water and shrink as they dry.

Expressway. A highway with full or partial control of access with some intersections at grade.

**Fair Market Rent.** The rent, including utility allowances, determined by the U.S. Department of Housing and Urban Development for purposes of administering the Section 8 Existing Housing Program.

**Family.** (1) Two or more persons related by birth, marriage, or adoption (as used by the U.S. Bureau of the Census). (2) One or more persons occupying a dwelling and living as a single, domestic housekeeping unit, as distinguished from a group occupying a hotel, motel, club, or fraternity or sorority house.

**Farmland Security Zone Contract.** A contract between a private landowner and a county that enforceably restricts land to agricultural uses. The minimum initial term is 20 years. This type of contract offers landowners greater property tax reduction than a Williamson Act contract. Land restricted by a farmland security contract is valued for the purpose of property assessment at 65% of its Williamson Act valuation or 65% of its Proposition 13 valuation, whichever is lower. *See "Williamson Act."* 

**Farmland Soils of Statewide Importance.** Soils similar to "Prime Farmland Soils," but with minor shortcomings, such as greater slopes, or with less ability to hold and store moisture.

**Fault.** A fracture or zone of closely associated fractures along which rocks on one side have been displaced with respect to those on the other side. A fault zone is a zone of related faults, which commonly are braided, but which may be branching. A fault trace is the line formed by the intersection of a fault and the earth's surface. *Active Fault:* A fault that has exhibited surface displacement within the Holocene Epoch (approximately the past 11,000 years). *Potentially Active Fault:* A fault that shows evidence of surface displacement during the Quaternary Period (the last 2 million years).

**Fee/Fee Waiver.** A fee or exaction charged on new commercial and/or residential development to generate funding for infrastructure development and public improvements, based on the impact that the new development would have on existing facilities. "Fee waivers" that reduce or eliminate local development fees are sometimes allowed for affordable housing developments or other types of development for which the fee would constitute a substantial hardship.

**Fill.** A deposit of earth material placed by artificial means; any act by which earth, sand, gravel, rock, or any other material is placed, pushed, dumped, pulled, transported, or moved to a new location above the natural surface of the ground, on top of the stripped surface, or in a submerged area.

**Finding(s).** The result(s) of an investigation and the basis upon which decisions are made. Findings are used by government agents and bodies to justify action taken by the entity.



**Fire Flow.** Term firefighters use to describe how much water can be delivered by a water system through one or more hydrants to fight a fire at a specific location. Also used to state the optimum amount (standard) of water flow that firefighters require for a theoretical fire at a specific location.

**Fire Hazard Zone.** An area where, due to slope, fuel, weather, or other fire-related conditions, the potential loss of life and property from a fire necessitates special fire protection measures and planning before development occurs.

**Five-Minute Response Time.** The critical time period for responding to a structural fire. Temperatures reach a level sufficient to cause significant damage within five minutes of a fire's outbreak.

**Flood Control.** Measures that are taken to increase the hydrologic capacity of a natural watercourse or to create new manmade channels or reservoirs to drain and contain precipitation that otherwise exceeds the capacity of the watercourse, in an effort to reduce flood damage, usually to manmade improvements.

**Flooding.** A rise in the level of a water body or the rapid accumulation of runoff, including related mudslides and land subsidence, that results in the temporary inundation of land that is usually dry. Riverine flooding, coastal flooding, mud flows, lake flooding, alluvial fan flooding, flash flooding, levee failures, tsunamis, and fluvial stream flooding are among the many forms that flooding takes.

**Flood Insurance Rate Map (FIRM).** For each community, the official map on which the Federal Emergency Management Agency has delineated areas of special flood hazard and the risk premium zones applicable to that community.

**Flood, 100 Year.** Based on historical data, the magnitude of a flood expected to occur on the average every 100 years. Hence, the 100-year flood has a 1% chance of occurring in any given year.

**Floodplain.** The relatively level land area on either side of the banks of a stream regularly subject to flooding. The part of the floodplain subject to a 1% chance of flooding in any given year is the 100-year floodplain and is designated as an "area of special flood hazard" by the Federal Emergency Management Agency.

**Floodway.** The channel of a river or other watercourse and the adjacent land areas that must be reserved in order to discharge the "base flood" without cumulatively increasing the water surface elevation more than one foot. No development is allowed in floodways.

**Floor Area Ratio (FAR).** The gross floor area permitted on a site divided by the total net area of the site, expressed in decimals to one or two places. For example, on a site with 10,000 net square feet of land area, a floor area ratio of 1.0 will allow a maximum of 10,000 gross square feet of building floor area to be built.

**Food Security.** Physical and economic access by all people at all times to sufficient safe and nutritious food to meet their dietary needs and food preferences for an active and healthy life.



Foothills. Low hills near the base of a mountain or mountain range.

**Footprint; Building Footprint.** The area in square feet occupied by the base of a structure, not including upper floors.

**Fragmentation.** The process of fragmenting or isolating similar habitat types, or opportunities for plant and animal dispersal across the landscape. Habitat can be fragmented naturally or from construction of barriers such as roadways, intensive development, filling of creeks, and obstruction of wildlife movement corridors.

**Freeway.** A highway serving high-speed traffic with no crossings interrupting the flow of traffic (i.e., no crossings at grade).

**Fuel Break.** Strategically located modification zone, usually a long strip, where vegetative fuels are reduced in volume and maintained so as to produce a reduction of fire intensity if a wildfire burns into it.

**General Plan (The Marin Countywide Plan).** A compendium of a city's or a county's policies regarding its long-term development in the form of maps and accompanying text. The general plan is a legal document required of each local agency by the State of California Government Code Section 65301 and adopted by the governing body of the local jurisdiction. In California, the general plan has seven mandatory elements (circulation, conservation, housing, land use, noise, open space, and safety) and may include any number of optional elements (such as community facilities and parks and recreation).

**Geographic Information System (GIS).** A system of computer hardware, software, and procedures designed to support the capture, management, manipulation, analysis, modeling, and display of spatially referenced data for solving complex planning and management problems.

Geological. Pertaining to rock or solid matter.

**Geologic Review.** The analysis of geologic hazards, including all potential seismic hazards, surface ruptures, liquefaction, landsliding, and the potential for erosion and sedimentation.

**Goal.** An expression of community values and desired outcomes; a sought-after end state that is not quantifiable or time dependent.

Granny Flat. See "Second Unit."

Grasslands. Land reserved for pasturing or mowing in which grasses are the predominant vegetation.

**Graywater.** Household wastewater from baths, showers, non-kitchen sinks, and washing machines. Installing a water treatment system for treating graywater for reuse requires a permit from the State. Wastewater from toilet and kitchen sink is considered "blackwater" and goes directly into the sewage system because it can contain viruses, bacteria, high levels of organic waste, or harmful chemicals.



Greenbelt. Significant open space lands at the periphery of an urbanized area.

**Green Building.** A whole-systems approach to building design, construction, and occupancy. Site, energy, water, resources, materials, indoor air quality, and financial feasibility are all analyzed for environmental impact, health effects, and cost effectiveness.

**Green Business.** A business that has been certified for reducing energy use, water use, and waste generation.

Ground Failure. Mudslide, landslide, liquefaction, or soil compaction.

**Groundwater.** Water under the earth's surface, often confined to aquifers capable of supplying wells and springs.

**Groundwater Recharge.** The natural process of infiltration and percolation of rainwater from land areas or streams through permeable soils into water-holding rocks that provide underground storage (aquifers).

**Guidelines.** General statements of policy direction around which specific details may later be established.

**Habitat.** The physical location or type of environment in which an organism or biological population lives or occurs.

Habitat Diversity. The number of different types of plant and wildlife habitats within a given area.

**Hazardous Building.** A building that may be hazardous to life in the event of an earthquake because of partial or complete collapse. Hazardous buildings may include (1) those constructed prior to the adoption and enforcement of local codes requiring earthquake-resistant building design; (2) those constructed of unreinforced masonry; (3) those which exhibit any of the following characteristics: exterior parapets or ornamentation that may fall on passersby; exterior walls that are not anchored to the floors, roof, or foundation; sheeting on roofs or floors incapable of withstanding lateral loads; large openings in walls that may cause damage from torsional forces; lack of an effective system to resist lateral forces; or nonductile concrete frame construction.

**Hazardous Material.** Any substance that, because of its quantity, concentration, or physical or chemical characteristics, poses a significant present or potential hazard to human health and safety or to the environment it is released into, in the workplace or the environment. The term includes but is not limited to hazardous substances and hazardous wastes.

**Heavy Rail Transit.** Service generally operates along a fully grade-separated rail line that crosses over or below city streets.

**High Occupancy Vehicle (HOV).** Any vehicle with two or more persons (e.g., a carpool, a vanpool, a bus).



**High Occupancy Vehicle (HOV) Lane.** A lane of a freeway or highway reserved during certain hours for high occupancy vehicles.

Highway. High-speed, high-capacity, limited-access roadway serving regional and countywide travel.

**Historic, Historical.** A building or site that is noteworthy for its significance in local, state, or national history or culture; its architecture or design; or its works of art, memorabilia, or artifacts.

**Historic Preservation.** The preservation of historically significant structures and neighborhoods in order to facilitate restoration and rehabilitation of the building(s) to a former condition.

**HOME Investment Partnership Act.** A formula-based federal block grant program with funds to be spent only on housing and intended to provide incentives for the acquisition, construction, and rehabilitation of affordable rental and home ownership. HOME requires local governments to provide matching funds, though the matching ratio depends on the specific uses to which HOME funds are to be put.

**Homeless.** Persons and families who lack a fixed, regular, and adequate nighttime residence. The homeless include those staying in temporary or emergency shelters or who are accommodated with friends or other with the understanding that shelter is being provided as a last resort.

**Home Occupation.** The conduct of a business within a dwelling, or within an accessory building located on the same site as the dwelling, employing the occupant of the dwelling, with the business activity being subordinate to the residential use of the property.

**HOPWA (Housing for Persons with AIDS).** A federal government program that provides funds for the acquisition, rehabilitation, conversion, lease, and repair of facilities to provide housing and services for persons with AIDS.

Hotel or Motel. A facility in which guest rooms or suites are offered to the general public for lodging for compensation.

Household. Persons, related or unrelated, who occupy a single housing unit.

Housing and Community Development Department of the State of California (HCD). The agency responsible for reviewing city and county housing elements, for administering federal funds for non-entitlement jurisdictions, and for various State programs funded through housing bonds.

Housing and Urban Development, U.S. Department of (HUD). A cabinet-level department of the federal government that administers housing and community development programs.

**Housing Element.** One of the seven State-mandated elements of a local general plan. It assesses the existing and projected housing needs of all economic segments of the community; identifies potential sites adequate to provide the amount and kind of housing needed; and contains adopted goals, policies,



and implementation programs for the preservation, improvement, and development of housing. Housing elements must be updated every five years.

**Housing Overlay Designation.** A land use overlay designation to encourage and facilitate the development of affordable housing on select commercial, multifamily residential, and public properties that meet the criteria established by the overlay. Its purpose is to facilitate the review and approval of such housing and mixed-use developments. The overlay designation occurs in conjunction with the underlying land use designation.

**Housing Trust Fund.** A fund to provide money for development or rehabilitation of affordable housing with revenue from a variety of sources, which may include in-lieu fees, linkage fees, grants, contributions, or dedicated revenue from a tax or other governmental source.

**Housing Unit.** The place of customary abode of a person or family. A housing unit may be a single-family dwelling, a multi-family dwelling, a condominium, a modular home, a mobile home, a cooperative, or any other residential unit considered real property under State law. *See "Dwelling or Dwelling Unit," "Family," and "Household."* 

**Hydrocarbons.** A family of compounds containing carbon and hydrogen in various combinations, emitted into the atmosphere from manufacturing, storage and handling, or combustion of petroleum products, and through natural processes. Certain hydrocarbons interact with nitrogen oxides in the presence of intense sunlight to form photochemical air pollution.

**Hydrology.** A science dealing with the properties, distribution, and circulation of water on the surface of the land, in the soil and underlying rocks, and in the atmosphere.

**Impact Fee.** A development fee levied on the developer of a project by a city, county, or other public agency as compensation for otherwise unmitigated impacts produced by the project.

**Impervious Surface.** Surface through which water cannot penetrate, such as a roof, road, sidewalk, or paved parking lot. The amount of impervious surface increases with development and establishes the need for drainage facilities to carry the increased runoff.

Implementation. Actions, procedures, programs, or techniques that carry out policies.

**Inclusionary Housing/Zoning.** Programs that require a percentage of low and moderate income housing to be provided in new market-rate residential developments.

**Indicator.** A nonbinding measurement taken at regular intervals using a similar methodology that will assist in demonstrating movement toward or away from the goals and policies of the Countywide Plan.

**Infill Development.** Development of vacant land (usually individual lots or leftover properties) within areas that are already largely developed.



**Income Levels.** Income categories defined with respect to the area median income and adjusted for household size.

- Very Low Income Households. Households earning less than 50% of the median household income.
- Low Income Households. Households earning 50% to 80% of the median household income.
- *Lower Income Households.* Defined by California Housing Element law as households earning less than 80% of the median income.
- Moderate Income Households. Households earning 80% to 120% of the median income.
- Above Moderate Income Households. Households earning over 120% of the median household income.

**Income Limit (Housing).** Maximum amounts that low or moderate income families may earn to qualify for subsidized rental housing or a low-interest mortgage. Limits are calculated by the U.S. Department of Housing and Urban Development (HUD) and are based on family size and geographic location.

**Infrastructure.** Public services and facilities such as sewage disposal systems, water supply systems, other utility systems, and roads.

**Integrated Pest Management.** A decision-making process for managing pests that uses monitoring to determine pest injury levels, and combines biological, cultural, mechanical, physical, or chemical tools and other management practices to control pests in a safe, cost-effective, and environmentally sound manner that contributes to the protection of public health. This method uses extensive knowledge about pests, such as infestations, thresholds, life histories, environmental requirements, and natural enemies to complement and facilitate biological and other natural control of pests. The method involves the use of non-chemical pest control methods, and careful use of least toxic chemical methods when non-chemical methods have been exhausted or are not feasible.

Interjurisdictional. Coordination between two or more jurisdictions.

Intermodal Transit Hub. A junction where transfers can be made from one mode of transportation to another.

**In-Lieu Fee.** A fee paid to the County by developers in lieu of providing required on-site improvements, inclusionary units or lots, or parkland, in compliance with the Marin County Code.

**Institutional Use.** A publicly owned structure accommodating a public facility, or a private structure designed and operated as a church, hospital, school, or similar facility, that cannot be considered a residential, commercial, or industrial activity.



**Interest, Fee.** Entitles a landowner to exercise complete control over use of land, subject only to government land use regulations.

Interest, Less-Than-Fee. The purchase of interest in land rather than outright ownership; includes the purchase of development rights via conservation, open space, or scenic easements. See "Development Rights," "Easement, Scenic," and "Lease."

Intermittent Stream. See "Stream."

**Issue.** An important unsettled community matter or problem that is identified in a community's general plan and addressed by the plan's policies and implementation programs.

**Jobs/Housing Balance.** The number of persons working and living in a community. The balance is often expressed as a ratio.

**Jobs/Housing Ratio.** A numerical relationship between the number of jobs and the number of housing units in a jurisdiction.

**Jobs/Housing Linkage Programs.** Fees or incentives that local governments place on new nonresidential developments to offset the impact that new employment has on housing needs within a community.

**Joint Powers Authority (JPA).** A legal arrangement that enables two or more units of government to share authority in order to plan and carry out a specific program or set of programs that serves both units.

**Land Banking.** The purchase of land by a local government for use or resale at a later date. "Banked lands" have been used for development of low and moderate income housing, expansion of parks, and development of industrial and commercial centers.

Land Capability Classification (U.S. Natural Resources Conservation Service). A grouping of soils into classes (I–VIII), subclasses, and units according to their suitability for agricultural use, based on soil characteristics and climatic conditions.

Landmark. (1) A building, site, object, structure, or significant tree, having historical, architectural, social, or cultural significance and marked for preservation by the local, State, or federal government. (2) A visually prominent or outstanding structure or natural feature that functions as a point of orientation or identification.

**Landscaping.** Planting, including trees, shrubs, and ground covers, suitably designed, selected, installed, and maintained to enhance a site or roadway permanently.

Landslide. Downslope movement of soil, rocks, water, and debris, which typically occurs during an earthquake or after heavy rainfall.

Land Use. The purpose for which land or a building thereon is occupied.



**Land Use Regulation.** A term encompassing the regulation of land in general and often used to mean those regulations incorporated in the general plan as distinct from zoning regulations, which are more specific.

Ldn. Day-Night Average Sound Level. The A-weighted average sound level for a given area (measured in decibels) during a 24-hour period with a 10 dB weighting applied to nighttime sound levels. The Ldn is approximately numerically equal to the CNEL for most environmental settings. *See "Community Noise Equivalent Level (CNEL)."* 

**Lead Agency.** Under CEQA, the public agency principally responsible for carrying out or approving a project.

**Lease.** A contractual agreement by which an owner of real property (the lessor) gives the right of possession to another (a lessee) for a specified period of time (term) and for a specified consideration (rent).

**Level of Service (LOS).** A qualitative measure describing the efficiency of a traffic stream and the way such conditions are perceived by persons traveling in a traffic stream. Level of service measurements describe variables such as speed and travel time, freedom to maneuver, traffic interruptions, traveler comfort and convenience, and safety. Measurements are graduated, ranging from level of service A (representing free flow and excellent comfort for the motorist, passenger, or pedestrian) to level of service F (reflecting highly congested conditions for motorists and pedestrians with stop-and-go traffic that exceeds the capacities of streets). Levels of service can be determined for freeways, multilane highways, two-lane highways, signalized intersections, intersections that are not signalized, arterials, and transit and pedestrian facilities.

**Life Estate.** The right of an individual to use or get income from property as long as that person is alive. The specific rights in a life estate are specified in a trust document that establishes the life estate.

**Light Rail Transit.** Streetcars or trolley cars that typically operate in mixed traffic and in nonexclusive, at-grade rights-of-way. Vehicles are self-propelled by electricity or diesel power and usually operate in one- or two-car trains.

Linkage. See "Jobs/Housing Linkage Programs."

**Liquefaction.** The transformation of loose, wet soil from a solid to a liquid state often as a result of strong ground shaking during an earthquake.

**Local Agency Formation Commission (LAFCO).** A commission within each county that reviews and evaluates all proposals for formation of special districts, incorporation of cities, annexation to special districts or cities, consolidation of districts, and merger of districts with cities.

Local Coastal Element. The portion of a general plan applicable to the coastal zone.



# **Local Coastal Program.** A local government's program for the coastal zone, including land use plans, zoning ordinances, zoning district maps, and, within sensitive coastal resource areas, other implementing actions.

**Local Scenic Highway.** A segment of a state or local highway or street that a city or county has designated as "scenic."

Local Street. A street providing direct access to properties and designed to discourage through-traffic.

**Manufactured Housing.** Residential structures that are constructed entirely in the factory. Manufactured homes built under guidelines of the U.S. Department of Housing and Urban Development (HUD) and installed on a foundation must be subject to the same permit approval process and criteria as a conventional dwelling on the same lot.

Marin County Development Code. See "Development Code."

**Master Plan.** A development application and process for phased projects, projects involving multiple parcels, and other projects in which a coordinated development plan is appropriate, as required by the Development Code.

Mean Sea Level. Average altitude of the sea surface for all tidal stages.

Median Household Income. The point at which half of the County's households earn more and half earn less.

**Micro-hydro.** Micro-hydro turbines use the energy of falling water to create electricity. MMWD and NMWD have hydro-power potential at their reservoirs.

**Mineral Resource.** Land on which known deposits of commercially viable mineral or aggregate deposits exist. This designation is applied to sites determined by the State Division of Mines and Geology as a resource of regional significance and is intended to help maintain quarrying operations and protect the quarry from encroachment of incompatible land uses. Minerals are any naturally occurring chemical element or compound, or groups of elements and compounds, formed from inorganic processes and organic substances, including, but not limited to, coal, peat, and bituminous rock, but excluding geothermal resources, natural gas, and petroleum. Gold, sand, gravel, clay, crushed stone, limestone, diatomite, salt, borate, and potash are examples of minerals.

**Mini-Park.** A very small park, play space, or sitting area serving a neighborhood or housing development.

Mitigation Measure. An action or series of actions designed to avoid or reduce the adverse impact or effect of a development or capital project.

**Mixed Use.** Properties on which various uses, such as office, commercial, institutional, and residential, are combined in a single building or co-located on a single site in an integrated development project



**Mobile Home.** A trailer, transportable in one or more sections, that is certified under the National Manufactured Housing Construction and Safety Standards Act of 1974, and that is over 8 feet in width and 40 feet in length, with or without a permanent foundation and not including recreational vehicles, commercial coaches, or factory-built housing. A mobile home on a permanent foundation is considered a Single Family Dwelling. *(See "Manufactured Housing" and "Modular Unit.")* 

**Mode Split.** The percentage share of total trips for each mode of transportation, such as drive alone, carpool, or public transit. For example, an "18% transit share" means that transit is used for 18 out of 100 trips from home to work.

**Modular Unit.** A factory-fabricated, transportable building or major component designed for use by itself or for incorporation with similar units on-site into a structure for residential, commercial, educational, or industrial use. A modular unit does not have any chassis or permanent hitch to allow future movement. *(See "Mobile Home" and "Manufactured Housing.")* 

**Monitoring Plan.** A jurisdiction's plan and annual progress report for monitoring progress made toward meeting Housing Element goals. The current update is for the period January 1999 through June 30, 2006.

**Multi-Family Housing.** Attached dwelling units available to multiple households, including duplexes, triplexes, apartments, and condominiums.

**Multiple Family Building.** A detached building designed and used exclusively as a dwelling by three or more families occupying separate suites.

**National Environmental Policy Act (NEPA).** A federal law establishing national environmental policy, a council on environmental quality, and requirements for environmental impact statements for development projects.

**National Flood Insurance Program.** A national program that authorizes the sale of federally subsidized flood insurance in communities where such flood insurance is not available privately.

**National Historic Preservation Act.** A federal law that established a National Register of Historic Places and the Advisory Council on Historic Preservation, and that authorized grants-in-aid for preserving historic properties.

**National Register of Historic Places.** The official list, established by the National Historic Preservation Act, of sites, districts, buildings, structures, and objects significant in the nation's history or whose artistic or architectural value is unique.

**Natural Resource.** A material source of wealth, such as timber, fresh water, or a mineral deposit, that occurs in a natural state.



**Natural Resource, Nonrenewable.** An inanimate resource that does not increase significantly with time and the use of which diminishes the total stock (e.g., minerals, fossil fuels, and fossil water).

**Natural Resource, Renewable.** A resource that can be replaced by natural ecological cycles or sound management practices (e.g., forests and plants).

**Neighborhood Park.** A park with a minimum size of 5 to 20 acres serving one or more neighborhoods with a population of 2,000 to 5,000 within a radius of one-half mile.

Nexus. A connection or linkage.

**Nitrogen Oxide.** A reddish brown gas that is a byproduct of combustion and ozone formation processes. Often referred as NO<sub>X</sub>, this gives smog its "dirty air" appearance.

Node. A hub of activity.

**Noise.** Any sound that is undesirable because it interferes with speech and hearing or is intense enough to damage hearing or is otherwise annoying.

**Noise Attenuation.** Reduction of the level of a noise source using a substance, material, or surface, such as earth berms and/or solid concrete walls.

Noise Contour. A line connecting points of equal noise level as measured on the same scale.

**National Scenic Byway.** A segment of a state or Interstate highway route that the United States Forest Service has designated as a scenic byway or that another federal agency has designated as a national scenic and recreational highway.

Nonrenewable Natural Resource. See "Natural Resource, Nonrenewable."

**Office Use.** The use of land by general business offices, medical and professional offices, or administrative or headquarters offices for large wholesaling or manufacturing operations, and research and development.

**Official County Scenic Highway.** A segment of a state highway identified in the Master Plan of State Highways eligible for Official Scenic Highway designation and designated by the director of the State Department of Transportation (Caltrans).

**Open Coastal Waters and Coastal Waters.** The open ocean overlying the continental shelf and its associated coastline. Salinities exceed 30 parts per thousand with little or no dilution except opposite mouths of estuaries.

**Ordinance.** A law or regulation set forth and adopted by a governmental authority, usually a city or county.



**Overlay.** A designation on the general plan land use map or zoning map that modifies the basic underlying designation in a specific manner.

**Ozone.** A tri-atomic form of oxygen  $(O_3)$  created naturally in the upper atmosphere by a photochemical reaction with solar ultraviolet radiation. In the lower atmosphere, ozone is a recognized air pollutant that is not emitted directly into the environment but is formed by complex chemical reactions between oxides of nitrogen and reactive organic compounds in the presence of sunlight, and becomes a major agent in the formation of smog.

**Paratransit.** Transportation systems such as jitneys, carpools, van pools, taxi service, and dial-a-ride arrangements.

**Parcel.** A lot or contiguous group of lots in single ownership or under single control that may be considered a unit for purposes of development.

Park. Open space land primarily for recreation.

**Parkway.** (1) An expressway or freeway designed for noncommercial traffic only, usually located within a strip of landscaped park or natural vegetation. (2) A landscaped strip between the curb and sidewalk, primarily found in residential neighborhoods for the purpose of planting street trees.

**Peak Hour/Peak Period.** A daily period during which traffic volume is highest on a roadway, usually in the morning and evening commute periods.

**Peak Load Water Supply.** The supply of water available to meet both domestic water and firefighting needs during the particular season and time of day when domestic water demand on a water system is at its peak.

**Permaculture.** An ecological design science that works on a restorative approach to watersheds, natural building, agriculture, wildlife, economics, and community.

**Permitted Use.** A use of land that is allowed within a particular general plan land use or zoning designation.

**Persons with Disabilities.** People with a wide range of physical and mental conditions that may affect their ability to function independently.

Persons per Household. The average number of persons in a household.

**Pesticide.** Any substance or mixture of substances intended for preventing, destroying, repelling, or mitigating any pest. The term *pesticide* applies to insecticides, herbicides, fungicides, and various other substances used to control pests. Under United States law, a pesticide is also any substance or mixture of substances intended for use as a plant regulator, defoliant, or desiccant.



**Photovoltaics.** Solar electric systems that use the sun's energy to generate electricity by moving electrons.

**Planning and Research, Office of (OPR).** A governmental division of the State of California that has among its responsibilities the preparation of a set of guidelines for use by local jurisdictions in drafting general plans.

**PM-2.5 and PM-10.** Inhalable particulate matter, which refers to a wide variety of solid or liquid particles in the atmosphere, including smoke, dust, aerosols, and metallic oxides. **PM-2.5** is particulate matter 2.5 microns or less in diameter, and **PM-10** is particulate matter 10 microns or less in diameter.

**Policy.** A statement derived from a goal that represents the jurisdiction's adopted position and guides action by decision-making bodies.

**Pollutant.** Any introduced gas, liquid, or solid that makes a resource unfit for its normal or usual purpose.

**Pollution.** The presence of matter or energy whose nature, location, or quantity produces undesired environmental effects.

**Pollution, Non-Point.** Sources for water pollution contributed from many small sources that cannot be easily identified individually but collectively degrade water quality.

**Pollution, Point.** In reference to water quality, a discrete source from which pollution is generated before it enters receiving waters, such as a sewer outfall or an industrial waste pipe.

**Poverty Level.** As defined by the U.S. Census, families and unrelated individuals are classified as being above or below the poverty level based on a poverty index that provides a range of income cutoffs or "poverty thresholds" varying by size of family, number of children, and age of householder. The income cutoffs are updated each year to reflect the change in the Consumer Price Index.

**Precautionary Principle.** A principle that, when an activity raises threats of harm to human health or the environment, precautionary measures should be taken to eliminate or minimize potential hazards and their onset. In this context, the proponent of an activity rather than the public should bear the burden of proof.

**Prescriptive Right.** A right of public access that is acquired over private lands through use and is essentially an easement over real property that comes into being without the explicit consent of the property owner. The term recognizes that the use must continue for the length of the prescriptive period before a public easement comes into being. In California the prescriptive period is five years.

**Preserve.** (Noun.) An area in which beneficial uses in their present condition are protected, such as a nature preserve or an agricultural preserve. *See "Agricultural Preserve."* 



**Prime Farmland Soils.** Soils with the best combination of physical and chemical features able to sustain long-term production of agricultural crops. These soils have the quality needed to produce sustained high yields given an appropriate growing season and adequate moisture supply.

**Program (or "Implementing Action").** A specific action, procedure, or technique used to carry out a Countywide Plan policy.

**Quimby Act.** A provision of the California Government Code (Section 66477) that permits a local jurisdiction to require the dedication of land for neighborhood and community recreation and/or the payment of an in-lieu fee as a condition of approval of a tentative or parcel map.

**Recreation, Active.** A type of recreation or activity that requires the use of organized play areas such as softball, baseball, football, and soccer fields; tennis and basketball courts; and children's play equipment.

**Recreational Trails.** Public areas that include pedestrian trails, bikeways, equestrian trails, boating routes, trails, and areas suitable for use by physically handicapped people; trails and areas for off-highway recreational vehicles; and cross-country skiing trails.

**Recreation, Passive.** A type of recreation or activity that does not require the use of organized play areas.

Recycle. The process of extraction and reuse of materials from waste products.

**Redevelop.** To demolish existing buildings or to increase the overall floor area existing on a property or both, irrespective of whether a change in land use occurs.

Redevelopment Agency. See "Community Redevelopment Agency."

**Redevelopment Project.** An activity undertaken by a redevelopment agency set up under California law to revitalize blighted areas as defined by the Health and Safety Code.

**Regional.** Activities or economies at a scale greater than that of a single jurisdiction and affecting a broad homogeneous area (e.g., the Bay Area region, which comprises the nine counties surrounding the San Francisco Bay).

**Regional Housing Needs.** A quantification by a regional or State agency of existing and projected housing need, by household income group, for all localities within a region.

**Regional Park.** A park typically 150 to 500 acres in size focusing on activities and natural features not included in most other types of parks, and often based on a specific scenic or recreational opportunity.

Rehabilitation. The repair, preservation, and/or improvement of substandard buildings.



**Renewable Energy.** Energy generated from naturally renewable sources such as the sun, wind, water, and earth's heat. Examples of renewable energy systems include solar thermal, photovoltaics (solar electric), wind energy conversion systems (WECS), geothermal (earth heating), hydroelectric under 30 kW, tidal power, and biodiesel.

Renewable Natural Resource. See "Natural Resource, Renewable."

**Residential.** Land designated in a general plan and zoning ordinance for buildings consisting of dwelling units.

Residential (Multiple Family). Usually three or more dwelling units on a single site.

Residential (Single-Family). A single dwelling unit on a building site.

**Responsible Agency.** As defined by CEQA guidelines, responsible agencies are public agencies other than the lead agency with the power of discretionary approval.

**Restoration.** Actions taken to improve a degraded ecosystem to achieve a desired, healthy, and functioning condition.

**Retrofit.** To add materials and/or devices to an existing building or system to improve its operation or efficiency.

**Revegetation.** The reestablishment of plant cover by either natural or artificial means, such as reseeding.

**Rezoning.** An amendment to the map and/or text of a zoning ordinance to effect a change in the nature, density, or intensity of uses allowed in a zoning district and/or on a designated parcel or land area.

**Richter Scale.** A measure of the size or energy release of an earthquake at its source. Because the scale is logarithmic, the wave amplitude of each number on the scale is 10 times greater than that of the previous whole number.

Rideshare. A travel mode other than driving alone, such as buses, rail transit, carpools, and vanpools.

**Ridge and Upland Greenbelt.** In the eastern portions of the county, the uppermost portions of ridges and hills, and associated wooded hillsides, identified in the Community Design Section of the Built Environment Element. *(See Map 3-4.)* 

**Ridgeline.** A line connecting the highest points along a ridge and separating drainage basins or small-scale drainage systems from one another.

**Right-of-Way.** A strip of land occupied or intended to be occupied by certain transportation and public use facilities, such as roadways, railroads, and utility lines.

Riparian. Associated with or dependent upon a river, stream, or other water body.



**Riparian Habitats.** Areas of riparian vegetation that are characterized by plant species that occur along and adjacent to fresh water courses, including perennial and intermittent streams, lakes, springs, and other bodies of fresh water. Riparian habitats include transitional zones between land and water and are distinguished by characteristic woody trees and shrubs, a variety of important ecological functions, and generally high wildlife habitat values.

**Riparian Lands.** The vegetative and wildlife areas adjacent to perennial and intermittent streams that are delineated by the existence of plant species normally found near fresh water.

**Riparian Vegetation.** Vegetation associated with a watercourse and relying on the higher level of water provided by the watercourse. Riparian vegetation can include trees, shrubs, and/or herbaceous plants. *Woody riparian vegetation* includes plants that have tough, fibrous stems and branches covered with bark and composed largely of cellulose and lignin. *Herbaceous riparian vegetation* includes grasses, sedges, rushes and forbs – broad-leaved plants that lack a woody skeleton.

**Runoff.** The portion of rain or snow that does not percolate into the ground and is discharged into streams.

**Rural Expressway.** A roadway in a non-urbanized area, which may or may not have multiple lanes in each direction, and designed for higher travel speeds with limited cross traffic and driveways.

Sanitary Sewer. A system of subterranean conduits that carries refuse liquids or waste matter to a facility where the sewage is treated. *See "Septic System" and "Storm Drainage."* 

**Satellite Work Center.** A remote office between residential neighborhoods and business locations established by a single employer or a group of employers where employees can work at the beginning and end of the day and thereby avoid commuting to the main office during hours of heavy traffic.

Scenic Easement. See "Easement, Scenic."

**Scenic Highway Corridor.** The visible area outside the highway's right-of-way, generally described as "the view from the road."

**Scenic Highway/Scenic Route.** A highway, road, drive, or street that in addition to its transportation function provides opportunities for the enjoyment of natural and manmade scenic resources, and access or direct views to areas or scenes of exceptional beauty or historic cultural interest.

Section 8 Rental Assistance Program. A federal rent subsidy program administered by the Department of Housing and Urban Development (HUD), which is one of the main sources of federal assistance for low income households. The program operates by providing "housing assistance payments" to owners, developers, and public housing agencies to make up the difference between the "fair market rent" of a unit (set by HUD) and the household's contribution toward the rent, which is calculated at 30% of the household's adjusted gross monthly income. Section 8 includes programs for new construction, existing housing, and substantial or moderate housing rehabilitation.



**Sea.** The Pacific Ocean and all harbors, bays, channels, estuaries, salt marshes, sloughs, and other areas subject to tidal action through any connection with the Pacific Ocean.

**Second Unit.** A self-contained living unit, either attached to or detached from, and in addition to, the primary residential unit on a single lot. Sometimes called a *granny flat* or *in-law unit*.

Sediment. Solid material settled from suspension in a liquid.

**Seiche.** An earthquake-induced wave in an enclosed body of water such as a lake, reservoir, bay, or harbor.

**Seismically Induced Surface Rupture.** A break in the ground's surface and associated deformation resulting from the movement of a fault.

**Seismic Hazard Zone.** A regulatory zone, delineated by the State Geologist, within which site-specific geologic, soils, and foundation engineering studies are required to identify and avoid earthquake-caused ground-failure hazards, or selected other earthquake hazards, prior to subdivision of land and for construction of most structures for human occupancy.

**Senior Housing.** Typically one- and two-bedroom apartments or condominiums designed to meet the needs of persons 62 years of age and older, and restricted to occupancy by them.

**Sensitive Coastal Resource Areas.** Identifiable and geographically bounded land and water areas within the coastal zone of vital interest and sensitivity. "Sensitive coastal resource areas" include the following: (a) special marine and land habitat areas, wetlands, lagoons, and estuaries; (b) areas possessing significant recreational value; (c) highly scenic areas; (d) archaeological sites referenced in the California Coastline and Recreation Plan or as designated by the State Historic Preservation Officer; (e) special communities or neighborhoods that are significant visitor destination areas; (f) areas that provide existing coastal housing or recreational opportunities for low and moderate income persons; (g) areas where divisions of land could substantially impair or restrict coastal access.

**Sensitive Receptor.** A facility in which a number of individuals are highly susceptible to the adverse effects of air pollutants or noise.

**Sensitive Resources.** Wildlife species that are officially designated as rare, threatened, or endangered, or as a species of special concern, pursuant to federal or State law, and sensitive wildlife habitats.

**Sensitive Wildlife Habitats.** Areas designated, precisely mapped, and officially adopted, pursuant to federal, State, or local law, that are essential to maintenance of native wildlife populations, providing breeding habitat, protective cover, and/or movement opportunities for dispersal, retreat, or foraging activities. Sensitive wildlife habitats include but are not limited to riparian corridors, estuaries, marshes, and other wetlands, nesting and roosting locations, and overwintering areas for migratory species.

**Septic System.** An on-site sewage disposal system consisting of a septic tank, and a soil infiltration leach field, evapotranspiration mound, or other approved disposal facility.



**Setback.** The distance by which a structure must be separated from a lot line. Setbacks from private streets and driveways are measured from the edge of the easement.

**Settlement.** (1) A drop in elevation of a ground surface caused by settling or compacting. (2) A gradual downward movement of an engineered structure due to compaction. Differential settlement is uneven settlement where one part of a structure settles more or at a different rate than another part.

**Shared Housing.** Shared housing generally means an arrangement in which two or more unrelated people, each with private sleeping quarters, share a house or an apartment for the purpose of social contact, mutual support, and assistance, to pursue a common purpose and/or to reduce housing expenses.

**Significant Effect.** A beneficial or detrimental impact on the environment, including but not limited to changes in an area's air, water, and land resources.

**Siltation.** (1) The accumulating deposition of eroded material. (2) The gradual filling in of streams and other bodies of water with sand, silt, and clay.

**Single-Family Dwelling (Attached).** A dwelling unit occupied or intended for occupancy by only one household that is structurally connected with at least one other such dwelling unit, including duets and townhomes.

**Single-Family Dwelling (Detached).** A dwelling unit occupied or intended for occupancy by only one household that is structurally independent from any other such dwelling unit or structure intended for residential or other use, excluding a second unit.

**Single Room Occupancy (SRO) Hotels and Efficiency Apartments.** Types of affordable private housing for single and elderly low income people and for new arrivals to an area. An SRO room usually is small, between 80 and 250 square feet. It typically has a sink and a closet but shares a bathroom, shower, and kitchen with other rooms. An efficiency apartment also is small but contains a small cooking area and bathroom.

**Site Assessment.** An analysis of the environmental setting of developed or undeveloped land, including but not limited to sensitive wildlife habitats and sensitive resources, such as baylands, wetlands, stream and riparian systems, and special-status species and species of concern. A site assessment may also include findings regarding potential environmental effects resulting from a development application, and recommendations for measures that may avoid or minimize such effects.

**Slope.** Land gradient described as the vertical rise divided by the horizontal run and expressed in percent.

**Slope Stability.** The likelihood of slope failure (a landslide) based on five factors: slope angle, soil characteristics, degree of saturation, human activity, and seismic activity.



**Smart Growth.** Refers to strategies that encourage development in existing communities where schools, shops, and transit already exist while conserving open space, farmland, and natural habitat. Through this model, medium to higher density and mixed-use development are encouraged along current and future transit corridors as well as downtown areas. Putting housing close to transit and shops is especially important for lower income workers, seniors, and others who can't drive or do not own a car. And increasing densities means transit systems will be better utilized.

**Soil.** The unconsolidated material on the immediate surface of the earth created by natural forces, which serves as a natural medium for growing land plants.

**Solar.** Solar energy uses the sun's energy to provide heat, light, hot water, and electricity for homes, businesses, and industry.

**Solar Access.** The provision of direct sunlight to an area specified for solar energy collection when the sun's azimuth is within 45 degrees of true south.

**Solar System, Active.** A system using a mechanical device, such as a pump or fan, and energy in addition to solar energy to transport a conductive medium (air or water) between a solar collector and the interior of a building for the purpose of heating or cooling.

**Solar System, Passive.** A system that uses direct heat transfer from thermal mass instead of mechanical power to distribute collected heat. Passive systems rely on building design and materials to collect and store heat and to create natural ventilation for cooling.

**Solid Waste.** Unwanted or discarded material that is not a liquid or gas, including organic wastes, paper products, metals, glass, plastics, cloth, brick, rock, soil, leather, rubber, yard wastes, and excluding sewage and hazardous materials.

**Special Districts.** Districts established under California law and through LAFCO in order to provide special services, such as water and sewer, street lighting, and other services, to the residents within the district boundaries.

**Special Needs.** Persons who are physically, mentally, and/or developmentally disabled; victims of domestic violence; homeless persons or those at risk of becoming homeless, including youth; chronic substance abusers; individuals exiting from institutional settings; chronically ill persons; persons disabled by HIV/AIDS or mental illness; and displaced teenage parents (or expectant teenage parents).

**Special-Status Species.** A species of animal or plant that is (1) listed in Title 50, Code of Federal Regulations, Section 17.11 or Section 17.2, pursuant to the Federal Endangered Species Act designating species as rare, threatened, or endangered; (2) listed in Sections 670.2 or 670.5, Title 14, California Administrative Code; or (3) considered rare or endangered under Section 15380 of the CEQA Guidelines.

**Specific Plan.** A detailed plan for the systematic implementation of the general plan, for all or part of the area covered by the general plan, as authorized by Government Code Sections 65450, et seq.



**Sphere of Influence (SOI).** The probable ultimate physical boundaries and service area of a local agency (city or district) as determined by the Local Agency Formation Commission (LAFCO) of a county.

**State.** State with a capital *S* refers to the State of California. *State* with a lowercase *s* refers to the geographical area of California.

**Storm Drainage.** A system of subterranean conduits that carries storm water runoff to an outfall in a body of water.

**Storm Water Runoff.** Surplus surface water generated by rainfall that does not seep into the earth but flows overland to flowing or stagnant bodies of water.

**Stream.** A natural or once natural flowing open drainage channel with an established bed and bank. These consist of perennial, intermittent, and ephemeral streams, including open waterways that have been restored, modified, or channelized, but do not include ditches, culverts, or other above- or below-ground conduits constructed specifically for storm drainage function. Perennial and intermittent streams, shown as solid or dashed blue lines (or purple lines) on the most recent appropriate USGS data, and ephemeral streams as defined below, are subject to Stream Conservation Area protection policies. *See "Stream Conservation Area (SCA)."* 

**Stream Conservation Area (SCA).** A setback from the bank of a natural watercourse, which is intended to protect the active channel, water quality, and flood control functions and associated fish and wildlife habitat values along streams.

**Stream, Ephemeral.** A watercourse that carries only surface runoff and flows during and immediately after periods of precipitation.

**Stream, Intermittent.** A watercourse that is temporally intermittent or seasonal and that flows during the wet season, continues to flow after the period of precipitation, and ceases surface flow during at least part of the dry season. Intermittent streams are typically shown as a dashed blue line on USGS quadrangle maps.

**Stream, Perennial.** A watercourse that flows throughout the year (except for infrequent or extended periods of drought), although surface water flow may be temporarily discontinuous in some reaches of the channel, such as between pools, typically shown as a solid blue line on USGS quadrangle maps. (Perennial streams can be spatially intermittent but flow all year.)

**Streets, Local.** Streets not shown on the Countywide Plan circulation map or diagram with the primary purpose of providing access to fronting properties.

**Streets, Major.** The transportation network that includes a hierarchy of freeways, arterials, and collectors servicing through traffic.



**Subdivision.** The division of any unit or portion of land shown on the latest equalized Marin County assessment roll as a unit or contiguous units, even if it is separated by roads, streets, utility easement, or railroad rights-of-way. Subdivision includes a condominium project, as defined in Section 1351 (f) of the Civil Code, and a community apartment project, as defined in Section 1351 (d) of the Civil Code.

**Subdivision Map Act.** Division 2 (Sections 66410, et seq.) of the California Government Code that vests in local legislative bodies the regulation and control of the design and improvement of subdivisions, including the requirement for tentative and final maps.

**Subsidence.** The gradual, local settling or sinking of the earth's surface with little or no horizontal motion. *See "Settlement."* 

**Succession.** The process of replacement of one plant community with another. Conditions of the prior plant community (or successional stage) create conditions that are favorable for the establishment of the next state. Human-induced changes, such as fire suppression or vegetation management, can sometimes alter the process of natural succession.

**Sustainability.** For the purpose of the Countywide Plan, sustainability is defined as aligning our built environment and socioeconomic activities with the natural systems that support life. In the long run, sustainability means adapting human activities to the constraints and opportunities of nature. Central to this definition is meeting the needs of both the present and the future.

**Target.** A nonbinding, quantifiable outcome that has been identified to help measure progress in achieving the objectives of the Countywide Plan.

**Telecommunications.** The transmission of information from one point to one or more points using radio frequency signals.

**Telecommuting.** Working at home or in a location other than the primary place of work and communicating with the workplace and conducting work via wireless or telephone lines, using moderns, fax machines, or other electronic devices in conjunction with computers.

**Tidal.** Tidal energy systems use the energy of waves, rising/falling tides, or the flow of water through a venturi to power a turbine. San Francisco is pursuing a tidal energy system, and Marin is exploring the idea with them.

**Tidelands.** Lands regularly subject to tidal action, between lowest low water and highest high water but exclusive of lands within floodplains, and subject to inundation due to potential upstream events.

Top of Bank. The elevation at which flow spills out of a stream channel and onto the floodplain.

**Topography.** Configuration of a surface, including its relief and the position of natural and built features.



**Total Maximum Daily Load (TMDL).** The amount of a contaminating substance allowed in a body of water according to criteria of the Federal Water Pollution Control Act. The TMDL regulations are designed to limit contaminants flowing into the San Francisco Bay Estuary.

**Townhouse; Townhome.** A one-family dwelling in a row of at least three such units in which each unit has its own front and rear access to the outside, no unit is located over another unit, and each unit is separated from any other unit by one or more common and fire-resistant walls. Townhouses usually have separate utilities; however, in some condominium situations, common areas are serviced by utilities purchased by a homeowners association on behalf of all townhouse members of the association.

**Traffic Mitigation Fee.** A fee levied on new development projects to collect funds to provide each development's fair share of traffic facilities needed to serve all new development.

**Traffic Model.** A mathematical representation of traffic movement within an area or region based on observed relationships between the kind and intensity of development in specific areas.

Trail. An unpaved, single-track pathway, generally traversing parklands or open space.

Trailhead. Trail connection to paved roadways.

**Transfer of Development Rights (TDR).** The process established by Chapter 22.34 (Transfer of Development Rights), which allows some or all of the number of dwelling units potentially allowed by the zoning applicable to a "donor" site to be transferred and built on another "receiving" site, in addition to the number of units potentially allowed by the zoning of the receiving site. *See "Density Transfer."* 

**Transitional Housing.** Shelter provided to the homeless for an extended period, often as long as 18 months, and generally integrated with other social services and counseling programs to assist in the transition to self-sufficiency through the acquisition of a stable income and permanent housing.

**Transit-Oriented Development (TOD).** Moderate to higher density development, located within easy walking distance of a major transit stop, generally with a mix of residential, employment, and shopping opportunities designed for pedestrians without excluding the auto. TOD can be new construction or redevelopment of one or more buildings whose design and orientation facilitate transit use.

**Transit, Public.** A system of regularly scheduled buses, trains, and/or ferryboats available to the public on a fee-per-ride basis.

**Transportation Demand Management (TDM).** A strategy for reducing demand on the road system by reducing the number of vehicles using the roadways and/or increasing the number of persons per vehicle. The **TDM** reduces the number of persons who drive alone on the roadway during the commute period and increases the number in carpools, in vanpools, in buses and trains, walking, and biking.

Transportation System Modeling. See "Traffic Model."



**Transportation Systems Management (TSM).** A comprehensive strategy developed to address the problems caused by additional development, increasing trips, and a shortfall in transportation capacity. TSM focuses on more efficiently utilizing existing highway and transit systems rather than expanding them. TSM measures are characterized by their low cost and quick implementation time frame, and may include computerized traffic signals, metered freeway ramps, and one-way streets.

**Trip.** A one-way journey that proceeds from an origin to a destination via a single mode of transportation; the smallest unit of movement considered in transportation studies. A trip has one "production end" (an origin often from home) and an "attraction end" (destination). *See "Traffic Model."* 

**Trip Generation.** The dynamics that account for people making trips in automobiles or by means of public transportation. Trip generation is the basis for estimating the level of use for a transportation system and the impact of additional development or transportation facilities on an existing local transportation system. Household trip generation is correlated with destinations that attract household members for specific purposes.

**Tsunami.** A wave, commonly called a tidal wave, caused by an underwater seismic disturbance, such as sudden faulting, landslide, or volcanic activity.

Uniform Building Code (UBC). A national, standard code that sets forth minimum standards for construction.

**Uniform Housing Code (UHC).** State housing regulations governing the condition of habitable structures with regard to health and safety standards and that provide for the conservation and rehabilitation of housing in accordance with the Uniform Building Code (UBC).

**Unincorporated Community.** A concentration of structures and population within the unincorporated areas of the county identified in the Countywide Plan as a community.

**United States Geological Survey (USGS).** An agency of the federal government that is the source for science about the earth, its natural and living resources, natural hazards, and the environment.

**Universal Design.** Consists of operating principles that promote the design of buildings and spaces to meet the needs of all people, young and old, abled and disabled, by creating comfortable surroundings that suit a lifetime of changing needs for the occupants. Universal design features are generally standard building products or features that that have been installed differently, selected carefully, or modified to allow ease of use by a larger population, including children, older people, and people with disabilities. Some examples of universal building design include eliminating steps at building entrances, designing wider doorways, using lever or loop-type handle designs for doors and drawers that require no gripping or twisting to operate, and placing electrical receptacles higher than usual above the floor.

**Urban Service Area.** An area that represents a legal, orderly expansion of urban development patterns and where a city or town is able to provide urban services.



Use Permit. The discretionary and conditional review of a land use on a site or in a building or facility.

User Fee. A fee levied on the user of a service or amenity.

**Vehicle Miles Traveled (VMT).** A key measure of overall street and highway use. Reducing VMT is often a major objective in efforts to reduce vehicular congestion and achieve regional air quality goals.

**View Corridor.** The line of sight – identified as the height, width, and distance – of an observer looking toward an object of significance to the community; the route that directs the viewer's attention.

Viewshed. The area within view from a defined observation point.

**Volume-to-Capacity Ratio (V/C).** A measure of the operating capacity of a roadway or intersection, in terms of the number of vehicles passing through divided by the number of vehicles that theoretically could pass through when the roadway or intersection is operating at its designated capacity. At a V/C ratio of 1.0 the roadway or intersection is operating at capacity. If the ratio is less than 1.0, the traffic facility has additional capacity. A ratio greater than 1.0 will cause the peak hour of traffic to elongate into a *peak period. See "Peak Hour/Peak Period" and "Level of Service."* 

Wastewater Irrigation. The process by which wastewater that has undergone primary, secondary, or tertiary treatment is used to irrigate land.

Watercourse. Natural or once-natural flowing (perennially or intermittently) water, including rivers, streams, and creeks. Includes natural waterways that have been channelized but does not include ditches, culverts, or other above- or below-ground conduits constructed for storm drainage function and sewage systems.

Watershed. The total area above a given point on a watercourse that contributes water to the flow of the watercourse; the entire region drained by a watercourse.

Weatherization. Provision of insulation, weatherstripping, caulking, and other improvements to reduce heat loss from buildings.

Wetland, Jurisdictional. An area that meets the criteria established by the U.S. Army Corps of Engineers (Corps or COE) for a Wetlands (as set forth in their Wetlands Delineation Manual). Such areas come under the jurisdiction of the Corps of Engineers for permitting certain actions such as dredge and fill operations.

Wetlands. Areas inundated or saturated by surface or groundwater at a frequency and duration sufficient to support and, under normal circumstances, that do support a prevalence of vegetation typically adapted for life in saturated soil conditions. These can include saltwater and freshwater marshes, brackish marshes, swamps, vernal pools, and seasonal wetlands. The Army Corps of Engineers and/or Regional Water Quality Control Board have jurisdiction over wetlands and unvegetated "other waters," which can include mudflats, lakes, ponds, and open waters of bays, lagoons, and ocean. The California Department of Fish and Game has jurisdiction over wetlands and other



waters associated with the bed and bank of creeks and streams, and the shoreline and open water habitat of nontidal water bodies. In the Coastal Zone, wetlands can include the presence of hydrophytes (plants typically found in wet habitats) and hydric (wet) soils.

**Wildland Fire.** A fire occurring in a suburban or rural area that contains uncultivated lands, timber, range, watershed, brush, or grasslands. This includes areas where there is a mingling of developed and undeveloped lands.

Wildlife Corridor. A continuous land area or natural feature such as a stream, shoreline, or ridge top used for wildlife movement.

Wildlife Nursery Area. Locations used for breeding and rearing of young by fish and wildlife.

Wildlife Refuge. An area maintained in a natural state for the preservation of both animal and plant life.

Williamson Act (California Land Conservation Act of 1965). Formally the California Land Conservation Act of 1965, this Act was designed as an incentive to retain prime agricultural land and open space in agricultural use, thereby slowing its conversion to urban and suburban development. The program entails a 10-year contract between the County and an owner of land whereby the land is taxed on the basis of its agricultural use rather than the market value. The land becomes subject to certain enforceable restrictions, and certain conditions need to be met prior to approval of an agreement.

**Wind.** Wind generators are turbines, which use the energy in the motion of the wind to make mechanical energy, which is then converted to electrical energy. Wind is the least-expensive method of generating electricity, and there is enough potential wind energy in the United States to power the entire country. (NREL)

Wind Energy Conversion System. Use of wind as an energy resource through a generation and conversion system such as a windmill.

**Workforce Housing.** Housing affordable to those in public service or quality of life occupations that are bound to a community and who earn less than 120% of the Area Median Income as defined by the U.S. Department of Housing and Urban Development.

**Xeriscape.** A sustainable landscape that conserves water and is based on sound horticultural practices. The word is derived from the Greek word *xeros*, meaning "dry," and *scape*, from the word *landscape*.

**Zoning.** The division of a city or county by legislative regulations into zones, which specify allowable uses for real property and size restrictions for buildings within these areas. Zoning implements land use policies of the Countywide Plan.

**Zoning District.** An area identified on the County Zoning Map within which certain uses of land and structures are permitted, and regulations are defined and specified by the Development Code, Title 22 of the Marin County Code.



**Zoning, Incentive.** The awarding of bonus credits to a development in the form of allowing more intensive use of land if public benefits — such as preservation of greater than the minimum required open space, provision for low and moderate income housing, or plans for public plazas and courts at ground level — are included in a project.

Zoning, Inclusionary. See "Inclusionary Housing/Zoning."

**Zoning Map.** The official map or maps of Marin County that identify the specific zoning districts located in the unincorporated areas of the county. The Zoning Map is on file with the Marin County Community Development Agency.

**Zoning Ordinance.** An ordinance authorized by Section 65850 of the Government Code, or, in the case of a charter city, a similar ordinance enacted pursuant to the authority of its charter. [California Coastal Act §30122.]



community meetings is scheduled for June 1 at a location that has yet to be determined. Meeting information will be made available soon on the Marin Flood District <u>website's project page</u>.

Meanwhile, a local task force will assure that community interests and concerns are taken into account during the 18-month development of the stormwater plan. The task force was created in collaboration with the Marin City community and includes representatives from the Gateway Shopping Center (which manages the Marin City pond adjacent to southbound Highway 101), the Marin City Community Development Corporation, the Marin City Community Services District, the Sausalito Marin City School District, Caltrans, Coast Miwok Tribal Council of Marin,



During storm season and high tides, Marin City is regularly impacted by flooding tied to challenging drainage issues. A new stormwater plan will incorporate community feedback, and public meetings begin soon. (Google Earth photo)

Federated Indians of Graton Rancheria, Floating Homes Association, the National Park Service Golden Gate National Recreation Area, Marin Water, and Marin County Parks. Residents representing upper and lower watershed property owners, local students, and young people have been invited to the task force as well.

There will be a total of five task force meetings and three community meetings over the 18-month process. The first task force meeting is scheduled for Monday, April 3. Outcomes of the task force meetings and community meetings will be posted on the District <u>website's project page</u>.

"Public safety and awareness are top priorities, so community engagement is key to the success of the stormwater plan," said Berenice Davidson, Assistant Director of Marin County Department of Public Works and the District. "The task force is going to ensure that community interests and concerns are being heard and that potential design options are in alignment with that community input. The heightened focus on community engagement is what makes this plan distinct from past studies. By community, we mean property owners, residents, nonprofits and public agencies; collaboration is crucial."

#### Community Input Key to Marin City Stormwater Plan

The geography of the watershed creates complex challenges for developing flood risk reduction measures. During storms, rainfall channels its way from the surrounding Marin Headlands into Marin City and out to Richardson Bay. The densely populated area was built upon wetlands and stormwater runoff can overwhelm the neighborhood's aging drainage system and lead to localized flooding, particularly when storms and high tides coincide. The flooding can slow or prevent first-responders from assisting people in need and limit residents from accessing essential goods and services.

The Marin City Stormwater Plan will build upon findings from several prior studies, addressing the locations where water flow is hampered, and recommend improvements that reduce flood risk, improve drainage, and maintain public access. Plan completion is targeted for summer 2024.

Once complete, the stormwater plan will provide detailed recommendations for potential projects that could then be implemented by local or state agencies. The plan would also be used as a basis for grant funding applications and associated environmental review requirements.

The estimated \$785,000 cost of the plan is funded by the District's Flood Control Zone 3 (Richardson Bay) budget. The District plans to coordinate with Caltrans as it develops its sea level rise project for State Route 1 and U.S. Highway 101 between the Manzanita Park & Ride lot (owned by Caltrans) and Marin City.

For updates on the Marin City Stormwater Plan, please check the project webpage regularly for new information.

If you are a person with a disability and require an accommodation to participate in a County program, service, or activity, requests may be made by calling (415) 473-4381 (Voice), Dial 711 for CA Relay, or by <u>emailing the Disability Access program</u> at least five business days in advance of the event. We will do our best to fulfill requests received with less than five business days' notice. Copies of documents are available in alternative formats upon request.

How Do I? Government For Residents For Business Recreation Contacts Newsroom Board of Supervisors Email Updates 2 Social Media Webcasts Emergency Alerts 2 Website Accessibility Contact Webmaster Terms & Conditions Disclaimers Get Adobe Reader 🗹 Mobile Gallery

Copyright © 2022 County of Marin



with next steps, including working with Caltrans on the environmental permitting phase. It is anticipated that construction could begin sometime in fall/winter 2023.

The estimated cost of the project, which includes operations and maintenance for three years, is \$2.1 to \$2.8 million. It will be covered by the \$10 million in state funding secured by state Senator Mike McGuire in 2022.

Marin City, which is adjacent to Richardson Bay, can become flooded during significant storms, especially when those storms are occur during high tides. The localized flooding impacts Donahue Street and impedes access to the Highway 101 ramps, restricting mobility of Marin City residents and visitors who must wait until the floodwater slowly drains out to Richardson Bay after a storm.



The installation of a sump and portable pump system is planned for the intersection of Donahue Street and Drake Avenue in Marin City. If approved March 21, the effort would reduce the severity of localized flooding along the key thoroughfare in the unincorporated neighborhood, which has a history of flooding issues during storms and high tides.

This project includes installing a sump at the low point elevation along Donahue Street at the Drake Avenue intersection, which will be pumped by portable pumps that are brought to the location in advance of major storm events. During and after the storms, the portable pump station would convey floodwater through an above-ground pipeline along Donahue Street, ultimately discharging the floodwater into Richardson Bay.

The pump system will not be large enough to prevent roadway flooding at the location following significant storms and high tides. The functional benefits would be to reduce the depth and duration of the flooding on Donahue Street and the Highway 101 ramps (which are owned and maintained by Caltrans). The improvement would allow vehicle access after storms to return faster than before the pump system was installed.

The project on Donahue is intended to provide an immediate positive impact while the District pursues several other flood mitigation elements to be installed near the Marin City Pond. Those potential projects are awaiting federal
#### Portable Pump Station Proposed for Marin City

funding and the current grant request is for approximately \$10 million. The proposed improvements are a larger, permanent pump station, a floodwall and storm drain upgrades.

Separately from the efforts detailed above, the District is also initiating a watershed-wide <u>Marin City Stormwater</u> <u>Plan</u> to identify an updated list of other projects that could be constructed in Marin City by local and state agencies to alleviate flooding impacts. The Stormwater Plan is distinct from the prior studies in that it will result in a public stormwater infrastructure plan developed with direct input from Marin City residents and organizations in partnership with the various public agencies that have responsibilities within the watershed. More information will be available on the Stormwater Plan as the work progresses.

If you are a person with a disability and require an accommodation to participate in a County program, service, or activity, requests may be made by calling (415) 473-4381 (Voice), Dial 711 for CA Relay, or by <u>emailing the Disability Access program</u> at least five business days in advance of the event. We will do our best to fulfill requests received with less than five business days' notice. Copies of documents are available in alternative formats upon request.

How Do I? Government For Residents For Business Recreation Contacts Newsroom Board of Supervisors Email Updates 2 Social Media Webcasts Emergency Alerts 2 Website Accessibility Contact Webmaster Terms & Conditions Disclaimers Get Adobe Reader 🗹 Mobile Gallery

Copyright © 2022 County of Marin

Public Review Draft May 2021

# 2020 Urban Water Management Plan

# for Marin Municipal Water District









### TABLE OF CONTENTS

1.	INTRC	DUCTION				
	1.1	Backgrou	und and Purpose	1		
	1.2	Urban Water Management Planning and the California Water Code				
	1.3					
	1.4	Plan Org	anization	2		
	1.5	Demons	tration of Consistency with the Delta Plan for Participants in Covered Actions	3		
	1.6	Lay Desc	ription	3		
2.	PLAN	PREPARAT	ΓΙΟΝ	7		
	2.1	Basis for	Preparing a Plan	7		
	2.2	Coordina	ation and Outreach	9		
		2.2.1	Wholesale Coordination	9		
		2.2.2	Agency Coordination	10		
		2.2.3	Public Participation	12		
	2.3	UWMP S	Structure, Standard Units, and Basis for Reporting	12		
3.	SERVI	CE AREA A	ND SYSTEM DESCRIPTION	14		
	3.1	Populati	on and Employment Trends Within the Service Area	14		
		3.1.1	Future Population Growth	15		
		3.1.2	Future Employment Growth	15		
	3.2	Land Use	es within Service Area	16		
	3.3	Service A	Area Social, Economic, and Demographic Factors	17		
	3.4	Climate		18		
	3.5	Climate	Change Considerations	19		
	3.6	Water D	istribution System	20		
4.	SYSTE	M WATER	DEMANDS	25		
	4.1	Current	and Historic Total Water Demand	26		
		4.1.1	Current and Historical Potable Water Demand	26		
		4.1.2	Current and Historical Non-Potable Water Demand	30		
		4.1.3	Potable Water Make-Up to the Recycled Water System	31		

i



		4.1.4	Distribution System Water Loss	31
	4.2	Projected	d Total Water Demand	32
		4.2.1	Projected Potable Water Demand	32
		4.2.2	Projected Non-Potable Water Demand	34
		4.2.3	Water Use for Lower Income Households	35
		4.2.4	Water Savings from Codes, Standards, Ordinances, or Transportation and Land Use Plans	36
		4.2.5	Projected Total Water Demand	37
	4.3	Water Us	se Sectors Not Included in the Demand Projections	38
	4.4	Climate (	Change Impacts to Demand	38
	4.5	Urban W	ater Use Objectives (Future Requirements)	39
5.	BASELI	INE WATE	R USE AND SB X7-7 WATER CONSERVATION TARGETS	42
	5.1	Service A	rea Population	43
	5.2	Baseline	Water Use	44
	5.3	Water Us	se Targets	46
		5.3.1.	Individual Water Use Targets	47
		5.3.2	Regional Water Use Targets	47
	5.4	2020 Tar	get Compliance	49
6.	WATE	R SUPPLY	CHARACTERIZATION	50
	6.1	Purchase	ed Water	50
		6.1.1	Sonoma County Water Agency Surface Water Supply	50
		6.1.2	Sonoma County Water Agency Surface Water Rights	51
		6.1.3	Sonoma County Water Agency Groundwater Supply	52
	6.2	Groundw	vater	53
		6.2.1	Basin Description and Status	54
		6.2.2	Non-SGMA Groundwater Management	55
		6.2.3	SGMA Groundwater Management	55
		6.2.4	Coordination with Groundwater Supply Agencies	55
		6.2.5	Historical Pumping and Supply Sufficiency	56
	6.3	Surface \	Water	56
	6.4	Stormwa	iter	57



	6.5	Wastewater and Recycled Water			
		6.5.1	Recycled Water Coordination	57	
		6.5.2	Wastewater Collection, Treatment, and Disposal	58	
		6.5.3	Recycled Water System and Potential, Current, and Projected Uses of Recycled Water	65	
		6.5.4	Comparison of Previously Projected Use and Actual Use	68	
		6.5.5	Actions to Encourage and Optimize Future Recycled Water Use	69	
	6.6	Desalina	ated Water	71	
	6.7	Water E	xchanges and Transfers	71	
	6.8	Future \	Water Projects	72	
	6.9	Summa	ry of Existing and Planned Sources of Water	74	
	6.10	Special	Conditions		
		6.10.1	Climate Change Effects		
		6.10.2	Regulatory Conditions and Project Development		
		6.10.3	Other Locally Applicable Criteria		
	6.11	Energy l	Intensity	79	
7.	WATE	R SUPPLY	′ RELIABILITY	81	
	7.1	Constra	ints on Water Sources	81	
		7.1.1	Supply Availability	82	
		7.1.2	Water Quality Impacts on Reliability		
		7.1.3	Climate Change Impacts to Supply	85	
	7.2	Reliabili	ity by Type of Year	86	
		7.2.1	Purchased Water	87	
		7.2.2	Surface Water	88	
		7.2.3	Recycled Water	88	
		7.2.4	Raw Water	89	
	7.3	Supply a	and Demand Assessment	90	
	7.4	Water Management Tools and Options			
	7.5	Drought	t Risk Assessment	93	
		7.5.1	Characteristic Five-Year Water Use		
		7.5.2	Risk Assessment Projections		



8.	WATE	SHORTAGE CONTINGENCY PLANNING			
9.	DEMA	DEMAND MANAGEMENT MEASURES			
	9.1	Regional	Water Conservation	109	
		9.1.1	Funding	110	
		9.1.2	Annual Report	110	
		9.1.3	Water Education Program	110	
		9.1.4	Public Outreach Program	111	
		9.1.5	Regional Programs	111	
	9.2	Agency V	Vater Conservation	112	
		9.2.1	DMM 1 – Water Waste Prevention Ordinances	113	
		9.2.2	DMM 2 – Metering	114	
		9.2.3	DMM 3 – Conservation Pricing	114	
		9.2.4	DMM 4 – Public Education and Outreach	115	
		9.2.5	DMM 5 – Programs to Assess and Manage Distribution System Real Loss	118	
		9.2.6	DMM 6 – Water Conservation Program Coordination and Staffing Support	118	
		9.2.7	DMM 7 – Other DMMs	119	
	9.3	Planned	Implementation to Achieve Water Use Targets	119	
10.	PLAN A	DOPTION	I AND SUBMITTAL	120	
	10.1	Notificat	ion of UWMP Preparation	120	
	10.2	Notificat	ion of Public Hearing	120	
		10.2.1	Notice to Cities and Counties	120	
		10.2.2	Notice to the Public	121	
	10.3	Public He	earing and Adoption	121	
	10.4	Plan Submittal			
	10.5	5 Public Availability			
	10.6	Amendin	g an Adopted UWMP or Water Shortage Contingency Plan	123	
11.	REFERI	ENCES		124	



#### TABLES

- Table 2-1
   Public Water Systems (DWR Table 2-1)
- Table 2-2Plan Identification Type (DWR Table 2-2)
- Table 2-3
   Water Supplier Information Exchange (DWR Table 2-4)
- Table 2-4 Notification to Cities, Counties, and Other Agencies (DWR Table 10-1)
- Table 2-5Supplier Identification (DWR Table 2-3)
- Table 3-1 Population Current and Projected (DWR Table 3-1)
- Table 3-2 Employment Current and Projected
- Table 3-3 Demographic and Housing Characteristics
- Table 3-4 Climate Characteristics
- Table 4-1 Demands for Potable and Non-Potable Water Actual (DWR Table 4-1)
- Table 4-2 Historical and Current Potable Water Demand and Population
- Table 4-3 12 Month Water Loss Audit Reporting (DWR Table 4-4)
- Table 4-4
   Use for Potable and Non-Potable Projected (DWR Table 4-2)
- Table 4-5 Inclusion in Water Use Projections (DWR Table 4-5)
- Table 4-6 Projected Water Use for Lower Income Households
- Table 4-7 Projected Potable Water Demand and Projected Passive and Active Water Conservation
- Table 4-8 Gross Water Use (DWR Table 4-3)
- Table 4-9 Current and Projected Residential Per Capita Water Use
- Table 5-1 SB X7-7 Service Area Population (SB X7-7 Table 3)
- Table 5-2 RA1 Weighted Baseline
- Table 5-3
   Baselines and Targets Summary (DWR Table 5-1)
- Table 5-4 DWR Regional Alliance Weighted 2020 Target
- Table 5-5 2020 Compliance for North Marin-Sonoma Regional Alliance (DWR Table 5-2)
- Table 6-1 Groundwater Volume Pumped (DWR Table 6-1)
- Table 6-2
   District Surface Water Reservoir System
- Table 6-3 Wastewater Collected Within Area in 2020 (DWR Table 6-2)
- Table 6-4 Wastewater Treatment and Discharge Within Service Area in 2020 (DWR Table 6-3)
- Table 6-5Current and Projected Recycled Water Direct Beneficial Uses Within Service Area (DWR Table6-4)



- Table 6-6 2015 UWMP Recycled Water Use Projection Compared to 2020 Actual (DWR Table 6-5)
- Table 6-7 Methods to Expand Future Recycled Water Use (DWR Table 6-6)
- Table 6-8 Expected Future Water Supply Projects or Programs (DWR Table 6-7)
- Table 6-9 Water Supplies Actual (DWR Table 6-8)
- Table 6-10 Water Supplies Projected (DWR Table 6-9)
- Table 6-11 Recommended Energy Intensity Total Utility Approach (DWR Table O-1B)
- Table 7-1 Factors Resulting in Inconsistency of Supply
- Table 7-2
   Basis of Water Year Data (Reliability Assessment) (DWR Table 7-1)
- Table 7-3
   Projected Availability of SCWA Supply (Responds to DWR Table 7-1)
- Table 7-4
   Projected Availability of Local Surface Water Supply (Responds to DWR Table 7-1)
- Table 7-5
   Projected Availability of Recycled Water Supply (Responds to DWR Table 7-1)
- Table 7-6
   Projected Availability of Raw Water Supply (Responds to DWR Table 7-1)
- Table 7-7
   Normal Year Supply and Demand Comparison (DWR Table 7-2)
- Table 7-8Single Dry Year Supply and Demand Comparison (DWR Table 7-3)
- Table 7-9 Multiple Dry Years Supply and Demand Comparison (DWR Table 7-4)
- Table 7-10 Characteristic Five-Year Water Use
- Table 7-11 Five-Year Drought Risk Assessment Tables to Address Water Code 10635(b) (DWR Table 7-5)
- Table 8-1
   Water Shortage Contingency Plan Levels (DWR Table 8-1)
- Table 8-2 Demand Reduction Actions (DWR Table 8-2)
- Table 8-3Supply Augmentation and Other Actions (DWR Table 8-3)
- Table 9-1Current Conservation Programs

#### FIGURES

- Figure 3-1 Regional Vicinity
- Figure 3-2 District Service Area
- Figure 3-3 District Water System

#### APPENDICES

#### Appendix A. Completed UWMP Checklist



- Appendix B. Renewal of the Third Amended Offpeak Water Supply Agreement and the Amended Agreement for the Sale of Water Between the Sonoma County Water Agency and Marin Municipal Water District
   Appendix C. 2020 Urban Water Management Plan Water Demand Analysis and Water Conservation Measures Update, Marin Municipal Water District
- Appendix D. UWMP Agency Notification Letters.
- Appendix E. UWMP Public Notification Notices.
- Appendix F. SB X7-7 Compliance Tables
- Appendix G. Board Policy No. 2 Recycled Water
- Appendix H. Water Shortage Contingency Plan
- Appendix I. 2007 Water Conservation Master Plan
- Appendix J. Marin Municipal Water District Title 13 Water Service Conditions and Water Conservation Measures
- Appendix K. AWWA Water Loss Reports
- Appendix L. Resolution XXX on UWMP and WSCP 2020 Update
- Appendix M. Plan Submittal Documentation



#### ABBREVIATIONS

ABAG	Association of Bay Area Governments
AF	acre-feet
AFY	acre-feet per year
Alliance	North Marin-Sonoma Alliance
AMI	Advanced Metering Infrastructure
AWE	Alliance for Water Efficiency
AWWA	American Water Works Association
BMP	best management practices
BTTP	Bon Tempe Treatment Plant
CASGEM	California Statewide Groundwater Elevation Monitoring
CCR	California Code of Regulations
Census	United States Census
CEQA	California Environmental Quality Act
cfs	cubic feet per second
CII	commercial, industrial, and institutional
CIMIS	California Irrigation Management Information System
CMSA	Central Marin Sanitation Agency
CUWCC	California Urban Water Conservation Council
CWC	California Water Code
CY	calendar years
DDW	Division of Drinking Water
Demand Report	2020 Water Demand Analysis and Water Conservation Measure Update, Marin
	Municipal Water District
DMM	demand management measures
DOF	Department of Finance
DWR	Department of Water Resources
EBMUD	East Bay Municipal Utility District
EIR	Environmental Impact Report
EKI	EKI Environment & Water, Inc.
ETo	reference evapotranspiration
ETWU	estimating total water use
FY	fiscal year
GGNRA	Golden Gate National Recreation Area
GMP	Groundwater Management Plan
GPCD	gallons per capita per day
gpf	gallons per flush
gpm	
GSA	gallons per minute
	gallons per minute Groundwater Sustainability Agency
GSP	gallons per minute Groundwater Sustainability Agency Groundwater Sustainability Plan
GSP HECW	gallons per minute Groundwater Sustainability Agency Groundwater Sustainability Plan High-Efficiency Clothes Washer
GSP HECW HET	gallons per minute Groundwater Sustainability Agency Groundwater Sustainability Plan High-Efficiency Clothes Washer High-Efficiency Toilet

# Table of Contents2020 Urban Water Management PlanMarin Municipal Water District



HVAC	Heating, ventilation, and air conditioning
LGVSD	Las Gallinas Valley Sanitary District
LHMP	Local Hazard Mitigation Plan
MAWA	maximum applied water allowance
MCL	Maximum Contaminant Level
Methodologies	Methodologies for Calculating Baseline and Compliance Urban Per Capita Water Use,
	California Department of Water Resources Division of Statewide Integrated Water
	Management Water Use and Efficiency Branch
MFR	multi-family residential
MG	million gallons
mgd	million gallons per day
MMWD	Marin Municipal Water District
MWELO	Model Water Efficient Landscape Ordinance
NBWRA	North Bay Water Reuse Authority
NMFS	National Marine Fisheries Service
NMWD	North Marin Water District
PG&E	Pacific Gas & Energy
PWS	public water system
R-GPCD	residential gallons per capita per day
RUWMP	Regional Urban Water Management Plan
RWTF	Recycled Water Treatment Facility
SASM	Sewage Agency of Southern Marin
SB	Senate Bill
SCWA	Sonoma County Water Agency
SFR	single family residential
SGMA	Sustainable Groundwater Management Act
SGTP	San Geronimo Treatment Plant
SMCSD	Sausalito-Marin City Sanitary District
SMSWP	Sonoma-Marin Saving Water Partnership
Sonoma Water	Sonoma County Water Agency
SWRCB	State Water Resources Control Board
TAC	Technical Advisory Committee
Title 22	California Code of Regulations, Title 22
TUCP	Temporary Urgency Change Petition
USACE	U.S Army Corps of Engineers
USEPA	United States Environmental Protection Agency
USGS	United States Geological Survey
UWMP	Urban Water Management Plan
UWMP Act	Urban Water Management Planning Act
UWMP	2020 Urban Water Management Plans Guidebook for Urban Water Suppliers
Guidebook 2020	
VOMWD	Valley of the Moon Water District
WAC	Water Advisory Committee
WRCC	Western Regional Climate Center

ix

Table of Contents2020 Urban Water Management PlanMarin Municipal Water District



WSCPWater Shortage Contingency PlanWUEwater use efficiencyWWTPwaste water treatment plant

х



#### 1. INTRODUCTION

This chapter discusses the importance and uses of this Urban Water Management Plan (UWMP or Plan), the relationship of this Plan to the California Water Code (CWC), the relationship of this Plan to other local and regional planning efforts, and how this Plan is organized and developed in general accordance with the 2020 Urban Water Management Plan's Guidebook for Urban Water Suppliers (UWMP Guidebook 2020).<sup>1</sup>

#### **1.1** Background and Purpose

This UWMP addresses the Marin Municipal Water District (MMWD or District) water system. Most of the District's water supply comes from a network of seven local, rain-fed reservoirs. This supply is supplemented with water from Sonoma County Water Agency (SCWA or Sonoma Water), which provides surface water from the Russian River and to a lesser extent groundwater from the Santa Rosa Plain Subbasin of the Santa Rosa Valley Basin (California Department of Water Resources [DWR] Basin No. 1-55.01). Some recycled water is also used for non-potable uses such as landscape irrigation, cooling towers, car washes, and toilet flushing.

This UWMP is a foundational document and source of information about the District's historical and projected water demands, water supplies, supply reliability and potential vulnerabilities, water shortage contingency planning, and demand management programs.

The District's last UWMP was completed in 2016, referred to herein as the "2015 UMWP." This Plan is an update to the 2015 UWMP and carries forward information from that plan that remains current and is relevant to this Plan and provides additional information as required by amendments to the Urban Water Management Planning Act (UWMP Act; CWC §10610 – 10657). Although this Plan is an update to the 2015 UWMP, it was developed to be a self-contained, stand-alone document.

#### **1.2** Urban Water Management Planning and the California Water Code

The UWMP Act requires urban water suppliers to prepare an UWMP every five years and to submit this plan to the DWR, the California State Library, and any city or county within which the supplier provides water. All urban water suppliers, either publicly or privately owned, providing water for municipal purposes either directly or indirectly to more than 3,000 customers or supplying more than 3,000 acrefeet annually are required to prepare an UWMP (CWC §10617).

The UWMP Act was enacted in 1983. Over the years it has been amended in response to water resource challenges and planning imperatives confronting California. A significant amendment was made in 2009 as a result of the governor's call for a statewide 20% reduction in urban water use by 2020, referred to as "20x2020," the Water Conservation Act of 2009, and "Senate Bill (SB) X7-7." This amendment required urban retail water suppliers to establish water use targets for 2015 and 2020 that would result in

<sup>&</sup>lt;sup>1</sup> The UWMP Guidebook 2020 is available at: https://water.ca.gov/Programs/Water-Use-And-Efficiency/Urban-Water-Use-Efficiency/Urban-Water-Management-Plans.

# Introduction 2020 Urban Water Management Plan Marin Municipal Water District



statewide water savings of 20% by 2020. Beginning in 2016, urban retail water suppliers were required to comply with the water conservation requirements in SB X7-7 to be eligible for state water grants or loans. Chapter 5 of this Plan contains the data and calculations used to determine compliance with these requirements.

A subsequent substantial revision to the UWMP Act was made in 2018 through a pair of bills (i.e., Assembly Bill 1668 and Senate Bill 606), referred to as "Making Water Conservation a California Way of Life" or the "2018 Water Conservation Legislation." These changes include, among other things, additional requirements for Water Shortage Contingency Plans (WSCPs), expansion of dry year supply reliability assessments to a five-year drought period, establishment of annual drought risk assessment procedures and reporting, and discussion of new conservation targets referred to as "annual water use objectives," which will require retailers to continue to reduce water use beyond the 2020 SB X7-7 targets. The UWMP Act contains numerous other requirements that an UWMP must satisfy. **Appendix A** to this Plan lists each of these requirements and where in the Plan they are addressed.

#### **1.3** Relationship to Other Planning Efforts

This Plan provides information specific to water management and planning by the District. However, water management does not happen in isolation; there are other planning processes that integrate with the UWMP to accomplish urban planning. Some of these relevant planning documents include relevant city and county General Plans, Water Master Plans, and others.

This Plan is informed by and helps to inform these other planning efforts. In particular, this Plan utilizes information contained county General Plans and local and regional water resource plans to the extent data from these plans are applicable and available.

#### 1.4 Plan Organization

The organization of this Plan follows the same sequence as outlined in the UWMP Guidebook 2020.

- Chapter 1 Introduction
- Chapter 2 Plan Preparation
- Chapter 3 Service Area and System Description
- Chapter 4 System Water Demands
- Chapter 5 Baseline Water Use and SB X7-7 Water Conservation Targets
- Chapter 6 Water Supply Characterization
- Chapter 7 Water Supply Reliability
- Chapter 8 Water Shortage Contingency Planning
- Chapter 9 Demand Management Measures
- Chapter 10 Plan Adoption and Submittal
- Chapter 11 References

# Introduction 2020 Urban Water Management Plan Marin Municipal Water District



In addition to these eleven chapters, this Plan includes a number of appendices providing supporting documentation and supplemental information. Pursuant to CWC §10644(a)(2), this Plan utilizes the standardized forms, tables, and displays developed by DWR for the reporting of water use and supply information required by the UWMP Act. This Plan also includes additional tables, figures, and maps to augment the set developed by DWR, as appropriate. The table headers indicate if the table is part of DWR's standardized set of submittal tables.

#### **1.5** Demonstration of Consistency with the Delta Plan for Participants in Covered Actions

Although not required by the UWMP Act, in the UWMP Guidebook 2020, DWR recommends that all suppliers that are participating in, or may participate in, receiving water from a proposed project that is considered a "covered action" under the Delta Plan—such as a (1) multiyear water transfer; (2) conveyance facility; or (3) new diversion that involves transferring water through, exporting water from, or using water in the Sacramento-San Joaquin Delta (Delta)—provide information in their UWMP to demonstrate consistency with the Delta Plan policy WR P1, Reduce Reliance on the Delta Through Improved Regional Water Self-Reliance (California Code of Regulations, Title 23, Section 5003).

The District's source of water supply is from a network of local, rain-red reservoirs, imported water purchased from the SCWA, and recycled water, and therefore the District does not receive water or plan to receive water from a "covered action" under the Delta Plan. As such, demonstration of consistency with the Delta Plan is not applicable.

#### 1.6 Lay Description

#### **☑** *CWC* § 10630.5

Each plan shall include a simple lay description of how much water the agency has on a reliable basis, how much it needs for the foreseeable future, what the agency's strategy is for meeting its water needs, the challenges facing the agency, and any other information necessary to provide a general understanding of the agency's plan.

This Urban Water Management Plan (UWMP or Plan) is prepared for Marin Municipal Water District (also referred to as MMWD or District), which serves drinking water to a population of approximately 191,269 in Marin County, California. This UWMP serves as a foundational planning document and includes descriptions of historical and projected water demands and water supplies and reliability over the UWMP planning horizon. This document also describes the actions the District is taking to promote water conservation, both by the District and by its customers (referred to as "demand management measures"), and includes a plan to address potential water supply shortages such as drought or other impacts to supply availability (the "Water Shortage Contingency Plan"). This UWMP is updated every five years in accordance with state requirements under the Urban Water Management Planning Act (UWMP Act) and amendments (Division 6 Part 2.6 of the California Water Code [CWC] §§10610 – 10656). Past plans developed for the District are available on the California Department of Water Resources (DWR) Water Use Efficiency Data Portal website: <u>https://wuedata.water.ca.gov/</u>. This Plan includes 11 chapters, which are summarized below.



#### Chapter 1 - Introduction

This chapter presents the background and purpose of the UWMP, identifies the Plan organization, and provides this lay description overview of the document.

#### Chapter 2 - Plan Preparation

This chapter discusses key structural aspects related to the preparation of the UWMP, and describes the coordination and outreach conducted as part of the preparation of the Plan, including coordination with local agencies (i.e., members of the Sonoma-Marin Saving Water Partnership [SMSWP] and Marin County) and the public.

#### Chapter 3 - Service Area and System Description

This chapter provides a description of the District's water system and service area, including information related to the climate, population, and demographics. The District serves a population of approximately 191,269 and has a moderate climate characterized by mild dry summers and cool wet winters. The majority of precipitation falls during late autumn, winter, and spring, averaging 47 inches of rainfall annually.

#### Chapter 4 - System Water Demands

This chapter provides a description and quantifies the District's current and projected demands through the year 2045. The District provides drinking water (also referred to as "potable water"), raw water, and recycled water to its customers. Water demands refer not only to the water used by customers, but also includes the water used as part of the system maintenance and operation, as well as unavoidable losses inherent in the operation of a water distribution system. The District also provides raw water environmental releases from Kent and Soulajule Lakes to meet environmental flow requirements that benefit silver salmon and steelhead populations in Walker Creek and Lagunitas Creek. Total potable and raw water demand within the District (excluding environmental releases) was 25,319 acre-feet per year (AFY) on average between 2016 and 2020.<sup>2</sup> Taking into account historical water use, expected population increase and other growth, climatic variability, and other assumptions, total potable and raw water (excluding environmental releases) demand within the District is projected to increase to 26,915 AFY by 2045, a change of 5.6% compared to the 2016-2020 average.

The District also provides recycled water to customers in the Terra Linda area of San Rafael for nonpotable uses, including irrigation, cooling towers, car washes, and toilet flushing. Recycled water demand was 638 acre-feet per year (AFY) on average between 2016 and 2020, and is 2020 and to increase to 771 AFY by 2045, an increase of 21% compared to the 2016-2020 average. The District's recycled water system and demands are described in Chapter 6.

#### Chapter 5 - Baseline Water Use and SB X7-7 Water Conservation Targets

In this chapter, the District compares its per capita water use with its water use target for the year 2020. The Water Conservation Act of 2009 (Senate Bill [SB] X7-7) was enacted in November 2009 and requires the state of California to achieve a 20% reduction in urban per capita water use by 31 December 2020. In

<sup>&</sup>lt;sup>2</sup> This water use includes potable water used to meet demands by the recycled water system while the recycled water system was offline to allow for infrastructure upgrades.

# Introduction 2020 Urban Water Management Plan Marin Municipal Water District



order to achieve this, each urban retail water supplier was required to establish water use targets for 2015 and 2020 using methodologies established by DWR. The District is a member of a "Regional Alliance," which allows retailers to demonstrate compliance with SB X7-7 on a regional basis. In 2020, the North Marin-Sonoma Alliance used 113 GPCD, which is in compliance with and below its 2020 target of 129 GPCD.

#### Chapter 6 - Water Supply Characterization

This chapter presents an analysis of the District's water supplies, as well as an estimate of water-related energy consumption. The intent of this chapter is to present a comprehensive overview of the District's water supplies, estimate the volume of available supplies over the UWMP planning horizon, and assess the sufficiency of the District's supplies to meet projected demands under "normal" hydrologic conditions.

Most of the District's water supply comes from a network of seven local, rain-fed reservoirs. Total reservoir storage operated by the District is 25.9 billion gallons (79,566 acre-feet). This supply is supplemented with water imported from the Russian River and purchased from the Sonoma County Water Agency (SCWA or Sonoma Water). The District has contracted with SCWA for this source of water since 1975. The current contract allows for the District to purchase up to 14,300 AF; however, the District's ability to accept this volume is currently limited by infrastructure constraints. The District is projecting use of up to 10,000 AFY by 2030 subject to capital improvements that would increase conveyance capacity beyond the 10,000 AFY. To treat this supply, the District operates three water treatment plants, including the Bon Tempe Treatment Plant, the San Geronimo Treatment Plant, and the Ignacio treatment facility.

There are also five wastewater treatment plants within the District's service area that collectively treat roughly 17,700 AFY of wastewater. The District produces its own recycled water by treating secondary effluent provided to the District by the Las Gallinas Valley Sanitary District to tertiary levels before being distributed to customers. MMWD's recycled water system consists of nearly 25 miles of pipeline and delivered about 748 AFY through approximately 300 service connections in 2020.<sup>3</sup>

Based on comparison of demands and available supplies, the District's water supply is expected to be sufficient to support the District's projected water demand through 2045 during normal hydrologic years.

Calculation and reporting of water system energy intensity is a new requirement for the 2020 UWMPs. Energy intensity is defined as the net energy used for water treatment, conveyance, and distribution for all water entering the distribution system and does not include the energy used to convey or treat wastewater. The energy intensity for the District is estimated to be 495 kilowatt hours per acre-foot of water (kWh/AF).

#### Chapter 7 - Water Supply Reliability

This chapter assesses the reliability of the District's water supplies, with a specific focus on potential constraints such as water supply availability, water quality, and climate change. The intent of this chapter is to identify any potential constraints that could affect the reliability of the District's supply (such as drought conditions) to support the District's planning efforts. Water service reliability is assessed during normal, single dry-year, and multiple dry-year hydrologic conditions. Based on this analysis, the District

<sup>&</sup>lt;sup>3</sup> The recycled water plant was non-operational in 2019 and 2020 to allow for infrastructure upgrades. During this time, water demands by the recycled water system were met with potable water.

# Introduction 2020 Urban Water Management Plan Marin Municipal Water District



expects the available supplies to be sufficient to meet projected demands in all hydrologic conditions, including a five-year drought period, and considering the impacts of climate change.

Further, potential water quality issues are not expected to affect the quality of water served to the District's customers, as water quality is routinely monitored and the District is able to make all appropriate adjustments to its treatment and distribution system to ensure only high-quality drinking water is served.

#### Chapter 8 - Water Shortage Contingency Planning

This chapter describes the Water Shortage Contingency Plan (WSCP) for the District. The WSCP serves as a standalone document to be engaged in the case of a water shortage event, such as a drought or supply interruption, and defines specific policies and actions that will be implemented at various shortage level scenarios. For example, implementing customer water budgets and surcharges, or restricting landscape irrigation to specific days and/or times. Consistent with DWR requirements, the WSCP includes six levels to address shortage conditions ranging from up to 10% to greater than 50% shortage.

#### Chapter 9 - Demand Management Measures

This chapter includes descriptions of past and planned conservation programs that the District and the Sonoma-Marin Saving Water Partnership (SMSWP) operate within each demand management measure (DMM) category outlined in the UWMP Act, specifically: (1) water waste prevention ordinances, (2) metering, (3) conservation pricing, (4) public education and outreach, (5) distribution system water loss management, (6) water conservation program coordination and staffing support, and (7) "other" DMMs. The District has developed a suite of conservation programs and policies which address each DMM category.

#### Chapter 10 - Plan Adoption and Submittal

This chapter provides information on a public hearing, the adoption process for the UWMP and WSCP, the adopted UWMP and WSCP submittal process, UWMP and WSCP implementation, and the process for amending the adopted UWMP and WSCP. Prior to adopting the Plan, the District held a formal public hearing to present information on its UWMP and WSCP on 15 July 2021 at 7:30 pm. The UWMP and WSCP were submitted to DWR within 30 days of adoption and by the 1 July 2021 deadline.

#### Chapter 11 - References

This chapter contains key references and sources used throughout the Plan.



#### 2. PLAN PREPARATION

This section provides information on the process for developing Marin Municipal Water District's (MMWD's or District's) 2020 Urban Water Management Plan (UWMP or Plan), including an overview of coordination with other agencies and a description of public outreach.

Text from the UWMP Act has been included in grey text boxes with italicized font at beginning of relevant sections of this UWMP. The information presented in the respective UWMP sections and the associated text, figures, and tables are collectively intended to fulfill the requirements of that sub-section of the UWMP Act. To the extent practicable, supporting documentation has also been provided in **Appendix A** through **Appendix M**. Other sources for the information contained herein are provided in the references section of this document.

Per CWC §10644(a)(2), selected information for the 2020 UWMP updates must be presented in standardized tables for electronic submittal to DWR. The tables presented in this UWMP have been renumbered, but the content has been preserved and the original DWR table numbers are included in parentheses in the table titles.

#### 2.1 Basis for Preparing a Plan

#### ☑ CWC § 10617

"Urban water supplier" means a supplier, either publicly or privately owned, providing water for municipal purposes either directly or indirectly to more than 3,000 customers or supplying more than 3,000 acre-feet of water annually. An urban water supplier includes a supplier or contractor for water, regardless of the basis of right, which distributes or sells for ultimate resale to customers. This part applies only to water supplied from public water systems subject to Chapter 4 (commencing with Section 116275) of Part 12 of Division 104 of the Health and Safety Code.

#### ☑ CWC § 10620

Every person that becomes an urban water supplier shall adopt an urban water management plan within one year after it has become an urban water supplier.

#### ☑ CWC § 10621 (a)

Each urban water supplier shall update its plan at least once every five years on or before July 1, in years ending in six and one, incorporating updated and new information from the five years preceding each update.

#### **☑** CWC § 10621 (f)(1)

Each urban water supplier shall update and submit its 2020 plan to the department by July 1, 2021.

#### **☑** CWC § 10644 (a)(2)

The plan, or amendments to the plan, submitted to the department ... shall include any standardized forms, tables, or displays specified by the department.

In 1983, the California Legislature enacted the UWMP Act (CWC §10610 - §10657). The UWMP Act states that every urban water supplier that provides water to 3,000 or more connections or that provides over 3,000 acre-feet of water per year (AFY) should make every effort to ensure the appropriate level of water service reliability to meet the needs of its customers during normal, dry, and multiple dry years.

# Plan Preparation 2020 Urban Water Management Plan Marin Municipal Water District



As a water system that provides drinking water for human consumption, the District is regulated as a Public Water System (PWS) by the State Water Resources Control Board (SWRCB), Division of Drinking Water. **Table 2-1** lists the District's PWS identification number. The SWRCB requires that water agencies report water usage and other relevant PWS information via the electronic Annual Reports to the Drinking Water Program (eARDWP). These data are used by the state to determine, among other things, whether an urban retail water supplier has reached the threshold (3,000 or more connections or 3,000 acre-feet [AF] of water supplied) for submitting an UWMP.

As shown in **Table 2-1**, the District served approximately 61,700 connections in 2020 and is therefore subject to the requirements of the UWMP Act.

Public Water System Number	Public Water System Name	Number of Municipal Connections 2020	Volume of Water Supplied 2020	
2110003	Marin Municipal Water District	61,700	40,149	
<b>TOTAL</b> 61,700 40,149				
NOTES: (a) Units are in AF.				

Table 2-1	Public Water Systems (DWR Table 2-1)
-----------	--------------------------------------

The District's 2020 UWMP describes how the current and future water resources and demands within the District's service area will be managed to provide an adequate and reliable water supply. Additionally, and as applicable, the District's 2020 UWMP reflects the significant revisions to the UWMP Act that have been made since 2015.

As with the 2010 and 2015 UWMPs, the District's 2020 UWMP has been prepared as an individual rather than a regional plan, as shown in **Table 2-2** (DWR Table 2-2). However, the 2020 UWMP was developed with close coordination with its wholesaler, Sonoma County Water Agency (SCWA or Sonoma Water), and other parties that receive water from the SCWA (Water Contractors). Furthermore, a regional alliance was formed in 2011 among these agencies including the cities of Santa Rosa, Rohnert Park, Sonoma, Cotati, Petaluma, Town of Windsor, Marin Municipal Water District, Valley of the Moon Water District, and North Marin Water District to comply with Senate Bill (SB) X7-7, the Water Conservation Act of 2009. This regional alliance, referred to in **Table 2-2** as the "North Marin-Sonoma Alliance" but more typically referred to as the Sonoma-Marin Saving Water Partnership (SMSWP), is used within the 2020 UWMP for reporting on regional 2015 and 2020 water use targets (see Chapter 5). All other elements of the CWC requirements are addressed in the District's Individual Plan.

8

## Plan Preparation 2020 Urban Water Management Plan Marin Municipal Water District



Tab	le	2-2

Plan Identification Type (DWR Table 2-2)

Select Only One	Type of Plan		Name of RUWMP or Regional Alliance if applicable
х	Individu	al UWMP	
		Water Supplier is also a member of a RUWMP	
	х	Water Supplier is also a member of a Regional Alliance	North Marin-Sonoma Alliance
	Regiona (RUWM	l Urban Water Management Plan P)	
NOTES:			

#### 2.2 Coordination and Outreach

Coordination with other water suppliers, cities, counties, and other community organizations in the region is an important part of preparing a UWMP and Water Shortage Contingency Plan (WSCP). This section identifies the agencies and organizations the District sought to coordinate with during preparation of this Plan.

#### 2.2.1 <u>Wholesale Coordination</u>

#### ☑ CWC § 10631 (h)

An urban water supplier that relies upon a wholesale agency for a source of water shall provide the wholesale agency with water use projections from that agency for that source of water in five-year increments to 20 years or as far as data is available. The wholesale agency shall provide information to the urban water supplier for inclusion in the urban water supplier's plan that identifies and quantifies, to the extent practicable, the existing and planned sources of water as required by subdivision (b), available from the wholesale agency to the urban water supplier over the same five-year increments, and during various water-year types in accordance with subdivision (f). An urban water supplier may rely upon water supply information provided by the wholesale agency in fulfilling the plan informational requirements of subdivisions (b) and (f).

Urban retail water suppliers relying on one or more wholesalers for water supply are required to provide these wholesalers with information regarding projected water supply and demand. The District meets regularly with other water purveyors to discuss water supply and demand planning. In particular, the District meets at least monthly with its water wholesaler, the SCWA, and with other Water Contractors who purchase water from the SCWA. These monthly meetings occur through the District's participation in the SCWA Technical Advisory Committee (TAC). The primary mission of the TAC is to provide input and guidance to the SCWA regarding technical issues that may have an impact on the Water Contractors (i.e., UMWP coordination, capital projects, operational changes, etc.). Additionally, the District participates in quarterly meetings of the Water Advisory Committee (WAC). The WAC's objectives are to advise the SCWA's Board of Directors on policy and fiscal matters affecting the Water Contractors. The District's

# Plan Preparation 2020 Urban Water Management Plan Marin Municipal Water District



participation in the TAC and WAC has been instrumental in coordinating water supply and demand analyses for the preparation of this Plan.

The District's local water supply is supplemented with water purchased from the SCWA. The District's water supply agreement is separate from that of the other eight Water Contractors that purchase water from SCWA. A copy of the agreement is provided in **Appendix B**. As indicated in **Table 2-3**, the District has provided demand projections through 2045 to the SCWA.

Table 2-3	Water Supplier Information Exchange (DWR Table 2-4)
The retail Supplier h water use in accorda	as informed the following wholesale supplier(s) of projected ance with Water Code Section 10631.
Wholesale Water Su	ipplier Name
Sonoma County Wa	ter Agency
NOTES:	

As discussed in Section 4.2, the District's projected water demands were developed as part of a planning effort in 2020 that was implemented in coordination with SCWA and the other Water Contractors (i.e., the 2020 Urban Water Management Plan Water Demand Analysis and Water Conservation Measures Update included as **Appendix C**). The SCWA was provided with the District's water use projections through this process. The District will continue to coordinate with the SCWA to determine the timing of capital improvement projects that may need to be implemented in order to meet the District's projected future water demands.

Additionally, as described in more detail in Chapter 7, the District has relied upon the water supply reliability projections provided by SCWA for the purposes of analyzing the reliability of its Russian River water supplies during normal and dry years through 2045.

#### 2.2.2 Agency Coordination

#### ☑ CWC § 10620 (d) (2)

Each urban water supplier shall coordinate the preparation of its plan with other appropriate agencies in the area, including other water suppliers that share a common source, water management agencies, and relevant public agencies, to the extent practicable.

As part of the development of this Plan, the District coordinated closely with the other SCWA Water Contractors. Among other methods, this coordination occurred through regular meetings of the TAC and WAC (see Section 2.2.1). These agencies also coordinated as part of the regional water conservation partnership, the SMSWP. On 9 November 2020, a letter was sent to each of these entities advising that the District was reviewing and updating its UWMP. The agencies, cities, and counties that were notified by the District during the development of this Plan are listed in **Table 2-4**. A sample copy of the notices is provided in **Appendix D**.



Table	2-4
-------	-----

#### 2-4 Notification to Cities, Counties, and Other Agencies (DWR Table 10-1)

City Name	60 Day Notice	Notice of Public Hearing
City of Belvedere	х	х
City of Larkspur	х	х
City of Mill Valley	х	х
City of Novato	х	х
City of San Rafael	х	х
City of Sausalito	х	х
Town of Corte Madera	х	х
Town of Fairfax	х	х
Town of Ross	х	х
Town of San Anselmo	х	х
Town of Tiburon	Х	Х
County Name	60 Day Notice	Notice of Public Hearing
Marin County	х	х
Other Agency Name	60 Day Notice	Notice of Public Hearing
North Marin Water District	Х	Х
Sonoma County Water Agency	X	X
NOTES:		



#### 2.2.3 <u>Public Participation</u>

#### ☑ CWC § 10642

Each urban water supplier shall encourage the active involvement of diverse social, cultural, and economic elements of the population within the service area prior to and during the preparation of both the plan and the water shortage contingency plan. Prior to adopting either, the urban water supplier shall make both the plan and the water shortage contingency plan available for public inspection and shall hold a public hearing or hearings thereon. Prior to any of these hearings, notice of the time and place of the hearing shall be published within the jurisdiction of the publicly owned water supplier pursuant to Section 6066 of the Government Code. The urban water supplier shall provide notice of the time and place of a hearing to any city or county within which the supplier provides water supplies. Notices by a local public agency pursuant to this section shall be provided pursuant to Chapter 17.5 (commencing with Section 7290) of Division 7 of Title 1 of the Government Code. A privately owned water supplier shall provide an equivalent notice within its service area. After the hearing or hearings, the plan or water shortage contingency plan shall be adopted as prepared or as modified after the hearing or hearings.

Water suppliers are required by the UWMP Act to encourage active involvement of the community within the service area prior to and during the preparation of its UWMP and WSCP. The UWMP Act also requires water suppliers to make a draft of the UWMP and WSCP available for public review and to hold a public hearing regarding the findings of the UWMP and WSCP prior to their adoption. In addition to sending notices of the District's intent to prepare its UWMP and WSCP to the various agencies listed in Section 2.2.2, the District also included a public notice in the Marin Independent Journal and on the District's website notifying the public that the draft UWMP and WSCP were available for review and that the District was seeking public input and comments, including during the public hearing. Public participation in the development of the District's 2020 UWMP and WSCP are summarized in **Appendix E**.

The Public Review Draft 2020 UWMP and WSCP was available for public review at the District's office, on the District's website (https://www.marinwater.org/water-supply), and at local libraries.

#### 2.3 UWMP Structure, Standard Units, and Basis for Reporting

As summarized in **Table 2-5**, the District is a water retailer and unless otherwise indicated, the data included in the following sections is presented in units of AF or AFY; annual values represent calendar years (CY) spanning from 1 January to 31 December.

Further, consistent with the Guidebook, the terms "water use", "water consumption", and "water demand" are used interchangeably in this UWMP.



#### Table 2-5Supplier Identification (DWR Table 2-3)

Type of Supplier			
	Supplier is a wholesaler		
Х	Supplier is a retailer		
Fiscal c	Fiscal or Calendar Year		
Х	UWMP Tables are in calendar years		
UWMP Tables are in fiscal years			
If using fiscal years provide month and date that the fiscal year begins (mm/dd)			
Units o	f measure used in UWMP		
Unit	AF		
NOTES:			



#### 3. SERVICE AREA AND SYSTEM DESCRIPTION

*CWC* § 10631 (a) A plan shall be adopted in accordance with this chapter that shall do all of the following:

Describe the service area of the supplier, including current and projected population, climate, and other social, economic, and demographic factors affecting the supplier's water management planning. The projected population estimates shall be based upon data from the state, regional, or local service agency population projections within the service area of the urban water supplier and shall be in five-year increments to 20 years or as far as data is available. The description shall include the current and projected land uses within the existing or anticipated service area affecting the supplier's water management planning. Urban water suppliers shall coordinate with local or regional land use authorities to determine the most appropriate land use information, including, where appropriate, land use information obtained from local or regional land use authorities, as developed pursuant to Article 5 (commencing with Section 65300) of Chapter 3 of Division 1 of Title 7 of the Government Code.

As shown in **Figure 3-1**, Marin Municipal Water District (MMWD or District) serves the populous eastern corridor of Marin County from the Golden Gate Bridge northward up to, but not including, the City of Novato. The District is bounded by the San Francisco Bay on the east and stretches through the San Geronimo Valley in the west. The incorporated cities and towns of San Rafael, Mill Valley, Fairfax, San Anselmo, Ross, Larkspur, Corte Madera, Tiburon, Belvedere and Sausalito are within the District's service area.

The District covers approximately 147 square miles and serves customers through about 61,700 active service connections. Five of the seven District reservoirs (Alpine, Bon Tempe, Kent, Lagunitas, and Phoenix Lake) are located on the north slope of Mt. Tamalpais. The remaining two District reservoirs (Nicasio and Soulajule) are outside the District's service area in western Marin County. The District's service area is shown in **Figure 3-2**.

#### 3.1 Population and Employment Trends Within the Service Area

Residential growth in Marin County boomed during the period following World War II up to the early years of the 1970s. Growth during the last two decades has averaged less than 1% per year, and the County Planning Department indicates that only 4% of lands within the County remain available for new development (MMWD, 2016). The population within the District remained level during the 1970s and 1980s. During that same period, the number of water service connections increased by 21%, from 46,000 to 58,000, with the majority being residential services, and the number of people per household declined from 3.1 to 2.5 (MMWD, 2016). Given the above, the District's service area is generally considered to be built-out, and very low population growth is expected within the area.

The 2020 population was calculated per methodologies outlined for use in the Senate Bill (SB) X7-7 analysis, based on Department of Finance (DOF) estimates (DOF, 2020; see Chapter 5). Using this methodology, the District's 2020 service area population is estimated to be 191,269.



#### 3.1.1 **Future Population Growth**

Table 3-1 and its associated chart provides the current and projected population for the District's service area through the year 2045. Population projections were developed as described in the 2020 Water Demand Analysis and Water Conservation Measure Update (Demand Report; EKI, 2020), which is included as Appendix C.

Population	2020	2025	2030	2035	2040	2045
Served	191,269	197,939	201,987	204,750	208,324	211,961
NOTES						

Table 3-1	Population - Current and Projected (DWR Table 3-1)

(a) 2020 population was calculated based on DOF estimates using methodologies outlined for use in the SB X7-7 analysis.

(b) Projected population growth was calculated by adjusting Marin County Association of Bay Area Governments (ABAG; 2018) projections for unincorporated area served by the District (i.e., 76% of unincorporated population).

(c) ABAG (2018) includes projections through 2040. 2045 population is estimated based on the 2035-2040 growth rate (1.7%).



**Chart 3-1 Current and Projected Population** 

#### 3.1.2 **Future Employment Growth**

Table 3-2 and its associated chart provides the current and projected employment for the District's service area through the year 2045. Current and projected employment estimates were developed as described in **Appendix C**.

# Service Area and System Description 2020 Urban Water Management Plan



**Marin Municipal Water District** 

Table 3-2

**Employment - Current and Projected** 

Service Area	2020	2025	2030	2035	2040	2045
Employment	98,019	98,822	100,449	101,246	101,474	101,703
NOTES:						
(a) Current and projected employment was calculated by adjusting Marin County ABAG						
(2018) projections for unincorporated area served by the District (i.e., 76% of						
unincorporated employment).						
(b) ABAG (2018) includes projections through 2040. 2045 employment is estimated based on						
the 2035-2040 growth rate (0.23%).						

104,000 102,000 100,000 Employment 98,000 96,000 94,000 92,000 \*Axis adjusted to show detail 90,000 2020 2025 2030 2035 2040 2045 Year

#### Chart 3-2 Current and Projected Employment

#### 3.2 Land Uses within Service Area

Land use within the District is primarily residential, but also includes agricultural, industrial, commercial, and recreational land uses. Current land use maps within the District, per the Land Use Element of the Marin Countywide Plan (County of Marin, 2014), could be found in Section 3.12 of the Countywide Plan.<sup>4</sup> Future land uses are expected to remain generally consistent with current land uses.

16

<sup>&</sup>lt;sup>4</sup> The Marin Countywide Plan could be found in the county's website: <u>https://www.marincounty.org/-</u> /media/files/departments/cd/planning/currentplanning/publications/county-wideplan/cwp 2015 update r.pdf?la=en.



#### **3.3** Service Area Social, Economic, and Demographic Factors

The District service area includes the majority of the population within Marin County, excluding the City of Novato and selected unincorporated areas. Demographics for Marin County are summarized in **Table 3-3**. The same data are also provided for the state of California as a whole and were obtained from the U.S. Census Bureau QuickFacts website (U.S. Census, 2021). Relative to the rest of California, Marin County's population is slightly older and somewhat less racially diverse. Educational attainment and median household income in Marin County are higher than for the state as a whole, while persons below the poverty level is comparatively lower.

Demographics (a)	Marin County	California	
Age and Sex			
Persons under 5 years	4.5%	6.0%	
Persons under 18 years	19.8%	22.5%	
Persons 65 years and older	23.0%	14.8%	
Female persons	51.1%	50.3%	
Race and Hispanic Origin			
White alone	85%	71.9%	
Black or African American alone	2.8%	6.5%	
American Indian and Alaska Native alone	1.0%	1.6%	
Asian alone	6.6%	15.5%	
Native Hawaiian and Other Pacific Islander alone	0.30%	0.5%	
Two or More Races	4.0%	4.0%	
Hispanic or Latino	16%	39.4%	
White alone, not Hispanic or Latino	71%	36.5%	
Families & Living Arrangements			
Persons per household	2.4	2.95	
Living in same house 1 year ago, percent of persons age 1 year+	86%	87.1%	
Language other than English spoken at home, age 5 years+	22%	44.2%	
Education			
High school graduate or higher, persons age 25 years+	93%	83.3%	
Bachelor's degree or higher, persons age 25 years+	60%	33.9%	
Income & Poverty			
Median Household Income (2019 dollars)	\$115,246	\$75,235	
Per capita income in past 12 months (2019 dollars)	\$72,466	\$36,955	
Persons in poverty 6.9%			
NOTES:			
(a) Demographic data per the U.S. Census Bureau QuickFacts website	e (U.S. Census, 2	021).	

#### Table 3-3Demographic and Housing Characteristics



#### 3.4 Climate

The District has a Mediterranean coastal climate. Summers are mild and dry, and winters are cool and wet, with an annual average of 47 inches of precipitation in the service area. The region is subject to wide variations in annual precipitation and contains a multitude of microclimates. Summer fog helps reduce summer irrigation requirements. **Table 3-4** and its associated chart present the monthly average evapotranspiration (ETo), precipitation, and maximum and minimum temperature for the service area.

	Average Temperature		Standard	Average
Month	Min (°F)	Max (°F)	Average ETo (inches)	Rainfall (inches)
January	38.6	55.6	1.4	10.45
February	41	60.5	2.0	8.36
March	42	64.5	3.2	6.36
April	43.8	69.3	4.5	2.81
May	46.5	74.1	5.5	1.31
June	49.5	79.8	6.2	0.36
July	50.2	83.4	6.3	0.05
August	50.1	82.7	5.5	0.08
September	49.6	81.4	4.5	0.46
October	47	74.8	3.3	2.46
November	42.8	64.2	1.8	5.82
December	39.1	56.3	1.3	8.89
Annual 45.0 70.6 45 47				
NOTES: (a) Temperature and precipitation data taken from the Kentfield climate				

Table 3-4 C	imate Characteristics
-------------	-----------------------

station (044500) from the Western Regional Climate Center (WRCC) forthe period 1 January 1902 through 10 June 2016.(b) ETo data taken from the California Irrigation Management Information

System (CIMIS) Point San Pedro station (157) and has a period of record from December 2002 through November 2020.

18





#### 3.5 Climate Change Considerations

#### ☑ CWC § 10630

It is the intention of the Legislature, in enacting this part, to permit levels of water management planning... while accounting for impacts of climate change.

Water managers throughout California are beginning to see the effects of a changing climate and recognize the risks associated with these changes. A portion of the District's service area is located along the coast of the San Francisco Bay, which, with a changing climate, could see rising sea levels. Over the last 30 years, mean sea level around the San Francisco Bay area, including the District's service area, has been observed to change between 0 and 3 millimeters per year (NOAA, 2021). There are a number of sensitive habitats, as well as man-made structures, located along this coastline that could be affected by a rise in sea level. The Corte Madera Marsh State Marine Park and Bothin Marsh Preserve, as well as several other marsh lands, are located near the coastline. The towns of Tiburon, Sausalito, and San Rafael have marinas that rely on the protection of breakwaters, and developed areas in San Rafael rely on a levee structure for protection from storm surges. There are also a number of ferries that run from San Francisco to Marin County which bring tourists and economic interests to the region.

As the climate changes, less frequent, more intense storms are expected. These storms have an increased potential to cause flooding, both in areas with historical flooding and in new areas. While the District is not located within a 200-year floodplain, it does rely on a network of reservoirs to capture runoff from the local watersheds. A change in precipitation patterns will necessitate water management strategies that are able to capture and store the precipitation from these storms while managing flood risk. With less frequent, more intense storms, there will also be extended dry periods that may have significant impact on available water supplies for the District. These potential impacts are discussed further in Section 6.10.1. The watersheds that MMWD relies on for water supply are heavily wooded. Under climate change, soils and wooded areas are expected to become increasingly dry, which creates additional fuel load for wildfires.

## Service Area and System Description 2020 Urban Water Management Plan Marin Municipal Water District



The District's service area provides rich habitat for a healthy ecosystem, which could be affected by a changing climate. Marin County is home to 17 endangered, threatened, or rare fauna species, including the California Tiger Salamander, Western snowy plover, and Longfin smelt. There are also 20 endangered, threatened, or rare flora in Marin County, including the Tiburon paintbrush, two-fork clover, and Sonoma spineflower. As the climate changes, climate variability could lead to changes in habitat distribution for these species, furthering endangering them. The San Francisco Bay is a key estuary for California that relies on freshwater draining from the Sacramento and San Joaquin rivers. Changes in snowmelt and rain/snow patterns in the Sierra Nevada Mountains would alter the flows in both of these rivers, ultimately affecting the seasonal freshwater flow patterns for the Bay.

Impacts associated with climate change are further discussed in the *Marin County Multi-Jurisdictional Local Hazard Mitigation Plan*, dated 2018, which is incorporated into this UWMP by reference (County LHMP; Marin County, 2018). The County LHMP assesses Marin County's vulnerabilities to various hazards and presents mitigation strategies that are planned over the next five years. Risks described in the current County LHMP include flooding, severe storms, wildfires, and landslides that are anticipated to occur due to climate change. The District is also preparing its own LHMP, which will address some of these climate change related risks and vulnerabilities and is expected to be completed by June 2022.

A discussion of climate change impacts specific to the SCWA water system is provided in the *Sonoma County Water Agency Local Hazard Mitigation Plan,* dated 16 October 2018, which is also incorporated into this UWMP by reference (SCWA LHMP; SCWA, 2018). The SCWA LHMP specifically assesses SCWA's natural hazard risks and vulnerabilities facing the SCWA infrastructure and provides a plan of action to address these vulnerabilities. As described in the SCWA LHMP, the most significant climate change\_related vulnerabilities for SCWA are associated with floods, wildfires, landslides, and drought.

The District's *Water Resources Plan 2040* dated March 2017 (Water Resources Plan; MMWD, 2017) also considered drought scenarios associated with climate change, as well as other water supply reliability threats such as wildfires, landslides, and other water quality changes that could result from climate change. The Water Resources Plan is also incorporated into this UWMP by reference.

Climate change impacts on the District's water demands are discussed in Section 4.4, and climate change impacts on the District's water supply are discussed in Section 6.10.1.

#### 3.6 Water Distribution System

The District's service area boundaries, location of water treatment plants, and potable water distribution system are illustrated in **Figure 3-3**.

The District's potable and raw water distribution system includes approximately 886 miles of water mains, 94 pump stations, and 121 treated water storage tanks<sup>5</sup> with a total storage capacity of 74.9<sup>6</sup> million gallons (MG). The District treats water at its three treatment plants, the Bon Tempe Treatment Plant

20

<sup>&</sup>lt;sup>5</sup> This total does not include 3 clearwells and 7 hydropneumatic tanks.

<sup>&</sup>lt;sup>6</sup> Excludes 7 MG from clearwells and 12,000 gallons from hydropneumatic tanks.

# Service Area and System Description 2020 Urban Water Management Plan Marin Municipal Water District



(BTTP) near Ross, the San Geronimo Treatment Plant (SGTP) in Woodacre, and the Ignacio treatment facility in Novato. Together, these facilities have a combined design capacity of 71 million gallons per day (mgd). Observed high flows have reached 45 mgd in July 2006; however, the average daily maximum flow is approximately 22.4 mgd over the last 10 years. In 2019, the total production of the three plants averaged 22.8 mgd.

The District's potable water system includes pipelines ranging in size from 3/4-inch pipes connecting customers' water meters to the District's mains, to the 42-inch transmission mains that carry source water to the treatment plants.

In addition to the District's potable water system, the District also owns and operates a recycled water system, which is described in detail in Section 6.4.





- Marin Municipal Water District
- Marin Municipal Water District Reservoirs

# Path: X:\C00055\_MMWD\Maps\.01\2021\05\Fig3-2\_MMWD\_ServiceArea.mxd **Cities and Towns**

- Belvedere Corte Madera
- Fairfax
- Larkspur
- Mill Valley
- Ross
- San Anselmo
- San Rafael
- Sausalito
  - Tiburon

- Sources
- 1. Service area boundary provided by Marin Municipal Water District.
- 2. Basemap provided by ESRI.

#### Notes

1. All locations are approximate.



DRAFT

Marin Municipal Water District May 2021 environment & water C00055.01

Figure 3-2



<u>Sources</u> 1.Marin Municipal Water District 2015 Urban Water Management Plan, prepared by RMC, dated June 2016.

#### Notes

1. All locations are approximate. 2. Not to scale. **Abbreviations** 

MMWD = Marin Municipal Water District WTP = Water Treatment Plant



**District Water System** 



Marin Municipal Water District May 2021 Men† C00055.01 Figure 3-3


#### 4. SYSTEM WATER DEMANDS

**CWC § 10631 (d) (1)** A plan shall be adopted in accordance with this chapter that shall do all of the following:

For an urban retail water supplier, quantify, to the extent records are available, past and current water use, over the same five-year increments described in subdivision (a), and projected water use, based upon information developed pursuant to subdivision (a), identifying the uses among water use sectors, including, but not necessarily limited to, all of the following:

- (A) Single-family residential.
- (B) Multifamily.
- (C) Commercial.
- (D) Industrial.
- (E) Institutional and governmental.
- (F) Landscape.
- (G) Sales to other agencies.

(H) Saline water intrusion barriers, groundwater recharge, or conjunctive use, or any combination thereof.

- (I) Agricultural.
- (J) Distribution system water loss.
- (2) The water use projections shall be in the same five-year increments described in subdivision (a).

This section describes and quantifies past and current water use and future water use projections through the year 2045. For purposes of this Urban Water Management Plan (UWMP or Plan), "potable water demand" is defined as the volume of water produced by the Marin Municipal Water District (MMWD or District), including both local surface water treated by the District and purchased water from Sonoma County Water Agency (SCWA or Sonoma Water). The District also serves raw and recycled water to customers, referred to herein as "non-potable water demand."<sup>7</sup>

Among other factors, water demand is dependent on climate, population, industry, and the types of development present in a community. Sections 4.1 and 4.2 describe MMWD's historical and projected water uses for residential, commercial, institutional, and landscape irrigation purposes (water use sectors A, B, C, E, and F, per California Water Code [CWC] §10631(e)(1)), as well as raw and recycled water uses. Distribution system water loss (water use sector J) is discussed in Section 4.1.4. As described in Section 4.3, this discussion does not include demands for water use sectors D, G, H, and I as they are not applicable or present within the District's service area. Section 4.4 describes anticipated climate change impacts to demand, and Section 4.5 discusses the District's urban water use objective requirements. Note that future water demand projections are subject to review and revision every five years as part of the UWMP update process.

<sup>&</sup>lt;sup>7</sup> Given the way the DWR required submittal tables are structured, raw and potable water demands are presented in Chapter 4, and recycled water supply and demand is presented separately, in Chapter 6.



#### 4.1 Current and Historic Total Water Demand

The following sections of the UWMP present the District's current and historical water demands, as well as the projected future demand in five-year increments from 2025 through 2045.

#### 4.1.1 Current and Historical Potable Water Demand

Current and historical potable water demand by water use sector from 2016 through 2020 is provided in **Table 4-1** and its associated charts. Water demand within the District is primarily measured using water meters that are installed at each customer account. Records of water use are maintained by the District and are based on billing data. Water use within the District is tracked for the following sectors:

- Single Family Residential (SFR);
- Multi-Family Residential (MFR);
- Commercial;
- Institutional/Governmental;
- Landscape; and
- Other.

Water use within the District's service area is predominantly associated with residential use, with 54% of the water use between 2016 and 2020 from SFR accounts and 12% from MFR accounts. Commercial accounts comprised 10% of total water use, landscape accounts comprised 5.5%, and institutional/governmental comprised 5.3%.

As shown in **Table 4-2** and its associated charts, the total and per capita water use increased from 2011 through 2013, then declined from 2014 through 2016. These trends were likely influenced by the historic drought conditions, mandatory state-wide restrictions in urban water use imposed by the State Water Resources Control Board (SWRCB), and local drought response. Total and per capita water use has remained lower than pre-drought conditions, with an increase beginning in 2017, indicating a degree of rebound following the drought. **Table 4-2** and associated charts show total and per capita water use for all potable and raw water (excluding environmental releases), and for potable and raw water adjusted to remove the volume used to meet demands of the recycled water system. Due to infrastructure upgrades at the recycled water plant, all demands by the recycled water system were met by potable water in 2019 and 2020, resulting in increased potable water use by 661 AF in 2019 and 748 AF in 2020; potable water is not anticipated to be needed to supplement the recycled water system going forward. Per capita potable and raw water use in 2020 was 128 GPCD, and adjusted potable water use (excluding recycled water system backup) was 125 GPCD.

In addition to the demands shown below, SCWA also provides a limited amount of water to the North Marin Water District's Point Reyes Public Water System (PWS) as a supplement to the system's groundwater supplies, with the transfer being performed by the District. The Point Reyes PWS has approximately 766 service connections and serves a population of 1,700 (SDWIS, 2021). Historically, transfers to the Point Reyes PWS have been minimal, up to a maximum of 9 AFY in a given year. Given that this water use is *de minimis* (i.e., less than 0.03% of total demand), it is not accounted for separately

#### System Water Demands

#### 2020 Urban Water Management Plan



Marin Municipal Water District

in **Table 4-1** or the demand projections presented in Section 4.2.5, and are not considered to be substantive within the level of precision of these estimates.

	Additional	Level of	Volume				
Use Type	Description (as needed)	Treatment When Delivered	2016	2017	2018	2019	2020
Single Family		Drinking Water	12,419	13,337	13,886	13,579	15,287
Multi-Family		Drinking Water	2,946	3,004	3,065	3,063	3,311
Commercial	Business/Industrial	Drinking Water	2,583	2,628	2,671	2,634	2,282
Institutional/Governmental		Drinking Water	1,295	1,374	1,365	1,386	1,323
Landscape		Drinking Water	1,248	1,369	1,417	1,348	1,525
Other Potable	Fireline/Hydrant	Drinking Water	28	35	38	50	62
Losses	(b)	Drinking Water	2,113	2,795	2,623	2,788	2,732
Other Non-Potable	Water sold to San Geronimo Golf Course and the Meadow Club	Raw Water	301	310	309	164	180
Other Potable	Other non- revenue water (c)	Drinking Water	13	16	13	13	
Other Potable	Potable water make-up to recycled water system (d)	Drinking Water	65	101	91	661	748
Wetlands or wildlife habitat	Environmental releases from Kent and Soulajule Lakes	Raw Water	10,314	8,377	13,245	8,080	12,699
	33,324	33,345	38,722	33,769	40,149		

Table 4-1Demands for Potable and Non-Potable Water - Actual (DWR Table 4-1)



(a) Volumes are in units of AF.

(b) "Losses" for 2016 through 2019 are the "water losses" estimated using the AWWA Free Water Audit Software and includes both real and apparent losses. Losses for 2016 through 2018 may include potable water make-up to the recycled water system. For 2020 where the AWWA Water Loss Worksheet was unavailable, total non-revenue water is used, calculated as the difference between 2020 potable water production and consumption.

(c) "Other non-revenue water" includes authorized but unbilled, unmetered consumption that does not fall under the category of "losses", such as fire flow, system flushing, hydrant leaks, etc. Other non-revenue water is calculated as the difference between "water losses" and "non-revenue water" as reported in the AWWA Water Loss Worksheets.

(d) Potable water has historically been used to supplement the recycled water system. The recycled water plant was taken offline in 2019 and 2020 to allow for upgrades, and all demands by the recycled water system were met by potable water during this time. Due to these upgrades, potable water is not anticipated to be needed to supplement the recycled water system going forward.



### Chart 4-1A Annual Water Demand by Sector: 2016-2020







#### Chart 4-1B Percentage of Total Water Demand by Sector: 2016-2020

 Table 4-2
 Historical and Current Potable Water Demand and Population

Year	Potable and Raw Water Demand (AF) (b)	Adjusted Potable and Raw Water Demand (AF) (b) (c)	Service Area Population	Per Capita Potable and Raw Water Demand (GPCD) (b)	Adjusted Per Capita Potable and Raw Water Demand (GPCD) (b) (c)
2010	25,727	25,641	183,716	125	125
2011	25,476	25,295	185,389	123	122
2012	26,893	26,722	187,089	128	128
2013	28,725	28,644	188,218	136	136
2014	24,846	24,790	190,267	117	116
2015	23,207	23,122	191,575	108	108
2016	23,010	22,945	192,402	107	106
2017	24,968	24,868	192,328	116	115
2018	25,477	25,386	192,277	118	118
2019	25,688	25,027	192,138	119	116
2020	27,451	26,703	191,269	128	125

NOTES:

(a) Unless otherwise noted, volumes are in units of AF.

(b) Excludes environmental releases from Kent and Soulajule Lakes.

(c) "Adjusted" and "Adjusted Per Capita" values are demands that are adjusted to remove

potable water used to meet recycled water system demands.







---- Per Capita Potable and Raw Water Demand ---- Adjusted Per Capita Potable and Raw Water Demand

#### 4.1.2 Current and Historical Non-Potable Water Demand

As described below, there are two sources of non-potable demand within the District: (1) raw water and (2) recycled water demands.

The District serves raw water to two customers: the San Geronimo Golf Course and the Meadow Club. The District also releases water from the Kent and Soulajule Lakes to meet environmental flow requirements that benefit silver salmon and steelhead populations in Walker Creek and Lagunitas Creek. In total, as shown in **Table 4-1**, raw water served to customers averaged 253 AFY, or approximately 1.0%, from 2016 through 2020 and environmental releases averaged 10,543 AFY, or approximately 29% of total water use over the same period.



The District made an agreement with the California Department of Fish and Game in 1976, with an amendment in 1985, to release a volume of water from Soulajule reservoir that maintains a constant streamflow in Walker Creek of 20 cubic feet per second (cfs) during the winter and spring months. The amount of water released is decreased in the summer and fall months and when the reservoir level is low.

The District releases water from Kent reservoir in accordance with the 1995 SWRCB Order 95-17 to maintain the streamflow in Lagunitas Creek of 20 to 25 cfs during winter months in wet years, with decreased flows during the summer and during dry years. Increased upstream migration flows are released from Kent for four three-day periods between November and February to provide for the upstream migration of anadromous fish.

Annual recycled water use represents approximately 2.7% of total water use within the District in 2020. It should be noted that recycled water demands are primarily associated with outdoor irrigation and therefore are highest between the months of April and October. As such, the actual maximum day supply of recycled water can represent up to 7.7% of the District's total demand during summer months. For additional information regarding the recycled water system, refer to Section 6.4.

#### 4.1.3 Potable Water Make-Up to the Recycled Water System

The recycled water system is supplemented with potable water to meet demands, as necessary. This potable make-up represented up to 748 AFY or 2.7% of total potable water demand from 2016 through 2020. In 2019 and 2020, the recycled water system was shut down to allow for treatment plant upgrades and 100% of recycled water demand was met by potable water. Following the completion of treatment plant upgrades, MMWD anticipates that all demand by the recycled water system will be met by recycled water. The recycled water system is discussed in further detail in Section 6.4.

#### 4.1.4 Distribution System Water Loss

#### ☑ CWC § 10631 (3)

(A) The distribution system water loss shall be quantified for each of the five years preceding the plan update, in accordance with rules adopted pursuant to Section 10608.34.

(B) The distribution system water loss quantification shall be reported in accordance with a worksheet approved or developed by the department through a public process. The water loss quantification worksheet shall be based on the water system balance methodology developed by the American Water Works Association.

(C) In the plan due July 1, 2021, and in each update thereafter, data shall be included to show whether the urban retail water supplier met the distribution loss standards enacted by the board pursuant to Section 10608.34.

Distribution system water losses for the previous five years are summarized in **Table 4-3**. Water loss is the sum of apparent and real losses. Apparent losses include metering inaccuracies, systematic data handling errors, and unauthorized consumption. Real losses represent water loss attributable to the distribution system and include physical water losses from the pressurized system and storage tanks up to the point of customer consumption. Since 2016, urban retail water suppliers have been required under CWC § 10608.34 and California Code of Regulations (CCR) § 638.1 et seq to quantify distribution system water losses using the American Water Works Association (AWWA) Free Water Audit Software (referred to as the "AWWA Water Loss Worksheet").



Total water losses calculated in the most recent AWWA Water Loss Worksheets are provided in **Table 4-3**, and are available through DWR's Water Use Efficiency Data Portal.<sup>8</sup> During calendar year 2019, the District's real losses were estimated to be 1,964 AF, or 7.6% of the total volume of water supplied; apparent losses were estimated to be 823 AF, or approximately 3.2% of the total volume of water supplied.

CWC § 10631 (3)(c) requires that this UWMP demonstrate whether the distribution loss standards enacted by the SWRCB pursuant to § 10608.34 have been met. However, the SWRCB has yet to establish these standards, and thus consistency with these standards cannot be demonstrated herein.

Reporting Period Start Date	Volume of Water Loss				
07/2014	1,500				
01/2016	2,113				
01/2017	2,795				
01/2018	2,623				
01/2019	2,788				
NOTES:					
(a) Volumes are in units of AF.					

#### Table 4-312 Month Water Loss Audit Reporting (DWR Table 4-4)

### 4.2 Projected Total Water Demand

The District's water demand projections were prepared as part of the 2020 Urban Water Management Plan Water Demand Analysis and Water Conservation Measures Update (Demand Report; EKI, 2020), which is provided in **Appendix C**. Projected total water demand is summarized in the following subsections based on the methods and assumptions included in the Demand Report.

#### 4.2.1 Projected Potable Water Demand

As described in more detail below and in the Demand Report (**Appendix C**), projected water demands for the District were estimated by:

- Applying an estimated growth rate to the number of accounts within each water use sector based on projected population and employment growth rates,
- Identifying known planned developments within the District to verify that account growth projections consider all currently anticipated growth,
- Evaluating and selecting water demand factors for each water use sector based on review of recent average per account water use representing three scenarios (i.e., pre-drought conditions, post-drought conditions, and a partial drought rebound scenario),

<sup>&</sup>lt;sup>8</sup> DWR's Water Use Efficiency Data Portal: <u>https://wuedata.water.ca.gov/awwa\_plans.</u>



- Estimating future passive savings using the Alliance for Water Efficiency (AWE) Water Conservation Tracking Tool (AWE model), and
- Calculating estimated future water demand that incorporates the anticipated account growth, water demand factors, and estimated future passive water savings.

This methodology is consistent with CWC § 10631(d)(4)(A), which requires that "Water use projections, where available, shall display and account for the water savings estimated to result from adopted codes, standards, ordinances, or transportation and land use plans identified by the urban water supplier, as applicable to the service area." The assumptions used as the basis for the demand projections were developed in close coordination with the District and reflect a land-use based approach consistent with community planning within the District.

Projected customer water demands through 2045 are presented in **Table 4-4** and its associated chart. These demands are broken down by sector, including water loss, raw water, and potable make-up to the recycled water system. Recycled water demands are not included in **Table 4-4** and are discussed in Section 6.4. As indicated in **Table 4-5**, the water use projections presented in **Table 4-4** include assumptions about future water savings due to passive conservation (Section 4.2.4) and water use by lower income households (Section 4.2.3).

	Additional	Projected Water Use					
Use Type	Description (as needed)	2025	2030	2035	2040	2045	
Single Family		14,710	14,855	14,928	15,120	15,349	
Multi-Family		2,998	2,846	2,715	2,628	2,568	
Commercial	Business/ Industrial	2,813	2,812	2,797	2,770	2,748	
Institutional/ Governmental		1,568	1,570	1,563	1,549	1,538	
Landscape		1,659	1,686	1,700	1,704	1,707	
Other Potable	Fireline/ Hydrant	49	50	51	51	51	
Losses	(c)	2,758	2,760	2,753	2,761	2,777	
Other Non-Potable	Raw Water sold to San Geronimo Golf Course and the Meadow Club	171	174	176	176	176	
Wetlands or wildlife habitat	Environmental releases from Kent and Soulajule Lakes	10,543	10,543	10,543	10,543	10,543	
	TOTAL	37,269	37,296	37,225	37,301	37,458	
NOTES:							

Table 4-4Use for Potable and Non-Potable - Projected (DWR Table 4-2)

#### **System Water Demands**

#### 2020 Urban Water Management Plan

#### **Marin Municipal Water District**



- (a) Volumes are in units of AF.
- (b) Projected water demands were estimated using methodology described in **Appendix C** and incorporate passive conservation savings as described in Section 4.2.4.
- (c) Losses represent all non-revenue water, which includes both real and apparent losses, as well as other non-revenue water.



#### Chart 4-4 Current and Projected Water Demand by Sector

 Table 4-5
 Inclusion in Water Use Projections (DWR Table 4-5)

Are Future Water Savings Included in Projections?	Yes
If "Yes" to above, state the section or page number, in the cell to the right, where citations of the codes, ordinances, or otherwise are utilized in demand projections are found.	Refer to Section 4.4 and Table 4-6 of <b>Appendix C</b> .
Are Lower Income Residential Demands Included In Projections?	Yes
NOTES:	

#### 4.2.2 Projected Non-Potable Water Demand

As shown in **Table 4-4** and described in **Appendix C**, the projected volume of raw water served to customers is anticipated to be 176 AFY by 2045. Projections for environmental releases from Kent and Soulajule Lakes are projected to be 10,543 AFY by 2045.



As discussed in **Appendix C**, recycled water demand was projected using the same methodology as potable water and is projected to be 771 AFY by 2045. Due to the format of the required DWR submittal tables, projected recycled water demand is not included in **Table 4-4**, but is included in **Table 4-8** in Section 4.2.5.

#### 4.2.3 <u>Water Use for Lower Income Households</u>

#### ☑ CWC § 10631.1

(a) The water use projections required by Section 10631 shall include projected water use for single-family and multifamily residential housing needed for lower income households, as defined in Section 50079.5 of the Health and Safety Code, as identified in the housing element of any city, county, or city and county in the service area of the supplier.

(b) It is the intent of the Legislature that the identification of projected water use for single-family and multifamily residential housing for lower income households will assist a supplier in complying with the requirements under Section 65589.7 of the Government Code to grant a priority for the provision of service to housing units affordable to lower income households.

California Senate Bill No. 1087 (SB 1087), Chapter 727, was passed in 2005 and amended by Government Code Section 65589.7 and Water Code Section 10631.1. SB 1087 requires local governments to provide a copy of their adopted housing element to water and sewer providers. In addition, it requires water providers to grant priority for service allocations to proposed developments that include housing units for lower income families and workers. Subsequent revisions to the UWMP Act require water providers to develop water demand projections for lower income single and multi-family households.

MMWD is required to serve any development that occurs within its service area, regardless of the income level of the future residents and does not discriminate in terms of supplying water. It is ultimately the City's or County's responsibility to approve or not approve developments within the service area.

As indicated in **Table 4-5**, the water use projections presented in Section 4.2.1 and **Table 4-4** include projected water use by lower income households. A "lower income household" is defined under California Health and Safety Code §50079.5(a) to be a household with less than 80% of median income, adjusted for family size. The Association of Bay Area Governments' (ABAG's) 2023-2031 Regional Housing Needs Allocation Proposed Methodology for the San Francisco Bay Area (ABAG, 2020) was used to estimate the proportion of new lower income households anticipated within the District. New lower income households for all cities within the District's service area, as well as the unincorporated area within Marin County, were estimated to comprise approximately 46% of all new households needed in the District's service area. **Table 4-6** shows the projected water demands for lower income households based on 46% of the total single-family and multi-family residential projected water uses included in **Table 4-4**.



#### Table 4-6 Projected Water Use for Lower Income Households

Lower Income Water Demand Sector	Projected Water Use (AFY)				
	2025	2030	2035	2040	2045
Single Family Residential	6,774	6,841	6,875	6,963	7,069
Multi-Family Residential	1,381	1,311	1,251	1,211	1,183
Total	8,155	8,152	8,125	8,174	8,252
NOTES: Volumes are in units of AF.					

#### 4.2.4 <u>Water Savings from Codes, Standards, Ordinances, or Transportation and Land Use Plans</u>

#### ☑ CWC § 10631 (d) (4)

(A) Water use projections, where available, shall display and account for the water savings estimated to result from adopted codes, standards, ordinances, or transportation and land use plans identified by the urban water supplier, as applicable to the service area.

(B) To the extent that an urban water supplier reports the information described in subparagraph (A), an urban water supplier shall do both of the following:

(i) Provide citations of the various codes, standards, ordinances, or transportation and land use plans utilized in making the projections.

(ii) Indicate the extent that the water use projections consider savings from codes, standards, ordinances, or transportation and land use plans. Water use projections that do not account for these water savings shall be noted of that fact.

"Passive conservation" refers to water savings resulting from actions and activities that do not depend on direct financial assistance or educational programs from the District. As described further in **Appendix C**, these savings result primarily from: (1) the natural replacement of existing plumbing fixtures with water-efficient models required under current plumbing code standards, and (2) the installation of water-efficient fixtures and equipment in new buildings and retrofits as required under CALGreen Building Code Standards. The water use projections discussed in Section 4.2.1 and summarized in **Table 4-4** included water savings associated with these codes and standards. Specifically, passive water savings for the District were calculated using the Alliance for Water Efficiency (AWE) Water Conservation Tracking Tool (referred to as the AWE model; AWE, 2016). The AWE model is an industry standard tool that incorporates historical population, residential building stock, number of accounts, and projected population and account growth to estimate future passive savings. More information regarding the passive savings estimates using the AWE model can be found in **Appendix C**. Projected potable water demand with and without estimated passive savings is shown in **Table 4-7** and its associated chart.

While projections account for passive savings, the District has taken a more conservative approach to demand projections by not accounting for savings associated with future active conservation measures. However, savings associated with all past active conservation efforts are embedded into the demand projections. This approach is conservative, as it projects a higher estimate of probable demand. Active conservation thus increases resiliency for District customers by further increasing efficient utilization of



available supplies. It is important to be conservative in estimating potential future shortfalls in water supply planning efforts so that the District can be confident in its ability to meet state-mandated water use reductions in addition to planning for sufficient supply to meet future demands.

# Table 4-7Projected Potable Water Demand and Projected Passive and Active Water<br/>Conservation

		Projected Potable Water Demand (AFY)					
Water Conservation Type	2025	2030	2035	2040	2045		
Projected Potable Water Demand Without Conservation	27,288	27,879	28,254	28,707	29,175		
Projected Passive Conservation Savings	733	1,301	1,749	2,125	2,437		
Projected Potable Water Demand after Passive Conservation Savings	26,555	26,579	26,506	26,582	26,738		
NOTES							

NOTES:

(a) The District also implements a number of active conservation programs, as discussed in Chapter 9 and **Appendix C**.

(b) Volumes exclude raw water and environmental releases.



#### Chart 4-7 Projected Potable Water Demand and Conservation

Passive Conservation Projected Potable Water Demand after Passive Conservation Savings

#### 4.2.5 Projected Total Water Demand

The District's total projected water demands are summarized in Table 4-8.

#### **System Water Demands**

#### 2020 Urban Water Management Plan





2045

37,458

750

38,207

TUNIC	-0 0	ioss water		ubic 4 0j	
	2020	2025	2030	2035	204
lator Raw Other					

Gross Water Use (DWP Table 1-2)

38.046

37,974

38,051

	2020	2025	2030	2035	2040	
Potable Water, Raw, Other Non-potable From DWR Tables 4-1 and 4-2	40,149	37,269	37,296	37,225	37,301	
Recycled Water Demand	748	750	750	750	750	

38,019

NOTES:

From DWR Table 6-4

(a) Volumes are in units of AF.

**TOTAL WATER USE** 

#### 4.3 Water Use Sectors Not Included in the Demand Projections

40.897

Table 1-8

Several water use sectors listed in CWC §10631(d)(1) are not included in the water demand projections described in Sections 4.2.1 and 4.2.2 because they are not applicable to the District. The following sectors were not included in the demand projections in this Plan:

- Industrial (CWC §10631(d)(1)(D)) Limited industrial water use occurs within the District, and this water use is tracked within the commercial water use sector, and thus demands by industrial users are captured in the historical and projected water demands described in Sections 4.2.1 and 4.2.2.
- <u>Sales to Other Agencies (CWC  $\frac{10631(d)(1)(G)}{1000}$  The District sells water to the National Parks</u> Service Golden Gate National Recreation Area (GGNRA), which operates its own small public water system of approximately 227 service connections (SDWIS, 2021). The District tracks sales to this system as a normal customer account, and thus demands associated with the GGNRA are captured in the historical and projected water demands described in Sections 4.2.1 and 4.2.2. The District also transfers water to the Point Reyes PWS operated by North Marin Water District. As discussed in Section 4.1.1, this constitutes less than 0.03% of MMWD's demands, and is not tracked separately.
- Saline Water Intrusion Barriers, Groundwater Recharge, or Conjunctive Use (CWC §10631(d)(1)(H)) – The District does not currently use, nor does it plan to use, water for saline water intrusion barriers, groundwater recharge, or conjunctive use.
- Agricultural (CWC §10631(d)(1)(I)) The District does not currently, nor does it plan to, provide water for agricultural uses.

#### 4.4 Climate Change Impacts to Demand

#### ☑ CWC § 10635(b)

(4) Considerations of the historical drought hydrology, plausible changes on projected supplies and demands under climate change conditions, anticipated regulatory changes, and other locally applicable criteria.



The methodology used to develop demand projections herein considers the impacts of climate change on projected demands. California experienced a historic drought between 2011-2017. In 2014, Governor Brown issued Executive Order B-26-14 declaring a Drought State of Emergency and requested all Californians to voluntarily reduce water use by 20%. In 2015, the SWRCB implemented emergency conservation regulations that, among other things, required water agencies to reduce their water use and prohibited certain types of water uses. As a result, the District experienced an overall decrease in demands during the historic drought, most significantly during 2015. As explained further in **Appendix C**, the demand factors evaluated herein consider both the 2011-2013 period, in which customers increased their water use (in part due to the drought conditions, prior to the imposed restrictions), as well as the observed rebound in demand following the drought (2017-2019). Thus, the periods used to develop the demand projections reflect conditions representative of the hotter, drier weather expected as a result of climate change.

### 4.5 Urban Water Use Objectives (Future Requirements)

#### ☑ CWC § 10609.20

(a) Each urban retail water supplier shall calculate its urban water use objective no later than January 1, 2024, and by January 1 every year thereafter.

(b) The calculation shall be based on the urban retail water supplier's water use conditions for the previous calendar or fiscal year....

#### ☑ CWC § 10609.22

(a) An urban retail water supplier shall calculate its actual urban water use no later than January 1, 2024, and by January 1 every year thereafter.

(b) The calculation shall be based on the urban retail water supplier's water use for the previous calendar or fiscal year....

#### ☑ CWC § 10609.24

(a) An urban retail water supplier shall submit a report to the department no later than January 1, 2024, and by January 1 every year thereafter. The report shall include all of the following:

(1) The urban water use objective calculated pursuant to Section 10609.20 along with relevant supporting data.

(2) The actual urban water use calculated pursuant to Section 10609.22 along with relevant supporting data.

(3) Documentation of the implementation of the performance measures for CII water use.

(4) A description of the progress made towards meeting the urban water use objective.

(5) The validated water loss audit report conducted pursuant to Section 10608.34.

(b) The department shall post the reports and information on its internet website.

(c) The board may issue an information order or conservation order to, or impose civil liability on, an entity or individual for failure to submit a report required by this section.

Beginning in 2023, urban water retailers will be required to report on "annual water use objectives" by 1 November of each year and to achieve these objectives by 1 January 2027. The annual water use objectives will be calculated based on standards for indoor residential water use, outdoor residential



water use, and distribution system water loss. Additionally, it is anticipated that performance-based standards for the commercial, industrial, and institutional sectors, separate from the annual water use objectives, will also be developed by DWR and implemented in the future. However, the specific standards that will be used to determine a retailer's annual urban water use objectives are currently under development by DWR, and thus, the annual urban water use objectives for the District cannot be calculated or estimated. Once the urban water use objectives are released, the District will evaluate its historical and current water use compared to the new objectives and will evaluate the need to adjust its conservation and water loss management measures to meet the new objectives.

One of the components for calculating the future water use objectives is provided for in CWC § 10609.4.(a), which states "(1) Until January 1, 2025, the standard for indoor residential water use shall be 55 gallons per capita daily. (2) Beginning January 1, 2025, and until January 1, 2030, the standard for indoor residential water use shall be the greater of 52.5 gallons per capita daily or a standard recommended pursuant to subdivision (b). (3) Beginning January 1, 2030, the standard for indoor residential water use shall be the greater of 50 gallons per capita daily or a standard recommended pursuant to subdivision (b). (3) Beginning January 1, 2030, the standard for indoor residential water use shall be the greater of 50 gallons per capita daily or a standard recommended pursuant to subdivision (b)."<sup>9</sup> **Table 4-9** shows an estimate of future per capita residential water use, broken out by estimated indoor and outdoor water use, per the analysis provided in **Appendix C**. Based on these estimates, per capita indoor residential potable water use within the District is expected to be at or below the indoor use standards presented in the legislation. Although indoor residential water use is expected to be within the indoor residential water use standard, it should be noted that because standards have not yet been developed for the outdoor water use or water loss components of the future water use objectives, it cannot be known whether projected demands for the District will be in compliance with the pending requirements.

Year	Residential Potable Water Demand	Service Area Population	Per Capita Residential Potable Water Use (GPCD)	Approximate Per Capita Indoor Residential Potable Water Use (GPCD)	Approximate Per Capita Outdoor Residential Potable Water Use (GPCD)
2020	18,599	191,269	87	56	31
2025	17,708	197,939	80	52	28
2030	17,700	201,987	78	50	28
2035	17,643	204,750	77	50	27
2040	17,748	208,324	76	49	27
2045	17,917	211,961	75	49	27
NOTES:					

Table 4-9	Current and Projected Residential Per Capita Water Use
-----------	--

<sup>&</sup>lt;sup>9</sup> While the legislation appears to be clear on the method to calculate the indoor residential water use component, the SWRCB has begun the California Environmental Quality Act (CEQA) process for the new water use objective requirements and has expressed concern that using the 55 gallons per capita per day (GPCD) number in the legislation will constitute "backsliding" (compared to the reduction required by SB X7-7) and thus may need to be lowered.



- (a) Unless otherwise noted, volumes are in units of AF.
- (b) Indoor and outdoor residential water use estimates are based on the estimated residential indoor and outdoor water use proportions of 64% residential indoor water use and 36% residential outdoor water use, as documented in **Appendix C**.

Baseline Water Use and SB X7-7 Water Conservation Targets 2020 Urban Water Management Plan



Marin Municipal Water District

#### 5. BASELINE WATER USE AND SB X7-7 WATER CONSERVATION TARGETS

#### **☑** CWC § 10608.24 (b)

Each urban retail water supplier shall meet its urban water use target by December 31, 2020.

#### ☑ CWC § 10608.28

(a) An urban retail water supplier may meet its urban water use target within its retail service area, or through mutual agreement, by any of the following:

(1) Through an urban wholesale water supplier.

(2) Through a regional agency authorized to plan and implement water conservation, including, but not limited to, an agency established under the Bay Area Water Supply and Conservation Agency Act (Division 31 (commencing with Section 81300)).

(3) Through a regional water management group as defined in Section 10537.

(4) By an integrated regional water management funding area.

(5) By hydrologic region.

(6) Through other appropriate geographic scales for which computation methods have been developed by the department.

(b) A regional water management group, with the written consent of its member agencies, may undertake any or all planning, reporting, and implementation functions under this chapter for the member agencies that consent to those activities. Any data or reports shall provide information both for the regional water management group and separately for each consenting urban retail water supplier and urban wholesale water supplier.

With the adoption of the Water Conservation Act of 2009, also known as Senate Bill (SB) X7-7, the state is required to reduce urban water use by 20% by the year 2020. Each urban retail water supplier was required to develop a baseline daily per capita water use ("baseline water use") in their 2010 Urban Water Management Plan (UWMP or Plan) and establish per capita water use targets for 2015 and 2020 in order to help the state achieve the 20% reduction. Under SB X7-7 urban retail water suppliers may either comply with their 2020 targets on an individual basis or as part of a regional alliance. As identified in **Table 2-2**, Marin Municipal Water District (MMWD or District) is part of the North Marin-Sonoma Alliance (Alliance), which includes eight other regional water retailers.

In support of implementing the requirements of SB X7-7, the California Department of Water Resources (DWR) produced a set of methodologies for developing baseline and compliance water use and targets, which are included in *Methodologies for Calculating Baseline and Compliance Urban Per Capita Water, California Department of Water Resources Division of Statewide Integrated Water Management Water Use and Efficiency Branch* (Methodologies; DWR, 2016b). The District and its regional Alliance have not made any changes to the information pertaining to the baseline water use or interim 2015 target compliance reported in the District's 2015 UWMP.

In this chapter, the District demonstrates compliance with its 2020 per capita water use target based on its participation in the regional Alliance. Per the Methodologies, regional alliances may calculate their baseline and targets using several methods. The Alliance has opted to use Option 1, in which individual

#### Baseline Water Use and SB X7-7 Water Conservation Targets

2020 Urban Water Management Plan



#### **Marin Municipal Water District**

agencies calculate their own baseline and target year per capita water use, and the regional alliance values are calculated as a population-weighted average of all members' water use.

As part of the compliance reporting for SB X7-7, water suppliers are required to complete and submit a set of standardized verification tables in their 2020 UWMPs. The information in these tables is discussed and summarized in the following subsections, and the complete set of SB X7-7 standardized tables is included in **Appendix F**.

#### 5.1 Service Area Population

#### ☑ CWC § 10608.20 (e)

An urban retail water supplier shall include in its urban water management plan due in 2010 pursuant to Part 2.6 (commencing with Section 10610) the baseline daily per capita water use, urban water use target, interim urban water use target, and compliance daily per capita water use, along with the bases for determining those estimates, including references to supporting data.

#### ☑ CWC § 10608.20 (g)

An urban retail water supplier may update its 2020 urban water use target in its 2015 urban water management plan required pursuant to Part 2.6 (commencing with Section 10610).

#### Methodology 2 Service Area Population.

DWR will examine discrepancy between the actual population estimate and DOF's projections for 2010; if significant discrepancies are discovered, DWR may require some or all suppliers to update their baseline population estimates. (DWR, 2016b)

As reported in its 2015 UWMP, the District calculated its baseline population using the California Department of Finance (DOF) data method described in the Methodologies (i.e., Methodology 2 – Service Area Population) in its 2010 UWMP and updated the calculations in its 2015 UWMP. The District's service area includes several city boundaries within Marin County. Populations for these cities were taken directly from DOF reports. The District's service area also includes a portion of unincorporated Marin County. As reported in the 2015 UWMP, the population within unincorporated Marin County served by the District was estimated as a proportion (76%) of the total unincorporated County population.

As required by California Water Code (CWC) §10680.20 and the Methodologies, the District recalculated its baseline population using 2010 Census data (which was made available in 2012) in its 2015 UWMP. The District's 5- and 10-year baseline populations, per the 2015 UWMP are presented in **Table 5-1** below. The District's 2020 population is estimated to be 191,269 (**Table 5-1**).

Baseline Water Use and SB X7-7 Water Conservation Targets 2020 Urban Water Management Plan Marin Municipal Water District



Year		Population		
	10 to 15 Year E	Baseline Population		
Year 1	1995	172,700		
Year 2	1996	174,300		
Year 3	1997	176,400		
Year 4	1998	178,000		
Year 5	1999	179,300		
Year 6	2000	181,300		
Year 7	2001	181,400		
Year 8	2002	181,300		
Year 9	2003	181,400		
Year 10	2004	180,600		
	5 Year Base	line Population		
Year 1	2003	181,400		
Year 2	2004	180,600		
Year 3	2005	179,700		
Year 4	2006	179,800		
Year 5	2007	180,100		
2020 Compliance Year Population				
2	020	191,269		
NOTES:				
(a) Historical population numbers have been rounded				
to the nea	rest hundred.			

#### Table 5-1SB X7-7 Service Area Population (SB X7-7 Table 3)

#### 5.2 Baseline Water Use

The baseline water use is the water supplier's average gross daily water use per capita measured in gallons. This baseline includes all water entering the delivery system, including water losses.<sup>10</sup> Water suppliers were required to define a 10- or 15-year base (or baseline) period for water use that was then used to develop their future target per capita water use in their 2010 and 2015 UWMPs.<sup>11</sup> Water suppliers were also required to calculate their water use over a five-year baseline period and use that value to determine a minimum required reduction in water use by 2020. For the development of the District's baseline water use, a 10-year average was used from 1995 to 2004. The 10-year baseline water use calculated and reported in the District's 2015 UWMP was 149 gallons per capita per day (GPCD). No deductions were made to the District's gross water use. The District was also required to determine its five-year base daily per capita water use in its prior UWMPs, which was determined to be 148 GPCD.

<sup>&</sup>lt;sup>10</sup> A water supplier may deduct from its gross water use water conveyed to other urban water suppliers, water placed into long-term storage, recycled water delivered within the supplier's service area, water delivered for agricultural use, water conveyed to other urban water suppliers, and water used for industrial processes.

<sup>&</sup>lt;sup>11</sup> Utilizing a 15-year baseline period is only allowed for water suppliers that meet at least 10 percent of their 2008 measured retail water demand through recycled water; the District does not meet this criterion and thus selected a 10-year baseline.

#### Baseline Water Use and SB X7-7 Water Conservation Targets

2020 Urban Water Management Plan



#### Marin Municipal Water District

**Table 5-2** below shows the weighted baseline as calculated for the Alliance in 2015. Taking the population-weighted average of all of its members, the Alliance's baseline was calculated to be 156 GPCD.

SB X7-7 RA1 - Weighted Baseline						
Participating Member Agency Name	10-15 year Baseline GPCD*	Average Population During 10-15 Year Baseline Period	(Baseline GPCD) X (Population)	Regional Alliance Weighted Average 10-15 Year Baseline GPCD		
Marin Municipal Water District	149	178,670	1,043,146			
North Marin Water District	159	6,559	26,690,318			
City of Petaluma	173	54,061	9,370,435			
City of Rohnert Park	180	52,622	9,491,997			
City of Santa Rosa	161	40,811	6,582,847			
City of Sonoma	145	143,109	20,806,963			
Valley of the Moon Water District	225	9,679	2,173,212			
Town of Windsor	146	20,969	3,058,648			
City of Cotati (Under Threshold)	156	24,572	3,834,809			
Regional Alliance Total	1,495	531,051	83,052,375	156		

Table 5-2	RA1 – Weighted Baseline
-----------	-------------------------

\*All participating agencies must submit individual SB X7-7 Tables, as applicable, showing the individual agency's calculations. These tables are: SB X7-7 Tables 0 through 6, Table 7, any required supporting tables (as stated in SB X7-7 Table 7), and SB X7-7 Table 9, as applicable. These individual agency tables will be submitted with the individual or Regional Urban Water Management Plan.

NOTES

Baseline Water Use and SB X7-7 Water Conservation Targets 2020 Urban Water Management Plan Marin Municipal Water District



#### 5.3 Water Use Targets

#### ☑ CWC § 10608.20 (b)

An urban retail water supplier shall adopt one of the following methods for determining its urban water use target pursuant to subdivision (a):

(1) Eighty percent of the urban retail water supplier's baseline per capita daily water use.

(2) The per capita daily water use that is estimated using the sum of the following performance standards:

(A) For indoor residential water use, 55 gallons per capita daily water use as a provisional standard. Upon completion of the department's 2017 report to the Legislature pursuant to Section 10608.42, this standard may be adjusted by the Legislature by statute.

(B) For landscape irrigated through dedicated or residential meters or connections, water efficiency equivalent to the standards of the Model Water Efficient Landscape Ordinance set forth in Chapter 2.7 (commencing with Section 490) of Division 2 of Title 23 of the California Code of Regulations, as in effect the later of the year of the landscape's installation or 1992. An urban retail water supplier using the approach specified in this subparagraph shall use satellite imagery, site visits, or other best available technology to develop an accurate estimate of landscaped areas.

(C) For commercial, industrial, and institutional uses, a 10-percent reduction in water use from the baseline commercial, industrial, and institutional water use by 2020.

(3) Ninety-five percent of the applicable state hydrologic region target, as set forth in the state's draft 20x2020 Water Conservation Plan (dated April 30, 2009). If the service area of an urban water supplier includes more than one hydrologic region, the supplier shall apportion its service area to each region based on population or area.

(4) A method that shall be identified and developed by the department, through a public process, and reported to the Legislature no later than December 31, 2010. The method developed by the department shall identify per capita targets that cumulatively result in a statewide 20-percent reduction in urban daily per capita water use by December 31, 2020. In developing urban daily per capita water use targets, the department shall do all of the following:

- (A) Consider climatic differences within the state.
- (B) Consider population density differences within the state.
- (C) Provide flexibility to communities and regions in meeting the targets.

(D) Consider different levels of per capita water use according to plant water needs in different regions.

(E) Consider different levels of commercial, industrial, and institutional water use in different regions of the state.

(F) Avoid placing an undue hardship on communities that have implemented conservation measures or taken actions to keep per capita water use low.

#### ☑ CWC § 10608.22

Notwithstanding the method adopted by an urban retail water supplier pursuant to Section 10608.20, an urban retail water supplier's per capita daily water use reduction shall be no less than 5 percent of base daily per capita water use as defined in paragraph (3) of subdivision (b) of Section 10608.12. This section does not apply to an urban retail water supplier with a base daily per capita water use at or below 100 gallons per capita per day.



#### 5.3.1. Individual Water Use Targets

**Table 5-3** shows the District's 5- and 10-year baseline periods, its baseline GPCD for these periods, and its confirmed 2020 target, which were previously developed and reported in its 2015 UWMP. This individual target is used for the development of the Alliance's regional water use target, as shown in Section 5.3.2.

Baseline Period	Start Year	End Year	Average Baseline GPCD	Confirmed 2020 Target GPCD
10-15 year	1995	2004	149	124
5 Year	2003	2007	148	124
NOTES:				

#### Table 5-3Baselines and Targets Summary (DWR Table 5-1)

#### 5.3.2 <u>Regional Water Use Targets</u>

Instead of, or in addition to, individual water use targets, urban water retail suppliers may plan, comply, and report on SB X7-7 requirements on a regional basis as part of a "Regional Alliance." As described in Section 2.2.2, the District is one of eight Water Contractors that purchase water from the Sonoma County Water Agency (SCWA or Sonoma Water). As such, the Water Contractors have formed a Regional Alliance under the provisions of SB X7-7 because they are recipients of water from a common wholesale water supplier.

A water conservation regional partnership among the eight Water Contractors is already in existence in the form of the Sonoma Marin Saving Water Partnership (SMSWP). As identified in **Table 2-2**, this regional group, which collaborates on regional water conservation efforts, formed a regional alliance for the purposes of meeting regional water use targets. The members of the North Marin-Sonoma Alliance include: the District, City of Sonoma, Valley of the Moon Water District, City of Santa Rosa, Town of Windsor, City of Rohnert Park, City of Cotati, City of Petaluma, and North Marin Water District.<sup>12</sup>

The DWR established three options for calculating a regional alliance water use target. The District, along with the other Water Contractors in the regional alliance, selected Option 1, which preserves maximum flexibility at the supplier level. Under Option 1, each member of the regional alliance calculates their individual targets and then weighs the individual targets by each member's population. The weighted targets are then averaged to determine the regional alliance target. Detailed calculations conducted by the Alliance are included in **Appendix F**. The Alliance's 2015 Interim and 2020 targets, as reported to DWR by the Alliance in 2015 are provided in **Table 5-4** below.

<sup>&</sup>lt;sup>12</sup> The letter approving the District's membership in the regional alliance: <u>http://www.savingwaterpartnership.org/wp-content/uploads/20x2020-regional-alliance-agreement.doc.pdf</u>.

#### Baseline Water Use and SB X7-7 Water Conservation Targets

2020 Urban Water Management Plan



**Marin Municipal Water District** 

#### Table 5-4

### DWR Regional Alliance Weighted 2020 Target

SB X7-7 RA1 - Weighted 2020 Target							
Participating Member Agency Name	2020 Target GPCD*	2015 Population	(Target) X (Population)	Regional Alliance Weighted Average 2020 Target			
City of Cotati	130	7,288	947,440				
Marin Municipal Water District	124	189,000	23,436,000				
North Marin Water District	139	61,381	8,531,959				
City of Petaluma	141	61,798	8,713,518				
City of Rohnert Park	119	41,675	4,959,325				
City of Santa Rosa	126	173,071	21,806,946				
City of Sonoma	180	11,147	2,006,460				
Valley of the Moon Water District	124	23,478	2,911,272				
Town of Windsor	130	27,486	3,573,180				
Regional Alliance Total	1,213	596,324	76,886,100	129			
*All participating agencies must submit individual SR X7-7 Tables, as applicable showing the individual agency's							

\*All participating agencies must submit individual SB X7-7 Tables, as applicable, showing the individual agency's calculations. These tables are: SB X7-7 Tables 0 through 6, Table 7, any required supporting tables (as stated in SB X7-7 Table 7), and SB X7-7 Table 9, as applicable. These individual agency tables will be submitted with the individual or Regional Urban Water Management Plan.

NOTES:

Baseline Water Use and SB X7-7 Water Conservation Targets

2020 Urban Water Management Plan



**Marin Municipal Water District** 

#### 5.4 2020 Target Compliance

#### ☑ CWC § 10608.24 (b)

Each urban retail water supplier shall meet its urban water use target by December 31, 2020.

#### ☑ CWC § 10608.24 (d)

(1) When determining compliance daily per capita water use, an urban retail water supplier may consider the following factors:

(A) Differences in evapotranspiration and rainfall in the baseline period compared to the compliance reporting period.

(B) Substantial changes to commercial or industrial water use resulting from increased business output and economic development that have occurred during the reporting period.

(C) Substantial changes to institutional water use resulting from fire suppression services or other extraordinary events, or from new or expanded operations, that have occurred during the reporting period.

(2) If the urban retail water supplier elects to adjust its estimate of compliance daily per capita water use due to one or more of the factors described in paragraph (1), it shall provide the basis for, and data supporting, the adjustment in the report required by Section 10608.40.

#### **☑** CWC § 10608.40

Urban water retail suppliers shall report to the department on their progress in meeting their urban water use targets as part of their urban water management plans submitted pursuant to Section 10631. The data shall be reported using a standardized form developed pursuant to Section 10608.52.

**Table 5-5** demonstrates the Alliance's compliance with its 2020 GPCD target. The Alliance's 2020 GPCD of 113 is below the regional target of 129 GPCD and the Alliance and District are therefore in compliance with SB X7-7 requirements.

#### Table 5-5 2020 Compliance for North Marin-Sonoma Regional Alliance (DWR Table 5-2)

2020 GPCD				Did Supplier		
Actual 2020 GPCD	2020 TOTAL Adjustments	Adjusted 2020 GPCD (Adjusted if applicable)	2020 Confirmed Target GPCD	Achieve Targeted Reduction for 2020?		
113	0	113	129	Yes		
NOTES:						



#### 6. WATER SUPPLY CHARACTERIZATION

*CWC* § 10631 (b) A plan shall be adopted in accordance with this chapter that shall do all of the following:

*Identify and quantify, to the extent practicable, the existing and planned sources of water available to the supplier over the same five-year increments described in subdivision (a).* 

#### ☑ CWC § 10631 (b) (2)

When multiple sources of water supply are identified, a description of the management of each supply in correlation with the other identified supplies.

This section describes existing and future sources of water available to Marin Municipal Water District (MMWD or District). It includes a description of each water source, source limitations, water quality, and future opportunities for additional supply development. The District's water supplies presently come from a combination of local surface water supplies, imported water from Sonoma County Water Agency (SCWA or Sonoma Water), and recycled water. Each water supply is described further in the following sections.

#### 6.1 Purchased Water

**W CWC** § 10631 (h) A plan shall be adopted in accordance with this chapter and shall do all of the following:

An urban water supplier that relies upon a wholesale agency for a source of water shall provide the wholesale agency with water use projections from that agency for that source of water in five-year increments to 20 years or as far as data is available. The wholesale agency shall provide information to the urban water supplier for inclusion in the urban water supplier's plan that identifies and quantifies, to the extent practicable, the existing and planned sources of water as required by subdivision (b), available from the wholesale agency to the urban water supplier over the same five-year increments, and during various water-year types in accordance with subdivision (f). An urban water supplier may rely upon water supply information provided by the wholesale agency in fulfilling the plan informational requirements of subdivisions (b) and (f).

#### 6.1.1 Sonoma County Water Agency Surface Water Supply

The District receives its supplemental water supply from Sonoma County Water Agency's (SCWA's or Sonoma Water's) transmission system, which provides treated water purchased from SCWA's Russian River Project. The Russian River flows are augmented by Pacific Gas and Electric's (PG&E's) Potter Valley Project, which diverts a portion of the Eel River flows to the East Fork of the Russian River. Water is diverted and extracted from the stretch of the Russian River located just upstream of Wohler Bridge via six radial wells known as "Ranney collectors." The diverted river water percolates through sand and gravel and only needs the addition of chlorine to meet drinking water quality standards. Although the water extracted via Ranney collectors does percolate through the ground, due to the connection to the surface water rights to the Russian River and Dry Creek water, described further in Section 6.1.2 (SCWA, 2016). The SCWA supply also includes a relatively small amount of groundwater from groundwater supply wells located in the central Santa Rosa Plain subbasin (SCWA, 2016).

The District's present contract with SCWA is based on two documents: the 1975 Off-Peak Water Supply Agreement (Off-Peak Agreement) and its amendments, and the 1991 Agreement for the Sale of Water



between SCWA and the District. In 1996, these two contracts were combined into a single new agreement, the Supplemental Water Supply Agreement (Agreement).

In its original form, the Off-Peak Agreement allowed the District to take delivery, in the months of October through April, of up to 4,300 acre-feet (AF) of water surplus to the needs of all other SCWA customers. The contract was amended twice before its inclusion in the Supplemental Water Supply Agreement of 1996. The first amendment changed the basis of delivery of this water from "surplus" to "firm," meaning that the District's water deliveries would be as reliable as that provided to SCWA's other contractors. The second amendment allowed deliveries up to 360 AF per month from May to September.

The 1991 Agreement for Water Supply allowed the District to take deliveries of up to 10,000 AF of water per year beyond the amount included in the Off-Peak Agreement. These water deliveries were classified as "as available." This meant that the contracted water supply was secondary to water provided to SCWA's other contractors and to water provided under the Off-Peak Agreement but would be provided unless certain predefined conditions existed.

The Supplemental Water Supply Agreement combined the two prior agreements such that the District can now take deliveries of up to 14,300 acre-feet per year (AFY) from SCWA. All of these deliveries are also now classified as "firm" water. In addition to the annual delivery limit, the Agreement also places seasonal limitations on water delivery rates to the District<sup>7</sup> with deliveries limited to 23.1 million gallons per day (mgd) from December to March, 12.8 mgd from May to September, 20.1 mgd in April and November, and 17.1 mgd in October.

The Supplemental Water Supply Agreement expired on 30 June 2014, but a Temporary Extension of the Supplemental Water Supply Agreement extended the Off-Peak Agreement and Water Sale Agreement until 30 June 2015. A formal Agreement renewal was subsequently approved and became effective on 1 July 2015. This renewed Agreement, dated 16 June 2015, will remain in force through 30 June 2025, and includes a renewal provision that will extend the Agreement through 30 June 2040. A copy of the renewed Agreement is included in **Appendix B**.

In addition to contractual delivery limits, Russian River water deliveries to the District are subject to available pipeline capacity in facilities owned by SCWA and North Marin Water District (NMWD). The Interconnection Agreement from 2014 describes the District's rights to use the excess capacity in NMWD's facilities. The Interconnection Agreement runs contiguous with the SCWA Restructured Agreement for Water Supply, which will expire on 30 June 2040 and has renewal options.

Water imported from SCWA is naturally filtered in the deep sand and gravel below the riverbed and requires no further clarification. This water enters the District's system at the Ignacio Water Quality and Pumping Station, where water quality is monitored continually and adjusted as needed.

#### 6.1.2 Sonoma County Water Agency Surface Water Rights

According to SCWA's 2015 UWMP, currently, four water rights permits (Permits 12947A, 12949, 12950, and 16596) issued by the State Water Resources Control Board (SWRCB) authorize SCWA to store up to 122,500 AFY of water in Lake Mendocino and up to 245,000 AFY of water in Lake Sonoma, and to divert or redivert up to 180 cubic feet per second (cfs) of water from the Russian River with a limit of 75,000 AFY. The permits also establish minimum instream flow requirements for fish and wildlife protection and recreation. These minimum instream flow requirements vary based on the hydrologic classifications of normal, dry, and critical water supply conditions as defined by SCWA's water rights permits and SWRCB



Decision 1610, adopted in 1986 (SCWA, 2016). SCWA meets the various instream flow requirements by making releases from Coyote Valley Dam and Warm Springs Dam (SCWA, 2016). The Russian River Biological Opinion requires modification of minimum instream flow requirements on the Russian River and Dry Creek to maintain the Incidental Take Statement provided by the Biological Opinion (SCWA, 2016). SCWA's evaluation of future Russian River supply availability is based upon the assumption that that proposed changes to the minimum instream flow requirements under Decision 1610 set forth in the Biological Opinion are implemented, and that SCWA will obtain water rights approvals necessary to increase its total Russian River diversions above 75,000 AFY by 2035 (SCWA, 2016). The SCWA 2015 UWMP anticipates that SCWA would request at that time an additional 1,000 AFY to increase the overall supply from the Russian River to 76,000 AFY.

#### 6.1.3 Sonoma County Water Agency Groundwater Supply

SCWA pumps a portion of its supply from the Santa Rosa Plain Subbasin of the Santa Rosa Valley Basin (DWR Basin 1-55.01). Groundwater is used primarily as a drought period supply, or when Russian River supplies are otherwise constrained (SCWA, 2016). In 2015, groundwater made up less than 2% of SCWA's supplies; through 2045, groundwater is projected to make up 3% of SCWA's supplies in normal year conditions (SCWA, 2016). It cannot be discerned what specific amount of SCWA supply provided to the District consists of groundwater; however, it is assumed to be proportionate to the overall percentage of groundwater used within SCWA's system. SCWA's groundwater supply is discussed further in Section 6.2.



#### 6.2 Groundwater

#### **☑** CWC § 10631

(b) (4) If groundwater is identified as an existing or planned source of water available to the supplier, all of the following information:

(A) The current version of any groundwater sustainability plan or alternative adopted pursuant to Part 2.74 (commencing with Section 10720), any groundwater management plan adopted by the urban water supplier, including plans adopted pursuant to Part 2.75 (commencing with Section 10750), or any other specific authorization for groundwater management for basins underlying the urban water supplier's service area.

(B) A description of any groundwater basin or basins from which the urban water supplier pumps groundwater. For basins that a court or the board has adjudicated the rights to pump groundwater, a copy of the order or decree adopted by the court or the board and a description of the amount of groundwater the urban water supplier has the legal right to pump under the order or decree. For a basin that has not been adjudicated, information as to whether the department has identified the basin as a high- or medium-priority basin in the most current official departmental bulletin that characterizes the condition of the groundwater basin, and a detailed description of the efforts being undertaken by the urban water supplier to coordinate with groundwater sustainability agencies or groundwater management agencies listed in subdivision (c) of Section 10723 to maintain or achieve sustainable groundwater conditions in accordance with a groundwater sustainability plan or alternative adopted pursuant to Part 2.74 (commencing with Section 10720).

(C) A detailed description and analysis of the location, amount, and sufficiency of groundwater pumped by the urban water supplier for the past five years. The description and analysis shall be based on information that is reasonably available, including, but not limited to, historic use records.

The District does not pump groundwater and does not plan to use groundwater as a supply source in the future (see **Table 6-1**).

There are three groundwater basins identified in the California Department of Water Resources' (DWR)'s Bulletin 118 that are at least partially within the District's service area. These three basins include Ross Valley Groundwater Basin (DWR Basin 2-18), San Rafael Valley Groundwater Basin (DWR Basin 2-29), and part of the Novato Basin (DWR Basin 2-30). All three basins are categorized by the California Statewide Groundwater Elevation Monitoring (CASGEM) program as low or very low priority basins (DWR, 2019). Studies that have been conducted by the District over the last 40 years have determined that groundwater within the boundaries of the District's service area is very limited as it is either found in fractures in the Franciscan Formation (bedrock) or in shallow alluvial deposits in valleys. Therefore, groundwater is not currently or planned to be used as a water supply source by the District.



x	Supplier does not pump groundwater. The supplier will not complete the table below.						
	All or part of the groundwater described below is desalinated.						
Groundwater Type	Location or Basin         2016         2017         2018         2019         202           Name         2016         2017         2018         2019         202					2020	
	TOTAL						
NOTES:							

Table 6-1 Groundwater Volume Rumped (DWR Table 6-1)

Although the District does not pump groundwater directly, as noted in Section 6.1.3, a small portion of the SCWA water supply (i.e., less than 2%) is comprised of groundwater from the Santa Rosa Plain Subbasin of the Santa Rosa Valley Basin (DWR Basin 1-55.01). Given this, characteristics and groundwater management of the Santa Rosa Plain Subbasin are provided below.

#### 6.2.1 **Basin Description and Status**

The Santa Rosa Subbasin is not adjudicated, and in its recent evaluation of California groundwater basins, DWR determined that the Basin is not in a condition of critical overdraft (DWR, 2019). The Santa Rosa Plain subbasin is currently categorized by the DWR program as a medium priority basin (DWR, 2019).

Under DWR's prioritization process, basins are ranked on eight components, and if a basin is assigned more than 14 total points, but less than 21 total points, it is defined as "medium priority." The main factors driving the Santa Rosa Plain subbasin's designation include population density (3 out of 5 possible ranking points), population growth (3 out of 5 possible points), public supply well density (5 out of 5 possible points), total production well density (5 out of 5 possible points), groundwater reliance (5 out of 5 possible points), and groundwater reliance (3 out of 5 possible points) (DWR, 2019).

Geologically, the Santa Rosa Plain subbasin has one main water-bearing unit, the Merced Formation, and several units with lower water-bearing capacities, including the Glen Ellen Formation and the Alluvium. The shallow Alluvium consists of poorly sorted coarse sand and gravel and moderately-sorted fine sand, silt, and clay. The alluvial deposits are not perennially saturated, have low permeability, and are generally unconfined or slightly confined (DWR, 2006). The Glen Ellen Formation underlies the Alluvium and consists of partially cemented beds of poorly sorted gravel, sand, and silt, and clay that vary widely in thickness and extent, with thicknesses varying from 3,000 feet to less than 1,500 feet on the west side of the valley (DWR, 2006). Underlying the Glen Ellen Formation is the Merced Formation, which is a marine deposit of fine sand and sandstone with thin interbeds of clay and silty-clay and some lenses of gravel and localized fossils. The Merced Formation is Pliocene in age and its thickness is estimated to range from 300 feet to greater than 1,500 feet. Aquifer continuity and water quality in the Merced Formation are generally very good, with well yields from 100 to 1,500 gpm (DWR, 2006).

As mentioned above, DWR has designated the Santa Rosa Plain subbasin as a medium priority basin and thus subject to the requirements of the Sustainable Groundwater Management Act (SGMA), including the requirement to be covered by one or more Groundwater Sustainability Agencies (GSAs) and to prepare and submit to DWR one or more Groundwater Sustainability Plans (GSPs) by 31 January 2022. Actions



related to management of the Santa Rosa Plain subbasin both currently and under SGMA are described in the next section.

#### 6.2.2 Non-SGMA Groundwater Management

The Santa Rosa Plain subbasin is currently managed under the Santa Rosa Plain Watershed Groundwater Management Plan (GMP), developed by the Santa Rosa Plain Basin Advisory Panel (Santa Rosa Plain Basin Advisory Panel, 2014). The stated goal of the GMP is "to proactively coordinate public and private groundwater management efforts and leverage funding opportunities to maintain a sustainable, locally-managed, high-quality groundwater resource for current and future users, while sustaining natural groundwater and surface water functions." The GMP outlines eighteen Basin Management Objectives and groups these into seven key management components, including: (1) stakeholder involvement and public awareness, (2) monitoring and modeling program, (3) groundwater protection, (4) increasing water conservation and efficiency, (5) increasing groundwater discharge, (6) increasing water reuse, and (7) integrated groundwater management. The GMP is the groundwater management program for this area, until the SGMA GSP for the Santa Rosa Plain subbasin is adopted.

#### 6.2.3 SGMA Groundwater Management

In 2014, the California State Legislature enacted the SGMA, with subsequent amendments in 2015. The SGMA requires the formation of GSAs and the development and implementation of GSPs for groundwater basins that are designated by DWR as medium or high priority. Because the Santa Rosa Plain subbasin is designated by DWR as a medium basin (DWR, 2019), the Santa Rosa Plain subbasin is subject to the requirements of SGMA, which include the formation of a one or more GSAs and the development and implementation of one or more GSPs.

The Santa Rosa Plain GSA was formed in June 2017 through a Joint Powers Agreement entered into by the SCWA, City of Cotati, City of Rohnert Park, City of Santa Rosa, City of Sebastopol, Town of Windsor, County of Sonoma, Gold Ridge Resource Conservation District, Sonoma Resource Conservation District, Branger Mutual Water Company, California American Water, Willowside Mutual Water Company, and Penngrove Water Company, and covers the entire subbasin. The Santa Rosa Plain GSA is governed by a nine-member Board of Directors, which includes a position held by SCWA. The Board of Directors is advised by an Advisory Committee that includes eighteen members appointed by the Board of Directors, representing various stakeholders. The GSP for the Santa Rosa Plain subbasin is currently under preparation and is anticipated to be complete and submitted to DWR by the statutory deadline of 31 January 2022. As of December 2020, initial drafts of sections describing the plan area and portions of the basin setting have been prepared and are available for public review on the Santa Rosa Plain GSA website: https://santarosaplaingroundwater.org/.

#### 6.2.4 <u>Coordination with Groundwater Supply Agencies</u>

Because the District does not directly pump groundwater, it does not coordinate with any GSAs. However, as noted above, the SCWA is a member of Santa Rosa Plain GSA and MMWD has coordinated with SCWA on its demand projections through 2045.



#### 6.2.5 <u>Historical Pumping and Supply Sufficiency</u>

As indicated in **Table 6-1**, the District does not pump any groundwater. SCWA's 2020 UWMP provides historical pumping and supply sufficiency information related to their use of groundwater and has factored this into the supply reliability information provided to the District and other Water Contractors.

#### 6.3 Surface Water

The District's primary water supply is local surface water. Until 1976, all of the District's water supply was obtained solely from rainfall collected from a watershed of approximately 28 square miles of District owned lands, and 36 square miles not owned by the District. Six reservoirs in the watershed had a storage capacity of 17.3 billion gallons (53,100 AF). Through a bond issue authorized during the drought of the 1970s, a seventh reservoir was completed in 1980, the Soulajule Reservoir, which added 3.4 billion gallons (10,400 AF) to the total storage. The District's Kent Lake facility was expanded in 1982 by raising Peters Dam 45 feet, increasing the storage capacity from 5.4 billion gallons (16,600 AF) to 10.6 billion gallons (32,500 AF). Presently, the total reservoir storage operated by the District is 25.9 billion gallons (79,566 AF).

A chronology of the District's water rights, and development of its reservoir system is provided in **Table 6-2**. A map of the reservoir system is shown in **Figure 3-2**.

Reservoir Name	Year Constructed	Storage Capacity (AF)	Water Rights
Lake Lagunitas	1873	350	Pre-1914
Phoenix Lake	1905	411	Pre-1914
Bon Tempe Reservoir	1948	4,017	Appropriative Permit No. 05633
	1918	3,069	Pre-1914
Alpine Lake	1924	4,600	
	1941	8,891	Appropriative Permit No. 05633
Kantlaka	1953	16,050	Appropriative Permit No. 05633,
Kent Lake	1982	32,895	09390, 18546
Nicasio Reservoir	1960	29,000(a)	Appropriative Permit No. 12800
Soulaiula Pasarvair	1090	10 572	Appropriative License 12807 and
	1980	10,572	Permit No. 16892
Total Existing Reservoir Storage		79,566	

 Table 6-2
 District Surface Water Reservoir System

NOTES: (a) Under the water right for storage, 6,570 AF of water from Nicasio Creek can be transferred from Nicasio Reservoir to Kent Lake to fill Kent Lake; this is in addition to any inflows from Lagunitas Creek into Kent Lake. This would free up capacity in Nicasio Reservoir for additional storage up to the total of 29,000 AF (22,430 AF stored in Nicasio Reservoir plus 6,750 transferred and stored in Kent Lake). However, new infrastructure would be required to transfer this supply since it cannot be conveyed via Lagunitas Creek.



On average, the District has an average annual runoff of 83,000 AF (10-year average). The range of annual runoff received into the reservoirs ranges widely from a low of 4,000 AF in 1977 to over 212,000 AF in 2017. Surface water from the Mt. Tamalpais watershed is aerated seasonally in the reservoirs to maintain adequate dissolved oxygen concentration. From the reservoirs, the water is conveyed to either the Bon Tempe Treatment Plant (BTTP) near Ross or the San Geronimo Treatment Plant (SGTP) in Woodacre. Suspended matter is removed in clarifiers, microscopic particles are removed in deep-bed, multi-media filters, and bacteria and pathogens are inactivated by disinfectants. The water is then treated for control corrosion. The District has been fluoridating its water since 1973 as required by a voter approved ballot measure in 1972.

#### 6.4 Stormwater

There are no plans to divert stormwater for beneficial uses in the District.

### 6.5 Wastewater and Recycled Water

#### ☑ CWC § 10633

The plan shall provide, to the extent available, information on recycled water and its potential for use as a water source in the service area of the urban water supplier. The preparation of the plan shall be coordinated with local water, wastewater, groundwater, and planning agencies that operate within the supplier's service area.

This section provides information on the amount of generated wastewater and existing disposal of wastewater to determine the potential for recycled water use by the District. The amount of recycled water currently used, potentially available, and future potential uses for recycled water for the District are also described.

#### 6.5.1 <u>Recycled Water Coordination</u>

The majority of recycled water used within the District's service area is distributed by the District. The Sewage Agency of Southern Marin (SASM) produces approximately 30 AFY of tertiary-treated recycled water that is used to irrigate playing fields situated adjacent to the SASM treatment plant. SASM treats and distributes this water. Recycled water production occurs at the Las Gallinas Valley Sanitary District (LGVSD) Recycled Water Treatment Facility (RWTF). The wastewater originates from within the LGVSD service area, which is also within the District's service area. The collected wastewater is treated to secondary levels at LGVSD's wastewater treatment plant and then receives further treatment at the RWTF before being distributed to customers.

In 2014 the LGVSD began supplying approximately 150 AFY of tertiary-treated recycled water produced at the newly constructed Las Gallinas Valley RWTF to NMWD. In 2019, LGVSD began construction on a major expansion and upgrade to the RWTF, which will expand the facility's capacity from 1.4 mgd to over 5 mgd. The RWTF expansion is now complete and the RWTF has been producing water since April 2021.

The District has a close working relationship with the LGVSD. Prior to each irrigation season (April through October), the District provides a written estimate of the quantity of recycled water that will be needed for the season. The LGVSD attempts to provide the District with enough recycled water to meet its



projected demand, and if the LGVSD is not able to meet the entire demand of the recycled water system, the system is supplemented by potable water.

#### 6.5.2 <u>Wastewater Collection, Treatment, and Disposal</u>

#### ☑ CWC § 10633 (a)

A description of the wastewater collection and treatment systems in the supplier's service area, including a quantification of the amount of wastewater collected and treated and the methods of wastewater disposal.

#### ☑ CWC § 10633 (b)

A description of the quantity of treated wastewater that meets recycled water standards, is being discharged, and is otherwise available for use in a recycled water project.

Within the District's service area, there are 16 wastewater collection entities. Of the 16 wastewater collection entities, 11 are wastewater collection entities and five have treatment facilities.

Three of the five treatment entities utilize secondary effluent for landscape irrigation at their wastewater treatment plant. The Richardson Bay Sanitary District irrigates an adjacent park with secondary effluent. This water does not meet current recycled water regulations, but the existing practice has been "grandfathered." SASM has a small tertiary treatment facility and irrigates an adjacent park; however, saltwater intrusion limits this operation to low tide cycles only. **Table 6-3** summarizes the 16 wastewater collection and treatment entities within the District's service area. Development within Marin County is limited due to space constraints; therefore, wastewater projections are estimated to remain steady over the planning horizon.

**Table 6-4** provides a summary of the volumes of treated effluent discharged and recycled within the District's service area in 2020.



#### There is no wastewater collection system. The supplier will not complete the table below. Percentage of 2020 service area covered by wastewater collection system (optional) Percentage of 2020 service area population covered by wastewater collection system (optional) Wastewater Collection **Recipient of Collected Wastewater** Volume of Is WWTP Wastewater Name of Wastewater Is WWTP Operation Wastewater Name of Wastewater Volume Treatment Agency **Treatment Plant** Located Collected from Contracted to **Receiving Collected** Collection Agency Metered or Name Within a Third Party? UWMP Service Estimated? Wastewater UWMP Area? (optional) Area 2020 Sewerage Agency of **Almonte Sanitary** Estimated 128 Southern Marin SASM Plant Yes No District (SASM) 93 SASM SASM Plant Alto Sanitary District Estimated Yes No Homestead Valley Estimated 177 SASM SASM Plant Yes No Sanitary District City of Mill Valley SASM SASM Plant Estimated 1,199 Yes No **Richardson Bay Sanitary** Estimated 795 SASM SASM Plant Yes No District SASM; Sausalito-Tamalpais Community SASM Plant: Estimated 290 Marin City Sanitary Yes No Services District SMCSD Plant District (SMCSD) Central Marin San Quentin State Metered 527 Sanitation Agency CMSA Plant Yes No Prison (CMSA) San Rafael Sanitation Metered 4,437 CMSA CMSA Plant Yes No

#### Table 6-3Wastewater Collected Within Area in 2020 (DWR Table 6-2)

District



	There is no wastewater collection system. The supplier will not complete the table below.						
	Percentage of 2020 service area covered by wastewater collection system (optional)						
	Percentage of 2020 service area population covered by wastewater collection system (optional)						
Wastev	Wastewater Collection Recipient o				Wastewater		
Name of Wastewater Collection Agency	Wastewater Volume Metered or Estimated?	Volume of Wastewater Collected from UWMP Service Area 2020	Name of Wastewater Treatment Agency Receiving Collected Wastewater	Treatment Plant Name	Is WWTP Located Within UWMP Area?	Is WWTP Operation Contracted to a Third Party? <i>(optional)</i>	
Sanitary District No. 1 (Ross Valley)	Metered	5,596	CMSA	CMSA Plant	Yes	No	
Sanitary District No. 2 (Corte Madera)	Metered	1,190	CMSA	CMSA Plant	Yes	No	
City of Sausalito	Estimated	510	SMCSD	SMCSD Plant	Yes	No	
National Park Service (Fort Baker)	Estimated	59	SMCSD	SMCSD Plant	Yes	No	
SMCSD	Metered	366	SMCSD	SMCSD Plant	Yes	No	
Sanitary District No. 5 (Tiburon)	Metered	666	Sanitary District No. 5	Tiburon Main Treatment Plant	Yes	No	
Sanitary District No. 5 (Paradise Cove)	Metered	18	Sanitary District No. 5	Paradise Cove Treatment Plant	Yes	No	
Las Gallinas Valley Sanitary District (LGVSD)	Metered	2,435	LGVSD	LGVSD Plant	Yes	No	
Total Wastewater	Total Wastewater Collected from     18,486       Service Area in 2020:     18,486						
(a) $(a)$	£ ^ F						

(a) Volumes are in units of AF.


	There is no was	There is no wastewater collection system. The supplier will not complete the table below.								
	Percentage of 2	Percentage of 2020 service area covered by wastewater collection system (optional)								
	Percentage of 2020 service area population covered by wastewater collection system (optional)									
Wastewater Collection			Re	cipient of Collected '	Wastewater					
Name of Wastewater Collection Agency	Wastewater Volume Metered or Estimated?	Volume of Wastewater Collected from UWMP Service Area 2020	Name of Wastewater Treatment Agency Receiving Collected Wastewater	Treatment Plant Name	Is WWTP Located Within UWMP Area?	Is WWTP Operation Contracted to a Third Party? (optional)				
<ul> <li>(b) Data reported on a fiscal year (FY) basis.</li> <li>(c) Volumes for SMCSD National Park Service (Fort Baker), City of Sausalito, and Tamalpais Community Services District were estimated based on proportions of total SMCSD 2015 volumes.</li> </ul>										



### Table 6-4Wastewater Treatment and Discharge Within Service Area in 2020 (DWR Table 6-3)

	No wastewater is treated or disposed of within the UWMP service area. The supplier will not complete the table below.											
					Does This Plant Treat Wastewater Generated Outside the Service Area?		2020 volumes					
Wastewater Treatment Plant Name	Discharge Location Name or Identifier	Discharge Location Description	Wastewater Discharge ID Number (optional)	Method of Disposal		Treatment Level	Wastewater Treated	Discharged Treated Wastewater	Recycled Within Service Area	Recycled Outside of Service Area	Instream Flow Permit Requirement	
Central Marin Sanitation Agency Plant	San Francisco Bay	Shallow estuary draining roughly 40% of California's water	2 215116001	Bay or estuary outfall	No	Secondary, Disinfected - 23	10,680	9,477	0	1,203	0	
Las Gallinas Valley Sanitary Agency Plant	Miller Creek	7.6 mile long stream draining into San Pablo Bay east of Marinwood	2 215012001	River or creek outfall	No	Secondary, Disinfected - 23	2,453	2,731	0	0	0	
Las Gallinas Valley Sanitary Agency Plant	Miller Creek	7.6 mile long stream draining into San Pablo Bay east of Marinwood	2 215012001	River or creek outfall	No	Tertiary	0	0	748 (c)	0	0	



	No wastewater is treated or disposed of within the UWMP service area. The supplier will not complete the table below.											
					Does This Plant Treat Wastewater Generated Outside the Service Area?		2020 volumes					
Wastewater Treatment Plant Name	Discharge Location Name or Identifier	Discharge Location Description	Wastewater Discharge ID Number <i>(optional)</i>	Method of Disposal		Treatment Level	Wastewater Treated	Discharged Treated Wastewater	Recycled Within Service Area	Recycled Outside of Service Area	Instream Flow Permit Requirement	
Paradise Cove Facility Sanitary District #5 (Tiburon)	San Francisco Bay	Shallow estuary draining roughly 40% of California's water	2 215021002	Bay or estuary outfall	No	Secondary, Disinfected - 23	18	17	0	0	0	
Main Plant Sanitary District #5 (Tiburon)	San Francisco Bay	Shallow estuary draining roughly 40% of California's water	2 215021001	Bay or estuary outfall	No	Secondary, Disinfected - 23	657	635	0	0	0	
Sausalito Marin City Sanitary District Plant	San Francisco Bay	Shallow estuary draining roughly 40% of California's water	2 215023001	Bay or estuary outfall	No	Secondary, Disinfected - 23	1,106	1,106	0	0	0	



	No wastewater is treated or disposed of within the UWMP service area. The supplier will not complete the table below.											
				Method of Disposal	Does This Plant Treat		2020 volumes					
Wastewater Treatment Plant Name	Discharge Location Name or Identifier	Discharge Location Description	Discharge ID Number (optional)		Wastewater Generated Outside the Service Area?	Treatment Level	Wastewater Treated	Discharged Treated Wastewater	Recycled Within Service Area	Recycled Outside of Service Area	Instream Flow Permit Requirement	
Sewerage Agency of Southern Marin Plant	Raccoon Straits	A part of the San Francisco Bay located between Angel Island and Tiburon Peninsula	2 215015001	River or creek outfall	No	Secondary, Disinfected - 23	2,280	2,403	0	0	0	
Sewerage Agency of Southern Marin Plant	Raccoon Straits	A part of the San Francisco Bay located between Angela Island and Tiburon Peninsula	2 215015001	River or creek outfall	No	Tertiary	0	0	0	0	18	
						Total	17,193	16,368	748	1,203	18	
NOTES:		-										

(a) Volumes are in units of AF.

(b) Data reported on a calendar year basis.

(c) Volume for LGVSD shown is reported on a fiscal year (FY) basis and may include an unknown amount of recycled water served outside of District service area.



### 6.5.3 <u>Recycled Water System and Potential, Current, and Projected Uses of Recycled Water</u>

#### ☑ CWC § 10633 (c-g)

(c) A description of the recycled water currently being used in the supplier's service area, including, but not limited to, the type, place, and quantity of use.

(d) A description and quantification of the potential uses of recycled water, including, but not limited to, agricultural irrigation, landscape irrigation, wildlife habitat enhancement, wetlands, industrial reuse, groundwater recharge, indirect potable reuse, and other appropriate uses, and a determination with regard to the technical and economic feasibility of serving those uses.

(e) The projected use of recycled water within the supplier's service area at the end of 5, 10, 15, and 20 years and a description of the actual use of recycled water in comparison to uses previously projected pursuant to this subdivision.

(f) A description of actions, including financial incentives, which may be taken to encourage the use of recycled water, and the projected results of these actions in terms of acre-feet of recycled water used per year.

(g) A plan for optimizing the use of recycled water in the supplier's service area, including actions to facilitate the installation of dual distribution systems, to promote recirculating uses, to facilitate the increased use of treated wastewater that meets recycled water standards, and to overcome any obstacles to achieving that increased use.

The District started water recycling during the drought of 1976-77 when a pilot plant was commissioned for drought relief. The pilot plant demonstrated that recycled water was available when other sources were not. Increased concern about limited potable supplies led to a joint effort with the LGVSD to build a permanent facility. In 1981, a 1.0 mgd direct filtration plant was completed and served 60 AFY to nearby McInnis Park and to highway landscaping. Plans for further expansion were set back when the state water reclamation regulations (Title 22) were made more stringent. The water quality that the plant was capable of producing was no longer adequate for irrigating parks, playgrounds, and greenbelts.

In 1989, the District upgraded the recycled water facility from direct filtration to full conventional treatment and increased capacity to 2.0 mgd. With improved water quality, the District was able to expand its recycled water market. From 1990 through 1994, the distribution system was enlarged using District funds plus a \$5 million low-interest loan from the State Revolving Fund. Today, the District's recycled water system serves about 748 AF of recycled water per year through 333 service connections.

Over the past 25 years, the District has consistently strived to expand the use of recycled water, number of sites served, and the total amount of recycled water put to use. The system is located in the northern part of the District's service area from the Marin County Civic Center through Marinwood. All customers on the system are served recycled water from the District's Las Gallinas Valley RWTF, operated in conjunction with the LGVSD.

Marin County experiences a relatively wet rainy season which decreases the demand season for landscape irrigation to about seven months of the year (April through October). Historically, the RWTF has not operated during the winter months when water demand is low, however the newly expanded RWTF is expected to produce water all year even when demand is low.



Although irrigation remains the primary use (95%) for recycled water in the District's service area, there are virtually no large-scale irrigation accounts and no major industrial water users in the area. Therefore, the District has turned to alternative uses of recycled water to optimize the efficiency of the system, including the following.

- Flushing Toilets with Recycled Water As a result of efforts launched in 1991, there are now 38 buildings in the District's service area that use recycled water to flush toilets and urinals. This includes the 330-bed Marin County Jail, which was the first indoor use of recycled water in a penal institution. All new buildings in the recycled water service area are now required to be constructed with dual plumbing to use recycled water indoors as well as for landscape irrigation. In September 2011, the District began providing recycled water to 33 San Pedro, the first residential condominium complex in California to be dual plumbed to use recycled water for toilet flushing.
- Car Washes with Recycled Water In 1995, the District was the first in California to use recycled water in a car wash. Building on that success, two new car washes were constructed to use recycled water. All new car washes in the recycled water service area are now required to use recycled water.
- Heating, ventilation, and air conditioning (HVAC) Cooling Towers with Recycled Water The District successfully pioneered the first use of recycled water in a HVAC cooling tower in 1995. Since then, two more buildings have had the HVAC system converted to recycled water.
- **Commercial Laundries with Recycled Water** In 1998, the District was the first water District in California to convert a commercial laundry to use tertiary-treated recycled water.

**Table 6-5** below provides the anticipated future recycled water within the District. As discussed in further detail in **Appendix C**, total recycled water use is projected to be 750 AFY by 2045.<sup>13</sup> Due to infrastructure upgrades at the Las Gallinas Valley RWTF, all demands by the recycled water system were met by potable water in 2020. The volume of potable water makeup used to supplement the recycled water system from 2016 through 2020 is provided in Section 4.1. Potable water is not anticipated to be needed to supplement the recycled water system going forward, following plant upgrades.

<sup>&</sup>lt;sup>13</sup> Recycled water demands were rounded from the totals shown in Appendix C.



### Table 6-5 Current and Projected Recycled Water Direct Beneficial Uses Within Service Area (DWR Table 6-4)

Recycled water is not used and is not planned for use within the service area of the supplier. The supplier will not complete the table below.											
Name of Supplier Producing	(Treating) the Recycled Water:	Marin Municipal Water District									
Name of Supplier Operat	ing the Recycled Water Distribution System:	Marin Municipal Wat	Marin Municipal Water District								
Supplemental Water A	dded in 2020 (volume)	748 AF									
Source of 202	20 Supplemental Water	Potable Water									
Beneficial Use Type	<b>Potential</b> Beneficial Uses of Recycled Water (Describe)	Amount of <b>Potential</b> Uses of Recycled Water (Quantity)	General Description of 2020 Uses	Level of Treatment	2020	2025	2030	2035	2040	2045	
Landscape irrigation (excludes golf courses)				Tertiary	441	443	443	443	443	443	
Golf course irrigation				Tertiary	88	88	88	88	88	88	
Commercial use				Tertiary	162	163	163	163	163	163	
Industrial use				Tertiary	33	33	33	33	33	33	
Other (Provide General Description)			Toilet/urinal flushing in commercial and residential settings	Tertiary	23	23	23	23	23	23	
	Total:						750	750	750	750	
			2020 Int	ernal Reuse							
NOTES: (a) Volumes are in units of AF	=.										



### 6.5.4 Comparison of Previously Projected Use and Actual Use

#### ☑ CWC § 10633 (e)

A description of the actual use of recycled water in comparison to uses previously projected pursuant to this subdivision.

The District's 2015 UWMP projected recycled water demand and production in 2020 to be 520 AF. Actual water use by the recycled water system in 2020 was 748 AF,<sup>14</sup> which is 228 AF more than the total demand projected in the 2015 UWMP. **Table 6-6** provides a comparison of the 2015 UWMP projection for 2020 demand to actual 2020 use by the system.

### Table 6-62015 UWMP Recycled Water Use Projection Compared to 2020 Actual (DWR Table 6-5)

	Recycled water was supplier will not com	Recycled water was not used in 2015 nor projected for use in 2020. The supplier will not complete the table below.								
Benefic	ial Use Type	2015 Projection for 2020	2020 Actual Use							
Landscape irrigation (excludes golf courses)		307	441							
Golf course irriga	ation	61	88							
Commercial use		113	162							
Industrial use		23	33							
Other		16	23							
	Total	520	748							

NOTES:

(a) Volumes are in units of AF.

(b) Given that Las Gallinas RWTF was shut down during 2020 due to infrastructure upgrades, all demands by the recycled water system in 2020 was met by potable water.

<sup>&</sup>lt;sup>14</sup> During 2020, the Las Gallinas RWTF was offline due to infrastructure upgrades and all demands by the recycled water system were met by potable water. Potable water is not anticipated to be needed to supplement the recycled water system going forward, following plant upgrades.



### 6.5.5 Actions to Encourage and Optimize Future Recycled Water Use

### ☑ CWC § 10633 (e-g)

(e) The projected use of recycled water within the supplier's service area at the end of 5, 10, 15, and 20 years and a description of the actual use of recycled water in comparison to uses previously projected pursuant to this subdivision.

(f) A description of actions, including financial incentives, which may be taken to encourage the use of recycled water, and the projected results of these actions in terms of acre-feet of recycled water used per year.

(g) A plan for optimizing the use of recycled water in the supplier's service area, including actions to facilitate the installation of dual distribution systems, to promote recirculating uses, to facilitate the increased use of treated wastewater that meets recycled water standards, and to overcome any obstacles to achieving that increased use.

The District uses a variety of incentives to encourage recycled water use. One of the more positive marketing aspects of recycled water is reliability. The droughts of 1976-77 and 1987-92 necessitated severe rationing with resultant damage to customers' landscape plantings. With recycled water used for landscaping during droughts, customers' investments in landscaping are protected.

Several District policies also encourage the use of recycled water. The rates for recycled water were originally established at half of the Tier 1 potable rate and, based on the rate structure changes that were put into effect during May 2016, are at 69% of the Tier 1 potable rate. The District also requires use of recycled water, where it is available, as a condition of potable water service. For existing potable water customers, the conversion to recycled water is provided without a charge or fee from the District. Board Policy No. 2, included in **Appendix G**, contains the District policy on recycled water.

**Table 6-7** provides a summary of estimated recycled water use that are expected to be realized by implementing these methods to encourage recycled water use.



	Supplier does not plan to expand recycled water use in the future. Supplier will not complete the table below but will provide narrative explanation.								
70 (Section 6.5.5)	Provide page location of narrative in U	Provide page location of narrative in UWMP							
Name of Action	Description	Planned Implementation Year	Expected Increase in Recycled Water Use						
Expand recycled water use	Explore opportunities to expand use of recycled water, pending grant funding opportunities	Unknown	Unknown						
		Total							
NOTES: (a) Volumes are ir	n units of AF.								

The District has thoroughly explored ways to expand water recycling. However, with few large users of non-potable water (such as golf courses and heavy industry) within the District's service area, the District's remaining water recycling options are more expensive and less feasible than continuing to use potable infrastructure. As more customers improve irrigation efficiency, reduce turf areas, switch to native and drought-tolerant landscapes, and convert parks and athletic fields to artificial turf, there are fewer opportunities and lower demand for recycled water. For example, the total water entitlement of all recycled water customers connected to the recycled water system is 956 AF.

Another obstacle to increased water recycling in the District's service area is severe saltwater intrusion into the sewer collection systems of most of the local sanitation agencies, including that of the community's largest wastewater agency, CMSA. Rehabilitation of the sewer collection system was explored, but intrusion is so widespread that the cost to repair the sewers is much more than the cost of the water recycling project. Reducing salt in wastewater using membrane processes has also been explored, but the high cost and increased concentration of contaminants remaining in the wastewater discharge render that alternative infeasible from both a cost and regulatory perspective.

Satellite recycling plants, which can intercept sewage above the zones of saltwater intrusion and process the sewage to tertiary recycled water, appeared to be a promising avenue for enlarging the District's recycled water program. In 2001, the District conducted a study, partially funded through a grant from the DWR, to investigate the viability and cost of incorporating satellite water recycling plants into the District's distribution system. However, while the study concluded that this approach was technically feasible, it would cost over \$3,000 per AF, making it prohibitively expensive. Recent feasibility studies



have also considered satellite treatment plants and the estimated costs are far greater than \$3,000 per AF.

The District continues to investigate the potential for expansion of recycled water, which, along with conservation, is a high priority in the District's vision of sustainable water resource management. As part of its 2040 Water Resources Plan, the District evaluated recycled water and potable reuse opportunities for its various treatment plants. At this time, the District is exploring opportunities for grant funding to offset the cost of recycled water projects.

### 6.6 Desalinated Water

*CWC* § 10631 (g) A plan shall be adopted in accordance with this chapter and shall do all of the following:

Describe the opportunities for development of desalinated water, including, but not limited to, ocean water, brackish water, and groundwater, as a long-term supply.

During August of 2010, the District adopted Ordinance 420, which states that the District shall not approve construction, or financing for construction, of a desalination facility unless such construction is approved by a majority of District voters voting in an election held within the District's service area for that purpose. While the District has, in the past, explored desalination as a potential supply option, the District does not intend to pursue desalination to augment water supplies at this time.

### 6.7 Water Exchanges and Transfers

**CWC § 10631 (c)** A plan shall be adopted in accordance with this chapter and shall do all of the following:

Describe the opportunities for exchanges or transfers of water on a short-term or long-term basis.

The District's service area is entirely within Marin County. Marin County is a peninsula surrounded on three sides by water. The physical barriers imposed by these water bodies severely limit the water transfer opportunities available to the District.

However, during the drought of the 1970s, the District made use of water transfers to augment its supplies. Emergency pipeline connections were made to SCWA, north of the District, and to the East Bay Municipal Utility District (EBMUD) water system to the east. Water was delivered to the District from the State Water Project through the EBMUD system and from the Russian River via the SCWA and NMWD water systems.

The temporary connection with EBMUD was installed in the emergency pull-out lane of the Richmond-San Rafael Bridge. It was removed from the bridge in the early 1980s when traffic increased making the pipeline a safety hazard. However, the connection to the NMWD and SCWA was improved into a permanent connection and contracts allowing delivery of water during non-emergency periods have been implemented as described in Section 6.1.

MMWD assessed, as one of many resiliency options, the feasibility of transfers with EBMUD and other local water suppliers as part of its 2040 Water Resources Plan, and as part of its involvement in the Bay Area Regional Reliability Drought Contingency Plan. The 2040 Water Resources Plan recommended that



the District explore groundwater partnering opportunities with a partner that uses groundwater supplies to implement an in-lieu groundwater recharge program. The conjunctive groundwater use is part of the SCWA's resiliency planning efforts and the District hopes to participate in a future project.

### 6.8 Future Water Projects

**CWC § 10631** A plan shall be adopted in accordance with this chapter and shall do all of the following:

(b) (3) For any planned sources of water supply, a description of the measures that are being undertaken to acquire and develop those water supplies.

(f) Include a description of all water supply projects and water supply programs that may be undertaken by the urban water supplier to meet the total projected water use, as established pursuant to subdivision (a) of Section 10635. The urban water supplier shall include a detailed description of expected future projects and programs that the urban water supplier may implement to increase the amount of the water supply available to the urban water supplier in normal and single-dry water years and for a period of drought lasting five consecutive water years. The description shall identify specific projects and include a description of the increase in water supply that is expected to be available from each project. The description shall include an estimate with regard to the implementation timeline for each project or program.

The District's 2040 Water Resources Plan evaluated various potential projects to increase supply reliability in the future. Future projects that may contribute to the District's water supply are summarized in **Table 6-8**.



	No ex agenc	No expected future water supply projects or programs that provide a quantifiable increase to the agency's water supply. Supplier will not complete the table below.								
	Some table	or all of the supplier's fu and are described in a n	uture water suppl arrative format.	y projects or progra	ims are not compa	tible with this				
Page 73 (Section 6.8)	Provid	de page location of narra	ative in the UWM	þ						
Name of Future Projects	Joii	nt Project with other suppliers?	Description	Planned Implementation	Planned for Use in Year	Expected Increase in Water Supply to Supplier				
or Programs	Y/N	If Yes, Supplier Name	(if needed)	Year	Туре					
Expand recycled water use	Ν		Pipeline to Peacock Gap Golf Course	2022	All types	166 AF				
Rehabilitation of Kastania Pump Station	Ν		Pump station renewal	2022	All	0				
Change to environmental releases	Y	Temporary Urgency Change Petition (TUCP) to State Water Board	TUCP	2021	Critically Dry	Unknown				
Rental of generator for Soulajule Reservoir	Ν		Generator to power pump station	2021	Dry	0				
Phoenix Lake	Ν		Pump water from lake to treatment plant	2021	Dry	0				
NOTES: (a) Volumes are in units of A	۶.									

### Table 6-8Expected Future Water Supply Projects or Programs (DWR Table 6-7)



### 6.9 Summary of Existing and Planned Sources of Water

**CWC** § 10631 (b) Identify and quantify, to the extent practicable, the existing and planned sources of water available to the supplier over the same five-year increments described in subdivision (a).

CWC § 10631 (b) (4) (D) A detailed description and analysis of the amount and location of groundwater that is projected to be pumped by the urban water supplier. The description and analysis shall be based on information that is reasonably available, including, but not limited to, historic use records.

**Table 6-9** summarizes the actual source and water supply volume for 2020. **Table 6-10** summarizes the projected source and water supply volume in five-year increments over the next 25 years.

Projected water supplies from SCWA reflect the anticipated purchases from SCWA. The District's contract with SCWA allows for purchases of up to 14,300 AFY. Projected surface water supplies are based on modelling performed by Woodard and Curran. Recycled and raw water projections were developed as part of the 2020 Urban Water Management Plan Water Demand Analysis and Water Conservation Measures Update and are discussed in more detail in **Appendix C**.

Water Supply	Additional Detail on Water Supply	Actual Volume	Water Quality	Total Right or Safe Yield <i>(optional)</i>					
Purchased or Imported Water	Sonoma County Water Agency	6,822	Drinking Water						
Surface water (not desalinated)		20,449	Drinking Water						
Surface water (not desalinated)	Environmental Releases from Kent and Soulajule Lakes	12,699	Other Non- Potable Water						
Other	Water sold to the Meadow Club	180	Other Non- Potable Water						
	Total	40,149							
NOTES:									
Other NOTES: (a) Volumes are in units o	Meadow Club <b>Total</b> f AF.	180 40,149	Potable Water						

Table 6-9	Water Supplies - Actual (DWR Table 6-8)
-----------	---



### Table 6-10 Water Supplies – Projected (DWR Table 6-9)

	Additional Detail on Water Supply	Projected Water Supply									
		2025		2030		2035		2040		2045	
Water Supply		Reasonably Available Volume	Total Right or Safe Yield (optional)								
Purchased or Imported Water	Sonoma County Water Agency	5,300		5,300		5,300		5,300		5,300	
Surface water (not desalinated)		78,540		78,793		78,525		78,558		78,626	
Recycled Water		750		750		750		750		750	
Other	Raw Water	171		174		176		176		176	
	Total	84,761		85,017		84,751		84,784		84,852	

NOTES:

(a) Volumes are in units of AF.

(b) Reasonably available volume for SCWA reflects the anticipated purchases from SCWA. MMWD's contract with SCWA allows for purchases of up to 14,300 AFY.

(c) Surface water volumes are estimated based on modeling performed by Woodard and Curran.



### 6.10 Special Conditions

### 6.10.1 Climate Change Effects

As discussed in SCWA's Draft 2020 Urban Water Management Plan (SCWA, 2021), SCWA has been evaluating the effects of climate change. As stated by SCWA:

Sonoma Water has investigated whether existing downscaled climate models can be used or modified to provide reliable estimates of the effects of increased concentrations of carbon dioxide and other greenhouse gases on temperatures and precipitation patterns within Sonoma Water's service area and within the watersheds from which Sonoma Water obtains its water supply during the 25-year planning horizon. As of this time, no detailed analysis exists of potential climate change impacts that takes into consideration the influence of marine layers, whose effects on the region are difficult to model. Given the uncertainties between various downscaled models, Sonoma Water evaluates ensembles of downscaled models for general water supply planning purposes. However, there is not one model that can be selected with any confidence to be analyzed for the required format of this Plan. For these reasons, this Plan assumes that the climatic patterns and associated hydrology experienced over the past 108 years of record (1910 – 2017) provide a reasonable basis for the 25-year planning horizon that would impact the water supply and water demand analysis set forth in the Plan.

As discussed in Section 5.9, however, the United States Geological Survey (USGS) conducted a study for Sonoma Water on the potential effects of climate change on Sonoma Water's water supply, which has provided additional information on the potential impacts of climate change on Sonoma Water's service area. Furthermore, Sonoma Water has embarked on development of a Climate Adaptation Plan which studies the potential impacts of climate change in regards to both water supply reliability and Sonoma Water's transmission system facilities. This planning process analyzes the results of multiple climate models to determine a range of potential climate related impacts. A risk based analysis of the potential impacts to the watershed and Sonoma Water facilities will be used to identify courses of action that can be pursued to mitigate the effects of climate change. The work plan was developed in 2015 and a robust planning process began in 2016. Sonoma Water expects to bring the Climate Adaptation Plan to its Board for approval in summer 2021.

Additionally, SGMA requires that GSPs include basin-wide water budget models under various climate change scenarios, including future conditions that account for the effects of estimated climate change. The water budget sections have not yet been developed by the Sonoma Valley GSA but are expected to include consideration of the effects of climate change for groundwater sustainability planning purposes.

Coping with inter-annual variability has always been a challenge for long-term water supply planning in the Bay Area, and climate change may intensify variability in coming decades. With potential additional changes imposed by climate change, there will be a heightened need to evaluate and respond to increased water supply variability.



The District's water supply comes from local runoff and the Russian River, which is dependent on local rainfall. Precipitation is stored in local reservoirs and released during the drier summer months. The District is currently storage-limited; existing storage capacity represents only about two years of demand. There are no remaining economically feasible sites for new surface water storage facilities, and the underlying groundwater basins are not considered feasible as a municipal supply source.

Historically, MMWD has been able to meet demands during prior periods of extreme drought with rationing, conservation, and increased purchases from SCWA. Given that the District is storage-limited, it can and has experienced changes in storage very quickly. For example, from December 2012 to January 2014, MMWD experienced a period of very low precipitation, and its reservoirs reached significantly low storage conditions that nearly triggered significant mandatory reductions. Water supply circumstances then changed in early February 2014 when the district received 15 inches of rain, more than the total rain during the prior 400 days combined.

Over the last 30 years, mean sea level around the San Francisco Bay area has been observed to change between 0 and 3 millimeters per year (NOAA, 2021). Sea level rise associated with climate change also has the potential to negatively impact the District's service area. There are a number of developed low-lying areas within the District's service area that could become inundated with sea level rise. Other areas may see an increase in the regularity of flood events. This could result in reduced overall water use in the service area, or shifts in the location of uses, as customers are either forced or opt to move elsewhere.

As climate change advances, there is also a potential for increased ecological vulnerability to currently identified invasive species as well as to new invasive species. According to Cal WeedMapper, Marin County has over 150 invasive flora species, including water hyacinth (*elchhornia crassipes*), European beachgrass (*ammophilia arenaria*), and red brome (*bromus madritensus* ssp. *rubens*). There are also 19 invasive flora species that, while not yet in Marin County, have been identified within 50 miles of the County. These include spotted knapweed (*centaurea maculosa*), alligator weed (*alternanthera philoxeroides*), and the South American spongeplant (*limnobium laevigatum*). There are also 11 invasive fauna species within Marin County, including the Asian estuarine mudsnail (*batillaria attramentaria*), the amethyst gem clam (*gemma gemma*), and the American oyster drill (*urosalpinx cinerea*). Invasive species could impact the District's water supply by negatively affecting water quality and infrastructure systems.

Climate change is expected to affect the District's supply as follows:

- Total precipitation is not projected to change significantly, although there may be less precipitation in the spring.
- Timing of runoff is expected to shift to earlier in the year, affecting reservoir storage, especially in the spring and summer months.
- Variability in annual precipitation is expected to continue, with vulnerability to droughts and dry periods.
- More intense storms anticipated that may affect surface water runoff and storage and stored water quality.
- Sea level rise could inundate some of the developed low-lying areas and increase flooding regularity in other areas.

The District's Water Resources Plan considered the impact of climate change on the District's supply. As part of the Water Resources Plan, climate change scenarios were developed and modeled using Marin



WaterSim, a dynamic systems model built within the commercial software GoldSim, to evaluate the District's water supply resiliency. The modeling used precipitation and inflow data for 2010 through 2099 for each of the District's lakes under four climate change models. Based on the modeling performed in 2017, there were no predicted shortages through 2040 under the modeled climate change scenarios, although overall reservoir levels were projected to significantly decrease under the modeled climate change conditions. These findings were incorporated into the recommendations in the Water Resources Plan.

### 6.10.2 <u>Regulatory Conditions and Project Development</u>

Emerging regulatory conditions may affect planned future projects and the characterization of future water supply availability and analysis. As the District moves forward with plans to develop supply projects, emerging regulatory conditions will be considered, and the associated water supply reliability impacts will be assessed in future UWMP updates.

### 6.10.3 Other Locally Applicable Criteria

Other locally applicable criteria may affect characterization and availability of an identified water supply (e.g., changes in regional water transfer rules may alter the availability of a water supply that had historically been readily available). The District does not have any current plans to develop additional supply sources. If the District does move forward ahead with any plans to develop supply projects, locally applicable criteria will be considered, and the associated water supply reliability impacts will be assessed in future UWMP updates.



### 6.11 Energy Intensity

### ☑ CWC § 10631.2

(a) In addition to the requirements of Section 10631, an urban water management plan shall include any of the following information that the urban water supplier can readily obtain:

(1) An estimate of the amount of energy used to extract or divert water supplies.

(2) An estimate of the amount of energy used to convey water supplies to the water treatment plants or distribution systems.

(3) An estimate of the amount of energy used to treat water supplies.

(4) An estimate of the amount of energy used to distribute water supplies through its distribution systems.

(5) An estimate of the amount of energy used for treated water supplies in comparison to the amount used for nontreated water supplies.

(6) An estimate of the amount of energy used to place water into or withdraw from storage.

(7) Any other energy-related information the urban water supplier deems appropriate.

(b) The department shall include in its guidance for the preparation of urban water management plans a methodology for the voluntary calculation or estimation of the energy intensity of urban water systems. The department may consider studies and calculations conducted by the Public Utilities Commission in developing the methodology.

(c) The Legislature finds and declares that energy use is only one factor in water supply planning and shall not be considered independently of other factors.

Within the service area, the District uses energy to distribute water supplies through its distribution systems. The energy used by the distribution systems is metered and documented in monthly PG&E bills. From January 2019 to December 2019, the District used 12,726,330 kilowatt hours (kWh) of energy to operate the water supply system and delivered 25,688 AFY of potable and non-potable water (excluding environmental releases) to customers in the service area, for a total energy intensity of 495 kWh/AF (**Table 6-11**). SCWA uses energy to treat and distribute water before delivery to the District. However, the energy is used outside of the District's service area, and the energy consumption information is not typically shared with the District.



### Table 6-11 Recommended Energy Intensity – Total Utility Approach (DWR Table O-1B)

Urban Water Supplier:

Marin Municipal Water District

Water Delivery Product

Multiple Products (unable to use table O-1C)

Enter Start Date for Reporting Period	1/1/2019	Urban Matar Su	Urban Water Supplier Operational Contro			
End Date	12/31/2019		pplier Operatio	mer Operational Control		
Is upstream embedded in the values reported?	No	Sum of All Water Management Processes	Non-Cons Hydroj	nsequential opower		
Water Volume Units Used	AF	Total Utility	Hydropower	Net Utility		
Volume of Water Entering	25,688	0	25,688			
En	ergy Consumed (kWh)	12,726,330	0	12,726,330		
Energy In	tensity (kWh/volume)	495.4	0.0	495.4		
Quantity of Self-Generated Renewable Energy 0 kWh Data Quality						

Metered Data

Data Quality Narrative:

Utility bills for the associated time period are used as the source for energy consumption data.

Narrative:

Total energy consumption represents the energy consumed during pumping, treatment, conveyance, and distribution.



### 7. WATER SUPPLY RELIABILITY

### ☑ CWC § 10620 (f)

An urban water supplier shall describe in the plan water management tools and options used by that entity that will maximize resources and minimize the need to import water from other regions.

### ☑ CWC § 10630.5

Each plan shall include a simple lay description of how much water the agency has on a reliable basis, how much it needs for the foreseeable future, what the agency's strategy is for meeting its water needs, the challenges facing the agency, and any other information necessary to provide a general understanding of the agency's plan.

This chapter describes the reliability of the Marin Municipal Water District's (MMWD's or District's) water supplies. Assessment of water supply reliability is complex and dependent upon a number of factors, such as the number of water sources, regulatory and legal constraints, hydrological and environmental conditions, climate change, and expected growth, among others. Based on available historical information and projections of future water uses, regulatory and legal constraints, and hydrological and environmental conditions, including climate change, the District has made its best determination of future water supply reliability of for the District, as described below.

### 7.1 Constraints on Water Sources

Purchased water from Sonoma County Water Agency (SCWA or Sonoma Water) and local surface water are the primary supply sources for the District. Several factors pose potential constraints on the District's water supply, including limits on the amount available, water quality, climatic conditions, or a combination of these. **Table 7-1** lists the District's sources of water supply and the potential factors that could impact the District's supply. These constraints, along with associated management strategies, are summary in the following sections.

### Water Supply Reliability 2020 Urban Water Management Plan Marin Municipal Water District



Factors	Sonoma County Water Agency (Imported Water)	District Produced Surface Water	Recycled Water
Limitation Quantification	Quantity limited by contractual limits and available pipeline capacity. When pipeline capacity impacts deliveries, the district/SCWA will have to construct new conveyance facilities to supplement capacity.	Climatic variation could result in limited storage carryover.	None
Legal	Supplies could potentially be reduced subject to unanticipated regulatory requirements.	None	None
Environmental	Fish habitat protection could result in summertime diversion curtailments on the Russian River. Future supply increases may not be consistent due to delays in construction, approval of water rights applications, or environmental documentation.	Future increases in instream flow requirements could decrease supply.	None
Water Quality	None	None	Salt-water intrusion in low-lying areas of sewer collection system could impact recycled water quality.
Climatic	Drought could reduce available surface water supply.	Drought could reduce available surface water supply.	None

### Table 7-1 Factors Resulting in Inconsistency of Supply

### 7.1.1 Supply Availability

### Purchased Water

The water available to SCWA's customers is constrained by both physical and legal constraints. The capacity of SCWA's transmission system is a physical constraint that can limit the District's water supply from SCWA. The District receives the SCWA supply through the District's 8-mile-long aqueduct, which is a 30, 36, and 42-inch diameter steel transmission main that runs from the Agency's Petaluma Aqueduct near Kastania Tank in south Petaluma to a connection located at the northern end of the District's pipeline facilities in Novato.

Legal constraints include the Agreement, SCWA Water Rights, and the Russian River Biological Opinion. These legal constraints are described below.

## Water Supply Reliability 2020 Urban Water Management Plan Marin Municipal Water District



- The Agreement includes specific maximum amounts of water that SCWA is obligated to supply to its Water Contractors, including the District. The Agreement states that SCWA is not obligated to provide the District with more than 14,300 acre-feet per year (AFY). The Agreement also places seasonal limitations on water delivery rates, with deliveries limited to 23.1 million gallons per day (mgd) from December to March, 12.8 mgd from May to September, 20.1 mgd in April and November, and 17.1 mgd in October.
- Four State Water Resources Control Board (SWRCB) permits (SWRCB Permit Numbers 12947A, 12949, 12950, and 1596) currently authorize SCWA to store water in Lake Mendocino (122,500 AFY) on the East Fork Russian River and Lake Sonoma (245,000 AFY) on Dry Creek, and to divert and redivert 180 cubic feet per second (cfs) of water from the Russian River, up to 75,000 AFY. SCWA estimates the existing annual diversion and rediversion limit of 75,000 AFY will be exceeded by 2035 (NMWD, 2016). Consequently, SCWA will need to file an application to SWRCB by around 2030 to increase its annual diversion and rediversion limit (NMWD, 2016). The permits also establish minimum instream flow requirements for fish and wildlife protection as well as for recreational considerations. These minimum instream flow requirements vary according to the hydrologic cycle as defined by SWRCB Decision 1610. SCWA meets the Decision 1610 flow requirements by making releases from Coyote Valley Dam at Lake Mendocino and Warm Springs Dam at Lake Sonoma (NMWD, 2016).
- On 24 September 2008, the National Marine Fisheries Service (NMFS) issued a 15-year biological opinion for water supply, flood control operations, and channel maintenance conducted by the U.S Army Corps of Engineers (USACE), SCWA, and Mendocino County Russian River Flood Control and Water Conservation Improvement District in the Russian River watershed. The Russian River Biological Opinion (Biological Opinion) concluded that the elevated river flows required by Decision 1610 were adversely affecting fish habitat and listed alternatives to reduce the effects. The alternatives included:
  - o Reducing summertime flows in the Russian River and Dry Creek;
  - o Enhancing six miles of habitat in Dry Creek;
  - Creating a freshwater lagoon in the estuary during summer months;
  - o Monitoring both habitat and fish in the Dry Creek, the estuary, and the Russian River; and
  - o Eliminating impediments to fish spawning or improving habitat in several streams.
- The Biological Opinion requires that summertime flows be permanently reduced to replicate river conditions in dry years. Since the biological opinion was released, SCWA has submitted a petition to the SWRCB requesting permanent changes to Decision 1610 minimum flow requirements in line with the Biological Opinion and is preparing an Environmental Impact Report (EIR) required by the California Environmental Quality Act (CEQA). Since 2010, SCWA has requested temporary changes to the Decision 1610 minimum flows annually based on the Biological Opinion recommendations.

In addition to these projects, SCWA is currently evaluating the feasibility of groundwater banking as a method of increasing water supply reliability. A Groundwater Banking Feasibility Study was completed in 2012. SCWA is currently working with the City of Sonoma to implement a pilot study using one of the City of Sonoma's municipal supply wells (SCWA, 2016).



In its 2020 UWMP, SCWA describes its reliability as follows:

"The overall conclusion is that Sonoma Water has adequate water supply through the 2045 planning horizon of this Plan, except for single-dry years, starting after 2025. For single-dry years, the model simulations predict that storage levels in Lake Sonoma will drop below 100,000 ac-ft prior to July 15th, thus requiring demand curtailments by Sonoma Water customers per Decision 1610 (Section 5.1.6.1 [of SCWA's 2020 UWMP]) for some portion of the year. In these circumstances, Sonoma Water will work with its Customers to reduce water demands as described in the Water Shortage Contingency Plan described in Section 7 [of SCWA's 2020 UWMP], or to utilize additional local sources, or both. Based on efforts over the last five years during dry conditions, Sonoma Water does not anticipate any difficulty in maintaining an adequate water supply during the single-dry year. The magnitude of these single-dry year potential shortfalls is estimated to be about 19% of average annual demand by 2045."

On the basis of this, and the associated tables provided by SCWA, projected availability of SCWA supplies to the District over the planning horizon are presented in Section 7.2.

Similar to the imported supply, the District's local surface water supply could also be impacted by future supply conditions and/or climate change. The reliability of the local surface water, as well as recycled water quality concerns are described later in this chapter (Sections 7.1.2 and 7.2.3).

### 7.1.2 <u>Water Quality Impacts on Reliability</u>

#### **☑** CWC § 10634

The plan shall include information, to the extent practicable, relating to the quality of existing sources of water available to the supplier over the same five-year increments as described in subdivision (a) of Section 10631, and the manner in which water quality affects water management strategies and supply reliability.

### Potable Water

The District is fortunate to have water of exceptionally high quality and has never exceeded a water quality regulatory limit or received a regulatory violation. All drinking water standards are set by the U.S. Environmental Protection Agency (USEPA) under the authorization of the Federal Safe Drinking Water Act of 1974. In California, the State Water Resources Control Board (SWRCB), Division of Drinking Water (DDW) can either adopt the USEPA standards or set more stringent standards, which are then codified in Title 22 of the California Code of Regulations. There are two general types of drinking water standards:

- Primary Maximum Contaminant Levels (MCLs) are health protective standards and are established using a very conservative risk-based approach for each constituent that takes into potential health effects, detectability and treatability, and costs of treatment. Public water systems may not serve water that exceeds Primary MCLs for any constituent.
- Secondary MCLs are based on the aesthetic qualities of the water such as taste, odor, color, and certain mineral content, and are considered limits for constituents that may affect consumer acceptance of the water.

The District routinely monitors the water that is treated and served to customers to ensure that water delivered to customers meets these drinking water standards. The results of this testing are reported to

### Water Supply Reliability 2020 Urban Water Management Plan Marin Municipal Water District



the SWRCB DDW following each test and are summarized annually in Water Quality Reports (also known as "Consumer Confidence Reports"), which are provided to customers by mail and made available on the District's website at <a href="https://www.marinwater.org/water-quality#button-1">https://www.marinwater.org/water-quality#button-1</a>.

Five of the seven local surface water reservoirs are located in a District-owned and protected watershed that substantially reduces the potential for contamination. The two reservoirs outside the protected watershed are located in rural areas with low population densities that are maintained by strict zoning requirements. In addition, the district has established Watershed Protection Agreements with landowners in these watersheds. Accordingly, the excellent water quality that the district has historically enjoyed is expected to continue into the future. There have been no instances when water quality issues have limited water supply or affected reliability.

Given the District's proactive monitoring and management of water quality in its source water supplies, water quality is not expected to impact the reliability of the District's available potable water supplies within the planning horizon (i.e., through 2045).

### **Recycled Water**

As described in Section 6.5, the District has developed an extensive recycled water program in the Las Gallinas area. The District has jointly investigated the feasibility of building water recycling systems in other areas as well, none of which were found to be economically feasible. An additional constraint to water recycling is saltwater intrusion into low-lying areas of the sewer collection systems that renders the water too salty to use for landscape irrigation, the primary market for recycled water in the District's service area. The District's existing recycled water system would also be vulnerable to saltwater intrusion in the event of a severe earthquake. Most of the low-lying areas subject to saltwater intrusion have soil conditions that would experience differential settlement in an earthquake and allow further saltwater inflow into the sewage collection system.

### 7.1.3 Climate Change Impacts to Supply

### **☑** *CWC* § 10635(b)

(4) Considerations of the historical drought hydrology, plausible changes on projected supplies and demands under climate change conditions, anticipated regulatory changes, and other locally applicable criteria.

Section 4.4 of this Urban Water Management Plan (UWMP or Plan) presents information on how the impacts of climate change are considered in projected demands in the District, and Section 6.10.1 provides a summary of potential climate change impacts on supplies.



### 7.2 Reliability by Type of Year

### ☑ CWC § 10631 (b)

Identify and quantify, to the extent practicable, the existing and planned sources of water available to the supplier over the same five-year increments described in subdivision (a), providing supporting and related information, including all of the following:

### ☑ CWC § 10631 (b)(1)

A detailed discussion of anticipated supply availability under a normal water year, single dry year, and droughts lasting at least five years, as well as more frequent and severe periods of drought, as described in the drought risk assessment. For each source of water supply, consider any information pertinent to the reliability analysis conducted pursuant to Section 10635, including changes in supply due to climate change.

### **☑** CWC § 10635 (a)

Every urban water supplier shall include, as part of its urban water management plan, an assessment of the reliability of its water service to its customers during normal, dry, and multiple dry water years. This water supply and demand assessment shall compare the total water supply sources available to the water supplier with the long-term total projected water use over the next 20 years, in five-year increments, for a normal water year, a single dry water year, and a drought lasting five consecutive water years. The water service reliability assessment shall be based upon the information compiled pursuant to Section 10631, including available data from state, regional, or local agency population projections within the service area of the urban water supplier.

Per the UWMP Guidebook 2020, the water service reliability assessment includes three unique year types:

- A <u>normal</u> hydrologic year represents the water supplies available under normal conditions, this could be an averaged range of years or a single representative year,
- A single dry year represents the lowest available water supply, and
- A <u>five-consecutive year drought</u> represents the driest five-year period in the historical record.

Identification of these dry year periods consistent with the UWMP Guidebook 2020 methodology is provided below.

SCWA reviewed historical hydrologic data for its system, and as indicated in **Table 7-2** below, has identified 2004 as the basis for its average year supply, 1977 as the basis for single-dry year supply, and the period of 1987-1991 as the basis for a five-year drought supply. The methodology used by SCWA is further discussed in SCWA's 2020 UWMP.



	Base Year	Available Supplies if Year Type Repeats			
Year Type		Quantification of available supplies is not compatible with this table and is provided elsewhere in the UWMP. Location: <b>Table 7-3</b> and <b>7-4</b>			
		Quantification of available supplies is provide in this table as either volume only, percent or or both.			
		Volume Available	% of Average Supply		
Average Year	2004	84,590	100%		
Single-Dry Year	1977	51,961	61%		
Consecutive Dry Years 1st Year	1987	79,385	94%		
Consecutive Dry Years 2nd Year	1988	84,150	99%		
Consecutive Dry Years 3rd Year	1989	86,262	102%		
Consecutive Dry Years 4th Year	1990	72,529	86%		
Consecutive Dry Years 5th Year	1991	69,270	82%		
NOTES: (a) Volumes are in units of AF.					

### 7.2.1 Purchased Water

Purchases from SCWA are assumed to increase in dry years, in order to supplement available local surface water supplies. In normal years, SCWA purchases are assumed to be 5,300 AF, but could increase during single and the early dry years of a drought, as long as MMWD's full allocation is available from SCWA. MMWD's current contract with SCWA allows for the District to purchase up to 14,300 AF; however, the District's ability to accept this volume is currently limited by infrastructure constraints. The District is projecting use of up to 10,000 AFY by 2030 subject to capital improvements that would increase conveyance capacity beyond the 10,000 AFY.

Based on the reliability information provided by SCWA, SCWA expects that its supply will be sufficient to meet all of its customers projected demands except for in the single-dry year scenario beginning in 2030. The projected shortfall under a single-dry year hydrology ranges from 16% to 19% of the total demands of SCWA's customers. For planning purposes, it is assumed that in the single-dry year and multiple dry-year scenarios, 7,200 AFY would be supplied in future years through 2045, which is within the projected supply with the projected cutbacks. This 7,200 AFY value is based on MMWD's estimates of reasonable hydraulic limits. The assumed purchases from SCWA are provided in **Table 7-3**, below, and are consistent with the reliability estimates provided by SCWA.



Table 7-3	Projected Availability	of SCWA Supply	(Responds to	DWR Table 7-1
	Trojectea Avanability	or seven suppry	(incopolitas to	Divit Tuble / 1

١	/ear Type	2025	2030	2035	2040	2045
Norma	l Year	5,300	5,300	5,300	5,300	5,300
Single-	Dry Year	7,200	7,200	7,200	7,200	7,200
	First year	7,200	7,200	7,200	7,200	7,200
led tht	Second year	7,200	7,200	7,200	7,200	7,200
enc oug	Third year	4,597	4,597	4,597	4,597	4,597
Dr Dr	Fourth year	4,300	4,300	4,300	4,300	4,300
	Fifth year	4,300	4,300	4,300	4,300	4,300
NOTES:						
(a) Vol	umes are in unit	s of AF.				

### 7.2.2 Surface Water

The projected availability of local surface water supplies by type are provided in **Table 7-4** below.

Table 7-4Projected Availability of Local Surface Water Supply (Responds to DWR Table 7-1)

١	/ear Type	2025	2030	2035	2040	2045
Norma	l Year	78,540	78,793	78,525	78,558	78,626
Single-	Dry Year	44,011	44,013	44,009	44,013	44,023
	First year	71,435	71,436	71,434	71,436	71,441
led ht	Second year	76,200	76,189	76,216	76,188	76,136
end oug	Third year	80,912	80,927	80,896	80,930	81,007
Dr Ext	Fourth year	67,479	67,471	67,502	67,470	67,401
	Fifth year	64,220	64,208	64,245	64,206	64,102
NOTES:						
(a) Vo	lumes are in unit	ts of AF.				

(b) Supplies are estimated based on modeling performed by Woodard and Curran.

### 7.2.3 <u>Recycled Water</u>

Supply availability for recycled water is not assumed to be impacted in dry years, as reflected in **Table 7-5** below, and is consistent with the demand projections identified in **Table 6-10**.



Table 7-5	Projected Availability	of Recycled Water Supply	(Responds to DWR Table 7-1)
	· · · · · · · · · · · · · · · · · · ·		

Y	ear Type	2025	2030	2035	2040	2045
Norma	l Year	750	750	750	750	750
Single-	Dry Year	750	750	750	750	750
	First year	750	750	750	750	750
led tht	Second year	750	750	750	750	750
enc oug	Third year	750	750	750	750	750
Dr.	Fourth year	750	750	750	750	750
	Fifth year	750	750	750	750	750
NOTES:						
(a) Vol	umes are in units	s of AF.				

#### 7.2.4 Raw Water

Supply availability for raw water is not assumed to be impacted in dry years, as reflected in Table 7-6 below, and is consistent with the demand projections identified in Table 4-4.

Table 7-6	Projected Availability of Raw Water Supply (Responds to DWR Table 7-1)
	responds to built rule rule supply (nesponds to built ruble r 1)

١	/ear Type	2025	2030	2035	2040	2045
Norma	l Year	171	174	176	176	176
Single-	Dry Year	171	174	176	176	176
	First year	171	174	176	176	176
led ht	Second year	171	174	176	176	176
end oug	Third year	171	174	176	176	176
Ext	Fourth year	171	174	176	176	176
	Fifth year	171	174	176	176	176
NOTES:						
(a) Vol	umes are in unit	s of AF.				



### 7.3 Supply and Demand Assessment

#### ☑ CWC § 10635 (a)

Every urban water supplier shall include, as part of its urban water management plan, an assessment of the reliability of its water service to its customers during normal, dry, and multiple dry water years. This water supply and demand assessment shall compare the total water supply sources available to the water supplier with the long-term total projected water use over the next 20 years, in five-year increments, for a normal water year, a single dry water year, and a drought lasting five consecutive water years. The water service reliability assessment shall be based upon the information compiled pursuant to Section 10631, including available data from state, regional, or local agency population projections within the service area of the urban water supplier.

As identified in Section 7.2 and associated tables, water supply availability differs for the District's supply sources during normal, single dry, and multiple dry years. **Table 7-7** shows the projected supply and demand totals for a normal year, consistent with those in **Table 6-10** and **Table 4-4**. **Table 7-8** shows the projected supply and demand totals for a single dry year, and **Table 7-9** shows the projected supply and demand totals for a single dry year, and **Table 7-9** shows the projected supply and demand totals for multiple dry year periods extending five years. The District is projected to have sufficient supplies to meet projected demands in normal years, single dry years, and multiple dry years through 2045.

	2025	2030	2035	2040	2045
Supply totals From DWR Table 6-9	84,761	85,017	84,751	84,784	84,852
Demand totals From DWR Table 4-3	38,019	38,046	37,974	38,051	38,207
Difference	46,742	46,972	46,777	46,733	46,645
NOTES:					

Table 7-7Normal Year Supply and Demand Comparison (DWR Table 7-2)

(a) Volumes are in units of AF.

Table 7-8Single Dry Year Supply and Demand Comparison (DWR Table 7-3)

	2025	2030	2035	2040	2045
Supply totals	52,132	52,137	52,135	52,139	52,149
Demand totals	38,019	38,046	37,974	38,051	38,207
Difference	14,113	14,091	14,161	14,088	13,942
NOTES:					
(a) Volumes are in units of AF.					



**Marin Municipal Water District** 

		2025	2030	2035	2040	2045
First year	Supply totals	79,556	79,560	79,560	79,562	79,567
	Demand totals	38,019	38,046	37,974	38,051	38,207
	Difference	41,537	41,514	41,586	41,511	41,360
Second	Supply totals	84,321	84,313	84,342	84,314	84,262
voar	Demand totals	38,019	38,046	37,974	38,051	38,207
year	Difference	46,302	46,267	46,368	46,263	46,055
Third	Supply totals	86,430	86,448	86,419	86,453	86,530
Inira	Demand totals	38,019	38,046	37,974	38,051	38,207
year	Difference	48,411	48,402	48,445	48,402	48,323
Fourth	Supply totals	72,700	72,695	72,728	72,696	72,627
Fourth	Demand totals	38,019	38,046	37,974	38,051	38,207
year	Difference	34,681	34,649	34,754	34,645	34,420
C:fth	Supply totals	69,441	69,432	69,471	69,432	69,328
FITTN	Demand totals	38,019	38,046	37,974	38,051	38,207
Difference		31,422	31,386	31,497	31,381	31,121
NOTES:						
(a) Volumes are in units of AF.						

Table 7-9 Multiple Dry Years Supply and Demand Comparison (DWR Table 7-4)

#### 7.4 Water Management Tools and Options

The District's Water Resources Plan 2040 dated March 2017 (Water Resources Plan; MMWD, 2017) evaluated various options and tools to improve water supply resiliency. A total of 40 resiliency options were developed and grouped into categories, including (1) water use efficiency, (2) reuse, (3) expanded SCWA facilities, (4) expanded storage, (5) water purchases and groundwater, (6) desalination, and (7) groundwater. The alternatives were evaluated based on a number of criteria, including reliability, technical complexity, environmental stewardship, local control, institutional complexity, public support, and project readiness. The resiliency options were then grouped into five alternatives, corresponding with specific themes, which included:

- Expand Existing Programs (consisting of enhanced conservation, Santa Rosa Plain conjunctive use, and watershed management);
- Minimize Infrastructure (consisting of enhanced conservation, SCWA Kastania Pump Station Upgrade, and Santa Rosa Plain Conjunctive Use);
- Dry Year Actions (consisting of enhanced conservation and spot market transfer;
- Maximize Reuse (consisting of regional indirect potable reuse); and
- Maximize Resiliency (consisting of enhanced conservation, regional indirect potable reuse, SCWA Kastania Pump Station upgrade, and watershed management).



Each of these alternatives was analyzed to determine their relative effectiveness and costs. Based on the analysis, the Water Resources Plan recommended the "Expand Existing Programs" alternative, which includes following actions:

- <u>Evaluate Increased Conservation</u>: The Water Resources Plan recommended that the District evaluate implementing an enhanced level of water conservation beyond the current level of commitment referred to internally as Program A.
- <u>Invest in Watershed Management</u>: The Water Resources Plan recommended that the District expand the implementation of a watershed management action to reduce accumulated fuels and brush in the watershed, which was projected to increase yield by approximately 210 AFY.
- Explore Groundwater Partnering Opportunities: The Water Resources Plan recommended that the District explore partnering with a SCWA customer that also uses groundwater supplies to implement an in-lieu groundwater recharge program. Under such a program, the District would allow a portion of its SCWA supply to be used by a partner agency in normal and wet years to offset local groundwater pumping, allowing the basin to recharge and store additional water om those years. The partner agency would then rely on this replenished groundwater supply in dry years, sending some or all of its SCWA supply to the District. This would allow the District to functionally "store" water in the groundwater basin for use during dry years.

The District is pursuing additional actions to optimize water supply during dry years, including ongoing projects to utilize reservoirs that are typically not utilized, including Phoenix Lake.

In addition, the District is a member of the North Bay Water Reuse Authority (NBWRA) which is a regional water recycling organization formed to put recycled water to its broadest and most beneficial use. NBWRA consists of ten local agencies covering 315 square miles in the portions of Marin, Sonoma and Napa counties that surround the northern rim of the San Francisco Bay. As part of NBWRA, the District has made great strides to expand recycled water use.

In addition, the District continues to work to expand the use of recycled water to include non-irrigation uses. Another option being evaluated includes upgrades to the Kastania Pump Station to increase the capacity of the pipeline from SCWA by installing new variable speed pumps.



### 7.5 Drought Risk Assessment

### ☑ CWC § 10635(b)

Every urban water supplier shall include, as part of its urban water management plan, a drought risk assessment for its water service to its customers as part of information considered in developing the demand management measures and water supply projects and programs to be included in the urban water management plan. The urban water supplier may conduct an interim update or updates to this drought risk assessment within the fiveyear cycle of its urban water management plan update. The drought risk assessment shall include each of the following:

(1) A description of the data, methodology, and basis for one or more supply shortage conditions that are necessary to conduct a drought risk assessment for a drought period that lasts five consecutive water years, starting from the year following when the assessment is conducted.

(2) A determination of the reliability of each source of supply under a variety of water shortage conditions. This may include a determination that a particular source of water supply is fully reliable under most, if not all, conditions.

(3) A comparison of the total water supply sources available to the water supplier with the total projected water use for the drought period.

(4) Considerations of the historical drought hydrology, plausible changes on projected supplies and demands under climate change conditions, anticipated regulatory changes, and other locally applicable criteria.

### 7.5.1 Characteristic Five-Year Water Use

As a first step to the Drought Risk Assessment, water suppliers are advised to estimated unconstrained water demand for the next five years (2021-2025). Unconstrained water demand is the expected water use in the absence of drought water use restrictions. The forecast of unconstrained demand for the next five-years is shown in **Table 7-10** below.

	2021	2022	2023	2024	2025
Total Projected Use During Drought Period	34,329	34,898	35,476	36,064	36,661

### Table 7-10 Characteristic Five-Year Water Use

### 7.5.2 <u>Risk Assessment Projections</u>

Based on information provided by SCWA, SCWA does not anticipate any supply shortfalls in the 2021 through 2025 period, even under hydrologically dry conditions. Based on the current reservoir levels MMWD's local supply system, however, supplies are expected to be constrained in the near term.



# Table 7-11Five-Year Drought Risk Assessment Tables to Address Water Code10635(b) (DWR Table 7-5)

2021	Total
Total Water Use	34,329
Total Supplies	79,556
Surplus/Shortfall w/o WSCP Action	45,227
Planned WSCP Actions (use reduction and supply augmentation)	
WSCP - supply augmentation benefit	
WSCP - use reduction savings benefit	
Revised Surplus/(shortfall)	
Resulting % Use Reduction from WSCP action	

2022	Total
Total Water Use	34,898
Total Supplies	84,321
Surplus/Shortfall w/o WSCP Action	49,423
Planned WSCP Actions (use reduction and supply augmentation)	
WSCP - supply augmentation benefit	
WSCP - use reduction savings benefit	
Revised Surplus/(shortfall)	
Resulting % Use Reduction from WSCP action	

2023	Total
Total Water Use	35,476
Total Supplies	86,430
Surplus/Shortfall w/o WSCP Action	50,954
Planned WSCP Actions (use reduction and supply augmentation)	
WSCP - supply augmentation benefit	
WSCP - use reduction savings benefit	
Revised Surplus/(shortfall)	
Resulting % Use Reduction from WSCP action	



# Table 7-11Five-Year Drought Risk Assessment Tables to Address Water Code10635(b) (DWR Table 7-5)

2024	Total
Total Water Use	36,064
Total Supplies	72,700
Surplus/Shortfall w/o WSCP Action	36,636
Planned WSCP Actions (use reduction and supply augmentation)	
WSCP - supply augmentation benefit	
WSCP - use reduction savings benefit	
Revised Surplus/(shortfall)	
Resulting % Use Reduction from WSCP action	

2025	Total
Total Water Use	36,661
Total Supplies	69,441
Surplus/Shortfall w/o WSCP Action	32,780
Planned WSCP Actions (use reduction and supply augmentation)	
WSCP - supply augmentation benefit	
WSCP - use reduction savings benefit	
Revised Surplus/(shortfall)	
Resulting % Use Reduction from WSCP action	
NOTES:	
(a) Volumes are in units of AF.	



### 8. WATER SHORTAGE CONTINGENCY PLANNING

### ☑ CWC § 10640

(a) Every urban water supplier required to prepare a plan pursuant to this part shall prepare its plan pursuant to Article 2 (commencing with Section 10630). The supplier shall likewise periodically review the plan as required by Section 10621, and any amendments or changes required as a result of that review shall be adopted pursuant to this article.

(b) Every urban water supplier required to prepare a water shortage contingency plan shall prepare a water shortage contingency plan pursuant to Section 10632. The supplier shall likewise periodically review the water shortage contingency plan as required by paragraph (10) of subdivision (a) of Section 10632 and any amendments or changes required as a result of that review shall be adopted pursuant to this article.

The Water Shortage Contingency Plan (WSCP) for Marin Municipal Water District (MMWD or District) is included in this Urban Water Management Plan (UWMP) as **Appendix H**. The WSCP serves as a standalone document to be engaged in the case of a water shortage event, such as a drought or supply interruption, and defines specific policies and actions that will be implemented at various shortage level scenarios. The primary objective of the WSCP is to ensure that the District has in place the necessary resources and management responses needed to protect health and human safety, minimize economic disruption, and preserve environmental and community assets during water supply shortages and interruptions. Consistent with CWC §10632, the WSCP includes six levels to address shortage conditions ranging from up to 10% to greater than 50% shortage, identifies a suite of demand mitigation measures for the District to implement at each level, and identifies procedures for the District to annually assess whether or not a water shortage is likely to occur in the coming year, among other things.

A summary of the key elements of the WSCP including water shortage levels and demand-reduction actions is shown in **Table 8-1**, **Table 8-2** and **Table 8-3**. Additional details are provided in **Appendix H.** 

Shortage Level	Percent Shortage Range	Shortage Response Actions
0	0%	<ul> <li>Includes water waste prohibitions effective at all times.</li> </ul>
1	Up to 10%	<ul> <li>If rainfall is 30% below average for the water year<sup>1</sup> as of April 1<sup>st</sup></li> <li>Includes implementation of mandatory restrictions on end uses (see Table 8-2) as well as agency actions (see Table 8-3).</li> </ul>
2	Up to 20%	<ul> <li>Total reservoir storage is in the vicinity of 45,000 acre-feet on January 1<sup>st</sup>.</li> <li>Includes implementation of mandatory restrictions on end uses (see Table 8-2) as well as agency actions (see Table 8-3).</li> </ul>
3	Up to 30%	<ul> <li>Total reservoir storage is in the vicinity of 50,000 AF on February 1<sup>st</sup></li> </ul>

 Table 8-1
 Water Shortage Contingency Plan Levels (DWR Table 8-1)


Shortage Level	Percent Shortage Range	Shortage Response Actions
		<ul> <li>Includes implementation of mandatory restrictions on end uses (see Table 8-2) as well as agency actions (see Table 8-3).</li> </ul>
4	Up to 40%	<ul> <li>Total reservoir storage is in the vicinity of 55,000 AF on April 30<sup>th</sup> and/or storage level projections using average rainfall indicate December 1<sup>st</sup> storage in vicinity of 30,000 AF.</li> <li>Includes implementation of mandatory restrictions on end uses (see <b>Table 8-2</b>) as well as agency</li> </ul>
5	Up to 50%	<ul> <li>actions (see Table 8-3).</li> <li>Total reservoir storage on December 1 is less than 30,000 AF.</li> <li>Includes implementation of mandatory restrictions on end uses (see Table 8-2) as well as agency actions (see Table 8-3).</li> </ul>
6	>50%	<ul> <li>Continued drought and forecasted decreasing storage levels or catastrophic event beyond Stage 5, or total reservoir storage is projected to be in the vicinity of 30,000 AF on April 1st.</li> <li>Includes implementation of mandatory restrictions on end uses (see Table 8-2) as well as agency actions (see Table 8-3).</li> </ul>
NOTES: 30%	below average rain	fall is defined as less than 32 inches of rainfall at the
Lagunitas rai	nfall gauge for the	time period of October 1 thru March 31.



Table 8-2	Demand Reduction Actions (DWR Table 8-2)	

Shortage Level	Demand Reduction Actions	How much is this going to reduce the shortage gap? (a)	Additional Explanation or Reference (optional)	Penalty, Charge, or Other Enforcement?
Landscape Irr	igation			
1, 2, 3, 4, 5, 6	Other landscape restriction or prohibition	5%	The application of potable water to outdoor landscapes during and within 48 hours after measurable rainfall is prohibited.	Y
1, 2, 3, 4, 5, 6	Restrict or prohibit runoff from landscape irrigation	5%	Irrigation shall not be conducted in a manner or to an extent that allows water to run off or overspray the areas being watered. Every customer is required to have his or her water distribution lines and facilities under control at all times to avoid water waste.	Y



Shortage Level	Demand Reduction Actions	How much is this going to reduce the shortage gap? (a)	Additional Explanation or Reference (optional)	Penalty, Charge, or Other Enforcement?
1, 2, 3, 4, 5, 6	Limit landscape irrigation to specific times	5%	Any landscape irrigation between the hours of 9:00 a.m. and 7:00 p.m. is prohibited. Necessary testing and repair of irrigation systems for the purpose of eliminating water waste is permitted during the hours of 9:00 a.m. and 7:00 p.m. Customers shall maintain appropriate documentation of any necessary testing and repairs for these purposes. For example, this documentation may include, but not be limited to, any applicable reports, invoices, photos, videos, and/or receipts for materials and labor related to the testing and repairs.	Y
1, 2, 3	Limit landscape irrigation to specific days	5%	Operating outdoor irrigation systems using potable water for the purpose of irrigating ornamental landscape areas or turf areas more than three days, within any calendar week is prohibited. For the purposes of this section, "calendar week" shall mean a period running from Monday- Sunday.	Υ



Shortage Level	Demand Reduction Actions	How much is this going to reduce the shortage gap? (a)	Additional Explanation or Reference (optional)	Penalty, Charge, or Other Enforcement?
3, 4, 5, 6	Limit landscape irrigation to specific days	25%	Increase restrictions on irrigation. For example, operating outdoor irrigation systems using potable water for the purpose of irrigating with overhead spray more than two days, as assigned by the District, within any calendar week may be prohibited. For the purposes of this section, "calendar week" shall mean a period running from Monday- Sunday.	Y
1, 2, 3, 4, 5, 6	Other landscape restriction or prohibition	5%	Irrigating ornamental turf on public street medians is prohibited.	Υ
4, 5, 6	Prohibit certain types of landscape irrigation	35%	Golf course irrigation with potable or raw water of any areas beyond the greens and tee areas is prohibited	Y
4	Limit landscape irrigation to specific days	35%	Further increase restrictions on irrigation. For example, irrigation may be limited to one day per week, except for minimal irrigation to maintain tree health.	Y

Table 8-2	Demand Reduction Actions (DWR Table 8-2)
-----------	--



Shortage Level	Demand Reduction Actions	How much is this going to reduce the shortage gap? (a)	Additional Explanation or Reference (optional)	Penalty, Charge, or Other Enforcement?
5	Other landscape restriction or prohibition	45%	Further increase restrictions on irrigation. For example, all landscape irrigation may be prohibited, except minimal irrigation to maintain tree health.	Y
6	Other	55%	Water allotments go into effect (60% of last three-year average January usage)	Y
6	Other	55%	The use of potable water for any purpose other than human health and sanitation is prohibited.	Y
<b>Restrictions o</b>	n Consumer Products			
1, 2	Other	5%	The installation of reverse osmosis water purifying systems not equipped with an automatic shutoff unit is prohibited.	Y
3, 4, 5, 6	Pools and Spas - Require covers for pools and spas	25%	Pool covers are required for all outdoor swimming pools.	Y
<b>Restrictions o</b>	n New Connections and Landscap	ing		
1, 2	Other	5%	(A) Single pass cooling systems for air conditioning or other cooling system applications are prohibited, unless required for health or safety reasons.	Y

Table 8-2	Demand Reduction Actions	(DWR Table 8-2)	
-----------	--------------------------	-----------------	--



Shortage Level	Demand Reduction Actions	How much is this going to reduce the shortage gap? (a)	Additional Explanation or Reference (optional)	Penalty, Charge, or Other Enforcement?
4, 5, 6	Other	35%	No installation of new landscapes including no expansion of existing landscapes.	Y
5	Other	45%	Consider restrictions on new water service applications such that they will be granted upon the condition that water shall be used for interior purposes and landscaping shall be delayed until the District determines that Stage 5 rationing levels are no longer needed.	Y
5	Other	45%	New non-recirculating systems for conveyer car wash applications is prohibited.	Y
6	Other	55%	Potable water shall only be used for health and safety.	Y
6	Moratorium or Net Zero Demand Increase on New Connections	55%	New water service applications will not be granted.	Y
<b>Restrictions o</b>	n Commercial Operations			
1, 2, 3, 4, 5, 6	Lodging establishment must offer opt out of linen service	5%	Lodging establishments must provide patrons the option of not having towels and linen laundered daily.	Υ



Shortage Level	Demand Reduction Actions	How much is this going to reduce the shortage gap? (a)	Additional Explanation or Reference (optional)	Penalty, Charge, or Other Enforcement?
1, 2, 3, 4, 5, 6	Restaurants may only serve water upon request	5%	Drinking water served upon request only.	Y
Other				
1, 2, 3, 4, 5, 6	Prohibit use of potable water for washing hard surfaces	5%	The washing of sidewalks, walkways, driveways, parking lots and all other hard surfaced areas by direct hosing Is prohibited, except as may be permitted by current regulations pertaining to urban water runoff pollution prevention as defined by the Marin County Stormwater Pollution Prevention Program and other controlling agencies, or as determined necessary by any public agency for the health and safety of the public.	Y



Shortage Level	Demand Reduction Actions	How much is this going to reduce the shortage gap? (a)	Additional Explanation or Reference (optional)	Penalty, Charge, or Other Enforcement?
1, 2, 3, 4, 5, 6	Customers must repair leaks, breaks, and malfunctions in a timely manner	5%	The escape of water through breaks or leaks within the customer's plumbing or private distribution system for any substantial period of time within which such break or leak should reasonably have been discovered and corrected is prohibited. A period of forty-eight hours after the customer discovers such a leak or break, or receives notice from the District of such leak or break, whichever occurs first, shall constitute a "reasonable time" within which to correct such leak or break for the purposes of this section.	Y
3, 4, 5, 6	Other	25%	Gutter flooding is prohibited.	Y
4, 5, 6	Other	35%	Decorative water fountains or pools, including the refilling or make-up of any decorative fountain or pool (excludes recreational pools) is prohibited.	Υ
4	Other	35%	Consider implementation of drought rates and water surcharges	Ν



Shortage Level	Demand Reduction Actions	How much is this going to reduce the shortage gap? (a)	Additional Explanation or Reference (optional)	Penalty, Charge, or Other Enforcement?
4	Other	35%	Begin discussion on limiting or excluding new service connections during drought	Ν
5, 6	Other landscape restriction or prohibition	45%	Filling recreational pools with potable water is prohibited	Y
5	Other	45%	Enact Drought Rates and Water Surcharges	
1, 2, 3, 4, 5, 6	Require automatic shut-off hoses	5%	Using a garden hose without a shut- off nozzle is prohibited.	Ν
1, 2, 3, 4, 5, 6	Other	5%	Any excess water runoff flowing onto the public right-of-way at a rate of one gallon per minute or greater not caused by storm water or naturally occurring groundwater is prohibited.	Y
1, 2, 3, 4, 5, 6	Other	5%	Use of private fire lines or private fire taps for any purposes other than fire suppression and necessary testing is prohibited.	Y
3, 4, 5, 6	Other	25%	Using potable water to wash vehicles except at commercial carwash facilities that use recycled water, is prohibited.	Y
4, 5, 6	Other	35%	Powerwashing of any structure is prohibited.	Y

	Table 8-2	Demand Reduction Actions	(DWR Table 8-2
--	-----------	--------------------------	----------------



Table 8-2	<b>Demand Reduction Actions</b>	(DWR Table 8-2)
-----------	---------------------------------	-----------------

Shortage Level	Demand Reduction Actions	How much is this going to reduce the shortage gap? (a)	Additional Explanation or Reference (optional)	Penalty, Charge, or Other Enforcement?			
4, 5, 6	Limit use of potable water for construction and dust control	35%	Dust control, compaction, sewer flushing, street cleaning, or any other use, as determined by the District, which can be met with disinfected tertiary recycled water is prohibited.	Y			
4,5,6Other35%Request that local fire departments limit training exercises that use potable water and cease hydrant testing.Y							
Notes: (a) The reductions shown are the combined reductions for all the actions associated with the particular shortage level.							



## Table 8-3Supply Augmentation and Other Actions (DWR Table 8-3)

Shortage Level	Supply Augmentation Methods and Other Actions by Water Supplier	How much is this going to reduce the shortage gap?	Additional Explanation or Reference (optional)
1	Increase supplemental water imports and closely monitor storage levels and weather conditions	Up to 5,000 AF	Range dependent on hydraulic capabilities
1	Enact dry year stream release flow reductions	Up to 100 AF	As defined under Water Right Order 95-17 and agreements with NMWD
2	Minimize system flushing	Unknown	Focus on water quality improvements
3	Increase water waste patrols	Unknown	
3	Restrict line flushing to include only regulatory compliance actions	Unknown	Flush for violations of water quality regulations or required disinfection for new construction and repairs.
3	Increase system leak repair to prioritize class 2 leaks	Unknown	
3	Access Stored Emergency Supply	Unknown	Rental of generator for Soulajule Reservoir to transfer water to Nicasio Reservoir
3	Access Stored Emergency Supply	Unknown	Initiate adjustments to pipe configuration to transfer water from Phoenix Lake to Bon Tempe Lake
4	Initiate Study/Application for Temporary Urgency Change Petition	Up to 2,500 AF	Change to environmental releases – possible reduction of migration flow volumes.
4	Increase water waste patrols	Unknown	
4	Increase system leak repair to prioritize class 1-3 leaks	Unknown	
5	Access Stored Emergency Supply	500 AF	Develop access to unusable water in reservoir
5	Initiate water transfers	Unknown	Pending investigation
5	Initiate groundwater banking agreement for increased imports	Unknown	Long term



Shortage Level	Supply Augmentation Methods and Other Actions by Water Supplier	How much is this going to reduce the shortage gap?	Additional Explanation or Reference (optional)
5	Access Emergency Water Treatment skids for desalination	Unknown	Conducting feasibility analysis
NOTES:			



## 9. DEMAND MANAGEMENT MEASURES

#### ☑ CWC § 10631 (e)

*Provide a description of the supplier's water demand management measures. This description shall include all of the following:* 

(1) (A) For an urban retail water supplier, as defined in Section 10608.12, a narrative description that addresses the nature and extent of each water demand management measure implemented over the past five years. The narrative shall describe the water demand management measures that the supplier plans to implement to achieve its water use targets pursuant to Section 10608.20.

(B) The narrative pursuant to this paragraph shall include descriptions of the following water demand management measures:

(i) Water waste prevention ordinances.

(ii) Metering.

(iii) Conservation pricing.

(iv) Public education and outreach.

(v) Programs to assess and manage distribution system real loss.

(vi) Water conservation program coordination and staffing support.

(vii) Other demand management measures that have a significant impact on water use as measured in gallons per capita per day, including innovative measures, if implemented.

Demand management measures (DMMs) are specific actions a water supplier takes to support its water conservation efforts. Implementation of DMMs over the past a couple decades has helped the Marin Municipal Water District (District) achieve its 2015 Interim and 2020 water use targets under SB X7-7 (Section 5). The following sections provide a comprehensive description of the District's water conservation programs, including programs currently being implemented and planned programs.

### 9.1 Regional Water Conservation

The Sonoma County Water Agency (SCWA), along with the cities of Santa Rosa, Rohnert Park, Sonoma, Cotati, and Petaluma, the Town of Windsor, and North Marin Water District (NMWD), the District and Valley of the Moon Water District (VOMWD) (the Partners), formed the Sonoma-Marin Saving Water Partnership (SMSWP) in 2010. The SMSWP's Memorandum of Understanding was amended in May 2018, extending the term another ten years, and adding language to streamline the addition of members to the SMSWP. Two new Partners have subsequently joined, with California American Water-Larkfield joining in January 2019 and the City of Healdsburg joining in August 2019.

SCWA coordinates the work of the SMSWP in conjunction with the Water Advisory Committee (WAC), which provides input to SCWA and holds certain powers and responsibilities enumerated in the Restructured Agreement for Water Supply between SCWA and SMSWP. The SMSWP is committed to continued water conservation and is in compliance with the final 2020 gallons per capita targets established by Senate Bill X7-7.

Demand Management Measures 2020 Urban Water Management Plan Marin Municipal Water District



## 9.1.1 <u>Funding</u>

SCWA's wholesaler water conservation programs are funded by the Partners annually through a WAC recommended budget that allocates a Water Conservation sub-charge for each acre-foot of water sold. The Partners have agreed to expend \$15 million dollars on water conservation implementation from July 2018 through June 2028. They have also agreed to maintain membership in good standing with the California Water Efficiency Partnership (CalWEP) and implement or use best efforts to secure the implementation of any water conservation requirements added as terms or conditions of SCWA's appropriative water rights or other regulation or law.

SCWA pursues grant funding on behalf of the SMSWP to off-set some of the programmatic costs associated with water use efficiency (WUE) programs and to test new technology. In the last five years, the Agency was awarded over \$1.46 million dollars for implementing WUE programs in our region.

### 9.1.2 Annual Report

The Partners are committed to remain as members in good standing of CalWEP and to implement water conservation measures that provide regional benefits and/or that may exceed the targets established from time to time by the Partners or the state. The Partners will implement or use best efforts to secure the implementation of any water conservation requirements and will publish an Annual Report to track progress. The Annual Report will track program implementation, highlight program milestones, and reinforce the importance of protecting and preserving water resources for future generations. The 2019/2020 Annual Report for the SMSWP could be found in the SMSWP's website.<sup>15</sup>

### 9.1.3 <u>Water Education Program</u>

The Water Education Program is a comprehensive approach to helping educators teach students the "value" of water as an important natural resource. Water and energy conservation and stewardship of our local watersheds is promoted throughout the program. Students are encouraged to use water wisely and make environmentally sustainable choices to help secure a reliable source of freshwater now and in the future. The program includes free curriculum materials aligned with the existing California State Frameworks and the California Science Standards, a lending library of videos, interactive models and printed materials, production of a newsletter for teachers and endorsement, participation and financial sponsorship of events, assemblies and workshops. All of the education programs and materials are free to teachers in the service area. An online assembly program was launched to support educators during the COVID-19 pandemic.

<sup>&</sup>lt;sup>15</sup> The 2019/2020 Annual Report for the SMSWP: http://www.savingwaterpartnership.org/wp-content/uploads/SMSWP-Annual-Report-2020-FINAL.pdf.



### 9.1.4 Public Outreach Program

The SMSWP develops an annual regional outreach campaign that aligns with our current water supply conditions and promotes water use efficiency programs. Over the last few years, the campaigns have included the following:

- Saving Water Ensures Water for What You Love (2020),
- Together Making Water Conservation a California Way of Life (2019),
- There's Never Enough to Waste. (2017 & 2018),
- Thank You for Doing Your Part (2016), and
- Take it From the Tap (2016).

SCWA, in collaboration with the members of the SMSWP, produces collateral material that aligns with the specific campaign. SCWA coordinates an annual media buy that includes outreach in English and Spanish. Each member of the SMSWP can choose to supplement the campaign with their own media buys. The buys generally include the following:

- Radio (streaming and broadcast),
- Newsprint and online digital media placements in 14 various local publications,
- Sonoma County Fair presence,
- Social Media (Facebook, Twitter, Instagram, YouTube, NextDoor),
- Mall banners, and
- Movie theater trailers.

### 9.1.5 <u>Regional Programs</u>

SCWA on behalf of the SMSWP implements numerous regional programs. This includes offering staff support for interested Partners as a cost-effective way to offer local programs to customers of smaller agencies. Some of these programs are:

- High Efficiency Clothes Washer Water Rebate a rebate for replacing a top-loading clothes washer with a qualifying front-loading clothes washer.
- Qualified Water Efficient Landscaper Training Program A low-cost professional certification program that educates landscapers about irrigation system auditing, while providing customers with a trusted source for knowledgeable hired help that can save them water.
- Eco-Friendly Garden Tour An annual self-guided garden tour in Sonoma County and North Marin that promotes sustainable landscaping practices. This tour transitioned to an online video format in 2020 to adapt to the COVID-19 pandemic.
- Landscape Design Templates These free, front yard designs are scalable to fit landscaped areas up to 2,500 square feet, ready-to-permit, and in compliance with local Water Efficient Landscape Ordinances.
- Water Smart Plant Label A free water smart plant labeling program to local nurseries. The water smart plant label highlights low water use plants to nursery customers and promotes sustainable landscaping practices in Sonoma and Marin counties.
- Water-Energy Rebates for Restaurants and Food Service Facilities a rebate program for replacing inefficient commercial kitchen equipment with new water and energy efficient models.

# Demand Management Measures 2020 Urban Water Management Plan



### Marin Municipal Water District

SCWA supports promoting new and innovative models to increase water use efficiency in our region. Some of the pilot projects we have collaborated with in the past include:

- PAYS Program (Windsor) An on-bill financing program that allows water customers to fund their own water and energy improvements with a long-term payback on their water bill.
- SmartMarkets Pilot (VOMWD) A water market that allows for 'eco-shares' to be earned for reducing demand and redeemed for various incentives.
- Water Smart Software (Cotati) A community based social marketing platform that compares a customer's water use to their neighbors to encourage behavioral change.
- Barnacle Pilot Program (All) An online platform that provides real-time water use data to the customer outside of the water utilities billing infrastructure.
- Unmetered Flow Reducer (NMWD) An in-line device that is placed between the meter and the customer connection that allows small leaks to be 'batched' through the meter, thus reducing unaccounted for water from low flow leaks and allowing the customer to be notified that a leak in occurring.

SCWA participates in numerous regional and statewide initiatives to ensure the SMSWP is on the forefront of water use efficiency, legislation, and conservation planning, such as:

- California Water Efficiency Partnership (Programs Subcommittee, Research Subcommittee), successor organization of the former California Urban Water Conservation Council,
- Association of California Water Agencies (Water Management and Water Use Efficiency Subcommittees),
- Russian River Watershed Association, and
- California Landscape Contractors Association.

The SMSWP has received notable recognition for effective collaboration and program implementation. Below are the awards the SMSWP has received.

- EPA Water Sense Excellence Award 2020 and Sustained Excellence Award 2020,
- EPA Water Sense Excellence Award 2019 and Sustained Excellence Award 2019,
- EPA Water Sense Excellence Award 2018 and Sustained Excellence Award 2018,
- EPA Water Sense Excellence Award 2017 and Sustained Excellence Award 2017, and
- EPA Water Sense Partner of the Year 2016.

# 9.2 Agency Water Conservation

### ☑ CWC § 10631 (e)

*Provide a description of the supplier's water demand management measures. This description shall include all of the following:* 

(1) (A) For an urban retail water supplier, as defined in Section 10608.12, a narrative description that addresses the nature and extent of each water demand management measure implemented over the past five years.

The District's programs for demand management through water conservation began in 1971 when water conservation literature from the American Water Works Association was inserted into water bills. By the

# Demand Management Measures 2020 Urban Water Management Plan Marin Municipal Water District



mid-1970s, the District's programs had expanded to include retrofits of water-using fixtures and have continued to expand over the last 45 years.

When the District was embarking on its Integrated Water Resources Management Program in 1991, a review of water demands found that an estimated 11% reduction in water use had occurred during the period from 1970 to 1987 after taking into account the additional services installed during the period. A similar review in 1999 found that the demand had been reduced by an estimated 25% during the period from 1970 to 1998.

In June 2007, the District adopted its 2007 Water Conservation Master Plan. By reference, the Conservation Master Plan is incorporated herein, and included in **Appendix I**.

In late 2010, the District joined with Sonoma County Water Agency (SCWA or Sonoma Water) and the Water Contractors to form the Sonoma Marin Saving Water Partnership (SMSWP). SMSWP allows the District, SCWA, and the Water Contractors to maximize cost-effective water conservation by identifying projects and programs that can be implemented regionally. SMSWP committed each signatory to remain in good standing with the California Urban Water Conservation Council (CUWCC), to be on track with implementing the best management practices (BMPs), and to budget and spend \$15 million over 10 years for implementing water conservation programs. Since SMSWP was formed, a regional water conservation educational campaign was launched, and regional conservation programs are being promoted. The CUWCC has been dissolved since the 2015 UWMP, and thus CUWCC BMP annual reports are no longer prepared. However, the District continues to implement DMMs in general accordance with the CUWCC BMPs.

The District is currently preparing a Water Efficiency Master Plan to assess and identify opportunities for future program implementation. This plan is anticipated to be complete in 2021 and will be used to inform and guide the District's conservation efforts over the planning horizon.

The following sections describe the District's efforts in implementing each of the DMMs as listed in CWC § 10631(f). Additional information regarding DMM implementation is provided in the District's 2020 Urban Water Management Plan Water Demand Analysis and Water Conservation Measures Update (**Appendix C**).

### 9.2.1 DMM 1 – Water Waste Prevention Ordinances

Title 13 of the District's Code, Water Service Conditions and Water Conservation Measures, includes a section on water waste prohibitions (Section 13.04.020). This section was updated in 2021 to explicitly state that the waste of water is to be prohibited. The section prohibits nonessential uses, places restrictions on irrigation watering times, limits days per week of allowed irrigation and reverse-osmosis units, and includes prohibitions on single-pass cooling systems and non-recirculating systems for conveyor carwash applications for new connections. The current Title 13 of the District's Code is included in **Appendix J**.



#### 9.2.2 DMM 2 – Metering

#### ☑ CWC § 526 (a)

Notwithstanding any other provision of law, an urban water supplier that, on or after January 1, 2004, receives water from the federal Central Valley Project under a water service contract or subcontract ... shall do both of the following:

(1) On or before January 1, 2013, install water meters on all service connections to residential and nonagricultural commercial buildings constructed prior to January 1, 1992, located within its service area.

(2) On and after March 1, 2013, or according to the terms of the Central Valley Project water contract in operation, charge customers for water based on the actual volume of deliveries, as measured by a water meter.

#### ☑ CWC § 527 (a)

(a) An urban water supplier that is not subject to Section 526 shall do both of the following:

(1) Install water meters on all municipal and industrial service connections located within its service area on or before January 1, 2025.

The District is fully metered and requires that all new connections be metered. The District requires separate landscape meters for qualified customers and has policies about the use of installed meters. For instance, Section 13.02.065 of the District's Code states that "...use of any District water not metered is prohibited." Additionally, Section 13.020.070 states that "it is unlawful [...] to remove, replace, alter, or damage any water meter or components thereof." If a violator is found, the District may impose penalties as outlined in the ordinance.

The District continues to replace meters as they age. The continued objective of the meter replacement program is to achieve a 20-year life cycle for all meters. Tests performed indicate that meters older than 20 years run, on average, 6% slower than new meters. By replacing the oldest meters in the system on an ongoing basis, the meter change program improves overall meter accuracy and retains revenue that would otherwise be lost by inaccurate meters.

The District has completed three Advanced Metering Infrastructure (AMI) pilot projects resulting in the installation of over 5,000 AMI meters on all dedicated irrigation accounts and many of the highest water users. An AMI Feasibility Study was completed in October 2020 to evaluate districtwide deployment of AMI technology.

#### 9.2.3 DMM 3 – Conservation Pricing

Conservation pricing sends a signal to customers about their water use and is always in place. MMWD's rate structure encourages conservation by including both a fixed service charge and a per-unit charge, reflecting both fixed costs and costs based on the amount of water used. The per-unit charge includes the cost of water supply, treatment and distribution, and watershed maintenance.

In May 2019, the District Board of Directors approved Ordinance No. 442, which enacts changes to the water rates and rate structure through 1 July 2023. This Ordinance includes "(1) an increase to the Bi-Monthly Fixed Service Charge, Fixed Watershed Management Fee and Fixed Private Fire Service Line Charges; (2) implementation of the Capital Maintenance Fee that will automatically increase or decrease according to the change in the Engineering News-Record Construction Cost Index and will not exceed 4%

# Demand Management Measures 2020 Urban Water Management Plan Marin Municipal Water District



per annum; (3) adjustments to the Tier Rates for all customer classes; (4) changes to the Low Income Discount Program that enhance the availability of the program; and (5) the adoption of another element of the District's water conservation program pursuant to Water Code Section 375 by the creation of a Super Water Saver Program" (MMWD, 2019).

To determine the per unit charge, the District uses a system of three or four billing tiers, depending on the customer class. The current rates went into effect on 1 July 2019 and are available on the District's website: <a href="https://www.marinwater.org/understanding-your-bill#button-1">https://www.marinwater.org/understanding-your-bill#button-1</a>

For residential customers, the District uses four tiers, with different allotments (volume ranges) for summer months (June through November) and winter months (December through May). For non-residential potable water, single-family residential irrigation, recycled, and raw water customers, the District uses three tiers, which do not vary by season.

For non-residential customers, the breakpoints of the three tiers are determined using a more site-specific method. Each non-residential customer has a water entitlement and a water budget, and the tiers are scaled relative to the baseline water budget. The water entitlement is the maximum amount of water the District is committed to supply any individual customer on an annual basis. The District has unique methods for determining the water entitlement for accounts that were in service prior to 1991, for new accounts, and for new dedicated irrigation accounts. The water budget is the District's determination of the actual consumption requirement of the customer.

## 9.2.4 DMM 4 – Public Education and Outreach

The District offers free water audits to residential and non-residential customers. During these water use surveys, a District representative evaluates the existing water-using fixtures, landscape irrigation system, and water use patterns to identify water savings opportunities. After the visit, the customer receives a written report of the existing water use patterns and a list of recommendations to increase efficiency. For large irrigation systems, the District offers a specialized large landscape water use survey.

For all schools located within the District's service area, MMWD offers several free programs, including watershed field trips, classroom presentations, and various materials intended to guide curriculum for various ages. All school programs are designed to support education standards while fostering water conservation and environmental stewardship. Between 2007 and 2020, 111,024 students have been reached (MMWD, 2020). The District also has a list of resources with links on their website that customers can access. For residential customers, the District has a Do-It-Yourself home water survey that customers can print and work through the activities.

In addition to water use audits and school education, the District offers a number of rebates and discounts to its customers for water-saving fixtures and systems, which also serve to educate and inform customers of conservation opportunities. Table 9-1 below shows the current conservation programs offered by the District.

# Demand Management Measures 2020 Urban Water Management Plan Marin Municipal Water District



Table 9-1	
-----------	--

# Current Conservation Programs

Program	Description	Eligible Customer Class	Program Run Dates
CII Water Use Evaluation Program	CII customers are provided on site water use evaluations, recommendations to improve efficiency, and pre-qualified for applicable rebates.	CII	1995 - Current
AMI Leak Notifications Program	Customers with AMI meters receive notifications of water use patterns indicative of leaks. Rather than having to wait until their next water bill, customers are able to receive timely information and stop leaks much faster. Water Efficiency staff runs automated reports to monitor water use of AMI customers. AMI meters record water use in 15- minute intervals as compared to every other month as is typical for most residential meter reads.	SFR, CII, Ag. & Irr.	2018 - Current
HET Rebate Program	2013-Current: Up to \$100 rebate to replace a 3.5 gpf or more per flush or a 1.6 gpf model year 2001 or older with an HET.	SFR	2013 - Current
Landscape Plan Review Program	MMWD has developed water conservation requirements for landscape professionals and homeowners when designing and installing landscapes and irrigation systems. Plan review requirements apply to all new construction and rehabilitated (renovations or changes made to sites with an existing irrigation system) landscape projects requiring a building permit, plan check, or design review.	SFR, CII	1986 - Current
Rain Barrel Rebate Program	2020: Residential and commercial customers with active potable water service can apply for a rebate up to \$0.50 per gallon of storage when they install rain barrels and/or cisterns at their sites. Total rebates for rain barrels and cisterns may not exceed \$1,000 per site.	SFR	2020 - Current

# Demand Management Measures 2020 Urban Water Management Plan



Marin Municipal Water District

Program	Description	Eligible Customer Class	Program Run Dates	
Residential HECW Rebate Program	MMWD customers can apply for a rebate towards the cost of installing a qualifying residential high-efficiency clothes washer that meets current water and energy efficiency requirements.	SFR	2013 – Current	
SFR Water Use Surveys/Audits Program	A free service for SFR customers that involves a visit to their property to review water use and identify ways to save water indoors and outside.	SFR	1995 - Current	
WaterSense Smart Controller Rebate	Residential and commercial customers can apply for a rebate up to \$100 towards the cost of a qualifying EPA WaterSense weather-based irrigation controller.	SFR, CII	2020-Current	
Water Use Surveys/Audits Program (Excluding SFR Participants)	Consultation activities include a review, evaluation and report of indoor plumbing devices and/or a review and report of the landscape irrigation system.	MFR, CII, Ag. & Irr.	1995 - Current	
Water Waste Report Program	The general public can report water waste situations to MMWD online or over the phone. These contacts are logged into a database and followed up on by field staff to research and notify properties about water waste situations.	SFR, CII	1998, 2000 - Current	
Abbreviations:				
Ag. = agricultural		HET = high-efficie	ency toilet	
AIVII = Advanced Met	Lering intrastructure	MER - multi fam	aily residential	
EPA = Environmental	Protection Agency	MMWD = Marin N	Aunicinal Water	
gof = gallons per flue	sh	District		
HECW = high efficienc	y clothes washer	SFR = single family residential		

During the period of 2016-2019, the conservation department:

- completed a total of 2,800 audits/surveys;
- provided program activities for 31,000 students;
- interacted with over 59,000 customers at public outreach events;
- incentivized installation of 990 high efficiency clothes washers, 3,900 high efficiency toilets, and 1,100 irrigation-related devices including high efficiency spray nozzles, rain sensors, drip system components, weather-based irrigation controllers, mulch, rain barrels, greywater system components at residential and commercial sites;



- conducted 800 water waste investigations; and
- responded to over 10,000 customer phone and email inquiries.

The District's 2020 Urban Water Management Plan Water Demand Analysis and Water Conservation Measures Update (**Appendix C**) provides additional detail on implementation of these conservation programs.

## 9.2.5 DMM 5 – Programs to Assess and Manage Distribution System Real Loss

The District performs annual water loss audits in accordance with CWC § 10608.34. These audits are prepared using the American Water Works Association (AWWA) Free Water Audit Software version 5.0 and validated by an AWWA California-Nevada Section-Certified California Water Audit validator prior to submission to DWR. The results of the last five years of water audit data are summarized in Section 4.1.4, and copies of the audit reports are provided in **Appendix K**. The District has an active water loss control program, including primary and secondary response. Primary response is by Utility Systems Specialists on an incident-by-incident basis. Utility Systems Specialists are first responders to all reported leaks, mainline shutdowns for the crews and contractors, consumer calls, and meter turn-on/offs. Secondary to any reported problem, the District has implemented a program to assess and manage system loss through leak survey. Leak survey is handled by Utility System. Using sonic leak detection equipment, technicians are able to locate leaks that may be hidden from view because they have not yet surfaced. Utility System Technicians also assist in pinpointing water leaks. The process of pinpointing water leaks involves the narrowing of a specific leak location. This process assists staff in completing repairs with a minimal amount of excavation.

From 2015 through 2020, the District surveyed 503 miles of pipeline. As a result, the team detected 791 customer service leaks and 214 District service leaks. In addition, the District has an ongoing program (as part of its Capital Improvement Program) to replace aging distribution pipeline; on average, the District replaced 4 miles of pipeline per year over the last five years. Including fire flow improvements, the District replaced approximately 8 miles per year over the last 5 years.

### 9.2.6 DMM 6 – Water Conservation Program Coordination and Staffing Support

The District has full-time staff members within the Water Conservation Department dedicated to implementing and overseeing the conservation program. The primary responsibilities of the Conservation Department include designing, developing, and implementing conservation programs for all MMWD customers. The Conservation Department also provides support to customers, including completing water use surveys and reports, answering any questions about conservation programs, and providing printed materials related to the District's conservation programs. The Conservation Department and its activities are funded by the general fund, which is supported by revenue from water sales.

Contact information for the District's Water Efficiency Manager is listed below:

Name: Carrie Pollard Phone: (415) 945-1522 Email: <u>capollard@marinwater.org</u>



## 9.2.7 DMM 7 – Other DMMs

#### 9.2.7.1 Landscape Plan Review Services

Water conservation staff provide landscape plan review services for all municipal jurisdictions in MMWD's service area at no cost. Title 13 of the District's Code includes a section on water efficient landscaping that requires projects needing a building or landscape permit, plan check or design review. The plan review process consists of reviewing construction documents to verify compliance with efficiency standards, calculating a maximum applied water allowance (MAWA) and estimating total water use (ETWU), conducting a site inspection, and issuing a final letter of approval. On 15 December 2015, the District's Board of Directors approved changes to the landscape plan review requirements to be consistent with the Governor's executive order revising the State of California's Model Water Efficient Landscape plan reviews.

### 9.2.7.2 Other Water Efficiency Codes

Title 13 of the District's Code documents all adopted codes related to water efficiency and conservation, codes. Current versions of these codes can be viewed here: <u>http://www.qcode.us/codes/marin-water/view.php?topic=13</u>.

## 9.3 Planned Implementation to Achieve Water Use Targets

#### ☑ CWC § 10631 (e)

*Provide a description of the supplier's water demand management measures. This description shall include all of the following:* 

(1) (A) ... The narrative shall describe the water demand management measures that the supplier plans to implement to achieve its water use targets pursuant to Section 10608.20.

Beginning in 2023, urban water retailers will be required to report on "annual water use objectives" by November 1 of each year and to achieve these objectives by 1 January 2027 (per CWC § 10609). The annual water use objectives will be calculated based on standards for indoor residential water use, outdoor residential water use, and distribution system water loss. Additionally, it is anticipated that performance-based standards for the commercial, industrial, and institutional sectors, separate from the annual water use objectives, will also be developed by Department of Water Resources (DWR) and implemented in the future. However, the specific standards that will be used to determine a retailer's annual urban water use objectives are currently under development by DWR, and thus, the annual urban water use objectives for the District cannot be calculated or estimated. Therefore, the District intends to continue implementing DMMs both locally and through the SMSWP and will evaluate potential adjustments needed to these programs as the annual water use objective standard methodologies are developed in the coming years. The District's 2020 Urban Water Management Plan Water Demand Analysis and Water Conservation Measures Update, provided in **Appendix C**, evaluates several options for future conservation program implementation.



## **10. PLAN ADOPTION AND SUBMITTAL**

This chapter provides information on a public hearing, the adoption process for the Urban Water Management Plan (UWMP) and Water Shortage Contingency Plan (WSCP), the adopted UWMP and WSCP submittal process, plan implementation, and the process for amending the adopted UWMP or WSCP.

## 10.1 Notification of UWMP Preparation

### ☑ CWC § 10621 (b)

Every urban water supplier required to prepare a plan pursuant to this part shall, at least 60 days before the public hearing on the plan required by Section 10642, notify any city or county within which the supplier provides water supplies that the urban water supplier will be reviewing the plan and considering amendments or changes to the plan. The urban water supplier may consult with, and obtain comments from, any city or county that receives notice pursuant to this subdivision.

Marin Municipal Water District (MMWD or District) sent a letter to relevant entities including Marin County and other local agencies informing them that the District was in the process of updating its UWMP and WSCP and soliciting their input in the update process. A listing of the entities contacted is provided in **Table 2-4**; the notices are included in **Appendix D** for reference. The letter was sent more than 60 days before the public hearing as required by code.

### **10.2** Notification of Public Hearing

#### **☑** *CWC* § 10642

Each urban water supplier shall encourage the active involvement of diverse social, cultural, and economic elements of the population within the service area prior to and during the preparation of both the plan and the water shortage contingency plan. Prior to adopting either, the urban water supplier shall make both the plan and the water shortage contingency plan available for public inspection and shall hold a public hearing or hearings thereon. Prior to any of these hearings, notice of the time and place of the hearing shall be published within the jurisdiction of the publicly owned water supplier pursuant to Section 6066 of the Government Code. The urban water supplier shall provide notice of the time and place of a hearing to any city or county within which the supplier provides water supplies. Notices by a local public agency pursuant to this section shall be provided pursuant to Chapter 17.5 (commencing with Section 7290) of Division 7 of Title 1 of the Government Code. A privately owned water supplier shall provide an equivalent notice within its service area. After the hearing or hearings, the plan or water shortage contingency plan shall be adopted as prepared or as modified after the hearing or hearings.

### 10.2.1 Notice to Cities and Counties

At least two weeks prior to the public hearing, the entities listed in **Table 2-4** were notified that the UWMP and WSCP public hearing would be occurring on 15 July 2021 at 7:30 pm. The letter informed them of the locations the Public Review Draft 2020 UWMP and the updated WSCP would be available for review and welcoming their input and comments on the document. The Public Review Draft 2020 UWMP and the

# Plan Adoption and Submittal 2020 Urban Water Management Plan Marin Municipal Water District



WSCP were available for public review on the District's website. **Table 2-4** lists the cities, counties, and other agencies that were notified. Copies of these letters are provided in **Appendix D**.

#### 10.2.2 Notice to the Public

The District issued public notifications soliciting public input during the preparation of 2020 UWMP and the WSCP. On XX XXX 2021 and XX XXX 2021, the District published a notice in \_\_\_\_\_ informing the public that the 2020 UWMP and WSCP would be available for public review on the District's website, consistent with requirements of California Government Code 6066. The notice also informed the public that the 2020 UWMP and WSCP public hearing would be held virtually on 15 July 2021. A copy of this notice is included in **Appendix E**.

### **10.3** Public Hearing and Adoption

#### ☑ CWC § 10608.26

(a) In complying with this part, an urban retail water supplier shall conduct at least one public hearing to accomplish all of the following:

(1) Allow community input regarding the urban retail water supplier's implementation plan for complying with this part.

(2) Consider the economic impacts of the urban retail water supplier's implementation plan for complying with this part.

(3) Adopt a method, pursuant to subdivision (b) of Section 10608.20, for determining its urban water use target.

Prior to adopting the Plan, the District held a formal public hearing to present information on the 2020 UWMP and WSCP on 15 July 2021 at 7:30 pm, in a virtual meeting.

As part of the public hearing, the District provided the audience with information on compliance with the Senate Bill (SB) X7-7, including its baseline daily per capita water use, water use targets, implementation plan, and 2020 compliance.

This UWMP was adopted by Resolution No. XXXX by the District Board during its XX MONTH 2021 board meeting. The WSCP included as **Appendix H** was adopted by **Resolution No. XXXX** during the same meeting. Copies of the resolutions are included in **Appendix L**.



#### 10.4 Plan Submittal

#### **☑** CWC § 10621

(f) (1) Each urban water supplier shall update and submit its 2020 plan to the department by July 1, 2021.

#### **☑** CWC § 10635 (c)

The urban water supplier shall provide that portion of its urban water management plan prepared pursuant to this article to any city or county within which it provides water supplies no later than 60 days after the submission of its urban water management plan.

#### **CWC** § 10644

(a) (1) An urban water supplier shall submit to the department, the California State Library, and any city or county within which the supplier provides water supplies a copy of its plan no later than 30 days after adoption. Copies of amendments or changes to the plans shall be submitted to the department, the California State Library, and any city or county within which the supplier provides water supplies within 30 days after adoption.

(2) The plan, or amendments to the plan, submitted to the department pursuant to paragraph (1) shall be submitted electronically and shall include any standardized forms, tables, or displays specified by the department.

(b) If an urban water supplier revises its water shortage contingency plan, the supplier shall submit to the department a copy of its water shortage contingency plan prepared pursuant to subdivision (a) of Section 10632 no later than 30 days after adoption, in accordance with protocols for submission and using electronic reporting tools developed by the department.

This UWMP and WSCP were submitted to DWR within 30 days of adoption and by the 1 July 2021 deadline. The submittal was done electronically through Water Use Efficiency Data Portal, an online submittal tool. The adopted Plan was also sent to the California State Library and to the cities and counties listed in **Table 2-4**. Plan Submittal Documentation is included in **Appendix M**.

### 10.5 Public Availability

#### **☑** *CWC* § 10645

(a) Not later than 30 days after filing a copy of its plan with the department, the urban water supplier and the department shall make the plan available for public review during normal business hours.

(b) Not later than 30 days after filing a copy of its water shortage contingency plan with the department, the urban water supplier and the department shall make the plan available for public review during normal business hours.

On or about 15 July 2021, a printed hard copy of the draft 2020 UWMP and WSCP were made available for review during normal business hours at the District's office. Electronic versions were also made available by visiting the District's website (<u>https://www.marinwater.org/water-supply</u>).



## 10.6 Amending an Adopted UWMP or Water Shortage Contingency Plan

#### **CWC** § 10644 (b)

If an urban water supplier revises its water shortage contingency plan, the supplier shall submit to the department a copy of its water shortage contingency plan prepared pursuant to subdivision (a) of Section 10632 no later than 30 days after adoption, in accordance with protocols for submission and using electronic reporting tools developed by the department.

If the Plan is amended, each of the steps for notification, public hearing, adoption and submittal will also be followed for the amended document.



#### 11. REFERENCES

- ABAG, 2018. Association of Bay Area Governments, Plan Bay Area Projections 2040, released on November 2018.
- ABAG, 2020. Association of Bay Area Governments, Regional Housing Needs Allocation Proposed Methodology: San Francisco Bay Area, 2023-2031, released on October 2020: <u>https://abag.ca.gov/sites/default/files/rhna\_methodology\_report\_2023-2031\_finalposting.pdf</u>.
- California Building Standards Commission. CAL Green Code, effective 2014 with supplements effective 2015: <u>http://www.bsc.ca.gov/Home/CALGreen.aspx</u>
- County of Marin, 2014. *Marin Countywide Plan*, adopted 6 November 2007, amended 24 September 2013, reprinted October 2014: <u>https://www.marincounty.org/-/media/files/departments/cd/planning/currentplanning/publications/county-wide-plan/cwp\_2015\_update.pdf</u>.
- DOF, 2020. California Department of Finance Demographic Research Unit, Population Estimates for Cities, Counties, and the State, 2011 2020, with 2010 Benchmark, Report E 4, released on 1 May 2020.
- DWR, 2006. California Groundwater Bulletin 118, dated January 2006.
- DWR, 2015. Department of Water Resources Model Water Efficient Landscape Ordinance (California Code of Regulations, Title 23, Division 2, Chapter 2.7), dated July 9, 2015.
- DWR, 2016a. *Guidebook for Urban Water Suppliers, 2015 Urban Water Management Plan,* dated March 2016.
- DWR, 2016b. Methodologies for Calculating Baseline and Compliance Urban Per Capita Water, California Department of Water Resources Division of Statewide Integrated Water Management Water Use and Efficiency Branch, updated March 2016.
- DWR, 2019. Sustainable Groundwater Management Act 2018 Basin Prioritization, State of California, January 2019.
- EKI, 2020. 2020 Water Demand Analysis and Water Conservation Measure Update, prepared for Marin Municipal Water District, dated December 2020.
- Marin County, 2018. Marin County Multi-Jurisdictional Local Hazard Mitigation Plan, dated 2018.
- MMWD, 2015. *Summary of Landscape Plan Review Changes,* updated on 31 December 2015, provided by MMWD on 13 February 2020.
- MMWD, 2016. Marin Municipal Water District 2015 Urban Water Management Plan, prepared by RMC, dated June 2016.

#### References

#### 2020 Urban Water Management Plan



#### Marin Municipal Water District

- MMWD, 2017. *Water Resources Plan 2040*, Marin Municipal Water District, prepared by RMC, dated March 2017.
- MMWD, 2019. Ordinance No. 442. An Ordinance Amending and Adding Certain Provisions of Title 6, Chapter 6.01 Of The Marin Municipal Water District Code, Entitled "Water Rates And Charges," adopted on 28 May 2019, http://www.gcode.us/codes/marin-water/revisions/442.pdf.
- NMWD, 2016. 2015 Urban Water Management Plan, North Marin Water District, June 2016.
- NOAA, 2021. Map of Sea level trends, NOAA Tides & Currents, data obtained on 10 February 2021, https://tidesandcurrents.noaa.gov/sltrends/.
- Santa Rosa Plain Basin Advisory Panel, 2014. *Santa Rosa Plain Watershed Groundwater Management Plan*, Santa Rosa Plain Basin Advisory Panel, 2014.
- SCWA, 2016. 2015 Urban Water Management Plan, prepared by Brown and Caldwell on June 2016.
- SCWA, 2018. Sonoma County Water Agency Local Hazard Mitigation Plan, dated 16 October 2018.
- SDWIS, 2021. Safe Drinking Water Information System, California Drinking Water Watch Water System Details for the NPS GGRA and North Marin WD Pt. Reyes systems, accessed 16 March 2021. <u>https://sdwis.waterboards.ca.gov/PDWW/JSP/WaterSystemDetail.jsp?tinwsys is number=2740&ti</u> <u>nwsys st code=CA&wsnumber=CA2110350</u> <u>https://sdwis.waterboards.ca.gov/PDWW/JSP/WaterSystemDetail.jsp?tinwsys is number=2728&ti</u> nwsys st code=CA&wsnumber=CA2110006

U.S. Census, 2021. U.S. Census Bureau QuickFacts website, obtained on 16 February 2021, <u>https://www.census.gov/quickfacts/fact/table/US/PST045219</u>



2001 Junipero Serra Blvd., Suite 300 | Daly City, CA 94014 (650) 292-9100 | Fax (650) 552-9012 | Ekiconsult.com Appendices 2020 Urban Water Management Plan Marin Municipal Water District



# Appendix A

# **Completed UWMP Checklist**



Retail	Wholesale	2020 Guidebook Location	Water Code Section	Summary as Applies to UWMP	Subject	2020 UWMP Location
x	x	Chapter 1	10615	A plan shall describe and evaluate sources of supply, reasonable and practical efficient uses, reclamation and demand management activities.	Introduction and Overview	Chapter 1
x	x	Chapter 1	10630.5	Each plan shall include a simple description of the supplier's plan including water availability, future requirements, a strategy for meeting needs, and other pertinent information. Additionally, a supplier may also choose to include a simple description at the beginning of each chapter.	Summary	Section 1.6
x	x	Section 2.2	10620(b)	Every person that becomes an urban water supplier shall adopt an urban water management plan within one year after it has become an urban water supplier.	Plan Preparation	Section 2.1
x	x	Section 2.6	10620(d)(2)	Coordinate the preparation of its plan with other appropriate agencies in the area, including other water suppliers that share a common source, water management agencies, and relevant public agencies, to the extent practicable.	Plan Preparation	Section 2.2.2 and Table 2-4



Retail	Wholesale	2020 Guidebook Location	Water Code Section	Summary as Applies to UWMP	Subject	2020 UWMP Location
x	x	Section 2.6.2	10642	Provide supporting documentation that the water supplier has encouraged active involvement of diverse social, cultural, and economic elements of the population within the service area prior to and during the preparation of the plan and contingency plan.	Plan Preparation	Section 1.3 Section 2.2.3
x		Section 2.6, Section 6.1	10631(h)	Retail suppliers will include documentation that they have provided their wholesale supplier(s) - if any - with water use projections from that source.	System Supplies	Section 2.2.1
	x	Section 2.6	10631(h)	Wholesale suppliers will include documentation that they have provided their urban water suppliers with identification and quantification of the existing and planned sources of water available from the wholesale to the urban supplier during various water year types.	System Supplies	N/A
x	x	Section 3.1	10631(a)	Describe the water supplier service area.	System Description	Chapter 3
x	x	Section 3.3	10631(a)	Describe the climate of the service area of the supplier.	System Description	Section 3.4
x	x	Section 3.4	10631(a)	Provide population projections for 2025, 2030, 2035, 2040 and optionally 2045.	System Description	Section 3.1.1 and Table 3-1
x	x	Section 3.4.2	10631(a)	Describe other social, economic, and demographic factors affecting the supplier's water management planning.	System Description	Section 3.3 and Table 3-3



Retail	Wholesale	2020 Guidebook Location	Water Code Section	Summary as Applies to UWMP	Subject	2020 UWMP Location
x	x	Sections 3.4 and 5.4	10631(a)	Indicate the current population of the service area.	System Description and Baselines and Targets	Sections 3.1 and 5.1, Table 3-1
x	x	Section 3.5	10631(a)	Describe the land uses within the service area.	System Description	Section 3.2
x	x	Section 4.2	10631(d)(1)	Quantify past, current, and projected water use, identifying the uses among water use sectors.	System Water Use	Chapter 4, Tables 4-1 to 4- 4
x	x	Section 4.2.4	10631(d)(3)(C)	Retail suppliers shall provide data to show the distribution loss standards were met.	System Water Use	Section 4.1.4, Table 4-3
x	x	Section 4.2.6	10631(d)(4)(A)	In projected water use, include estimates of water savings from adopted codes, plans, and other policies or laws.	System Water Use	Section 4.2.1, Table 4-4
x	x	Section 4.2.6	10631(d)(4)(B)	Provide citations of codes, standards, ordinances, or plans used to make water use projections.	System Water Use	Section 4.2.4, Table 4-7
x	optional	Section 4.3.2.4	10631(d)(3)(A)	Report the distribution system water loss for each of the 5 years preceding the plan update.	System Water Use	Section 4.1.4, Table 4-3
x	optional	Section 4.4	10631.1(a)	Include projected water use needed for lower income housing projected in the service area of the supplier.	System Water Use	Section 4.2.3, Table 4-6
x	x	Section 4.5	10635(b)	Demands under climate change considerations must be included as part of the drought risk assessment.	System Water Use	Section 4.4



Retail	Wholesale	2020 Guidebook Location	Water Code Section	Summary as Applies to UWMP	Subject	2020 UWMP Location
x		Chapter 5	10608.20(e)	Retail suppliers shall provide baseline daily per capita water use, urban water use target, interim urban water use target, and compliance daily per capita water use, along with the bases for determining those estimates, including references to supporting data.	Baselines and Targets	Chapter 5
x		Chapter 5	10608.24(a)	Retail suppliers shall meet their water use target by December 31, 2020.	Baselines and Targets	Section 5.4, Table 5-5
	x	Section 5.1	10608.36	Wholesale suppliers shall include an assessment of present and proposed future measures, programs, and policies to help their retail water suppliers achieve targeted water use reductions.	Baselines and Targets	N/A
x		Section 5.2	10608.24(d)(2)	If the retail supplier adjusts its compliance GPCD using weather normalization, economic adjustment, or extraordinary events, it shall provide the basis for, and data supporting the adjustment.	Baselines and Targets	Section 5.4, Table 5-5
x		Section 5.5	10608.22	Retail suppliers' per capita daily water use reduction shall be no less than 5 percent of base daily per capita water use of the 5-year baseline. This does not apply if the suppliers base GPCD is at or below 100.	Baselines and Targets	Section 5.3



Retail	Wholesale	2020 Guidebook Location	Water Code Section	Summary as Applies to UWMP	Subject	2020 UWMP Location
x		Section 5.5 and Appendix E	10608.4	Retail suppliers shall report on their compliance in meeting their water use targets. The data shall be reported using a standardized form in the SBX7-7 2020 Compliance Form.	Baselines and Targets	Appendix F
x	x	Sections 6.1 and 6.2	10631(b)(1)	Provide a discussion of anticipated supply availability under a normal, single dry year, and a drought lasting five years, as well as more frequent and severe periods of drought.	System Supplies	Section 7.2
x	x	Sections 6.1	10631(b)(1)	Provide a discussion of anticipated supply availability under a normal, single dry year, and a drought lasting five years, as well as more frequent and severe periods of drought, <i>including changes in supply due to climate</i> <i>change.</i>	System Supplies	Section 6.10.1, Section 7.1.3
x	x	Section 6.1	10631(b)(2)	When multiple sources of water supply are identified, describe the management of each supply in relationship to other identified supplies.	System Supplies	Chapter 6
x	x	Section 6.1.1	10631(b)(3)	Describe measures taken to acquire and develop planned sources of water.	System Supplies	Sections 6.6 to 6.9
x	x	Section 6.2.8	10631(b)	Identify and quantify the existing and planned sources of water available for 2020, 2025, 2030, 2035, 2040 and optionally 2045.	System Supplies	Section 6.9 and Table 6-9


Retail	Wholesale	2020 Guidebook Location	Water Code Section	Summary as Applies to UWMP	Subject	2020 UWMP Location
x	x	Section 6.2	10631(b)	Indicate whether groundwater is an existing or planned source of water available to the supplier.	System Supplies	Section 6.2
x	x	Section 6.2.2	10631(b)(4)(A)	Indicate whether a groundwater sustainability plan or groundwater management plan has been adopted by the water supplier or if there is any other specific authorization for groundwater management. Include a copy of the plan or authorization.	System Supplies	Section 6.2.3
x	x	Section 6.2.2	10631(b)(4)(B)	Describe the groundwater basin.	System Supplies	Section 6.2.1
x	x	Section 6.2.2	10631(b)(4)(B)	Indicate if the basin has been adjudicated and include a copy of the court order or decree and a description of the amount of water the supplier has the legal right to pump.	System Supplies	Section 6.2.1
x	x	Section 6.2.2.1	10631(b)(4)(B)	For unadjudicated basins, indicate whether or not the department has identified the basin as a high or medium priority. Describe efforts by the supplier to coordinate with sustainability or groundwater agencies to achieve sustainable groundwater conditions.	System Supplies	Section 6.2.4



Retail	Wholesale	2020 Guidebook Location	Water Code Section	Summary as Applies to UWMP	Subject	2020 UWMP Location
x	x	Section 6.2.2.4	10631(b)(4)(C)	Provide a detailed description and analysis of the location, amount, and sufficiency of groundwater pumped by the urban water supplier for the past five years	System Supplies	Section 6.2.5 and Table 6-1
x	x	Section 6.2.2	10631(b)(4)(D)	Provide a detailed description and analysis of the amount and location of groundwater that is projected to be pumped.	System Supplies	Section 6.2
x	x	Section 6.2.7	10631(c)	Describe the opportunities for exchanges or transfers of water on a short-term or long-term basis.	System Supplies	Section 6.7
x	x	Section 6.2.5	10633(b)	Describe the quantity of treated wastewater that meets recycled water standards, is being discharged, and is otherwise available for use in a recycled water project.	System Supplies (Recycled Water)	Section 6.5.3 and Tables 6-4 and 6-5
x	x	Section 6.2.5	10633(c)	Describe the recycled water currently being used in the supplier's service area.	System Supplies (Recycled Water)	Section 6.5.3 and Table 6-4
x	x	Section 6.2.5	10633(d)	Describe and quantify the potential uses of recycled water and provide a determination of the technical and economic feasibility of those uses.	System Supplies (Recycled Water)	Section 6.5.3 and Table 6-4

7



Retail	Wholesale	2020 Guidebook Location	Water Code Section	Summary as Applies to UWMP	Subject	2020 UWMP Location
x	x	Section 6.2.5	10633(e)	Describe the projected use of recycled water within the supplier's service area at the end of 5, 10, 15, and 20 years, and a description of the actual use of recycled water in comparison to uses previously projected.	System Supplies (Recycled Water)	Section 6.5.3 and Table 6-5
x	x	Section 6.2.5	10633(f)	Describe the actions which may be taken to encourage the use of recycled water and the projected results of these actions in terms of acre-feet of recycled water used per year.	System Supplies (Recycled Water)	Section 6.5.5
x	x	Section 6.2.5	10633(g)	Provide a plan for optimizing the use of recycled water in the supplier's service area.	System Supplies (Recycled Water)	Section 6.5.5 and Table 6-7
x	x	Section 6.2.6	10631(g)	Describe desalinated water project opportunities for long-term supply.	System Supplies	Section 6.6
x	x	Section 6.2.5	10633(a)	Describe the wastewater collection and treatment systems in the supplier's service area with quantified amount of collection and treatment and the disposal methods.	System Supplies (Recycled Water)	Section 6.5.2 and Table 6-3
x	x	Section 6.2.8, Section 6.3.7	10631(f)	Describe the expected future water supply projects and programs that may be undertaken by the water supplier to address water supply reliability in average, single-dry, and for a period of drought lasting 5 consecutive water years.	System Supplies	Sections 6.9 and 6.10 and Table 6-10



Retail	Wholesale	2020 Guidebook Location	Water Code Section	Summary as Applies to UWMP	Subject	2020 UWMP Location
x	x	Section 6.4 and Appendix O	10631.2(a)	The UWMP must include energy information, as stated in the code, that a supplier can readily obtain.	System Suppliers, Energy Intensity	Section 6.11 and Table 6-11
x	x	Section 7.2	10634	Provide information on the quality of existing sources of water available to the supplier and the manner in which water quality affects water management strategies and supply reliability	Water Supply Reliability Assessment	Chapter 7
x	x	Section 7.2.4	10620(f)	Describe water management tools and options to maximize resources and minimize the need to import water from other regions.	Water Supply Reliability Assessment	Section 7.4
x	x	Section 7.3	10635(a)	Service Reliability Assessment: Assess the water supply reliability during normal, dry, and a drought lasting five consecutive water years by comparing the total water supply sources available to the water supplier with the total projected water use over the next 20 years.	Water Supply Reliability Assessment	Section 7.2
x	x	Section 7.3	10635(b)	Provide a drought risk assessment as part of information considered in developing the demand management measures and water supply projects.	Water Supply Reliability Assessment	Section 7.5



Retail	Wholesale	2020 Guidebook Location	Water Code Section	Summary as Applies to UWMP	Subject	2020 UWMP Location
x	x	Section 7.3	10635(b)(1)	Include a description of the data, methodology, and basis for one or more supply shortage conditions that are necessary to conduct a drought risk assessment for a drought period that lasts 5 consecutive years.	Water Supply Reliability Assessment	Section 7.5.1
x	x	Section 7.3	10635(b)(2)	Include a determination of the reliability of each source of supply under a variety of water shortage conditions.	Water Supply Reliability Assessment	Section 7.5.2 and Table 7-9
x	x	Section 7.3	10635(b)(3)	Include a comparison of the total water supply sources available to the water supplier with the total projected water use for the drought period.	Water Supply Reliability Assessment	Section 7.3 and Tables 7-5 to 7- 7
x	x	Section 7.3	10635(b)(4)	Include considerations of the historical drought hydrology, plausible changes on projected supplies and demands under climate change conditions, anticipated regulatory changes, and other locally applicable criteria.	Water Supply Reliability Assessment	Sections 6.10.1, and 7.1.3
x	x	Chapter 8	10632(a)	Provide a water shortage contingency plan (WSCP) with specified elements below.	Water Shortage Contingency Planning	Chapter 8 and Appendix H
x	x	Chapter 8	10632(a)(1)	Provide the analysis of water supply reliability (from Chapter 7 of Guidebook) in the WSCP	Water Shortage Contingency Planning	Appendix H, Chapter 2



Retail	Wholesale	2020 Guidebook Location	Water Code Section	Summary as Applies to UWMP	Subject	2020 UWMP Location
x	x	Section 8.10	10632(a)(10)	Describe reevaluation and improvement procedures for monitoring and evaluation the water shortage contingency plan to ensure risk tolerance is adequate and appropriate water shortage mitigation strategies are implemented.	Water Shortage Contingency Planning	Appendix H, Chapter 13
x	x	Section 8.2	10632(a)(2)(A)	Provide the written decision- making process and other methods that the supplier will use each year to determine its water reliability.	Water Shortage Contingency Planning	Appendix H, Chapter 4
x	x	Section 8.2	10632(a)(2)(B)	Provide data and methodology to evaluate the supplier's water reliability for the current year and one dry year pursuant to factors in the code.	Water Shortage Contingency Planning	Appendix H, Chapter 4
x	x	Section 8.3	10632(a)(3)(A)	Define six standard water shortage levels of 10, 20, 30, 40, 50 percent shortage and greater than 50 percent shortage. These levels shall be based on supply conditions, including percent reductions in supply, changes in groundwater levels, changes in surface elevation, or other conditions. The shortage levels shall also apply to a catastrophic interruption of supply.	Water Shortage Contingency Planning	Appendix H, Chapter 5
x	x	Section 8.3	10632(a)(3)(B)	Suppliers with an existing water shortage contingency plan that uses different water shortage levels must cross reference their categories with the six standard categories.	Water Shortage Contingency Planning	Appendix H, Chapter 5



Retail	Wholesale	2020 Guidebook Location	Water Code Section	Summary as Applies to UWMP	Subject	2020 UWMP Location
x	x	Section 8.4	10632(a)(4)(A)	Suppliers with water shortage contingency plans that align with the defined shortage levels must specify locally appropriate supply augmentation actions.	Water Shortage Contingency Planning	Appendix H, Section 6.6 and Table 5-2
x	x	Section 8.4	10632(a)(4)(B)	Specify locally appropriate demand reduction actions to adequately respond to shortages.	Water Shortage Contingency Planning	Appendix H, Section 6.1 and Table 5-1
x	x	Section 8.4	10632(a)(4)(C)	Specify locally appropriate operational changes.	Water Shortage Contingency Planning	Appendix H, Section 6.1
x	x	Section 8.4	10632(a)(4)(D)	Specify additional mandatory prohibitions against specific water use practices that are in addition to state-mandated prohibitions are appropriate to local conditions.	Water Shortage Contingency Planning	Appendix H, Section 6.3
x	x	Section 8.4	10632(a)(4)(E)	Estimate the extent to which the gap between supplies and demand will be reduced by implementation of the action.	Water Shortage Contingency Planning	Appendix H, Section 6.1 and Table 5-1
x	x	Section 8.4.6	10632.5	The plan shall include a seismic risk assessment and mitigation plan.	Water Shortage Contingency Plan	Appendix H, Chapter 7
x	x	Section 8.5	10632(a)(5)(A)	Suppliers must describe that they will inform customers, the public and others regarding any current or predicted water shortages.	Water Shortage Contingency Planning	Appendix H, Chapter 8



Retail	Wholesale	2020 Guidebook Location	Water Code Section	Summary as Applies to UWMP	Subject	2020 UWMP Location
x	x	Section 8.5 and 8.6	10632(a)(5)(B) 10632(a)(5)(C)	Suppliers must describe that they will inform customers, the public and others regarding any shortage response actions triggered or anticipated to be triggered and other relevant communications.	Water Shortage Contingency Planning	Appendix H, Chapter 8
x		Section 8.6	10632(a)(6)	Retail supplier must describe how it will ensure compliance with and enforce provisions of the WSCP.	Water Shortage Contingency Planning	Appendix H, Chapter 9
x	x	Section 8.7	10632(a)(7)(A)	Describe the legal authority that empowers the supplier to enforce shortage response actions.	Water Shortage Contingency Planning	Appendix H, Chapter 10
x	x	Section 8.7	10632(a)(7)(B)	Provide a statement that the supplier will declare a water shortage emergency Water Code Chapter 3.	Water Shortage Contingency Planning	Appendix H, Chapter 10
x	x	Section 8.7	10632(a)(7)(C)	Provide a statement that the supplier will coordinate with any city or county within which it provides water for the possible proclamation of a local emergency.	Water Shortage Contingency Planning	Appendix H, Chapter 10
x	x	Section 8.8	10632(a)(8)(A)	Describe the potential revenue reductions and expense increases associated with activated shortage response actions.	Water Shortage Contingency Planning	Appendix H, Chapter 11
x	x	Section 8.8	10632(a)(8)(B)	Provide a description of mitigation actions needed to address revenue reductions and expense increases associated with activated shortage response actions.	Water Shortage Contingency Planning	Appendix H, Chapter 11



Retail	Wholesale	2020 Guidebook Location	Water Code Section	Summary as Applies to UWMP	Subject	2020 UWMP Location
x		Section 8.8	10632(a)(8)(C)	Retail suppliers must describe the cost of compliance with Water Code Chapter 3.3: Excessive Residential Water Use During Drought	Water Shortage Contingency Planning	Appendix H, Chapter 11
x		Section 8.9	10632(a)(9)	Retail suppliers must describe the monitoring and reporting requirements and procedures that ensure appropriate data is collected, tracked, and analyzed for purposes of monitoring customer compliance.	Water Shortage Contingency Planning	Appendix H, Chapter 12
x		Section 8.11	10632(b)	Analyze and define water features that are artificially supplied with water, including ponds, lakes, waterfalls, and fountains, separately from swimming pools and spas.	Water Shortage Contingency Planning	Appendix H, Section 6.5
x	x	Sections 8.12 and 10.4	10635(c)	Provide supporting documentation that Water Shortage Contingency Plan has been, or will be, provided to any city or county within which it provides water, no later than 30 days after the submission of the plan to DWR.	Plan Adoption, Submittal, and Implementation	Appendix H, Chapter 14
x	x	Section 8.14	10632(c)	Make available the Water Shortage Contingency Plan to customers and any city or county where it provides water within 30 after adopted the plan.	Water Shortage Contingency Planning	Appendix H, Chapter 14



Retail	Wholesale	2020 Guidebook Location	Water Code Section	Summary as Applies to UWMP	Subject	2020 UWMP Location
	x	Sections 9.1 and 9.3	10631(e)(2)	Wholesale suppliers shall describe specific demand management measures listed in code, their distribution system asset management program, and supplier assistance program.	Demand Management Measures	N/A
x		Sections 9.2 and 9.3	10631(e)(1)	Retail suppliers shall provide a description of the nature and extent of each demand management measure implemented over the past five years. The description will address specific measures listed in code.	Demand Management Measures	Chapter 9
x		Chapter 10	10608.26(a)	Retail suppliers shall conduct a public hearing to discuss adoption, implementation, and economic impact of water use targets (recommended to discuss compliance).	Plan Adoption, Submittal, and Implementation	Section 10.3
x	x	Section 10.2.1	10621(b)	Notify, at least 60 days prior to the public hearing, any city or county within which the supplier provides water that the urban water supplier will be reviewing the plan and considering amendments or changes to the plan. Reported in Table 10-1.	Plan Adoption, Submittal, and Implementation	Section 10.1
x	x	Section 10.4	10621(f)	Each urban water supplier shall update and submit its 2020 plan to the department by July 1, 2021.	Plan Adoption, Submittal, and Implementation	Section 10.4



Retail	Wholesale	2020 Guidebook Location	Water Code Section	Summary as Applies to UWMP	Subject	2020 UWMP Location
x	x	Sections 10.2.2, 10.3, and 10.5	10642	Provide supporting documentation that the urban water supplier made the plan and contingency plan available for public inspection, published notice of the public hearing, and held a public hearing about the plan and contingency plan.	Plan Adoption, Submittal, and Implementation	Section 10.5
x	x	Section 10.2.2	10642	The water supplier is to provide the time and place of the hearing to any city or county within which the supplier provides water.	Plan Adoption, Submittal, and Implementation	Section 10.2.1
x	x	Section 10.3.2	10642	Provide supporting documentation that the plan and contingency plan has been adopted as prepared or modified.	Plan Adoption, Submittal, and Implementation	Section 10.3
x	x	Section 10.4	10644(a)	Provide supporting documentation that the urban water supplier has submitted this UWMP to the California State Library.	Plan Adoption, Submittal, and Implementation	Section 10.4
x	x	Section 10.4	10644(a)(1)	Provide supporting documentation that the urban water supplier has submitted this UWMP to any city or county within which the supplier provides water no later than 30 days after adoption.	Plan Adoption, Submittal, and Implementation	Section 10.4
x	x	Sections 10.4.1 and 10.4.2	10644(a)(2)	The plan, or amendments to the plan, submitted to the department shall be submitted electronically.	Plan Adoption, Submittal, and Implementation	Section 10.6



Retail	Wholesale	2020 Guidebook Location	Water Code Section	Summary as Applies to UWMP	Subject	2020 UWMP Location
x	x	Section 10.5	10645(a)	Provide supporting documentation that, not later than 30 days after filing a copy of its plan with the department, the supplier has or will make the plan available for public review during normal business hours.	Plan Adoption, Submittal, and Implementation	Section 10.5
x	x	Section 10.5	10645(b)	Provide supporting documentation that, not later than 30 days after filing a copy of its water shortage contingency plan with the department, the supplier has or will make the plan available for public review during normal business hours.	Plan Adoption, Submittal, and Implementation	Section 10.5
x	x	Section 10.6	10621(c)	If supplier is regulated by the Public Utilities Commission, include its plan and contingency plan as part of its general rate case filings.	Plan Adoption, Submittal, and Implementation	N/A
x	x	Section 10.7.2	10644(b)	If revised, submit a copy of the water shortage contingency plan to DWR within 30 days of adoption.	Plan Adoption, Submittal, and Implementation	Section 10.4



# **Appendix B**

Renewal of the Third Amended Offpeak Water Supply Agreement and the Amended Agreement for the Sale of Water Between the Sonoma County Water Agency and Marin Municipal Water District

### <u>Renewal of the Third Amended Offpeak Water Supply Agreement and the</u> <u>Amended Agreement for the Sale of Water between the Sonoma County Water</u> <u>Agency and Marin Municipal Water District</u>

٦

This Renewal of the Third Amended Offpeak Water Supply Agreement and the Amended Agreement for the Sale of Water between the Sonoma County Water Agency and Marin Municipal Water District ("Renewal Agreement") is made by and between the Marin Municipal Water District ("MMWD") and the Sonoma County Water Agency ("Agency") and is effective as of July 1, 2015.

#### <u>Recitals</u>

A. Agency and MMWD are parties to a Supplemental Water Supply Agreement dated January 25, 1996, which incorporates two agreements between the Agency and MMWD that are attached as exhibits to the Supplemental Water Supply Agreement, the Third Amended Offpeak Water Supply Agreement (Exhibit A, referred to herein as the "Offpeak Agreement") and the Amended Agreement for the Sale of Water between the Sonoma County Water Agency and Marin Municipal Water District (Exhibit B, referred to herein as the "Water Sale Agreement").

B. The Offpeak Agreement and the Water Sale Agreement by their terms remain in effect until June 30, 2015 as provided in the Amendment To and Temporary Extension of the Supplemental Water Supply Agreement between MMWD and the Agency dated June 27, 2014.

C. MMWD has requested that the Offpeak Agreement and the Water Sale Agreement be renewed pursuant to Section 7 of the Offpeak Agreement and Section 18 of the Water Supply Agreement.

D. In September 2008, the National Marine Fisheries Service issued its *Russian River Biological Opinion*, which requires the Agency to implement certain activities and projects in the Russian River watershed to improve habitat for listed salmonid species as a condition to receiving "incidental take" authorization in connection with its operation of the Russian River Project and the Water transmission system. MMWD also has listed salmonid species present in streams within its water service area. The protection and recovery of listed salmonid species is a high priority for the Agency and for MMWD.

E. The Supplemental Water Supply Agreement recognized that one of the reasons for that agreement was "to accommodate the Agency's efforts to attempt to ensure a continuation of Pacific Gas and Electric Company's historic diversions of Eel

River water to the Russian River and to increase the reliability of the water supply deliverable to [MMWD]." The need to ensure the continuation of such flows was heightened by the amendments made by the Federal Energy Regulatory Commission in 2004 to Pacific Gas & Electric Company's license for the Potter Valley Project, which reduced such flows. By entering into this Renewal Agreement, the parties recognize the ongoing importance of the continuation of such flows.

-

#### **Agreement**

Wherefore, in consideration of the foregoing and the mutual promises set forth herein, Agency and MMWD agree as follows:

1. This Renewal Agreement shall be effective on July 1, 2015. On the effective date of this Renewal Agreement, the Offpeak Agreement shall be amended as indicated in the attached Exhibit A and the Water Sale Agreement shall be amended as indicated in the attached Exhibit B.

2. Pursuant to section 15051(d) of the CEQA Guidelines, MMWD is designated as the lead agency under the California Environmental Quality Act for the execution of this Renewal Agreement, and for any projects south of Kastania Reservoir that MMWD may construct to implement this Renewal Agreement. Agency is designated as the lead agency for any projects north of Kastania Reservoir that Agency may construct to implement this Renewal Agreement.

3. MMWD shall indemnify Agency and its officers, agents and employees for any and all liabilities, claims, demands, damages, losses, disabilities and expenses (including attorney fees and litigation costs) of every nature arising out of, or in connection with, any legal proceeding that is commenced against the Agency and that alleges that Agency's execution of this Renewal Agreement violates the California Environmental Quality Act or the CEQA Guidelines. MMWD shall provide such indemnification whether or not such liabilities, claims, demands, damages, losses, disabilities or expenses are based on Agency's negligence, unless such liabilities, claims, demands, damages, losses, disabilities or expenses are based on Agency's sole negligence, willful misconduct or violation of law. The provisions in this section regarding attorney fees shall not apply to any other section of this Renewal Agreement.

4. This writing is intended both as the final expression of the agreement between the parties hereto with respect to the included terms and conditions and as a complete and exclusive statement of the terms of the Renewal Agreement. Pursuant to Code of Civil Procedure section 1856, no modification of the Renewal Agreement shall be effective unless and until such modification is evidenced by a writing signed by all parties. 5. No third party beneficiaries are intended or established by this Renewal Agreement.

6. Agency and District each was represented by independent counsel in the negotiation and execution of this Renewal Agreement. For the purposes of interpretation of this Renewal Agreement, neither party shall be deemed to have been the drafter of this Renewal Agreement.

ATTEST:

MARIN MUNICIPAL WATER DISTRICT

10-7 Secretarv

By:

John C. Gibson, President Board of Directors

Reviewed As To Substance By Marin Municipal Water District\_

Krishna Kumar General Manager

Deputy Clerk of the Board of Directors Reviewed As To Form:

nou Mary R. Casey

General Counsel

ATTEST:

By:

SONOMA COUNTY WATER AGENCY

Chair, Board of Directors

DATE

o Substance By Reviewed County Water Agency: Sonoma

Grant Davis, General Manager

Reviewed As To Form:

AurShupe

Steven S. Shupe Deputy County Counsel

## Exhibit A

### Amendments To Third Amended Offpeak Water Supply Agreement

#### FOURTH AMENDED

#### OFFPEAK WATER SUPPLY AGREEMENT

This Fourth Amended Offpeak Water Supply Agreement ("Offpeak Agreement") is made and entered into effective this July 1, 2015, by and between the Marin Municipal Water District, a public body, hereinafter called "Marin" and the Sonoma County Water Agency, a public body, hereinafter called "Agency".

#### WITNESSETH:

WHEREAS, Agency entered into contracts with the United States Government to secure the construction of Coyote Valley Dam on the East Fork Russian River and Warm Springs Dam on Dry Creek (hereinafter "Russian River Project"); and

WHEREAS, Agency entered into a contract with the cities of Cotati, Petaluma, Rohnert Park, Santa Rosa, and Sonoma, the Forestville County Water District, the Valley of the Moon Water District, and the North Marin Water District (hereinafter "North Marin") entitled, Agreement for Water Supply and Construction of Russian River-Cotati Intertie Project, dated October 25, 1974, and last amended June 23, 2006 (hereinafter "Restructured Agreement for Water Supply").

WHEREAS, said Restructured Agreement for Water Supply authorizes the use of certain water transmission facilities (hereinafter

- Exh. A, page 1 -

"Transmission System") for the purpose of delivering to Marin, under prescribed terms and conditions, water which is available in the Russian River; and

WHEREAS, Agency entered into a contract with Marin entitled Second Amended Offpeak Water Supply Agreement, dated May 3, 1988, which provides for the delivery to Marin by Agency, under prescribed terms and conditions, of not to exceed 4,300 acre feet of Russian River water per fiscal year; and

WHEREAS, parts of the property taxes paid since 1971 by the taxpayers of Sonoma County have been paid into a sinking fund that was established to fund Agency's payment obligations to the United States Government for the Warm Springs Dam Project and parts of the property taxes that will be paid in the future by the taxpayers of Sonoma County will likewise be paid into this sinking fund; and WHEREAS pursuant to the Second Amended Offpeak Water Supply Agreement Marin has paid Agency a Russian River Conservation Charge, the proceeds of which were in the past paid into the sinking fund described in the previous recital; and

WHEREAS Agency entered into a contract with Marin dated October 22, 1991 and entitled Agreement for the Sale of Water between the Sonoma County Water Agency and the Marin Municipal Water District (now referred to as the "Water Sale Agreement"), which provides for the delivery to Marin by Agency, under prescribed terms and conditions, of up to 10,000 acre-feet of water per fiscal year; and

WHEREAS, Agency and Marin entered into a Supplemental Water Supply Agreement dated January 25, 1996, which amended and

incorporated two agreements between the Agency and Marin that are attached as exhibits to the Supplemental Water Supply Agreement, those agreements being known as the Third Amended Offpeak Water Supply Agreement and the Amended Agreement for the Sale of Water between the Sonoma County Water Agency and Marin Municipal Water District, to accommodate the Agency's efforts to attempt to ensure a continuation of the Pacific Gas and Electric Company's historic diversions of Eel River water to the Russian River and to increase the reliability of the water supply deliverable to Marin; and

WHEREAS, the Third Amended Offpeak Water Supply Agreement by its terms remain in effect until June 30, 2015 as provided in the Amendment To and Temporary Extension of the Supplemental Water Supply Agreement between Marin and the Agency dated June 27, 2014; and

WHEREAS, Marin has requested that the Third Amended Offpeak Water Supply Agreement be renewed.

NOW, THEREFORE, the parties hereto agree as follows:

1. Subject to all the terms and conditions of this Offpeak Agreement and the relevant provisions of the Restructured Agreement for Water Supply, Agency shall deliver water to Marin either through separately metered turnout at Agency's Kastania Reservoir or through North Marin's meter(s) at Marin's option. Should Marin desire a separate metered turnout, it shall pay to Agency the actual cost of installation of such metered turnout. Such water shall be made available to Marin in an amount not to exceed 4,300 acre feet per fiscal year and at delivery rates as requested by Marin but not to exceed 760 acre feet per calendar month. During the five month period May 1 through September 30 the delivery rate shall not exceed 360 acre feet per calendar month without the prior written consent of the Agency and the total quantity of water delivered shall not exceed 1,800 acre feet. "Fiscal year" means the year beginning on July 1 and ending on the following June 30.

2. Marin shall pay for water delivered pursuant to this Offpeak Agreement at a rate equal to the highest rate per acre-foot then charged by the Agency to any party to the Restructured Agreement for Water Supply (or pursuant to any amendment or successor agreement to the Restructured Agreement for Water Supply) for water taken from either the Petaluma Aqueduct or the Santa Rosa Aqueduct, multiplied by 1.11; provided, however, that the 20% surcharge imposed on the Town of Windsor under Section 4.17(a) of the Restructured Agreement shall not be included in determining the highest rate per acre-foot for water taken from the Santa Rosa Agueduct or Petaluma Agueduct. Seven and four hundred thirty-two one-thousandths percent (7.432%) of this per-acre-foot charge shall be placed in the Russian River Projects Fund; two and four hundred seventy-seven thousandths percent (2.477%) of this per-acre-foot charge shall be used, at the discretion of the Agency, to pay for the costs of Common Facilities, to pay the Capital Costs of Aqueduct Facilities relating to the Santa Rosa or Petaluma Aqueducts, or to pay operations and maintenance costs; and the remainder shall be applied pursuant to the Restructured Agreement for Water Supply (or pursuant to any amendment or successor agreement to the Restructured Agreement for Water Supply), with the water delivered to Marin considered to be delivered from the Petaluma Aqueduct. Agency shall bill Marin monthly for the quantity of water

- Exh. A, page 4 -

measured from Marin's metered turnout or from the accounting provided by North Marin and Marin shall pay the amount of such billing within 30 days after receipt of the bill. Notwithstanding any dispute between Agency and Marin, Marin shall pay all its bills when due and shall not withhold all or any part of any payment pending the final resolution of such dispute. In the event of a dispute, Marin may pay its bills under protest and if the resolution of the dispute results in a refund to Marin, Agency shall make refund plus any interest earned by investment of the disputed funds.

3. If the total amount of water delivered to Marin pursuant to this Offpeak Agreement is less than 4,300 acre-feet in any fiscal year, then Agency shall include in the bill for the month of July the difference between the actual amount of water delivered during the previous fiscal year and 4,300 acre-feet, and Marin shall pay for such amount at the same rate as though such water had been delivered, if and to the extent that such water was available to Marin. Water shall be deemed to have been available to Marin during the previous fiscal year if Marin could have taken delivery of such water at any time during such fiscal year at delivery rates not exceeding the rates specified in Section 1 of this Offpeak Agreement. Agency shall keep and make available for review by Marin operating records indicating the availability of water to Marin.

4. Except as otherwise provided herein, Agency shall release water from storage in Lake Mendocino or Lake Sonoma when necessary to make available in the Russian River sufficient water to make the deliveries provided for herein. If by reason of drought or other physical cause or legal impediment beyond the control of the Agency, a shortage in the water

- Exh. A, page 5 -

available from the Russian River or the Russian River Project occurs, then Agency shall not be liable to Marin for any damage resulting therefrom. In the event of shortage of water in the Russian River or the Russian River Project that requires the Agency to apportion available water, Agency shall apportion the available water so that it may make deliveries as follows:

ж

First, Agency shall deliver to each of its regular customers, not in excess of the respective entitlements set forth in Sections 3.1, 3.2 and 3.3 of the Restructured Agreement for Water Supply or any amendments or successor agreements thereto, authorize Agency's Russian River customers to divert or redivert not in excess of the amounts for which those customers have contracted to purchase from the Agency, and deliver to Marin not in excess of the amounts set forth in Section 1 of this Offpeak Agreement and paragraph (f) of Section 8 of the Water Sale Agreement, the quantities of water required by each for human consumption, sanitation and fire protection, as determined by the Agency after taking into consideration all other sources of potable water then available to said customer;

Second, to the extent additional water is available to the Agency, Agency shall deliver such water to Agency's regular customers, authorize Agency's Russian River customers, to divert or redivert such water and deliver such water to Marin in proportion to the respective entitlements set forth in Sections 3.1, 3.2 and 3.3 of the Restructured Agreement for Water Supply, or any amendments or successor agreements, thereto, the agreements between the Agency and its Russian River customers, and Section 1 of this Offpeak Agreement and paragraph (f) of Section 8 of the Water Sale Agreement, provided, however, that no customer shall receive under subdivisions "First" and "Second" hereof a total quantity of water in excess of its reasonable requirements or its said entitlement or contracted amount, whichever is less.

Agency shall make all reasonable efforts to deliver water to Marin at Kastania Reservoir or at North Marin's meter, as specified in Section 1 of this Offpeak Agreement, using all facilities available to Agency However, Marin acknowledges that priorities of use of delivery capacity in Agency's facilities have been allocated to others who are not parties to this Offpeak Agreement, and that capacity to make deliveries of water to Marin at the rates in Section 1 of this Offpeak Agreement may not always be available to Agency. Agency shall keep, and make available to Marin on an annual basis in July of each year, records indicating the availability of capacity to deliver water to Marin, up to the limits set forth in Section 1 of this Offpeak Agreement and in Section 2 of the Water Sale Agreement, for each month of the previous fiscal year. During July of each year Agency also shall provide Marin with an estimate of the capacity that will be available to deliver water to Marin pursuant to this Offpeak Agreement and the Water Sale Agreement during each remaining month of the then present fiscal year. If these reports indicate that delivery capacity is insufficient, or soon will be insufficient, to provide deliveries at the rates specified in Section 1 of this Offpeak Agreement, then Agency shall use its best efforts to secure an alternative means, solely at Marin's expense, to accomplish the purposes of this Offpeak Agreement.

In the event of an impairment of or limitation on the use or capacity of the Transmission System, or other facility that affects the Agency's ability to deliver water to Marin pursuant to this Offpeak Agreement, by reason of natural disaster, sabotage, legal impediment or other cause beyond the control of the Agency, the Agency shall not be liable to Marin for any damage

ţ

- Exh. A, page 7 -

arising therefrom. In such event, the Agency shall use the available Transmission System capacity to make deliveries as follows:

. .

First, deliver to each of its regular customers the quantity of water, not in excess of its respective entitlement set forth in Sections 3.1 and 3.2 of the Restructured Agreement for Water Supply, or any amendments or successor agreements thereto, required by it for human consumption, sanitation and fire protection as determined by the Agency after taking into consideration all other sources of potable water then available to said customer;

Second, to the extent additional Transmission System capacity is available to the Agency, deliver a quantity of water to the regular customers in proportion to their respective entitlements set forth in Section 3.1 and 3.2 of the Restructured Agreement for Water Supply, or any amendments or successor agreements thereto, provided, however, that no regular customer shall receive under the paragraphs "first" and "second" a total quantity of water in excess of its reasonable requirements or its said entitlement, whichever is less;

Third, to the extent additional Transmission System capacity is available, deliver water to regular customers in excess of their entitlements, pursuant to section 3.3 of the Restructured Agreement for Water Supply;

Fourth, to the extent additional Transmission System capacity is available, deliver water to Marin Municipal Water District, not in excess of the delivery limitations in Section 1 of this Offpeak Agreement;

Fifth, to the extent additional Transmission System capacity is available, deliver surplus water to the water contractors;

Sixth, to the extent additional Transmission System capacity is available, delivery surplus water to other Agency customers;

- Exh. A, page 8 -

The terms "regular customers", "Russian River customers" and "water contractors" in this Offpeak Agreement shall have the same meanings that they have in the Restructured Agreement for Water Supply.

5. In addition to the rates and charges required to be paid by Marin pursuant to Section 2 hereof, Marin shall pay the following additional charges on or before each September 30 during the term of this Offpeak Agreement.

a. Marin shall pay a Russian River Conservation Charge in lieu of the property taxes levied by the Agency on property in Sonoma County, to pay the capital, operation and maintenance costs associated with the Warm Springs Dam Project. The Russian River Conservation Charge shall be a charge per acre foot which shall be applied to Marin's total 4,300 acre feet per annum entitlement under this Offpeak Agreement. The charge shall be determined annually on or before April 30 preceding each September 30. The Russian River Conservation Charge shall be determined by multiplying the tax rate levied by the Agency in the then current fiscal year to pay the costs associated with the Warm Springs Dam Project times the total assessed value of secured and unsecured property situated within the cities of Cotati, Petaluma, Rohnert Park, Santa Rosa and Sonoma, the Forestville County Water District, and the Valley of the Moon Water District, and dividing the product by the total number of acre feet of water delivered to said public agencies pursuant to Section 3.1 and 3.3 of the Restructured Agreement for Water Supply, or any amendments or successor agreements thereto, during the twelve month period ending on March 31.

b. Marin shall pay a Russian River Projects Charge in lieu of the property taxes levied on property in Sonoma County and other Agency

- Exh. A, page 9 -

general fund monies which are transferred by the Agency to the Agency's Russian River Projects Fund and expended to pay for or partially pay for: (1) carrying out the Agency's Coyote Valley Dam Project and Warm Springs Dam Project channel-stabilization works obligations to the United States Government and the State of California under Agency Board of Directors Resolutions No. 6847 adopted May 24, 1955, No. 7798 adopted September 27, 1955, No. DR00793-1 adopted September 25, 1961 and Resolution No DR68485 adopted December 23, 1980; (2) securing and defending appropriative water rights which are necessary for the realization of the full benefits of the Coyote Valley Dam and Warm Springs Dam Projects; (3) the Agency's share of the United States Government's investment, operation and maintenance costs associated with the Coyote Valley Dam and Warm Springs Dam Projects; (4) the acquisition of all or part of the Potter Valley Project, or contributions made to the Project owner to insure the continued operation of all or part of the Project; and (5) fishery mitigation and enhancement projects undertaken by the Agency in the Russian River and Eel River and their tributaries. The Russian River Projects Charge shall be determined by dividing the total amount of Agency monies expended from the Agency's Russian River Projects Fund in the preceding ten fiscal years, exclusive of the funds contributed to the Fund by Marin and North Marin Water Districts and interest earnings attributable to funds contributed by Marin and North Marin Water District, by the sum of the total acre-feet of water delivered by the Agency to the cities of Cotati, Petaluma, Rohnert Park, Santa Rosa, and Sonoma, the Forestville County Water District, and the Valley of the Moon Water District pursuant to Section 3.1 and 3.3 of the Restructured Agreement for Water Supply, or any amendments or successor agreements thereto, during the preceding ten

- Exh. A, page 10 -

fiscal years and multiplying the quotient by the ratio that the assessed value of secured and unsecured property situated within the cities of Cotati, Petaluma, Rohnert Park, Santa Rosa and Sonoma, the Forestville County Water District, and the Valley of the Moon Water District bears to the assessed value of all secured and unsecured property within Sonoma County, provided, however, in no event shall the Russian River Projects Charge exceed \$20.00 per acre-foot. Agency shall keep proper books, records and accounts in which complete and accurate entries shall be made of all Agency general fund monies transferred to the Agency's Russian River Projects Fund and all expenditures made from the fund for the purposes described in this paragraph. The Agency shall maintain a separate account within the Russian River Projects Fund for monies contributed by Marin and North Marin Water District. Monies expended from the Russian River Projects Fund shall be deemed to have been expended from the Marin and North Marin Water District account in the proportion that the balance of the account bears to the total Russian River Projects Fund balance at the end of the fiscal year quarter preceding the expenditure.

c. All money received by the Agency from the Russian River Conservation Charge and the Russian River Projects Charge on water sold to Marin shall be credited to the Agency's Russian River Projects Fund and shall be used only for the purposes set forth in Paragraph (b) of this section.

6. The annual quantity of water provided to Marin by Agency pursuant to this Offpeak Agreement shall be in addition to the annual quantity of water provided to Marin by Agency pursuant to the Sale of Water Agreement.

7. This Offpeak Agreement supersedes the Third Amended Offpeak Water Supply Agreement between Marin and Agency. This Offpeak Agreement shall be effective on July 1, 2015 and shall remain in effect until June 30, 2025. Upon the request of Marin, Agency agrees to enter into renewal agreements for periods not to exceed the then-existing term of the Restructured Agreement for Water Supply or any renewals, amendments or replacement agreements to the Restructured Agreement for Water Supply upon the same terms and conditions contained in this Offpeak Agreement, except that the Agency may make reasonable adjustments to the charges under Section 5 of this Offpeak Agreement, and any such reasonable adjustments then shall be included in any renewal agreement. If the Restructured Agreement for Water Supply has been terminated or has expired without being renewed, amended or replaced by another agreement pertaining to water supply, then upon the request of Marin, the Agency shall enter into renewal agreements for periods not to exceed forty (40) years upon the same terms and conditions contained in this Offpeak Agreement, except that the Agency may make reasonable adjustments to the charges under Section 5 of this Offpeak Agreement and any such reasonable adjustment then shall be included in any renewal agreement.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed as of the date hereinabove first written.

ATTEST:

Secretarv

MARIN MUNICIPAL WATER DISTRICT

By:

Board of Directors

- Exh. A, page 12 -

Reviewed As To Substance By Marin Municipal Water District

Krishna Kumar **General Manager** 

Reviewed As To Form:

nay axil Many R. Casey General Counsel

ATTEST:

By: Deputy Clerk of the

Board of Directors

SONOMA COUNTY WATER AGENCY

Chair.

Board of Directors

15 DATE:

Reviewed As To Substance By Sonoma County Water Agency:

Grant Davis, General Manager

Reviewed As To Form:

AurShupe

Steven S. Shupe Deputy County Counsel

- Exh. A, page 13 -

## Exhibit B

سو کا ا

### Amendments to Amended Agreement for the Sale of Water between the Sonoma County Water Agency and Marin Municipal Water District

# SECOND AMENDED AGREEMENT FOR THE SALE OF WATER BETWEEN THE SONOMA COUNTY WATER AGENCY AND THE MARIN MUNICIPAL WATER DISTRICT

This agreement is made this 22nd day of October, 1991, and is amended and effective this July 1, 2015, by and between the Sonoma County Water Agency, hereinafter called "Agency," and the Marin Municipal Water District, hereinafter called "District" hereinafter ("Water Sale Agreement.")

## **RECITALS**

A. District operates a municipal water system to supply water to customers within its boundaries. District's water system is supplied by water from reservoirs owned and operated by District, and by water purchased from Agency pursuant to Fourth Amended Offpeak Water Supply Agreement ("Offpeak Agreement") and this Water Sale Agreement.

B. District and Agency now wish to revise this Water Sale Agreement to accommodate the Agency's effort to attempt to ensure a continuation of the Pacific Gas and Electric Company's historic diversions of Eel River water to the Russian River and to increase the reliability of the water supply deliverable to District. C. The Coyote Valley Project was authorized by the Flood Control Act of 1950 (Pub. L. No. 81-516), and was completed by the U. S. Army Corps of Engineers in 1958. This project includes Lake Mendocino, which has a capacity of 122,500 acre-feet, of which 70,000 acre-feet is allocated to storage for water supply. K ( )

D. In 1955, Agency (then called the "Sonoma County Flood Control and Water Conservation District") sold general obligation bonds to raise \$5,650,000, which it then paid to the United States for the reimbursable costs of the water-storage element of the Coyote Valley Project. Parts of the property taxes paid since 1955 by the taxpayers of Sonoma County have been used to make payments on these bonds.

E. The Warm Springs Dam Project was authorized by the Flood Control Act of 1962 (Pub. L. No. 87-874), and was completed by the United States Army Corps of Engineers in 1984. This project includes Lake Sonoma, which has a capacity of 381,000 acre-feet, of which 212,000 acre-feet is allocated to storage for water supply.

F. Agency contracted with the United States to make annual payments to the United States for portions of the construction, operation, maintenance, major-replacement and major-rehabilitation costs of the Warm Springs Dam Project. Parts of the property taxes paid since 1971 by the taxpayers of Sonoma County have been paid into a sinking fund that was established to fund Agency's payment obligations to the United States for this project. Parts of the property taxes that will be paid in the future by the taxpayers of Sonoma County will likewise be paid into this sinking fund.

G. Pursuant to the Restructured Agreement for Water Supply, the North Marin Water District has paid Agency a Russian River Conservation Charge. Pursuant to the Offpeak Water Supply Agreement and its two amendments, District also has paid Agency a Russian River Conservation Charge. The proceeds of these payments were in the past paid into the sinking fund described in the preceding paragraph.

يو دي ا

H. Agency currently utilizes the Transmission System to supply water to the Water Contractors and some of Agency's Other Customers and Contractors pursuant to the Restructured Agreement for Water Supply and other agreements. Payments made by the Water Contractors pursuant to the Restructured Agreement for Water Supply were and are being used to fund payments on the bonds sold to finance the construction of parts of the Transmission System and to fund directly the construction of other parts of the Transmission System.

I. Some of Agency's Other Customers and Contractors divert water purchased from Agency directly from the Russian River.

J. Agency holds Permits 12947A, 12949, 12950 and 16596, which were issued by the State Water Resources Control Board and its predecessors pursuant to Applications 12919A, 12920A, 15736, 15737 and 19351. These permits authorize Agency to divert Russian River water, and to redivert water previously stored in Lake Mendocino and Lake Sonoma, subject to specified terms and conditions.

K. On July 3, 1975, Agency and District entered into an agreement entitled, "Offpeak Water Supply Agreement." Agency and

- Exh. B, page 3 -

District amended that agreement on August 28, 1984, May 3, 1988, January 25, 1995, and June 27, 2014. That agreement, now entitled "Fourth - Amended Offpeak Water Supply Agreement" and referred to herein as the "Offpeak Agreement" provides for Agency to deliver up to 4,300 acre-feet of water per year pursuant to Permits 12947A, 12949, 12950 and 16596, subject to specified terms and conditions. L. In June 1980, Agency certified its final environmental impact report on Proposed Amendments of Permits on Applications 12919A, 15736, 15737 and 19351. In July 1984, Agency certified its supplemental environmental impact report covering Proposed Coordinated Use of the Water Supply of Lake Mendocino and Lake Sonoma, Russian River Project. On November 14, 1990, District certified its Water Supply Plan Program Final Environmental Impact Report. On September 24, 1991, District certified its Water Supply Project Final Environmental Impact Report. These reports together satisfy the environmental-impact-report requirements of the California Environmental Quality Act for this Water Sale Agreement.

M. Agency's January 1991 "Urban Water Management Plan" concluded that Agency's direct-diversion rights and its share of the Coyote Valley and Warm Springs Dam Projects together will have yield sufficient to supply District with additional water according to the terms of this Water Sale Agreement.

N. Agency has the capacity to produce water from groundwater wells and intends, if authorized by an amendment to the Restructured Agreement for Water Supply, to develop additional emergency groundwater wells, aquifer storage and recovery wells or other offstream

- Exh. B, page 4 -

water-production facilities, to improve the reliability of the delivery capacity of the Transmission System.

•

O. At the present time and under normal circumstances, the Transmission System has excess capacity that may be used by Agency to supply District with water pursuant to this Water Sale Agreement. In order for the Transmission System in the future to continue to have such excess capacity under normal circumstances, Agency will need to construct elements of a new aqueduct from the Russian River to the Ely Pumping Plant, roughly paralleling portions of the existing Cotati Intertie and Petaluma Aqueduct, at an earlier date than would be necessary absent this Water Sale Agreement, and Agency will need to construct a new aqueduct from the Ely Pumping Plant to Kastania Reservoir, roughly paralleling portions of the existing Petaluma Aqueduct, which would not be necessary absent this Water Sale Agreement.

P. Agency and District entered into a Supplemental Water Supply Agreement dated January 25, 1996, which amended and incorporated two agreements between the Agency and MMWD that are attached as exhibits to the Supplemental Water Supply Agreement, the Third Amended Offpeak Water Supply Agreement and the Amended Agreement for the Sale of Water between the Sonoma County Water Agency and Marin Municipal Water District. The Amended Agreement for the Sale of Water by its terms remain in effect until June 30, 2015 as provided in the Amendment To and Temporary Extension of the Supplemental Water Supply Agreement between MMWD and the Agency dated June 27, 2014.

Q. District has requested that the Third Amended Offpeak Water Supply Agreement be renewed. Agency is willing to sell water to District, and District is willing to purchase such water from Agency, pursuant to the terms and conditions of this Water Sale Agreement.

. .

NOW, THEREFORE, in consideration of these recitals and the mutual promises made herein, Agency and District agree as follows:

Section 1. Definitions.

When used in this Water Sale Agreement, unless otherwise distinctly expressed or manifestly incompatible with the intent of this Water Sale Agreement, the following terms shall have the following meanings:

a. "Restructured Agreement for Water Supply" means the agreement between Agency and the Water Contractors entitled <u>Restructured Agreement for Water Supply</u>, executed on June 25, 2006.

b. "Fiscal Year" means each year that begins on July 1 and ends on the following June 30.

c. "Maximum Delivery Limit" means the maximum amount of water that Agency is required to deliver and that District may take pursuant to this Water Sale Agreement in any Fiscal Year, and which is described in Section 4 of this Water Sale Agreement.

d. "Other Customers and Contractors" means the North Marin Water District and all present and future Agency customers within Sonoma County, including all entities within Sonoma County that receive or will receive water from the Transmission System or other water conveyance facilities, and all entities within Sonoma County that divert or will divert water purchased from Agency directly from the Russian River or Dry Creek.

e. "Permits 12947A, 12949, 12950 and 16596" mean water rights permits 12947A, 12949, 12950 and 16596 which the State Water Resources Control Board and its predecessors issued to Agency pursuant to Applications 12919A and 12920A, 15736, 15737 and 19351, as such permits now exist or in the future may exist (including any licenses that may be issued to replace these permits).

2 g

f. "Prudent Storage Reserve" means a quantity of water in storage in Lake Sonoma, Lake Mendocino and Lake Pillsbury sufficient to assure both that the level of Lake Sonoma would not drop below 292 feet and the level of Lake Mendocino would not drop below 685 feet, with reference to the National Geodetic Vertical Datum of 1929, during the fall of the calendar year following the calendar year in which the determination of availability is made under Section 7 of this Water Sale Agreement, if the most severe hydrologic period of record were to recur.

g. "Russian River Conservation Charge" means the charge that District pays and will pay Agency in lieu of the property taxes levied by the Agency on property in Sonoma County to pay the capital, operation and maintenance costs associated with the Warm Springs Dam Project, and which is described In Section 10 of this Water Sale Agreement.

"Russian River Projects Charge" means the charge which is h. described in Section 10 of this Water Sale Agreement that the District will pay Agency in lieu of the property taxes levied on property in Sonoma County and other Agency general fund monies which are transferred by the Agency to the Agency's Russian River Projects Fund and expended to pay for or partially pay for: (1) carrying out the Agency's Coyote Valley Dam Project and Warm Springs Dam Project channel-stabilization works obligations to the United States Government and the State of California under Agency Board of Directors Resolutions No. 6847 adopted May 24, 1955, No. 7798 adopted September 27, 1955, 1961 Resolution adopted September 25, and No. DR00793-1
No. DR68485 adopted December 23, 1980; (2) securing and defending appropriative water rights which are necessary for the realization of the full benefits of the Coyote Valley Dam and Warm Springs Dam Projects; (3) the Agency's share of the United States Government's investment, operation and maintenance costs associated with the Coyote Valley Dam and Warm Springs Dam Projects; (4) the acquisition of all or part of the Potter Valley Project, or contributions made to the Project owner to insure the continued operation of all or part of the Project; and (5) fishery mitigation and enhancement projects undertaken by the Agency in the Russian River and Eel River and their tributaries.

i. "Offpeak Agreement" means the July 1, 2015 agreement between Agency and District that is described in recital K of this Water Sale Agreement.

j. "Transmission System" means the water-supply facilities financed and constructed pursuant to the Restructured Agreement for Water Supply and its predecessor agreements, including the remaining features of the Russian River-Cotati Intertie authorized by the Restructured Agreement for Water Supply but not yet constructed.

k. "Water Contractors" means the Cities of Cotati, Petaluma, Rohnert Park, Santa Rosa and Sonoma, the Town of Windsor, the Valley of the Moon Water District, and the North Marin Water District.

Section 2. Agency Deliveries of Water to District.

Subject to all of the terms and conditions of this Water Sale Agreement, and subject to all relevant present and future provisions of the Restructured Agreement for Water Supply, District may purchase water from Agency pursuant to this Water Sale Agreement in any Fiscal Year in an amount not to exceed either 10,000 acre-feet or the District's Maximum Delivery Limit for that Fiscal Year. Upon request by District and subject to all of the terms of this Water Sale Agreement, Agency shall make such water available to District at Agency's Kastania Reservoir at delivery rates specified by District, but not to exceed rates calculated by dividing the District's Maximum Delivery Limit by 10,000 acre feet and multiplying this quotient by the following amounts: (a) 9 million gallons per day between May 1 and October 31 of each year; (b) 12 million gallons per day during the months of April and November of each year; and (c) 15 million gallons per day during the months of April and November of each year; and (c) 15 million gallons per day during the months of April and November of each year; and (c) 15 million gallons per day during the months of April and November of each year; and (c) 15 million gallons per day during the months of April and November of each year; and (c) 15 million gallons per day during the months of April and November of each year; and (c) 15 million gallons per day during the months of April and November of each year; and (c) 15 million gallons per day during the months of April and November of each year; and (c) 15 million gallons per day during the months of April and November of each year; and (c) 15 million gallons per day during the months of April and November of each year; and (c) 15 million gallons per day during the months of April and November of each year; and (c) 15 million gallons per day during the months of April and November of each year; and (c) 15 million gallons per day during the months of April and November of each year; and (c) 15 million gallons per day during the months during all other times.

To the extent permissible under the Offpeak Agreement, the first 360 acre-feet of water received by District from Agency in any month shall be accounted for as being received pursuant to that agreement.

District may not receive any water pursuant to this Water Sale Agreement in any month unless or until one of the following two events has occurred: (1) District already has received in that month at least 360 acre-feet pursuant to the Offpeak Agreement; or (2) District already has received in that Fiscal Year the maximum amount of water that it may receive pursuant to that Offpeak Agreement.

## Section 3. Meter, Connections and Delivery.

• • • • •

Agency will install a turnout, a meter, and appropriate appurtenances at its Kastania Reservoir so that the water delivered to District pursuant to this Water Sale Agreement may be accurately measured and delivered to District's water system. District shall take delivery of water purchased from Agency pursuant to this Water Sale Agreement immediately after the water has passed through this meter. After installation of this turnout, meter and appurtenances, Agency shall bill District for their cost. District shall pay Agency the amount of this bill within 30 days after receipt of the bill.

Agency shall keep accurate records of the amounts of water delivered to District at this meter, and shall allow District to inspect these records during regular business hours.

## Section 4. Maximum Delivery Limit.

a. In the Fiscal Year commencing July 1, 1995, District's Maximum Delivery Limit shall be 5,000 acre-feet. If District takes delivery of, or makes payments in lieu of delivery pursuant to Section 5 of this Water Sale Agreement for, at least 90 percent of the Maximum Delivery Limit in any Fiscal Year, then its Maximum Delivery Limit for the following Fiscal Year shall be increased by 1,000 acre-feet, but never to exceed 10,000 acre-feet.

b. Beginning with the Fiscal Year immediately following District's completion of transmission facilities sufficient to take delivery of at least 5,000 acre-feet of water pursuant to this Water Sale Agreement, or with the July 1, 1996 to June 30, 1997 Fiscal Year, whichever occurs first, if District does not take delivery pursuant to this Water Sale Agreement of, or make payments in lieu of delivery for, at least 90 percent of its Maximum Delivery Limit in any Fiscal Year, then District's Maximum Delivery Limit for the following Fiscal Year shall be decreased by 1,000 acre-feet, except:

The Maximum Delivery Limit shall not decrease for the Fiscal Year following any Fiscal Year in which District requests delivery of at least 90 percent of the Maximum Delivery Limit and either (1) Agency determines, pursuant to Section 7 of this Water Sale Agreement, that less than 90 percent of the Maximum Delivery Limit is available; or (2) Agency determines pursuant to Section 9 of this Water Sale Agreement that Transmission System capacity will not be available to deliver District pursuant to this Water Sale Agreement at least 90 percent of the Maximum Delivery Limit.

### Section 5. <u>Payments In Lieu of Delivery.</u>

- - - - - **-**-

District may elect to make payments to Agency in lieu of delivery to prevent any reduction of District's Maximum Delivery Limit for the following Fiscal Year.

Within 10 days after the end of any month in which District makes such an election, it shall notify Agency in writing of the number of acre-feet for which District elects to make payments in lieu of delivery. The total of this number of acre-feet plus the number of acre-feet of water actually delivered during the month pursuant to this Water Sale Agreement shall not exceed the volume equal to the number of days in the month times the applicable maximum daily delivery limit specified in Section 2 of this Water Sale Agreement. If at the end of May of any fiscal year District has not taken delivery pursuant to this Water Sale Agreement of at least 172 acre-feet of water during the fiscal year, District shall be deemed to have elected to make a payment in lieu of delivery during the month of May for the difference between 172 acre-feet and the amount of water for which delivery already has been taken, or for which payments in lieu of delivery already have been made, pursuant to this Water Sale Agreement. If at the end of June of any fiscal year District has not taken delivery pursuant to this Water Sale Agreement of at least 1000 acre-feet of water during the fiscal year, District

shall be deemed to have elected to make a payment in lieu of delivery during the month of June for the difference between 1000 acre-feet and the amount of water for which delivery already has been taken or payments in lieu of delivery already have been made, pursuant to this Water Sale Agreement. . .

ġ.

After receipt of such notice from District, and in instances when District shall be deemed to have elected to make payments in lieu of deliveries, Agency shall bill District for the appropriate number of acre-feet at the rate determined pursuant to Section 10 of this Water Sale Agreement, less Agency's estimated average, system wide per-acre-foot costs (as calculated by Agency) for the energy and chemicals necessary to operate the Transmission system during the preceding month. When the actual average, system wide per-acre-foot costs (as calculated by Agency) for the energy and chemicals necessary to operate the Transmission System during any month in which District has elected to make payments in lieu of delivery are known, Agency shall make an appropriate adjustment to the amount previously billed to District pursuant to this section, and credit or debit the District for the adjustment.

District shall pay the amount of each bill made pursuant to this section within 30 days after receipt of the bill. No such billing or payment shall affect District's obligations to make payments to Agency pursuant to any other section of this Water Sale Agreement.

In calculating District's Maximum Delivery Limit for the following Fiscal Year, Agency shall add to the amount of water that District received pursuant to this Water Sale Agreement during the current Fiscal Year the number of acre-feet for which District made such payments in lieu of delivery during the current Fiscal Year.

#### Section 6. <u>Annual Request by District.</u>

On or before May 1 of each year, District shall submit a written request to Agency, stating the amount of water that District requests to purchase pursuant to this Water Sale Agreement and the Offpeak Agreement during the following Fiscal Year.

#### Section 7. Availability.

/15 m

Except as otherwise provided in Section 8 of this Water Sale Agreement, water shall be deemed to be available for sale by Agency to District pursuant to this Water Sale Agreement if Agency has water in excess of the amounts that it needs to supply all of its Other Customers and Contractors, to meet its obligations under the Offpeak Agreement, to meet its obligations under any contract between Agency and the Mendocino County Russian River Flood Control and Water Conservation Improvement District in existence on October 22, 1991, to meet its obligations to appropriative water rights owners, to meet instream flow requirements, and to maintain a Prudent Storage Reserve, considering hydrologic conditions, Agency's offstream water production capacity, the probable effects of water conservation efforts on water demand and other relevant factors.

On or before March 1 of each year, Agency shall estimate the amount of water that will be available during the next Fiscal Year for sale to District pursuant to this Water Sale Agreement, and advise the District of this estimate. On or before June 1 of each year, Agency shall, to the extent that it can, make a final determination of the amount of water that will be available the next Fiscal Year for sale to District pursuant to this Water Sale Agreement, and advise the District of this amount. If Agency notifies District pursuant to this section on March 1 of any year that the amount of water that will be available for delivery by Agency to District pursuant to this Water Sale Agreement during the following Fiscal Year will be less than District's Maximum Delivery Limit for that following Fiscal Year, then District may elect in writing to defer until July, August, September or October of the following Fiscal Year delivery of up to one-half of the undelivered water that District otherwise would be entitled to receive pursuant to this Water Sale Agreement during the current Fiscal Year. If District makes such an election, then none of the maximum daily delivery limits specified in Section 2 of this Water Sale Agreement shall be exceeded by the total of all water delivered by Agency to District pursuant to this Water Sale Agreement. If District makes such an election and receives the deferred-delivery water during the following Fiscal Year, then for the purposes of Section 4 of this Water Sale Agreement such water shall be treated as if District had taken delivery of it during the current Fiscal Year.

Section 8. Firm Water Supply.

a. No later than July 31, 1996, District shall pay to the Agency a lump sum representing a share of the Warm Springs Dam sinking fund and principal payments made to the United States Government by Agency for Lake Sonoma water storage space corresponding to 5,000 acre-feet per annum of the 75,000 acre-feet per annum net firm yield of the Warm Springs Dam Project.

b. At any time that is after District has made the payment described in paragraph (a) of this section and prior to July 1, 2005 District shall have the option to pay to the Agency a lump sum representing a share of the Warm Springs Dam sinking fund and principal payments made to the United States Government by Agency for Lake Sonoma water storage space corresponding to an additional 5,000 acre-feet per annum of the 75,000 acre-feet per annum net firm yield of the Warm Springs Dam Project.

1.5

c. The amount of the lump sum payments referred to in paragraphs (a) and (b) of this section shall be determined by the following formula:

 $C = (A + B) \times O / (75,000 \text{ acre-feet} - O)$ 

In this formula *C* is the lump sum payment representing a share of the Warm Springs sinking fund corresponding to an additional 5,000 acre-feet per annum of the 75,000 acre-feet per annum net firm yield of the Warm Springs Dam Project; *A* is the balance of cash and market value of investments held by the Agency's Warm Springs Dam Sinking fund on the last day of the fiscal year quarter immediately preceding the date of the payment provided for in paragraph (a) or (b) of this section; *B* is the total amount of the principal payments which have been made by the Agency to the United States Government for water storage space in Lake Sonoma as of the date of the payment provided for in paragraph (a) or (b) of this section; and *O* is the 5,000 acre-feet of annual firm water to be made available to MMWD pursuant to paragraphs (a) or (b) of this section.

d. In lieu of the determination of availability provided for in Section 7 of this Water Sale Agreement, Agency shall release water from storage in Lake Mendocino or Lake Sonoma when necessary to make available in the Russian River sufficient water to make deliveries of the first 5,000 acre-feet per annum of water provided for by this Water Sale Agreement, but subject to the provisions of paragraph (f) of this section.

e. If District makes the payment pursuant to paragraph (b) of this section, then in lieu of the determination of availability provided for in Section 7 of this Water Sale Agreement, Agency shall release water from storage in

- Exh. B, page 15 -

Lake Mendocino or Lake Sonoma when necessary to make available in the Russian River sufficient water to make the deliveries of all the water provided for by this Water Sale Agreement, but subject to the provisions of paragraph (f) of this section.

f. With respect to the releases provided for in paragraphs (d) and (e) of this section, if by reason of drought or other physical cause or legal impediment beyond the control of the Agency, a shortage in the water available from the Russian River or the Russian River Project occurs, then Agency shall not be liable to District for any damage resulting therefrom. In the event of a shortage of water in the Russian River or the Russian River or the Russian River Project that requires the Agency to apportion available water, Agency shall apportion the available water so that it may make deliveries as follows:

First, Agency shall deliver to each of its regular customers, not in excess of the respective entitlements set forth in Sections 3.1, 3.2 and 3.3 of the Restructured Agreement for Water Supply, or any amendments or successor agreements thereto, authorize Agency's Russian River customers to divert or redivert not in excess of the amounts for which those customers have contracted to purchase from the Agency, and deliver to District not in excess of the amounts set forth in Section 1 of the Offpeak Agreement, paragraph (d) of this section, and (if District makes the payment pursuant to paragraph (b) of this section) paragraph (e) of this section, the quantities of water required by each for human consumption, sanitation and fire protection, as determined by the Agency after taking into consideration all other sources of potable water then available to said customer;

Second, to the extent additional water is available to the Agency, Agency shall deliver such water to Agency's regular customers, authorize Agency's Russian River customers to divert or redivert such water, and deliver such water to the District, in proportion to the respective entitlements set forth in Sections 3.1, 3.2 and 3.3 of the Restructured Agreement For Water Supply, or any amendments or successor agreements thereto, the agreements between the Agency and its Russian River customers, Section 1 of the Offpeak Agreement, paragraph (d) of this section and (if District makes the payment pursuant to paragraph (b) of this section) paragraph (e) of this section, provided, however, that no customer shall receive under subdivisions "First" and "Second" hereof a total quantity of water in excess of its reasonable requirements or it's said entitlement or contracted amount, whichever is less.

The terms "regular customers", "Russian River customers" and "water contractors" in this Water Sale Agreement shall have the same meanings that they have in the Restructured Agreement for Water Supply.

#### Section 9. Delivery Schedule.

Notwithstanding a determination of availability of water pursuant to Section 7 or Section 8 of this Water Sale Agreement, Agency shall be obligated to release water from storage or deliver water to District pursuant to this Water Sale Agreement only to the extent that the Transmission System has capacity in excess of that required by Agency to supply all of its Other Customers and Contractors, and District under the Offpeak Agreement. However, Agency shall not curtail deliveries pursuant to this section because of inadequate capacity in any new Ely-to-Kastania Aqueduct constructed pursuant to Section 13 of this Water Sale Agreement.

On or before June 1 of each year, Agency shall submit a written estimate to District of the capacity of the Transmission System that is expected to be available to deliver water pursuant to this Water Sale Agreement for each month of the following Fiscal Year. Fifteen days before the first day of each month, District shall notify the Agency of its planned rate of delivery for that month pursuant to this Water Sale Agreement and the Offpeak Agreement. Such notice may be by telephone unless written notice is requested by Agency.

## Section 10. <u>Water Charges.</u>

On or before April 30 of each year, Agency shall establish a per-acre-foot water rate applicable to all water delivered pursuant to this Water Sale Agreement for the following Fiscal Year. This rate shall be the sum of the following three per-acre-foot charges:

A per acre foot charge equal to the highest rate per а. acre-foot then charged by the Agency to any party to the Restructured Agreement for Water Supply (or pursuant to any amendment or successor agreement to the Restructured Agreement for Water Supply) for water taken from either the Petaluma Aqueduct or the Santa Rosa Aqueduct, multiplied by 1.11; provided, however, that the 20% surcharge imposed on the Town of Windsor under Section 4.17(a) of the Restructured Agreement shall not be included in determining the highest rate per acre-foot for water taken from the Santa Rosa Aqueduct or Petaluma Aqueduct. Seven and four hundred thirty-two one-thousandths percent (7.432%) of this per-acre-foot charge shall be placed in the Russian River Projects Fund; two and four hundred seventy-seven thousandths percent (2.477%) of this per-acre-foot charge shall be used, at the discretion of the Agency, to pay for the costs of Common Facilities, to pay the Capital Costs of Aqueduct Facilities relating to the Santa Rosa or Petaluma Aqueducts, or to pay

operations and maintenance costs; and the remainder shall be applied pursuant to the Restructured Agreement for Water Supply (or pursuant to any amendment or successor agreement to the Restructured Agreement for Water Supply), with the water delivered to Marin considered to be delivered from the Petaluma Aqueduct.

1. 76

b. A Russian River Conservation charge which shall be paid in lieu of the property taxes levied by the Agency on property in Sonoma County, to pay the capital, operation and maintenance costs associated with the Warm Springs Dam Project. The Russian River Conservation Charge shall be determined by multiplying the tax rate levied by the Agency in the then current fiscal year to pay the costs associated with the Warm Springs Dam Project times the assessed value of secured and unsecured property situated within the cities of Cotati, Petaluma, Rohnert Park, Santa Rosa and Sonoma, the Forestville County Water District, and the Valley of the Moon Water District, and dividing the product by the total number of acre-feet of water delivered to said public agencies pursuant to Section 3.1 and 3.3 of the Restructured Agreement for Water Supply, or any amendments or successor agreements thereto, during the twelve month period ending on March 31.

c. A Russian River Projects Charge which shall be paid in lieu of the property taxes levied on property in Sonoma County and other Agency general fund monies which are transferred to the Agency's Russian River Projects Fund and expended for the purposes enumerated in subsection (h) of Section 1. The Russian River Projects Charge shall be determined by dividing the total amount of Agency monies expended from the Agency's Russian River Projects Fund in the preceding ten fiscal years, exclusive of the funds contributed to the Fund by District and North Marin Water District,

- Exh. B, page 19 -

and interest earnings attributable to funds contributed by District and North Marin Water District, by the sum of the total acre-feet of water delivered by the Agency to the cities of Cotati, Petaluma, Rohnert Park, Santa Rosa and Sonoma, the Forestville County Water District, and the Valley of the Moon Water District pursuant to Sections 3.1 and 3.3 of the Restructured Agreement for Water Supply, or any amendments or successor agreements thereto, during the preceding ten fiscal years and multiplying the quotient by the ratio that the assessed value of the secure and unsecured property situated within the cities of Cotati, Petaluma, Rohnert Park, Santa Rosa and Sonoma, the Forestville County Water District and the Valley of the Moon Water District bears to the assessed value of all secured and unsecured property within Sonoma County, provided, however, in no event shall the Russian River Projects Charge exceed \$20.00 per acre-foot. Agency shall keep proper books, records and accounts in which complete and accurate entries shall be made of all Agency general fund monies transferred to the Agency's Russian River Projects Fund and all expenditures made from the fund for the purposes described in this paragraph. The Agency shall maintain separate account within the Russian River Projects Fund for monies contributed by District and North Marin Water District. Monies expended from the Russian River Projects Fund shall be deemed to have been expended from the District and North Marin Water District account in the proportion that the balance of that account bears to the total Russian River Projects Fund balance at the end of the fiscal year quarter preceding the expenditure.

Agency shall bill District each month for all of these charges, based on the quantity of water delivered to District pursuant to this Water Sale Agreement during the preceding month. District shall pay the amount of each such billing within 30 days after receipt of the bill.

A 1 4 4

If in any fiscal year the sum of the total amount of water delivered under this Water Sale Agreement plus the total amount of water for which payments are made in lieu of delivery pursuant to Section 5 of this Water Sale Agreement is less than the total amount of water for which District has made firm water supply payments pursuant to subsections (a) and (b) of Section 8 of this Water Sale Agreement, District shall pay the following additional payments:

d. An additional lump sum payment which shall be determined by multiplying the Russian River Conservation Charge times the difference between the total acre-feet of water for which District has made firm water supply payments pursuant to subsections (a) and (b) Section 8 of this Water Sale Agreement and the sum of the total acre-feet of water delivered under this Water Sale Agreement during the prior fiscal year plus the total acre-feet of water for which payments have been made in lieu of delivery pursuant to Section 5 of this Water Sale Agreement during the prior fiscal year plus the prior fiscal year.

e. An additional lump sum payment which shall be determined by multiplying the Russian River Projects Charge times the difference between the total acre-feet of water for which District has made firm water supply payments pursuant subsection (a) and (b) of Section 8 of this Water Sale Agreement and the sum of the total acre-feet of water delivered under this Water Sale Agreement during the prior fiscal year plus the total acre-feet of water for which payments have been made in lieu of delivery pursuant to Section 5 of this Water Sale Agreement during the prior fiscal year.

Agency shall bill District for these additional lump sum payments within 30 days after the end of each fiscal year. District shall pay the amount of each such billing within 30 days after receipt of the bill. 

## Section 11. Payment.

Notwithstanding any dispute between Agency and District, District shall pay all bills made by Agency pursuant to this Water Sale Agreement when due and shall not withhold all or any part of any amount billed pending the final resolution of such dispute. In the event of a dispute, District may pay its bills under protest, and if necessary under the ultimate resolution of the dispute, Agency shall make an appropriate refund to District, including interest on the overpaid amount at the rate obtained by Agency as a result of investment of the disputed amount. If District does not pay any bill by the due date for such bill, then, in addition to the principal amount due, District also shall pay Agency interest on this principal amount due, calculated from the due date until the payment date at the legal rate per annum established pursuant to Section 685.010 of the Code of Civil Procedure.

## Section 12. Acceleration of Construction of New Aqueduct Elements.

If the financing and construction by Agency of a new aqueduct from the Russian River to the Ely Pumping Plant, roughly paralleling portions of the existing Cotati Intertie and Petaluma Aqueduct, is authorized by an amendment to the Restructured Agreement for Water Supply or successor agreement, then Agency shall annually provide District with a copy of its long-range capital improvement program. District may from time to time make written requests to Agency, requesting Agency to construct one or more elements of this new aqueduct at dates earlier than the dates planned by Agency. If Agency, after considering the availability of funds and other relevant matters, agrees to accelerate the construction of any element or elements of this new aqueduct, then District shall commit in writing to reimburse Agency for the interest revenue foregone, or interest paid by Agency, as the result of such accelerated construction. Upon receipt of such written commitment acceptable to Agency in form and substance, Agency shall construct such element or elements. Agency shall bill District annually at least 30 days in advance of each bill's due date pursuant to such commitment and District shall pay each bill by its due date.

. . . .

Construction of the new aqueduct elements shall be financed by Agency. Title to all elements of this new aqueduct shall vest with Agency, which shall operate and maintain this new aqueduct as part of the Transmission System. Construction of this new aqueduct shall not change any of the delivery limits, availability provisions or other terms of this Water Sale Agreement.

## Section 13. Construction of New Ely-To-Kastania Aqueduct.

If Agency initiates construction of a new aqueduct roughly paralleling the portion of the existing Petaluma Aqueduct that extends from the junction of that aqueduct and the Cotati Intertie to the Ely Pumping Plant, then District may make a written request to Agency, requesting Agency to construct at District's expense an aqueduct roughly paralleling the portion of the existing Petaluma Aqueduct that extends from the Ely Pumping Plant to Kastania Reservoir, with sufficient capacity to meet the maximum delivery limits specified in Section 2 of this Water Sale Agreement and the maximum delivery limit specified in the Offpeak Agreement, taking into account surplus capacity in that portion of the existing Petaluma Aqueduct.

- Exh. B, page 23 -

Construction of this new Ely-To-Kastania aqueduct shall be financed by cash deposits and payments from District. Title to this new aqueduct shall vest with Agency, which shall operate and maintain this new aqueduct as part of the Transmission system. Construction of this new aqueduct shall not change any of the delivery limits, availability provisions or other terms of this Water Sale Agreement. ها يور

Upon receipt of such a written request from District, Agency shall prepare and deliver to District design plans and a schedule for the construction of this new aqueduct, and a schedule for payment by District to Agency of appropriate deposits and payments for such plans and construction. If District commits in writing to pay such deposits and payments to Agency according to this payment schedule, then Agency shall construct this new aqueduct, and shall bill District for such deposits and payments according to this payment schedule. Agency shall bill District at least 30 days in advance of each bill's due date and District shall pay each bill by its due date.

#### Section 14. Place of Use.

ł

Water received by District pursuant to this Water Sale Agreement and the Offpeak Agreement shall only be used within the sphere of influence (as such term is defined in Section 56076 of the Government Code) of the District on [insert effective date].

ł.

## Section 15. <u>Water Conservation.</u>

• • • • • •

District and Agency will comply with all applicable state laws related to water conservation planning and implementation, including the Urban Water Management Planning Act, Water Code Sections 10610-10655 (as such act now exists or in the future may exist). Whenever District files any urban water management plan, or any amendment or change to such a plan, with the Department of Water Resources or any other state agency, District shall, upon request from Agency, at the same time submit a copy of the plan amendment or change to Agency. Whenever Agency files any urban water management plan, or any amendment or change to such a plan, with the Department of Water Resources or any other state agency, Agency shall, upon request from District, at the same time submit a copy of the plan, amendment or change to District.

## Section 16. California Environmental Quality Act.

Pursuant to Section 15051(d) of the State CEQA Guidelines, District is designated as the lead agency under the California Environmental Quality Act for the execution of this Water Sale Agreement, and for any projects south of Kastania Reservoir that District may construct to implement this Water Sale Agreement. Agency is designated as the lead agency for any projects north of Kastania Reservoir that Agency may construct to implement this Water Sale Agreement.

## Section 17. <u>Future Agency Application to Increase Limit on Diversions</u> and Rediversions in Permit 16596.

District acknowledges that the 75,000 acre foot per year limit on diversions and rediversions in term 5 of Permit 16596 does not allow

Agency to utilize the entire yield of the Warm Springs Dam Project, and that Agency intends to file in the future an application with the State Water Resources Control Board to increase this limit. District recognizes and agrees that it will be in the public interest for the State Board to grant any such application.

## Section 18. Effective Date, Term and Termination.

This Water Sale Agreement shall be effective July 1, 2015 and shall remain in effect until June 30, 2025. Upon the request of District, Agency shall enter into renewal agreements for periods not to exceed the then-existing term of the Restructured Agreement for Water Supply or any renewals, amendments or replacement agreements to the Restructured Agreement for Water Supply upon the same terms and conditions contained in this Water Sale Agreement, except that the Agency may make reasonable adjustments to the charges under Section 10 of this Water Sale Agreement, and any such reasonable adjustments then shall be included in any renewal agreement. If the Restructured Agreement for Water Supply has been terminated or has expired without being renewed, amended or replaced by another agreement pertaining to water supply, then upon the request of District, the Agency shall enter into renewal agreements for periods not to exceed forty (40) years upon the same terms and conditions contained in this Water Sale Agreement, except that the Agency may make reasonable adjustments to the charges under Section 10 of this Water Sale Agreement and any such reasonable adjustment then shall be included in any renewal agreement.

## Section 19. Merger.

1.5 ml /

This writing is intended both as the final expression of the agreement between the parties hereto with respect to the included terms and conditions and as a complete and exclusive statement of the terms of the Water Sale Agreement. Pursuant to Code of Civil Procedure Section 1856, no modification of the Water Sale Agreement shall be effective unless and until such modification is evidenced by a writing signed by all parties.

#### Section 20. Equitable Remedies.

Because water is a scarce and precious resource, District will not have an adequate remedy at law, and thus may request a court of competent jurisdiction to order equitable remedies, to compel Agency to deliver the water that District is entitled to receive pursuant to this Water Sale Agreement. Such equitable remedies shall be District's sole and exclusive remedies in actions brought by District relating to the amounts of water that Agency must deliver to District pursuant to this Water Sale Agreement.

## Section 21. <u>Waiver and Indemnification.</u>

District waives, releases and forever discharges Agency, its officers, agents and employees from any and all liabilities, claims, demands, losses and costs relating to any of the following: (1) any property damage or personal injury arising from any non-delivery of water requested by District pursuant to this Water Sale Agreement, or for any property damage or personal injury arising from the quality of water delivered pursuant to this Water Sale Agreement, if such property damage or personal injury is caused by lack of available capacity in the Transmission System, drought, earthquake or other Act of God, strike or other labor dispute, partial or total dam, gate or tunnel loss, dam, gate or tunnel repairs, water pollution, or any factor beyond the control of Agency (whether or not such factor is listed in this sentence); and (2) any property damage or personal injury arising from any decision of Agency or the Water Contractors regarding:

(a) determinations of the availability of water for sale by Agency to District pursuant to this Water Sale Agreement; (b) allocation of Transmission System capacity; (c) proposed expansions of the Transmission System; or (d) repair (or non-repair) of the Potter Valley Project, Coyote Valley Dam or Warm Springs Dam. District's waiver, release and discharge described in this paragraph shall apply to any of the property damages or personal injuries described in this paragraph, whether or not such property damages or personal injuries were caused by Agency's negligence, unless such property damages or personal injuries resulted from Agency's sole negligence, willful misconduct or violation of law.

District shall indemnify, hold harmless, protect and defend Agency, its officers, agents and employees from and against any and all liabilities, claims, demands, damages, losses, disabilities or expenses (including attorney fees and litigation costs) of every nature arising out of, or in connection with: (1) the lack of quantity of water that has been delivered by Agency to District pursuant to this Water Sale Agreement; or (2) the control, conveyance and disposition of water that has been delivered by Agency to District pursuant to this Water Sale Agreement. For the purposes of this section, the point of delivery shall be as specified in Section 3 of this Water Sale Agreement. District shall provide such indemnification, holding harmless, protection and defense whether or not such liabilities, claims, demands, damages, losses, disabilities or expenses are based on Agency's negligence, unless such liabilities, claims,

demands, damages, losses, disabilities or expenses are based on Agency's sole negligence, willful misconduct or violation of law.

This section shall not apply to any construction activities, or construction contracts, relating to the construction described in Section 13 of this Water Sale Agreement. The provisions in this section regarding attorney fees shall not apply to any other section of this Water Sale Agreement.

#### Section 22. <u>Water Quality.</u>

A Stand A

Water delivered by Agency to District pursuant to this Water Sale Agreement shall be of the same general quality as water delivered by Agency from the Transmission System to the Water Contractors. Except as expressly stated in the preceding sentence, Agency does not make any express or implied warranty regarding the quality of the water delivered pursuant to this Water Sale Agreement.

#### Section 23. <u>No Effect on Offpeak Agreement.</u>

Except as stated in Section 14 of this Water Sale Agreement, nothing in this Water Sale Agreement shall be construed as affecting or amending the Offpeak Agreement.

#### Section 24. Method and Place of Giving Notice and Making Payments.

All notices shall be in writing and notices and payments may be given by personal delivery or by mail. Notices and payments sent by mail shall be addressed as follows;

Agency:	General Manager Sonoma County Water Agency 404 Aviation Boulevard Santa Rosa, CA 95403
D:_1.:	

District: General Manager Marin Municipal Water District 220 Nellen Avenue Corte Madera, CA 94925

When so addressed, a notice shall be deemed given upon deposit in the United States Mail, postage prepaid. In all other instances, notices and payments shall be deemed given at the time of actual delivery. Changes may be made in the names and addresses of the person to who notices and payments are to be given by giving notice pursuant to this section. .

## Section 25. Demand for Assurance.

When reasonable grounds for insecurity arise with respect to the performance of either party, the other party may, in writing, demand adequate assurance of due performance and, until it receives such assurance, may, if commercially reasonable, suspend any performance for which the agreed return performance has not been received. "Commercially reasonable" includes not only the conduct of either party with respect to performance under this Water Sale Agreement, but also conduct with respect to other agreements with the other party to this Water Sale Agreement or others. After receipt of a justified demand, the failure to provide within a reasonable time, but not exceeding 30 days, such assurance of due performance as is adequate under the circumstances of the particular case is a repudiation of this Water Sale Agreement. Acceptance of any improper delivery, service, or payment does not

prejudice the aggrieved party's right to demand adequate assurance of future performance.

Section 26. Third Party Beneficiaries.

No third party beneficiaries are intended or established by this Water Sale Agreement.

Section 27. Representation by Counsel.

Agency and District each was represented by independent counsel in the negotiation and execution of this Water Sale Agreement. For the purposes of interpretation of this Water Sale Agreement, neither party shall be deemed to have been the drafter of this Water Sale Agreement.

ATTEST:

A 34 4

Secretary

Reviewed as to Substance Marin Municipal Water District

Krishna Kumar General Manager

Reviewed As To Form:

Mu Mary(R./Casey

General Counsel

MARIN MUNICIPAL WATER DISTRICT

By:

John C. Gibson, President Board of Directors

ATTEST:

By:

Deputy Clerk of the Board of Directors

SONOMA COUNTY WATER AGENCY

Chair, Board of Directors

15 DATE:

Reviewed As To Substance By Sonoma County Water Agency:

Grant Davis, General Manager

Reviewed As To Form:

AurShupe

Steven S. Shupe Deputy County Counsel



CF/60-62-21.5 MARIN MUNICIPAL WATER DISTRICT (NON-PRIME WATER SERVICE AGREE) TW NO (ID 1576)

June 23, 2015

Marin Municipal Water District Attn: Krishna Kumar, General Manager 220 Nellen Avenue Corte Madera, CA 94925-1169

Dear Mr. Kumar:

## RE: RENEWAL OF WATER SUPPLY AGREEMENTS

On June 16, 2015, the Sonoma County Water Agency Board of Directors approved the water supply agreements that are part of the *Renewal of the Third Offpeak Water Supply Agreement and the Amended Agreement for the Sale of Water between the Sonoma County Water Agency and Marin Municipal Water District*. The agreements are effective July 1, 2015 and shall remain in effect until June 30, 2025. Your copy of the executed agreements is enclosed.

If you have any questions, please contact me at (707) 524-1173.

Sincerely,

Todd J. Schram, P.E. Senior Water Agency Engineer

Encl

C. Davis, P. Jeane, J. Jasperse, D. Seymour — Sonoma County Water Agency
 C. O'Donnell, S. Shupe—Sonoma County Counsel
 Alan Lilly—Bartkiewicz, Kronick & Shanahan

pa\\T:\Pinks\06-22-15\Transmittal\_ExecutedCopy\_22jun15.docx

Appendices 2020 Urban Water Management Plan Marin Municipal Water District



## Appendix C

2020 Urban Water Management Plan Water Demand Analysis and Water Conservation Measures Update, Marin Municipal Water District



December 2020 (EKI C00004.00)

## Prepared by:

EKI Environment & Water, Inc. 2001 Junipero Serra Boulevard, Suite 300 Daly City, California 94014 (650) 292-9100



#### **Table of Contents**

1.	INTRODUCTION	1-1	
2.	REGULATORY CONTEXT		
	2.1. 2020 UWMP Demand Projections Requirements	2-1	
	2.2. New Requirements for 2020 UWMPs and Future Demand Planning	2-2	
	2.2.1. Annual Urban Water Use Objectives	2-2	
	2.2.2. Supply Reliability	2-4	
	2.2.3. Water Shortage Contingency Plans	2-4	
3.	WATER USE CHARACTERISTICS	3-1	
	3.1. Historical Total and Per Capita Water Use	3-1	
	3.2. Historical Average Water Use Per Account	3-1	
	3.3. Change in Residential Water Use Pre- and Post-Drought	3-2	
	3.4. Residential Water Use by Dwelling Unit and Age of Construction	3-2	
	3.5. Estimated Indoor and Outdoor Water Use	3-3	
4.	WATER DEMAND PROJECTIONS	4-1	
	4.1. Basis for Account Growth Projections	4-1	
	4.2. Change in Number of Accounts based on Projected Growth	4-2	
	4.3. Water Demand Factors	4-2	
	4.3.1. Potable, Raw, and Recycled Water	4-3	
	4.3.2. Non-Revenue Water (Potable Water System)	4-5	
	4.4. Passive Water Savings Estimates	4-5	
	4.5. Projected Water Demand Through 2045	4-5	
5.	CONSERVATION PROGRAM PARTICIPATION	5-1	
	5.1. Conservation Programs	5-1	
	5.2. Historical Conservation Program Participation	5-4	
	5.3. Estimated Savings from Past Programs	5-5	
	5.3.1. Estimated Water Savings Based on AWE Model	5-5	
	5.3.2. Estimated Water Savings for Selected Programs Based on Customer Billing Data	5-5	
	5.4. Spatial Trends in Program Participation	5-7	
	5.5. Building Stock Characteristics	5-8	
	5.6. Demographic Characteristics of Residential Conservation Program Participation	5-9	
	5.6.1. Household Income Trends	5-9	
	5.6.2. Homeownership Trends	5-10	
	5.6.3. Household Age Trends	5-10	
	5.7. Summary	5-11	
6.	CONSERVATION PROGRAM UPDATE	6-1	
	6.1. Methodology for Screening of Potential Water Conservation Programs	6-1	
	6.2. Screening of Regional Conservation Measures	6-2	



#### **Table of Contents**

REFERENCES			8-1	
CONCLUSION			7-1	
6.4. Evaluation of Future Water Conservation Programs				
6.3. Screening of Local Conservation Measures				
(	6.2.5.	Regional Program Screening Findings	. 6-4	
(	6.2.4.	Policy and Regulation Based Conservation Programs	. 6-3	
(	6.2.3.	Device and Financial Incentive Based Conservation Programs	. 6-3	
(	6.2.2.	Public Outreach and Education Based Conservation Programs	. 6-2	
(	5.2.1.	Retailer Actions and Water Rate Based Conservation Programs	.6-2	

#### TABLES

7. 8.

Table 3-2. Water Use by Cust	omer Sector
------------------------------	-------------

- Table 3-3Number of Accounts by Customer Sector
- Table 3-4a.Per Account Water Use by Customer Sector
- **Table 3-4b.**Per Dwelling Unit Water Use for Residential Sectors
- **Table 3-5.**Residential Water Use by Age of Construction
- Table 3-6.Monthly Water Use
- **Table 3-7.**Estimated Indoor and Outdoor Water Use
- **Table 4-1.** Population and Employment Growth Projections
- **Table 4-2.**Historical and Projected Account Growth Rate by Customer Sector
- **Table 4-3.**Change in Number of Accounts based on Projected Growth
- **Table 4-4.**Potential Potable Water Demand Factors Considered
- **Table 4-5.**Selected Water Demand Factors
- Table 4-6.Projected Water Demand
- **Table 5-1.**Description of Conservation Programs
- **Table 5-2.**Summary of Conservation Program Participation
- **Table 5-3.** Summary of Conservation School Education Program Participation
- Table 5-4.
   Estimated Water Savings Achieved by Conservation Programs and Passive Savings
- Table 5-5.Average Estimated Water Savings Achieved by Selected Conservation Programs from<br/>2010-2018
- Table 5-6a.
   Estimated Water Savings Achieved by the AMI Leak Notifications Program
- Table 5-6b.
   Estimated Water Savings Achieved by the Residential HECW Rebate Program
- Table 5-6c.
   Estimated Water Savings Achieved by the SFR Water Use Surveys/Audits Program
- Table 5-6d.
   Estimated Water Savings Achieved by the SFR WBIC Rebate Program



#### **Table of Contents**

- **Table 5-7.**Building Stock Characteristics by Program Participants
- Table 5-8a.
   Residential Customer Program Participation by Median Household Income
- Table 5-8b.
   Residential Customer Program Participation by Percentage of Renters
- Table 5-8c.
   Residential Customer Program Participation by Median Household Age
- **Table 6-1.** Regional Prioritization of Conservation Measures and Programs
- Table 6-2.
   Marin Municipal Water District Prioritization of Conservation Measures and Programs
- Table 6-3a.
   Conservation Program Scenarios
- Table 6-3b.
   Costs and Savings of Potential Conservation Programs
- Table 6-3c.
   Comparison of Program Scenarios Costs and Savings

#### FIGURES

- Figure 1-1. Participating Sonoma-Marin Saving Water Partnership Members
- Figure 1-2. Marin Municipal Water District Service Area
- Figure 3-1. SFR Water Use over Time
- Figure 5-1a. Participation Density for the AMI Leak Notifications Program
- Figure 5-1b. Participation Density for AMI Leak Notifications Program Belvedere and Tiburon
- Figure 5-1c. Participation Density for the Residential HECW Rebate Program
- Figure 5-1d. Participation Density for the Rain Barrel Rebate Program
- Figure 5-1e. Participation Density for the SFR Water Use Surveys/Audits Program
- Figure 5-1f. Participation Density for the SFR WBIC Rebate Program
- Figure 5-2. Age of Building Stock
- Figure 5-3a. Median Household Income
- Figure 5-3b. Percentage of Renters
- Figure 5-3c. Median Household Age
- Figure 5-4. SFR Customers to Potentially Target with SFR WBIC Rebate Program Outreach

#### APPENDICES

- Appendix A. California Water Code Revisions per AB-1668, SB-606, and SB-664, Redlines prepared by DWR
- Appendix B. AWE Model Assumptions
- **Appendix C.** Methodology for Conservation Program Savings Analyses
- Appendix D. Prioritization and Screening of Future Water Conservation Measures



#### ABBREVIATIONS AND ACRONYMS

AB	Assembly Bill
ABAG	Association of Bay Area Governments
AFY	acre-feet per year
Ag.	agricultural
AMI	advanced metering infrastructure
AWE	Alliance for Water Efficiency
CA	California
CEQA	California Environmental Quality Act
CII	commercial, industrial, and institutional
CWC	California Water Code
DMM	demand management measure
DOF	Department of Finance
DRA	drought risk assessment
DSS	Decision Support System
d.u.	dwelling unit
DWR	Department of Water Resources
GPCD	gallons per capita day
GPD	gallons per day
gpf	gallons per flush
HECW	High Efficiency Clothes Washer
HET	High Efficiency Toilet
Irr.	irrigation
MF	multi-family
MFR	multi-family residential
MMWD	Marin Municipal Water District
psi	pounds per square inch
QWEL	Qualified Water Efficient Landscaper
SB	Senate Bill
SFR	single family residential
SMSWP	Sonoma-Marin Saving Water Partnership
Sonoma Water	Sonoma County Water Agency
sq ft	square feet
SWRCB	State Water Resources Control Board
ULFT	ultra low flow toilet
UWMP	Urban Water Management Plan
WBIC	Weather Based Irrigation Controller
WSA	Water Supply Assessment
WSCP	Water Shortage Contingency Plan



## **1. INTRODUCTION**

In preparation for development of their 2020 Urban Water Management Plan (UWMP) updates, nine members of the Sonoma-Marin Saving Water Partnership (SMSWP or Water Contractors) coordinated to conduct a joint update of their water demand projections and water conservation planning efforts (i.e., the *2020 Water Demand and Conservation Project*). The participating SMSWP members include: City of Cotati, City of Petaluma, City of Rohnert Park, City of Santa Rosa, City of Sonoma, Marin Municipal Water District, North Marin Water District, Town of Windsor, and Valley of the Moon Water District. These nine Water Contractors are shown on **Figure 1-1**.

The goals of the *2020 Water Demand and Conservation Project* were to apply a common methodology to conduct the following analysis for each Water Contractor:

- Evaluate and document recent historical water use characteristics and trends, including population and account growth;
- Estimate projected water demands for the years 2025 through 2045 to support both the 2020 UWMP update and coordination and planning efforts with Sonoma County Water Agency (Sonoma Water);
- Update the suite of common regional conservation measures that are being considered for implementation in the future;
- Review and document past participation in water conservation programs; and
- Estimate the potential water savings associated with future water conservation program implementation.

This 2020 Water Demand and Conservation report presents the results for the Marin Municipal Water District (District), which is located in Marin County and serves a population of approximately 192,138 people (**Figure 1-2**). The District's water supplies include surface water purchased from the Sonoma County Water Agency (Sonoma Water), surface water collected from the Mt. Tamalpais watershed, and recycled water produced by the Las Gallinas Sanitary District. Potable water is supplied to District customers, and recycled water is provided to a variety of uses in the Terra Linda area of San Rafael including for irrigation, cooling towers, car washes and toilet flushing. Over the years, the District has worked to increase water efficiency (conservation) in response to both the SB X7-7 UWMP requirements and as part of the regional SMSWP. Demand reductions have been achieved through the implementation of the plumbing code and water conservation programs, including some administered by the District and some administered through the regional SMSWP.

This 2020 Water Demand and Conservation report is organized as follows:

- Section 1 identifies the goals and objectives of this report;
- Section 2 provides the regulatory context for the demand projections described in this report as well as new requirements related to UWMPs and long-term demand planning that agencies will need to consider in development of their 2020 UWMPs;
- Section 3 describes historical water use patterns and characteristics within the District;



- Section 4 describes the projected water demands through 2045, including the assumptions and methodology used;
- Section 5 documents past participation in conservation programs and estimated savings associated with program implementation, and presents the results of a detailed analysis of program participation trends for five select conservation programs;
- Section 6 documents the water conservation measure screening process, identifies individual programs and program scenarios for potential future implementation by the District, and presents the results of a benefit-cost analysis and an estimate of the potential water savings associated with these conservation programs;
- Section 7 provides conclusions regarding the main findings of the report; and
- Section 8 provides key references and sources.

Small tables are provided within text throughout the document. Figures and large tables and charts are provided at the end of each section.





Path: X:\C00004\_SonomaMarin\Map\2020\12\AgencyBoundaryMaps\_20201204.mxd


#### 2. REGULATORY CONTEXT

This section provides the regulatory background for the requirements to project future demand in the 2020 UWMP. In addition, it outlines requirements for elements of the District's 2020 UWMP that are beyond the scope of the *2020 Water Demand and Conservation Project*, such as consideration of supply reliability, water shortage contingency planning, and the annual urban water use objectives retailers will be required to report on in 2023 and meet by 2027.

#### 2.1. 2020 UWMP Demand Projections Requirements

California Water Code (CWC) § 10631, excerpted below, describes the requirements to develop water demand projections that consider water use by customer sector, incorporate distribution system water loss, and account for anticipated water savings. As described further in Section 4, water demand projections were developed for the District using a land-use based approach that is consistent with these requirements and previous UWMP demand projection methodologies, and can be incorporated into the District's 2020 UWMP.

#### CWC § 10631

A plan shall be adopted in accordance with this chapter that shall do all of the following:

•••

(d) (1) For an urban retail water supplier, quantify, to the extent records are available, past and current water use, over the same five-year increments described in subdivision (a), and projected water use, based upon information developed pursuant to subdivision (a), identifying the uses among water use sectors, including, but not necessarily limited to, all of the following:

(A) Single-family residential.

(B) Multifamily.

(C) Commercial.

(D) Industrial.

(E) Institutional and governmental.

(F) Landscape.

(G) Sales to other agencies.

*(H)* Saline water intrusion barriers, groundwater recharge, or conjunctive use, or any combination thereof.

(I) Agricultural.

(J) Distribution system water loss.

(2) The water use projections shall be in the same five-year increments described in subdivision (a).

...

(d)(4) (A) Water use projections, where available, shall display and account for the water savings estimated to result from adopted codes, standards, ordinances, or transportation and land use plans identified by the urban water supplier, as applicable to the service area.

(B) To the extent that an urban water supplier reports the information described in subparagraph (A), an urban water supplier shall do both of the following:



(i) Provide citations of the various codes, standards, ordinances, or transportation and land use plans utilized in making the projections.
(ii) Indicate the extent that the water use projections consider savings from codes, standards, ordinances, or transportation and land use plans. Water use projections that do not account for these water savings shall be noted of that fact.

#### 2.2. New Requirements for 2020 UWMPs and Future Demand Planning

Through the recent *Making Water Conservation a California Way of Life* (Assembly Bill [AB]-1668/Senate Bill [SB]-606) and other legislation, the State has made numerous changes to the requirements for UWMPs and related water conservation planning efforts. In many cases, the updated regulations reference details and methodologies to be developed by the California Department of Water Resources (DWR), and/or are somewhat vague and will benefit from the development of guidelines/further clarification by DWR. DWR is currently developing an updated guidebook to support the development of the 2020 UWMPs, which is expected to be complete by late 2020. This new guidebook is anticipated to provide direction to retailers with respect to many elements of the new legislation.

A summary of key changes to various elements of 2020 UWMP and related planning efforts is provided below. Copies of the revisions to relevant sections of the California Water Code per AB-1668, SB-606, and SB-664 are provided in **Appendix A**.

#### 2.2.1. Annual Urban Water Use Objectives

Beginning in 2023,<sup>1</sup> retailers will be required to report on "annual water use objectives" by November 1 of each year, per CWC § 10609. The specific standards that will be used to determine an retailer's annual urban water use objectives are currently under development and are the source of a great deal of uncertainty with respect to the long-term water conservation and demand planning as part of the 2020 UWMP. Although the 2020 UWMP will not identify or calculate these new annual urban water use objectives, the new standards will become effective within the UWMP planning horizon. Per CWC § 10609.25, retailers will be required to "provide a narrative that describes the water demand management measures that the supplier plans to implement to achieve its urban water use objective by January 1, 2027." Details regarding the annual urban water use objectives and other requirements are expected to evolve significantly over the next two years.

 Residential outdoor water use: Per CWC § 10609.6, DWR and California State Water Resources Control Board (SWRCB) "shall conduct necessary studies and investigations and recommend, no later than October 1, 2021, standards for outdoor residential use" which "incorporate the principles of the model water efficient landscape" and "apply to irrigable lands." DWR is currently working with a contractor to measure all of the single- and multi-family landscape (irrigable) area within urban water suppliers' service areas across the state based on aerial imagery. The result of these measurements will become the basis for each retailer's residential landscape water use component of the annual water use objectives. In order to accurately calculate and compare

<sup>&</sup>lt;sup>1</sup> DWR acknowledged publicly on 5 December 2019 that this and other related deadlines are likely to slip. DWR indicated that compliance with these objectives will most likely begin in 2024.



against this metric, retailers will be responsible for identifying dedicated irrigation accounts (water connections) associated with residential water use (including multi-family residential) and dedicated irrigation accounts associated with commercial, industrial and irrigation (CII) use. The landscape area measurement process is being developed through a stakeholder workgroup process with periodic public meetings.

- Residential indoor water use: Per CWC § 10609.4.(a), "(1) Until January 1, 2025, the standard for indoor residential water use shall be 55 gallons per capita daily. (2) Beginning January 1, 2025, and until January 1, 2030, the standard for indoor residential water use shall be the greater of 52.5 gallons per capita daily or a standard recommended pursuant to subdivision (b). (3) Beginning January 1, 2030, the standard for indoor residential water use shall be the greater of 50 gallons per capita daily or a standard recommended pursuant to subdivision (b). (3) Beginning January 1, 2030, the standard for indoor residential water use shall be the greater of 50 gallons per capita daily or a standard recommended pursuant to subdivision (b)." While the legislation appears to be clear on the method to calculate the indoor residential water use component, the SWRCB has begun the California Environmental Quality Act (CEQA) process for the new water use objective requirements and has expressed concern that using the 55 gallons per capita per day (GPCD) number in the legislation will constitute "backsliding" and thus will need to be ratcheted down.
- Water loss: Per CWC § 10608.34.(i), "No earlier than January 1, 2019, and no later than July 1, 2020, the board shall adopt rules requiring urban retail water suppliers to meet performance standards for the volume of water losses. In adopting these rules, the board shall employ full life-cycle cost accounting to evaluate the costs of meeting the performance standards. The board may consider establishing a minimum allowable water loss threshold that, if reached and maintained by an urban water supplier, would exempt the urban water supplier from further water loss reduction requirements." The SWRCB is developing a complicated cost-benefit analysis methodology that would need to be conducted by retailers in order to determine what water loss controls are deemed cost-effective and thus required to be implemented. Water retailers and the California Municipal Utilities Association are advocating for an alternative methodology. The implementation of these requirements has been delayed beyond the 1 July 2020 deadline.
- **CII:** Rather than developing a water volume-based standard for the CII sector, DWR was tasked with developing a set of performance standards through a workgroup process to increase water efficiency, per CWC § 10609.10, with adoption of these performance measures by 30 June 2022. Based on this process, DWR has determined that it is impossible to set such standards today, but retailers will be required to report on progress towards key actions related to potential future standards, such as conversion of mixed CII meters to dedicated irrigation meters, performance of water audits for CII accounts, development of water management plans for CII accounts, detailed classification of CII accounts by industry, etc. The specific actions that retailers will be required to report are not yet known.
- Recycled Water Use: In previous UWMPs, calculations of SB X7-7 baselines, targets, and gross water use for compliance were based only on potable water use, and thus the use of recycled water to offset potable water use was an effective method to help retailers conserve potable water and meet their SB X7-7 targets. However, under CWC § 10609.(b)(2)(F), the benefit of recycled water for compliance with annual water use objectives is much more limited: "Provides a bonus incentive for the amount of potable recycled water used the previous year when comparing the previous year's water use with the urban water use objective, of up to 10 percent of the urban water use objective." Thus, adoption and expansion of recycled water use only provides a compliance benefit if it constitutes direct potable reuse, indirect potable reuse, or reservoir



augmentation (CWC § 10608.12.(o)).

#### 2.2.2. <u>Supply Reliability</u>

• Retailers will be required to develop procedures to conduct annual water supply and demand assessments to determine its water supply reliability for the current year and one dry year and to conduct these assessments annually beginning in 2022 (CWC § 10632(a)(2)). These procedures are required to include the following (emphasis added):

(A) The **written decision making process** that an urban water supplier will use each year to determine its water supply reliability.

(B) The key data inputs and assessment methodology used to evaluate the urban water supplier's water supply reliability for the current year and one dry year, including all of the following:

(i) Current year unconstrained demand, <u>considering weather, growth, and other</u> <u>influencing factors</u>, such as policies to manage current supplies to meet demand objectives in future years, as applicable.

(ii) Current year available supply, considering <u>hydrological and regulatory conditions</u> <u>in the current year and one dry year</u>. The annual supply and demand assessment may consider more than one dry year solely at the discretion of the urban water supplier.

(iii) Existing infrastructure capabilities and plausible constraints.

(iv) <u>A defined set of locally applicable evaluation criteria</u> that are consistently relied upon for each annual water supply and demand assessment.

(v) A description and **<u>quantification of each source</u>** of water supply.

In addition, the requirement to analyze supply reliability for a period of multiple consecutive drought years has been extended from a 3-year period to a 5-year period, per CWC §10631(f) and §10635(a). Specifically, retailers are now required to "compare the total water supply sources available to the water supplier with the long-term total projected water use over the next 20 years, in five-year increments, for a normal water year, a single dry water year, and a drought lasting five consecutive water years."

#### 2.2.3. <u>Water Shortage Contingency Plans</u>

The new regulations also add new requirements related to drought planning and Water Shortage Contingency Plans (WSCPs):

Retailers will now be required to conduct a drought risk assessment (DRA) as part of their UWMPs to assess water supply reliability (or vulnerability) for a period of drought lasting <u>five consecutive</u> <u>water years</u> (defined by CWC § 10612 as "the driest five-year historical sequence for the agency's water supply"),<sup>2</sup> starting from the year following that of the UWMP, and to compare water supplies (assessing each source of supply separately) with total projected water use (CWC § 10635(b))

<sup>&</sup>lt;sup>2</sup> While the corresponding Water Supply Assessment (WSA) regulations have not been updated to require analysis of a five-year period, retailers should consider including a five-year drought period in their supply reliability assessment in any new WSAs.



during that period. The DRA five-year period for this 2020 UWMP is 2021-2025. During the 10 March 2020 workshop, DWR indicated that retailers will be expected to identify supply and demand on a monthly basis for this purpose, although it is noted that this does not appear to be an explicit requirement of the regulations.

- Per CWC § 10632.5 retailers' WSCPs "shall include a seismic risk assessment and mitigation plan to assess the vulnerability of each of the various facilities of a water system and mitigate those vulnerabilities" and a water supplier may submit "a copy of the most recent adopted local hazard mitigation plan or multihazard mitigation plan under the federal Disaster Mitigation Act of 2000 (Public Law 106-390) if the local hazard mitigation plan or multihazard mitigation plan addresses seismic risk."
- WSCPs will be required to use "Six standard water shortage levels corresponding to progressive ranges of up to 10, 20, 30, 40, and 50 percent shortages and greater than 50 percent shortage," or to provide a "cross-reference relating its existing categories to the six standard water shortage levels."



#### **3. WATER USE CHARACTERISTICS**

This section describes historical water use by customers within the District, including changes in use observed during and after the historic 2014 - 2016 drought, changes in average per account water use over time, and estimates of indoor and outdoor water use, based on data provided by the District. This information is used to provide context and background to support the projections of future demands (Section 4) and estimates of potential conservation program benefits (Section 6).

#### **3.1.** Historical Total and Per Capita Water Use

**Table 3-1** summarizes the District's historical water use, service area population, and per capita water use for the years 2010 through 2019 (Marin Municipal Water District, 2020). Water use is described both in terms of total water produced and average per capita water use. It should be noted that the per capita water use for purposes of comparing water use to SB X7-7 water conservation targets may be different, due to the prescriptive methodology DWR has established for determining an agencies compliance population and total water use<sup>3</sup>. SB X7-7 compliance will need to be separately addressed by the District's 2020 UWMP.

Total water use, including potable, raw, and recycled water,<sup>4</sup> ranged from 23,680 acre-feet per year (AFY) to 29,847 AFY over this period. Total per capita water use (i.e., including potable, raw, and recycled water use) ranged from 110 GPCD to 142 GPCD.

Both the total and per capita water use declined from 2013 through 2015, likely influenced by the historic drought conditions, mandatory state-wide restrictions in urban water use imposed by the SWRCB, and local drought response. Total and per capita water use has remained lower than pre-drought conditions, with an increase from 2016 through 2019, indicating a degree of rebound following the drought.

Historical water use by customer sector is provided in **Table 3-2**. The single family residential (SFR) sector comprises the largest proportion of the District's total water use (i.e., 53% in 2019). By comparison, in 2019, multi-family residential (MFR) accounts comprised 12% of total water use; business/industrial accounts comprised 10% of total water use; the combined agricultural/irrigation, raw water, and recycled water accounts comprised 8.5% of total water use; and institutional accounts comprised 5.4% of total water use. In 2019, non-revenue water was estimated to be 11% of potable water demand.

#### **3.2.** Historical Average Water Use Per Account

The total number of accounts varies over time due to growth and development within the District and shifts in land use.

<sup>&</sup>lt;sup>3</sup> In previous years, DWR has preferred that the DWR population tool be used for purposes of estimating service area population for purposes of SB X7-7 compliance, so that a uniform method is applied across retailers. The updated 2020 DWR population tool has not yet been released. The population reported by this tool may therefore be somewhat different than the population estimates used herein.

<sup>&</sup>lt;sup>4</sup> Water use data is per District-provided billing data. The recycled water system is supplemented with potable water to meet demands, as necessary. Recycled water use discussed herein reflects all water served through the recycled water system.



The total number of accounts by customer sector for the 2010 to 2019 period is shown in **Table 3-3**, including a pie chart illustrating the relative proportion of accounts (Marin Municipal Water District, 2020). The SFR sector comprised the highest proportion of accounts in 2019 (85%), followed by the MFR sector (6.3%), business/industrial sector (5.3%), agricultural/irrigation sector (1.4%), recycled water sector (0.49%), and institutional sector (0.38%). From 2010 to 2019, most sectors experienced between 0.5% and 0.9% total growth. However, business/industrial accounts decreased by 0.61% over the same time period, while agricultural/irrigation accounts increased by 2.4%.

Average water use per account is presented in **Table 3-4a**. For most sectors, per account water usage has followed the same general trends over time as total water use in the District (per **Table 3-1**).

**Table 3-4b** presents average water use for the residential sectors normalized by number of dwelling units. SFR accounts, on average, use approximately 100% to 130% more water per dwelling unit than MFR accounts. It should be noted that many larger MFR developments have dedicated irrigation meters.

#### 3.3. Change in Residential Water Use Pre- and Post-Drought

Over time, customer water use becomes more efficient due to participation in conservation programs, passive savings,<sup>5</sup> and other behavioral or cultural changes. The more efficient customers become, the less opportunity there is for customers to save more water, which is referred to as "demand hardening." The SFR sector comprises the largest proportion of the District's total water use (approximately 53% in 2019). Therefore, in order to observe demand hardening over time, histograms illustrating the distribution of water use by SFR customers for three separate years (2010, 2013, and 2019) are shown in **Figure 3-1**.

The median SFR account water use has shifted from 207 GPD to 227 GPD between 2010 and 2013, reflecting a 9.7% increase in median water use. Following the drought, water use was reduced to a median of 193 GPD in 2019, reflecting a 17% reduction from 2013 water use. In 2010, the middle 50% of accounts used 131 GPD to 307 GPD. In 2019, this range has slightly broadened, with the middle 50% of accounts using between 117 GPD and 301 GPD. Based on this (and taken with the **Table 3-5** results discussed below), it appears that customers are continuing to increase their efficiency, which is expected to be a combination of both passive and active savings, as well as effects of the drought. Water savings achieved during drought conditions are typically driven by behavioral changes, rather than device changeouts (AWE, 2015). Given the limited rebound observed since the drought (**Table 3-4a**), it may be that behavioral changes during the drought have resulted in permanent changes in customers' water use.

#### **3.4.** Residential Water Use by Dwelling Unit and Age of Construction

It is commonly assumed that new residential construction is inherently more water efficient than older construction due to changes in plumbing codes and the increased efficiency of water using devices available on the market today. However, in some areas it has been observed that newer construction can actually have higher rates of water use, which is an important consideration when evaluating future water

<sup>&</sup>lt;sup>5</sup> Passive savings refers to the water savings associated with the natural replacement of older toilets, showerheads, clothes washers, and other water using appliances with newer high efficiency devices that are available due to both market shifts and increasing efficiency mandated by the building code, plumbing code, and other regulatory requirements.



demands associated with new development. In order to evaluate water use relative to the age of residential construction within the District, water use by SFR and MFR accounts is summarized in **Table 3-5** by units constructed: (1) prior to 1994, (2) from 1994 through 2009, and (3) 2010 and later.

Water use by SFR units constructed from 1994-2009 had on average 18% higher water use than units constructed 2010 and later and 43% higher water use than units constructed prior to 1994. Water use for buildings constructed 2010 and later had on average 22% higher water use than pre-1994 construction. Given this, as discussed in Section 4.3.1, a water demand factor representative of newer construction (1994 and later) is used as the basis for demand projections for new SFR accounts.

MFR units appear generally more consistent across construction age than SFR units. Newer construction (2010 and later) shows a larger range in water use across the time period but is likely driven by the relatively low number of accounts in that age group. Given this general consistency, the demand projections for new MFR accounts discussed in Section 4.3.1 are based on all MFR units regardless of construction age.

#### **3.5. Estimated Indoor and Outdoor Water Use**

When designing and estimating the benefits of potential water conservation programs, it is important to understand the relative proportion of water use that is used indoors versus outdoors.

As shown in the first chart in **Table 3-6**, potable water use within the District varies seasonally, and water use in the summer is two to three times greater than water use during the winter. This seasonality is typically driven by increased irrigation needs in the summer, as compared to the more limited irrigation water use during the wetter and cooler winter months. The second chart in **Table 3-6** shows the seasonality of recycled water use, which is used primarily for irrigation.<sup>6</sup> Based on the recycled water use patterns, irrigation rates appear to be nearly zero during winter months, confirming that it is reasonable and conservative to assume that minimal irrigation with potable water occurs during winter months. This is a high-level estimate of indoor and outdoor water use, which errs on the side of estimating higher indoor water use.

Given the water use patterns presented in **Table 3-6**, the minimum average daily water use during winter months (November – April due to bi-monthly billing data) was used to estimate the indoor water use for all non-irrigation customer sectors. The results of this estimate are shown in **Table 3-7**. Approximately 64% of all potable water use (excluding potable water served through the recycled water system) within the District is estimated to be indoor use, and 36% to be outdoor water use. For SFR users (i.e., the largest water using sector within the District), approximately 59% of water use is estimated to be indoor, and 41% outdoor water use. Total water use (including recycled and raw water) is approximately 61% indoor water use and 39% outdoor use.

Aside from the "other," raw water, irrigation, and recycled water sectors (presumed 100% outdoor water use), the SFR sector is estimated to have the highest proportion of outdoor water use at 41%, followed by business/industrial at 17%, institutional at 17%, and MFR at 10%. It should be noted that landscape areas

<sup>&</sup>lt;sup>6</sup> Recycled water is also used for toilet flushing, car washes, cooling towers, and commercial laundry facilities and averages about 100,000 gallons per day of demand during winter months.



for larger multi-family developments tend to have dedicated irrigation accounts. Further, some industries within the CII sector, such as restaurants and manufacturing, may also experience some degree of seasonality in indoor use, with increased business and production during summer months. Thus, these should be considered high-level estimates of indoor and outdoor use proportions.

### Table 3-1Water Use and PopulationMarin Municipal Water District

Year (a)	Potable Water Use (Sales) (AFY) (b)	Recycled Water Use (AFY) (b)	Raw Water Use (AFY) (b)	Non- Revenue Water (AFY) (c)	Total Water Use (AFY)	Service Area Population (d)	Per Capita Potable Water Use (GPCD) (e)	Per Capita Total Water Use (GPCD) (e)
2010	22,597	514	258	2,872	26,241	183,716	110	128
2011	22,340	432	220	2,916	25,908	185,389	108	125
2012	23,864	507	301	2,728	27,400	187,089	114	131
2013	25,458	684	351	3,354	29,847	188,218	121	142
2014	22,435	579	323	2,088	25,425	190,267	105	119
2015	20,624	520	304	2,279	23,727	191,575	96	111
2016	20,584	491	301	2,304	23,680	192,402	96	110
2017	21,847	512	310	2,708	25,377	192,328	101	118
2018	22,533	522	309	2,511	25,875	192,277	105	120
2019	22,723	0	164	2,765	25,652	192,138	106	119



#### Abbreviations:

AFY = acre-feet per year

- DOF = Department of Finance
- GPCD = gallons per capita per day

#### Notes:

- (a) Data are presented on a calendar year basis.
- (b) Water use data based on customer sales, per Reference 2. The recycled water system is supplemented with potable water to meet demands, as necessary. Recycled water use shown here reflects all water served through the recycled water system.
- (c) Estimated non-revenue water (potable) per Table 3-2.

#### Table 3-1 Water Use and Population

#### Marin Municipal Water District

- (d) Population estimates are adjusted from DOF county estimates using a conversion factor provided by the district, per Reference 3. 2016-2019 estimates were updated using 2020 DOF population estimates, per Reference 1.
- (e) Per capita water use is calculated by dividing the annual water use by service area population and the number of days in a year.

#### References:

- 1. DOF, 2020. California Department of Finance Demographic Research Unit, Population Estimates for Cities, Counties, and the State, 2011-2020, with 2010 Benchmark, Report E-4, released on 1 May 2020.
- 2. Marin Municipal Water District, 2020a. 2010-2019 Urban Water Management Plan Water Use Data, provided by Marin Municipal Water District on 9 July 2020.
- 3. Marin Municipal Water District, 2020b. MMWD Population 2019 Demand Analysis.xls, provided by Marin Municipal Water District on 9 April 2020.

### Table 3-2Water Use by Customer SectorMarin Municipal Water District

Water Use Sector				W	ater Use (AF	Y) (a) (b)				
water use sector	2010	2011	2012	2013	2014	2015	2016	2017	2018 13,886 3,065 2,671 1,365 1,417 38 309 91 522 9,7% 2,511 25,875	2019
Single Family Residential (c)	13,747	13,482	14,672	15,525	13,824	12,403	12,419	13,337	13,886	13,579
Multi-Family Residential (d)	3,185	3,244	3,329	3,367	3,128	2,984	2,946	3,004	3,065	3,063
Business/Industrial	2,716	2,651	2,788	2,867	2,655	2,577	2,583	2,628	2,671	2,634
Institutional	1,639	1,571	1,467	1,523	1,406	1,311	1,295	1,374	1,365	1,386
Agricultural/Irrigation	1,185	1,179	1,391	1,612	1,330	1,230	1,248	1,369	1,417	1,348
Other (e)	40	32	45	44	36	34	28	35	38	50
Raw Water	258	220	301	351	323	304	301	310	309	164
Recycled Water System (f)										
Potable Water	86	181	171	81	56	85	65	101	91	661
Recycled Water	514	432	507	684	579	520	491	512	522	0
Non revenue Water (Detable) (g)	11%	11%	10%	11%	8.2%	9.6%	9.7%	11%	9.7%	11%
Non-revenue water (Potable) (g)	2,872	2,916	2,728	3,354	2,088	2,279	2,304	2,708	2,511	2,765
Total Water Use	26,241	25,908	27,400	29,409	25,425	23,727	23,680	25,377	25,875	25,652
Total Potable Water Use	25,469	25,256	26,592	28,374	24,523	22,903	22,888	24,555	25,044	25,488



### Table 3-2Water Use by Customer SectorMarin Municipal Water District

#### Abbreviations:

AFY = acre-feet per year

#### Notes:

- (a) Data are presented on a calendar year basis.
- (b) Water use by sector per Reference 1.
- (c) Single-family residential water use includes dedicated single-family irrigation accounts.
- (d) Multi-family residential includes duplexes and 3-10+ unit apartments.
- (e) "Other" includes fireline and hydrant sectors.
- (f) The recycled water system is supplemented with potable water to meet demands, as necessary. The recycled water plant was non-operational in 2019 to allow for infrastructure upgrades.
- (g) Non-revenue water was calculated by subtracting total potable water use (including recycled water system makeup water) from total potable water production,

#### References:

- 1. Marin Municipal Water District, 2020a. 2010-2019 Urban Water Management Plan Water Use Data, provided by Marin Municipal Water District on 9 July 2020.
- 2. Marin Municipal Water District, 2020b. Production data provided by Marin Municipal Water District via email on 14 September 2020.

# Table 3-3Number of Accounts by Customer SectorMarin Municipal Water District

Water Use Sector				Num	nber of A	ccounts (	a) (b)			
Water Ose Sector	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019
Single Family Residential (c)	51,241	51,242	51,286	51,327	51,421	51,474	51,511	51,518	51,558	51,593
Multi-Family Residential (d)	3,778	3,771	3,773	3,779	3,788	3,801	3,802	3,798	3,797	3,797
Business/Industrial	3,254	3,246	3,247	3,257	3,249	3,246	3,247	3,245	3,249	3,234
Institutional	225	225	226	227	227	230	228	227	228	227
Agricultural/Irrigation	838	833	825	842	857	850	845	852	853	858
Other (e)	479	472	456	447	466	442	460	491	495	490
Raw Water	2	2	2	2	2	2	2	2	2	1
Recycled Water	300	302	301	303	302	303	302	297	297	294
Total Accounts	60,117	60,093	60,116	60,184	60,312	60,348	60,397	60,430	60,479	60,494



### Table 3-3Number of Accounts by Customer Sector

Marin Municipal Water District

#### Notes:

- (a) Data are presented on a calendar year basis.
- (b) Number of accounts by sector per Reference 1. Number of accounts reflects active accounts for each year.
- (c) Number of single-family residential accounts does not include dedicated single-family irrigation accounts.
- (d) Multi-family residential includes duplexes and 3-10+ unit apartments.
- (e) "Other" includes fireline and hydrant sectors.

#### References:

1. Marin Municipal Water District, 2020. 2010-2019 Urban Water Management Plan Water Use Data, provided by Marin Municipal Water District on 9 July 2020.

#### Table 3-4a Per Account Water Use by Customer Sector Marin Municipal Water District

Water Use Sector				v	Vater Use per Ac	count (GPD) (a) (	b)			
water use sector	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019
Single Family Residential (c)	239	235	255	270	240	215	215	231	240	235
Multi-Family Residential (d)	752	767	787	795	737	700	691	706	720	720
Business/Industrial	745	729	766	785	729	708	710	722	734	727
Institutional	6,500	6,228	5,789	5,985	5,527	5,085	5,067	5,399	5,341	5,449
Agricultural/Irrigation	1,261	1,263	1,504	1,708	1,385	1,291	1,317	1,433	1,482	1,402
Other (e)	75	61	89	88	68	69	55	63	68	92
Raw Water	115,184	98,107	134,286	156,569	144,270	135,628	134,245	138,449	137,692	146,440
Recycled Water System (f)	1,784	1,811	2,010	2,252	1,876	1,781	1,642	1,841	1,841	2,006



(C 200,000 Mater Nse (CbD) 150,000 50,000 0 Raw Water

2010 2011 2012 2013 2014 2015 2015 2015 2016 2017 2018 2019



(Figure 1) (Control (Contro) (Contro) (Contro) (Contro) (Contro) (Contro) (Contro) (

Abbreviations:

GPD = gallons per day

#### Notes:

- (a) Data are presented on a calendar year basis.
- (b) Water use and number of accounts by sector per Tables 3-2 and 3-3.
- (c) Single-family residential use includes dedicated use by single-family irrigation accounts.
- (d) Multi-family residential includes duplexes and 3-10+ unit apartments.
- (e) "Other" includes fireline and hydrant sectors.
- (f) The recycled water system is supplemented with potable water to meet demands, as necessary. Potable make-up water volume is shown in Table 3-2.

Recycled Water System (f)

2010 2011 2012 2013 2014 2015 2015 2015 2016 2017 2018 2019

Xater Use (GPD) 2,000 (GPD) 1,500 500 0 0

# Table 3-4b Per Dwelling Unit Water Use for Residential Sectors Marin Municipal Water District

Water Use Sector	Water Use per Dwelling Unit (GPD/DU) (a) (b)												
Water Use Sector	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019			
Single Family Residential	246	240	261	275	245	221	220	237	246	241			
Multi-Family Residential (c)	116	116	121	123	113	105	104	108	109	108			





#### Abbreviations:

DU = dwelling unit

GPD = gallons per day

#### Notes:

- (a) Data are presented on a calendar year basis.
- (b) Per dwelling unit water use is calculated based on the number of residential dwelling units per account provided in customer billing data. Data included in this analysis is limited to accounts that received six bills in the specified year.
- (c) Multi-family residential includes duplexes and 3-10+ unit apartments.

Table 3-5Residential Water Use by Age of ConstructionMarin Municipal Water District

Construction Age		Ave	rage W	ater Us	e (GPD	per Dw	velling l	Jnit) (a	) (b)		Number of
Construction Age	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	Accounts, 2019
Single Family Resident	ial										
Pre-1994	234	230	249	262	232	208	208	223	232	227	39,003
1994-2009	323	315	348	370	338	305	304	328	339	327	6,776
2010 and Later	276	252	278	295	279	267	270	287	295	306	732
Multi-Family Residenti	al										
Pre-1994	116	117	121	123	112	105	104	107	109	109	2,466
1994-2009	111	110	120	121	115	114	104	121	111	114	209
2010 and Later	132	174	200	185	150	99	81	97	105	115	33





## Table 3-5 Residential Water Use by Age of Construction Marin Municipal Water District

#### Abbreviations:

-- = not available

DU = dwelling unit

GPD = gallons per day

#### Notes:

- (a) Data are presented on a calendar year basis.
- (b) Average water use per dwelling unit is shown for residential sectors based on billing data, per Reference 2. Accounts included in this analysis are limited to that for which construction year is available, based on Marin County Assessor data, and that received 6 bills in the specified year per Reference 1.

#### References:

- 1. Marin County, 2020. County Wide Parcel Data ConservationJan2020.gdb, provided by Marin Municipal Water District on 13 February 2020.
- 2. Marin Municipal Water District, 2020. 2010-2019 Urban Water Management Plan Water Use Data, provided by Marin Municipal Water District on 9 July 2020.

# Table 3-6Monthly Water UseMarin Municipal Water District

		Monthly Water Use (AF) (a)												
Month	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019				
Potable Wate	er Use													
January	1,102	1,050	1,197	991	1,316	1,034	989	956	1,079	1,052				
February	1,591	1,616	1,829	1,664	1,948	1,630	1,436	1,438	1,684	1,508				
March	902	958	1,077	1,064	1,041	1,046	870	867	1,009	859				
April	1,602	1,619	1,643	1,926	1,595	1,859	1,474	1,408	1,596	1,440				
May	1,207	1,307	1,251	1,609	1,273	1,309	1,177	1,119	1,222	1,219				
June	2,246	2,375	2,577	2,889	2,430	2,157	2,262	2,286	2,352	2,211				
July	1,898	1,796	2,083	2,146	1,935	1,575	1,836	1,994	1,945	1,804				
August	3,162	2,882	3,252	3,127	2,846	2,360	2,797	2,928	2,928	2,862				
September	2,246	2,111	2,269	2,175	1,908	1,733	2,029	2,121	2,084	2,152				
October	3,058	2,864	2,950	2,991	2,583	2,387	2,530	2,854	2,730	2,878				
November	1,633	1,593	1,580	1,908	1,559	1,481	1,404	1,733	1,657	1,809				
December	1,790	1,914	1,889	2,361	1,855	1,888	1,646	1,950	2,071	2,186				
Recycled Wa	ter Syste	m Use (b	)											
January								<sup> </sup>						
February	11	10	32	16	40	16	12	11	24	14				
March								, <sup> </sup>						
April	19	21	23	66	26	72	30	16	27	14				
May			<sup> </sup>					·						
June	125	150	156	205	163	132	149	151	160	138				
July			<sup> </sup>					·						
August	241	202	240	202	209	177	207	225	220	220				
September			<sup> </sup>					·						
October	194	163	176	192	163	159	157	193	163	192				
November								, <sup> </sup>						
December	17	43	25	73	46	48	22	49	62	82				



### Table 3-6Monthly Water UseMarin Municipal Water District



#### Abbreviations:

-- = not availableAF = acre-feetMMWD = Marin Municipal Water District

#### Notes:

- (a) Monthly potable and recycled water use per Reference 1. Customers are billed on a bi-monthly basis, and data are presented based on billing cycle. Due to a limitation of the MMWD billing data system, monthly water use data shown is between approximately 0.3% and 0.4% less than the actual total water use shown in Table 3-2.
- (b) The recycled water system is supplemented with potable water to meet demands, as necessary. Recycled water use shown here reflects all water served through the recycled water system. Potable make-up water volume is shown in Table 3-2.

#### References:

1. Marin Municipal Water District, 2020. 2010-2019 Urban Water Management Plan Water Use Data, provided by Marin Municipal Water District on 9 July 2020.

### Table 3-7 Estimated Indoor and Outdoor Water Use Marin Municipal Water District

		20	17			20	18			20	19		Avera	ge Pct.
Water Use Sector (a)	Indoor Water Use (AFY)	Outdoor Water Use (AFY)	Pct. Indoor	Pct. Outdoor	Indoor Water Use (AFY)	Outdoor Water Use (AFY)	Pct. Indoor	Pct. Outdoor	Indoor Water Use (AFY)	Outdoor Water Use (AFY)	Pct. Indoor	Pct. Outdoor	Indoor Use	Outdoor Use
Single Family Residential	7,378	5,892	56%	44%	8,984	4,835	65%	35%	7,470	6,046	55%	45%	59%	41%
Multi-Family Residential	2,674	321	89%	11%	2,782	274	91%	9%	2,776	281	91%	9%	90%	10%
Business/Industrial	2,121	499	81%	19%	2,277	388	85%	15%	2,174	451	83%	17%	83%	17%
Institutional	1,144	228	83%	17%	1,153	212	85%	15%	1,112	273	80%	20%	83%	17%
Agricultural/Irrigation	0	1,369	0%	100%	0	1,417	0%	100%	0	1,348	0%	100%	0%	100%
Other (b)	0	35	0%	100%	0	38	0%	100%	0	50	0%	100%	0%	100%
Total (Potable)	13,318	8,344	61%	39%	15,197	7,164	68%	32%	13,532	8,450	62%	38%	64%	36%
Raw Water	0	310	0%	100%	0	309	0%	100%	0	164	0%	100%	0%	100%
Recycled Water System (c)	0	647	0%	100%	0	657	0%	100%	0	661	0%	100%	0%	100%
Total (Potable, Raw & Recycled)	13,318	9,300	59%	41%	15,197	8,129	65%	35%	13,532	9,274	59%	41%	61%	39%

#### 2019 Estimated Indoor and Outdoor Water Use by Sector (d)



Abbreviations:

AFY = acre-feet per year

Pct. = Percentage

MMWD = Marin Municipal Water District

### Table 3-7 Estimated Indoor and Outdoor Water Use

Marin Municipal Water District

#### Notes:

- (a) Due to a limitation of the MMWD billing data system, data shown is approximately 0.4% less than the actual total water use shown in Table 3-2.
- (b) "Other" includes fireline and hydrant sectors.
- (c) The recycled water system is supplemented with potable water to meet demands, as necessary. Recycled water use shown here reflects all water served through the recycled water system. Potable make-up water volume is shown in Table 3-2.
- (d) The minimum average daily water use from November through April was used to estimate indoor water use for all non-irrigation customer sectors. This method is used to assess relative proportion of indoor and outdoor use, and conservatively errs on the side of estimating more indoor water use, so that the potential for outdoor water savings is not over-estimated. Raw water, recycled water, agricultural/irrigation and "other" sectors are considered outdoor use only.

#### References:

1. Marin Municipal Water District, 2020. 2010-2019 Urban Water Management Plan Water Use Data, provided by Marin Municipal Water District on 9 July 2020.





#### 4. WATER DEMAND PROJECTIONS

The purpose of this section is to document the basis, methodology, and resulting projected demands for the District through 2045. As described in more detail below, the future water demands for the District were estimated by:

- 1. Applying an estimated growth rate to accounts within each water use sector based on projected population and employment growth rates,
- 2. Identifying known planned developments within the District to verify that account growth projections consider all anticipated growth,
- 3. Evaluating and selecting water demand factors for each water use sector based on review of recent average per account water use representing three scenarios,
- 4. Estimating future passive savings using the Alliance for Water Efficiency (AWE) Water Conservation Tracking Tool (AWE model), and
- 5. Calculating estimated future water demand that incorporates the anticipated account growth, water demand factors, and estimated future passive water savings.

This methodology is consistent with California Water Code (CWC) § 10631(d)(4)(A), which requires that "Water use projections, where available, shall display and account for the water savings estimated to result from adopted codes, standards, ordinances, or transportation and land use plans identified by the urban water supplier, as applicable to the service area." The assumptions used as the bases for demand projections were developed in close coordination with the District and reflect a land-use based approach consistent with the District's community planning.

#### 4.1. Basis for Account Growth Projections

Water demand increases as new accounts are added to the system, among other factors. In order to estimate how accounts will grow within the District, recent historical account growth within the District was considered, as well as projected future growth in population and employment. As described below, it was assumed, that depending on the customer sector, the number of accounts will grow at the same *rate* as the projected population or employment growth.

**Table 4-1** presents historical population and 2018 Association of Bay Area Governments (ABAG) Plan Bay Area Projections 2040 population and employment growth projections for the District, in context with recent historical population estimates.<sup>7</sup>

**Table 4-2**, identifies which growth projection was applied to each potable water use sector (population or employment) at the District's direction, identifies the average annual growth rate in accounts observed

<sup>&</sup>lt;sup>7</sup> Several growth projections were evaluated as potential bases for growth assumptions, including previous 2013 ABAG Plan Bay Area Projections (ABAG, 2013), ABAG Plan Bay Area Projections 2040 (ABAG, 2018), and 2020 Department of Finance (DOF) Total Estimated and Projected Population for California and Counties (DOF, 2020). The DOF (2020) projections are only available at the County-wide level and show a decline in population over the planning horizon and given the recent historical growth observed in the District, are not considered appropriately conservative for planning purposes. Although anticipated to be released in 2020, updated ABAG projections are not yet available. Therefore ABAG (2018) projections were selected as the basis for growth assumptions for the District.



within the District (based on data presented in **Table 3-3**), and the associated average annual growth rate projected by ABAG (2018). With the exception of agricultural/irrigation accounts, recent historical growth rates have been lower than the projected growth rates by ABAG (2018). Recycled water actually decreased by 0.22% over the recent historical time period, and raw water accounts decreased by half.<sup>8</sup> At the District's direction, ABAG (2018) projected growth rates were used and are considered to be reasonably conservative for planning purposes.

The planning horizon for the 2020 UWMP is 2045; however, the ABAG (2018) projections extend only through 2040. For purposes of demand projections, it is therefore assumed that the projected growth rates from 2035 through 2040 extend through 2045.

	Basis for	Average Annu	al Growth (a)	
Water Use Sector	Account Growth	Historic (2010-2019)	ABAG 2018 (2020-2045)	
Single Family Residential				
Existing Accounts	population	0.076%	0.34%	
New Accounts				
Multi-Family Residential	population	0.056%	0.34%	
Business/Industrial	employment	-0.068%	0.15%	
Institutional	employment	0.10%	0.15%	
Agricultural/Irrigation	employment	0.27%	0.15%	
Other	employment	0.26%	0.15%	
Raw Water <sup>8</sup>	employment	-50%	0.15%	
Recycled Water	employment	-0.22%	0.15%	

### Table 4-2 Historical and Projected Account Growth Rate by Customer Sector

#### Abbreviations:

ABAG = Association of Bay Area Governments

#### Notes:

(a) Growth is presented on an average annual basis over the indicated period. When applied to account growth, the specific growth rate between each 5-year period, per ABAG (2018) was applied.

#### 4.2. Change in Number of Accounts based on Projected Growth

**Table 4-3**, presents the projected increase in accounts over the planning horizon as well as the incremental increase in accounts from 2019 per sector. There are no known major developments within the District's service area, and thus the projected increase in accounts reflects the assumed level of growth described in Section 4.1.

#### 4.3. Water Demand Factors

Water use rates are influenced by a variety of factors, including weather, economic recession, and state and local regulations, among other drivers. Given this, selecting a "representative" baseline year is

<sup>&</sup>lt;sup>8</sup> The number of raw water accounts dropped from 2 to 1 over this time period.



important to developing the land-use based water demand factors to estimate baseline water use by existing customers, which can then be extrapolated and applied to future growth within the District.

Water demand factors based on historical use within the District were used as the basis of future demand projections for potable water accounts, considering in particular the range of water use associated with pre-drought conditions, post-drought conditions, and a midpoint scenario that assumes water use partially rebounds to pre-drought conditions. **Table 3-2** provides historical water use by sector within the District. To more fully capture total water use within the District, non-revenue water is estimated as a percentage of water production as discussed in 4.3.2.

#### 4.3.1. Potable, Raw, and Recycled Water

As shown in **Table 4-4**, the District evaluated a range of water demand factors for each water use sector using three water use scenarios, based primarily on recent historical average per account water use for selected time periods,<sup>9</sup> representing pre-drought water use rates, post-drought water use rates, and a partial rebound to pre-drought water use rates. Specifically:

- 1. *Pre-drought demand factors* based on the maximum per account water use by sector for 2011 through 2013 (**Table 3-4a**), generally representing higher water use before drought restrictions were put in place.
- 2. *Post-drought demand factors* based on the maximum per account water use by sector for 2017 through 2019 (**Table 3-4a**), generally representing lower water use than pre-drought conditions but with some amount of rebound.
- 3. *Partial rebound demand factors* estimated as the midpoint of the pre-drought and post-drought demand factors, representing an average of the two scenarios.

<sup>&</sup>lt;sup>9</sup> Given the results discussed in Section 3.4, water demand factors for new SFR accounts are based on water use for homes constructed in 1994 and later.



	Water Den	nand Factor (GPI	D/account)
Water Use Sector	Pre-Drought (2011-2013)	Partial Rebound	Post-Drought (2017-2019)
Single Family Residential			
Existing Accounts	270	255	240
New Accounts	365	350	335
Multi-Family Residential	795	758	720
Business/Industrial	785	759	734
Institutional	6,228	5,839	5,449
Agricultural/Irrigation	1,708	1,595	1,482
Other	89	90	92
Raw Water	156,569	147,509	138,449
Recycled Water	2,252	2,129	2,006

 Table 4-4

 Potential Water Demand Factors Considered

Abbreviations:

GPD = gallons per day

As shown in **Table 4-5**, below, for purposes of developing the District's 2045 demand projections, the District directed EKI to apply partial rebound demand factors to residential and raw water sectors and pre-drought demand factors to all other sectors.

Water Use Sector	Water Demand Factor (GPD/account)	Basis for Demand Factor
Single Family Residential		
Existing Accounts	255	Partial rebound
New Accounts	350	Partial rebound
Multi-Family Residential	758	Partial rebound
Business/Industrial	785	Pre-drought
Institutional	6,228	Pre-drought
Agricultural/Irrigation	1,708	Pre-drought
Other	89	Pre-drought
Raw Water	147,509	Partial rebound
Recycled Water	2,252	Pre-drought

Table 4-5 Selected Water Demand Factors

Abbreviations:

GPD = gallons per day

The recycled water system is supplemented with potable water to meet demands, as necessary. Between 2010 and 2018, potable water was used to meet between 9% and 30% of recycled water system demand; in 2019, the recycled water plant was shut down due to upgrades and 100% of recycled water demand was met by potable water. However, following plant upgrades, it is anticipated that all demand by the recycled



water system will be met by recycled water, thus demand projections are based on total recycled water system use, and do not include a projection of potable make-up water.

#### 4.3.2. Non-Revenue Water (Potable Water System)

Non-revenue water is water that has been produced but not billed, and thus does not generate revenue for the supplier. Non-revenue water includes unbilled authorized uses (such as water for fighting fires and flushing mains) and water losses (including real losses due to distribution system leaks and apparent losses due to metering inaccuracies). For the purposes of this assessment, non-revenue water is estimated as total potable water produced minus the total billed potable water use on an annual basis. As shown in **Table 4-6**, potable non-revenue water is projected to range from 2,758 AFY to 2,777 AFY through 2045, based on the average percentage of non-revenue reported from 2017 to 2019 (10%, see **Table 3-2**).

#### 4.4. Passive Water Savings Estimates

Passive water savings are the water savings associated with the natural replacement of older toilets, showerheads, clothes washers, and other water using appliances with newer high efficiency devices that are available due to both market shifts and increasing efficiency mandated by the building code and other regulatory requirements. The AWE model<sup>10</sup> was used to estimate future passive savings within the District. The AWE model takes into account estimates of historical population, residential building stock, number of accounts, and projected population and account growth to estimate future passive savings. The estimated passive savings are presented in **Table 4-6** and are subtracted from the water demand projected based on the water demand factors described in Section 4.3 above. Passive savings are only applied to potable water use.

#### 4.5. Projected Water Demand Through 2045

Future water demand was projected for each sector based on their respective demand factors, and is shown in **Table 4-6**. Potable and raw water demand is projected to increase to 26,915 AFY in 2045, which is a 4.9% increase over 2019 potable and raw water demand. Recycled water demand is projected to increase to 771 AFY, which is a 17% increase over the 2019 recycled water system demand. Both potable and raw water, and recycled water demand projections are higher than the District's 2015 UWMP demand projections.

<sup>&</sup>lt;sup>10</sup> Alliance for Water Efficiency, Water Conservation Tracking Tool Version 3, released in July 2016.

### Table 4-1 Population and Employment Growth Projections Marin Municipal Water District

Category					Grov	vth Projec	tions					Total Growth	Average Annual
	2015	2016	2017	2018	2019	2020	2025	2030	2035	2040	2045 (c)	Rate 2020-2045	2020-2045
Population													
Historical Population Estimates (a)	191,575	192,402	192,328	192,277	192,138								
2018 ABAG Population Projections (b)						195,360	197,939	201,987	204,750	208,324	211,961	8.5%	0.34%
Employment													
2018 ABAG Employment Projections (b)						98,019	98,822	100,449	101,246	101,474	101,703	3.8%	0.15%



Abbreviations:

-- = not available

ABAG = Association of Bay Area Governments

DOF = California Department of Finance

MMWD = Marin Municipal Water District

#### Table 4-1

#### **Population and Employment Growth Projections**

Marin Municipal Water District

#### Notes:

- (a) Historical population estimates are adjusted from DOF county estimates using a conversion factor provided by the District, per Reference 3. 2016-2020 estimates were updated using 2020 DOF population estimates, per Reference 2.
- (b) 2018 ABAG population and employment projections per Reference 1. Unincorporated county population and employment is adjusted for proportion within the MMWD service area using a conversion factor provided by the District (i.e., 76% of unincorporated population/employment), per Reference 3.
- (c) ABAG 2018 includes projections through 2040. 2045 population and employment projections are calculated based on 2035-2040 growth rates (1.7% and 0.23%, respectively).

#### References:

- 1. ABAG, 2018. Association of Bay Area Governments, Plan Bay Area Projections 2040, released on November 2018.
- 2. DOF, 2020. California Department of Finance Demographic Research Unit, Population Estimates for Cities, Counties, and the State, 2011-2020, with 2010 Benchmark, Report E-4, released on 1 May 2020.
- 3. Marin Municipal Water District, 2020. MMWD Population 2019 Demand Analysis.xls, provided by Marin Municipal Water District on 9 April 2020.

## Table 4-3 Change in Number of Accounts based on Projected Growth Marin Municipal Water District

#### **Projected Number of Accounts**

Water Use Sector	Number of Accounts (a)					
	2025	2030	2035	2040	2045 (b)	
Single Family Residential (c)	52,410	53 <i>,</i> 482	54,214	55,160	56,123	
Multi-Family Residential (d)	3,857	3,936	3,990	4,060	4,130	
Business/Industrial	3,266	3,320	3,346	3,353	3,361	
Institutional	229	233	235	235	236	
Agricultural/Irrigation	866	881	888	890	892	
Other (e)	495	503	507	508	509	
Raw Water	1	1	1	1	1	
Recycled Water	297	302	304	305	306	
Total Accounts	61,422	62,657	63,484	64,512	65,558	

#### **Incremental Increase in Accounts from 2019**

Water Use Sector	Number of Accounts					
	2025	2030	2035	2040	2045	
Single Family Residential (c)	817	1,889	2,621	3,567	4,530	
Multi-Family Residential (d)	60	139	193	263	333	
Business/Industrial	32	86	112	119	127	
Institutional	2	6	8	8	9	
Agricultural/Irrigation	8	23	30	32	34	
Other (e)	5	13	17	18	19	
Raw Water	0	0	0	0	0	
Recycled Water	3	8	10	11	12	
Total New Accounts	928	2,163	2,990	4,018	5,064	

#### **Estimate of Known Planned Development**

Water Use Sector	Number of Accounts (f)				
	2025	2030	2035	2040	2045
Single Family Residential (c)					
Multi-Family Residential (d)					
Business/Industrial					
Institutional					
Agricultural/Irrigation					
Other (e)					
Raw Water					
Recycled Water					
Total New Accounts					

#### Table 4-3 Change in Number of Accounts based on Projected Growth Marin Municipal Water District

Abbreviations:

-- = not available

ABAG = Association of Bay Area Governments

Notes:

- (a) Growth in number of accounts is estimated based on ABAG 2018 projected growth rates for population and employment. Residential sectors are estimated relative to population growth, and growth in all other account types are estimated relative to employment growth.
- (b) ABAG 2018 includes projections through 2040. For the purposes of demand and account projections, it is assumed that the growth rate remains constant from 2036 through 2045.
- (c) Single-family residential includes irrigation.
- (d) Multi-family residential includes duplexes and 3-10+ unit apartments.
- (e) "Other" includes fireline and hydrant sectors.
- (f) No new developments are currently known/anticipated.

References:

1. ABAG, 2018. Association of Bay Area Governments, Plan Bay Area Projections 2040, released on November 2018.

# Table 4-6Projected Water DemandMarin Municipal Water District

Water Use Sector	Projected Demand (AFY) (a)					
	2025	2030	2035	2040	2045	
Potable and Raw Water						
Single Family Residential (b)						
Existing Accounts	14,751	14,751	14,751	14,751	14,751	
New Accounts (c)	321	741	1,028	1,399	1,777	
Multi-Family Residential (d)	3,275	3,342	3,388	3,447	3,507	
Business/Industrial	2,875	2,922	2,945	2,952	2,959	
Institutional	1,600	1,627	1,640	1,643	1,647	
Agricultural/Irrigation	1,659	1,686	1,700	1,704	1,707	
Other (e)	49	50	51	51	51	
Potable Water Served through Recycled	0	0	0	0	0	
Water System (f)	0	0	0	0	0	
Raw Water	171	174	176	176	176	
Non-revenue Water (Potable) (g)	10%	10%	10%	10%	10%	
	2,758	2,760	2,753	2,761	2,777	
Estimated Passive Savings (h)	-733	-1,301	-1,749	-2,125	-2,437	
Total Potable and Raw Water Demand	26,726	26,753	26,682	26,758	26,915	
Recycled Water						
Recycled Water	750	762	768	770	771	
Total Recycled Water Demand	750	762	768	770	771	



## Table 4-6Projected Water DemandMarin Municipal Water District



#### Abbreviations:

- ABAG = Association of Bay Area Governments
- AFY = acre-feet per year
- AWE = Alliance for Water Efficiency
- UWMP = Urban Water Management Plan

#### Notes:

- (a) Water demand projections are estimated based on partial rebound demand factors for residential and raw water sectors and pre-drought demand factors for all other sectors, based on recent historical use. Growth in accounts is based on ABAG 2018 projections, as identified in Table 4-1.
- (b) Single-family residential includes irrigation.
- (c) Water demand factors for new single family residential accounts are based on water use per dwelling unit for buildings constructed in 1994 and later.
- (d) Multi-family residential includes duplexes and 3-10+ unit apartments.
- (e) "Other" includes fireline and hydrant sectors.
- (f) The recycled water system is supplemented with potable water to meet demands, as necessary. The recycled water plant was shut off in 2019 to allow for infrastructrure upgrades. Following the upgrades, potable make-up water to the system, if any, is expected to be de minimis.
- (g) Estimates of potable and raw non-revenue water are based on the average percentage non-revenue water for 2017 through 2019, per Table 3-2.
- (h) Passive water savings are based on the AWE Conservation Tracking Tool.
- (i) 2015 UWMP projections per Reference 2 and include raw water demand.

#### Table 4-6 Projected Water Demand

#### Marin Municipal Water District

#### References:

- 1. ABAG, 2018. Association of Bay Area Governments, Plan Bay Area Projections 2040, released on November 2018.
- 2. Marin Municipal Water District, 2016. Urban Water Management Plan 2015 Update, prepared by RMC Water and Environment, dated June 2016.


### 5. CONSERVATION PROGRAM PARTICIPATION

The following section evaluates historical participation in water conservation programs by District customers and the estimated water savings associated with that participation. This information is used to inform future program selection and implementation assumptions, and to support the demand management measure (DMM) reporting required in UWMPs under CWC § 10631.(e).<sup>11</sup>

For five water conservation programs selected by the District, additional analyses have been conducted, including: (1) a refined estimate of the actual water conservation savings achieved by District customers based on customer billing data (Section 5.3.2), and (2) program participation trends in relation to spatial distribution (Section 5.4), property characteristics (Section 5.5), and customer demographics (Section 5.6). The following five programs were included in the detailed analyses:

- 1. Advanced Metering Infrastructure (AMI) Leak Notifications Program
- 2. Rain Barrel Rebate Program
- 3. Residential High Efficiency Clothes Washer (HECW) Rebate Program
- 4. SFR Water Use Surveys/Audits Program
- 5. SFR Weather-Based Irrigation Controller (WBIC) Rebate Program

The goals of these more detailed analyses are to identify participation drivers and to help the District better understand which customers are participating in which programs. The District can accordingly use this information to inform the strategic design, selection, and marketing of future conservation programs and services.

### 5.1. Conservation Programs

The District currently provides a broad variety of conservation programs directly to customers. These programs are described in **Table 5-1** below.

Program	Description	Eligible Customer Class	Program Run Dates			
CII Water Use Evaluation	CII customers are provided on site water use	CII	1995 - Current			
Program	evaluations, recommendations to improve efficiency,					
	and pre-qualified for applicable rebates.					
AMI Leak Notifications	Customers with AMI meters receive notifications of	SFR, CII, Ag. &	2018 - Current			
Program	water use patterns indicative of leaks. Rather than	Irr.				
	having to wait until their next water bill, customers are					
	able to receive timely information and stop leaks much					
	faster. Water Efficiency staff runs automated reports					

### Table 5-1Description of Conservation Programs

<sup>&</sup>lt;sup>11</sup> The information presented herein supports a portion of the required DMM analysis, focusing on device and education-focused programs. Additional details regarding customer billing rates and structure, conservation staffing levels, customer metering, etc. are required under CWC § 10631.(e), but not addressed herein.



Table 5-1Description of Conservation Programs

Program	Description	Eligible Customer Class	Program Run Dates
	to monitor water use of AMI customers. AMI meters record water use in 15 minute intervals as compared to every other month as is typical for most residential meter reads.		
HET Rebate Program	<ul> <li>2007-2010: Up to a \$250 rebate when customers upgrade a tank-style 3.5 gpf or higher to a new, highefficiency model.</li> <li>2013-Current: Up to \$100 rebate to replace a 3.5 gpf or more per flush or a 1.6 gpf model year 2001 or older with an HET.</li> </ul>	SFR	2007 - 2010 2013 - Current
HET Direct Install Program	Free HET and installation to replace existing 3.5 gpf toilets.	MFR, CII	2013 - 2015
Hot Water Recirculating System	Single family or duplex residential customers can apply for a rebate up to \$50 towards the cost of a hot water recirculating system to create a looped system to recirculate cold water back to the water heater.	SFR	2014 - 2019
Irrigation Improvement Equipment Program	Commercial and multi-family customers can apply for a rebate up to \$1,500 when they install qualifying irrigation equipment to improve outdoor water use efficiency.	Ag. & Irr.	2015 - 2017
Landscape Plan Review Program	MMWD has developed water conservation requirements for landscape professionals and homeowners when designing and installing landscapes and irrigation systems. Plan review requirements apply to all new construction and rehabilitated (renovations or changes made to sites with an existing irrigation system) landscape projects requiring a building permit, plan check, or design review.	SFR, CII	1986 - Current
Large Landscape WBIC Rebate Program	Commercial customers can apply for a rebate up to \$30 per active irrigation station when they purchase and install a new weather based irrigation controller to replace an existing standard controller.	Ag. & Irr.	2013 - 2015
Laundry-to-Landscape System	Single family or duplex residential customers can apply for a rebate up to \$50 towards the cost of a 3-way diverter valve and/or air vent. Products must be installed as part of a residential laundry to landscape system that uses discharge water from a single domestic clothes washer in a one or two family dwelling.	SFR	2015 - 2019
Organic Mulch Rebate Program	Single family or duplex residential customers can apply for a rebate up to \$50 towards the cost of organic mulch. By applying mulch to their gardens, customers will be able to adjust their irrigation schedules as they	SFR	2014 - 2019



Table 5-1
<b>Description of Conservation Programs</b>

Program	Description	Eligible Customer Class	Program Run Dates
	realize mulched areas of their gardens can remain		
	healthy with less water.		
Pool Cover Rebate	Single family or duplex residential customers can apply	SFR	2014 - 2019
Program	for a rebate up to \$50 towards the cost of a pool cover.		
	Covering a swimming pool when it is not in use is the		
	most effective means of reducing evaporative water		
	loss.		
Rain Barrel Rebate	<b>2014-2019</b> : Single family or duplex residential	SFR	2014 – 2019
Program	customers can apply for a rebate up to \$50 towards		2020 - Current
	the cost of a rain barrel to collect rainwater that can		
	be used to supplement landscape irrigation.		
	<b>2020</b> : Residential and commercial customers with		
	active potable water service can apply for a rebate up		
	to \$0.50 per gallon of storage when they install rain		
	rain barrels and cisterns may not exceed \$1,000 per		
	cito		
Residential HECW Rebate	MMWD customers can apply for a rebate towards the	SER	1998 - 2010
Program	cost of installing a qualifying residential high-efficiency	511	2013 – Current
1 rogium	clothes washer that meets current water and energy		2015 Current
	efficiency requirements.		
SFR Turf Removal	MMWD customers participated in the state Save Our	SFR	2015 - 2016
Program	Water turf replacement program, which offered \$2/sq		
_	ft of turf removed.		
SFR Water Use	A free service for SFR customers that involves a visit to	SFR	1995 - Current
Surveys/Audits Program	their property to review water use and identify ways		
	to save water indoors and outside.		
SFR WBIC Rebate	Residential customers can apply for a rebate up to \$20	SFR	2013 - 2015
Program	per active irrigation station when they purchase and		
	install a new weather based irrigation controller to		
	replace an existing standard controller.		2024 2245
Tier 4 Exemption	An incentive program designed to help save water and	SFR	2004 - 2015
Program	lower customer's water bill. To qualify, customers		
	must meet the District's current water conservation		
	suctomor's property then becomes event from Tier		
	Four water rates for a two-year period. This means		
	their water will be billed at no higher than Tier Three		
	rates.		
Time of Sale (Toilet	An ordinance-based program that required the	SFR	2002 - 2006
Retrofit) Program	installation of low flow plumbing (toilets no greater		
,	than 1.6 gpf and pressure reducing valves set at no		
	greater than 50 psi unless required for irrigation) at		



Table 5-1
<b>Description of Conservation Programs</b>

Program	Description	Eligible Customer Class	Program Run Dates
	the time of resale. Ordinance applied to SFR, multi- family and hotel/motel structures.		
ULFT Rebate Program	Offered a rebate of up to \$75 or \$100 when customers upgrade a tank-style 3.5 gpf or higher to a new, 1.6 gpf model.	SFR	1993 - 2007
WaterSense Smart Controller Rebate	Residential and commercial customers can apply for a rebate up to \$100 towards the cost of a qualifying EPA WaterSense weather-based irrigation controller.	SFR, CII	2020-Current
Water Use Surveys/Audits Program (Excluding SFR Participants)	Consultation activities include a review, evaluation and report of indoor plumbing devices and/or a review and report of the landscape irrigation system.	MFR, CII, Ag. & Irr.	1995 - Current
Water Waste Report Program	The general public can report water waste situations to MMWD online or over the phone. These contacts are logged into a database and followed up on by field staff to research and notify properties about water waste situations.	SFR, CII	1998, 2000 - Current
Water-wise Community Garden Program	The Water-Wise Community Gardens program provided resources and rebates for school and community gardens to improve water efficiency. Implementation of a water-related project or improvement to the irrigation system could qualify for a rebate of up to \$1,000 per site.	CII, Ag. & Irr.	2017

Abbr	eviations:	MF	= multi-family
Ag.	= agricultural	MFR	= multi-family residential
AMI	= Advanced Metering Infrastructure	MMW	D = Marin Municipal Water District
CII	<ul> <li>commercial, industrial and institutional</li> </ul>	psi	= pounds per square inch
EPA	= Environmental Protection Agency	SFR	= single family residential
gpf	= gallons per flush	sq ft	= square feet
HECV	V = high efficiency clothes washer	ULFT	= ultra low flow toilet
HET	= high-efficiency toilet	WBIC	= Weather Based Irrigation Controller
Irr.	= irrigation		

In addition to programs offered by the District, several regional-based programs are offered through the SMSWP, including: (1) education and outreach to schools, (2) public outreach and educational workshops, (3) Qualified Water Efficient Landscaper (QWEL) Training, and (4) garden tours. Currently, the District implements its own school education and outreach programs separate from the SMSWP.

### **5.2.** Historical Conservation Program Participation

As shown in **Table 5-2**, the District has implemented 22 different conservation programs offered directly to customers from 1986 through 2020. Of the programs implemented by the District, the ultra low flow toilet (ULFT) Rebate Program, Residential High Efficiency Clothes Washer (HECW) Rebate Program, and SFR Water



Use Surveys/Audits Program had the highest participation, with 27,269, 17,807, and 12,837 participants, respectively. Through the SFR Turf Removal Program, over 55,000 sq ft of turf has been removed.

**Table 5-3** summarizes District participation in the regional SMSWP water conservation school education and outreach programs during the 2007-2008 through 2019-2020 school years. Over this period, over 74,000 students were reached through 20 different programs, including assemblies, presentations, workshops and other educational materials.

### **5.3.** Estimated Savings from Past Programs

### 5.3.1. Estimated Water Savings Based on AWE Model

The AWE model<sup>12</sup> was used to estimate water savings associated with the implementation of all device or turf replacement and audit programs identified in **Table 5-2** for the period of 2010 to 2020. Water savings estimates were based on District-specific values calculated per Section 5.3.2, AWE model default values, values developed for the District in 2015, and other literature values, as needed. The specific assumptions used in this assessment are presented in **Appendix B**. The results of this analysis are presented in **Table 5-4**.

Based on the record of water conservation program participation within the District and application of the AWE Model, it is estimated that the District conservation programs included in this assessment resulted in a savings of between 1,441 AFY and 6,366 AFY between 2010 and 2020.<sup>13</sup> In addition, over this period, it is estimated that the District saved 8,352 AFY through passive savings. Thus, the total active and passive savings achieved by the District between 2010 and 2020 is estimated to be between 13,277 AFY and 14,718 AFY.

### 5.3.2. Estimated Water Savings for Selected Programs Based on Customer Billing Data

Water use savings associated with implementation of specific water conservation programs are typically estimated based on literature values, which may or may not accurately capture the specific ways customers in a specific area (i.e., the District) use water. Therefore, District customer billing data were analyzed using a modified *Difference in Difference Estimation Method* (Columbia Public Health, 2013) to assess the amount of water typically saved through implementation of the selected programs. As described further in **Appendix C**, a version of this method is used to compare the water use patterns in a participant group to that of a cohort group to isolate the impact (in terms of water savings) of participation in a specific water conservation program.

<sup>&</sup>lt;sup>12</sup> Alliance for Water Efficiency, Water Conservation Tracking Tool Version 3, released in July 2016.

<sup>&</sup>lt;sup>13</sup> Free ridership refers to customers who participate in a conservation program, but who would have taken the water saving action (e.g., replace a toilet) regardless of whether the conservation program incentive was available. The amount of free ridership is unknown, and thus a range of savings is shown, assuming 0% to 100% free ridership for programs, as appropriate.



**Table 5-5** summarizes the average estimated water savings for each selected conservation program from 2010-2018.<sup>14</sup> The AMI Leak Notifications Program demonstrated the most savings at 29,023 gallons per account per year (gal/acct/yr), followed by the SFR WBIC Rebate Program at 17,258 gal/acct/yr.

Table 5-5
Average Estimated Water Savings Achieved by Selected Conservation Programs from 2010-2018

Conservation Program (a)	Number of Participants in Analysis	Default AWE Model Unit Savings Factors				
AMI Leak Notifications Program (b)	62	29,023	29,023 gal/event/yr	n/a (e)		
Residential HECW Rebate Program (b)	1,031	4,281	4,276 gal/unit/yr	5,000 gal/unit/yr		
SFR Water Use Surveys/Audits Program (b)	1,682	6,273	6,273 gal/survey/yr	12,373 gal/survey/yr		
SFR WBIC Rebate Program (c)	108	17,258	17,258 gal/WBIC/yr	5,639 gal/WBIC/yr (f)		

Abbreviations:

acct = account	MMWD = Marin Municipal Water District
AMI = Advanced Metering Infrastructure	n/a = not available
AWE = Alliance for Water Efficiency	SFR = single family residential
DSS = Decision Support System	sq ft = square feet
gal = gallon	WBIC = Weather-Based Irrigation Controller
HECW = high efficiency clothes washer	yr = year

### Notes:

(a) This analysis was also performed for the Rain Barrel Rebate Program. However, due to the limited sample size the results were not considered robust and thus are not presented herein.

(b) Program participants included in this analysis are limited to those that: (1) have only participated in the specified program, (2) have only participated in the program in the specified year, and (3) have sufficient water use data within the study periods.

(c) All of the participants have participated in more than one conservation program, thus the analysis is not limited to those that only participated in this program.

(d) Estimated annual water savings associated with the program are calculated as the incremental amount of water saved by the program participants over that of the comparison cohort accounts, as shown in **Tables 5-6a** through **5-6d**. Water savings comparison cohorts for SFR customers are stratified geographically based on Census Block Groups.

(e) Not available for either the AWE or DSS models.

(f) Default value not available in the AWE model. Water savings factor shown is per the District's 2015 DSS Model, and is based on a program that provides multiple types of landscape rebates and other equipment upgrades.

**Tables 5-6a** through **5-6d** summarize the detailed results of these analyses, including the number of participants included in the analysis for each year, the total amounts rebated, the change in water use by participants and their comparison cohort groups, and the estimated savings values by year and in total.

<sup>&</sup>lt;sup>14</sup> This time period was selected so that at least two full years of water use billing data could be analyzed following the program participation year.



**Table 5-5** also shows the default water savings factors included in the AWE model,<sup>15</sup> which are based on available literature values and other assumptions. Water savings for District customers for the Residential HECW Rebate Program are generally consistent with AWE model default values. However, based on analysis of District customers specifically, water savings for the SFR Water Use Surveys/Audits Program are lower than the default values, and therefore evaluation of potential savings for future programs would be significantly overestimated for District customers if default values are used. Conversely, savings for the SFR WBIC Rebate Program are higher than default DSS model values, resulting in a potential for underestimation of program savings if the default values were used.

### 5.4. Spatial Trends in Program Participation

Given the large amount of program participation data for some programs, it can be difficult to ascertain whether participation in these programs has been evenly distributed across the service area, or if participation tends to be clustered in certain regions. In order to identify program participation density for conservation programs in the District service area, a geostatistical spatial analysis was performed.<sup>16</sup> This analysis identifies participation "hot spots," which are areas where a higher density of participation is observed than would be expected by randomly distributed participation. Similarly, "cold spots," or areas of lower than expected participation, are identified. Ineligible parcels (i.e., parcels with no sector use relevant to each respective conservation program) were excluded from each analysis, as well as very large rural SFR parcels (e.g., greater than 10 acres), to reduce skewing of density mapping. High density participation areas are identified in red and low density participation areas are identified in blue on **Figures 5-1a** through **5-1f**.

**Figure 5-1a** shows the results of the AMI Leak Notifications Program, which includes SFR, CII, and agriculture/irrigation accounts. The rebate and survey programs require a customer to opt-in to participate. Participation in the AMI Leak Notifications Program, however, occurs when a customer is notified once a leak is detected for an account based on AMI data. Areas of higher participation (high incidence of leaks) were focused in the southern portion of the service area in the Cities of Belvedere and Tiburon, with no distinct areas of low participation. **Figure 5-1b** shows the results of this analysis when focused on just the Cities of Belvedere and Tiburon. Within these areas, distinct areas of high participation (or rate of leaks) are identified, primarily in the southern portion of the area, and areas of low participation (or rate of leaks) are identified in the northern and southern portions of the area.

**Figures 5-1c** through **5-1f** show the results of the participation destiny analysis for the Residential HECW Rebate Program, the Rain Barrel Rebate Program, the SFR Water Use Surveys/Audits Program, and the SFR WBIC Rebates Program, all of which target residential accounts. Participation was similar between

<sup>&</sup>lt;sup>15</sup> Default value not available in the SFR WBIC Rebate Program. Water savings factor shown is per the District's 2015 DSS Model, and is based on a program that provides multiple types of landscape rebates and other equipment upgrades.

<sup>&</sup>lt;sup>16</sup> The ESRI ArcGIS 10.8 Optimized Hot Spot Analysis tool was used for spatial hot spot analysis of program participation. The hot spot analysis calculates a Getis Ord GI\* statistic for each cell. This statistical z-score evaluates how the event (in this case, participation in the program) clusters spatially, by looking at the cell in the context of the neighboring cells. For the purposes of this study, hot and cold spots are identified as cells with a 90% or greater level of statistical confidence.



the Residential HECW Rebate and SFR Water Use Surveys/Audits Programs, with areas of high participation in the southern and central portions of the service area and low participation in the northwestern portion. The Residential HECW Rebate Program also had more areas of low participation along the western portion of the District. By contrast, the SFR WBIC Rebates Program had only one cluster of high participation towards the southern portion of the service area, while the Rain Barrel Rebate Program had one cluster of high participation in the northern portion of the service area. It should be noted that these programs only included 110 and 162 participants, respectively, and therefore produced less robust results than the other programs that were assessed.

Based on this information, the District could consider targeting outreach to the portions of its service area that have historically had lower program participation, particularly within the Residential HECW Rebate Program and SFR Water Use Surveys/Audits Program.

### 5.5. Building Stock Characteristics

Certain characteristics related to building age can influence, or at least be correlated with, water use. In general, older homes and businesses tend to have higher water using fixtures that were installed prior to passage of key changes to the Federal and California Plumbing, Energy, and Building Codes; these accounts present an opportunity for increasing water conservation. Homes and businesses with larger landscaped areas tend to use more water than those with smaller landscaped areas. Similarly, larger homes may have more occupants and therefore more water use.

In order to assess the distribution of housing stock and other key water use characteristics, service areawide data were evaluated based on Marin County Assessor parcel data. These data included lot sizes and building construction date for residential program participants. Building construction date for parcels within the District based on Marin County Assessor data is shown on **Figure 5-2**. This figure shows parcels for all land use types for which building construction date is available (e.g., residential, commercial, open space, etc.).

Building stock characteristics of conservation program participants for each of the selected programs are summarized in **Table 5-7**.<sup>17</sup> The first chart shows the total number of participants by program by age of building construction, while the second chart shows the results after controlling for the relative number of parcels within each age category.

The average year of building construction for each program ranged from 1954 to 1987. The majority of program participants are in homes built prior to 1994, for all programs. When the results are normalized based on total building stock, CII participants in the AMI Leak Notifications Program and Residential HECW Rebate Program had the highest rates of participation for homes constructed prior to 1994.<sup>18</sup> Participation by SFR customers in homes constructed between 1994 and 2009 was notably higher than that of participants with pre 1994 or 2010 and new homes.

<sup>&</sup>lt;sup>17</sup> Results for SFR, MFR, CII and Irr. participants are shown separately, given the diversity of building stock.

<sup>&</sup>lt;sup>18</sup> The rebate and survey programs require a customer to opt-in to participate. Participation in the AMI Leak Notifications Program, however, occurs when a customer is notified once a leak is detected for an account based on AMI data.



Based on this analysis, the District appears to be successfully reaching buildings in most age ranges for most programs. However, there appears to be some potential to increase participation in: (1) the Rain Barrel Rebate Program in buildings constructed in 1994 and later, and (2) SFR WBIC Rebates and Water Use Surveys/Audits for customers with homes built prior to 1994 and 2010 and later.

### 5.6. Demographic Characteristics of Residential Conservation Program Participation

Residential conservation programs are generally open to all residents in the District service area. Although the programs are available to all residents, those with certain demographic characteristics can tend to participate at higher rates than others in some programs. The analyses described in the following sections were performed for the five selected programs in order to better understand trends in customer demographics among residential conservation program participants in the District – specifically, income, whether the home occupants rent or own the property, and household age.

### 5.6.1. <u>Household Income Trends</u>

Household income data were based on the estimated 2017 median household income by Census Block Group (Census, 2019).<sup>19</sup> The following sections discuss the breakdown of program participation in residential programs by income classification. These income levels are defined as follows: low income (<\$94,850/year), moderate income (\$94,850-\$124,500), and high income (>\$124,500), based on Marin County income designations for a three-person household (HCD, 2017). Given that these classifications reflect the median of all households in a given Census Block Group, this reflects the predominant income for that area (neighborhood), but does not mean that every participant or household in that area falls within the same income group.

**Figure 5-3a** shows the distribution of income groups across the service area and **Table 5-8a** shows the distribution of residential program participants by income level. The first chart in **Table 5-8a** shows the percentage of participants in each program that live in areas of each income level grouping. Across all programs, participation was highest in the high income category, ranging from 53% to 66%, and lowest in the low income category, with participation ranging from 0.78% to 19%.

The second chart on **Table 5-8a** shows participation rates controlled for the number of parcels within the service area within each income group. Customers in all three income groups appear to be well represented in the Rain Barrel Rebate Program and the Residential HECW Programs. The AMI Leak Notifications Program, SFR WBIC Rebate Program, and SFR Water Use Surveys/Audits Program showed very low participation by customers in low income areas and high participation by customers in high income areas.

<sup>&</sup>lt;sup>19</sup> Census Block Group is the smallest geographical unit for which the United States Census Bureau publishes income data.



These results suggest that there are opportunities to increase program participation by lower income households in the SFR WBIC Rebate Program and Water Use Surveys/Audits Program.<sup>20</sup>

### 5.6.2. <u>Homeownership Trends</u>

In order to evaluate whether home ownership appears to be a driving factor in program participation, residential program participation was compared to the proportion of the population that live in renteroccupied homes, based on Census data. Rentership status was based on 2017 Census estimates of the population within a Census Block Group that live in a renter-occupied home versus an owner-occupied home (Census, 2019). Rentership is thus presented as the proportion of the population within a Census Block Group that lives in a renter-occupied home. A Census Block Group with a rentership of less than 25% indicates that the area consists primarily of owner-occupied homes, while a rentership population of greater than 75% indicates that the area is predominantly made up of those who rent their homes.

**Figure 5-3b** shows the distribution of renter-occupancy rate across the District. **Table 5-8b** shows the distribution of residential program participation by the percentage of the population that live in renter-occupied homes ("rentership").

The first chart in **Table 5-8b** shows the percentage of participants in each program that live in areas of each percent rentership grouping. Participation in conservation programs was higher in Census Block Groups with a lower percentage of rentership (high home ownership). Between 51% and 56% of participants across all conservation programs were in Census Block Groups that had less than or equal to 25% rentership, compared to between 0% and 1.0% of participants in the high rentership category ( $\geq$  75% rentership).

The second chart in **Table 5-8b** shows participation rates controlled for the number of customers within the District that fall within each rentership classification. When the relative proportion of number of customers within each rentership group is controlled for, participants in the low rentership (high home ownership) category are 2.6% to 8.1% higher than the overall percentage of customers in the same category. Conversely, participants in the moderate to high rentership groups ( $\leq$ 50%-75% rentership) were underrepresented by 3.2% to 15%.

These results suggest that there are opportunities to increase participation across all programs in areas with higher levels of rentership.

### 5.6.3. <u>Household Age Trends</u>

Median household age is based on 2017 Census estimates of the median age of household members by Census Block Group (Census, 2019). Median age is broken up as follows: <35 years old, 35-45 years old, 45-55 years old, and >55 years old. Given that these classifications reflect the median age of all household

<sup>&</sup>lt;sup>20</sup> While the AMI Leak Notifications Program also had lower levels of participation by low income households, given that participation in this program occurs when customers are notified of leaks using AMI data, it is not appropriate to specifically target selected customers for participation.



members in a given Census Block Group, this reflects the predominant age for that area but does not mean that every participant or household in that area falls within the same age group.

**Figure 5-3c** shows the distribution of median household age by Census Block Group across the service area and **Table 5-8c** shows the distribution of residential program participants by age group. The first chart in **Table 5-8c** shows the percentage of participants in each program that live in areas of each household age grouping. Across all programs, participation was highest for households whose median household member age was between 45-55 years, ranging from 51% to 64%. The lowest participation was in households with a median age of less than 35 years, comprising 0% to 1.2% of all participants.

The second chart in **Table 5-8c** shows participation rates controlled for the number of parcels within the service area within each median household age group. Compared to the overall distribution of customers, there was little difference among age groups for most conservation programs, with the exception of the AMI Leak Notifications Program,<sup>21</sup> which had a higher proportion of participants from households with a median age older than 55 years (19% higher) and a lower proportion of participants 35-45 years (12% lower), and the Rain Barrell Rebate Program, which had a higher proportion of participants from households with a median age between 45 and 55 years (6.3% higher).

These results suggest that, while there are some differences in participation rates across age groups, the District has been generally successful at reaching customers of all age groups in all programs.

### 5.7. Summary

Sections 5.4 through 5.6 above identify opportunities for the District to increase customer participation in each of the selected programs through targeted outreach to certain customer classes. The results of these analyses can be combined to identify specific customers by overlaying these results spatially. For example, one may identify SFR customers to target with the SFR WBIC Rebate Program by overlaying customers in areas: (1) outside of high participation as identified on **Figure 5-1f**, (2) within low income areas identified on **Figure 5-3a**, and (3) in areas with between 50% and 75% rentership as shown on **Figure 5-3b**. As show on **Figure 5-4**, by overlaying these key metrics, approximately 9,000 SFR customers are identified for potential targeting of SFR WBIC Rebate Program outreach materials.

<sup>&</sup>lt;sup>21</sup> Participation in the AMI Leak Notifications Program occurs when a customer is notified once a leak is detected for an account based on AMI data, and is not driven by a customer choice to participate.

## Table 5-2 Summary of Conservation Program Participation Marin Municipal Water District

End Use Number of Program Participants														Data of																						
Program Name	Sector (a)	Indoor/ Outdoor	1986	1987	1020	1990	1991	1992	1994	1995	1996	1997	1998	1999	2000	2001	2002	2003	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	Total	Accounts (b)
AMI Leak Notifications Program (c)	SFR, CII, Ag. & Irr.	Both																															690	1,187	1,877	3.4%
CII Water Use Evaluation Program	CII	Both								25	42	23	40	56	47	17	4	20	42	16	15	33	53	140	65	89	66	47	41	32	35	35	11	4	998	24%
HET Rebate Program	SFR	Indoor																				612	839	1,624	51			214	1,169	1,226	772	591	427	308	7,833	14%
HET Direct Install Program (d)	MFR, CI	Indoor																										42	124	68					234	3.80%
Hot Water Recirculating System	SFR	Outdoor											1																10	141	31	26	13	15	236	0.43%
Irrigation Improvement Equipment Program	Ag. & Irr.	Outdoor																												30	12	22			64	4.6%
Landscape Plan Review Program	SFR, CII	Outdoor	3	6 2	9 2	4 28	39	30 2	1 32	31	36	38	66	29	27	47	46	40	56	50	74	50	42	45	30	50	76	89	72	88	91	104	114	120	1,723	2.9%
Large Landscape WBIC Rebate Program (c)	Ag. & Irr.	Outdoor																										-	5	6					11	0.80%
Laundry-to-Landscape System	SFR	Outdoor																												13	1	5			19	0.035%
Organic Mulch Rebate Program	SFR	Outdoor																											87	782	203	107	98	80	1,357	2.5%
Pool Cover Rebate Program	SFR	Outdoor																											10	142	43	10	13	14	232	0.43%
Rain Barrel Rebate Program (c)	SFR	Outdoor																											19	100	20	11	6	6	162	0.30%
Residential HECW Rebate Program (c)(e)	SFR	Indoor											720	967	826	942	1,472	1,366	1,344	1,248	1,688	1,537	1,620	1,654	252			133	675	615	439	187	57	65	17,807	33%
SFR Turf Removal Program (c)	SFR	Outdoor																												11	44				55	0.10%
SFR Water Use Surveys/Audits Program (c)	SFR	Both								127	234	380	246	281	470	634	465	319	46	329	83	264	755	892	699	1,102	862	849	1,033	838	498	712	466	253	12,837	24%
SFR WBIC Rebate Program (c)	SFR	Outdoor																										2	66	42					110	0.20%
Tier 4 Exemption Program (c)	SFR	Both																	23	6	7	4	3	3	0	1	2	0	2	2					53	0.10%
Time of Sale (Toilet Retrofit) Program	SFR	Indoor															327	419	630	391	53														1,820	3.4%
ULFT Rebate Program	SFR	Indoor						4	3,552	2 3,425	3,877	5,112	2,779	2,629	2,078	942	2,580	15	56	174	45	1													27,269	50%
Water Use Surveys/Audits	MFR,																																			
Program (Excluding SFR	CII, Ag.	Both								49	10	37	26	9	1	6	2	4	2	3	228	362	558	317	463	311	504	504	364	415	128	125	71	22	4,521	58%
Participants)	& Irr.																																			
Water Waste Report Program	SFR, CII	Both											6		46	16	127	99	41	18	33	80	52	42	33	67	52	68	584	541	190	142	162	151	2,550	4.4%
Water-wise Community Garden Program	CII, Ag. & Irr.	Outdoor																														22			22	0.40%
																		Total	Turf R	emove	ed (sq f	ft)														
SFR Turf Removal Program (Turf Removed)	SFR	Outdoor																											:	10,364	45,348				55,712	
						<u> </u>		<u> </u>	· · · · · · · · · · · · · · · · · · ·									WB	IC Stat	ions Re	ebated					I						'				
SFR WBIC Rebate Program (Station Rebated) (f)	SFR	Outdoor																										20	719	394					1,133	
Large Landscape WBIC Rebate Program (Station Rebated) (f)	Ag. & Irr.	Outdoor																											191	123					314	

### Table 5-2 Summary of Conservation Program Participation

Marin Municipal Water District

### **Abbreviations**

Ag. & Irr. = Agricultural & Irrigation CII = Commercial, Industrial, and Institutional HECW = High Efficiency Clothes Washer HET = High Efficiency Toilet MFR = multi-family residential

#### Notes

(a) Each record provided in the sources below is assumed to be one participant. However, some customers may have participated multiple times, but program records do not include sufficient detail to identify this.

- (a) Predominant sector for program participants.
- (b) Participation is calculated as a percentage of total accounts of the predominant sector indicated.
- (c) Indicated program will be included in detailed program analysis.
- (d) Three participant records did not incude date and name; thus, they were not included herein.
- (e) Only residential washer models are qualified for rebate (i.e. no coin op models).
- (f) Number of WBIC rebates shown is based on the number of active valve stations.
- (g) Colored shading is added for visualization purposes. Green shading represents higher participation values.

Pct. = Percentage SFR = Single-family residential sq ft = Square feet ULFT = Ultra Low Flow Toilet WBIC = Weather-Based Irrigation Controller

EKI Environment Water, Inc. December 2020

# Table 5-3 Summary of Conservation School Education Program Participation Marin Municipal Water District

					Num	ber of St	tudents	Reached	by Scho	ol Year (	a)			
Activity	2007- 2008	2008- 2009	2009- 2010	2010- 2011	2011- 2012	2012- 2013	2013- 2014	2014- 2015	2015- 2016	2016- 2017	2017- 2018	2018- 2019	2019- 2020	Total
Assembly	1,175	2,967	5,942	5,138	6,955	6,962	7,418	6,161	6,890	6,127	6,579	8,122	3,690	74,126
BF Garden Registration		0												0
BF School Garden Registration	0													0
Bus Transportation Paid			92	75	50		218	242						677
Bus Transportation Pending			56											56
Event (Staffing)			90	824	172	120	85	610	330	830	1,250	250		4,561
Event (Support)				0										0
Field Trip (LHS)		1,151												1,151
Flyer/Material Distribution			4,951	2,324	1,344	1,052								9,671
Hands-on Student Activities	900		122											1,022
Other			1								1			2
Presentation (Classroom)	2,561	372	332	529	731	504	727	550	539	502	834	554	138	8,873
Presentation (Misc)		40	37	0	0	23	22		52	11	11			196
Presentation (Pre-Field Trip)			936	631	778	834								3,179
Restoration Field Trip			768	552	890	558								2,768
School Garden Grants		500												500
Student Projects	25	10												35
Teacher Workshop	1,500	0												1,500
Technical Garden Support		0												0
Water Walk Field Trip				149	190	169	285	443	456	338	481	130	66	2,707
Total	6,161	5,040	13,327	10,222	11,110	10,222	8,755	8,006	8,267	7,808	9,156	9,056	3,894	111,024

# Table 5-3 Summary of Conservation School Education Program Participation Marin Municipal Water District

### **Abbreviations**

BF = Bay-Friendly LHS = Lawrence Hall of Science Misc = miscellaneous

### <u>Notes</u>

- (a) School education program participation is presented by number of students reached, per Marin Municipal Water District, 2020. School year is assumed to start in September and end in August.
- (b) Colored shading is added for visualization purposes. Green shading represents higher participation values.

### <u>Source</u>

Marin Municipal Water District, 2020. School Education.xlsx, provided by MMWD on 1 April 2020.

### Table 5-4 Estimated Water Savings Achieved by Conservation Programs and Passive Savings

Marin Municipal Water District

	End U	se	Estimated Cumulative Water Savings (AFY) (b)										
Water Saving Type	Sector (a)	Indoor/ Outdoor	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020
Conservation Programs (c)					-		-			-			
AMI Leak Notifications Program	SFR, CII, Ag. & Irr.	Both	0	0	0	0	0	0	0	0	61	229	396
CII Water Use Evaluation Program	CII	Both	19	59	110	165	220	268	307	343	370	389	401
HET Rebate Program	SFR	Indoor	1.5	3.0	4.4	12	59	148	263	397	553	719	880
HET Direct Install Program	MFR, CII	Indoor	1.8	3.5	5.1	19	74	174	270	363	453	540	625
Hot Water Recirculating System	SFR	Outdoor	0	0	0	0	0.06	1.0	2.1	3.3	4.7	6.1	7.5
Irrigation Improvement Equipment Program	Ag. & Irr.	Outdoor	0	0	0	0	0	0.52	1.2	2.4	3.5	4.6	5.7
Large Landscape WBIC Rebate Program	Ag. & Irr.	Outdoor	0	0	0	0	0	1.5	4.7	8.0	11	15	18
Laundry-to-Landscape System	SFR	Outdoor	0	0	0	0	0	0.21	0.44	0.76	1.1	1.4	1.7
Organic Mulch Rebate Program	SFR	Outdoor	0	0	0	0	0	0.04	0.07	0.12	0.17	0.22	0.27
Pool Cover Rebate Program	SFR	Outdoor	0	0	0	0	0.21	3.3	7.3	12	16	21	22
Rain Barrel Rebate Program	SFR	Outdoor	0	0	0	0	0	0.11	0.22	0.34	0.5	0.6	0.7
Residential HECW Rebate Program	SFR	Indoor	3.3	6.4	9.4	14	27	47	72	99	124	149	172
SFR Turf Removal Program	SFR	Outdoor	0	0	0	0	0	0.38	2.4	4.5	6.5	9	11
SFR Water Use Surveys/Audits Program	SFR	Both	27	90	173	272	390	508	607	703	787	850	891
SFR WBIC Rebate Program	SFR	Outdoor	0	0	0	0	3.7	10	15	21	27	33	39
Water Use Surveys/Audits Program (Excluding SFR Participants)	MFR, CII, Ag. & Irr.	Both	133	328	630	1,015	1,428	1,834	2,166	2,420	2,597	2,710	2,761
Water Waste Report Program	SFR, CII	Both	0.70	2.7	5.4	9.0	24	48	70	91	110	125	133
Water-wise Community Garden Program	CII, Ag. & Irr.	Outdoor	0	0	0	0	0	0	0	0.38	0.76	1.1	1.5
Total Program Water Savings			185	493	937	1,506	2,226	3,043	3,790	4,468	5,126	5,801	6,366
Passive Water Savings (d)			0	135	398	785	1,351	2,106	3,035	4,128	5,384	6,794	8,352
Total Water Savings (100% Free Ridership) (e)			180	617	1,320	2,254	3,444	4,834	6,302	7,853	9,533	11,380	13,277
Total Water Savings (0% Free Ridership) (e)			185	628	1,336	2,290	3,577	5,149	6,824	8,596	10,510	12,595	14,718

 Table 5-4

 Estimated Water Savings Achieved by Conservation Programs and Passive Savings

 Marin Municipal Water District



### **Abbreviations**

Ag. & Irr. = Agricultural & Irrigation CII = Commercial, Industrial, and Institutional HECW = High Efficiency Clothes Washer HET = High Efficiency Toilet MFR = multi-family residential SFR = single-family residential WBIC = weather-based irrigation controller

### Table 5-4

### Estimated Water Savings Achieved by Conservation Programs and Passive Savings

Marin Municipal Water District

### <u>Notes</u>

- (a) Predominant sector for program participants.
- (b) Water savings are estimated per the AWE model.
- (c) The water savings associated with the landscape plan review program and the tier 4 exemption program are estimated as a part of passive savings. Additional programs with participation prior to 2010 are not included herein.
- (d) Passive water savings are water savings associated with the natural change out of water using fixtures and devices with higher efficiency ones, due to plumbing code and market changes. Passive savings are estimated for the whole service area.
- (e) Free ridership refers to customers who participate in a conservation program, but who would have taken the water saving action (e.g., replace a toilet) regardless of whether the conservation program incentive was available. The amount of free ridership is unknown, and thus a range is shown. Free ridership is applied to device, hot water recirculation systems, turf replacement, and other irrigation equipment programs only.

### Sources

1. Marin Municipal Water District, 2020. Program Participation Data, provided by Marin Municipal Water District on 1 April 2020 and 16 June 2020.

# Table 5-6a Estimated Water Savings Achieved by the AMI Leak Notifications Program Marin Municipal Water District

		Average Water L	Estimated Savings due	
	Number of	Participant Group	Cohort Group (c)	to Program (d)
Year	Participants (a)	(gal/yr)	(gal/yr)	(gal/acct/yr)
2018	62	15,291	-13,732	29,023
Total	62			
Avg (e)		15,291	-13,732	29,023

Abbreviations:

avg = average	MFR = multi-family residential
gal/acct/yr = gallons per account per year	SFR = single family residential
gal/yr = gallons per year	= not applicable

Notes:

- (a) Program participants included in this analysis are limited to those that: (1) have only participated in the specified program, (2) have only participated in the program in the specified year, and (3) have sufficient water use data within the study periods.
- (b) A negative value indicates that average water use increased following program participation.
- (c) Customers included in the comparison cohort groups are limited to those that: (1) have not participated in any water efficiency program based on available data and (2) have sufficient water use data within the study periods.
- (d) Estimated annual water savings associated with the program are calculated as the incremental amount of water saved by the program participants over that of the comparison cohort accounts.
   Water savings comparison cohorts for SFR and MFR customers are stratified geographically based on Census Block Groups. Cohorts for participants in other sectors are stratified by sector only.
- (e) The estimated savings are the weighted average based on the number of participants. Water use reduction averages are not weighted.

Sources:

# Table 5-6b Estimated Water Savings Achieved by the Residential HECW Rebate Program Marin Municipal Water District

				Average Water Use			
			Total	Reducti	Reduction (b)		
	Number of	<b>Total HECW</b>	Rebate	Participant	Cohort	Savings due to	Estimated
	Participants	Rebated	Amount	Group	Group (c)	Program (d)	Unit Savings
Year	(a)	(unit)	(\$)	(gal/yr)	(gal/yr)	(gal/acct/yr)	(gal/yr/unit)
2013	68	68	\$3,400	11,985	6,905	5,080	5,080
2014	354	355	\$17 <i>,</i> 750	16,344	11,441	4,904	4,890
2015	315	315	\$28,150	11,402	8,680	2,723	2,723
2016	186	186	\$27 <i>,</i> 900	5,852	2,356	3,496	3,496
2017	84	84	\$12,600	2,545	-4,824	7,369	7,369
2018	24	24	\$3 <i>,</i> 600	2,068	-6,486	8,553	8,553
Total	1,031	1,032	\$93,400				
Avg (e)				8,366	3,012	4,281	4,276

Abbreviations:

avg = average gal/acct/yr = gallons per account per year gal/yr = gallons per year gal/yr/unit = gallons per year per unit device rebated HECW = high efficiency clothes washer MFR = multi-family residential SFR = single family residential -- = not applicable

Notes:

- (a) Program participants included in this analysis are limited to those that: (1) have only participated in the specified program, (2) have only participated in the program in the specified year, and (3) have sufficient water use data within the study periods.
- (b) A negative value indicates that average water use increased following program participation.
- (c) Customers included in the comparison cohort groups are limited to those that: (1) have not participated in any water efficiency program based on available data and (2) have sufficient water use data within the study periods.
- (d) Estimated annual water savings associated with the program are calculated as the incremental amount of water saved by the program participants over that of the comparison cohort accounts. Water savings comparison cohorts for SFR and MFR customers are stratified geographically based on Census Block Groups. Cohorts for participants in other sectors are stratified by sector only.
- (e) The estimated savings are the weighted averages based on the number of participants. Water use reduction averages are not weighted.

Sources:

# Table 5-6c Estimated Water Savings Achieved by the SFR Water Use Surveys/Audits Program Marin Municipal Water District

		Average Water L	Estimated Savings due	
	Number of	Participant Group Cohort Group (c)		to Program (d)
Year	Participants (a)	(gal/yr)	(gal/yr)	(gal/acct/yr)
2013	311	14,784	7,320	7,464
2014	346	18,632	11,006	7,627
2015	299	16,028	8,114	7,914
2016	210	8,473	1,782	6,690
2017	311	-4,120	-5,694	1,574
2018	205	379	-6,112	6,491
Total	1,682			
Avg (e)		9,029	2,736	6,273 (f)

Abbreviations:

avg = average gal/acct/yr = gallons per account per year gal/yr = gallons per year MFR = multi-family residential MMWD = Marin Municipal Water District SFR = single family residential -- = not applicable

### Notes:

- (a) Program participants included in this analysis are limited to those that: (1) have only participated in the specified program, (2) have only participated in the program in the specified year, and (3) have sufficient water use data within the study periods. It is noted that participants was not limited to SFR customers.
- (b) A negative value indicates that average water use increased following program participation.
- (c) Customers included in the comparison cohort groups are limited to those that: (1) have not participated in any water efficiency program based on available data and (2) have sufficient water use data within the study periods.
- (d) Estimated annual water savings associated with the program are calculated as the incremental amount of water saved by the program participants over that of the comparison cohort accounts. Water savings comparison cohorts for SFR and MFR customers are stratified geographically based on Census Block Groups. Cohorts for participants in other sectors are stratified by sector only.
- (e) The estimated savings are the weighted average based on the number of participants. Water use reduction averages are not weighted.
- (f) MMWD often promoted rebates to SFR customers during the water use survey. As such, it is likely that many customers also participated in rebate programs, which eliminated them from the participant group. The participants who did not follow up with the rebate programs might have different water use habits than those who did. Thus, the estimated savings presented here is different than the expected value.

### Sources:

## Table 5-6d Estimated Water Savings Achieved by the SFR WBIC Rebate Program Marin Municipal Water District

		Total	Total	Total	Average Water Use Reduction (b)		Estimated		
	Number of	WBIC Robated	WBIC Station	Rebate	Participant	Cohort Group (c)	Savings due to	Estimated	Unit Sovings
Year	(a)	(unit)	(station)	(\$)	(gal/yr)	(gal/yr)	(gal/acct/yr)	(gal/yr/WBIC)	(gal/yr/station)
2014	66	66	719	\$12,937	27,226	11,131	16,095	16,095	1,477
2015	42	42	394	\$7,676	27,387	8,303	19,084	19,084	2,034
Total	108	108	1,113	\$20,613					
Avg (e)					27,307	9,717	17,258	17,258	1,694

Abbreviations:

avg = average	gal/yr/station = gallons per year per WBIC station
gal/acct/yr = gallons per account per year	SFR = single family residential
gal/yr = gallons per year	WBIC = weather-based irrigation controller
gal/yr/WBIC = gallons per year per WBIC rebated	= not applicable

Notes:

- (a) Program participants included in this analysis are limited to those that have sufficient water use data within the study periods. All the participants have participated in more than one conservation program, thus the analysis is not limited to those that only participated in this program.
- (b) A negative value indicates that average water use increased following program participation.
- (c) Customers included in the comparison cohort groups are limited to those that: (1) have not participated in any water efficiency program based on available data and (2) have sufficient water use data within the study periods.
- (d) Estimated annual water savings associated with the program are calculated as the incremental amount of water saved by the program participants over that of the comparison cohort accounts. Water savings comparison cohorts for SFR customers are stratified geographically based on Census Block Groups.
- (e) The estimated savings are the weighted averages based on the number of participants. Water use reduction averages are not weighted.

### Sources:

### Table 5-7 Building Stock Characteristics by Program Participants Marin Municipal Water District

			Aver Lat Circ		Year of Construction			
Water Efficiency Program (a)	Sector (b)	Avg Year Built	(sq ft)	ac)	pre-1994	1994-2009	2010 and Later	
	SFR	1987	24,291	0.56	64%	33%	2.7%	
AMI Leak Notifications Program	CII	1957	479,819	11	100%	0%	0%	
	Irrig.	1973	953,284	22	93%	7.5%	0%	
Rain Barrel Rebate Program	SFR	1971	11,669	0.27	91%	9.0%	0%	
	SFR	1975	13,834	0.32	84%	15%	0.67%	
Residential HECW Rebate Program	MFR	1979	47,317	1.1	89%	11%	0.24%	
	CII	1954	118,839	2.7	95%	1.3%	3.9%	
SFR Water Use Surveys/Audits Program	SFR	1977	19,155	0.44	81%	18%	0.99%	
SFR WBIC Rebate Program	SFR	1978	20,113	0.46	81%	19%	0%	



 Table 5-7

 Building Stock Characteristics by Program Participants

 Marin Municipal Water District



## Table 5-7 Building Stock Characteristics by Program Participants Marin Municipal Water District

#### Abbreviations:

ac = acre avg = average CII = commercial, industrial, and institutional Irrig. = Irrigation HECW = High Rfficiency Clothes Washer MFR = multi-family residential SFR = single family residential sq ft = square feet

### Notes:

- (a) Program participants included in this analysis are limited to those for which relevant parcel data are available. The analysis is also limited to sectors with more than 50 participants in a given program.
- (b) Program participants in the business / industrial and institutional sectors are grouped as "CII", and participants in the agricultural / irrigation sector are presented as "Irrig." in this analysis.
- (c) Relative difference is calculated as the percentage of program participation by year of construction minus the overall percentage of residential customers by year of construction within the service area.

#### Sources:

1. Marin County, 2020. Sonoma county Assessor Parcel Data, provided via Marin Municipal Water District, 13 February 2020.

# Table 5-8a Residential Customer Program Participation by Median Household Income Marin Municipal Water District

			Percentage of Participating Residential Customers (b)						
Median House	hold Income (a)	Percentage of Residential Customers in MMWD (b)	AMI Leak Notifications Program	Rain Barrel Rebate Program	Residential HECW Rebate Program	SFR Water Use Surveys/Audits Program	SFR WBIC Rebate Program		
Low Income	<\$94,850	20%	0.78%	19%	18%	15%	13%		
Moderate Income	\$94,850 - \$124,500	26%	33%	28%	26%	25%	23%		
High Income	>\$124,500	54%	66%	53%	57%	59%	65%		



Table 5-8aResidential Customer Program Participation by Median Household IncomeMarin Municipal Water District



### Abbreviations:

AMI = Advanced Metering Infrastructure	MMWD = Marin Municipal Water District
HECW = High Efficiency Clothes Washer	SFR = single family residential
HUD = United States Department of Housing and Urban Development	WBIC = weather-based irrigation controller

# Table 5-8a Residential Customer Program Participation by Median Household Income Marin Municipal Water District

### Notes:

- (a) Household income is based on estimated 2017 median household income by Census Block Group, per Census (2019). Income level groupings are based on California Department of Housing and Community Development (HCD) income levels for Marin County for a 3-person household in 2017 (HCD, 2017). The average persons per household is 2.4 for Marin County, based on Census data.
- (b) Residential customers include both single-family and multi-family customers. Participants included in this analysis are limited to those for which location data are available.
- (c) Relative difference is calculated as the percentage of program participation by income group minus the overall percentage of residential customers by income group within the service area.

References:

- 1. Census, 2019. 2013-2017 American Community Survey (ACS) 5-year estimates. TIGER/Line Shapefiles by Block Group, https://www.census.gov/geo/maps-data/data/tiger-data.html, United States Census Bureau, downloaded on 14 January 2020.
- 2. HCD, 2017. Memorandum: State Income Limits for 2017, California Department of Housing and Community Development, dated June 9, 2017.

# Table 5-8b Residential Customer Program Participation by Percentage of Renters Marin Municipal Water District

			Perc	Percentage of Participating Residential Customers (b)							
Percentage of Renters (a)		Percentage of Residential Customers in MMWD (b)	AMI Leak Notifications Program	Rain Barrel Rebate Program	Residential HECW Rebate Program	SFR Water Use Surveys/Audits Program	SFR WBIC Rebate Program				
Low Rentership	≤25%	48%	51%	56%	52%	53%	55%				
Low to Moderate Rentership	25.1%-50%	34%	48%	34%	34%	34%	36%				
Moderate to High Rentership	50.1%-75%	16%	1.1%	9.3%	13%	12%	8.2%				
High Rentership	≥75%	2.0%	0.11%	0.62%	1.0%	0.84%	0%				



 Table 5-8b

 Residential Customer Program Participation by Percentage of Renters

 Marin Municipal Water District



Abbreviations:

AMI = Advanced Metering Infrastructure HECW = High Efficiency Clothes Washer MMWD = Marin Municipal Water District SFR = single family residential WBIC = weather-based irrigation controller

# Table 5-8b Residential Customer Program Participation by Percentage of Renters Marin Municipal Water District

### Notes:

- (a) Percent rentership reflects the proportion of population within a given Census Block Group that lives in renter-occupied homes. Low rentership indicates an area consists predominantly of owner-occupied homes; high rentership indicates an area consists predominantly of renter-occupied homes. Rentership is based on estimated percentage of rentership by Census Block Group, per Census (2019).
- (b) Residential customers include both single-family and multi-family customers. Participants included in this analysis are limited to those for which location data are available.
- (c) Relative difference is calculated as the percentage of program participation by percent of renters group minus the overall percentage of residential customers by percent of renters group within the service area.

### References:

1. Census, 2019. 2013-2017 American Community Survey (ACS) 5-year estimates. TIGER/Line Shapefiles by Block Group, https://www.census.gov/geo/maps-data/data/tiger-data.html, United States Census Bureau, downloaded on 14 January 2020.

# Table 5-8c Residential Customer Program Participation by Median Household Age Marin Municipal Water District

Modian	Percentage of	Percentage of Participating Residential Customers (b)								
Median Household Age (a)	Residential Customers in	AMI Leak Notifications	Rain Barrel Rebate	Residential HECW Rebate	SFR Water Use Surveys/Audits	SFR WBIC Rebate				
	MMWD (b)	Program	Program	Program	Program	Program				
<35 Years	1.6%	0.11%	1.2%	0.97%	0.92%	0%				
35-45 Years	27%	15%	23%	27%	23%	30%				
45-55 Years	57%	51%	64%	58%	58%	55%				
>55 Years	14%	34%	12%	14%	17%	15%				



 Table 5-8c

 Residential Customer Program Participation by Median Household Age

 Marin Municipal Water District



Abbreviations:

AMI = Advanced Metering Infrastructure HECW = High Efficiency Clothes Washer MMWD = Marin Municipal Water District SFR = single family residential WBIC = weather-based irrigation controller

# Table 5-8cResidential Customer Program Participation by Median Household AgeMarin Municipal Water District

### Notes:

- (a) Median household age is based on the estimated median age of household members by Census Block Group, per Census (2019).
- (b) Residential customers include both single-family and multi-family customers. Participants included in this analysis are limited to those for which location data are available.
- (c) Relative difference is calculated as the percentage of program participation by household age group minus the overall percentage of residential customers by household age group within the service area.

### References:

 Census, 2019. 2013-2017 American Community Survey (ACS) 5-year estimates. TIGER/Line Shapefiles by Block Group, https://www.census.gov/geo/maps-data/data/tiger-data.html, United States Census Bureau, downloaded on 14 January 2020.



### Legend

### Participation Hot and Cold Spots

bath: X:\C00004\_SonomaMarin\Map\2020\12\_MMWVD\Fig5-1a\_MMWD\_HotSpot\_AMI\_Leak\_Notificat

Cold Spot - 99% Confidence

Cold Spot - 95% Confidence

Cold Spot - 90% Confidence

### Not Significant

Hot Spot - 90% Confidence Hot Spot - 95% Confidence

Hot Spot - 99% Confidence

### **Program Participation**

• AMI Leak Notifications

Service Area

### <u>Notes</u>

- 1. All locations are approximate.
- 2. Program participation hot and cold spots were evaluated using the Esri ArcGIS 10.8.0 Optimized Hot Spot Analysis tool, which calculates a Getis-Ord GI\* statistic. This statistic is a measure of the spatial distribution of incidents (participation) relative to a random, equally-spaced distribution.
- 3. Participants included in this analysis are limited to those for which detailed participation records and location data are available.

### Sources

- 1. Water use efficiency program data provided by Marin Municipal Water District on April 2020.
- 2. Basemaps provided by ESRI and Stamen Design, under CC BY 3.0. Data by OpenStreetMap, under ODbL.



### Participation Density for AMI Leak Notifications Program





### Legend

#### **Participation Hot and Cold Spots**

Cold Spot - 99% Confidence Cold Spot - 95% Confidence Cold Spot - 90% Confidence Not Significant Hot Spot - 90% Confidence Hot Spot - 95% Confidence Hot Spot - 99% Confidence

### <u>Notes</u>

- 1. All locations are approximate.
- 2. Program participation hot and cold spots were evaluated using the Esri ArcGIS 10.8.0 Optimized Hot Spot Analysis tool, which calculates a Getis-Ord GI\* statistic. This statistic is a measure of the spatial distribution of incidents (participation) relative to a random, equally-spaced distribution.
- 3. Participants included in this analysis are limited to those for which detailed participation records and location data are available.

### Sources

- Water use efficiency program data provided by Marin Municipal Water District on April 2020.
- 2. Basemaps provided by ESRI and Stamen Design, under CC BY 3.0. Data by OpenStreetMap, under ODbL.



Participation Density for AMI Leak Notifications Program -Belvedere and Tiburon



ath: X:\C00004 SonomaMarin\Map\2020\12 MMWD\Fig5-1b MMWD HotSpot AMI Leak Notifica

Service Area

• AMI Leak Notifications Program


#### Participation Hot and Cold Spots

Cold Spot - 99% Confidence

Cold Spot - 95% Confidence

Cold Spot - 90% Confidence

#### Not Significant

Hot Spot - 90% Confidence Hot Spot - 95% Confidence Hot Spot - 99% Confidence

#### **Program Participation**

• Residential HECW Rebate Program

Service Area Boundary

#### Abbreviation

CII = commercial, industrial, and institutional HECW = high efficiency clothes washer MFR = multi-family residential SFR = single family residential

#### <u>Notes</u>

1. All locations are approximate.

- 2. Program participation hot and cold spots were evaluated using the Esri ArcGIS 10.8.0 Optimized Hot Spot Analysis tool, which calculates a Getis-Ord GI\* statistic. This statistic is a measure of the spatial distribution of incidents (participation) relative to a random, equally-spaced distribution.
- 3. Participants included in this analysis are limited to those for which detailed participation records and location data are available.
- 4. The HECW Rebate Program is open to SFR, MFR, and CII customers, but only residential washers are rebated.

#### Sources

- 1. Water use efficiency program data provided by Marin Municipal Water District on April 2020.
- 2. Basemaps provided by ESRI and Stamen Design, under CC BY 3.0. Data by OpenStreetMap, under ODbL.



#### Participation Density for Residential HECW Rebate Program



Marin Municipal Water District December 2020 C00004.00 Figure 5-1c



#### Participation Hot and Cold Spots

Not Significant

• Rain Barrel Rebate

Service Area

bath: X:\C00004\_SonomaMarin\Map\2020\12\_MMWD\Fig5-1d\_MMWD\_HotSpot\_RainBarrel\_20201103 Program

Cold Spot - 99% Confidence

Cold Spot - 95% Confidence

Cold Spot - 90% Confidence

Hot Spot - 90% Confidence

Hot Spot - 95% Confidence

Hot Spot - 99% Confidence

#### <u>Notes</u>

- 1. All locations are approximate.
- 2. Program participation hot and cold spots were evaluated using the Esri ArcGIS 10.8.0 Optimized Hot Spot Analysis tool, which calculates a Getis-Ord GI\* statistic. This statistic is a measure of the spatial distribution of incidents (participation) relative to a random, equally-spaced distribution.
- 3. Participants included in this analysis are limited to those for which detailed participation records and location data are available.

#### Sources

- 1. Water use efficiency program data provided by Marin Municipal Water District on April 2020.
- 2. Basemaps provided by ESRI and Stamen Design, under CC BY 3.0. Data by OpenStreetMap, under ODbL.



#### **Participation Density for Rain Barrel Rebate Program**



#### Marin Municipal Water District December 2020 C00004.00 Figure 5-1d



Participation Hot and Cold Spots

Cold Spot - 99% Confidence Cold Spot - 95% Confidence Cold Spot - 90% Confidence

Not Significant

Hot Spot - 90% Confidence Hot Spot - 95% Confidence

Hot Spot - 99% Confidence

#### **Program Participation**

• SFR Water Use Surveys/Audits Program

Service Area Boundary

#### <u>Abbreviation</u> SFR = single family residential

#### <u>Notes</u>

- 1. All locations are approximate.
- 2. Program participation hot and cold spots were evaluated using the Esri ArcGIS 10.8.0 Optimized Hot Spot Analysis tool, which calculates a Getis-Ord GI\* statistic. This statistic is a measure of the spatial distribution of incidents (participation) relative to a random, equally-spaced distribution.
- 3. Participants included in this analysis are limited to those for which detailed participation records and location data are available.

#### Sources

- 1. Water use efficiency program data provided by Marin Municipal Water District on April 2020.
- 2. Basemaps provided by ESRI and Stamen Design, under CC BY 3.0. Data by OpenStreetMap, under ODbL.



#### Participation Density for SFR Water Use Surveys/Audits Program



Marin Municipal Water District December 2020 C00004.00 Figure 5-1e



Participation Hot and Cold Spots

Cold Spot - 99% Confidence Cold Spot - 95% Confidence Cold Spot - 90% Confidence Not Significant

Hot Spot - 90% Confidence Hot Spot - 95% Confidence Hot Spot - 99% Confidence

Program

• SFR WBIC Rebates

Service Area

#### Abbreviation

SFR = single family residential WBIC = Weather-Based Irrigation Controller

#### <u>Notes</u>

- 1. All locations are approximate.
- 2. Program participation hot and cold spots were evaluated using the Esri ArcGIS 10.8.0 Optimized Hot Spot Analysis tool, which calculates a Getis-Ord GI\* statistic. This statistic is a measure of the spatial distribution of incidents (participation) relative to a random, equally-spaced distribution.
- 3. Participants included in this analysis are limited to those for which detailed participation records and location data are available.

#### <u>Sources</u>

- 1. Water use efficiency program data provided by Marin Municipal Water District on April 2020.
- 2. Basemaps provided by ESRI and Stamen Design, under CC BY 3.0. Data by OpenStreetMap, under ODbL.



#### **Participation Density for** SFR WBIC Rebates Program



Marin Municipal Water District December 2020 C00004.00 Figure 5-1f







Path: X:\C00004

#### Legend County Boundary

Service Area Boundary

Residential Customers

#### Median Household Income

<\$94,850 (Low)

\$94,850 - \$124,500 (Medium)

>\$124,500 (High)

#### Abbreviations

HUD = Housing and Community Development

#### <u>Notes</u>

1. All locations are approximate.

2. Household income is based on estimated 2017 median household income by Census Block Group, per Census (2019). Income level groupings are based on California Department of Housing and Community Development (HCD) income levels for Marin County for a 3-person household in 2017 (HCD, 2017). The average persons per household is 2.4 for Marin County.

#### Sources

1. Census, 2019. 2013-2017 American Community Survey (ACS) 5- ear estimates. TIGER/Line Shapefiles by Block Group, https://www.census.gov/geo/maps-data/data/tiger-data.html, United States Census Bureau.

2. HCD, 2017. Memorandum: State Income Limits for 2017, California Department of Housing and Community Development, dated June 9, 2017.

3. Basemap provided by ESRI.



#### Median Household Income



Marin Municipal Water District December 2020 C00004.00 **Figure 5-3a** 







- All SFR Customers
- SFR Customers to Potentially Target with Outreach (8,891 customers)

#### Abbreviation

SFR = single family residential WBIC = Weather-Based Irrigation Controller

### <u>Notes</u>

- 1. All locations are approximate.
- 2. SFR customers to potentially target with outreach for the SFR WBIC Rebate Program are identified as those (1) outside areas of high participation, (2) within low income household areas, and (3) within areas of 50-75%rentership.

#### Sources

- 1. Water use efficiency program data provided by Marin Municipal Water District on April 2020.
- 2. Basemaps provided by ESRI and Stamen Design, under CC BY 3.0. Data by OpenStreetMap, under ODbL.



#### SFR Customers to Potentially Target with SFR WBIC Rebate Program Outreach



Marin Municipal Water District December 2020 C00004.00 Figure 5-4



#### 6. CONSERVATION PROGRAM UPDATE

The following section evaluates current and potential conservation programs for both the District and the SMSWP. The purpose of this section is to compile programs that are prioritized by both the District and by all Water Contractors in the SMSWP collectively in order to calculate the potential water savings and economic feasibility of those programs. Section 6.1 discusses the methodology used to prioritize conservation programs. Section 6.2 describes the programs given high priority for implementation by all nine Water Contractors collectively, and Section 6.3 describes programs given high priority by the District. Section 6.4 analyzes the potential water savings and cost-benefit for those programs selected by the District as both individual programs and in three implementation scenarios. By assessing the feasibility of these programs, the District can make more informed decisions regarding program selection and implementation.

#### 6.1. Methodology for Screening of Potential Water Conservation Programs

In order to evaluate the potential for new conservation programs, a comprehensive list of over 100 conservation programs was developed (**Appendix D**). Each of the nine Water Contractors were first asked to review and identify any additional programs to add to this list. Following receipt of feedback from the Water Contractors, each Water Contractor was asked to review the list and identify:

- Priority (on a scale of 1 to 5, with 5 being the highest priority) as a program to be implemented regionally through the SMSWP;
- Priority (on a scale of 1 to 5, with 5 being the highest priority) as a program to be implemented locally through their agency;
- Preference for the program to be implemented either regionally or locally; and
- Whether each program is currently or has previously been implemented by their agency.

The list of water conservation programs is organized into four categories, specifically: (1) retailer actions and water rates, (2) public outreach and education, (3) device-based and financial incentive programs, and (4) policies and regulations. The results of the water conservation program prioritization and screening are summarized for all Water Contractors combined, representing overall regional priorities and preferences (**Table 6-1**), and for each individual Water Contractor, representing each agencies local priorities and preferences. **Table 6-1** shows the average prioritization ranking for all Water Contractors for each program for regional and local implementation as well as the percentage of Water Contractors that prefer each program to be implemented at the local level or the regional level.<sup>22</sup> The results presented in **Table 6-1** are discussed below for each water conservation program category. **Table 6-2** provides the results of this screening for the Marin Municipal Water District, including priorities and preferences for each water conservation program, and identifies the target sector, whether the program addresses indoor or outdoor water use, and the primary end use.

<sup>&</sup>lt;sup>22</sup> Water Contractors were asked to provide a preference for local or regional implementation for all programs they ranked a priority score of 3 or above. Thus, the percentages of Water Contractors shown in **Table 6-1** does not sum to 100%.



#### 6.2. Screening of Regional Conservation Measures

#### 6.2.1. <u>Retailer Actions and Water Rate Based Conservation Programs</u>

Of the 15 retailer action and water rate based conservation programs included in the screening list, the Water Contractors identified the following ten programs as high priority (average score of three or higher) to implement at the local level:

- 1. Install Advanced Metering Infrastructure (AMI) for High Water Users and Large Landscape Accounts
- 2. Install AMI in New Development
- 3. Customer Water Loss Reduction (AMI Leak Notifications)
- 4. Install AMI for Existing Accounts
- 5. Tiered Water Rates (Conservation Pricing)
- 6. Water Budgeting/Monitoring for Large Landscape Accounts
- 7. Water Budget Based Billing for Only Irrigation Customers
- 8. Modification to or Implementation of Tiered Rate Conservation Pricing
- 9. Establish Separate Pricing Structure for Irrigation Accounts
- 10. Rate Structure Evaluation
- 11. Increase Enforcement of State Water Waste Regulations

By their nature as water retailer actions, these programs do not lend themselves to regional implementation. However, in some cases, such as the "Increase Enforcement of State Water Waste Regulations" program, there may be an opportunity to coordinate across the region at a policy or education level. For example, SB-407<sup>23</sup> requires older plumbing fixtures to be replaced with new, more efficient fixtures that meet current water efficiency standards; this requirement is supposed to be enforced at time of sale. If this or similar policies are being enforced differently across Water Contractor jurisdictions, it could result in confusion among customers. Thus, even for retailer action-based programs, there may be opportunity for the Water Contractors to coordinate these efforts and share staff education resources.

#### 6.2.2. <u>Public Outreach and Education Based Conservation Programs</u>

Of the 11 public outreach and education-based water conservation programs included in the screening, the Water Contractors identified the following six programs as high priority (average score of three or higher), with a preference for regional implementation through SMSWP:

- 1. Qualified Water Efficient Landscaper (QWEL) Training
- 2. Public Outreach through Print & Electronic Media Focused on Outdoor Irrigation
- 3. Educational Workshops
- 4. School Education Programs
- 5. Public Outreach through Print & Electronic Media Focused on Indoor Conservation
- 6. Garden tour

<sup>&</sup>lt;sup>23</sup> SB 407: <u>https://leginfo.legislature.ca.gov/faces/billNavClient.xhtml?bill\_id=200920100SB407</u>



All of these programs are currently being implemented by the SMSWP. In addition to these programs, the Water Contractors also indicated that water use surveys or audits for single-family residential and CII customers were a high priority; however, the Water Contractors generally expressed a preference for these programs to be implemented locally.

#### 6.2.3. <u>Device and Financial Incentive Based Conservation Programs</u>

Of the 61 device- and financial incentive- based water conservation programs included in the screening list, the Water Contractors identified the following 11 programs as high priority (average score of three or higher) to implement at either the regional or local level:

- 1. Landscape Conversion or Turf Removal multi-family residential (MFR) and CII
- 2. Landscape Conversion or Turf Removal single family residential (SFR)
- 3. High Efficiency Faucet Aerator / Showerhead Giveaway Residential Customers
- 4. Smart Irrigation Controller (Weather-Based Irrigation Controller) Rebates Large Landscape
- 5. Drip Irrigation Incentive for SFR
- 6. High Efficiency Faucet Aerator / Showerhead Giveaway CII Customers
- 7. Drip Irrigation Incentive for MFR and CII
- 8. High Efficiency Clothes Washer Rebate Residential
- 9. Smart Irrigation Controller (Weather-Based Irrigation Controller) Rebates SFR
- 10. Restaurant Spray Nozzle Rebates
- 11. Incentivize Irrigation Equipment Upgrades SFR

The above list includes four programs that focus on indoor water use ("High Efficiency Faucet Aerator / Showerhead Giveaway – Residential Customers", "High Efficiency Faucet Aerator / Showerhead Giveaway – CII Customers", "High Efficiency Clothes Washer Rebate – Residential," and "Restaurant Spray Nozzle Rebates"). The remaining preferred programs all focus on outdoor water use, including turf removal and methods to increase irrigation efficiency.

Of these preferred programs, the Water Contractors expressed a preference for two of the programs to be administered at a regional level rather than local level, specifically the "High Efficiency Clothes Washer Rebate – Residential" and the "Restaurant Spray Nozzle Rebates".

#### 6.2.4. Policy and Regulation Based Conservation Programs

Of the 29 policy- and regulation- based water conservation programs included in the screening list, the Water Contractors identified the following six programs as high priority (average score of three or higher) to implement at the local level:

- 1. Water Waste Ordinance
- 2. Require Submetering of Landscaping for New MFR and Commercial Developments
- 3. Require Water Efficiency Plan Reviews for New CII Development
- 4. Require High Efficiency Clothes Washers in New Development
- 5. Require Weather Adjusting Smart Irrigation Controllers, Rain Sensors, and/or Soil Moisture Sensors in New Development
- 6. Demand Offset/Water Neutral Policy for Large New Developments



Nearly all of the highest priority programs focus on ensuring efficiency in new developments, and target both indoor and outdoor water use. The Water Contractors expressed that the program "Require Irrigation Designers / Installers be Certified (QWEL)" is a high priority at the local level but were split equally as to whether they would prefer this program to be implemented at a local or regional level. Further, given the shift in state policy regarding recycled water use (i.e., that non-potable use of recycled water use will no longer be counted towards water conservation), some Water Contractors were conflicted as to how recycled water should be considered in policies regarding new development, in particular with respect to the program "Demand Offset/Water Neutral Policy for Large New Development."

#### 6.2.5. <u>Regional Program Screening Findings</u>

With some exceptions, the Water Contractors expressed a strong preference for water conservation programs to be implemented locally rather than regionally through the SMSWP, with the exception of programs that are already implemented regionally by the SMSWP. However, as listed above, there was general consensus among Water Contractors about which water conservation programs are a high priority, and thus important for the region. Given this consensus, while there is not an apparent desire to implement programs regionally, there may be opportunity for further coordination and collaboration on these programs, such as sharing of educational resources, training of staff (e.g., building permit and plan review staff), and collaboration on creating similar program structure and requirements (such as for financial incentive-based programs) across the region.

#### 6.3. Screening of Local Conservation Measures

**Table 6-2** shows the results of this screening for the Marin Municipal Water District, and lists the programs considered by the District to be medium or high priority to consider for the future. **Table 6-2** also identifies the target sector, whether the program addresses indoor or outdoor water use, and the primary targeted end use.

- **Retailer Actions and Water Rate Based Conservation Programs.** Eleven retailer action and water rate based conservation programs were identified as medium or high priority for potential future implementation, all of which are currently implemented locally by the District. Four programs target outdoor water use and seven target both indoor and outdoor water use.
- Public Outreach and Education Based Conservation Programs. The District ranked seven public outreach and education-based water conservation programs as medium to medium-high priority for potential future implementation, all of which are currently implemented by the District. Two programs target indoor water use, two target outdoor use, and three target both. Three were given a preference for local implementation, three were given regional preference, and one was given no preference.
- Device and Financial Incentive Based Conservation Programs. Twenty-five device and financial incentive based programs were ranked as medium to high priority for potential future implementation, eleven of which would be new programs for the District. Eight programs target indoor water use, fifteen target outdoor use, and two target both. Eighteen were given a



preference for local implementation and three were given no preference. The potential new programs are identified as follows, in general order of priority:

- Drip Irrigation Incentive for SFR
- o Incentivize Irrigation Equipment Upgrades SFR
- Landscape Conversion or Turf Removal MFR and CII
- o Landscape Conversion or Turf Removal -SFR
- Soil Moisture Sensor Rebate
- Water Savings Incentive Program for CII
- o Incentivize Gray Water Systems for New CII Development
- o Incentivize Replacement of Inefficient Commercial and Industrial Equipment
- Rain Sensor Rebate
- Rotating Sprinkler Nozzle Giveaway
- Rotating Sprinkler Nozzle Rebate
- Policy and Regulation Based Conservation Programs. Fourteen policy and regulation based programs were identified as highest priority for potential future implementation, nine of which are currently implemented by the District and five of which would be new programs. Three programs target indoor water use, eight target outdoor use, and three target both. All programs were given a preference for local implantation, except for "Waste Water Ordinance" (no preference). The potential new programs identified are as follows, in general order of priority:
  - Require Submetering of Landscaping for New MFR and Commercial Developments
  - o Demand Offset/Water Neutral Policy for Large New Developments
  - Prohibit Once through Cooling Systems
  - Require <1.0 gal/flush Toilets in New Development
  - Require Rain Barrels in New Development

#### 6.4. Evaluation of Future Water Conservation Programs

Based on the conservation screening process described in Sections 6.2 and 6.3 above, a suite of conservation programs to be considered for future implementation were evaluated. These programs were evaluated both individually and as components in three water conservation program scenarios, as shown in **Table 6-3a**. The three program scenarios represent three potential approaches or strategies for the District's future conservation programs, specifically:

- Scenario A represents a focus on programs that target outdoor water savings,
- Scenario B represents a more "business as usual" approach based on programs ranked most highly by the District, and
- Scenario C represents a focus on the programs that all nine Water Contractors collectively identified as highest priority.

**Table 6-3a** also identifies the customer sectors each program would target as well as whether the program focuses on indoor or outdoor water use, or both.



The benefits and costs associated with implementation of these programs were evaluated using the AWE model, using a series of assumptions documented in **Appendix B**.<sup>24</sup> Key assumptions and considerations related to the methodology used by the AWE model and in this analysis are provided below:

- Financial assumptions related to both costs to the utility and customer water rates were provided by the District.
- Financial assumptions related to energy costs to the customer were assumed based on typical PG&E rates (PG&E, 2020; PG&E and Marin Clean Energy, 2020).
- Water savings assumptions were based on a combination of District-specific water savings estimates per Section 5.3.2, AWE model default assumptions, assumptions developed for the District as a part of the 2015 conservation modeling, and water savings factors developed based on other published literature sources.
- Assumed rate of program implementation was based on historical participation levels by District customers in similar programs.
- For purposes of near-term conservation program analysis, it is assumed that all programs are active from 2021 through 2025; water savings projections beyond this period reflect cumulative savings achieved over time from implementation during this five-year period.
- Benefit-costs ratios are particularly sensitive to the assumed nominal rate of increase of the utility water cost.
- Lost revenue due to reduced water sales is not included as a cost.
- Additional program-specific considerations are provided as notes in the attached tables.

**Table 6-3b** presents a comparison of individual water conservation measures, and identifies the following information for each program:

- Net present value of costs and benefits represents the present value over the 25-year period discounted to current 2020 dollars.
- Benefit to cost ratio calculated as present value of costs divided by the present value of benefits.
- Water Utility Costs costs that the District as a water utility will incur to operate the program including administrative costs.
- **Customer Costs** costs customers will incur to implement a program in the Water Contractor's service area.
- **Utility Benefits** the avoided cost to the District to produce the volume of water saved.
- **Customer Benefits** the savings from reduced water/sewer utility bills and energy savings resulting from reduced use of hot water.
- **Total Water Utility Costs** includes costs to the District for program implementation from 2021-2025.

<sup>&</sup>lt;sup>24</sup> Alliance for Water Efficiency, Water Conservation Tracking Tool Version 3, released in July 2016.



- Water Savings in 2025 one-year estimated water savings in 2025.
- Water Utility Cost of Water Saved for individual programs cost of water saved dividing by the lifetime water savings of that program.
- Water Utility Cost of Water Saved for program scenarios weighted average of Water Utility Cost of Water Saved for the individual programs by the cumulative water savings through 2045.

This analysis estimates active program savings based on the AWE model, and does not include additional savings anticipated from passive savings (i.e., water savings associated with the natural replacement of less efficient water using fixtures and appliances due to both market shifts and increasing efficiency mandated by the building code and other regulatory requirements). Based on this analysis, and the assumptions presented in **Appendix B**, the benefit-cost ratios for the District range from 0.52 to 116.

**Table 6-3c** presents the results of the analysis of the three conservation program scenarios identified in **Table 6-3a**, and includes a summary of costs and benefits to the District and customers, estimated cumulative water savings through 2045 (based on assumed program implementation from 2021-2025), and the estimated cost of water saved to the District. Based on this, the approach of focusing water conservation measures on those ranked highest by the District (i.e., Scenario B) has a greater benefit to cost ratio than that of Scenarios A or C. The projected water savings associated with implementation of Scenario B is 584 AF by 2025 and 2,205 by 2045, at a cost of approximately \$332/AF. The high benefit-cost ratio in this scenario is driven primarily by the CII Water Savings Incentive Program, which has a guaranteed rate of water savings per agency cost.

Regional Prioritiz	ation of Conservation Measures a	and Programs
I	Marin Municipal Water District	
	Prioritization (a)	Broforonco (b)

	Priorit	zation (a)	Prefere	Current		
Conservation Measure/Program	Regional Local		Regional Program	Local Program	SMSWP Program	
RETAILER ACTIONS AND WATER RATES						
Install AMI for High Water Users and Large Landscape Accounts	2.5	4.7	11%	67%	No	×
Install AMI in New Development	2.4	4.7	0%	67%	No	×
Customer Water Loss Reduction (AMI Leak Detection)	2.4	4.4	0%	89%	No	×
Install AMI for Existing Accounts	2.4	4.0	0%	86%	No	×
Tiered Water Rates (Conservation Pricing)	2.0	3.6	0%	88%	No	×
Water Budgeting/Monitoring for Large Landscape Accounts	2.5	3.4	0%	83%	No	×
Water Budget Based Billing for Only Irrigation Customers	2.1	3.4	0%	86%	No	×
Modification to or Implementation of Tiered Rate Conservation Pricing	2.0	3.4	0%	88%	No	×
Establish Separate Pricing Structure for Irrigation Accounts	2.0	3.2	0%	83%	No	×
Rate Structure Evaluation	2.4	3.1	0%	78%	No	×
Increase Enforcement of State Water Waste Regulations	2.6	3.0	0%	86%	No	×
Water Budget Based Billing for All Customers	2.3	2.4	0%	50%	No	×
Increase Enforcement of Indoor Fixture Retrofit at Time of Sale	1.9	2.2	17%	67%	No	×
Increase Enforcement of Customer Pressure Reducing Valve (PRV) Requirement	1.6	1.9	0%	40%	No	×
Regional UHET and/or Urinal Bulk Purchase Program	1.9	1.7	75%	0%	No	×
Average by Program Type	2.2	3.3				
PUBLIC OUTREACH AND EDUCATION						
QWEL Training (Qualified Water Efficient Landscaper)	4.3	2.0	89%	0%	Yes	~
Public Outreach through Print & Electronic Media - Focused on Outdoor Irrigation	4.0	3.9	67%	0%	Yes	~
Educational Workshops	4.0	3.2	63%	0%	Yes	✓
School Education Programs	4.0	3.1	78%	0%	Yes	~

Marin Municipal Water District

	Prioritization (a)		Preference (b)		Current	
Conservation Measure/Program	Regional	Local	Regional Program	Local Program	SMS Prog	SWP gram
Water Use Surveys/Audits - SFR	3.5	3.9	22%	44%	No	×
Public Outreach through Print & Electronic Media - Focused on Indoor Conservation	3.6	3.3	57%	0%	Yes	~
Garden tour	3.6	1.9	86%	0%	Yes	~
Water Use Surveys/Audits - CII	3.0	3.4	38%	38%	No	×
Water Use Surveys/Audits - MFR	2.8	3.3	29%	43%	No	×
Promote Green Building and Certification	3.1	2.2	33%	17%	No	×
Provide Support with Smart Irrigation Controller Setup	2.9	2.3	60%	0%	No	×
Average by Program Type	3.5	3.0				
DEVICE-BASED AND FINANCIAL INCENTIVE PF	ROGRAMS					
Landscape Conversion or Turf Removal - MFR and CII	3.9	4.6	11%	78%	No	×
Landscape Conversion or Turf Removal -SFR	3.9	4.6	22%	67%	No	×
High Efficiency Faucet Aerator / Showerhead Giveaway - Residential Customers	3.0	3.9	11%	44%	No	×
Smart Irrigation Controller (Weather-Based Irrigation Controller) Rebates - Large Landscape	3.1	3.6	38%	38%	No	×
Drip Irrigation Incentive for SFR	2.4	3.6	25%	50%	No	×
High Efficiency Faucet Aerator / Showerhead Giveaway - CII Customers	2.9	3.4	14%	57%	No	×
Drip Irrigation Incentive for MFR and CII	2.4	3.4	25%	50%	No	×
High Efficiency Clothes Washer Rebate - Residential	3.3	3.3	44%	11%	Yes	~
Smart Irrigation Controller (Weather-Based Irrigation Controller) Rebates - SFR	2.9	3.2	14%	57%	No	×
Restaurant Spray Nozzle Rebates	3.1	2.8	50%	0%	No	×
Incentivize Irrigation Equipment Upgrades - SFR	2.1	3.0	17%	50%	No	×
Indoor Fixture Program For Schools	2.9	2.9	14%	71%	No	×
Rotating Sprinkler Nozzle Rebate	2.9	2.9	40%	20%	No	×

Marin Municipal Water District

	Prioritization (a)		Preference (b)		Current			
Conservation Measure/Program	Regi	ional Local		Regional Program		SMSWP Program		
High Efficiency Clothes Washer Rebate Program - CII	2.8		2.8		29%	29%	No	×
Direct Install of Efficient Indoor Fixtures - Low Income Residential	2.8		2.6		60%	0%	No	×
Indoor Fixture Program For Hotels & Motels	2.8		2.2		29%	43%	No	×
Mulch rebate	2.6		2.7		33%	50%	No	×
Rain Sensor Rebate	2.5		2.6		33%	50%	No	×
Incentivize Submetering for Existing Customers - CII	2.4		2.6		25%	25%	No	×
Incentivize Submetering for Existing Customers - MFR	2.4		2.6		25%	25%	No	×
Incentivize Gray Water Retrofit for Existing SFR Customers	2.3		2.6		20%	60%	No	×
Toilet Flapper Giveaway - SFR customers	2.1		2.6		40%	40%	No	×
Rotating Sprinkler Nozzle Giveaway	2.5		2.1		60%	0%	No	×
Incentivize Replacement of Inefficient Commercial and Industrial Equipment	2.4		2.4		33%	33%	No	×
Soil Moisture Sensor Rebate	2.4		2.4		60%	20%	No	×
High Efficiency Urinal (<0.25 gal/flush) Rebates - CII	2.4		2.4		25%	0%	No	×
Incentivize Gray Water Systems for New Cll Development	2.3		2.4		50%	25%	No	×
Incentivize Irrigation Equipment Upgrades - Large Landscapes	1.9		2.4		20%	40%	No	×
Direct Install of Efficient Indoor Fixtures - Residential	2.4		2.2		50%	0%	No	×
High Efficiency Clothes Washer Install - Low Income Residential Customers	2.4		2.2		50%	0%	No	×
Smart Irrigation Controller (Weather-Based Irrigation Controller) Giveaway - Large Landscape	2.4		2.0		80%	0%	No	×
Smart Irrigation Controller (Weather-Based Irrigation Controller) Giveaway - SFR	2.4		2.0		60%	20%	No	×
Incentivize Artificial Turf for Sports Fields	2.3		2.3		75%	0%	No	×
UHET <1.0 gal/flush Rebate - Residential	2.1		2.3		50%	17%	No	×
Water Savings Incentive Program for CII	2.1		2.2		40%	40%	No	×

# Table 6-1 Regional Prioritization of Conservation Measures and Programs Marin Municipal Water District

	Priori	tization (a)	Prefer	Curi	rent	
Conservation Measure/Program	Regional	Local	Regional Program	Local Program	SMS Prog	WP gram
Hot Water on Demand Pump System Rebate	2.0	2.2	60%	20%	No	×
UHET Direct Installation - CII	2.1	1.8	40%	0%	No	×
Plumber Initiated UHET and / or Urinal Retrofit Program	2.1	1.8	67%	0%	No	×
Direct Install of Efficient Indoor Fixtures - Government Buildings	2.1	1.6	50%	0%	No	×
Rain Barrel Rebate	1.9	2.1	40%	40%	No	×
Incentivize Replacement of Pressure Reducing Valves (PRVs) with 60-70 psi PRVs	2.0	2.0	33%	33%	No	×
Thermostatic Shut-Off Valve Showerheads/Tub Spouts Rebates	2.0	1.9	50%	0%	No	×
Dipper Well Rebates	2.0	1.8	50%	0%	No	×
Rain Sensor Giveaway	2.0	1.7	75%	0%	No	×
Rebates for Conductivity Controllers on Cooling Towers	2.0	1.6	75%	0%	No	×
Rainwater Catchment System Rebate for Large Landscapes	1.9	2.0	50%	25%	No	×
Nonresidential Incentive for Self-closing or Metering Faucets	1.9	1.9	33%	33%	No	×
Efficient (EnergyStar) Dishwasher Rebates	1.9	1.8	50%	0%	No	×
Rain Barrel Giveaway	1.9	1.7	75%	0%	No	×
UHET Direct Installation - Residential	1.9	1.7	50%	0%	No	×
Autoclave (Steam-Sterilizer) Retrofit Rebates	1.9	1.7	67%	0%	No	×
Connectionless Food Steamer Rebates	1.9	1.7	67%	0%	No	×
Dry Vacuum Pumps	1.9	1.6	33%	0%	No	×
Incentivize Cooling Tower Upgrades	1.9	1.6	50%	0%	No	×
UHET <1.0 gal/flush Rebate - CII	1.8	1.8	60%	20%	No	×
Soil Moisture Sensor Giveaway	1.8	1.7	67%	0%	No	×
Direct Install of Efficient Indoor Fixtures - Commercial and Industrial	1.8	1.7	67%	0%	No	×

### Table 6-1Regional Prioritization of Conservation Measures and Programs

Marin Municipal Water District

	Prioriti	zation (a)	Preference (b)		Current	
Conservation Measure/Program	Regional	Local	Regional Program	Local Program	SMSWP Program	
Swimming Pool and Hot Tub Cover Rebates	1.3	1.7	50%	25%	No 🗙	
Urinal Direct Installation - CII	1.5	1.4	50%	0%	No 🗙	
Tier 4 Exemption	1.3	1.4	25%	25%	No 🗙	
Incentivize Submetering of Cooling Towers for Existing Customers	1.3	1.4	50%	0%	No 🗙	
Average by Program Type	2.3	2.3				
POLICIES AND REGULATIONS						
Water Waste Ordinance	2.9	4.3	0%	63%	No 🗙	
Require Submetering of Landscaping for New MFR and Commercial Developments	2.8	4.0	0%	63%	No 🗙	
Require Water Efficiency Plan Reviews for New Cll Development	2.5	3.7	14%	57%	No 🗙	
Require High Efficiency Clothes Washers in New Development	2.8	3.3	17%	67%	No 🗙	
Require Weather Adjusting Smart Irrigation Controllers, Rain Sensors, and/or Soil Moisture Sensors in New Development	2.4	3.1	0%	80%	No 🗙	
Require Irrigation Designers / Installers be Certified (QWEL)	3.0	2.9	40%	40%	No 🗙	
Demand Offset/Water Neutral Policy for Large New Developments	2.4	3.0	0%	83%	No 🗙	
Require Efficient (EnergyStar) Dishwashers in New Development	2.8	2.9	20%	60%	No 🗙	
Require <0.25 gal/flush Urinals in New Development	2.3	2.8	0%	67%	No 🗙	
Water Conserving Landscape and Irrigation Codes, More Stringent than MWELO	1.6	2.8	0%	67%	No 🗙	
Require Swimming Pool and Hot Tub Covers	2.0	2.7	40%	20%	No 🗙	
Require Submetering by Unit for New Commercial Developments	2.3	2.6	0%	50%	No 🗙	
Require Submetering of Landscaping for Existing MFR and Commercial Customers	2.4	2.4	0%	67%	No 🗙	
Require Hot Water on Demand / Structured Plumbing in New Residential Development	2.3	2.4	25%	50%	No 🗙	
Require Submetering by Unit for Existing Commercial Customers	2.1	2.4	0%	25%	No 🗙	

# Table 6-1 Regional Prioritization of Conservation Measures and Programs Marin Municipal Water District

	Pric	oritization (a)	Prefere	Preference (b)		
Conservation Measure/Program	Regional	Local	Local Regional Program		SMSWP Program	
Require Submetering for New MFR Developments	1.9	2.4	0%	50%	No 🗙	
Require Plumbing for Recycled Water in New MFR Development	2.0	2.3	0%	60%	No 🗙	
Require <1.0 gal/flush Toilets in New Development	2.0	2.3	0%	80%	No 🗙	
Require Submetering for New Mobile Home Park Developments	2.0	2.3	0%	40%	No 🗙	
Prohibit Once through Cooling Systems	2.0	2.2	0%	50%	No 🗙	
Require Plumbing for Recycled Water in New Cll Development	1.9	2.2	0%	60%	No 🗙	
Require On-Site Water Reuse Systems (Grey Water or Black Water) for Large CII Developments	1.8	2.1	25%	50%	No 🗙	
Require Plumbing for Gray Water in New SFR Development	1.6	2.1	0%	75%	No 🗙	
Require Submetering of Cooling Towers for New Development	2.0	1.9	0%	33%	No 🗙	
Require Submetering of Existing MFR (and Mobile Home Park) Customers	1.9	1.9	0%	50%	No 🗙	
Restrict Landscape Irrigation to Designated Days/Times	1.6	1.8	33%	0%	No 🗙	
Require Rain Barrels in New Development	1.5	1.8	0%	67%	No 🗙	
Require Submetering of Cooling Towers for Existing Customers	1.8	1.6	0%	50%	No 🗙	
Require Cooling Tower Retrofits	1.5	1.4	0%	33%	No 🗙	
Average by Program Type	2.1	2.5				

Marin Municipal Water District

#### Abbreviations:

AMI = advanced metering infrastructure CII = commercial, industrial, institutional MFR = multi-family residential MWELO = Model Water Efficient Landscape Ordinance PRV = pressure reducing valve SFR = single-family residential SMSWP = Sonoma-Marin Saving Water Partnership UHET = ultra high efficiency toilet

#### Notes:

(a) Each Water Contractor was asked to rank each conservation program or measure in terms of priority as a regionally-administered program, and as a locallyadministered program, where 5 indicated highest priority and 1 indicated the lowest priority. Results are presented as an average of the responses of all nine Water Contractors.

(b) For each program a Water Contractor ranked as "3" or above, the Water Contractor was asked to indicate whether they would prefer the program to be administered regionally or locally. The results are presented as a percentage of the number of Water Contractors. Results of contractors who expressed "no preference" are not shown, and thus the total may not sum to 100% for a given measure.

Table 6-2
Prioritization of Conservation Measures and Programs
Marin Municipal Water District

Conservation Measure/Program	Prioritization (a)	Sector	Indoor	Outdoor	Primary End Use	Preference (b)	Local Program		
RETAILER ACTIONS AND WATER RATES									
Establish Separate Pricing Structure for Irrigation Accounts	5	IRR		x	Irrigation	Locally	Yes, currently		
Install AMI for High Water Users and Large Landscape Accounts	5	All		х	Water Loss	Locally	Yes, currently		
Install AMI in New Development	5	All	х	х	Water Loss	Locally	Yes, currently		
Modification to or Implementation of Tiered Rate Conservation Pricing	5	All	x	х	All	Locally	Yes, currently		
Tiered Water Rates (Conservation Pricing)	5	All	х	х	All	Locally	Yes, currently		
Water Budget Based Billing for Only Irrigation Customers	5	CII, IRR		х	Irrigation	Locally	Yes, currently		
Customer Water Loss Reduction (AMI Leak Detection)	4	All	x	х	Water Loss	Locally	Yes, currently		
Install AMI for Existing Accounts	4	All	х	х	Water Loss	Locally	Yes, currently		
Water Budgeting/Monitoring for Large Landscape Accounts	4	IRR	х	х	Irrigation	Locally	Yes, currently		
Increase Enforcement of State Water Waste Regulations	3	All		х	Irrigation	Locally	Yes, currently		
Rate Structure Evaluation	3	All	х	х	All	Locally	Yes, currently		
PUBLIC OUTREACH AND EDUCATION									
Water Use Surveys/Audits - Cll	4	CII	х	х	All	Locally	Yes, currently		
Water Use Surveys/Audits - MFR	4	MFR	х		All Indoor	Locally	Yes, currently		
Water Use Surveys/Audits - SFR	4	SFR	х	x	All	Locally	Yes, currently		
Educational Workshops	3	SFR		х	All Outdoor	No preference	Yes, currently		
Public Outreach through Print & Electronic Media - Focused on Indoor Conservation	3	All		х	All Indoor	Regionally	Yes, currently		
Public Outreach through Print & Electronic Media - Focused on Outdoor Irrigation	3	All	х		Irrigation	Regionally	Yes, currently		
School Education Programs	3	SFR, MFR	х	х	All	Regionally	Yes, currently		
DEVICE-BASED AND FINANCIAL INCENTIVE PRO	GRAMS								
High Efficiency Faucet Aerator / Showerhead Giveaway - CII Customers	5	CII	х		Faucet, Showerhead	Locally	Yes, currently		
High Efficiency Faucet Aerator / Showerhead Giveaway - Residential Customers	5	SFR, MFR	х		Faucet, Showerhead	Locally	Yes, currently		
Drip Irrigation Incentive for MFR and CII	4	MFR, CII		х	Irrigation	Locally	Yes, previously		
Drip Irrigation Incentive for SFR	4	SFR		х	Irrigation	Locally	No		

Table 6-2
Prioritization of Conservation Measures and Programs
Marin Municipal Water District

Conservation Measure/Program	Prioritization (a)	Sector	Indoor	Outdoor	Primary End Use	Preference (b)	Local Program
Incentivize Irrigation Equipment Upgrades - Large Landscapes	4	MFR, CII, IRR		х	Irrigation	Locally	Yes, previously
Incentivize Irrigation Equipment Upgrades - SFR	4	SFR		x	Irrigation	Locally	No
Landscape Conversion or Turf Removal - MFR and CII	4	MFR, CII		x	Irrigation	Locally	No
Landscape Conversion or Turf Removal -SFR	4	SFR		x	Irrigation	Locally	No
Smart Irrigation Controller (Weather-Based Irrigation Controller) Rebates - Large Landscape	4	MFR, CII		x	Irrigation	Locally	Yes, currently
Smart Irrigation Controller (Weather-Based Irrigation Controller) Rebates - SFR	4	SFR		x	Irrigation	Locally	Yes, currently
Soil Moisture Sensor Rebate	4	All		x	Irrigation	Locally	No
Water Savings Incentive Program for CII	4	CII	x		All Indoor	Locally	No
High Efficiency Clothes Washer Rebate - Residential	3	SFR, MFR	x		Clothes Washer	No preference	Yes, currently
High Efficiency Clothes Washer Rebate Program - CII	3	CII	х		Clothes Washer	Locally	Yes, currently
Incentivize Gray Water Retrofit for Existing SFR Customers	3	SFR		х	Irrigation / Gray Water	Locally	Yes, currently
Incentivize Gray Water Systems for New CII Development	3	CII	х	x	Irrigation / Gray Water	Locally	No
Incentivize Replacement of Inefficient Commercial and Industrial Equipment	3	CII	х		CII Equipment	Locally	No
Indoor Fixture Program For Hotels & Motels	3	CII	х		All Indoor	Locally	Yes, previously
Indoor Fixture Program For Schools	3	CII	х		All Indoor	Locally	Yes, previously
Rain Barrel Rebate	3	SFR		х	Irrigation	Locally	Yes, previously
Rain Sensor Rebate	3	All		x	Irrigation	Locally	No
Rainwater Catchment System Rebate for Large Landscapes	3	MFR, CII		x	Irrigation	Locally	Yes, previously
Rotating Sprinkler Nozzle Giveaway	3	All		x	Irrigation	No preference	No
Rotating Sprinkler Nozzle Rebate	3	All		x	Irrigation	No preference	No
Tier 4 Exemption	3	SFR	x	x	toilet, Faucet, Showerhead, clothes washer, irrigation	Locally	Yes, currently
POLICIES AND REGULATIONS							
Require <0.25 gal/flush Urinals in New Development	5	CII	x		Urinal	Locally	Yes, currently
Require High Efficiency Clothes Washers in New Development	5	SFR, MFR	x		Clothes Washer	Locally	Yes, currently

Table 6-2
Prioritization of Conservation Measures and Programs
Marin Municipal Water District

Conservation Measure/Program	Prioritization (a)	Sector	Indoor	Outdoor	Primary End Use	Preference (b)	Local Program
Require On-Site Water Reuse Systems (Grey Water or Black Water) for Large CII Developments	5	CII		х	Irrigation / Recycled Water	Locally	Yes, currently
Require Plumbing for Gray Water in New SFR Development	5	SFR		х	Irrigation / Gray Water	Locally	Yes, currently
Require Swimming Pool and Hot Tub Covers	5	SFR, MFR		х	Pool/Hot Tub	Locally	Yes, currently
Require Water Efficiency Plan Reviews for New Cll Development	5	CII	х	х	All Indoor	Locally	Yes, currently
Require Weather Adjusting Smart Irrigation Controllers, Rain Sensors, and/or Soil Moisture Sensors in New Development	5	All		x	Irrigation	Locally	Yes, currently
Water Conserving Landscape and Irrigation Codes, More Stringent than MWELO	5	All		x	Irrigation	Locally	Yes, currently
Water Waste Ordinance	5	All		х	All Outdoor	No preference	Yes, currently
Require Submetering of Landscaping for New MFR and Commercial Developments	4	CII		х	Irrigation	Locally	No
Demand Offset/Water Neutral Policy for Large New Developments	3	All	х	х	All	Locally	No
Prohibit Once through Cooling Systems	3	CII	х	х	CII Equipment	Locally	No
Require <1.0 gal/flush Toilets in New Development	3	All	х		Toilet	Locally	No
Require Rain Barrels in New Development	3	SFR		х	Irrigation	Locally	No

Abbreviations:

- AMI = advanced metering infrastructure
- CII = commercial, industrial, institutional
- COM = commercial
- IRR = irrigation account
- MFR = multi-family residential
- MWELO = Model Water Efficient Landscape Ordinance
- PRV = pressure reducing valve
- SFR = single-family residential
- SMSWP = Sonoma-Marin Saving Water Partnership
- UHET = ultra high efficiency toilet

#### Notes:

(a) Each Water Contractor was asked to rank each conservation program or measure in terms of priority as a locally-administered program, where 5 indicated highest priority and 1 indicated the lowest priority.

(b) For each program a Water Contractor ranked as "3" or above, the Water Contractor was asked to indicate whether they would prefer the program to be administered regionally or locally. N/A indicates no preference given for programs given a ranking lower than three for both local and regional priority.

## Table 6-3a Conservation Program Scenarios

Marin Municipal Water District

			Program Scenario (a)					
Program	Sector	Indoor/ Outdoor	A) Outdoor Programs B) Highly- Ranked Loca Programs		C) Highly- Ranked Regional Programs			
High Efficiency Clothes Washer Rebate - Residential	SFR, MFR	Indoor			Х			
High Efficiency Faucet Aerator / Showerhead Giveaway - Residential Customers	SFR, MFR	Indoor		x	x			
Incentivize Irrigation Equipment Upgrades - Large Landscapes	MFR, CII, IRR	Outdoor	Х	X				
Incentivize Irrigation Equipment Upgrades - SFR	SFR	Outdoor	Х	X				
Landscape Conversion or Turf Removal - MFR and CII	MFR, CII	Outdoor	Х	X	Х			
Landscape Conversion or Turf Removal -SFR	SFR	Outdoor	Х	X	Х			
Restaurant Spray Nozzle Rebates	CII	Indoor			х			
Smart Irrigation Controller (Weather-Based Irrigation Controller) Rebates - Large Landscape	MFR, CII	Outdoor	x	x	x			
Smart Irrigation Controller (Weather-Based Irrigation Controller) Rebates - SFR	SFR	Outdoor	х	x				
Soil Moisture Sensor Rebate	All	Outdoor	Х	X				
Water Savings Incentive Program for CII	CII	Indoor		X				
Water Use Surveys/Audits - CII	CII	Both	Х		Х			
Water Use Surveys/Audits - MFR	MFR	Indoor		X				
Water Use Surveys/Audits - SFR	SFR	Both	х		Х			

#### **Abbreviations**

CII = Commercial, Industrial, and Institutional MFR = multi-family residential SFR = Single-family residential

# Table 6-3aConservation Program ScenariosMarin Municipal Water District

#### <u>Notes</u>

(a) Program scenarios represent three potential approaches to program selection. Scenario A represents a focus on outdoor water savings, Scenario B represents a more "business as usual" approach based on programs ranked most highly by Marin Municipal Water District, and Scenario C represents a focus on the programs all nine Water Contractors collectively identified as highest priority.

### Table 6-3b Costs and Savings of Potential Conservation Programs Marin Municipal Water District

Program (a)	Sector	Indoor/	Note	Net Present Value of Benefits		Net Present Value of Cost		Benefit to Cost Ratio		Water Utility Costs	Water Savings in	Water Utility Cost of Water
		outdoor		Water Utility	Customers	Water Utility	Customers	Water Utility Customers		2021-2025 (b)	2025 (AFY)	Saved (\$/AF)
High Efficiency Clothes Washer Rebate - Residential	SFR, MFR	Indoor	(c)	\$433,978	\$1,579,895	\$323,234	\$1,037,663	1.3	1.5	\$295,425	18	\$1,369
High Efficiency Faucet Aerator /												
Showerhead Giveaway - Residential Customers	SFR, MFR	Indoor		\$530,195	\$1,215,639	\$155,152	\$248,641	3.4	4.9	\$141,804	56	\$500
Incentivize Irrigation Equipment Upgrades - Large Landscapes	MFR, CII, IRR	Outdoor	(d)	\$234,582	\$650,198	\$149,349	\$57,442	1.6	11	\$136,500	12	\$1,137
Incentivize Irrigation Equipment Upgrades - SFR	SFR	Outdoor		\$64,542	\$145,132	\$65,785	\$50,604	1.0	2.9	\$60,125	3.2	\$1,820
Landscape Conversion or Turf Removal - MFR and CII	MFR, CII	Outdoor	(e)	\$310,238	\$774,891	\$594,324	\$457,172	0.52	1.7	\$543,192	15	\$3,421
Landscape Conversion or Turf Removal - SFR	SFR	Outdoor	(e)	\$103,413	\$232,537	\$198,108	\$152,391	0.52	1.5	\$181,064	5.1	\$3,421
Restaurant Spray Nozzle Rebates	CII	Indoor		\$318,529	\$1,218,458	\$17,780	\$13,677	18	89	\$16,250	34	\$95
Smart Irrigation Controller (Weather- Based Irrigation Controller) Rebates - Large Landscape	MFR, CII	Outdoor		\$179,392	\$497,227	\$35,916	\$18,662	5.0	27	\$32,826	8.9	\$358
Smart Irrigation Controller (Weather- Based Irrigation Controller) Rebates - SFR	SFR	Outdoor	(c)	\$197,545	\$444,206	\$50,167	\$34,895	3.9	13	\$45,851	10	\$454
Soil Moisture Sensor Rebate	All	Outdoor		\$363,376	\$784,437	\$52,628	\$60,724	6.9	13	\$48,100	7.9	\$284
Water Savings Incentive Program for CII	CII	Indoor		\$2,964,596	\$7,257,526	\$25,584	\$29,520	116	246	\$23,383	92	\$16
Water Use Surveys/Audits - CII	CII	Both		\$388,829	\$993,499	\$470,477	\$588,096	0.83	1.7	\$430,000	41	\$2,051
Water Use Surveys/Audits - MFR	MFR	Indoor		\$388,829	\$640,597	\$470,477	\$588,096	0.83	1.1	\$430,000	41	\$2,051
Water Use Surveys/Audits - SFR	SFR	Both		\$874,527	\$2,334,034	\$842,200	\$199,953	1.04	11.7	\$769,743	93	\$1,632

#### **Abbreviations**

AFY = acre-feet per year CII = Commercial, Industrial, and Institutional MFR = multi-family residential MMWD = Marin Municipal Water District SFR = Single-family residential sq ft = square feet WBIC = weather-based irrigation controller \$/AF = dollars per acre-foot

#### Table 6-3b

#### Costs and Savings of Potential Conservation Programs

Marin Municipal Water District

#### <u>Notes</u>

- (a) Estimated water savings, benefits, and costs are calculated using the AWE model. Assumptions used are presented in Appendix B.
- (b) For purposes of near-term conservation program analysis, it is assumed that all programs are active from 2021 through 2025.
- (c) Program savings are based on MMWD-specific estimates, which are derived from participant water savings based on their water bills.
- (d) Cost-effectiveness of this program is largely driven by the cost and type of equipment replaced. If the program was focused on certain equipment types, its cost-effectiveness would likely be increased.
- (e) Evaluation of this program assumed a rebate amount of \$1/square foot plus administrative cost. It is noted that a lower rebate would result in a benefit-cost ratio of greater than 1.

## Table 6-3c Comparison of Program Scenarios – Costs and Savings Marin Municipal Water District

Scenario (a)	Present Valu	e of Benefits	Present Value of Cost		Benefit to Cost Ratio		Cumulative Water Savings (AF)					Water Utility Cost of Water
	Water Utility	Customers	Water Utility	Customers	Water Utility	Customers	2025	2030	2035	2040	2045	Saved (\$/AF) (b)
A) Outdoor Programs	\$2,406,206	\$6,081,271	\$1,864,629	\$1,162,765	1.3	5.2	603	1,047	1,164	1,203	1,219	\$1,529
B) Highly-Ranked Local Programs	\$4,637,641	\$11,226,903	\$732,689	\$652,879	6.3	17	584	1,389	1,966	2,189	2,205	\$332
C) Highly-Ranked Regional Programs	\$2,828,862	\$8,071,289	\$2,042,867	\$2,259,082	1.4	3.6	829	1,364	1,450	1,469	1,469	\$1,390

#### **Abbreviations**

AF = acre-feet

\$/AF = dollars per acre-foot

#### <u>Notes</u>

- (a) For purposes of near-term conservation program analysis, it is assumed that all programs are active from 2021 through 2025. Cumulative water savings achieved beyond 2025 reflect the ongoing benefit of program implementation.
- (b) The water utility cost is based on the cumulative savings achieved through 2045 cumulative water savings.



### 7. CONCLUSION

This report presents the results of demand analysis and projections, developed consistent with CWC § 10631(d)(4)(A), which requires that "Water use projections, where available, shall display and account for the water savings estimated to result from adopted codes, standards, ordinances, or transportation and land use plans identified by the urban water supplier, as applicable to the service area." The assumptions used as the bases for demand projections were developed in close coordination with the District and reflect a land-use based approach consistent with the District's community planning, using the best available information. It should be noted that all demand and conservation projections have limitations and should be considered estimates that require revisiting as factors that affect demands arise, such as significant economic or population shifts, extreme hydrological conditions, etc.

The methodology used to develop demand projections herein is also consistent with the CWC §10635(b)(4), requirement to consider climate change on projected demands.<sup>25</sup> California experienced a historic drought between 2011-2017. In 2014, Governor Brown issued Executive Order B-26-14 declaring a Drought State of Emergency and requested all Californians to voluntarily reduce water use by 20%. In 2015, the State Water Resources Control Board implemented emergency conservation regulations that, among other things, required water agencies to reduce their water use and prohibited certain types of water uses. As a result, the District experienced an overall decrease in demands during the historic drought, most significantly during 2014. The demand factors evaluated herein consider both the 2011-2013 period, in which customers increased their water use (in part due to the drought conditions, prior to the imposed restrictions), as well as the observed rebound in demand following the drought (2017-2019). Thus, the periods used to develop the demand projections reflect conditions representative of the hotter, drier weather expected as a result of climate change.

<sup>&</sup>lt;sup>25</sup> CWC §10635(b)(4) requires that suppliers consider plausible changes on projected supplies and demands under climate change conditions specific to their five-year drought risk assessments. Section 4.5 of the draft 2020 UWMP Guidebook more generally recommends that consideration of climate change be incorporated into all demand projections.



#### 8. REFERENCES

- ABAG, 2013. Association of Bay Area Governments, Plan Bay Area Projections 2013, adopted on 18 July 2013.
- ABAG, 2018. Association of Bay Area Governments, Plan Bay Area Projections 2040, released on November 2018.
- Census, 2019. 2013-2017 American Community Survey (ACS) 5-year estimates. TIGER/Line Shapefiles by Block Group, https://www.census.gov/geo/maps-data/data/tiger-data.html, United States Census Bureau, downloaded on 14 January 2020.
- DOF, 2020. California Department of Finance Demographic Research Unit, Total Estimated and Projected Population for California and Counties: July 1, 2010 to July 1, 2060 in 1-year Increments, Report P-1, released on 10 January 2020.
- Marin Municipal Water District, 2016. Urban Water Management Plan 2015 Update, prepared by RMC Water and Environment, dated June 2016.
- Marin Municipal Water District, 2020. 2010-2019 Urban Water Management Plan Water Use Data, provided by Marin Municipal Water District on 9 July 2020.
- PG&E, 2020. Gas Rate Finder, Volume 48-G, No.3, dated in March 2020 (<u>https://www.pge.com/tariffs/GRF0320.pdf</u>).
- PG&E and Marin Clean Energy, 2020. Joint Rate Comparisons, dated May 2020 (<u>https://www.pge.com/pge\_global/common/pdfs/customer-service/other-services/alternative-energy-providers/community-choice-aggregation/mce\_rateclasscomparison.pdf</u>).



### Appendix A

California Water Code Revisions per AB-1668, SB-606, and SB-664, Redlines prepared by DWR



SB-664 Water: urban water management planning. (2015-2016)

As Amends the Law Today	As Amends the Law on Nov 20, 2015
<b>SECTION 1.</b> Section 10632.5 is added to the Water Code, to read:	
<b>10632.5.</b> (a) In addition to the requirements of paragraph (3) of subdivisio January 1, 2020, the plan shall include a seismic risk assessment and mitiga of each of the various facilities of a water system and mitigate those vulnerab	n (a) of Section 10632, beginning tion plan to assess the vulnerability ilities.
(b) An urban water supplier shall update the seismic risk assessment and urban water management plan as required by Section 10621.	mitigation plan when updating its
(c) An urban water supplier may comply with this section by submitting, pu the most recent adopted local hazard mitigation plan or multihazard mitigat Mitigation Act of 2000 (Public Law 106-390) if the local hazard mitigation p addresses seismic risk.	rsuant to Section 10644, a copy of ion plan under the federal Disaster plan or multihazard mitigation plan



AB-1668 Water management planning. (2017-2018)

As Amends the Law Today

As Amends the Law on Nov 08, 2018

**SECTION 1.** Section 531.10 of the Water Code is amended to read:

**531.10.** (a) (1) An agricultural water supplier shall submit an annual report to the department that summarizes aggregated farm-gate delivery data, on a monthly or bimonthly basis, using best professional practices. The annual report for the prior year shall be submitted to the department by April 1 of each year. The annual report shall be organized by basin, as defined in Section 10721, within the service area of the agricultural water supplier, if applicable.

(2) The report, and any amendments to the report, submitted to the department pursuant to this subdivision shall be submitted electronically and shall include any standardized forms, tables, or displays specified by the department.

(3) The department shall post all reports on its Internet Web site in a manner that allows for comparisons across water suppliers. The department shall make the reports available for public viewing in a timely manner after it receives them.

(b) Nothing in this article shall be construed to require the implementation of water measurement programs or practices that are not locally cost effective.

(c) It is the intent of the Legislature that the requirements of this section shall complement and not affect the scope of authority granted to the department or the board by provisions of law other than this article.

**SEC. 2.** Section 1120 of the Water Code is amended to read:

**1120.** This chapter applies to any decision or order issued under this part or Section 275, Part 2 (commencing with Section 1200), Part 2 (commencing with Section 10500) of Division 6, Part 2.55 (commencing with Section 10608) of Division 6, or Chapter 11 (commencing with Section 10735) of Part 2.74 of Division 6, Article 7 (commencing with Section 13550) of Chapter 7 of Division 7, or the public trust doctrine.

**SEC. 3.** Section 1846.5 is added to the Water Code, to read:

**1846.5.** (*a*) An urban retail water supplier who commits any of the violations identified in subdivision (b) may be liable in an amount not to exceed the following, as applicable:

(1) If the violation occurs in a critically dry year immediately preceded by two or more consecutive below normal, dry, or critically dry years or during a period for which the Governor has issued a proclamation of a state of emergency under the California Emergency Services Act (Chapter 7 (commencing with Section 8550) of Division 1 of Title 2 of the Government Code) based on drought conditions, ten thousand dollars (\$10,000) for each day in which the violation occurs.

(2) For all violations other than those described in paragraph (1), one thousand dollars (\$1,000) for each day in which the violation occurs.

(b) Liability pursuant to this section may be imposed for any of the following violations:

(1) Violation of an order issued under Chapter 9 (commencing with Section 10609) of Part 2.55 of Division 6.

(2) Violation of a regulation issued under Chapter 9 (commencing with Section 10609) of Part 2.55 of Division 6, if the violation occurs after November 1, 2027.
(c) Civil liability may be imposed by the superior court. The Attorney General, upon the request of the board, shall petition the superior court to impose, assess, and recover those sums.

(d) Civil liability may be imposed administratively by the board pursuant to Section 1055.

SEC. 4. Section 10608.12 of the Water Code is amended to read:

10608.12. Unless the context otherwise requires, the following definitions govern the construction of this part:

(a) "Agricultural water supplier" means a water supplier, either publicly or privately owned, providing water to 10,000 or more irrigated acres, excluding recycled water. "Agricultural water supplier" includes a supplier or contractor for water, regardless of the basis of right, that distributes or sells water for ultimate resale to customers. "Agricultural water supplier" does not include the department.

(b) "Base daily per capita water use" means any of the following:

(1) The urban retail water supplier's estimate of its average gross water use, reported in gallons per capita per day and calculated over a continuous 10-year period ending no earlier than December 31, 2004, and no later than December 31, 2010.

(2) For an urban retail water supplier that meets at least 10 percent of its 2008 measured retail water demand through recycled water that is delivered within the service area of an urban retail water supplier or its urban wholesale water supplier, the urban retail water supplier may extend the calculation described in paragraph (1) up to an additional five years to a maximum of a continuous 15-year period ending no earlier than December 31, 2004, and no later than December 31, 2010.

(3) For the purposes of Section 10608.22, the urban retail water supplier's estimate of its average gross water use, reported in gallons per capita per day and calculated over a continuous five-year period ending no earlier than December 31, 2007, and no later than December 31, 2010.

(c) "Baseline commercial, industrial, and institutional water use" means an urban retail water supplier's base daily per capita water use for commercial, industrial, and institutional users.

(d) "CII water use" means water used by commercial water users, industrial water users, institutional water users, and large landscape water users.

(e) "Commercial water user" means a water user that provides or distributes a product or service.

(f) "Compliance daily per capita water use" means the gross water use during the final year of the reporting period, reported in gallons per capita per day.

(g) "Disadvantaged community" means a community with an annual median household income that is less than 80 percent of the statewide annual median household income.

(h) "Gross water use" means the total volume of water, whether treated or untreated, entering the distribution system of an urban retail water supplier, excluding all of the following:

(1) Recycled water that is delivered within the service area of an urban retail water supplier or its urban wholesale water supplier.

(2) The net volume of water that the urban retail water supplier places into long-term storage.

(3) The volume of water the urban retail water supplier conveys for use by another urban water supplier.

(4) The volume of water delivered for agricultural use, except as otherwise provided in subdivision (f) of Section 10608.24.

(i) "Industrial water user" means a water user that is primarily a manufacturer or processor of materials as defined by the North American Industry Classification System code sectors 31 to 33, inclusive, or an entity that is a water user primarily engaged in research and development.

(j) "Institutional water user" means a water user dedicated to public service. This type of user includes, among other users, higher education institutions, schools, courts, churches, hospitals, government facilities, and nonprofit research institutions.

(k) "Interim urban water use target" means the midpoint between the urban retail water supplier's base daily per capita water use and the urban retail water supplier's urban water use target for 2020.

(I) "Large landscape" means a nonresidential landscape as described in the performance measures for CII water use adopted pursuant to Section 10609.10.

(m) "Locally cost effective" means that the present value of the local benefits of implementing an agricultural efficiency water management practice is greater than or equal to the present value of the local cost of implementing that measure.

(n) "Performance measures" means actions to be taken by urban retail water suppliers that will result in increased water use efficiency by CII water users. Performance measures may include, but are not limited to, educating CII water users on best management practices, conducting water use audits, and preparing water management plans. Performance measures do not include process water.

(o) "Potable reuse" means direct potable reuse, indirect potable reuse for groundwater recharge, and reservoir water augmentation as those terms are defined in Section 13561.

(p) "Process water" means water used by industrial water users for producing a product or product content or water used for research and development. Process water includes, but is not limited to, continuous manufacturing processes, and water used for testing, cleaning, and maintaining equipment. Water used to cool machinery or buildings used in the manufacturing process or necessary to maintain product quality or chemical characteristics for product manufacturing or control rooms, data centers, laboratories, clean rooms, and other industrial facility units that are integral to the manufacturing or research and development process is process water. Water used in the manufacturing process that is necessary for complying with local, state, and federal health and safety laws, and is not incidental water, is process water. Process water does not mean incidental water uses.

(q) "Recycled water" means recycled water, as defined in subdivision (n) of Section 13050.

(r) "Regional water resources management" means sources of supply resulting from watershed-based planning for sustainable local water reliability or any of the following alternative sources of water:

(1) The capture and reuse of stormwater or rainwater.

(2) The use of recycled water.

(3) The desalination of brackish groundwater.

(4) The conjunctive use of surface water and groundwater in a manner that is consistent with the safe yield of the groundwater basin.

(s) "Reporting period" means the years for which an urban retail water supplier reports compliance with the urban water use targets.

(t) "Urban retail water supplier" means a water supplier, either publicly or privately owned, that directly provides potable municipal water to more than 3,000 end users or that supplies more than 3,000 acre-feet of potable water annually at retail for municipal purposes.

(u) "Urban water use objective" means an estimate of aggregate efficient water use for the previous year based on adopted water use efficiency standards and local service area characteristics for that year, as described in Section 10609.20.

(v) "Urban water use target" means the urban retail water supplier's targeted future daily per capita water use.

(w) "Urban wholesale water supplier" supplier, " means a water supplier, either publicly or privately owned, that provides more than 3,000 acre-feet of water annually at wholesale for potable municipal purposes.

SEC. 5. Section 10608.20 of the Water Code is amended to read:

**10608.20.** (a) (1) Each urban retail water supplier shall develop urban water use targets and an interim urban water use target by July 1, 2011. Urban retail water suppliers may elect to determine and report progress toward achieving these targets on an individual or regional basis, as provided in subdivision (a) of Section 10608.28, and may determine the targets on a fiscal year or calendar year basis.

(2) It is the intent of the Legislature that the urban water use targets described in paragraph (1) cumulatively result in a 20-percent reduction from the baseline daily per capita water use by December 31, 2020.

(b) An urban retail water supplier shall adopt one of the following methods for determining its urban water use target pursuant to subdivision (a):

(1) Eighty percent of the urban retail water supplier's baseline per capita daily water use.

(2) The per capita daily water use that is estimated using the sum of the following performance standards:

(A) For indoor residential water use, 55 gallons per capita daily water use as a provisional standard. Upon completion of the department's  $\frac{2017}{2016}$  report to the Legislature pursuant to Section 10608.42, this standard may be adjusted by the Legislature by statute.

(B) For landscape irrigated through dedicated or residential meters or connections, water efficiency equivalent to the standards of the Model Water Efficient Landscape Ordinance set forth in Chapter 2.7 (commencing with Section 490) of Division 2 of Title 23 of the California Code of Regulations, as in effect the later of the year of the landscape's installation or 1992. An urban retail water supplier using the approach specified in this subparagraph shall use satellite imagery, site visits, or other best available technology to develop an accurate estimate of landscaped areas.

(C) For commercial, industrial, and institutional uses, a 10-percent reduction in water use from the baseline commercial, industrial, and institutional water use by 2020.

(3) Ninety-five percent of the applicable state hydrologic region target, as set forth in the state's draft 20x2020 Water Conservation Plan (dated April 30, 2009). If the service area of an urban water supplier includes more than one hydrologic region, the supplier shall apportion its service area to each region based on population or area.

(4) A method that shall be identified and developed by the department, through a public process, and reported to the Legislature no later than December 31, 2010. The method developed by the department shall identify per capita targets that cumulatively result in a statewide 20-percent reduction in urban daily per capita water use by December 31, 2020. In developing urban daily per capita water use targets, the department shall do all of the following:

(A) Consider climatic differences within the state.

(B) Consider population density differences within the state.

(C) Provide flexibility to communities and regions in meeting the targets.

(D) Consider different levels of per capita water use according to plant water needs in different regions.

(E) Consider different levels of commercial, industrial, and institutional water use in different regions of the state.

(F) Avoid placing an undue hardship on communities that have implemented conservation measures or taken actions to keep per capita water use low.

(c) If the department adopts a regulation pursuant to paragraph (4) of subdivision (b) that results in a requirement that an urban retail water supplier achieve a reduction in daily per capita water use that is greater than 20 percent by December 31, 2020, an urban retail water supplier that adopted the method described in paragraph (4) of subdivision (b) may limit its urban water use target to a reduction of not more than 20 percent by December 31, 2020, by adopting the method described in paragraph (1) of subdivision (b).

(d) The department shall update the method described in paragraph (4) of subdivision (b) and report to the Legislature by December 31, 2014. An urban retail water supplier that adopted the method described in paragraph (4) of subdivision (b) may adopt a new urban daily per capita water use target pursuant to this updated method.

(e) An urban retail water supplier shall include in its urban water management plan due in 2010 pursuant to Part 2.6 (commencing with Section 10610) the baseline daily per capita water use, urban water use target, interim urban water use target, and compliance daily per capita water use, along with the bases for determining those estimates, including references to supporting data.

(f) When calculating per capita values for the purposes of this chapter, an urban retail water supplier shall determine population using federal, state, and local population reports and projections.

(g) An urban retail water supplier may update its 2020 urban water use target in its 2015 urban water management plan required pursuant to Part 2.6 (commencing with Section 10610).

(h) (1) The department, through a public process and in consultation with the California Urban Water Conservation Council, shall develop technical methodologies and criteria for the consistent implementation of this part, including, but not limited to, both of the following:

(A) Methodologies for calculating base daily per capita water use, baseline commercial, industrial, and institutional water use, compliance daily per capita water use, gross water use, service area population, indoor residential water use, and landscaped area water use.

(B) Criteria for adjustments pursuant to subdivisions (d) and (e) of Section 10608.24.

(2) The department shall post the methodologies and criteria developed pursuant to this subdivision on its internet website, Internet Web site, and make written copies available, by October 1, 2010. An urban retail water supplier shall use the methods developed by the department in compliance with this part.

(i) (1) The department shall adopt regulations for implementation of the provisions relating to process water in accordance with Section 10608.12, subdivision (e) of Section 10608.24, and subdivision (d) of Section 10608.26.

(2) The initial adoption of a regulation authorized by this subdivision is deemed to address an emergency, for purposes of Sections 11346.1 and 11349.6 of the Government Code, and the department is hereby exempted for that purpose from the requirements of subdivision (b) of Section 11346.1 of the Government Code. After the initial adoption of an emergency regulation pursuant to this subdivision, the department shall not request approval from the Office of Administrative Law to readopt the regulation as an emergency regulation pursuant to Section 11346.1 of the Government Code.

(j) (1) An urban retail water supplier is granted an extension to July 1, 2011, for adoption of an urban water management plan pursuant to Part 2.6 (commencing with Section 10610) due in 2010 to allow the use of technical methodologies developed by the department pursuant to paragraph (4) of subdivision (b) and subdivision (h). An urban retail water supplier that adopts an urban water management plan due in 2010 that does not use the methodologies developed by the department pursuant to subdivision (h) shall amend the plan by July 1, 2011, to comply with this part.

(2) An urban wholesale water supplier whose urban water management plan prepared pursuant to Part 2.6 (commencing with Section 10610) was due and not submitted in 2010 is granted an extension to July 1, 2011, to permit coordination between an urban wholesale water supplier and urban retail water suppliers.

**SEC. 6.** Section 10608.48 of the Water Code is amended to read:

**10608.48.** (a) On or before July 31, 2012, an agricultural water supplier shall implement efficient water management practices pursuant to subdivisions (b) and (c).

(b) Agricultural water suppliers shall implement both of the following critical efficient management practices:

(1) Measure the volume of water delivered to customers with sufficient accuracy to comply with subdivision (a) of Section 531.10 and to implement paragraph (2).

(2) Adopt a pricing structure for water customers based at least in part on quantity delivered.

(c) Agricultural water suppliers shall implement additional efficient management practices, including, but not limited to, practices to accomplish all of the following, if the measures are locally cost effective and technically feasible:

(1) Facilitate alternative land use for lands with exceptionally high water duties or whose irrigation contributes to significant problems, including drainage.

(2) Facilitate use of available recycled water that otherwise would not be used beneficially, meets all health and safety criteria, and does not harm crops or soils.

(3) Facilitate the financing of capital improvements for on-farm irrigation systems.

(4) Implement an incentive pricing structure that promotes one or more of the following goals:

(A) More efficient water use at the farm level.

(B) Conjunctive use of groundwater.

(C) Appropriate increase of groundwater recharge.

(D) Reduction in problem drainage.

(E) Improved management of environmental resources.

(F) Effective management of all water sources throughout the year by adjusting seasonal pricing structures based on current conditions.

(5) Expand line or pipe distribution systems, and construct regulatory reservoirs to increase distribution system flexibility and capacity, decrease maintenance, and reduce seepage.

(6) Increase flexibility in water ordering by, and delivery to, water customers within operational limits.

(7) Construct and operate supplier spill and tailwater recovery systems.

(8) Increase planned conjunctive use of surface water and groundwater within the supplier service area.

(9) Automate canal control structures.

(10) Facilitate or promote customer pump testing and evaluation.

(11) Designate a water conservation coordinator who will develop and implement the water management plan and prepare progress reports.

(12) Provide for the availability of water management services to water users. These services may include, but are not limited to, all of the following:

(A) On-farm irrigation and drainage system evaluations.

(B) Normal year and real-time irrigation scheduling and crop evapotranspiration information.

(C) Surface water, groundwater, and drainage water quantity and quality data.

(D) Agricultural water management educational programs and materials for farmers, staff, and the public.

(13) Evaluate the policies of agencies that provide the supplier with water to identify the potential for institutional changes to allow more flexible water deliveries and storage.

(14) Evaluate and improve the efficiencies of the supplier's pumps.

(d) Agricultural water suppliers shall include in the agricultural water management plans required pursuant to Part 2.8 (commencing with Section 10800) a report on which efficient water management practices have been implemented and are planned to be implemented, an estimate of the water use efficiency improvements that have occurred since the last report, and an estimate of the water use efficiency improvements estimated to occur five and 10 years in the future. If an agricultural water supplier determines that an efficient water management practice is not locally cost effective or technically feasible, the supplier shall submit information documenting that determination.

(e) The department shall require information about the implementation of efficient water management practices to be reported using a standardized form developed pursuant to Section 10608.52.

(f) An agricultural water supplier may meet the requirements of subdivisions (d) and (e) by submitting to the department a water conservation plan submitted to the United States Bureau of Reclamation that meets the requirements described in Section 10828.

(g) On or before December 31, 2013, December 31, 2016, and December 31, 2021, the department, in consultation with the board, shall submit to the Legislature a report on the agricultural efficient water management practices that have been implemented and are planned to be implemented and an assessment of the manner in which the implementation of those efficient water management practices has affected and will affect agricultural operations, including estimated water use efficiency improvements, if any.

(h) The department may update the efficient water management practices required pursuant to subdivision (c), in consultation with the Agricultural Water Management Council, the United States Bureau of Reclamation, and the board. All efficient water management practices for agricultural water use pursuant to this chapter shall be adopted or revised by the department only after the department conducts public hearings to allow participation of the diverse geographical areas and interests of the state.

(i) (1) The department shall adopt regulations that provide for a range of options that agricultural water suppliers may use or implement to comply with the measurement requirement in paragraph (1) of subdivision (b).

(2) The initial adoption of a regulation authorized by this subdivision is deemed to address an emergency, for purposes of Sections 11346.1 and 11349.6 of the Government Code, and the department is hereby exempted for that purpose from the requirements of subdivision (b) of Section 11346.1 of the Government Code. After the initial adoption of an emergency regulation pursuant to this subdivision, the department shall not request approval from the Office of Administrative Law to readopt the regulation as an emergency regulation pursuant to Section 11346.1 of the Government Code.

**SEC. 7.** Chapter 9 (commencing with Section 10609) is added to Part 2.55 of Division 6 of the Water Code, to read:

# CHAPTER 9. Urban Water Use Objectives and Water Use Reporting

**10609.** (a) The Legislature finds and declares that this chapter establishes a method to estimate the aggregate amount of water that would have been delivered the previous year by an urban retail water supplier if all that water had been used efficiently. This estimated aggregate water use is the urban retail water supplier's urban water use objective. The method is based on water use efficiency standards and local service area characteristics for that year. By comparing the amount of water actually used in the previous year with the urban water use objective, local urban water suppliers will be in a better position to help eliminate unnecessary use of water; that is, water used in excess of that needed to accomplish the intended beneficial use.

- (b) The Legislature further finds and declares all of the following:
- (1) This chapter establishes standards and practices for the following water uses:
- (A) Indoor residential use.
- (B) Outdoor residential use.
- (C) CII water use.
- (D) Water losses.

(E) Other unique local uses and situations that can have a material effect on an urban water supplier's total water use.

(2) This chapter further does all of the following:

- (A) Establishes a method to calculate each urban water use objective.
- (B) Considers recycled water quality in establishing efficient irrigation standards.

(C) Requires the department to provide or otherwise identify data regarding the unique local conditions to support the calculation of an urban water use objective.

(D) Provides for the use of alternative sources of data if alternative sources are shown to be as accurate as, or more accurate than, the data provided by the department.

(E) Requires annual reporting of the previous year's water use with the urban water use objective.

(F) Provides a bonus incentive for the amount of potable recycled water used the previous year when comparing the previous year's water use with the urban water use objective, of up to 10 percent of the urban water use objective.

(3) This chapter requires the department and the board to solicit broad public participation from stakeholders and other interested persons in the development of the standards and the adoption of regulations pursuant to this chapter.

(4) This chapter preserves the Legislature's authority over long-term water use efficiency target setting and ensures appropriate legislative oversight of the implementation of this chapter by doing all of the following:

(A) Requiring the Legislative Analyst to conduct a review of the implementation of this act, including compliance with the adopted standards and regulations, accuracy of the data, use of alternate data, and other issues the Legislative Analyst deems appropriate.

(*B*) Stating legislative intent that the director of the department and the chairperson of the board appear before the appropriate Senate and Assembly policy committees to report on progress in implementing this chapter.

(C) Providing one-time-only authority to the department and board to adopt water use efficiency standards, except as explicitly provided in this chapter. Authorization to update the standards shall require separate legislation.

(c) It is the intent of the Legislature that the following principles apply to the development and implementation of long-term standards and urban water use objectives:

(1) Local urban retail water suppliers should have primary responsibility for meeting standards-based water use targets, and they shall retain the flexibility to develop their water supply portfolios, design and implement water conservation strategies, educate their customers, and enforce their rules.

(2) Long-term standards and urban water use objectives should advance the state's goals to mitigate and adapt to climate change.

(3) Long-term standards and urban water use objectives should acknowledge the shade, air quality, and heatisland reduction benefits provided to communities by trees through the support of water-efficient irrigation practices that keep trees healthy.

(4) The state should identify opportunities for streamlined reporting, eliminate redundant data submissions, and incentivize open access to data collected by urban and agricultural water suppliers.

**10609.2.** (a) The board, in coordination with the department, shall adopt long-term standards for the efficient use of water pursuant to this chapter on or before June 30, 2022.

(b) Standards shall be adopted for all of the following:

(1) Outdoor residential water use.

(2) Outdoor irrigation of landscape areas with dedicated irrigation meters in connection with CII water use.

(3) A volume for water loss.

(c) When adopting the standards under this section, the board shall consider the policies of this chapter and the proposed efficiency standards' effects on local wastewater management, developed and natural parklands, and urban tree health. The standards and potential effects shall be identified by May 30, 2022. The board shall allow for public comment on potential effects identified by the board under this subdivision.

(d) The long-term standards shall be set at a level designed so that the water use objectives, together with other demands excluded from the long-term standards such as CII indoor water use and CII outdoor water use not connected to a dedicated landscape meter, would exceed the statewide conservation targets required pursuant to Chapter 3 (commencing with Section 10608.16).

(e) The board, in coordination with the department, shall adopt by regulation variances recommended by the department pursuant to Section 10609.14 and guidelines and methodologies pertaining to the calculation of an urban retail water supplier's urban water use objective recommended by the department pursuant to Section 10609.16.

**10609.4.** (a) (1) Until January 1, 2025, the standard for indoor residential water use shall be 55 gallons per capita daily.

(2) Beginning January 1, 2025, and until January 1, 2030, the standard for indoor residential water use shall be the greater of 52.5 gallons per capita daily or a standard recommended pursuant to subdivision (b).

(3) Beginning January 1, 2030, the standard for indoor residential water use shall be the greater of 50 gallons per capita daily or a standard recommended pursuant to subdivision (b).

(b) (1) The department, in coordination with the board, shall conduct necessary studies and investigations and may jointly recommend to the Legislature a standard for indoor residential water use that more appropriately reflects best practices for indoor residential water use than the standard described in subdivision (a). A report on the results of the studies and investigations shall be made to the chairpersons of the relevant policy committees of each house of the Legislature by January 1, 2021, and shall include information necessary to support the recommended standard, if there is one. The studies and investigations shall also include an analysis of the benefits and impacts of how the changing standard for indoor residential water use will impact water and wastewater management, including potable water usage, wastewater, recycling and reuse systems, infrastructure, operations, and supplies.

(2) The studies, investigations, and report described in paragraph (1) shall include collaboration with, and input from, a broad group of stakeholders, including, but not limited to, environmental groups, experts in indoor plumbing, and water, wastewater, and recycled water agencies.

**10609.6.** (a) (1) The department, in coordination with the board, shall conduct necessary studies and investigations and recommend, no later than October 1, 2021, standards for outdoor residential use for adoption by the board in accordance with this chapter.

(2) (A) The standards shall incorporate the principles of the model water efficient landscape ordinance adopted by the department pursuant to the Water Conservation in Landscaping Act (Article 10.8 (commencing with Section 65591) of Chapter 3 of Division 1 of Title 7 of the Government Code).

(B) The standards shall apply to irrigable lands.

(C) The standards shall include provisions for swimming pools, spas, and other water features. Ornamental water features that are artificially supplied with water, including ponds, lakes, waterfalls, and fountains, shall be analyzed separately from swimming pools and spas.

(b) The department shall, by January 1, 2021, provide each urban retail water supplier with data regarding the area of residential irrigable lands in a manner that can reasonably be applied to the standards adopted pursuant to this section.

(c) The department shall not recommend standards pursuant to this section until it has conducted pilot projects or studies, or some combination of the two, to ensure that the data provided to local agencies are reasonably accurate for the data's intended uses, taking into consideration California's diverse landscapes and community characteristics.

**10609.8.** (a) The department, in coordination with the board, shall conduct necessary studies and investigations and recommend, no later than October 1, 2021, standards for outdoor irrigation of landscape areas with dedicated irrigation meters or other means of calculating outdoor irrigation use in connection with CII water use for adoption by the board in accordance with this chapter.

(b) The standards shall incorporate the principles of the model water efficient landscape ordinance adopted by the department pursuant to the Water Conservation in Landscaping Act (Article 10.8 (commencing with Section 65591) of Chapter 3 of Division 1 of Title 7 of the Government Code).

(c) The standards shall include an exclusion for water for commercial agricultural use meeting the definition of subdivision (b) of Section 51201 of the Government Code.

**10609.9.** For purposes of Sections 10609.6 and 10609.8, "principles of the model water efficient landscape ordinance" means those provisions of the model water efficient landscape ordinance applicable to the establishment or determination of the amount of water necessary to efficiently irrigate both new and existing landscapes. These provisions include, but are not limited to, all of the following:

(a) Evapotranspiration adjustment factors, as applicable.

- (b) Landscape area.
- (c) Maximum applied water allowance.
- (d) Reference evapotranspiration.

(e) Special landscape areas, including provisions governing evapotranspiration adjustment factors for different types of water used for irrigating the landscape.

**10609.10.** (a) The department, in coordination with the board, shall conduct necessary studies and investigations and recommend, no later than October 1, 2021, performance measures for CII water use for adoption by the board in accordance with this chapter.

(b) Prior to recommending performance measures for CII water use, the department shall solicit broad public participation from stakeholders and other interested persons relating to all of the following:

(1) Recommendations for a CII water use classification system for California that address significant uses of water.

(2) Recommendations for setting minimum size thresholds for converting mixed CII meters to dedicated irrigation meters, and evaluation of, and recommendations for, technologies that could be used in lieu of requiring dedicated irrigation meters.

(3) Recommendations for CII water use best management practices, which may include, but are not limited to, water audits and water management plans for those CII customers that exceed a recommended size, volume of water use, or other threshold.

(c) Recommendations of appropriate performance measures for CII water use shall be consistent with the October 21, 2013, report to the Legislature by the Commercial, Industrial, and Institutional Task Force entitled "Water Use Best Management Practices," including the technical and financial feasibility recommendations provided in that report, and shall support the economic productivity of California's commercial, industrial, and institutional sectors.

(*d*) (1) The board, in coordination with the department, shall adopt performance measures for CII water use on or before June 30, 2022.

(2) Each urban retail water supplier shall implement the performance measures adopted by the board pursuant to paragraph (1).

**10609.12.** The standards for water loss for urban retail water suppliers shall be the standards adopted by the board pursuant to subdivision (i) of Section 10608.34.

**10609.14.** (a) The department, in coordination with the board, shall conduct necessary studies and investigations and, no later than October 1, 2021, recommend for adoption by the board in accordance with this chapter appropriate variances for unique uses that can have a material effect on an urban retail water supplier's urban water use objective.

(b) Appropriate variances may include, but are not limited to, allowances for the following:

(1) Significant use of evaporative coolers.

(2) Significant populations of horses and other livestock.

(3) Significant fluctuations in seasonal populations.

(4) Significant landscaped areas irrigated with recycled water having high levels of total dissolved solids.

(5) Significant use of water for soil compaction and dust control.

(6) Significant use of water to supplement ponds and lakes to sustain wildlife.

(7) Significant use of water to irrigate vegetation for fire protection.

(8) Significant use of water for commercial or noncommercial agricultural use.

(c) The department, in recommending variances for adoption by the board, shall also recommend a threshold of significance for each recommended variance.

(*d*) Before including any specific variance in calculating an urban retail water supplier's water use objective, the urban retail water supplier shall request and receive approval by the board for the inclusion of that variance.

(e) The board shall post on its Internet Web site all of the following:

(1) A list of all urban retail water suppliers with approved variances.

(2) The specific variance or variances approved for each urban retail water supplier.

(3) The data supporting approval of each variance.

**10609.15.** To help streamline water data reporting, the department and the board shall do all of the following:

(a) Identify urban water reporting requirements shared by both agencies, and post on each agency's Internet Web site how the data is used for planning, regulatory, or other purposes.

(b) Analyze opportunities for more efficient publication of urban water reporting requirements within each agency, and analyze how each agency can integrate various data sets in a publicly accessible location, identify priority actions, and implement priority actions identified in the analysis.

(c) Make appropriate data pertaining to the urban water reporting requirements that are collected by either agency available to the public according to the principles and requirements of the Open and Transparent Water Data Act (Part 4.9 (commencing with Section 12400)).

**10609.16.** The department, in coordination with the board, shall conduct necessary studies and investigations and recommend, no later than October 1, 2021, guidelines and methodologies for the board to adopt that identify how an urban retail water supplier calculates its urban water use objective. The guidelines and methodologies shall address, as necessary, all of the following:

(a) Determining the irrigable lands within the urban retail water supplier's service area.

(b) Updating and revising methodologies described pursuant to subparagraph (A) of paragraph (1) of subdivision (h) of Section 10608.20, as appropriate, including methodologies for calculating the population in an urban retail water supplier's service area.

(c) Using landscape area data provided by the department or alternative data.

(d) Incorporating precipitation data and climate data into estimates of a urban retail water supplier's outdoor irrigation budget for its urban water use objective.

(e) Estimating changes in outdoor landscape area and population, and calculating the urban water use objective, for years when updated landscape imagery is not available from the department.

(f) Determining acceptable levels of accuracy for the supporting data, the urban water use objective, and compliance with the urban water use objective.

**10609.18.** The department and the board shall solicit broad public participation from stakeholders and other interested persons in the development of the standards and the adoption of regulations pursuant to this chapter. The board shall hold at least one public meeting before taking any action on any standard or variance recommended by the department.

**SEC. 8.** Chapter 10 (commencing with Section 10609.40) is added to Part 2.55 of Division 6 of the Water Code, to read:

**CHAPTER 10. Countywide Drought and Water Shortage Contingency Plans 10609.40.** The Legislature finds and declares both of the following:

(a) Small water suppliers and rural communities are often not covered by established water shortage planning requirements. Currently, most counties do not address water shortages or do so minimally in their general plan or the local hazard mitigation plan.

(b) The state should provide guidance to improve drought planning for small water suppliers and rural communities.

**10609.42.** (a) No later than January 1, 2020, the department, in consultation with the board and other relevant state and local agencies and stakeholders, shall use available data to identify small water suppliers and rural communities that may be at risk of drought and water shortage vulnerability. The department shall notify counties and groundwater sustainability agencies of those suppliers or communities that may be at risk within its jurisdiction, and may make the information publicly accessible on its Internet Web site.

(b) The department shall, in consultation with the board, by January 1, 2020, propose to the Governor and the Legislature recommendations and guidance relating to the development and implementation of countywide drought and water shortage contingency plans to address the planning needs of small water suppliers and rural communities. The department shall recommend how these plans can be included in county local hazard

mitigation plans or otherwise integrated with complementary existing planning processes. The guidance from the department shall outline goals of the countywide drought and water shortage contingency plans and recommend components including, but not limited to, all of the following:

(1) Assessment of drought vulnerability.

(2) Actions to reduce drought vulnerability.

(3) Response, financing, and local communication and outreach planning efforts that may be implemented in times of drought.

(4) Data needs and reporting.

(5) Roles and responsibilities of interested parties and coordination with other relevant water management planning efforts.

(c) In formulating the proposal, the department shall utilize a public process involving state agencies, cities, counties, small communities, small water suppliers, and other stakeholders.

SEC. 9. Section 10801 of the Water Code is amended to read:

**10801.** The Legislature finds and declares all of the following:

(a) The waters of the state are a limited and renewable resource.

(b) The California Constitution requires that water in the state be used in a reasonable and beneficial manner.

(c) The efficient use of agricultural water supplies is of great statewide concern.

(d) There is a great amount of reuse of delivered water, both inside and outside the water service areas of agricultural water suppliers.

(e) Significant noncrop beneficial uses are associated with agricultural water use, including the preservation and enhancement of fish and wildlife resources.

(f) Significant opportunities exist in some areas, through improved irrigation water management, to conserve water or to reduce the quantity of highly saline or toxic drainage water.

(g) Changes in water management practices should be carefully planned and implemented to minimize adverse effects on other beneficial uses currently being served.

(h) Agricultural water suppliers that receive water from the federal Central Valley Project are required by federal law to prepare and implement water conservation plans.

(i) Agricultural water users applying for a permit to appropriate water from the board are required to prepare and implement water conservation plans.

SEC. 10. Section 10802 of the Water Code is amended to read:

**10802.** The Legislature finds and declares that all of the following are the policies of the state:

(a) The efficient use of water shall be pursued actively to protect both the people of the state and the state's water resources.

(b) The efficient use of agricultural water supplies shall be an important criterion in public decisions with regard to water.

(c) Agricultural water suppliers shall be required to prepare water management plans to achieve greater efficiency in the use of water.

SEC. 11. Section 10814 of the Water Code is amended to read:

**10814.** "Person" has the same meaning as defined in Section 10614.

SEC. 12. Section 10817 of the Water Code is amended to read:

**10817.** "Water use efficiency" means the efficient management of water resources for beneficial uses, preventing waste, or accomplishing additional benefits with the same amount of water.

**SEC. 13.** Section 10820 of the Water Code is amended to read:

**10820.** (a) (1) Except as provided in paragraph (2), an agricultural water supplier shall prepare and adopt an agricultural water management plan in the manner set forth in this chapter on or before December 31, 2012, and shall update that plan on December 31, 2015.

(2) (A) The agricultural water management plan shall be updated on or before April 1, 2021, and thereafter on or before April 1 in the years ending in six and one. The plan shall satisfy the requirements of Section 10826.

(B) An agricultural water supplier shall submit its plan to the department no later than 30 days after the adoption of the plan. The plan shall be submitted electronically and shall include any standardized forms, tables, or displays specified by the department.

(b) (1) The department shall review each plan that is due pursuant to paragraph (2) of subdivision (a). The department may coordinate its review with the Department of Food and Agriculture and the board.

(2) The department shall notify an agricultural water supplier that it is not in compliance with this part if the department determines that actions are required to comply with the requirements of this part or if a supplier fails to update a plan as provided in paragraph (2) of subdivision (a). The department shall identify the specific deficiencies and the supplier shall have 120 days to remedy an identified deficiency. The department may provide additional time to remedy a deficiency if it finds that a supplier is making substantial progress toward remedying the deficiency. An agricultural water supplier that fails to submit corrective actions or a completed plan shall not be in compliance with this part.

(3) If the department has not received a plan or the department has determined that the plan submitted does not comply with the requirements of this part, and a revised plan has not been submitted, the department may undertake the following actions:

(A) Contract with a state academic institution or qualified entity to prepare or complete an agricultural water management plan on behalf of the supplier. The costs and expenses related to preparation or completion of a plan, including the costs of the contract and contract administration, shall be recoverable by the department from the supplier.

(B) If a supplier does not provide data necessary for the preparation or completion of a plan to the department or the contracting entity as determined by the department in accordance with subparagraph (A), the department may assess a fine of one thousand dollars (\$1,000) per day, not to exceed twenty-five thousand dollars (\$25,000), until data is made available.

(4) (A) A plan prepared or completed pursuant to paragraph (3) shall be deemed the adopted plan for the supplier.

(B) Any action to challenge or invalidate the adequacy of the plan prepared or completed pursuant to paragraph(3) shall be brought against the supplier for whom the plan was prepared.

(c) Every supplier that becomes an agricultural water supplier after December 31, 2012, shall prepare and adopt an agricultural water management plan within one year after the date it has become an agricultural water supplier.

(d) A water supplier that indirectly provides water to customers for agricultural purposes shall not prepare a plan pursuant to this part without the consent of each agricultural water supplier that directly provides that water to its customers.

**SEC. 14.** Section 10825 of the Water Code is amended to read:

**10825.** (a) It is the intent of the Legislature in enacting this part to allow levels of water management planning commensurate with the numbers of customers served and the volume of water supplied.

(b) This part does not require the implementation of water use efficiency programs or practices that are not locally cost effective.

SEC. 15. Section 10826 of the Water Code is amended to read:

**10826.** An agricultural water management plan shall be adopted in accordance with this chapter. The plan shall do all of the following:

(a) Describe the agricultural water supplier and the service area, including all of the following:

- (1) Size of the service area.
- (2) Location of the service area and its water management facilities.
- (3) Terrain and soils.
- (4) Climate.
- (5) Operating rules and regulations.
- (6) Water delivery measurements or calculations.
- (7) Water rate schedules and billing.
- (8) Water shortage allocation policies.

(b) Describe the quantity and quality of water resources of the agricultural water supplier, including all of the following:

- (1) Surface water supply.
- (2) Groundwater supply.
- (3) Other water supplies, including recycled water.
- (4) Source water quality monitoring practices.
- (5) Water uses within the agricultural water supplier's service area, including all of the following:
- (A) Agricultural.
- (B) Environmental.
- (C) Recreational.
- (D) Municipal and industrial.

(E) Groundwater recharge, including estimated flows from deep percolation from irrigation and seepage.

(c) Include an annual water budget based on the quantification of all inflow and outflow components for the service area of the agricultural water supplier. Components of inflow shall include surface inflow, groundwater pumping in the service area, and effective precipitation. Components of outflow shall include surface outflow, deep percolation, and evapotranspiration. An agricultural water supplier shall report the annual water budget on a water-year basis. The department shall provide tools and resources to assist agricultural water suppliers in developing and quantifying components necessary to develop a water budget.

(d) Include an analysis, based on available information, of the effect of climate change on future water supplies.

(e) Describe previous water management activities.

(f) Identify water management objectives based on the water budget to improve water system efficiency or to meet other water management objectives. The agricultural water supplier shall identify, prioritize, and implement actions to reduce water loss, improve water system management, and meet other water management objectives identified in the plan.

(g) Include in the plan information regarding efficient water management practices required pursuant to Section 10608.48.

(h) Quantify the efficiency of agricultural water use within the service area of the agricultural water supplier using the appropriate method or methods from among the four water use efficiency quantification methods developed by the department in the May 8, 2012, report to the Legislature entitled "A Proposed Methodology for

Quantifying the Efficiency of Agricultural Water Use." The agricultural water supplier shall account for all water uses, including crop water use, agronomic water use, environmental water use, and recoverable surface flows.

**SEC. 16.** Section 10826.2 is added to the Water Code, to read:

**10826.2.** As part of its agricultural water management plan, each agricultural water supplier shall develop a drought plan for periods of limited water supply describing the actions of the agricultural water supplier for drought preparedness and management of water supplies and allocations during drought conditions. The drought plan shall contain both of the following:

(a) Resilience planning, including all of the following:

(1) Data, indicators, and information needed to determine the water supply availability and levels of drought severity.

(2) Analyses and identification of potential vulnerability to drought.

(3) A description of the opportunities and constraints for improving drought resilience planning, including all of the following:

(A) The availability of new technology or information.

(B) The ability of the agricultural water supplier to obtain or use additional water supplies during drought conditions.

(C) A description of other actions planned for implementation to improve drought resilience.

(b) Drought response planning, including all of the following:

(1) Policies and a process for declaring a water shortage and for implementing water shortage allocations and related response actions.

(2) Methods and procedures for the enforcement or appeal of, or exemption from, triggered shortage response actions.

(3) Methods and procedures for monitoring and evaluation of the effectiveness of the drought plan.

(4) Communication protocols and procedures to inform and coordinate customers, the public, interested parties, and local, regional, and state government.

(5) A description of the potential impacts on the revenues, financial condition, and planned expenditures of the agricultural water supplier during drought conditions that reduce water allocations, and proposed measures to overcome those impacts, including reserve-level policies.

**SEC. 17.** Section 10843 of the Water Code is amended to read:

**10843.** (a) An agricultural water supplier shall submit to the entities identified in subdivision (b) a copy of its plan no later than 30 days after review of the plan pursuant to subdivision (b) of Section 10820.

(b) An agricultural water supplier shall submit a copy of its plan to each of the following entities:

(1) The department.

(2) Any city, county, or city and county within which the agricultural water supplier provides water supplies.

(3) Any groundwater management entity within which jurisdiction the agricultural water supplier extracts or provides water supplies.

(4) The California State Library.

**SEC. 18.** Section 10845 of the Water Code is amended to read:

**10845.** (a) The department shall prepare and submit to the Legislature, on or before April 30, 2022, and thereafter in the years ending in seven and years ending in two, a report summarizing the status of the plans adopted pursuant to this part.

(b) The report prepared by the department shall identify the outstanding elements of any plan adopted pursuant to this part. The report shall include an evaluation of the effectiveness of this part in promoting efficient agricultural water management practices and recommendations relating to proposed changes to this part, as appropriate.

(c) The department shall provide a copy of the report to each agricultural water supplier that has submitted its plan to the department. The department shall also prepare reports and provide data for any legislative hearing designed to consider the effectiveness of plans submitted pursuant to this part.

(d) This section does not authorize the department, in preparing the report, to approve, disapprove, or critique individual plans submitted pursuant to this part.

**SEC. 19.** Section 10910 of the Water Code is amended to read:

**10910.** (a) Any city or county that determines that a project, as defined in Section 10912, is subject to the California Environmental Quality Act (Division 13 (commencing with Section 21000) of the Public Resources Code) under Section 21080 of the Public Resources Code shall comply with this part.

(b) The city or county, at the time that it determines whether an environmental impact report, a negative declaration, or a mitigated negative declaration is required for any project subject to the California Environmental Quality Act pursuant to Section 21080.1 of the Public Resources Code, shall identify any water system whose service area includes the project site and any water system adjacent to the project site that is, or may become as a result of supplying water to the project identified pursuant to this subdivision, a public water system, as defined in Section 10912, that may supply water for the project. If the city or county is not able to identify any public water system that may supply water for the project, the city or county shall prepare the water assessment required by this part after consulting with any entity serving domestic water supplies whose service area includes the project site, the local agency formation commission, and any public water system adjacent to the project site.

(c) (1) The city or county, at the time it makes the determination required under Section 21080.1 of the Public Resources Code, shall request each public water system identified pursuant to subdivision (b) to determine whether the projected water demand associated with a proposed project was included as part of the most recently adopted urban water management plan adopted pursuant to Part 2.6 (commencing with Section 10610).

(2) If the projected water demand associated with the proposed project was accounted for in the most recently adopted urban water management plan, the public water system may incorporate the requested information from the urban water management plan in preparing the elements of the assessment required to comply with subdivisions (d), (e), (f), and (g).

(3) If the projected water demand associated with the proposed project was not accounted for in the most recently adopted urban water management plan, or the public water system has no urban water management plan, the water supply assessment for the project shall include a discussion with regard to whether the public water system's total projected water supplies available during normal, single dry, and multiple dry water years during a 20-year projection will meet the projected water demand associated with the proposed project, in addition to the public water system's existing and planned future uses, including agricultural and manufacturing uses.

(4) If the city or county is required to comply with this part pursuant to subdivision (b), the water supply assessment for the project shall include a discussion with regard to whether the total projected water supplies, determined to be available by the city or county for the project during normal, single dry, and multiple dry water years during a 20-year projection, will meet the projected water demand associated with the proposed project, in addition to existing and planned future uses, including agricultural and manufacturing uses.

(d) (1) The assessment required by this section shall include an identification of any existing water supply entitlements, water rights, or water service contracts relevant to the identified water supply for the proposed project, and a description of the quantities of water received in prior years by the public water system, or the city or county if either is required to comply with this part pursuant to subdivision (b), under the existing water supply entitlements, water rights, or water service contracts.

(2) An identification of existing water supply entitlements, water rights, or water service contracts held by the public water system, or the city or county if either is required to comply with this part pursuant to subdivision (b), shall be demonstrated by providing information related to all of the following:

(A) Written contracts or other proof of entitlement to an identified water supply.

(B) Copies of a capital outlay program for financing the delivery of a water supply that has been adopted by the public water system.

(C) Federal, state, and local permits for construction of necessary infrastructure associated with delivering the water supply.

(D) Any necessary regulatory approvals that are required in order to be able to convey or deliver the water supply.

(e) If no water has been received in prior years by the public water system, or the city or county if either is required to comply with this part pursuant to subdivision (b), under the existing water supply entitlements, water rights, or water service contracts, the public water system, or the city or county if either is required to comply with this part pursuant to subdivision (b), shall also include in its water supply assessment pursuant to subdivision (c), an identification of the other public water systems or water service contractholders that receive a water supply or have existing water supply entitlements, water rights, or water service contracts, to the same source of water as the public water system, or the city or county if either is required to comply with this part pursuant to subdivision (b), has identified as a source of water supply within its water supply assessments.

(f) If a water supply for a proposed project includes groundwater, the following additional information shall be included in the water supply assessment:

(1) A review of any information contained in the urban water management plan relevant to the identified water supply for the proposed project.

(2) (A) A description of any groundwater basin or basins from which the proposed project will be supplied.

(B) For those basins for which a court or the board has adjudicated the rights to pump groundwater, a copy of the order or decree adopted by the court or the board and a description of the amount of groundwater the public water system, or the city or county if either is required to comply with this part pursuant to subdivision (b), has the legal right to pump under the order or decree.

(C) For a basin that has not been adjudicated that is a basin designated as high- or medium-priority pursuant to Section 10722.4, information regarding the following:

(i) Whether the department has identified the basin as being subject to critical conditions of overdraft pursuant to Section 12924.

(ii) If a groundwater sustainability agency has adopted a groundwater sustainability plan or has an approved alternative, a copy of that alternative or plan.

(D) For a basin that has not been adjudicated that is a basin designated as low- or very low priority pursuant to Section 10722.4, information as to whether the department has identified the basin or basins as overdrafted or has projected that the basin will become overdrafted if present management conditions continue, in the most current bulletin of the department that characterizes the condition of the groundwater basin, and a detailed description by the public water system, or the city or county if either is required to comply with this part pursuant to subdivision (b), of the efforts being undertaken in the basin or basins to eliminate the long-term overdraft condition.

(3) A detailed description and analysis of the amount and location of groundwater pumped by the public water system, or the city or county if either is required to comply with this part pursuant to subdivision (b), for the past five years from any groundwater basin from which the proposed project will be supplied. The description and analysis shall be based on information that is reasonably available, including, but not limited to, historic use records.

(4) A detailed description and analysis of the amount and location of groundwater that is projected to be pumped by the public water system, or the city or county if either is required to comply with this part pursuant to subdivision (b), from any basin from which the proposed project will be supplied. The description and analysis shall be based on information that is reasonably available, including, but not limited to, historic use records.

(5) An analysis of the sufficiency of the groundwater from the basin or basins from which the proposed project will be supplied to meet the projected water demand associated with the proposed project. A water supply assessment shall not be required to include the information required by this paragraph if the public water system

determines, as part of the review required by paragraph (1), that the sufficiency of groundwater necessary to meet the initial and projected water demand associated with the project was addressed in the description and analysis required by subparagraph (D) of paragraph (4) of subdivision (b) of Section 10631.

(g) (1) Subject to paragraph (2), the governing body of each public water system shall submit the assessment to the city or county not later than 90 days from the date on which the request was received. The governing body of each public water system, or the city or county if either is required to comply with this act pursuant to subdivision (b), shall approve the assessment prepared pursuant to this section at a regular or special meeting.

(2) Prior to the expiration of the 90-day period, if the public water system intends to request an extension of time to prepare and adopt the assessment, the public water system shall meet with the city or county to request an extension of time, which shall not exceed 30 days, to prepare and adopt the assessment.

(3) If the public water system fails to request an extension of time, or fails to submit the assessment notwithstanding the extension of time granted pursuant to paragraph (2), the city or county may seek a writ of mandamus to compel the governing body of the public water system to comply with the requirements of this part relating to the submission of the water supply assessment.

(h) Notwithstanding any other provision of this part, if a project has been the subject of a water supply assessment that complies with the requirements of this part, no additional water supply assessment shall be required for subsequent projects that were part of a larger project for which a water supply assessment was completed and that has complied with the requirements of this part and for which the public water system, or the city or county if either is required to comply with this part pursuant to subdivision (b), has concluded that its water supplies are sufficient to meet the projected water demand associated with the proposed project, in addition to the existing and planned future uses, including, but not limited to, agricultural and industrial uses, unless one or more of the following changes occurs:

(1) Changes in the project that result in a substantial increase in water demand for the project.

(2) Changes in the circumstances or conditions substantially affecting the ability of the public water system, or the city or county if either is required to comply with this part pursuant to subdivision (b), to provide a sufficient supply of water for the project.

(3) Significant new information becomes available that was not known and could not have been known at the time when the assessment was prepared.

(i) For the purposes of this section, hauled water is not considered as a source of water.

**SEC. 20.** This act shall become operative only if Senate Bill 606 of the 2017–18 Regular Session is enacted and becomes effective.



SB-606 Water management planning. (2017-2018)

As Amends the Law Today

As Amends the Law on Nov 08, 2018

**SECTION 1.** Section 350 of the Water Code is amended to read:

**350.** The governing body of a distributor of a public water supply, whether publicly or privately owned and including a mutual water company, shall declare a water shortage emergency condition to prevail within the area served by such distributor whenever it finds and determines that the ordinary demands and requirements of water consumers cannot be satisfied without depleting the water supply of the distributor to the extent that there would be insufficient water for human consumption, sanitation, and fire protection.

SEC. 2. Section 377 of the Water Code is amended to read:

**377.** (a) From and after the publication or posting of any ordinance or resolution pursuant to Section 376, a violation of a requirement of a water conservation program adopted pursuant to Section 376 is a misdemeanor. A person convicted under this subdivision shall be punished by imprisonment in the county jail for not more than 30 days, or by a fine not exceeding one thousand dollars (\$1,000), or by both.

(b) A court or public entity may hold a person civilly liable in an amount not to exceed ten thousand dollars (\$10,000) for a violation of any of the following:

(1) An ordinance or resolution adopted pursuant to Section 376.

(2) A regulation adopted by the board under Section 1058.5 or Chapter 9 (commencing with Section 10609) of Part 2.55 of Division 6, unless the board regulation provides that it cannot be enforced under this section or provides for a lesser applicable maximum penalty.

(c) Commencing on the 31st day after the public entity notified a person of a violation described in subdivision (b), the person additionally may be civilly liable in an amount not to exceed ten thousand dollars (\$10,000) plus five hundred dollars (\$500) for each additional day on which the violation continues.

(d) Remedies prescribed in this section are cumulative and not alternative, except that no liability shall be recoverable under this section for any violation of paragraph (2) of subdivision (b) if the board has filed a complaint pursuant to Section 1846 alleging the same violation.

(e) A public entity may administratively impose the civil liability described in subdivisions (b) and (c) after providing notice and an opportunity for a hearing. The public entity shall initiate a proceeding under this subdivision by a complaint issued pursuant to Section 377.5. The public entity shall issue the complaint at least 30 days before the hearing on the complaint and the complaint shall state the basis for the proposed civil liability order.

(f) (1) In determining the amount of civil liability to assess, a court or public entity shall take into consideration all relevant circumstances, including, but not limited to, the nature and persistence of the violation, the extent of the harm caused by the violation, the length of time over which the violation occurs, and any corrective action taken by the violator.

(2) The civil liability calculated pursuant to paragraph (1) for the first violation of subdivision (b) by a residential water user shall not exceed one thousand dollars (\$1,000) except in extraordinary situations where the court or public entity finds all of the following:

(A) The residential user had actual notice of the requirement found to be violated.

(B) The conduct was intentional.

(C) The amount of water involved was substantial.

(g) Civil liability imposed pursuant to this section shall be paid to the public entity and expended solely for the purposes of this chapter.

(h) An order setting administrative civil liability shall become effective and final upon issuance of the order and payment shall be made. Judicial review of any final order shall be pursuant to Section 1094.5 of the Code of Civil Procedure.

(i) In addition to the remedies prescribed in this section, a public entity may enforce water use limitations established by an ordinance or resolution adopted pursuant to this chapter, or as otherwise authorized by law, by a volumetric penalty in an amount established by the public entity.

**SEC. 3.** Section 1058.5 of the Water Code is amended to read:

**1058.5.** (a) This section applies to any emergency regulation adopted by the board for which the board makes both of the following findings:

(1) The emergency regulation is adopted to prevent the waste, unreasonable use, unreasonable method of use, or unreasonable method of diversion, of water, to promote water recycling or water conservation, to require curtailment of diversions when water is not available under the diverter's priority of right, or in furtherance of any of the foregoing, to require reporting of diversion or use or the preparation of monitoring reports.

(2) The emergency regulation is adopted in response to conditions which exist, or are threatened, in a critically dry year immediately preceded by two or more consecutive below normal, dry, or critically dry years or during a period for which the Governor has issued a proclamation of a state of emergency under the California Emergency Services Act (Chapter 7 (commencing with Section 8550) of Division 1 of Title 2 of the Government Code) based on drought conditions.

(b) Notwithstanding Sections 11346.1 and 11349.6 of the Government Code, any findings of emergency adopted by the board, in connection with the adoption of an emergency regulation under this section, are not subject to review by the Office of Administrative Law.

(c) An emergency regulation adopted by the board under this section may remain in effect for up to one year, as determined by the board, and is deemed repealed immediately upon a finding by the board that due to changed conditions it is no longer necessary for the regulation to remain in effect. An emergency regulation adopted by the board under this section may be renewed if the board determines that the conditions specified in paragraph (2) of subdivision (a) are still in effect.

(d) In addition to any other applicable civil or criminal penalties, any person or entity that who violates a regulation adopted by the board pursuant to this section is guilty of an infraction punishable by a fine of up to five hundred dollars (\$500) for each day in which the violation occurs.

(e) (1) Notwithstanding subdivision (b) of Section 1551 or subdivision (e) of Section 1848, a civil liability imposed under Chapter 12 (commencing with Section 1825) of Part 2 of Division 2 by the board or a court for a violation of an emergency conservation regulation adopted pursuant to this section shall be deposited, and separately accounted for, in the Water Rights Fund. Funds deposited in accordance with this subdivision shall be available, upon appropriation, for water conservation activities and programs.

(2) For purposes of this subdivision, an "emergency conservation regulation" means an emergency regulation that requires an end user of water, a water retailer, or a water wholesaler to conserve water or report to the board on water conservation. Water conservation includes restrictions or limitations on particular uses of water or a reduction in the amount of water used or served, but does not include curtailment of diversions when water is not available under the diverter's priority of right or reporting requirements related to curtailments.

**SEC. 4.** Section 1120 of the Water Code is amended to read:

**1120.** This chapter applies to any decision or order issued under this part or Section 275, Part 2 (commencing with Section 1200), Part 2 (commencing with Section 10500) of Division 6, Part 2.55 (commencing with Section 10608) of Division 6, or Chapter 11 (commencing with Section 10735) of Part 2.74 of Division 6, Article 7 (commencing with Section 13550) of Chapter 7 of Division 7, or the public trust doctrine.

**SEC. 5.** Section 10608.12 of the Water Code is amended to read:

**10608.12.** Unless the context otherwise requires, the following definitions govern the construction of this part:

(a) "Agricultural water supplier" means a water supplier, either publicly or privately owned, providing water to 10,000 or more irrigated acres, excluding recycled water. "Agricultural water supplier" includes a supplier or contractor for water, regardless of the basis of right, that distributes or sells water for ultimate resale to customers. "Agricultural water supplier" does not include the department.

(b) "Base daily per capita water use" means any of the following:

(1) The urban retail water supplier's estimate of its average gross water use, reported in gallons per capita per day and calculated over a continuous 10-year period ending no earlier than December 31, 2004, and no later than December 31, 2010.

(2) For an urban retail water supplier that meets at least 10 percent of its 2008 measured retail water demand through recycled water that is delivered within the service area of an urban retail water supplier or its urban wholesale water supplier, the urban retail water supplier may extend the calculation described in paragraph (1) up to an additional five years to a maximum of a continuous 15-year period ending no earlier than December 31, 2004, and no later than December 31, 2010.

(3) For the purposes of Section 10608.22, the urban retail water supplier's estimate of its average gross water use, reported in gallons per capita per day and calculated over a continuous five-year period ending no earlier than December 31, 2007, and no later than December 31, 2010.

(c) "Baseline commercial, industrial, and institutional water use" means an urban retail water supplier's base daily per capita water use for commercial, industrial, and institutional users.

(d) "CII water use" means water used by commercial water users, industrial water users, institutional water users, and large landscape water users.

(e) "Commercial water user" means a water user that provides or distributes a product or service.

(f) "Compliance daily per capita water use" means the gross water use during the final year of the reporting period, reported in gallons per capita per day.

(g) "Disadvantaged community" means a community with an annual median household income that is less than 80 percent of the statewide annual median household income.

(h) "Gross water use" means the total volume of water, whether treated or untreated, entering the distribution system of an urban retail water supplier, excluding all of the following:

(1) Recycled water that is delivered within the service area of an urban retail water supplier or its urban wholesale water supplier.

(2) The net volume of water that the urban retail water supplier places into long-term storage.

(3) The volume of water the urban retail water supplier conveys for use by another urban water supplier.

(4) The volume of water delivered for agricultural use, except as otherwise provided in subdivision (f) of Section 10608.24.

(i) "Industrial water user" means a water user that is primarily a manufacturer or processor of materials as defined by the North American Industry Classification System code sectors 31 to 33, inclusive, or an entity that is a water user primarily engaged in research and development.

(j) "Institutional water user" means a water user dedicated to public service. This type of user includes, among other users, higher education institutions, schools, courts, churches, hospitals, government facilities, and nonprofit research institutions.

(k) "Interim urban water use target" means the midpoint between the urban retail water supplier's base daily per capita water use and the urban retail water supplier's urban water use target for 2020.

(I) "Large landscape" means a nonresidential landscape as described in the performance measures for CII water use adopted pursuant to Section 10609.10.

(m) "Locally cost effective" means that the present value of the local benefits of implementing an agricultural efficiency water management practice is greater than or equal to the present value of the local cost of implementing that measure.

(n) "Performance measures" means actions to be taken by urban retail water suppliers that will result in increased water use efficiency by CII water users. Performance measures may include, but are not limited to, educating CII water users on best management practices, conducting water use audits, and preparing water management plans. Performance measures do not include process water.

(o) "Potable reuse" means direct potable reuse, indirect potable reuse for groundwater recharge, and reservoir water augmentation as those terms are defined in Section 13561.

(p) "Process water" means water used by industrial water users for producing a product or product content or water used for research and development. Process water includes, but is not limited to, continuous manufacturing processes, and water used for testing, cleaning, and maintaining equipment. Water used to cool machinery or buildings used in the manufacturing process or necessary to maintain product quality or chemical characteristics for product manufacturing or control rooms, data centers, laboratories, clean rooms, and other industrial facility units that are integral to the manufacturing or research and development process is process water. Water used in the manufacturing process that is necessary for complying with local, state, and federal health and safety laws, and is not incidental water, is process water. Process water does not mean incidental water uses.

(q) "Recycled water" means recycled water, as defined in subdivision (n) of Section 13050.

(r) "Regional water resources management" means sources of supply resulting from watershed-based planning for sustainable local water reliability or any of the following alternative sources of water:

(1) The capture and reuse of stormwater or rainwater.

(2) The use of recycled water.

(3) The desalination of brackish groundwater.

(4) The conjunctive use of surface water and groundwater in a manner that is consistent with the safe yield of the groundwater basin.

(s) "Reporting period" means the years for which an urban retail water supplier reports compliance with the urban water use targets.

(t) "Urban retail water supplier" means a water supplier, either publicly or privately owned, that directly provides potable municipal water to more than 3,000 end users or that supplies more than 3,000 acre-feet of potable water annually at retail for municipal purposes.

(u) "Urban water use objective" means an estimate of aggregate efficient water use for the previous year based on adopted water use efficiency standards and local service area characteristics for that year, as described in Section 10609.20.

(v) "Urban water use target" means the urban retail water supplier's targeted future daily per capita water use.

(w) "Urban wholesale water supplier" supplier," means a water supplier, either publicly or privately owned, that provides more than 3,000 acre-feet of water annually at wholesale for potable municipal purposes.

SEC. 6. Section 10608.20 of the Water Code is amended to read:

**10608.20.** (a) (1) Each urban retail water supplier shall develop urban water use targets and an interim urban water use target by July 1, 2011. Urban retail water suppliers may elect to determine and report progress toward achieving these targets on an individual or regional basis, as provided in subdivision (a) of Section 10608.28, and may determine the targets on a fiscal year or calendar year basis.

(2) It is the intent of the Legislature that the urban water use targets described in paragraph (1) cumulatively result in a 20-percent reduction from the baseline daily per capita water use by December 31, 2020.

(b) An urban retail water supplier shall adopt one of the following methods for determining its urban water use target pursuant to subdivision (a):

(1) Eighty percent of the urban retail water supplier's baseline per capita daily water use.

(2) The per capita daily water use that is estimated using the sum of the following performance standards:

(A) For indoor residential water use, 55 gallons per capita daily water use as a provisional standard. Upon completion of the department's  $\frac{2017}{2016}$  report to the Legislature pursuant to Section 10608.42, this standard may be adjusted by the Legislature by statute.

(B) For landscape irrigated through dedicated or residential meters or connections, water efficiency equivalent to the standards of the Model Water Efficient Landscape Ordinance set forth in Chapter 2.7 (commencing with Section 490) of Division 2 of Title 23 of the California Code of Regulations, as in effect the later of the year of the landscape's installation or 1992. An urban retail water supplier using the approach specified in this subparagraph shall use satellite imagery, site visits, or other best available technology to develop an accurate estimate of landscaped areas.

(C) For commercial, industrial, and institutional uses, a 10-percent reduction in water use from the baseline commercial, industrial, and institutional water use by 2020.

(3) Ninety-five percent of the applicable state hydrologic region target, as set forth in the state's draft 20x2020 Water Conservation Plan (dated April 30, 2009). If the service area of an urban water supplier includes more than one hydrologic region, the supplier shall apportion its service area to each region based on population or area.

(4) A method that shall be identified and developed by the department, through a public process, and reported to the Legislature no later than December 31, 2010. The method developed by the department shall identify per capita targets that cumulatively result in a statewide 20-percent reduction in urban daily per capita water use by December 31, 2020. In developing urban daily per capita water use targets, the department shall do all of the following:

(A) Consider climatic differences within the state.

(B) Consider population density differences within the state.

(C) Provide flexibility to communities and regions in meeting the targets.

(D) Consider different levels of per capita water use according to plant water needs in different regions.

(E) Consider different levels of commercial, industrial, and institutional water use in different regions of the state.

(F) Avoid placing an undue hardship on communities that have implemented conservation measures or taken actions to keep per capita water use low.

(c) If the department adopts a regulation pursuant to paragraph (4) of subdivision (b) that results in a requirement that an urban retail water supplier achieve a reduction in daily per capita water use that is greater than 20 percent by December 31, 2020, an urban retail water supplier that adopted the method described in paragraph (4) of subdivision (b) may limit its urban water use target to a reduction of not more than 20 percent by December 31, 2020, by adopting the method described in paragraph (1) of subdivision (b).

(d) The department shall update the method described in paragraph (4) of subdivision (b) and report to the Legislature by December 31, 2014. An urban retail water supplier that adopted the method described in paragraph (4) of subdivision (b) may adopt a new urban daily per capita water use target pursuant to this updated method.

(e) An urban retail water supplier shall include in its urban water management plan due in 2010 pursuant to Part 2.6 (commencing with Section 10610) the baseline daily per capita water use, urban water use target, interim urban water use target, and compliance daily per capita water use, along with the bases for determining those estimates, including references to supporting data.

(f) When calculating per capita values for the purposes of this chapter, an urban retail water supplier shall determine population using federal, state, and local population reports and projections.

(g) An urban retail water supplier may update its 2020 urban water use target in its 2015 urban water management plan required pursuant to Part 2.6 (commencing with Section 10610).

(h) (1) The department, through a public process and in consultation with the California Urban Water Conservation Council, shall develop technical methodologies and criteria for the consistent implementation of this part, including, but not limited to, both of the following:

(A) Methodologies for calculating base daily per capita water use, baseline commercial, industrial, and institutional water use, compliance daily per capita water use, gross water use, service area population, indoor residential water use, and landscaped area water use.

(B) Criteria for adjustments pursuant to subdivisions (d) and (e) of Section 10608.24.

(2) The department shall post the methodologies and criteria developed pursuant to this subdivision on its internet website, Internet Web site, and make written copies available, by October 1, 2010. An urban retail water supplier shall use the methods developed by the department in compliance with this part.

(i) (1) The department shall adopt regulations for implementation of the provisions relating to process water in accordance with Section 10608.12, subdivision (e) of Section 10608.24, and subdivision (d) of Section 10608.26.

(2) The initial adoption of a regulation authorized by this subdivision is deemed to address an emergency, for purposes of Sections 11346.1 and 11349.6 of the Government Code, and the department is hereby exempted for that purpose from the requirements of subdivision (b) of Section 11346.1 of the Government Code. After the initial adoption of an emergency regulation pursuant to this subdivision, the department shall not request approval from the Office of Administrative Law to readopt the regulation as an emergency regulation pursuant to Section 11346.1 of the Government Code.

(j) (1) An urban retail water supplier is granted an extension to July 1, 2011, for adoption of an urban water management plan pursuant to Part 2.6 (commencing with Section 10610) due in 2010 to allow the use of technical methodologies developed by the department pursuant to paragraph (4) of subdivision (b) and subdivision (h). An urban retail water supplier that adopts an urban water management plan due in 2010 that does not use the methodologies developed by the department pursuant to subdivision (h) shall amend the plan by July 1, 2011, to comply with this part.

(2) An urban wholesale water supplier whose urban water management plan prepared pursuant to Part 2.6 (commencing with Section 10610) was due and not submitted in 2010 is granted an extension to July 1, 2011, to permit coordination between an urban wholesale water supplier and urban retail water suppliers.

**SEC. 7.** Section 10608.35 is added to the Water Code, to read:

**10608.35.** (a) The department, in coordination with the board, shall conduct necessary studies and investigations and make a recommendation to the Legislature, by January 1, 2020, on the feasibility of developing and enacting water loss reporting requirements for urban wholesale water suppliers.

(b) The studies and investigations shall include an evaluation of the suitability of applying the processes and requirements of Section 10608.34 to urban wholesale water suppliers.

(c) In conducting necessary studies and investigations and developing its recommendation, the department shall solicit broad public participation from stakeholders and other interested persons.

**SEC. 8.** Section 10609.20 is added to the Water Code, immediately following Section 10609.18, to read:

**10609.20.** (a) Each urban retail water supplier shall calculate its urban water use objective no later than November 1, 2023, and by November 1 every year thereafter.

(b) The calculation shall be based on the urban retail water supplier's water use conditions for the previous calendar or fiscal year.

(c) Each urban water supplier's urban water use objective shall be composed of the sum of the following:

(1) Aggregate estimated efficient indoor residential water use.

(2) Aggregate estimated efficient outdoor residential water use.

(3) Aggregate estimated efficient outdoor irrigation of landscape areas with dedicated irrigation meters or equivalent technology in connection with CII water use.

(4) Aggregate estimated efficient water losses.

(5) Aggregate estimated water use in accordance with variances, as appropriate.

(d) (1) An urban retail water supplier that delivers water from a groundwater basin, reservoir, or other source that is augmented by potable reuse water may adjust its urban water use objective by a bonus incentive calculated pursuant to this subdivision.

(2) The water use objective bonus incentive shall be the volume of its potable reuse delivered to residential water users and to landscape areas with dedicated irrigation meters in connection with CII water use, on an acre-foot basis.

(3) The bonus incentive pursuant to paragraph (1) shall be limited in accordance with one of the following:

(A) The bonus incentive shall not exceed 15 percent of the urban water supplier's water use objective for any potable reuse water produced at an existing facility.

(B) The bonus incentive shall not exceed 10 percent of the urban water supplier's water use objective for any potable reuse water produced at any facility that is not an existing facility.

(4) For purposes of this subdivision, "existing facility" means a facility that meets all of the following:

(A) The facility has a certified environmental impact report, mitigated negative declaration, or negative declaration on or before January 1, 2019.

(B) The facility begins producing and delivering potable reuse water on or before January 1, 2022.

(C) The facility uses microfiltration and reverse osmosis technologies to produce the potable reuse water.

(e) (1) The calculation of the urban water use objective shall be made using landscape area and other data provided by the department and pursuant to the standards, guidelines, and methodologies adopted by the board. The department shall provide data to the urban water supplier at a level of detail sufficient to allow the urban water supplier to verify its accuracy at the parcel level.

(2) Notwithstanding paragraph (1), an urban retail water supplier may use alternative data in calculating the urban water use objective if the supplier demonstrates to the department that the alternative data are equivalent, or superior, in quality and accuracy to the data provided by the department. The department may provide technical assistance to an urban retail water supplier in evaluating whether the alternative data are appropriate for use in calculating the supplier's urban water use objective.

SEC. 9. Section 10609.22 is added to the Water Code, to read:

**10609.22.** (a) An urban retail water supplier shall calculate its actual urban water use no later than November 1, 2023, and by November 1 every year thereafter.

(b) The calculation shall be based on the urban retail water supplier's water use for the previous calendar or fiscal year.

(c) Each urban water supplier's urban water use shall be composed of the sum of the following:

(1) Aggregate residential water use.

(2) Aggregate outdoor irrigation of landscape areas with dedicated irrigation meters in connection with CII water use.

(3) Aggregate water losses.

SEC. 10. Section 10609.24 is added to the Water Code, to read:

**10609.24.** (a) An urban retail water supplier shall submit a report to the department no later than November 1, 2023, and by November 1 every year thereafter. The report shall include all of the following:

(1) The urban water use objective calculated pursuant to Section 10609.20 along with relevant supporting data.

(2) The actual urban water use calculated pursuant to Section 10609.22 along with relevant supporting data.

(3) Documentation of the implementation of the performance measures for CII water use.

(4) A description of the progress made towards meeting the urban water use objective.

(b) The department shall post the reports and information on its Internet Web site.

(c) The board may issue an information order or conservation order to, or impose civil liability on, an entity or individual for failure to submit a report required by this section.

SEC. 11. Section 10609.26 is added to the Water Code, to read:

**10609.26.** (a) (1) On and after November 1, 2023, the board may issue informational orders pertaining to water production, water use, and water conservation to an urban retail water supplier that does not meet its urban water use objective required by this chapter. Informational orders are intended to obtain information on supplier activities, water production, and conservation efforts in order to identify technical assistance needs and assist urban water suppliers in meeting their urban water use objectives.

(2) In determining whether to issue an informational order, the board shall consider the degree to which the urban retail water supplier is not meeting its urban water use objective, information provided in the report required by Section 10609.24, and actions the urban retail water supplier has implemented or will implement in order to help meet the urban water use objective.

(3) The board shall share information received pursuant to this subdivision with the department.

(4) An urban water supplier may request technical assistance from the department. The technical assistance may, to the extent available, include guidance documents, tools, and data.

(b) On and after November 1, 2024, the board may issue a written notice to an urban retail water supplier that does not meet its urban water use objective required by this chapter. The written notice may warn the urban retail water supplier that it is not meeting its urban water use objective described in Section 10609.20 and is not making adequate progress in meeting the urban water use objective, and may request that the urban retail water supplier address areas of concern in its next annual report required by Section 10609.24. In deciding whether to issue a written notice, the board may consider whether the urban retail water supplier has received an informational order, the degree to which the urban retail water supplier is not meeting its urban water use objective, information provided in the report required by Section 10609.24, and actions the urban retail water supplier has implemented or will implement in order to help meet its urban water use objective.

(c) (1) On and after November 1, 2025, the board may issue a conservation order to an urban retail water supplier that does not meet its urban water use objective. A conservation order may consist of, but is not limited to, referral to the department for technical assistance, requirements for education and outreach, requirements for local enforcement, and other efforts to assist urban retail water suppliers in meeting their urban water use objective.

(2) In issuing a conservation order, the board shall identify specific deficiencies in an urban retail water supplier's progress towards meeting its urban water use objective, and identify specific actions to address the deficiencies.

(3) The board may request that the department provide an urban retail water supplier with technical assistance to support the urban retail water supplier's actions to remedy the deficiencies.

(d) A conservation order issued in accordance with this chapter may include requiring actions intended to increase water-use efficiency, but shall not curtail or otherwise limit the exercise of a water right, nor shall it require the imposition of civil liability pursuant to Section 377.

SEC. 12. Section 10609.28 is added to the Water Code, to read:

**10609.28.** The board may issue a regulation or informational order requiring a wholesale water supplier, an urban retail water supplier, or a distributor of a public water supply, as that term is used in Section 350, to provide a monthly report relating to water production, water use, or water conservation.

**SEC. 13.** Section 10609.30 is added to the Water Code, to read:

**10609.30.** On or before January 10, 2024, the Legislative Analyst shall provide to the appropriate policy committees of both houses of the Legislature and the public a report evaluating the implementation of the water use efficiency standards and water use reporting pursuant to this chapter. The board and the department shall provide the Legislative Analyst with the available data to complete this report.

(a) The report shall describe all of the following:

(1) The rate at which urban retail water users are complying with the standards, and factors that might facilitate or impede their compliance.

(2) The accuracy of the data and estimates being used to calculate urban water use objectives.

(3) Indications of the economic impacts, if any, of the implementation of this chapter on urban water suppliers and urban water users, including CII water users.

(4) The frequency of use of the bonus incentive, the volume of water associated with the bonus incentive, value to urban water suppliers of the bonus incentive, and any implications of the use of the bonus incentive on water use efficiency.

(5) The early indications of how implementing this chapter might impact the efficiency of statewide urban water use.

(6) Recommendations, if any, for improving statewide urban water use efficiency and the standards and practices described in this chapter.

(7) Any other issues the Legislative Analyst deems appropriate.

**SEC. 14.** Section 10609.32 is added to the Water Code, to read:

**10609.32.** It is the intent of the Legislature that the chairperson of the board and the director of the department appear before the appropriate policy committees of both houses of the Legislature on or around January 1, 2026, and report on the implementation of the water use efficiency standards and water use reporting pursuant to this chapter. It is the intent of the Legislature that the topics to be covered include all of the following:

(a) The rate at which urban retail water suppliers are complying with the standards, and factors that might facilitate or impede their compliance.

(b) What enforcement actions have been taken, if any.

(c) The accuracy of the data and estimates being used to calculate urban water use objectives.

(*d*) Indications of the economic impacts, if any, of the implementation of this chapter on urban water suppliers and urban water users, including CII water users.

(e) The frequency of use of the bonus incentive, the volume of water associated with the bonus incentive, value to urban water suppliers of the bonus incentive, and any implications of the use of the bonus incentive on water use efficiency.

(f) An assessment of how implementing this chapter is affecting the efficiency of statewide urban water use.

SEC. 15. Section 10609.34 is added to the Water Code, to read:

**10609.34.** Notwithstanding Section 15300.2 of Title 14 of the California Code of Regulations, an action of the board taken under this chapter shall be deemed to be a Class 8 action, within the meaning of Section 15308 of Title 14 of the California Code of Regulations, provided that the action does not involve relaxation of existing water conservation or water use standards.

SEC. 16. Section 10609.36 is added to the Water Code, to read:

**10609.36.** (a) Nothing in this chapter shall be construed to determine or alter water rights. Sections 1010 and 1011 apply to water conserved through implementation of this chapter.

(b) Nothing in this chapter shall be construed to authorize the board to update or revise water use efficiency standards authorized by this chapter except as explicitly provided in this chapter. Authorization to update the standards beyond that explicitly provided in this chapter shall require separate legislation.

(c) Nothing in this chapter shall be construed to limit or otherwise affect the use of recycled water as seawater barriers for groundwater salinity management.

SEC. 17. Section 10609.38 is added to the Water Code, to read:

**10609.38.** The board may waive the requirements of this chapter for a period of up to five years for any urban retail water supplier whose water deliveries are significantly affected by changes in water use as a result of damage from a disaster such as an earthquake or fire. In establishing the period of a waiver, the board shall take into consideration the breadth of the damage and the time necessary for the damaged areas to recover from the disaster.

SEC. 18. Section 10610.2 of the Water Code is amended to read:

**10610.2.** (a) The Legislature finds and declares all of the following:

(1) The waters of the state are a limited and renewable resource subject to ever-increasing demands.

(2) The conservation and efficient use of urban water supplies are of statewide concern; however, the planning for that use and the implementation of those plans can best be accomplished at the local level.

(3) A long-term, reliable supply of water is essential to protect the productivity of California's businesses and economic climate, and increasing long-term water conservation among Californians, improving water use efficiency within the state's communities and agricultural production, and strengthening local and regional drought planning are critical to California's resilience to drought and climate change.

(4) As part of its long-range planning activities, every urban water supplier should make every effort to ensure the appropriate level of reliability in its water service sufficient to meet the needs of its various categories of customers during normal, dry, and multiple dry water years now and into the foreseeable future, and every urban water supplier should collaborate closely with local land-use authorities to ensure water demand forecasts are consistent with current land-use planning.

(5) Public health issues have been raised over a number of contaminants that have been identified in certain local and imported water supplies.

(6) Implementing effective water management strategies, including groundwater storage projects and recycled water projects, may require specific water quality and salinity targets for meeting groundwater basins water quality objectives and promoting beneficial use of recycled water.

(7) Water quality regulations are becoming an increasingly important factor in water agencies' selection of raw water sources, treatment alternatives, and modifications to existing treatment facilities.

(8) Changes in drinking water quality standards may also impact the usefulness of water supplies and may ultimately impact supply reliability.

(9) The quality of source supplies can have a significant impact on water management strategies and supply reliability.

(b) This part is intended to provide assistance to water agencies in carrying out their long-term resource planning responsibilities to ensure adequate water supplies to meet existing and future demands for water.

**SEC. 19.** Section 10610.4 of the Water Code is amended to read:

**10610.4.** The Legislature finds and declares that it is the policy of the state as follows:

(a) The management of urban water demands and efficient use of water shall be actively pursued to protect both the people of the state and their water resources.

(b) The management of urban water demands and efficient use of urban water supplies shall be a guiding criterion in public decisions.

(c) Urban water suppliers shall be required to develop water management plans to achieve the efficient use of available supplies and strengthen local drought planning.

SEC. 20. Section 10612 of the Water Code is amended and renumbered to read:

**10612. 10611.3.** "Drought risk assessment" "Customer" means a method that examines water shortage risks based on the driest five-year historic sequence for the agency's water supply, as described in subdivision (b) of Section 10635. purchaser of water from a water supplier who uses the water for municipal purposes, including residential, commercial, governmental, and industrial uses.

SEC. 21. Section 10612 is added to the Water Code, to read:

**10612.** "Drought risk assessment" means a method that examines water shortage risks based on the driest fiveyear historic sequence for the agency's water supply, as described in subdivision (b) of Section 10635.

SEC. 22. Section 10617.5 is added to the Water Code, to read:

**10617.5.** "Water shortage contingency plan" means a document that incorporates the provisions detailed in subdivision (a) of Section 10632 and is subsequently adopted by an urban water supplier pursuant to this article.

**SEC. 23.** Section 10618 is added to the Water Code, to read:

**10618.** "Water supply and demand assessment" means a method that looks at current year and one or more dry year supplies and demands for determining water shortage risks, as described in Section 10632.1.

SEC. 24. Section 10620 of the Water Code is amended to read:

**10620.** (a) Every urban water supplier shall prepare and adopt an urban water management plan in the manner set forth in Article 3 (commencing with Section 10640).

(b) Every person that becomes an urban water supplier shall adopt an urban water management plan within one year after it has become an urban water supplier.

(c) An urban water supplier indirectly providing water shall not include planning elements in its water management plan as provided in Article 2 (commencing with Section 10630) that would be applicable to urban water suppliers or public agencies directly providing water, or to their customers, without the consent of those suppliers or public agencies.

(d) (1) An urban water supplier may satisfy the requirements of this part by participation in areawide, regional, watershed, or basinwide urban water management planning where those plans will reduce preparation costs and contribute to the achievement of conservation, efficient water use, and improved local drought resilience.

(2) Notwithstanding paragraph (1), each urban water supplier shall develop its own water shortage contingency plan, but an urban water supplier may incorporate, collaborate, and otherwise share information with other urban water suppliers or other governing entities participating in an areawide, regional, watershed, or basinwide urban water management plan, an agricultural management plan, or groundwater sustainability plan development.

(3) Each urban water supplier shall coordinate the preparation of its plan with other appropriate agencies in the area, including other water suppliers that share a common source, water management agencies, and relevant public agencies, to the extent practicable.

(e) The urban water supplier may prepare the plan with its own staff, by contract, or in cooperation with other governmental agencies.

(f) An urban water supplier shall describe in the plan water management tools and options used by that entity that will maximize resources and minimize the need to import water from other regions.

SEC. 25. Section 10621 of the Water Code is amended to read:

**10621.** (a) Each urban water supplier shall update its plan at least once every five years on or before July 1, in years ending in six and one, incorporating updated and new information from the five years preceding each update.

(b) Every urban water supplier required to prepare a plan pursuant to this part shall, at least 60 days before the public hearing on the plan required by Section 10642, notify any city or county within which the supplier provides water supplies that the urban water supplier will be reviewing the plan and considering amendments or changes to the plan. The urban water supplier may consult with, and obtain comments from, any city or county that receives notice pursuant to this subdivision.

(c) An urban water supplier regulated by the Public Utilities Commission shall include its most recent plan and water shortage contingency plan as part of the supplier's general rate case filings.

(d) The amendments to, or changes in, the plan shall be adopted and filed in the manner set forth in Article 3 (commencing with Section 10640).

(e) Each urban water supplier shall update and submit its 2015 plan to the department by July 1, 2016.

(f) (1) Each urban water supplier shall update and submit its 2020 plan to the department by July 1, 2021.

(2) By January 1, 2024, each urban retail water supplier shall adopt and submit to the department a supplement to the adopted 2020 plan that includes information required pursuant to subparagraph (B) of paragraph (1) of subdivision (e) of Section 10631. This supplement is not an update or an amendment to the plan and, therefore, an urban water supplier is not required to comply with the public notice, hearing, and adoption requirements of Section 10642 before submitting the information to the department.

**SEC. 26.** Section 10630 of the Water Code is amended to read:

**10630.** It is the intention of the Legislature, in enacting this part, to permit levels of water management planning commensurate with the numbers of customers served and the volume of water supplied, while accounting for impacts from climate change.

SEC. 27. Section 10630.5 is added to the Water Code, to read:

**10630.5.** Each plan shall include a simple lay description of how much water the agency has on a reliable basis, how much it needs for the foreseeable future, what the agency's strategy is for meeting its water needs, the challenges facing the agency, and any other information necessary to provide a general understanding of the agency's plan.

SEC. 28. Section 10631 of the Water Code is amended to read:

**10631.** A plan shall be adopted in accordance with this chapter that shall do all of the following:

(a) Describe the service area of the supplier, including current and projected population, climate, and other social, economic, and demographic factors affecting the supplier's water management planning. The projected population estimates shall be based upon data from the state, regional, or local service agency population projections within the service area of the urban water supplier and shall be in five-year increments to 20 years or as far as data is available. The description shall include the current and projected land uses within the existing or anticipated service area affecting the supplier's water management planning. Urban water suppliers shall coordinate with local or regional land use authorities to determine the most appropriate land use information, including, where appropriate, land use information obtained from local or regional land use authorities, as developed pursuant to Article 5 (commencing with Section 65300) of Chapter 3 of Division 1 of Title 7 of the Government Code.

(b) Identify and quantify, to the extent practicable, the existing and planned sources of water available to the supplier over the same five-year increments described in subdivision (a), providing supporting and related information, including all of the following:

(1) A detailed discussion of anticipated supply availability under a normal water year, single dry year, and droughts lasting at least five years, as well as more frequent and severe periods of drought, as described in the drought risk assessment. For each source of water supply, consider any information pertinent to the reliability analysis conducted pursuant to Section 10635, including changes in supply due to climate change.

(2) When multiple sources of water supply are identified, a description of the management of each supply in correlation with the other identified supplies.

(3) For any planned sources of water supply, a description of the measures that are being undertaken to acquire and develop those water supplies.

(4) If groundwater is identified as an existing or planned source of water available to the supplier, all of the following information:

(A) The current version of any groundwater sustainability plan or alternative adopted pursuant to Part 2.74 (commencing with Section 10720), any groundwater management plan adopted by the urban water supplier, including plans adopted pursuant to Part 2.75 (commencing with Section 10750), or any other specific authorization for groundwater management for basins underlying the urban water supplier's service area.

(B) A description of any groundwater basin or basins from which the urban water supplier pumps groundwater. For basins that a court or the board has adjudicated the rights to pump groundwater, a copy of the order or decree adopted by the court or the board and a description of the amount of groundwater the urban water supplier has the legal right to pump under the order or decree. For a basin that has not been adjudicated, information as to whether the department has identified the basin as a high- or medium-priority basin in the most current official departmental bulletin that characterizes the condition of the groundwater basin, and a detailed description of the efforts being undertaken by the urban water supplier to coordinate with groundwater sustainability agencies or groundwater management agencies listed in subdivision (c) of Section 10723 to maintain or achieve sustainable groundwater conditions in accordance with a groundwater sustainability plan or alternative adopted pursuant to Part 2.74 (commencing with Section 10720).

(C) A detailed description and analysis of the location, amount, and sufficiency of groundwater pumped by the urban water supplier for the past five years. The description and analysis shall be based on information that is reasonably available, including, but not limited to, historic use records.

(D) A detailed description and analysis of the amount and location of groundwater that is projected to be pumped by the urban water supplier. The description and analysis shall be based on information that is reasonably available, including, but not limited to, historic use records.

(c) Describe the opportunities for exchanges or transfers of water on a short-term or long-term basis.

(d) (1) For an urban retail water supplier, quantify, to the extent records are available, past and current water use, over the same five-year increments described in subdivision (a), and projected water use, based upon information developed pursuant to subdivision (a), identifying the uses among water use sectors, including, but not necessarily limited to, all of the following:

(A) Single-family residential.

- (B) Multifamily.
- (C) Commercial.
- (D) Industrial.
- (E) Institutional and governmental.
- (F) Landscape.
- (G) Sales to other agencies.

(H) Saline water intrusion barriers, groundwater recharge, or conjunctive use, or any combination thereof.

- (I) Agricultural.
- (J) Distribution system water loss.

(2) The water use projections shall be in the same five-year increments described in subdivision (a).

(3) (A) The distribution system water loss shall be quantified for each of the five years preceding the plan update, in accordance with rules adopted pursuant to Section 10608.34.

(B) The distribution system water loss quantification shall be reported in accordance with a worksheet approved or developed by the department through a public process. The water loss quantification worksheet shall be based on the water system balance methodology developed by the American Water Works Association.

(C) In the plan due July 1, 2021, and in each update thereafter, data shall be included to show whether the urban retail water supplier met the distribution loss standards enacted by the board pursuant to Section 10608.34.

(4) (A) Water use projections, where available, shall display and account for the water savings estimated to result from adopted codes, standards, ordinances, or transportation and land use plans identified by the urban water supplier, as applicable to the service area.

(B) To the extent that an urban water supplier reports the information described in subparagraph (A), an urban water supplier shall do both of the following:

(i) Provide citations of the various codes, standards, ordinances, or transportation and land use plans utilized in making the projections.

(ii) Indicate the extent that the water use projections consider savings from codes, standards, ordinances, or transportation and land use plans. Water use projections that do not account for these water savings shall be noted of that fact.

(e) Provide a description of the supplier's water demand management measures. This description shall include all of the following:

(1) (A) For an urban retail water supplier, as defined in Section 10608.12, a narrative description that addresses the nature and extent of each water demand management measure implemented over the past five years. The narrative shall describe the water demand management measures that the supplier plans to implement to achieve its water use targets pursuant to Section 10608.20.

(B) For the supplement required of urban retail water suppliers by paragraph (2) of subdivision (f) of Section 10621, a narrative that describes the water demand management measures that the supplier plans to implement to achieve its urban water use objective by January 1, 2027, pursuant to Chapter 9 (commencing with Section 10609) of Part 2.55.

(B) (C) The narrative pursuant to this paragraph shall include descriptions of the following water demand management measures:

- (i) Water waste prevention ordinances.
- (ii) Metering.
- (iii) Conservation pricing.
- (iv) Public education and outreach.
- (v) Programs to assess and manage distribution system real loss.
- (vi) Water conservation program coordination and staffing support.

(vii) Other demand management measures that have a significant impact on water use as measured in gallons per capita per day, including innovative measures, if implemented.

(2) For an urban wholesale water supplier, as defined in Section 10608.12, a narrative description of the items in clauses (ii), (iv), (vi), and (vii) of subparagraph (B) (C) of paragraph (1), and a narrative description of its distribution system asset management and wholesale supplier assistance programs.

(f) Include a description of all water supply projects and water supply programs that may be undertaken by the urban water supplier to meet the total projected water use, as established pursuant to subdivision (a) of Section 10635. The urban water supplier shall include a detailed description of expected future projects and programs that the urban water supplier may implement to increase the amount of the water supply available to the urban water supplier in normal and single-dry water years and for a period of drought lasting five consecutive water years. The description shall identify specific projects and include a description of the increase in water supply that is expected to be available from each project. The description shall include an estimate with regard to the implementation timeline for each project or program.

(g) Describe the opportunities for development of desalinated water, including, but not limited to, ocean water, brackish water, and groundwater, as a long-term supply.

(h) An urban water supplier that relies upon a wholesale agency for a source of water shall provide the wholesale agency with water use projections from that agency for that source of water in five-year increments to 20 years or as far as data is available. The wholesale agency shall provide information to the urban water supplier for inclusion in the urban water supplier's plan that identifies and quantifies, to the extent practicable, the existing and planned sources of water as required by subdivision (b), available from the wholesale agency to the urban water supplier over the same five-year increments, and during various water-year types in accordance with subdivision (f). An urban water supplier may rely upon water supply information provided by the wholesale agency in fulfilling the plan informational requirements of subdivisions (b) and (f).

**SEC. 29.** Section 10631.2 of the Water Code is amended to read:

**10631.2.** (a) In addition to the requirements of Section 10631, an urban water management plan shall include any of the following information that the urban water supplier can readily obtain:

(1) An estimate of the amount of energy used to extract or divert water supplies.

(2) An estimate of the amount of energy used to convey water supplies to the water treatment plants or distribution systems.

(3) An estimate of the amount of energy used to treat water supplies.

(4) An estimate of the amount of energy used to distribute water supplies through its distribution systems.

(5) An estimate of the amount of energy used for treated water supplies in comparison to the amount used for nontreated water supplies.

(6) An estimate of the amount of energy used to place water into or withdraw from storage.

(7) Any other energy-related information the urban water supplier deems appropriate.

(b) The department shall include in its guidance for the preparation of urban water management plans a methodology for the voluntary calculation or estimation of the energy intensity of urban water systems. The department may consider studies and calculations conducted by the Public Utilities Commission in developing the methodology.

(c) The Legislature finds and declares that energy use is only one factor in water supply planning and shall not be considered independently of other factors.

SEC. 30. Section 10631.7 of the Water Code is repealed.

SEC. 31. Section 10632 of the Water Code is repealed.

**10632.** (a) Every urban water supplier shall prepare and adopt a water shortage contingency plan as part of its urban water management plan that consists of each of the following elements:

(1) The analysis of water supply reliability conducted pursuant to Section 10635.

(2) The procedures used in conducting an annual water supply and demand assessment that include, at a minimum, both of the following:

(A) The written decisionmaking process that an urban water supplier will use each year to determine its water supply reliability.

(B) The key data inputs and assessment methodology used to evaluate the urban water supplier's water supply reliability for the current year and one dry year, including all of the following:

(i) Current year unconstrained demand, considering weather, growth, and other influencing factors, such as policies to manage current supplies to meet demand objectives in future years, as applicable.

(ii) Current year available supply, considering hydrological and regulatory conditions in the current year and one dry year. The annual supply and demand assessment may consider more than one dry year solely at the discretion of the urban water supplier.

(iii) Existing infrastructure capabilities and plausible constraints.

(iv) A defined set of locally applicable evaluation criteria that are consistently relied upon for each annual water supply and demand assessment.

(v) A description and quantification of each source of water supply.

(3) (A) Six standard water shortage levels corresponding to progressive ranges of up to 10, 20, 30, 40, and 50 percent shortages and greater than 50 percent shortage. Urban water suppliers shall define these shortage levels based on the suppliers' water supply conditions, including percentage reductions in water supply, changes in groundwater levels, changes in surface elevation or level of subsidence, or other changes in hydrological or other local conditions indicative of the water supply available for use. Shortage levels shall also apply to catastrophic interruption of water supplies, including, but not limited to, a regional power outage, an earthquake, and other potential emergency events.

(B) An urban water supplier with an existing water shortage contingency plan that uses different water shortage levels may comply with the requirement in subparagraph (A) by developing and including a cross-reference relating its existing categories to the six standard water shortage levels.

(4) Shortage response actions that align with the defined shortage levels and include, at a minimum, all of the following:

(A) Locally appropriate supply augmentation actions.

(B) Locally appropriate demand reduction actions to adequately respond to shortages.

(C) Locally appropriate operational changes.

(D) Additional, mandatory prohibitions against specific water use practices that are in addition to state mandated prohibitions and appropriate to the local conditions.

(E) For each action, an estimate of the extent to which the gap between supplies and demand will be reduced by implementation of the action.

(5) Communication protocols and procedures to inform customers, the public, interested parties, and local, regional, and state governments, regarding, at a minimum, all of the following:

(A) Any current or predicted shortages as determined by the annual water supply and demand assessment described pursuant to Section 10632.1.

(B) Any shortage response actions triggered or anticipated to be triggered by the annual water supply and demand assessment described pursuant to Section 10632.1.

(C) Any other relevant communications.

(6) For an urban retail water supplier, customer compliance, enforcement, appeal, and exemption procedures for triggered shortage response actions as determined pursuant to Section 10632.2.

(7) (A) A description of the legal authorities that empower the urban water supplier to implement and enforce its shortage response actions specified in paragraph (4) that may include, but are not limited to, statutory authorities, ordinances, resolutions, and contract provisions.

(B) A statement that an urban water supplier shall declare a water shortage emergency in accordance with Chapter 3 (commencing with Section 350) of Division 1.

(C) A statement that an urban water supplier shall coordinate with any city or county within which it provides water supply services for the possible proclamation of a local emergency, as defined in Section 8558 of the Government Code.

(8) A description of the financial consequences of, and responses for, drought conditions, including, but not limited to, all of the following:

(A) A description of potential revenue reductions and expense increases associated with activated shortage response actions described in paragraph (4).

(B) A description of mitigation actions needed to address revenue reductions and expense increases associated with activated shortage response actions described in paragraph (4).

(C) A description of the cost of compliance with Chapter 3.3 (commencing with Section 365) of Division 1.

(9) For an urban retail water supplier, monitoring and reporting requirements and procedures that ensure appropriate data is collected, tracked, and analyzed for purposes of monitoring customer compliance and to meet state reporting requirements.

(10) Reevaluation and improvement procedures for systematically monitoring and evaluating the functionality of the water shortage contingency plan in order to ensure shortage risk tolerance is adequate and appropriate water shortage mitigation strategies are implemented as needed.

(b) For purposes of developing the water shortage contingency plan pursuant to subdivision (a), an urban water supplier shall analyze and define water features that are artificially supplied with water, including ponds, lakes,

waterfalls, and fountains, separately from swimming pools and spas, as defined in subdivision (a) of Section 115921 of the Health and Safety Code.

(c) The urban water supplier shall make available the water shortage contingency plan prepared pursuant to this article to its customers and any city or county within which it provides water supplies no later than 30 days after adoption of the water shortage contingency plan.

SEC. 32. Section 10632 is added to the Water Code, to read:

**10632.** (a) Every urban water supplier shall prepare and adopt a water shortage contingency plan as part of its urban water management plan that consists of each of the following elements:

(1) The analysis of water supply reliability conducted pursuant to Section 10635.

(2) The procedures used in conducting an annual water supply and demand assessment that include, at a minimum, both of the following:

(A) The written decisionmaking process that an urban water supplier will use each year to determine its water supply reliability.

(*B*) The key data inputs and assessment methodology used to evaluate the urban water supplier's water supply reliability for the current year and one dry year, including all of the following:

(*i*) Current year unconstrained demand, considering weather, growth, and other influencing factors, such as policies to manage current supplies to meet demand objectives in future years, as applicable.

(ii) Current year available supply, considering hydrological and regulatory conditions in the current year and one dry year. The annual supply and demand assessment may consider more than one dry year solely at the discretion of the urban water supplier.

(iii) Existing infrastructure capabilities and plausible constraints.

(*iv*) A defined set of locally applicable evaluation criteria that are consistently relied upon for each annual water supply and demand assessment.

(v) A description and quantification of each source of water supply.

(3) (A) Six standard water shortage levels corresponding to progressive ranges of up to 10, 20, 30, 40, and 50 percent shortages and greater than 50 percent shortage. Urban water suppliers shall define these shortage levels based on the suppliers' water supply conditions, including percentage reductions in water supply, changes in groundwater levels, changes in surface elevation or level of subsidence, or other changes in hydrological or other local conditions indicative of the water supply available for use. Shortage levels shall also apply to catastrophic interruption of water supplies, including, but not limited to, a regional power outage, an earthquake, and other potential emergency events.

(B) An urban water supplier with an existing water shortage contingency plan that uses different water shortage levels may comply with the requirement in subparagraph (A) by developing and including a cross-reference relating its existing categories to the six standard water shortage levels.

(4) Shortage response actions that align with the defined shortage levels and include, at a minimum, all of the following:

(A) Locally appropriate supply augmentation actions.

(B) Locally appropriate demand reduction actions to adequately respond to shortages.

(C) Locally appropriate operational changes.

(D) Additional, mandatory prohibitions against specific water use practices that are in addition to state-mandated prohibitions and appropriate to the local conditions.

(E) For each action, an estimate of the extent to which the gap between supplies and demand will be reduced by implementation of the action.

(5) Communication protocols and procedures to inform customers, the public, interested parties, and local, regional, and state governments, regarding, at a minimum, all of the following:

(A) Any current or predicted shortages as determined by the annual water supply and demand assessment described pursuant to Section 10632.1.

(B) Any shortage response actions triggered or anticipated to be triggered by the annual water supply and demand assessment described pursuant to Section 10632.1.

(C) Any other relevant communications.

(6) For an urban retail water supplier, customer compliance, enforcement, appeal, and exemption procedures for triggered shortage response actions as determined pursuant to Section 10632.2.

(7) (A) A description of the legal authorities that empower the urban water supplier to implement and enforce its shortage response actions specified in paragraph (4) that may include, but are not limited to, statutory authorities, ordinances, resolutions, and contract provisions.

(B) A statement that an urban water supplier shall declare a water shortage emergency in accordance with Chapter 3 (commencing with Section 350) of Division 1.

(C) A statement that an urban water supplier shall coordinate with any city or county within which it provides water supply services for the possible proclamation of a local emergency, as defined in Section 8558 of the Government Code.

(8) A description of the financial consequences of, and responses for, drought conditions, including, but not limited to, all of the following:

(A) A description of potential revenue reductions and expense increases associated with activated shortage response actions described in paragraph (4).

(*B*) A description of mitigation actions needed to address revenue reductions and expense increases associated with activated shortage response actions described in paragraph (4).

(C) A description of the cost of compliance with Chapter 3.3 (commencing with Section 365) of Division 1.

(9) For an urban retail water supplier, monitoring and reporting requirements and procedures that ensure appropriate data is collected, tracked, and analyzed for purposes of monitoring customer compliance and to meet state reporting requirements.

(10) Reevaluation and improvement procedures for systematically monitoring and evaluating the functionality of the water shortage contingency plan in order to ensure shortage risk tolerance is adequate and appropriate water shortage mitigation strategies are implemented as needed.

(b) For purposes of developing the water shortage contingency plan pursuant to subdivision (a), an urban water supplier shall analyze and define water features that are artificially supplied with water, including ponds, lakes, waterfalls, and fountains, separately from swimming pools and spas, as defined in subdivision (a) of Section 115921 of the Health and Safety Code.

(c) The urban water supplier shall make available the water shortage contingency plan prepared pursuant to this article to its customers and any city or county within which it provides water supplies no later than 30 days after adoption of the water shortage contingency plan.

**SEC. 33.** Section 10632.1 is added to the Water Code, to read:

**10632.1.** An urban water supplier shall conduct an annual water supply and demand assessment pursuant to subdivision (a) of Section 10632 and, on or before June 1 of each year, submit an annual water shortage assessment report to the department with information for anticipated shortage, triggered shortage response actions, compliance and enforcement actions, and communication actions consistent with the supplier's water shortage contingency plan. An urban water supplier that relies on imported water from the State Water Project or the Bureau of Reclamation shall submit its annual water supply and demand assessment within 14 days of receiving its final allocations, or by June 1 of each year, whichever is later.

**SEC. 34.** Section 10632.2 is added to the Water Code, to read:

**10632.2.** An urban water supplier shall follow, where feasible and appropriate, the prescribed procedures and implement determined shortage response actions in its water shortage contingency plan, as identified in subdivision (a) of Section 10632, or reasonable alternative actions, provided that descriptions of the alternative

actions are submitted with the annual water shortage assessment report pursuant to Section 10632.1. Nothing in this section prohibits an urban water supplier from taking actions not specified in its water shortage contingency plan, if needed, without having to formally amend its urban water management plan or water shortage contingency plan.

**SEC. 35.** Section 10632.3 is added to the Water Code, to read:

**10632.3.** It is the intent of the Legislature that, upon proclamation by the Governor of a state of emergency under the California Emergency Services Act (Chapter 7 (commencing with Section 8550) of Division 1 of Title 2 of the Government Code) based on drought conditions, the board defer to implementation of locally adopted water shortage contingency plans to the extent practicable.

SEC. 36. Section 10635 of the Water Code is amended to read:

**10635.** (a) Every urban water supplier shall include, as part of its urban water management plan, an assessment of the reliability of its water service to its customers during normal, dry, and multiple dry water years. This water supply and demand assessment shall compare the total water supply sources available to the water supplier with the long-term total projected water use over the next 20 years, in five-year increments, for a normal water year, a single dry water year, and a drought lasting five consecutive water years. The water service reliability assessment shall be based upon the information compiled pursuant to Section 10631, including available data from state, regional, or local agency population projections within the service area of the urban water supplier.

(b) Every urban water supplier shall include, as part of its urban water management plan, a drought risk assessment for its water service to its customers as part of information considered in developing the demand management measures and water supply projects and programs to be included in the urban water management plan. The urban water supplier may conduct an interim update or updates to this drought risk assessment within the five-year cycle of its urban water management plan update. The drought risk assessment shall include each of the following:

(1) A description of the data, methodology, and basis for one or more supply shortage conditions that are necessary to conduct a drought risk assessment for a drought period that lasts five consecutive water years, starting from the year following when the assessment is conducted.

(2) A determination of the reliability of each source of supply under a variety of water shortage conditions. This may include a determination that a particular source of water supply is fully reliable under most, if not all, conditions.

(3) A comparison of the total water supply sources available to the water supplier with the total projected water use for the drought period.

(4) Considerations of the historical drought hydrology, plausible changes on projected supplies and demands under climate change conditions, anticipated regulatory changes, and other locally applicable criteria.

(c) The urban water supplier shall provide that portion of its urban water management plan prepared pursuant to this article to any city or county within which it provides water supplies no later than 60 days after the submission of its urban water management plan.

(d) Nothing in this article is intended to create a right or entitlement to water service or any specific level of water service.

(e) Nothing in this article is intended to change existing law concerning an urban water supplier's obligation to provide water service to its existing customers or to any potential future customers.

**SEC. 37.** Section 10640 of the Water Code is amended to read:

**10640.** (a) Every urban water supplier required to prepare a plan pursuant to this part shall prepare its plan pursuant to Article 2 (commencing with Section 10630). The supplier shall likewise periodically review the plan as required by Section 10621, and any amendments or changes required as a result of that review shall be adopted pursuant to this article.

(b) Every urban water supplier required to prepare a water shortage contingency plan shall prepare a water shortage contingency plan pursuant to Section 10632. The supplier shall likewise periodically review the water
#### Today's Law As Amended

shortage contingency plan as required by paragraph (10) of subdivision (a) of Section 10632 and any amendments or changes required as a result of that review shall be adopted pursuant to this article.

**SEC. 38.** Section 10641 of the Water Code is amended to read:

**10641.** An urban water supplier required to prepare a plan or a water shortage contingency plan may consult with, and obtain comments from, any public agency or state agency or any person who has special expertise with respect to water demand management methods and techniques.

SEC. 39. Section 10642 of the Water Code is amended to read:

**10642.** Each urban water supplier shall encourage the active involvement of diverse social, cultural, and economic elements of the population within the service area prior to and during the preparation of both the plan and the water shortage contingency plan. Prior to adopting either, the urban water supplier shall make both the plan and the water shortage contingency plan available for public inspection and shall hold a public hearing or hearings thereon. Prior to any of these hearings, notice of the time and place of the hearing shall be published within the jurisdiction of the publicly owned water supplier pursuant to Section 6066 of the Government Code. The urban water supplier shall provide notice of the time and place of a hearing to any city or county within which the supplier provides water supplies. Notices by a local public agency pursuant to this section shall be provided pursuant to Chapter 17.5 (commencing with Section 7290) of Division 7 of Title 1 of the Government Code. A privately owned water supplier shall provide an equivalent notice within its service area. After the hearing or hearings, the plan or water shortage contingency plan shall be adopted as prepared or as modified after the hearing or hearings.

**SEC. 40.** Section 10644 of the Water Code is amended to read:

**10644.** (a) (1) An urban water supplier shall submit to the department, the California State Library, and any city or county within which the supplier provides water supplies a copy of its plan no later than 30 days after adoption. Copies of amendments or changes to the plans shall be submitted to the department, the California State Library, and any city or county within which the supplier provides water supplies water supplies within 30 days after adoption.

(2) The plan, or amendments to the plan, submitted to the department pursuant to paragraph (1) shall be submitted electronically and shall include any standardized forms, tables, or displays specified by the department.

(b) If an urban water supplier revises its water shortage contingency plan, the supplier shall submit to the department a copy of its water shortage contingency plan prepared pursuant to subdivision (a) of Section 10632 no later than 30 days after adoption, in accordance with protocols for submission and using electronic reporting tools developed by the department.

(c) (1) (A) Notwithstanding Section 10231.5 of the Government Code, the department shall prepare and submit to the Legislature, on or before July 1, in the years ending in seven and two, a report summarizing the status of the plans and water shortage contingency plans adopted pursuant to this part. The report prepared by the department shall identify the exemplary elements of the individual plans and water shortage contingency plans. The department shall provide a copy of the report to each urban water supplier that has submitted its plan and water shortage contingency plan to the department. The department shall also prepare reports and provide data for any legislative hearings designed to consider the effectiveness of plans and water shortage contingency plans submitted pursuant to this part.

(B) The department shall prepare and submit to the board, on or before September 30 of each year, a report summarizing the submitted water supply and demand assessment results along with appropriate reported water shortage conditions and the regional and statewide analysis of water supply conditions developed by the department. As part of the report, the department shall provide a summary and, as appropriate, urban water supplier specific information regarding various shortage response actions implemented as a result of annual supplier-specific water supply and demand assessments performed pursuant to Section 10632.1.

(C) The department shall submit the report to the Legislature for the 2015 plans by July 1, 2017, and the report to the Legislature for the 2020 plans and water shortage contingency plans by July 1, 2022.

(2) A report to be submitted pursuant to subparagraph (A) of paragraph (1) shall be submitted in compliance with Section 9795 of the Government Code.

#### Today's Law As Amended

(d) The department shall make available to the public the standard the department will use to identify exemplary water demand management measures.

**SEC. 41.** Section 10645 of the Water Code is amended to read:

**10645.** (a) Not later than 30 days after filing a copy of its plan with the department, the urban water supplier and the department shall make the plan available for public review during normal business hours.

(b) Not later than 30 days after filing a copy of its water shortage contingency plan with the department, the urban water supplier and the department shall make the plan available for public review during normal business hours.

SEC. 42. Section 10650 of the Water Code is amended to read:

**10650.** Any actions or proceedings, other than actions by the board, to attack, review, set aside, void, or annul the acts or decisions of an urban water supplier on the grounds of noncompliance with this part shall be commenced as follows:

(a) An action or proceeding alleging failure to adopt a plan or a water shortage contingency plan shall be commenced within 18 months after that adoption is required by this part.

(b) Any action or proceeding alleging that a plan or water shortage contingency plan, or action taken pursuant to either, does not comply with this part shall be commenced within 90 days after filing of the plan or water shortage contingency plan or an amendment to either pursuant to Section 10644 or the taking of that action.

SEC. 43. Section 10651 of the Water Code is amended to read:

**10651.** In any action or proceeding to attack, review, set aside, void, or annul a plan or a water shortage contingency plan, or an action taken pursuant to either by an urban water supplier on the grounds of noncompliance with this part, the inquiry shall extend only to whether there was a prejudicial abuse of discretion. Abuse of discretion is established if the supplier has not proceeded in a manner required by law or if the action by the water supplier is not supported by substantial evidence.

**SEC. 44.** Section 10653 of the Water Code is amended to read:

**10653.** The adoption of a plan shall satisfy any requirements of state law, regulation, or order, including those of the board and the Public Utilities Commission, for the preparation of water management plans, water shortage contingency plans, or conservation plans; provided, that if the board or the Public Utilities Commission requires additional information concerning water conservation, drought response measures, or financial conditions to implement its existing authority, nothing in this part shall be deemed to limit the board or the commission in obtaining that information. The requirements of this part shall be satisfied by any urban water demand management plan that complies with analogous federal laws or regulations after the effective date of this part, and which substantially meets the requirements of this part.

**SEC. 45.** Section 10654 of the Water Code is amended to read:

**10654.** An urban water supplier may recover in its rates the costs incurred in preparing its urban water management plan, its drought risk assessment, its water supply and demand assessment, and its water shortage contingency plan and implementing the reasonable water conservation measures included in either of the plans.

SEC. 46. Section 10656 of the Water Code is amended to read:

**10656.** An urban water supplier is not eligible for a water grant or loan awarded or administered by the state unless the urban water supplier complies with this part.

#### **SEC. 47.** Section 10657 is added to the Water Code, to read:

**10657.** The department may adopt regulations regarding the definitions of water, water use, and reporting periods, and may adopt any other regulations deemed necessary or desirable to implement this part. In developing regulations pursuant to this section, the department shall solicit broad public participation from stakeholders and other interested persons.

**SEC. 48.** This act shall become operative only if Assembly Bill 1668 of the 2017–18 Regular Session is enacted and becomes effective.



# Appendix B

**AWE Model Assumptions** 

# AWE CONSERVATION TRACKING TOOL: COMMON ASSUMPTIONS WORKSHEET

**Getting Started**: On this worksheet you enter information the tracking tool needs to operate. This includes specifying whether to use English or Metric units, setting up customer classes, specifying the first year for forecasts, entering forecasted population, housing, and customer accounts, setting financial assumptions, providing information needed to calculate water and energy savings due to appliance and plumbing standards for toilets, clothes washers, and dishwashers, and providing information needed to calculate water savings for landscape conservation measures included in the conservation measure

			State	CA		Model will	use CA plumbin	g standards	
			Volume Units		•	Flow Units Will Be:		MGD	]
Population, Housing,	and Accou	nt Forecas	sts r	new WU data -	- DU scaled b	ased on ABAG	i Pop, 2025-20	)45 Sector sca	led based on A
Enter Starting Year for Forecasts	i	2010	2	2020 sectors b	ased on pop	only			
Population & Housing	2010	2015	2020	2025	2030	2035	2040	2045	
Population	183,716	191,575	191,269	197,939	201,987	204,750	208,324	211,961	
Single Family Dwelling Units	55,038	55,291	55,202	57,127	58,296	59,093	60,125	61,174	
Multi Family Dwelling Units	26,036	26,338	26,296	27,213	27,769	28,149	28,640	29,140	
Number of Accounts									
Single Family	51,241	51,474	51,392	52,410	53,482	54,214	55,160	56,123	
Multi Family	3,778	3,801	3,795	3,857	3,936	3,990	4,060	4,130	
CII	3,479	3,476	3,470	3,495	3,553	3,581	3,589	3,597	Business/Indus
Irrigation	838	850	849	866	881	888	890	892	
Other	479	442	441	495	503	507	508	509	
Not in use									
Not in use									
Not in use									
Not in use									

# **Financial Assumptions**

These inputs are used by the tracking tool to standardize costs and benefits, calculate present values, and estimate utility and customer benefits of conservation.

Dollar Base Year Annual Inflation Rate Nominal Interest Rate

201	9
3.09	%
2.39	%

Utility Rates in 2010

Average Class Rate (2019 Dollars)				Annual Rate of Increase				
Water	Sewer	Electricity	Gas	Water	Sewer	Electricity	Gas	

Scenario "'Empty" loaded into model on 7/27/2

BAG Pop and Employment,

trial + Institutional = CII

Customer Class	(\$/Thou Gal)	(\$/Thou Gal)	(\$/KWh)	(\$/Therm)	(%/Yr)	(%/Yr)	(%/Yr)	(%/Yr)
Single Family	\$10.36		\$0.28	\$2.00	4.0%		3.0%	3.0%
Multi Family	\$7.42		\$0.28	\$2.00	4.0%		3.0%	3.0%
CII	\$11.51		\$0.24	\$0.80	4.0%		3.0%	3.0%
Irrigation	\$12.77		\$0.31		4.0%		3.0%	3.0%
Other					4.0%		3.0%	3.0%
Not in use								
Not in use								
Not in use								
Not in use								

### Information Needed to Calculate Water/Energy Savings from Plumbing/Appliance Standards

These inputs are used by the tracking tool to estimate water and energy savings for national toilet and showerhead standards, which first took effect in 1994, and clothes washer and dishwasher appliance standards, which first included maximum allowable water factors in 2011 and 2010, respectively. Toilet standards took effect in 1992 in California and Texas.

	Single Family	Multi Family
Persons per household	2.4	2.40
Full Baths/Dwelling Unit	2.01	1.68
Half Baths/Dwelling Unit	0.24	0.59
Dwellling Units in 1992	50,156	23,953
Population in 1990	164,249	

# Information Needed to Calculate Water Savings for Landscape Measures in Library

Average landscape water use for residential and non-residential sites is used by the model to calculate water savings for various landscape conservation measures included in the program library. Average landscape water use is calculated using the following equation. Alternatively, you can use your own landscape water use estimate by selecting the "Use My Own Estimate" option.

use per site = 
$$\left(\frac{1}{irr.eff.}\right) \times (ET_0 \times K_L - R_e) \times Area \times C_v$$
, where

irr.eff. = typical irrigation efficiency

 $ET_0 = reference \ evapotranspiration$ 

 $K_L = landscape \ coefficient \ (\% \ of ET_0 \ needed \ by \ crop)$ 

 $R_e = effective rainfall (\% of annual rainfall contributing to plant water requirement)$ 

 $C_v = coefficient$  that converts water use to appropriate volume units (gal for english units,  $M^3$  for metric units)

O Use my own landscape water use estimates

• Use model's landscape water use calculator

Reference ET

in/yr

45.20

Avg Annual Rainfall	in/yr	47.41
Effective Rainfall	%	25%

Landscape Water Requirement Coefficient ( $K_L$ )

Turf	% of $ET_0$	80%
Other than turf	% of $ET_0$	40%

			Non
		Residential	Residential
Avg Landscape Area Per Site	ft^2		
Avg Turf Area (% of Total)	%		
Avg Irrigation Efficiency (%)	%	75%	75%

Irrigation Requirement		Residential	Non Residential
Turf Area	in/ft^2/yr	32	32
Other	in/ft^2/yr	8	8

			Non
Avg Landscape Water Use Per S	Site	Residential	Residential
Turf Area	Gal/Yr	0	0
Other	Gal/Yr	0	0
Total	Gal/Yr	0	0

# AWE CONSERVATION TRACKING TOOL: ENTER UTILITY AVOIDED COSTS WORKSHEET

Enter utility avoided costs: The primary benefit of conservation to the utility and its ratepayers is avoiding future water supply and wastewater costs. A utility avoids cost by not having to purchase (or otherwise acquire), transport, treat and distribute water supply, and by not having to collect, treat, and dispose of wastewater. The variable costs of these activities are major components of avoided cost. Conservation, if done at sufficient scale, may also allow the utility to defer or even entirely avoid future expansion of system capacity. This can be a major source of benefit in some cases.

The tracking tool comes with a calculator you can use to estimate your avoided costs. Alternatively, you can enter you own avoided cost estimates by selecting

O Use my own avoided cost estimates

• Use model's avoided cost calculator

### Tracking Tool Utility Avoided Cost Calculator

### Water and Wastewater System Variable Costs (2019 Dollars)

	Wa	ater	Waste	ewater
		Nominal		Nominal
	\$/AF	Increase	\$/AF	Increase
Water purchase	\$1,151	5.0%	NA	NA
Energy	\$115	3.0%		
Chemicals	\$50	3.0%		
Other variable cost				
Total \$1,316		4.8%	\$0	0.0%

### Variable Cost Forecast

Variable Cost	Units	2010	2015	2020	2025	2030	2035	2040	2045	2040-2045	Ann Grwt
Water Supply	\$/AF	\$1,316	\$1,435	\$1,564	\$1,705	\$1,858	\$2,026	\$2,208	\$2,407	1.7%	
Wastewater	\$/AF	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	0.0%	

Scenario "'Empty" loaded into model on 7/27/2016 5:44:04 PM

Download CUWCC Avoided Cost & Environmental Benefits Models

th %

#### AWE CONSERVATION TRACKING TOOL: DEFINE CONSERVATION ACTIVITIES WORKSHEET

Define conservation activities: Click the Define/Edit/Delete button to setup and edit conservation activities. You can use the form to define your own activities or import activities from the tracking tool's library. Once imported, library activities can be customized. Conservation activity specifications are stored in a table on this worksheet. This table is hidden by default. You can unhide the table by clicking the "Show Activities Table" button. You can edit activities directly in the table if you find this easier than using the form. HOWEVER, DO NOT DELETE TABLE ROWS. ONLY USE THE FORM TO DELETE CONSERVATION ACTIVITIES.

Scenario "'Empty" loaded into model on 7/27/2016 5:44:04 PM

#### NOTE: You can define activities in the table rather than using the form. BUT ONLY USE THE FORM TO DELETE ACTIVITIES.

				Savings,	Sourings Dook	Covinas	Savings,	Utility Costs	Utility Costs	Utility Costs	Utility Costs,	Litility Costs	Utility Costs,	Portisipant	Douticinont	Douticinont	Dorticinont	Dortisinant	Douticinout	Participant	Dlumbing	Dlumbing Code	Plumbing Code,
Activity			Savings, Per	Rate of	Period (% of	Useful Life	Riders (% of	Year	Initial Fixed	Initial Variable	Follow-up	Follow-up	Variable	Costs, Year	Costs Initial	Costs, Years of	Costs, On-	Savings	Savings, Gas	Flectricity	Code, Year	Unit Savings	Replacement
ID	Activity Name	Class	Unit (gpy)	Decay (%)	Annual Savings)	(yrs)	Participants)	Denominated	(\$)	(\$/unit)	(yrs)	Fixed (\$/yr)	(\$/unit/yr)	Denominated	(\$)	On-going (yrs)	going (\$/Yr)	Sewer (gpy)	(Therms/Gal)	(KWh/Gal)	Effective	(gpy)	Rate NRR (%)
1	High Efficiency Clothes Washer Rebate - Residentia	Single Family	4,276	0%	0%	15	0%	2019		\$195.00				2019	\$626.00			5,000.00	0.0035	0.0036	2011	3500	7.14%
2	Water Use Surveys/Audits - SFR	Single Family	12,373	20%	68%	5	0%	2019		\$210.60				2019	\$50.00			4,949.20	0.0010	0.0000	0	0	0.00%
3	Landscape Conversion or Turf Removal -SFR	Single Family	12	0%	80%	10	0%	2019		\$1.30				2019	\$1.00			0.00	0.0000	0.0000	0	0	0.00%
4	Smart Irrigation Controller (Weather-Based Irrigation	Single Family	17,258	0%	70%	10	0%	2019		\$247.84				2019	\$172.39			0.00	0.0000	0.0000	0	0	0.00%
5	Water Use Surveys/Audits - CII	CII	93,521	20%		5	0%	2019		\$2,000.00				2019	\$2,500.00								
6	Incentivize Irrigation Equipment Upgrades - Large La	Irrigation	36,108	0%		10	0%	2019		\$1,300.00				2019	\$500.00								
7	Smart Irrigation Controller (Weather-Based Irrigation	Irrigation	96,645	0%	70%	10	0%	2019		\$1,094.21				2019	\$568.54			0.00	0.0000	0.0000	0	0	0.00%
8	Water Use Surveys/Audits - MFR	Multi Family	93,521	20%		5	0%	2019		\$2,000.00				2019	\$2,500.00								
9	High Efficiency Faucet Aerator / Showerhead Giveav	Single Family	4,013	0%		5		2019		\$31.20				2019	\$50.00								
10	Incentivize Irrigation Equipment Upgrades - SFR	Single Family	5,639	0%		10		2019		\$325.00				2019	\$250.00								
11	Landscape Conversion or Turf Removal - MFR and	CII	12	0%		10		2019		\$1.30				2019	\$1.00								
12	Soil Moisture Sensor Rebate	Single Family	13,964	0%		20		2019		\$260.00				2019	\$300.00								
13	Water Savings Incentive Program for CII	CII	5,995,666	0%		15		2019		\$4,676.62				2019	\$5,396.10								
14	Restaurant Spray Nozzle Rebates	CII	43,830	0%		5	0%	2019		\$65.00				2020	\$50.00			43,830.00	0.0083				
15																							
16																							

#### AWE CONSERVATION TRACKING

Enter annual conservation activity: Use this worksheet to enter the annual activity levels for the conservation activities you defined on the 4. Define Activities worksheet. You can enter activity through the end of your forecast period, but this is not required. It is okay to enter activity for shorter periods. You also can start an activity in

Activity II	D Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1	Single Family	y High Efficiency Clothes Washer Rebate - Res	303	303	303	303	303																				
2	Single Famil	y Water Use Surveys/Audits - SFR	731	731	731	731	731																				
3	Single Famil	y Landscape Conversion or Turf Removal -SFF	27856	27856	27856	27856	27856																				
4	Single Famil	y Smart Irrigation Controller (Weather-Based Ir	37	37	37	37	37																				
5	CII	Water Use Surveys/Audits - CII	43	43	43	43	43																				
6	Irrigation	Incentivize Irrigation Equipment Upgrades - L	21	21	21	21	21																				
7	Irrigation	Smart Irrigation Controller (Weather-Based Ir	6	6	6	6	6																				
8	Multi Family	Water Use Surveys/Audits - MFR	43	43	43	43	43																				
9	Single Famil	y High Efficiency Faucet Aerator / Showerhead	909	909	909	909	909																				
10	Single Famil	y Incentivize Irrigation Equipment Upgrades - S	37	37	37	37	37																				
11	CII	Landscape Conversion or Turf Removal - MF	83568	83568	83568	83568	83568																				
12	Single Famil	y Soil Moisture Sensor Rebate	37	37	37	37	37																				
13	CII	Water Savings Incentive Program for CII	1	1	1	1	1																				
14	CII	Restaurant Spray Nozzle Rebates	50	50	50	50	50																				
	·	· · · · ·							· · · · ·												· · · · ·	· · · · ·					

Annual Program Overhead Cost (2019 dollars)	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
Enter additional program cost not included in activity definitions																									

#### Model calculation tables below this line. Do not delete or modify.

#### Effective Conservation Activity

Activity ID	) Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1	Single Family	High Efficiency Clothes Washer Rebate - Res	303	606	909	1,212	1,515	1,515	1,515	1,515	1,515	1,515	1,515	1,515	1,515	1,515	1,515	1,212	909	606	303	0	0	0	0	0	0
2	Single Family	Water Use Surveys/Audits - SFR	731	1,316	1,784	2,158	2,457	1,726	1,142	674	299	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
3	Single Family	Landscape Conversion or Turf Removal -SFF	27,856	55,712	83,568	111,424	139,280	139,280	139,280	139,280	139,280	139,280	111,424	83,568	55,712	27,856	0	0	0	0	0	0	0	0	0	0	0
4	Single Family	Smart Irrigation Controller (Weather-Based Ir	37	74	111	148	185	185	185	185	185	185	148	111	74	37	0	0	0	0	0	0	0	0	0	0	0
5	CII	Water Use Surveys/Audits - CII	43	77	105	127	145	102	67	40	18	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
6	Irrigation	Incentivize Irrigation Equipment Upgrades - L	21	42	63	84	105	105	105	105	105	105	84	63	42	21	0	0	0	0	0	0	0	0	0	0	0
7	Irrigation	Smart Irrigation Controller (Weather-Based Ir	6	12	18	24	30	30	30	30	30	30	24	18	12	6	0	0	0	0	0	0	0	0	0	0	0
8	Multi Family	Water Use Surveys/Audits - MFR	43	77	105	127	145	102	67	40	18	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
9	Single Family	High Efficiency Faucet Aerator / Showerhead	909	1,818	2,727	3,636	4,545	3,636	2,727	1,818	909	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
10	Single Family	Incentivize Irrigation Equipment Upgrades - S	37	74	111	148	185	185	185	185	185	185	148	111	74	37	0	0	0	0	0	0	0	0	0	0	0
11	CII	Landscape Conversion or Turf Removal - MF	83,568	167,136	250,704	334,272	417,840	417,840	417,840	417,840	417,840	417,840	334,272	250,704	167,136	83,568	0	0	0	0	0	0	0	0	0	0	0
12	Single Family	Soil Moisture Sensor Rebate	37	74	111	148	185	185	185	185	185	185	185	185	185	185	185	185	185	185	185	185	148	111	74	37	0
13	CII	Water Savings Incentive Program for CII	1	2	3	4	5	5	5	5	5	5	5	5	5	5	5	4	3	2	1	0	0	0	0	0	0
14	CII	Restaurant Spray Nozzle Rebates	50	100	150	200	250	200	150	100	50	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0

#### Gross Water Savings (AF)

Activity ID	Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1	Single Family	High Efficiency Clothes Washer Rebate - Res	4.0	8.0	11.9	15.9	19.9	19.9	19.9	19.9	19.9	19.9	19.9	19.9	19.9	19.9	19.9	15.9	11.9	8.0	4.0	0.0	0.0	0.0	0.0	0.0	0.0
2	Single Family	Water Use Surveys/Audits - SFR	27.8	50.0	67.7	81.9	93.3	65.6	43.3	25.6	11.4	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
3	Single Family	Landscape Conversion or Turf Removal -SFF	1.0	2.1	3.1	4.1	5.1	5.1	5.1	5.1	5.1	5.1	4.1	3.1	2.1	1.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
4	Single Family	Smart Irrigation Controller (Weather-Based Ir	2.0	3.9	5.9	7.8	9.8	9.8	9.8	9.8	9.8	9.8	7.8	5.9	3.9	2.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
5	CII	Water Use Surveys/Audits - CII	12.3	22.2	30.1	36.4	41.5	29.1	19.3	11.4	5.1	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
6	Irrigation	Incentivize Irrigation Equipment Upgrades - L	2.3	4.7	7.0	9.3	11.6	11.6	11.6	11.6	11.6	11.6	9.3	7.0	4.7	2.3	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
7	Irrigation	Smart Irrigation Controller (Weather-Based Ir	1.8	3.6	5.3	7.1	8.9	8.9	8.9	8.9	8.9	8.9	7.1	5.3	3.6	1.8	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
8	Multi Family	Water Use Surveys/Audits - MFR	12.3	22.2	30.1	36.4	41.5	29.1	19.3	11.4	5.1	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
9	Single Family	High Efficiency Faucet Aerator / Showerhead	11.2	22.4	33.6	44.8	56.0	44.8	33.6	22.4	11.2	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
10	Single Family	Incentivize Irrigation Equipment Upgrades - S	0.6	1.3	1.9	2.6	3.2	3.2	3.2	3.2	3.2	3.2	2.6	1.9	1.3	0.6	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
11	CII	Landscape Conversion or Turf Removal - MF	3.1	6.2	9.2	12.3	15.4	15.4	15.4	15.4	15.4	15.4	12.3	9.2	6.2	3.1	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
12	Single Family	Soil Moisture Sensor Rebate	1.6	3.2	4.8	6.3	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	6.3	4.8	3.2	1.6	0.0
13	CII	Water Savings Incentive Program for CII	18.4	36.8	55.2	73.6	92.0	92.0	92.0	92.0	92.0	92.0	92.0	92.0	92.0	92.0	92.0	73.6	55.2	36.8	18.4	0.0	0.0	0.0	0.0	0.0	0.0
14	CII	Restaurant Spray Nozzle Rebates	6.7	13.5	20.2	26.9	33.6	26.9	20.2	13.5	6.7	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
	Total Gross W	Vater Savings	105.1	199.8	286.0	365.6	439.7	369.4	309.5	258.0	213.3	173.9	163.0	152.2	141.4	130.6	119.8	97.4	75.1	52.7	30.3	7.9	6.3	4.8	3.2	1.6	0.0

#### Activity ID Class Activity Name <u>2023</u> 2024 2025 2026 2027 2028 2029 2030 2031 2032 2033 2034 2035 2036 2037 2021 2022 1 Single Family High Efficiency Clothes Washer Rebate - Res 2 Single Family Water Use Surveys/Audits - SFR 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 \_\_\_\_ 34.0 18.9 46.1 55.7 63.4 44.6 29.5 17.4 7.7 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 3 Single Family Landscape Conversion or Turf Removal -SFF 0.8 1.6 2.5 3.3 4.1 4.1 4.1 4.1 4.1 4.1 3.3 2.5 1.6 0.8 0.0 0.0 0.0


2038	2039	2040	2041	2042	2043	2044	2045
0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0

4       Single Family       Smart Irrigation Controller (Weather-Based Ir       1.4       2.7       4.1       5.6       6.9 <th><u>34.6</u> <u>24.9</u> <u>17.2</u> <u>13.8</u> <u>10.3</u> <u>6.9</u> <u>3.4</u> <u>0.0</u> <u>0.0</u></th> <th>0.0</th>	<u>34.6</u> <u>24.9</u> <u>17.2</u> <u>13.8</u> <u>10.3</u> <u>6.9</u> <u>3.4</u> <u>0.0</u>	0.0
4       Single Family       Smart Irrigation Controller (Weather-Based Ir       1.4       2.7       4.1       5.5       6.9       6.9       6.9       6.9       5.5       4.1       2.7       1.4       0.0 <td><u>0.0</u> 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0</td> <td>0.0</td>	<u>0.0</u> 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	0.0
4       Single Family       Smart Irrigation Controller (Weather-Based Ir       1.4       2.7       4.1       5.5       6.9       6.0       0.0	0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	0.0
4       Single Family       Smart Irrigation Controller (Weather-Based Ir       1.4       2.7       4.1       5.5       4.1       2.7       1.4       0.0	0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	0.0
4       Single Family       Smart Irrigation Controller (Weather-Based Ir       1.4       2.7       4.1       5.5       6.9       6.9       6.9       6.9       6.9       6.9       5.5       4.1       2.7       1.4       0.0	0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	0.0
4       Single Family       Smart Irrigation Controller (Weather-Based Ir       1.4       2.7       4.1       5.5       6.9       6.9       6.9       6.9       6.9       6.9       5.5       4.1       2.7       1.4       0.0	0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	0.0
4       Single Family       Smart Irrigation Controller (Weather-Based Ir       1.4       2.7       4.1       5.5       6.9       6.0       6.0       6.0       6.0       6.0       6.0       6.0       6.0       6.0       6.0       6.0       6.0       6.0       6.0       6.0       6.0       6.0       6.0       6.0	0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	0.0
4       Single Family       Smart Irrigation Controller (Weather-Based Ir       1.4       2.7       4.1       5.5       6.9       6.9       6.9       6.9       6.9       6.9       5.5       4.1       2.7       1.4       0.0 <td>0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0</td> <td>0.0</td>	0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	0.0
4       Single Family       Smart Irrigation Controller (Weather-Based Ir       1.4       2.7       4.1       5.5       6.9	6.2         6.2         6.2         5.0         3.7         2.5         1.2         0.0 <td>0.0</td>	0.0
4         Single Family         Smart Irrigation Controller (Weather-Based Ir         1.4         2.7         4.1         5.5         6.9         6.9         6.9         6.9         6.9         5.5         4.1         2.7         1.4         0.0 <th< td=""><td>0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0</td><td>0.0</td></th<>	0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	0.0
4 Single Family Smart Irrigation Controller (Weather-Based Ir 1.4 2.7 4.1 5.5 6.9 6.9 6.9 6.9 6.9 6.9 5.5 4.1 2.7 1.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	0.0
	6.9         6.9         6.9         5.5         4.1         2.7         1.4         0.0 <td>0.0</td>	0.0

#### Off Peak Gross Water Savings (AF)

Activity II	D Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1	Single Family	High Efficiency Clothes Washer Rebate - Res	4.0	8.0	11.9	15.9	19.9	19.9	19.9	19.9	19.9	19.9	19.9	19.9	19.9	19.9	19.9	15.9	11.9	8.0	4.0	0.0	0.0	0.0	0.0	0.0	0.0
2	Single Family	Water Use Surveys/Audits - SFR	8.9	16.0	21.7	26.2	29.9	21.0	13.9	8.2	3.6	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
3	Single Family	Landscape Conversion or Turf Removal -SFF	0.2	0.4	0.6	0.8	1.0	1.0	1.0	1.0	1.0	1.0	0.8	0.6	0.4	0.2	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
4	Single Family	Smart Irrigation Controller (Weather-Based Ir	0.6	1.2	1.8	2.4	2.9	2.9	2.9	2.9	2.9	2.9	2.4	1.8	1.2	0.6	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
5	CII	Water Use Surveys/Audits - CII	12.3	22.2	30.1	36.4	41.5	29.1	19.3	11.4	5.1	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
6	Irrigation	Incentivize Irrigation Equipment Upgrades - L	2.3	4.7	7.0	9.3	11.6	11.6	11.6	11.6	11.6	11.6	9.3	7.0	4.7	2.3	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
7	Irrigation	Smart Irrigation Controller (Weather-Based Ir	0.5	1.1	1.6	2.1	2.7	2.7	2.7	2.7	2.7	2.7	2.1	1.6	1.1	0.5	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
8	Multi Family	Water Use Surveys/Audits - MFR	12.3	22.2	30.1	36.4	41.5	29.1	19.3	11.4	5.1	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
9	Single Family	High Efficiency Faucet Aerator / Showerhead	11.2	22.4	33.6	44.8	56.0	44.8	33.6	22.4	11.2	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
10	Single Family	Incentivize Irrigation Equipment Upgrades - S	0.6	1.3	1.9	2.6	3.2	3.2	3.2	3.2	3.2	3.2	2.6	1.9	1.3	0.6	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
11	CII	Landscape Conversion or Turf Removal - MF	3.1	6.2	9.2	12.3	15.4	15.4	15.4	15.4	15.4	15.4	12.3	9.2	6.2	3.1	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
12	Single Family	Soil Moisture Sensor Rebate	1.6	3.2	4.8	6.3	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	6.3	4.8	3.2	1.6	0.0
13	CII	Water Savings Incentive Program for CII	18.4	36.8	55.2	73.6	92.0	92.0	92.0	92.0	92.0	92.0	92.0	92.0	92.0	92.0	92.0	73.6	55.2	36.8	18.4	0.0	0.0	0.0	0.0	0.0	0.0
14	CII	Restaurant Spray Nozzle Rebates	6.7	13.5	20.2	26.9	33.6	26.9	20.2	13.5	6.7	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
	Total Gross V	Vater Savings	82.8	158.9	229.7	296.1	359.1	307.6	262.8	223.4	188.3	156.7	149.3	141.9	134.6	127.2	119.8	97.4	75.1	52.7	30.3	7.9	6.3	4.8	3.2	1.6	0.0

#### Active Water Savings (AF)

Activity ID	Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1	Single Family	High Efficiency Clothes Washer Rebate - Res	4.0	7.7	11.2	14.6	17.7	16.7	15.8	14.9	14.1	13.3	12.7	12.0	11.4	10.8	10.3	8.1	5.9	3.8	1.9	0.0	0.0	0.0	0.0	0.0	0.0
2	Single Family	Water Use Surveys/Audits - SFR	27.8	50.0	67.7	81.9	93.3	65.6	43.3	25.6	11.4	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
3	Single Family	Landscape Conversion or Turf Removal -SFF	1.0	2.1	3.1	4.1	5.1	5.1	5.1	5.1	5.1	5.1	4.1	3.1	2.1	1.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
4	Single Family	Smart Irrigation Controller (Weather-Based Ir	2.0	3.9	5.9	7.8	9.8	9.8	9.8	9.8	9.8	9.8	7.8	5.9	3.9	2.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
5	CII	Water Use Surveys/Audits - CII	12.3	22.2	30.1	36.4	41.5	29.1	19.3	11.4	5.1	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
6	Irrigation	Incentivize Irrigation Equipment Upgrades - L	2.3	4.7	7.0	9.3	11.6	11.6	11.6	11.6	11.6	11.6	9.3	7.0	4.7	2.3	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
7	Irrigation	Smart Irrigation Controller (Weather-Based Ir	1.8	3.6	5.3	7.1	8.9	8.9	8.9	8.9	8.9	8.9	7.1	5.3	3.6	1.8	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
8	Multi Family	Water Use Surveys/Audits - MFR	12.3	22.2	30.1	36.4	41.5	29.1	19.3	11.4	5.1	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
9	Single Family	High Efficiency Faucet Aerator / Showerhead	11.2	22.4	33.6	44.8	56.0	44.8	33.6	22.4	11.2	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
10	Single Family	Incentivize Irrigation Equipment Upgrades - S	0.6	1.3	1.9	2.6	3.2	3.2	3.2	3.2	3.2	3.2	2.6	1.9	1.3	0.6	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
11	CII	Landscape Conversion or Turf Removal - MF	3.1	6.2	9.2	12.3	15.4	15.4	15.4	15.4	15.4	15.4	12.3	9.2	6.2	3.1	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
12	Single Family	Soil Moisture Sensor Rebate	1.6	3.2	4.8	6.3	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	6.3	4.8	3.2	1.6	0.0
13	CII	Water Savings Incentive Program for CII	18.4	36.8	55.2	73.6	92.0	92.0	92.0	92.0	92.0	92.0	92.0	92.0	92.0	92.0	92.0	73.6	55.2	36.8	18.4	0.0	0.0	0.0	0.0	0.0	0.0
14	CII	Restaurant Spray Nozzle Rebates	6.7	13.5	20.2	26.9	33.6	26.9	20.2	13.5	6.7	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
	Total Active W	Vater Savings	105.1	199.5	285.3	364.2	437.6	366.2	305.4	253.0	207.5	167.3	155.8	144.4	133.0	121.6	110.3	89.6	69.0	48.6	28.2	7.9	6.3	4.8	3.2	1.6	0.0

#### Peak Active Water Savings (AF)

Activity I	Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1	Single Family	High Efficiency Clothes Washer Rebate - Res	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
2	Single Family	Water Use Surveys/Audits - SFR	18.9	34.0	46.1	55.7	63.4	44.6	29.5	17.4	7.7	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
3	Single Family	Landscape Conversion or Turf Removal -SFF	0.8	1.6	2.5	3.3	4.1	4.1	4.1	4.1	4.1	4.1	3.3	2.5	1.6	0.8	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
4	Single Family	Smart Irrigation Controller (Weather-Based Ir	1.4	2.7	4.1	5.5	6.9	6.9	6.9	6.9	6.9	6.9	5.5	4.1	2.7	1.4	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
5	CII	Water Use Surveys/Audits - CII	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
6	Irrigation	Incentivize Irrigation Equipment Upgrades - L	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
7	Irrigation	Smart Irrigation Controller (Weather-Based Ir	1.2	2.5	3.7	5.0	6.2	6.2	6.2	6.2	6.2	6.2	5.0	3.7	2.5	1.2	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
8	Multi Family	Water Use Surveys/Audits - MFR	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
9	Single Family	High Efficiency Faucet Aerator / Showerhead	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
10	Single Family	Incentivize Irrigation Equipment Upgrades - S	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
11	CII	Landscape Conversion or Turf Removal - MF	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
12	Single Family	Soil Moisture Sensor Rebate	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
13	CII	Water Savings Incentive Program for CII	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
14	CII	Restaurant Spray Nozzle Rebates	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
	<b>Total Active V</b>	Vater Savings	22.3	40.9	56.4	69.5	80.6	61.8	46.7	34.6	24.9	17.2	13.8	10.3	6.9	3.4	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0

#### Off Peak Active Water Savings (AF)

Activity I	D Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1	Single Family	High Efficiency Clothes Washer Rebate - Res	4.0	7.7	11.2	14.6	17.7	16.7	15.8	14.9	14.1	13.3	12.7	12.0	11.4	10.8	10.3	8.1	5.9	3.8	1.9	0.0	0.0	0.0	0.0	0.0	0.0
2	Single Family	Water Use Surveys/Audits - SFR	8.9	16.0	21.7	26.2	29.9	21.0	13.9	8.2	3.6	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
3	Single Family	Landscape Conversion or Turf Removal -SFF	0.2	0.4	0.6	0.8	1.0	1.0	1.0	1.0	1.0	1.0	0.8	0.6	0.4	0.2	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
4	Single Family	Smart Irrigation Controller (Weather-Based Ir	0.6	1.2	1.8	2.4	2.9	2.9	2.9	2.9	2.9	2.9	2.4	1.8	1.2	0.6	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
5	CII	Water Use Surveys/Audits - CII	12.3	22.2	30.1	36.4	41.5	29.1	19.3	11.4	5.1	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0

	Total Active V	Nater Savings	82.8	158.7	229.0	294.8	356.9	304.4	258.7	218.5	182.6	150.1	142.1	134.0	126.1	118.1	110.3	89.6	69.0	48.6	28.2	7.9	6.3	4.8	3.2	1.6	0.0
14	CII	Restaurant Spray Nozzle Rebates	6.7	13.5	20.2	26.9	33.6	26.9	20.2	13.5	6.7	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
13	CII	Water Savings Incentive Program for CII	18.4	36.8	55.2	73.6	92.0	92.0	92.0	92.0	92.0	92.0	92.0	92.0	92.0	92.0	92.0	73.6	55.2	36.8	18.4	0.0	0.0	0.0	0.0	0.0	0.0
12	Single Family	Soil Moisture Sensor Rebate	1.6	3.2	4.8	6.3	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	7.9	6.3	4.8	3.2	1.6	0.0
11	CII	Landscape Conversion or Turf Removal - MF	3.1	6.2	9.2	12.3	15.4	15.4	15.4	15.4	15.4	15.4	12.3	9.2	6.2	3.1	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
10	Single Family	Incentivize Irrigation Equipment Upgrades - S	0.6	1.3	1.9	2.6	3.2	3.2	3.2	3.2	3.2	3.2	2.6	1.9	1.3	0.6	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
9	Single Family	High Efficiency Faucet Aerator / Showerhead	11.2	22.4	33.6	44.8	56.0	44.8	33.6	22.4	11.2	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
8	Multi Family	Water Use Surveys/Audits - MFR	12.3	22.2	30.1	36.4	41.5	29.1	19.3	11.4	5.1	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
7	Irrigation	Smart Irrigation Controller (Weather-Based Ir	0.5	1.1	1.6	2.1	2.7	2.7	2.7	2.7	2.7	2.7	2.1	1.6	1.1	0.5	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
6	Irrigation	Incentivize Irrigation Equipment Upgrades - L	2.3	4.7	7.0	9.3	11.6	11.6	11.6	11.6	11.6	11.6	9.3	7.0	4.7	2.3	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0

#### Passive Water Savings (AF)

		-3-()																									
Activity I	D Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1	Single Family	High Efficiency Clothes Washer Rebate - Res	0.0	0.2	0.7	1.3	2.2	3.2	4.1	5.0	5.8	6.5	7.2	7.9	8.5	9.0	9.5	7.8	6.0	4.1	2.1	0.0	0.0	0.0	0.0	0.0	0.0
2	Single Family	Water Use Surveys/Audits - SFR	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
3	Single Family	Landscape Conversion or Turf Removal -SFF	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
4	Single Family	Smart Irrigation Controller (Weather-Based Ir	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
5	CII	Water Use Surveys/Audits - CII	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
6	Irrigation	Incentivize Irrigation Equipment Upgrades - L	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
7	Irrigation	Smart Irrigation Controller (Weather-Based Ir	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
8	Multi Family	Water Use Surveys/Audits - MFR	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
9	Single Family	High Efficiency Faucet Aerator / Showerhead	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
10	Single Family	Incentivize Irrigation Equipment Upgrades - S	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
11	CII	Landscape Conversion or Turf Removal - MF	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
12	Single Family	Soil Moisture Sensor Rebate	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
13	CII	Water Savings Incentive Program for CII	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
14	CII	Restaurant Spray Nozzle Rebates	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
	Total Passiv	e Water Savings	0.0	0.2	0.7	1.3	2.2	3.2	4.1	5.0	5.8	6.5	7.2	7.9	8.5	9.0	9.5	7.8	6.0	4.1	2.1	0.0	0.0	0.0	0.0	0.0	0.0

#### Peak Passive Water Savings (AF)

Activity ID	) Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1	Single Family	High Efficiency Clothes Washer Rebate - Res	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
2	Single Family	Water Use Surveys/Audits - SFR	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
3	Single Family	Landscape Conversion or Turf Removal -SFF	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
4	Single Family	Smart Irrigation Controller (Weather-Based Ir	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
5	CII	Water Use Surveys/Audits - CII	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
6	Irrigation	Incentivize Irrigation Equipment Upgrades - L	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
7	Irrigation	Smart Irrigation Controller (Weather-Based Ir	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
8	Multi Family	Water Use Surveys/Audits - MFR	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
9	Single Family	High Efficiency Faucet Aerator / Showerhead	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
10	Single Family	Incentivize Irrigation Equipment Upgrades - S	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
11	CII	Landscape Conversion or Turf Removal - MF	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
12	Single Family	Soil Moisture Sensor Rebate	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
13	CII	Water Savings Incentive Program for CII	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
14	CII	Restaurant Spray Nozzle Rebates	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
	Total Passive	Water Savings	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0

#### Off Peak Passive Water Savings (AF)

Activity ID	Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1	Single Family	High Efficiency Clothes Washer Rebate - Res	0.0	0.2	0.7	1.3	2.2	3.2	4.1	5.0	5.8	6.5	7.2	7.9	8.5	9.0	9.5	7.8	6.0	4.1	2.1	0.0	0.0	0.0	0.0	0.0	0.0
2	Single Family	Water Use Surveys/Audits - SFR	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
3	Single Family	Landscape Conversion or Turf Removal -SFF	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
4	Single Family	Smart Irrigation Controller (Weather-Based Ir	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
5	CII	Water Use Surveys/Audits - CII	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
6	Irrigation	Incentivize Irrigation Equipment Upgrades - L	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
7	Irrigation	Smart Irrigation Controller (Weather-Based Ir	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
8	Multi Family	Water Use Surveys/Audits - MFR	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
9	Single Family	High Efficiency Faucet Aerator / Showerhead	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
10	Single Family	Incentivize Irrigation Equipment Upgrades - S	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
11	CII	Landscape Conversion or Turf Removal - MF	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
12	Single Family	Soil Moisture Sensor Rebate	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
13	CII	Water Savings Incentive Program for CII	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
14	CII	Restaurant Spray Nozzle Rebates	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
	Total Passive	Water Savings	0.0	0.2	0.7	1.3	2.2	3.2	4.1	5.0	5.8	6.5	7.2	7.9	8.5	9.0	9.5	7.8	6.0	4.1	2.1	0.0	0.0	0.0	0.0	0.0	0.0

#### Customer Water Bill Savings (2019 dollars)

Activity II	) Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1	Single Family	High Efficiency Clothes Washer Rebate - Res	\$14,959	\$29,331	\$43,158	\$56,479	\$69,331	\$66,034	\$62,953	\$60,074	\$57,384	\$54,872	\$52,527	\$50,339	\$48,297	\$46,393	\$44,619	\$35,149	\$25,975	\$17,073	\$8,422	\$0	\$0	\$0	\$0	\$0	\$0
2	Single Family	Water Use Surveys/Audits - SFR	\$104,427	\$189,830	\$259,873	\$317,516	\$365,152	\$259,067	\$173,003	\$103,111	\$46,281	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
3	Single Family	Landscape Conversion or Turf Removal -SFF	\$3,859	\$7,795	\$11,809	\$15,901	\$20,073	\$20,271	\$20,472	\$20,675	\$20,879	\$21,086	\$17,036	\$12,903	\$8,687	\$4,387	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
4	Single Family	Smart Irrigation Controller (Weather-Based Ir	\$7,372	\$14,891	\$22,558	\$30,375	\$38,344	\$38,724	\$39,107	\$39,494	\$39,885	\$40,280	\$32,543	\$24,649	\$16,595	\$8,380	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
5	CII	Water Use Surveys/Audits - CII	\$51,573	\$93,751	\$128,343	\$156,812	\$180,337	\$127,945	\$85,441	\$50,923	\$22,857	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
6	Irrigation	Incentivize Irrigation Equipment Upgrades - L	\$10,791	\$21,796	\$33,018	\$44,460	\$56,125	\$56,681	\$57,242	\$57,809	\$58,381	\$58,959	\$47,634	\$36,079	\$24,291	\$12,266	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0

	Total Wastewa	ater Savings	\$402.839	\$774.545	\$1.121.860	\$1.450.248	\$1.764.141 \$	1.505.045	\$1.281.090	\$1.085.265	\$911.901	\$756.411	\$710.750	\$664.320	\$617.096	\$569.055	\$520.175	\$426.275	\$330.953	\$234,159	\$135.846	\$35.965	\$29.057	\$22.008	\$14.818	\$7.482	\$0
14	CII	Restaurant Spray Nozzle Rebates	\$28,105	\$56,767	\$85,994	\$115,794	\$146,175	\$118,098	\$89,450	\$60,224	\$30,410	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
13	CII	Water Savings Incentive Program for CII	\$76,893	\$155,308	\$235,269	\$316,797	\$399,916	\$403,876	\$407,874	\$411,912	\$415,990	\$420,108	\$424,267	\$428,468	\$432,709	\$436,993	\$441,319	\$356,551	\$270,060	\$181,823	\$91,811	\$0	\$0	\$0	\$0	\$0	\$0
12	Single Family	Soil Moisture Sensor Rebate	\$5,965	\$12,048	\$18,252	\$24,576	\$31,025	\$31,332	\$31,642	\$31,955	\$32,271	\$32,591	\$32,914	\$33,239	\$33,569	\$33,901	\$34,236	\$34,575	\$34,918	\$35,263	\$35,613	\$35,965	\$29,057	\$22,008	\$14,818	\$7,482	\$0
11	CII	Landscape Conversion or Turf Removal - MF	\$12,861	\$25,976	\$39,350	\$52,986	\$66,889	\$67,551	\$68,220	\$68,895	\$69,577	\$70,266	\$56,769	\$42,998	\$28,949	\$14,618	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
10	Single Family	Incentivize Irrigation Equipment Upgrades - S	\$2,409	\$4,865	\$7,370	\$9,924	\$12,528	\$12,652	\$12,777	\$12,904	\$13,031	\$13,160	\$10,633	\$8,053	\$5,422	\$2,738	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
9	Single Family	High Efficiency Faucet Aerator / Showerhead	\$42,116	\$85,067	\$128,863	\$173,518	\$219,045	\$176,971	\$134,042	\$90,246	\$45,570	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
8	Multi Family	Water Use Surveys/Audits - MFR	\$33,254	\$60,450	\$82,754	\$101,110	\$116,280	\$82,498	\$55,091	\$32,835	\$14,738	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
7	Irrigation	Smart Irrigation Controller (Weather-Based Ir	\$8,252	\$16,668	\$25,250	\$34,000	\$42,921	\$43,346	\$43,775	\$44,208	\$44,646	\$45,088	\$36,427	\$27,591	\$18,576	\$9,380	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0

#### Wastewater Savings (AF)

Activity ID	) Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1	Single Family	High Efficiency Clothes Washer Rebate - Res	4.6	9.0	13.2	17.0	20.7	19.5	18.4	17.4	16.5	15.6	14.8	14.0	13.3	12.7	12.1	9.4	6.9	4.5	2.2	0.0	0.0	0.0	0.0	0.0	0.0
2	Single Family	Water Use Surveys/Audits - SFR	11.1	20.0	27.1	32.8	37.3	26.2	17.3	10.2	4.5	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
3	Single Family	Landscape Conversion or Turf Removal -SFF	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
4	Single Family	Smart Irrigation Controller (Weather-Based Ir	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
5	CII	Water Use Surveys/Audits - CII	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
6	Irrigation	Incentivize Irrigation Equipment Upgrades - L	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
7	Irrigation	Smart Irrigation Controller (Weather-Based Ir	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
8	Multi Family	Water Use Surveys/Audits - MFR	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
9	Single Family	High Efficiency Faucet Aerator / Showerhead	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
10	Single Family	Incentivize Irrigation Equipment Upgrades - S	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
11	CII	Landscape Conversion or Turf Removal - MF	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
12	Single Family	Soil Moisture Sensor Rebate	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
13	CII	Water Savings Incentive Program for CII	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
14	CII	Restaurant Spray Nozzle Rebates	6.7	13.5	20.2	26.9	33.6	26.9	20.2	13.5	6.7	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
	Total Wastew	vater Savings	22.5	42.5	60.4	76.7	91.7	72.7	56.0	41.1	27.8	15.6	14.8	14.0	13.3	12.7	12.1	9.4	6.9	4.5	2.2	0.0	0.0	0.0	0.0	0.0	0.0

#### Customer Sewer Bill Savings (2019 dollars)

Activity ID	) Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1	Single Family	High Efficiency Clothes Washer Rebate - Res	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
2	Single Family	Water Use Surveys/Audits - SFR	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
3	Single Family	Landscape Conversion or Turf Removal -SFF	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
4	Single Family	Smart Irrigation Controller (Weather-Based Ir	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
5	CII	Water Use Surveys/Audits - CII	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
6	Irrigation	Incentivize Irrigation Equipment Upgrades - L	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
7	Irrigation	Smart Irrigation Controller (Weather-Based Ir	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
8	Multi Family	Water Use Surveys/Audits - MFR	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
9	Single Family	High Efficiency Faucet Aerator / Showerhead	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
10	Single Family	Incentivize Irrigation Equipment Upgrades - S	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
11	CII	Landscape Conversion or Turf Removal - MF	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
12	Single Family	Soil Moisture Sensor Rebate	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
13	CII	Water Savings Incentive Program for CII	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
14	CII	Restaurant Spray Nozzle Rebates	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
	Total Wastew	ater Savings	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0

#### Customer Electricity Savings (KWh)

Activity ID	) Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1	Single Family	High Efficiency Clothes Washer Rebate - Res	4,664.3	9,055.8	13,194.2	17,097.4	20,782.2	19,600.1	18,502.4	17,483.1	16,536.6	15,657.7	14,841.6	14,083.8	13,380.1	12,726.7	12,120.0	9,454.0	6,917.9	4,502.5	2,199.2	0.0	0.0	0.0	0.0	0.0	0.0
2	Single Family	Water Use Surveys/Audits - SFR	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
3	Single Family	Landscape Conversion or Turf Removal -SFF	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
4	Single Family	Smart Irrigation Controller (Weather-Based Ir	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
5	CII	Water Use Surveys/Audits - CII	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
6	Irrigation	Incentivize Irrigation Equipment Upgrades - L	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
7	Irrigation	Smart Irrigation Controller (Weather-Based Ir	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
8	Multi Family	Water Use Surveys/Audits - MFR	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
9	Single Family	High Efficiency Faucet Aerator / Showerhead	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
10	Single Family	Incentivize Irrigation Equipment Upgrades - S	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
11	CII	Landscape Conversion or Turf Removal - MF	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
12	Single Family	Soil Moisture Sensor Rebate	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
13	CII	Water Savings Incentive Program for CII	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
14	CII	Restaurant Spray Nozzle Rebates	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
	Total Electric	ity Savings	4,664.3	9,055.8	13,194.2	17,097.4	20,782.2	19,600.1	18,502.4	17,483.1	16,536.6	15,657.7	14,841.6	14,083.8	13,380.1	12,726.7	12,120.0	9,454.0	6,917.9	4,502.5	2,199.2	0.0	0.0	0.0	0.0	0.0	0.0

#### Customer Electricity Bill Savings (2019 dollars)

Activity	ID Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1	Single Famil	<ul> <li>High Efficiency Clothes Washer Rebate - Res</li> </ul>	\$1,298	\$2,519	\$3,670	\$4,756	\$5,781	\$5,452	\$5,147	\$4,863	\$4,600	\$4,356	\$4,129	\$3,918	\$3,722	\$3,540	\$3,372	\$2,630	\$1,924	\$1,253	\$612	\$0	\$0	\$0	\$0	\$0	\$0
2	Single Family	<ul> <li>Water Use Surveys/Audits - SFR</li> </ul>	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
3	Single Famil	<ul> <li>Landscape Conversion or Turf Removal -SFF</li> </ul>	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
4	Single Famil	Smart Irrigation Controller (Weather-Based Ir	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
5	CII	Water Use Surveys/Audits - CII	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
6	Irrigation	Incentivize Irrigation Equipment Upgrades - L	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
7	Irrigation	Smart Irrigation Controller (Weather-Based Ir	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
8	Multi Family	Water Use Surveys/Audits - MFR	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0

9	Single Family	High Efficiency Faucet Aerator / Showerhead	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
10	Single Family	Incentivize Irrigation Equipment Upgrades - S	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
11	CII	Landscape Conversion or Turf Removal - MF	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
12	Single Family	Soil Moisture Sensor Rebate	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
13	CII	Water Savings Incentive Program for CII	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
14	CII	Restaurant Spray Nozzle Rebates	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
	Total Electric	ity Savings	\$1,298	\$2,519	\$3,670	\$4,756	\$5,781	\$5,452	\$5,147	\$4,863	\$4,600	\$4,356	\$4,129	\$3,918	\$3,722	\$3,540	\$3,372	\$2,630	\$1,924	\$1,253	\$612	\$0	\$0	\$0	\$0	\$0	\$0

#### Customer Gas Savings (Therms)

Activity ID	Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1	Single Family	High Efficiency Clothes Washer Rebate - Res	4,535	8,804	12,828	16,622	20,205	19,056	17,988	16,997	16,077	15,223	14,429	13,693	13,008	12,373	11,783	9,191	6,726	4,377	2,138	0	0	0	0	0	0
2	Single Family	Water Use Surveys/Audits - SFR	8,661	15,590	21,133	25,568	29,115	20,454	13,525	7,982	3,548	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
3	Single Family	Landscape Conversion or Turf Removal -SFF	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
4	Single Family	Smart Irrigation Controller (Weather-Based Ir	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
5	CII	Water Use Surveys/Audits - CII	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
6	Irrigation	Incentivize Irrigation Equipment Upgrades - L	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
7	Irrigation	Smart Irrigation Controller (Weather-Based Ir	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
8	Multi Family	Water Use Surveys/Audits - MFR	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
9	Single Family	High Efficiency Faucet Aerator / Showerhead	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
10	Single Family	Incentivize Irrigation Equipment Upgrades - S	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
11	CII	Landscape Conversion or Turf Removal - MF	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
12	Single Family	Soil Moisture Sensor Rebate	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
13	CII	Water Savings Incentive Program for CII	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
14	CII	Restaurant Spray Nozzle Rebates	18,263	36,525	54,788	73,050	91,313	73,050	54,788	36,525	18,263	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
	Total Gas Sav	rings	31,458	60,919	88,748	115,240	140,633	112,560	86,301	61,505	37,887	15,223	14,429	13,693	13,008	12,373	11,783	9,191	6,726	4,377	2,138	0	0	0	0	0	0

#### Customer Gas Savings (2019 dollars)

ouston																											
Activity	ID Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1	Single Famil	High Efficiency Clothes Washer Rebate - Res	\$9,079	\$17,627	\$25,682	\$33,279	\$40,452	\$38,151	\$36,014	\$34,030	\$32,188	\$30,477	\$28,889	\$27,414	\$26,044	\$24,772	\$23,591	\$18,402	\$13,465	\$8,764	\$4,281	\$0	\$0	\$0	\$0	\$0	\$0
2	Single Famil	Water Use Surveys/Audits - SFR	\$17,340	\$31,212	\$42,310	\$51,188	\$58,291	\$40,951	\$27,079	\$15,981	\$7,103	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
3	Single Famil	/ Landscape Conversion or Turf Removal -SFF	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
4	Single Famil	Smart Irrigation Controller (Weather-Based Ir	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
5	CII	Water Use Surveys/Audits - CII	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
6	Irrigation	Incentivize Irrigation Equipment Upgrades - L	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
7	Irrigation	Smart Irrigation Controller (Weather-Based Ir	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
8	Multi Family	Water Use Surveys/Audits - MFR	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
9	Single Famil	High Efficiency Faucet Aerator / Showerhead	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
10	Single Famil	/ Incentivize Irrigation Equipment Upgrades - S	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
11	CII	Landscape Conversion or Turf Removal - MF	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
12	Single Famil	Soil Moisture Sensor Rebate	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
13	CII	Water Savings Incentive Program for CII	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
14	CII	Restaurant Spray Nozzle Rebates	\$14,683	\$29,365	\$44,048	\$58,730	\$73,413	\$58,730	\$44,048	\$29,365	\$14,683	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
	Total Gas S	avings	\$41,102	\$78,204	\$112,040	\$143,198	\$172,155	\$137,832	\$107,140	\$79,376	\$53,973	\$30,477	\$28,889	\$27,414	\$26,044	\$24,772	\$23,591	\$18,402	\$13,465	\$8,764	\$4,281	\$0	\$0	\$0	\$0	\$0	\$0

#### User Entered Utility Avoided Water System Cost (2019 dollars)

			,																								
Activity	ID Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1	Single Family	High Efficiency Clothes Washer Rebate - Res	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
2	Single Family	Water Use Surveys/Audits - SFR	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
3	Single Family	Landscape Conversion or Turf Removal -SFF	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
4	Single Family	Smart Irrigation Controller (Weather-Based Ir	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
5	CII	Water Use Surveys/Audits - CII	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
6	Irrigation	Incentivize Irrigation Equipment Upgrades - L	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
7	Irrigation	Smart Irrigation Controller (Weather-Based Ir	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
8	Multi Family	Water Use Surveys/Audits - MFR	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
9	Single Family	High Efficiency Faucet Aerator / Showerhead	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
10	Single Family	Incentivize Irrigation Equipment Upgrades - S	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
11	CII	Landscape Conversion or Turf Removal - MF	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
12	Single Family	Soil Moisture Sensor Rebate	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
13	CII	Water Savings Incentive Program for CII	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
14	CII	Restaurant Spray Nozzle Rebates	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
	Total Avoided	d Cost	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0

#### User Entered Utility Avoided Wastewater System Cost (2019 dollars)

			/																								
Activity	D Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1	Single Family	/ High Efficiency Clothes Washer Rebate - Res	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
2	Single Family	/ Water Use Surveys/Audits - SFR	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
3	Single Famil	/ Landscape Conversion or Turf Removal -SFF	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
4	Single Family	/ Smart Irrigation Controller (Weather-Based Ir	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
5	CII	Water Use Surveys/Audits - CII	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
6	Irrigation	Incentivize Irrigation Equipment Upgrades - L	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
7	Irrigation	Smart Irrigation Controller (Weather-Based Ir	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
8	Multi Family	Water Use Surveys/Audits - MFR	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
9	Single Famil	/ High Efficiency Faucet Aerator / Showerhead	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
10	Single Famil	Incentivize Irrigation Equipment Upgrades - S	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0

11	CII	Landscape Conversion or Turf Removal - MF	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
12	Single Family	y Soil Moisture Sensor Rebate	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
13	CII	Water Savings Incentive Program for CII	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
14	CII	Restaurant Spray Nozzle Rebates	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
	Total Avoide	ed Cost	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0

#### User Entered Other Utility Avoided Cost (2019 dollars)

Ober Linter																											
Activity ID C	Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1 S	Single Family	High Efficiency Clothes Washer Rebate - Res	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
2 S	Single Family	Water Use Surveys/Audits - SFR	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
3 5	Single Family	Landscape Conversion or Turf Removal -SFF	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
4 S	Single Family	Smart Irrigation Controller (Weather-Based Ir	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
5 C	CII	Water Use Surveys/Audits - CII	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
6 li	rrigation	Incentivize Irrigation Equipment Upgrades - L	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
7 li	rrigation	Smart Irrigation Controller (Weather-Based Ir	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
8 N	∕lulti Family	Water Use Surveys/Audits - MFR	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
9 S	Single Family	High Efficiency Faucet Aerator / Showerhead	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
10 S	Single Family	Incentivize Irrigation Equipment Upgrades - S	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
11 C	CII	Landscape Conversion or Turf Removal - MF	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
12 S	Single Family	Soil Moisture Sensor Rebate	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
13 C	CII	Water Savings Incentive Program for CII	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
14 C		Restaurant Spray Nozzle Rebates	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
Т	otal Avoided	d Cost	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0

#### Model Calculator Utility Water System Avoided Cost (2019 dollars)

mouel o		inty water bystern Avolueu bost (2015	uonaraj																								
Activity II	) Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1	Single Family	High Efficiency Clothes Washer Rebate - Res	\$6,330	\$12,508	\$18,541	\$24,436	\$30,202	\$28,997	\$27,858	\$26,781	\$25,764	\$24,805	\$23,936	\$23,116	\$22,343	\$21,615	\$20,931	\$16,621	\$12,378	\$8,196	\$4,072	\$0	\$0	\$0	\$0	\$0	\$0
2	Single Family	Water Use Surveys/Audits - SFR	\$44,190	\$80,950	\$111,641	\$137,377	\$159,068	\$113,763	\$76,557	\$45,967	\$20,779	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
3	Single Family	Landscape Conversion or Turf Removal -SFF	\$1,633	\$3,324	\$5,073	\$6,880	\$8,744	\$8,902	\$9,059	\$9,217	\$9,374	\$9,532	\$7,763	\$5,925	\$4,019	\$2,044	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
4	Single Family	Smart Irrigation Controller (Weather-Based Ir	\$3,120	\$6,350	\$9,691	\$13,142	\$16,703	\$17,004	\$17,306	\$17,607	\$17,908	\$18,209	\$14,829	\$11,319	\$7,677	\$3,904	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
5	CII	Water Use Surveys/Audits - CII	\$19,648	\$35,992	\$49,638	\$61,080	\$70,724	\$50,581	\$34,038	\$20,438	\$9,239	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
6	Irrigation	Incentivize Irrigation Equipment Upgrades - L	\$3,705	\$7,541	\$11,508	\$15,606	\$19,835	\$20,193	\$20,550	\$20,908	\$21,265	\$21,622	\$17,610	\$13,441	\$9,117	\$4,636	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
7	Irrigation	Smart Irrigation Controller (Weather-Based Ir	\$2,833	\$5,767	\$8,800	\$11,934	\$15,169	\$15,442	\$15,715	\$15,989	\$16,262	\$16,535	\$13,467	\$10,279	\$6,972	\$3,545	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
8	Multi Family	Water Use Surveys/Audits - MFR	\$19,648	\$35,992	\$49,638	\$61,080	\$70,724	\$50,581	\$34,038	\$20,438	\$9,239	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
9	Single Family	High Efficiency Faucet Aerator / Showerhead	\$17,822	\$36,275	\$55,360	\$75,075	\$95,421	\$77,712	\$59,316	\$40,232	\$20,460	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
10	Single Family	Incentivize Irrigation Equipment Upgrades - S	\$1,019	\$2,075	\$3,166	\$4,294	\$5,457	\$5,556	\$5,654	\$5,752	\$5,851	\$5,949	\$4,845	\$3,698	\$2,508	\$1,276	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
11	CII	Landscape Conversion or Turf Removal - MF	\$4,900	\$9,973	\$15,219	\$20,639	\$26,232	\$26,705	\$27,178	\$27,650	\$28,123	\$28,596	\$23,289	\$17,776	\$12,057	\$6,131	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
12	Single Family	Soil Moisture Sensor Rebate	\$2,524	\$5,138	\$7,841	\$10,633	\$13,515	\$13,759	\$14,002	\$14,246	\$14,489	\$14,733	\$14,998	\$15,264	\$15,529	\$15,795	\$16,060	\$16,350	\$16,639	\$16,929	\$17,218	\$17,507	\$14,258	\$10,883	\$7,382	\$3,754	\$0
13	CII	Water Savings Incentive Program for CII	\$29,293	\$59,624	\$90,992	\$123,396	\$156,838	\$159,665	\$162,491	\$165,318	\$168,144	\$170,970	\$174,052	\$177,133	\$180,214	\$183,295	\$186,376	\$151,788	\$115,856	\$78,581	\$39,962	\$0	\$0	\$0	\$0	\$0	\$0
14	CII	Restaurant Spray Nozzle Rebates	\$10,707	\$21,793	\$33,259	\$45,103	\$57,327	\$46,688	\$35,636	\$24,170	\$12,292	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
	Total Avoided	Cost	\$167,372	\$323,301	\$470,364	\$610,674	\$745,960	\$635,546	\$539,399	\$454,711	\$379,189	\$310,952	\$294,789	\$277,951	\$260,436	\$242,242	\$223,367	\$184,758	\$144,873	\$103,706	\$61,252	\$17,507	\$14,258	\$10,883	\$7,382	\$3,754	\$0

#### Model Calculator Utility Wastewater System Avoided Cost (2019 dollars)

				-,																							
Activity ID	Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1	Single Family	<ul> <li>High Efficiency Clothes Washer Rebate - Res</li> </ul>	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
2	Single Family	Water Use Surveys/Audits - SFR	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
3	Single Family	<ul> <li>Landscape Conversion or Turf Removal -SFF</li> </ul>	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
4	Single Family	Smart Irrigation Controller (Weather-Based Ir	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
5	CII	Water Use Surveys/Audits - CII	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
6	Irrigation	Incentivize Irrigation Equipment Upgrades - L	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
7	Irrigation	Smart Irrigation Controller (Weather-Based Ir	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
8	Multi Family	Water Use Surveys/Audits - MFR	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
9	Single Family	High Efficiency Faucet Aerator / Showerhead	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
10	Single Family	Incentivize Irrigation Equipment Upgrades - S	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
11	CII	Landscape Conversion or Turf Removal - MF	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
12	Single Family	Soil Moisture Sensor Rebate	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
13	CII	Water Savings Incentive Program for CII	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
14	CII	Restaurant Spray Nozzle Rebates	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
	Total Avoide	d Cost	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0

#### Total Avoided Water and Wastewater Production Cost (2019 dollars)

		· · · · · · · · · · · · · · · · · · ·	,																								
Activity ID	Class	Activity Name	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031	2032	2033	2034	2035	2036	2037	2038	2039	2040	2041	2042	2043	2044	2045
1	Single Family	High Efficiency Clothes Washer Rebate - Res	\$6,330	\$12,508	\$18,541	\$24,436	\$30,202	\$28,997	\$27,858	\$26,781	\$25,764	\$24,805	\$23,936	\$23,116	\$22,343	\$21,615	\$20,931	\$16,621	\$12,378	\$8,196	\$4,072	\$0	\$0	\$0	\$0	\$0	\$0
2	Single Family	Water Use Surveys/Audits - SFR	\$44,190	\$80,950	\$111,641	\$137,377	\$159,068	\$113,763	\$76,557	\$45,967	\$20,779	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
3	Single Family	Landscape Conversion or Turf Removal -SFF	\$1,633	\$3,324	\$5,073	\$6,880	\$8,744	\$8,902	\$9,059	\$9,217	\$9,374	\$9,532	\$7,763	\$5,925	\$4,019	\$2,044	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
4	Single Family	Smart Irrigation Controller (Weather-Based Ir	\$3,120	\$6,350	\$9,691	\$13,142	\$16,703	\$17,004	\$17,306	\$17,607	\$17,908	\$18,209	\$14,829	\$11,319	\$7,677	\$3,904	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
5	CII	Water Use Surveys/Audits - CII	\$19,648	\$35,992	\$49,638	\$61,080	\$70,724	\$50,581	\$34,038	\$20,438	\$9,239	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
6	Irrigation	Incentivize Irrigation Equipment Upgrades - L	\$3,705	\$7,541	\$11,508	\$15,606	\$19,835	\$20,193	\$20,550	\$20,908	\$21,265	\$21,622	\$17,610	\$13,441	\$9,117	\$4,636	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
7	Irrigation	Smart Irrigation Controller (Weather-Based Ir	\$2,833	\$5,767	\$8,800	\$11,934	\$15,169	\$15,442	\$15,715	\$15,989	\$16,262	\$16,535	\$13,467	\$10,279	\$6,972	\$3,545	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
8	Multi Family	Water Use Surveys/Audits - MFR	\$19,648	\$35,992	\$49,638	\$61,080	\$70,724	\$50,581	\$34,038	\$20,438	\$9,239	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
9	Single Family	High Efficiency Faucet Aerator / Showerhead	\$17,822	\$36,275	\$55,360	\$75,075	\$95,421	\$77,712	\$59,316	\$40,232	\$20,460	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
10	Single Family	Incentivize Irrigation Equipment Upgrades - S	\$1,019	\$2,075	\$3,166	\$4,294	\$5,457	\$5,556	\$5,654	\$5,752	\$5,851	\$5,949	\$4,845	\$3,698	\$2,508	\$1,276	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
11	CII	Landscape Conversion or Turf Removal - MF	\$4,900	\$9,973	\$15,219	\$20,639	\$26,232	\$26,705	\$27,178	\$27,650	\$28,123	\$28,596	\$23,289	\$17,776	\$12,057	\$6,131	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
12	Single Family	Soil Moisture Sensor Rebate	\$2,524	\$5,138	\$7,841	\$10,633	\$13,515	\$13,759	\$14,002	\$14,246	\$14,489	\$14,733	\$14,998	\$15,264	\$15,529	\$15,795	\$16,060	\$16,350	\$16,639	\$16,929	\$17,218	\$17,507	\$14,258	\$10,883	\$7,382	\$3,754	\$0

13	CII	Water Savings Incentive Program for CII	\$29,293	\$59,624	\$90,992	\$123,396	\$156,838	\$159,665	\$162,491	\$165,318	\$168,144	\$170,970	\$174,052	\$177,133	\$180,214	\$183,295	\$186,376	\$151,788	\$115,856	\$78,581	\$39,962	\$0	\$0	\$0	\$0	\$0	\$0
14	CII	Restaurant Spray Nozzle Rebates	\$10,707	\$21,793	\$33,259	\$45,103	\$57,327	\$46,688	\$35,636	\$24,170	\$12,292	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0
	Total Avoid	ded Cost	\$167,372	\$323,301	\$470,364	\$610,674	\$745,960	\$635,546	\$539,399	\$454,711	\$379,189	\$310,952	\$294,789	\$277,951	\$260,436	\$242,242	\$223,367	\$184,758	\$144,873	\$103,706	\$61,252	\$17,507	\$14,258	\$10,883	\$7,382	\$3,754	\$0



# Appendix C

Methodology for Water Conservation Program Savings Analyses



## Appendix C

## Methodology for Water Conservation Program Savings Analyses

This Appendix describes the methodology used to estimate water conservation program savings based on customer billing data, for the analyses presented in report Section 5.3.2, *Estimated Water Savings for Selected Programs Based on Customer Billing Data*.

Water use savings associated with conservation programs are typically estimated based on literature values, which may or may not accurately capture the specific ways customers in a specific area (i.e., Marin Municipal Water District [District]) use water. Therefore, District customer billing data were analyzed in order to assess the amount of water typically saved through implementation of each of the four selected conservation programs. Water use by program participants was compared to water use by a representative cohort over the same time period, that was stratified based on key criteria. Water use savings were estimated for the four conservation programs identified below:<sup>1</sup>

- 1. Advanced Metering Infrastructure (AMI) Leak Notifications Program
- 2. Residential High Efficiency Clothes Washer (HECW) Rebate Program
- 3. SFR Water Use Surveys/Audits Program
- 4. SFR Weather-Based Irrigation Controller (WBIC) Rebate Program

Specifically, water use before and after implementation of a given action (e.g., device replacement or turf removal) by program participants is compared to the water use by a cohort of accounts who have not participated in the same or other programs in the given time frame. The incremental volume of water saved by program participants compared to that of the cohort group can then be attributed to program participation, as other factors have been normalized. This analytical technique is a version of the "Difference-in-Differences Estimation" method. The Difference-in-Differences Estimation method is a standard method used in economics and social science for quantitatively evaluating observational study data by studying the differential effect of a treatment, or in this case participation in a given program as compared to a "control group," when a true controlled experiment cannot be performed (Columbia Public Health, 2013).

By comparing water use over time by program participants to a cohort group and identifying the incremental change in water use due to program participation, this methodology controls for variations in water use due to climatic, economic, and other temporally related factors. By stratifying (or weighting) the cohort group based on key factors (i.e., Census Block Group or neighborhood), this method also effectively controls for geographic-linked water use influencing factors, such as house and yard size, housing age, general socio-economic factors, general landscape management factors, etc.

<u>Participant Sample Groups</u>: In order to estimate water savings attributable to a single conservation program, participant sample groups for this analysis were limited to accounts that participated in only one program, and who participated in that program in only one year (e.g., did not receive several rebates from the same program over several years), except as indicated in savings results tables, as appropriate. The

<sup>&</sup>lt;sup>1</sup> This analysis was also performed for the Rain Barrel Rebate Program. However, due to the limited sample size and likely other human factors, the results were not considered robust and thus are not presented herein.

Appendix C December 2020



participant sample groups were further limited to just those accounts that had active water use over the study period. Active accounts were identified as those who received six water bills and had non-zero water use in a given year.

<u>Comparison Cohort Sample Groups</u>: Accounts included in the cohort groups are limited to those accounts that had not participated in any program based on available data and that meet the same active account thresholds as described above for the participant sample groups (i.e., received six bills per year and nonzero annual water use). It is possible that members of the cohort group participated in a program that was not included in this study; however, given the large number of accounts included in these cohort groups the effect of participation in other programs would be expected to be minimal. Although not participants in a specific program, a portion of the cohort group members would be expected to have changed out water using devices with more efficient ones through natural replacement. Given this, the program savings identified by this method may actually be somewhat higher than estimated herein, resulting in a more conservative program savings estimate.

<u>Study Periods</u>: Since account-level water use billing data are available from 2004 to 2019, the participation data from 2010 to 2018 are analyzed so that one to three years of water use data can be used to capture the average water use before and after the participation year. The AMI Leak Notifications Program started in 2018, and thus even though there is only one year of billing data (2019) available to represent the average water use after the participation year, participation year 2018 is included.

<u>Stratification</u>: The water savings calculations for SFR accounts were stratified (or weighted) based on the Census Block Group (except as indicated in savings results table notes, as appropriate), as a way to control for geographically linked variables such as house and yard size, housing age, and general socio-economic factors, among others.

<u>Water Savings Calculation</u>: For each active account, the average annual water use for a period of three years prior to program participation is compared to the average annual water use in the year following program participation, dependent on available data. The change in water use by program participants is then compared to that of the cohort group over the same time period. The difference between the change in water use of the participants and the change in water use of the cohorts is the water savings due to the given conservation program. A positive average water savings suggests the program resulted in water.

### References

Columbia Public Health, 2013. Difference in Difference Estimation. Columbia Public Health, <u>https://www.publichealth.columbia.edu/research/population-health-methods/difference-difference-estimation#Overview</u>, accessed 28 September 2020.



# Appendix D

Prioritization and Screening of Future Water Conservation Measures

# Prioritization and Screening of Future Water Conservation Measures

Marin-Sonoma Saving Water Partnership

INSTRUCTIONS: Please review and complete the orange highlighted cells using the provided dropdown lists. Comments and clarifications may be added to the comments column on the right. You may use the filter features to help navigate this list. "Preference for Implementation" need only be completed when a program is given a priorty of 3 or greater. See READ ME tab for additional information.

Conservation Measure/Program	Туре	Indoor / Outdoor	Primary End Use	
Agency Actions and Water Rates				
Customer Water Loss Reduction (AMI Leak Detection)	Agency action	Both	Water Loss	All
Increase Enforcement of Customer Pressure Reducing Valve (PRV) Requirement	Agency action	Both	Water loss; Irrigation	All
Increase Enforcement of Indoor Fixture Retrofit at Time of Sale	Agency action	Indoor	Toilet, Urinal, Faucet,	All
Increase Enforcement of State Water Waste Regulations	Agency action	Outdoor	Irrigation	All
Install ANAL for Evisting Assounts	Agongy action	Poth	Water Locc	
Install AMI for High Water Users and Large Landscape Accounts		Outdoor	Water Loss	
Install AMI in New Development	Agency action	Both	Water Loss	
Rate Structure Evaluation	Agency action	Both	All	All
Regional UHET and/or Urinal Bulk Purchase Program	Agency action	Indoor	Toilet / Urinal	All
Water Budgeting/Monitoring for Large Landscape Accounts	Agency action	Both		IRR
Establish Separate Pricing Structure for Irrigation Accounts	Water Rates	Outdoor	Irrigation	IRR
Violification to or Implementation of Tiered Rate Conservation Pricing	Water Rates	Both	All	
Water Rudget Raced Billing for All Customers	Water Rates	Both		
Water Budget Based Billing for Only Irrigation Customers	Water Pater	Outdoor	All	
Public Outrooch and Education	Water Rates	Outdoor	Ingation	(CII,
Public Outreach and Education		D. ul		CII
Water Use Surveys/Audits - Cli	Audit/ Survey	Both	All	
Water Use Surveys/Audits - MFR	Audit/ Survey	Both		
Educational Workshops	Public Outreach/ Workshop	Outdoor		
Garden tour	Public Outreach/ Workshop	Outdoor	Outdoor	SFR
Promote Green Building and Certification	Public Outreach/ Workshop	Both		
Provide Support with Smart Irrigation Controller Setup	Public Outreach/ Workshop	Outdoor	Irrigation	All
Public Outreach through Print & Electronic Media - Focused on Indoor Conservation	Public Outreach/ Workshop	Outdoor	All Indoor	All
Public Outreach through Print & Electronic Media - Focused on Outdoor Irrigation	Public Outreach/ Workshop	Indoor	Irrigation	All
QWEL Training (Qualified Water Efficient Landscaper)	Public Outreach/ Workshop	Outdoor	Irrigation	All
School Education Programs	Public Outreach/ Workshop	Both	All	SFR
Device-Based and Financial Incentive Programs			·	
Direct Install of Efficient Indoor Fixtures - Commercial and Industrial	Direct Install/ No-Cost Device	Indoor	Toilet, Urinal, Faucet, Showerhead	CII
Direct Install of Efficient Indoor Fixtures - Government Buildings	Direct Install/ No-Cost Device	Indoor	Toilet, Urinal, Faucet, Showerhead	CII
Direct Install of Efficient Indoor Fixtures - Low Income Residential	Direct Install/ No-Cost Device	Indoor	Toilet, Faucet, Showerhead	SFR
Direct Install of Efficient Indoor Fixtures - Residential	Direct Install/ No-Cost Device	Indoor	Toilet, Faucet, Showerhead	SFR
High Efficiency Clothes Washer Install - Low Income Residential Customers	Direct Install/ No-Cost Device	Indoor	Clothes Washer	SFR
High Efficiency Faucet Aerator / Showerhead Giveaway - CII Customers	Direct Install/ No-Cost Device	Indoor	Faucet, Showerhead	CII
High Efficiency Faucet Aerator / Showerhead Giveaway - Residential Customers	Direct Install/ No-Cost Device	Indoor	Faucet, Showerhead	SFR
Rain Barrel Giveaway	Direct Install/ No-Cost Device	Outdoor	Irrigation	SFR
Rain Sensor Giveaway	Direct Install/ No-Cost Device	Outdoor	Irrigation	All
Rotating Sprinkler Nozzle Giveaway	Direct Install/ No-Cost Device	Outdoor	Irrigation	All
Smart Irrigation Controller (Weather-Based Irrigation Controller) Giveaway - Large	Direct Install/ No-Cost Device	Outdoor	Irrigation	ME
Smart Irrigation Controller (Weather-Based Irrigation Controller) Giveaway - SFR	Direct Install/ No-Cost Device	Outdoor	Irrigation	SFR
Soil Moisture Sensor Giveaway	Direct Install/ No-Cost Device	Outdoor	Irrigation	All
Toilet Flapper Giveaway - SFR customers	Direct Install/ No-Cost Device	Indoor	Toilet	SFR
UHET Direct Installation - CII	Direct Install/ No-Cost Device	Indoor	Toilet	CII
UHET Direct Installation - Residential	Direct Install/ No-Cost Device	Indoor	Toilet	SFR
Urinal Direct Installation - CII	Direct Install/ No-Cost Device	Indoor	Urinal	CII
Autoclave (Steam-Sterilizer) Retrofit Rebates	Rebate/ Financial Incentive	Indoor	CII Equipment	CII
Connectionless Food Steamer Rebates	Rebate/ Financial Incentive	Indoor	CII Equipment	CII
Dipper Well Rebates	Rebate/ Financial Incentive	Indoor	CII Equipment	CII
Drip Irrigation Incentive for MFR and CII	Rebate/ Financial Incentive	Outdoor	Irrigation	MFI
Drip Irrigation Incentive for SFR	Rebate/ Financial Incentive	Outdoor	Irrigation	SFR
Dry Vacuum Pumps	Rebate/ Financial Incentive	Indoor	CII Equipment	CII
Efficient (EnergyStar) Dishwasher Rebates	Rebate/ Financial Incentive	Indoor	Dishwashers	SFR
High Efficiency Clothes Washer Rebate - Residential	Rebate/ Financial Incentive	Indoor	Clothes Washer	SFR
High Efficiency Clothes Washer Rebate Program - CII	Rebate/ Financial Incentive	Indoor	Clothes Washer	CII
High Efficiency Urinal (<0.25 gal/flush) Rebates - CII	Rebate/ Financial Incentive	Indoor	Urinal	CII

Sector	Priority as a Regional Program	Priority as a Local Program	Preference for Implementation	Current Implementation Status	Notes / Comments
					Enforcement of SB 407 at time of sale.
					Assumes water waste regulations per Executive Order B-40-17 rulemaking is completed largely as currently proposed.
					Fixtures are purchased in bulk at a discounted rate and then sold to customers at the discounted rate
RR					
MFR					
MFR					
MFR					
MFR					
MFR					
CII					
, Cli					
MFR					Could be used for CII customers, but hasn't been vet
MFR					
					More info: https://www.energy.gov/eere/femp/water-efficient-technology-opportunity-steam- sterilizer-condensate-retrofit-kit
					More info: https://www.energy.gov/eere/femp/water-efficient-technology-opportunity- connectionless-food-steamer
					Incentivize replacement of perpetual-flow holders for ice cream dippers & utensils; https://server- products.com/equipment/conservewell/utensil-holder/87740.htm
. CII					
-					
MFR					

	Source	Added By
	2015 Screening	EKI
	Added 2020	EKI
	2015 Screening	EKI
	Added 2020	EKI
	2015 Screening	EKI
e	2015 Screening	ΕΚΙ
	2015 Screening	EKI
	2015 Screening	EKI
	2015 Screening	EKI
	2015 Screening	EKI
	Added 2020	MMWD
	Added 2020	MMWD
	2015 Screening	EKI
	Added 2020	EKI
	2015 Screening	EKI
	2015 Screening	EKI
	Added 2020	EKI
	2015 Screening	EKI
	2015 Screening	ΕΚΙ
	2015 Screening	ΕΚΙ
	2015 Screening	EKI
	2015 Screening	EKI
	2015 Screening Added 2020	EKI EKI
	2015 Screening Added 2020 2015 Screening	EKI EKI EKI
	2015 Screening Added 2020 2015 Screening 2015 Screening	EKI EKI EKI
	2015 Screening Added 2020 2015 Screening 2015 Screening Added 2020	EKI EKI EKI EKI
	2015 Screening Added 2020 2015 Screening 2015 Screening Added 2020 2015 Screening	EKI EKI EKI EKI EKI
	2015 Screening Added 2020 2015 Screening 2015 Screening Added 2020 2015 Screening Added 2020 2015 Screening	EKI EKI EKI EKI EKI EKI
	2015 Screening Added 2020 2015 Screening 2015 Screening Added 2020 2015 Screening Added 2020 2015 Screening	EKI EKI EKI EKI EKI EKI EKI
	2015 Screening Added 2020 2015 Screening 2015 Screening Added 2020 2015 Screening Added 2020 2015 Screening Added 2020	EKI EKI EKI EKI EKI EKI EKI
	2015 Screening Added 2020 2015 Screening 2015 Screening Added 2020 2015 Screening Added 2020 2015 Screening Added 2020 Added 2020	EKI EKI EKI EKI EKI EKI EKI EKI
	2015 Screening Added 2020 2015 Screening 2015 Screening Added 2020 2015 Screening Added 2020 2015 Screening Added 2020 Added 2020 Added 2020	EKI EKI EKI EKI EKI EKI EKI EKI Santa Rosa
	2015 Screening Added 2020 2015 Screening 2015 Screening Added 2020 2015 Screening Added 2020 2015 Screening Added 2020 Added 2020 2015 Screening 2015 Screening	EKI EKI EKI EKI EKI EKI EKI Santa Rosa EKI EKI
	2015 Screening Added 2020 2015 Screening 2015 Screening Added 2020 2015 Screening Added 2020 2015 Screening Added 2020 Added 2020 Added 2020 2015 Screening 2015 Screening	EKI EKI EKI EKI EKI EKI EKI EKI Santa Rosa EKI EKI
	2015 Screening Added 2020 2015 Screening 2015 Screening Added 2020 2015 Screening Added 2020 2015 Screening Added 2020 Added 2020 2015 Screening 2015 Screening 2015 Screening Added 2020 Added 2020	EKI EKI EKI EKI EKI EKI EKI EKI EKI EKI
	2015 Screening Added 2020 2015 Screening 2015 Screening Added 2020 2015 Screening Added 2020 2015 Screening Added 2020 Added 2020 2015 Screening 2015 Screening Added 2020 Added 2020 Added 2020 Added 2020	EKI EKI EKI EKI EKI EKI EKI EKI EKI EKI
	2015 Screening Added 2020 2015 Screening 2015 Screening Added 2020 2015 Screening Added 2020 2015 Screening Added 2020 Added 2020 2015 Screening 2015 Screening 2015 Screening Added 2020 Added 2020 Added 2020	EKI EKI EKI EKI EKI EKI EKI EKI EKI EKI
	2015 Screening Added 2020 2015 Screening 2015 Screening Added 2020 2015 Screening Added 2020 2015 Screening Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 Added 2020	EKI EKI EKI EKI EKI EKI EKI EKI EKI EKI
	2015 Screening Added 2020 2015 Screening 2015 Screening Added 2020 2015 Screening Added 2020 2015 Screening Added 2020 Added 2020 2015 Screening 2015 Screening Added 2020 Added 2020 Added 2020 Added 2020 2015 Screening 2015 Screening	EKI EKI EKI EKI EKI EKI EKI EKI EKI EKI
	2015 Screening Added 2020 2015 Screening 2015 Screening Added 2020 2015 Screening Added 2020 2015 Screening Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 2015 Screening 2015 Screening 2015 Screening 2015 Screening 2015 Screening	EKI EKI EKI EKI EKI EKI EKI EKI EKI EKI
	2015 Screening Added 2020 2015 Screening 2015 Screening Added 2020 2015 Screening Added 2020 2015 Screening Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 2015 Screening 2015 Screening 2015 Screening 2015 Screening 2015 Screening 2015 Screening 2015 Screening 2015 Screening	EKI EKI EKI EKI EKI EKI EKI EKI EKI EKI
	2015 Screening Added 2020 2015 Screening 2015 Screening Added 2020 2015 Screening Added 2020 2015 Screening Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 2015 Screening 2015 Screening 2015 Screening 2015 Screening 2015 Screening 2015 Screening 2015 Screening 2015 Screening 2015 Screening 2015 Screening	EKI EKI EKI EKI EKI EKI EKI EKI EKI EKI
	2015 Screening Added 2020 2015 Screening 2015 Screening Added 2020 2015 Screening Added 2020 2015 Screening Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 Added 2020 2015 Screening 2015 Screening	EKI EKI EKI EKI EKI EKI EKI EKI EKI EKI

# Prioritization and Screening of Future Water Conservation Measures

Marin-Sonoma Saving Water Partnership

INSTRUCTIONS: Please review and complete the orange highlighted cells using the provided dropdown lists. Comments and clarifications may be added to the comments column on the right. You may use the filter features to help navigate this list. "Preference for Implementation" need only be completed when a program is given a priorty of 3 or greater. See READ ME tab for additional information.

		Priority as a									
		Indoor /			Regional	Priority as a Local	Preference for	Current Implementation			
Conservation Measure/Program	Туре	Outdoor	Primary End Use	Sector	Program	Program	Implementation	Status	Notes / Comments	Source	Added By
Hot Water on Demand Pump System Rebate	Rebate/ Financial Incentive	Indoor	Hot Water	SFR, MFR						2015 Screening	EKI
Incentivize Artificial Turf for Sports Fields	Rebate/ Financial Incentive	Outdoor	Irrigation	CII						2015 Screening	EKI
Incentivize Cooling Tower Upgrades	Rebate/ Financial Incentive	Indoor	Cooling Towers	CII						Added 2020	EKI
Incentivize Gray Water Retrofit for Existing SFR Customers	Rebate/ Financial Incentive	Outdoor	Irrigation / Gray Water	SFR						2015 Screening	EKI
Incentivize Gray Water Systems for New CII Development	Rebate/ Financial Incentive	Both	Irrigation / Gray Water	CII						2015 Screening	EKI
Incentivize Irrigation Equipment Upgrades - Large Landscapes	Rebate/ Financial Incentive	Outdoor	Irrigation	MFR, CII, IRR						2015 Screening	EKI
Incentivize Irrigation Equipment Upgrades - SFR	Rebate/ Financial Incentive	Outdoor	Irrigation	SFR						2015 Screening	EKI
Incentivize Replacement of Inefficient Commercial and Industrial Equipment	Rebate/ Financial Incentive	Indoor	CII Equipment	CII					Example: SoCal Water Smart Water Savings Incentive Program:	2015 Screening	EKI
									https://socalwatersmart.com/en/commercial/water-savings-incentive-program/		
Incentivize Replacement of Pressure Reducing Valves (PRVs) with 60-70 psi PRVs	Rebate/ Financial Incentive	Both	Water loss; Irrigation	All					PRVs must be installed by customers with pressure exceeding 80 psi, per the plumbing code	2015 Screening	EKI
Incentivize Submetering for Existing Customers - CII	Rebate/ Financial Incentive	Both	All Indoor	MFR, COM, IRR						2015 Screening	EKI
Incentivize Submetering for Existing Customers - MFR	Rebate/ Financial Incentive	Both	All Indoor	MFR						2015 Screening	EKI
Incentivize Submetering of Cooling Towers for Existing Customers	Rebate/ Financial Incentive	Indoor	Cooling Towers	CII						2015 Screening	EKI
Indoor Fixture Program For Hotels & Motels	Rebate/ Financial Incentive	Indoor	All Indoor	CII						2015 Screening	EKI
Indoor Fixture Program For Schools	Rebate/ Financial Incentive	Indoor	All Indoor	CII						2015 Screening	EKI
Landscape Conversion or Turf Removal - MFR and CII	Rebate/ Financial Incentive	Outdoor	Irrigation	MFR, CII						2015 Screening	EKI
Landscape Conversion or Turf Removal -SFR	Rebate/ Financial Incentive	Outdoor	Irrigation	SFR						2015 Screening	EKI
Mulch rebate	Rebate/ Financial Incentive	Outdoor	Irrigation	SFR						Added 2020	MMWD
Nonresidential Incentive for Self-closing or Metering Faucets	Rebate/ Financial Incentive	Indoor	Faucet	CII						Added 2020	Sonoma
Plumber Initiated UHET and / or Urinal Retrofit Program	Rebate/ Financial Incentive	Indoor	Toilet	All						2015 Screening	EKI
Rain Barrel Rebate	Rebate/ Financial Incentive	Outdoor	Irrigation	SFR						2015 Screening	EKI
Rain Sensor Rebate	Rebate/ Financial Incentive	Outdoor	Irrigation	All						2015 Screening	EKI
Rainwater Catchment System Rebate for Large Landscapes	Rebate/ Financial Incentive	Outdoor	Irrigation	MFR, CII						2015 Screening	EKI
Rebates for Conductivity Controllers on Cooling Towers	Rebate/ Financial Incentive	Indoor	Cooling Towers	CII						2015 Screening	EKI
Restaurant Spray Nozzle Rebates	Rebate/ Financial Incentive	Indoor	CII Equipment	CII						2015 Screening	EKI
Rotating Sprinkler Nozzle Rebate	Rebate/ Financial Incentive	Outdoor	Irrigation	All						2015 Screening	EKI
Smart Irrigation Controller (Weather-Based Irrigation Controller) Rebates - Large	Rebate/ Financial Incentive	Outdoor	Irrigation	MFR, CII						2015 Screening	EKI
Landscape											
Smart Irrigation Controller (Weather-Based Irrigation Controller) Rebates - SFR	Rebate/ Financial Incentive	Outdoor	Irrigation	SFR						2015 Screening	EKI
Soil Moisture Sensor Rebate	Rebate/ Financial Incentive	Outdoor	Irrigation	All						2015 Screening	EKI
Swimming Pool and Hot Tub Cover Rebates	Rebate/ Financial Incentive	Outdoor	Pool/Hot Tub	SFR, MFR						Added 2020	EKI
Thermostatic Shut-Off Valve Snowerneads/Tub Spouts Rebates	Rebate/ Financial Incentive	Indoor	Shower	SFR, MFR, CII					Reduce not water use before snowering https://www.thinkevolve.com/	Added 2020	EKI
Tier 4 Exemption	Rebate/ Financial Incentive	BOTH	tollet, Faucet, Snowernead,	SFR					Exemption from high tier water rates w/installation of devices	Added 2020	MINIVUD
LILLET <1.0 gol/fluch Pohoto CIL	Robato / Einancial Incontivo	Indoor	Ciotries washer, imgation							201E Scrooning	EVI
UHET <1.0 gal/flush Rebate - Cli	Rebate/ Financial Incentive	Indoor	Toilet							2015 Screening	
Water Savings Incentive Program for CII	Rebate/ Financial Incentive	Indoor	All Indoor						Einancial incentive to reward demonstrated water savings and offset capital improvement costs:	2015 Screening	EKI
		muoor	All Indoor						Example: SoCal Water Smart Water Savings Incentive Program:	2013 Screening	
									https://socalwatersmart.com/en/commercial/water-savings-incentive-program/		
Policies and Regulations									https://socalwatersmart.com/en/commercial/water savings incentive program/		
Policies and Regulations	Delias / De avlation	Deth									EKI
Demand Offset/ water Neutral Policy for Large New Developments	Policy/ Regulation	Both		All						Added 2020	EKI
Prohibit Once through Cooling Systems	Policy/ Regulation	Both								2015 Screening	EKI
Require <0.25 gal/flush Unitals in New Development	Policy/ Regulation	Indoor							State minimum officiancy is 1.28 gal/fluch	2015 Screening	EKI
Require <1.0 gal/itusii Tollets ili New Development	Policy/ Regulation	Indoor								2015 Scrooping	
Require Efficient (EnergyStar) Dishwashers in New Development	Policy/ Regulation	Indoor	Dishwashers	SER MER						2015 Screening	EKI
Require High Efficiency Clothes Washers in New Development	Policy/ Regulation	Indoor	Clothes Washer	SER MER						2015 Screening	FKI
Require High Enclored Clothes Washers in New Development	Policy/ Regulation	Indoor	Shower/Sink	SFR, MFR						2015 Screening	FKI
	l oney, regulation									2010 001001116	
Require Irrigation Designers / Installers be Certified (OWFL)	Policy/ Regulation	Outdoor	Irrigation	All						2015 Screening	EKI
Require On-Site Water Reuse Systems (Grev Water or Black Water) for Large CI	Policy/ Regulation	Outdoor	Irrigation / Recycled Water						Example: https://sfwater.org/index.aspx?page=686	Added 2020	FKI
Developments											
Require Plumbing for Gray Water in New SER Development	Policy/ Regulation	Outdoor	Irrigation / Gray Water	SER						2015 Screening	EKI
Require Plumbing for Recycled Water in New CII Development	Policy/ Regulation	Outdoor	Irrigation / Recycled Water							Added 2020	FKI
Require Plumbing for Recycled Water in New MFR Development	Policy/ Regulation	Outdoor	Irrigation / Recycled Water	MFR						Added 2020	FKI
Require Rain Barrels in New Development	Policy/ Regulation	Outdoor	Irrigation	SFR						2015 Screening	EKI
Require Submetering by Unit for Existing Commercial Customers	Policy/ Regulation	Indoor	All Indoor	CII						Added 2020	EKI
Require Submetering by Unit for New Commercial Developments	Policy/ Regulation	Indoor	All Indoor	CII						Added 2020	EKI
Require Submetering for New MFR Developments	Policy/ Regulation	Indoor	All Indoor	MFR						2015 Screening	EKI
Require Submetering for New Mobile Home Park Developments	Policy/ Regulation	Indoor	All Indoor	MFR						2015 Screening	EKI
Require Submetering of Cooling Towers for Existing Customers	Policy/ Regulation	Indoor	Cooling Towers	CII						Added 2020	EKI
Require Submetering of Cooling Towers for New Development	Policy/ Regulation	Indoor	Cooling Towers	CII						Added 2020	EKI
Require Submetering of Existing MFR (and Mobile Home Park) Customers	Policy/ Regulation	Indoor	All Indoor	MFR						Added 2020	EKI
Require Submetering of Landscaping for Existing MFR and Commercial Customers	Policy/ Regulation	Outdoor	Irrigation	MFR. CII						Added 2020	EKI
Require Submetering of Landscaping for New MFR and Commercial Developments	Policy/ Regulation	Outdoor	Irrigation	CII						Added 2020	EKI
Require Swimming Pool and Hot Tub Covers	Policy/ Regulation	Outdoor	Pool/Hot Tub	SFR, MFR						2015 Screening	EKI
			· ·							- 0	

# Prioritization and Screening of Future Water Conservation Measures

Marin-Sonoma Saving Water Partnership

INSTRUCTIONS: Please review and complete the orange highlighted cells using the provided dropdown lists. Comments and clarifications may be added to the comments column on the right. You may use the filter features to help navigate this list. "Preference for Implementation" need only be completed when a program is given a priorty of 3 or greater. See READ ME tab for additional information.

					Priority as a						
		Indoor /			Regional	Priority as a Local	Preference for	Current Implementation			
Conservation Measure/Program	Туре	Outdoor	Primary End Use	Sector	Program	Program	Implementation	Status	Notes / Comments	Source	Added By
Require Water Efficiency Plan Reviews for New CII Development	Policy/ Regulation	Both	All Indoor	CII						2015 Screening	EKI
Require Weather Adjusting Smart Irrigation Controllers, Rain Sensors, and/or Soil	Policy/ Regulation	Outdoor	Irrigation	All						2015 Screening	EKI
Moisture Sensors in New Development											
Restrict Landscape Irrigation to Designated Days/Times	Policy/ Regulation	Outdoor	Irrigation	All					Under all conditions, not just drought	2015 Screening	EKI
Water Conserving Landscape and Irrigation Codes, More Stringent than MWELO	Policy/ Regulation	Outdoor	Irrigation	All						2015 Screening	EKI
Water Waste Ordinance	Policy/ Regulation	Outdoor	All Outdoor	All						Added 2020	MMWD

### Abbreviations:

AMI = advanced metering infrastructure CII = commercial, industrial, institutional COM = commercial HET = high efficiency toilet HEU = high efficiency urinal Info = information IRR = irrigation account MFR = multi-family residential MWELO = Model Water Efficient Landscape Ordinance PRV = pressure reducing valve SFR = single-family residential SMSWP = Sonoma-Marin Saving Water Partnership UHET = ultra high efficiency toilet

Appendices 2020 Urban Water Management Plan Marin Municipal Water District



# Appendix D

# **UWMP Agency Notification Letters**



November 9, 2020

City of Belvedere Mr. Craig Middleton City Manager 450 San Rafael Avenue Belvedere, CA 94920

### Re: Notice of Preparation of Urban Water Management Plan - 2020 Update

The Urban Water Management Planning Act (California Water Code §10608–10656) requires the Marin Municipal Water District ("MMWD") to update its Urban Water Management Plan ("UWMP") every 5 years. MMWD is currently reviewing its existing UWMP, which was updated in 2016, and considering revisions to the document. The updated UWMP is due by July 1, 2021. We invite your agency's participation in this revision process.

A draft of the 2020 UWMP will be made available for public review and a public hearing will be scheduled for spring of next year. In the meantime, if you would like more information regarding the District's 2015 UWMP and the schedule for preparing the 2020 UWMP, or if you would like to participate in the preparation of the 2020 UWMP, please contact Carrie Pollard at:

Marin Municipal Water District 220 Nellen Avenue Corte Madera, CA 94925 Phone: (415) 945-1522 capollard@marinwater.org

Carrier Pollad

Carrie Pollard Water Efficiency Manager



November 9, 2020

Town of Corte Madera Mr. Todd Cusimano Town Manager 300 Tamalpais Drive Corte Madera, CA 94925-0159

### Re: Notice of Preparation of Urban Water Management Plan - 2020 Update

The Urban Water Management Planning Act (California Water Code §10608–10656) requires the Marin Municipal Water District ("MMWD") to update its Urban Water Management Plan ("UWMP") every 5 years. MMWD is currently reviewing its existing UWMP, which was updated in 2016, and considering revisions to the document. The updated UWMP is due by July 1, 2021. We invite your agency's participation in this revision process.

A draft of the 2020 UWMP will be made available for public review and a public hearing will be scheduled for spring of next year. In the meantime, if you would like more information regarding the District's 2015 UWMP and the schedule for preparing the 2020 UWMP, or if you would like to participate in the preparation of the 2020 UWMP, please contact Carrie Pollard at:

Marin Municipal Water District 220 Nellen Avenue Corte Madera, CA 94925 Phone: (415) 945-1522 capollard@marinwater.org

Carrier Pollad

Carrie Pollard Water Efficiency Manager



November 9, 2020

Town of Fairfax Mr. Garrett Toy Town Manager 142 Bolinas Road Fairfax, CA 94930

#### Re: Notice of Preparation of Urban Water Management Plan - 2020 Update

The Urban Water Management Planning Act (California Water Code §10608–10656) requires the Marin Municipal Water District ("MMWD") to update its Urban Water Management Plan ("UWMP") every 5 years. MMWD is currently reviewing its existing UWMP, which was updated in 2016, and considering revisions to the document. The updated UWMP is due by July 1, 2021. We invite your agency's participation in this revision process.

A draft of the 2020 UWMP will be made available for public review and a public hearing will be scheduled for spring of next year. In the meantime, if you would like more information regarding the District's 2015 UWMP and the schedule for preparing the 2020 UWMP, or if you would like to participate in the preparation of the 2020 UWMP, please contact Carrie Pollard at:

Marin Municipal Water District 220 Nellen Avenue Corte Madera, CA 94925 Phone: (415) 945-1522 capollard@marinwater.org

Carrier Pollad

Carrie Pollard Water Efficiency Manager



November 9, 2020

City of Larkspur Mr. Dan Schwarz City Manager 400 Magnolia Avenue Larkspur, CA 94939

### Re: Notice of Preparation of Urban Water Management Plan - 2020 Update

The Urban Water Management Planning Act (California Water Code §10608–10656) requires the Marin Municipal Water District ("MMWD") to update its Urban Water Management Plan ("UWMP") every 5 years. MMWD is currently reviewing its existing UWMP, which was updated in 2016, and considering revisions to the document. The updated UWMP is due by July 1, 2021. We invite your agency's participation in this revision process.

A draft of the 2020 UWMP will be made available for public review and a public hearing will be scheduled for spring of next year. In the meantime, if you would like more information regarding the District's 2015 UWMP and the schedule for preparing the 2020 UWMP, or if you would like to participate in the preparation of the 2020 UWMP, please contact Carrie Pollard at:

Marin Municipal Water District 220 Nellen Avenue Corte Madera, CA 94925 Phone: (415) 945-1522 capollard@marinwater.org

Carrier Pollad

Carrie Pollard Water Efficiency Manager



November 9, 2020

Marin County Administrator Mr. Matthew Hymel Administrator 3501 Civic Center Drive, Suite 325 San Rafael, CA 94903

### Re: Notice of Preparation of Urban Water Management Plan - 2020 Update

The Urban Water Management Planning Act (California Water Code §10608–10656) requires the Marin Municipal Water District ("MMWD") to update its Urban Water Management Plan ("UWMP") every 5 years. MMWD is currently reviewing its existing UWMP, which was updated in 2016, and considering revisions to the document. The updated UWMP is due by July 1, 2021. We invite your agency's participation in this revision process.

A draft of the 2020 UWMP will be made available for public review and a public hearing will be scheduled for spring of next year. In the meantime, if you would like more information regarding the District's 2015 UWMP and the schedule for preparing the 2020 UWMP, or if you would like to participate in the preparation of the 2020 UWMP, please contact Carrie Pollard at:

Marin Municipal Water District 220 Nellen Avenue Corte Madera, CA 94925 Phone: (415) 945-1522 capollard@marinwater.org

Marie Bellad

Carrie Pollard Water Efficiency Manager



November 9, 2020

City of Mill Valley Mr. Alan Piombo City Manager 26 Corte Madera Avenue Mill Valley, CA 94941

#### Re: Notice of Preparation of Urban Water Management Plan - 2020 Update

The Urban Water Management Planning Act (California Water Code §10608–10656) requires the Marin Municipal Water District ("MMWD") to update its Urban Water Management Plan ("UWMP") every 5 years. MMWD is currently reviewing its existing UWMP, which was updated in 2016, and considering revisions to the document. The updated UWMP is due by July 1, 2021. We invite your agency's participation in this revision process.

A draft of the 2020 UWMP will be made available for public review and a public hearing will be scheduled for spring of next year. In the meantime, if you would like more information regarding the District's 2015 UWMP and the schedule for preparing the 2020 UWMP, or if you would like to participate in the preparation of the 2020 UWMP, please contact Carrie Pollard at:

Marin Municipal Water District 220 Nellen Avenue Corte Madera, CA 94925 Phone: (415) 945-1522 capollard@marinwater.org

Marie Bellal

Carrie Pollard Water Efficiency Manager



November 9, 2020

City of Novato Mr. Adam McGill City Manager 922 Machin Avenue Novato, CA 94945

#### Re: Notice of Preparation of Urban Water Management Plan - 2020 Update

The Urban Water Management Planning Act (California Water Code §10608–10656) requires the Marin Municipal Water District ("MMWD") to update its Urban Water Management Plan ("UWMP") every 5 years. MMWD is currently reviewing its existing UWMP, which was updated in 2016, and considering revisions to the document. The updated UWMP is due by July 1, 2021. We invite your agency's participation in this revision process.

A draft of the 2020 UWMP will be made available for public review and a public hearing will be scheduled for spring of next year. In the meantime, if you would like more information regarding the District's 2015 UWMP and the schedule for preparing the 2020 UWMP, or if you would like to participate in the preparation of the 2020 UWMP, please contact Carrie Pollard at:

Marin Municipal Water District 220 Nellen Avenue Corte Madera, CA 94925 Phone: (415) 945-1522 capollard@marinwater.org

Carrier Pollad

Carrie Pollard Water Efficiency Manager



November 9, 2020

Sonoma County Water Agency Mr. Grant Davis General Manager 404 Aviation Blvd Santa Rosa, CA 95403

### Re: Notice of Preparation of Urban Water Management Plan - 2020 Update

The Urban Water Management Planning Act (California Water Code §10608–10656) requires the Marin Municipal Water District ("MMWD") to update its Urban Water Management Plan ("UWMP") every 5 years. MMWD is currently reviewing its existing UWMP, which was updated in 2016, and considering revisions to the document. The updated UWMP is due by July 1, 2021. We invite your agency's participation in this revision process.

A draft of the 2020 UWMP will be made available for public review and a public hearing will be scheduled for spring of next year. In the meantime, if you would like more information regarding the District's 2015 UWMP and the schedule for preparing the 2020 UWMP, or if you would like to participate in the preparation of the 2020 UWMP, please contact Carrie Pollard at:

Marin Municipal Water District 220 Nellen Avenue Corte Madera, CA 94925 Phone: (415) 945-1522 capollard@marinwater.org

Marie Bellal

Carrie Pollard Water Efficiency Manager



November 9, 2020

North Marin Water District Mr. Chris DeGabriele General Manager 999 Rush Creek Place Novato, CA 94945

### Re: Notice of Preparation of Urban Water Management Plan - 2020 Update

The Urban Water Management Planning Act (California Water Code §10608–10656) requires the Marin Municipal Water District ("MMWD") to update its Urban Water Management Plan ("UWMP") every 5 years. MMWD is currently reviewing its existing UWMP, which was updated in 2016, and considering revisions to the document. The updated UWMP is due by July 1, 2021. We invite your agency's participation in this revision process.

A draft of the 2020 UWMP will be made available for public review and a public hearing will be scheduled for spring of next year. In the meantime, if you would like more information regarding the District's 2015 UWMP and the schedule for preparing the 2020 UWMP, or if you would like to participate in the preparation of the 2020 UWMP, please contact Carrie Pollard at:

Marin Municipal Water District 220 Nellen Avenue Corte Madera, CA 94925 Phone: (415) 945-1522 capollard@marinwater.org

Janei Pollad

Carrie Pollard Water Efficiency Manager



November 9, 2020

Town of Tiburon Mr. Greg Chanis Town Manager 1505 Tiburon Boulevard Tiburon, CA 94920

### Re: Notice of Preparation of Urban Water Management Plan - 2020 Update

The Urban Water Management Planning Act (California Water Code §10608–10656) requires the Marin Municipal Water District ("MMWD") to update its Urban Water Management Plan ("UWMP") every 5 years. MMWD is currently reviewing its existing UWMP, which was updated in 2016, and considering revisions to the document. The updated UWMP is due by July 1, 2021. We invite your agency's participation in this revision process.

A draft of the 2020 UWMP will be made available for public review and a public hearing will be scheduled for spring of next year. In the meantime, if you would like more information regarding the District's 2015 UWMP and the schedule for preparing the 2020 UWMP, or if you would like to participate in the preparation of the 2020 UWMP, please contact Carrie Pollard at:

Marin Municipal Water District 220 Nellen Avenue Corte Madera, CA 94925 Phone: (415) 945-1522 capollard@marinwater.org

Carrier Pollad

Carrie Pollard Water Efficiency Manager



November 9, 2020

City of Sausalito Mr. Adam Politzer City Manager 420 Litho Street Sausalito, CA 94965

### Re: Notice of Preparation of Urban Water Management Plan - 2020 Update

The Urban Water Management Planning Act (California Water Code §10608–10656) requires the Marin Municipal Water District ("MMWD") to update its Urban Water Management Plan ("UWMP") every 5 years. MMWD is currently reviewing its existing UWMP, which was updated in 2016, and considering revisions to the document. The updated UWMP is due by July 1, 2021. We invite your agency's participation in this revision process.

A draft of the 2020 UWMP will be made available for public review and a public hearing will be scheduled for spring of next year. In the meantime, if you would like more information regarding the District's 2015 UWMP and the schedule for preparing the 2020 UWMP, or if you would like to participate in the preparation of the 2020 UWMP, please contact Carrie Pollard at:

Marin Municipal Water District 220 Nellen Avenue Corte Madera, CA 94925 Phone: (415) 945-1522 capollard@marinwater.org

Carrier Pollad

Carrie Pollard Water Efficiency Manager



November 9, 2020

City of San Rafael Mr. Jim Schutz City Manager 1400 Fifth Avenue San Rafael, CA 94915-1560

### Re: Notice of Preparation of Urban Water Management Plan - 2020 Update

The Urban Water Management Planning Act (California Water Code §10608–10656) requires the Marin Municipal Water District ("MMWD") to update its Urban Water Management Plan ("UWMP") every 5 years. MMWD is currently reviewing its existing UWMP, which was updated in 2016, and considering revisions to the document. The updated UWMP is due by July 1, 2021. We invite your agency's participation in this revision process.

A draft of the 2020 UWMP will be made available for public review and a public hearing will be scheduled for spring of next year. In the meantime, if you would like more information regarding the District's 2015 UWMP and the schedule for preparing the 2020 UWMP, or if you would like to participate in the preparation of the 2020 UWMP, please contact Carrie Pollard at:

Marin Municipal Water District 220 Nellen Avenue Corte Madera, CA 94925 Phone: (415) 945-1522 capollard@marinwater.org

Carrier Pollad

Carrie Pollard Water Efficiency Manager


220 Nellen Avenue Corte Madera CA 94925-1169 marinwater.org

November 9, 2020

Town of San Anselmo Mr. David Donery Town Administrator 525 San Anselmo Avenue San Anselmo, CA 94960

### Re: Notice of Preparation of Urban Water Management Plan - 2020 Update

The Urban Water Management Planning Act (California Water Code §10608–10656) requires the Marin Municipal Water District ("MMWD") to update its Urban Water Management Plan ("UWMP") every 5 years. MMWD is currently reviewing its existing UWMP, which was updated in 2016, and considering revisions to the document. The updated UWMP is due by July 1, 2021. We invite your agency's participation in this revision process.

A draft of the 2020 UWMP will be made available for public review and a public hearing will be scheduled for spring of next year. In the meantime, if you would like more information regarding the District's 2015 UWMP and the schedule for preparing the 2020 UWMP, or if you would like to participate in the preparation of the 2020 UWMP, please contact Carrie Pollard at:

Marin Municipal Water District 220 Nellen Avenue Corte Madera, CA 94925 Phone: (415) 945-1522 capollard@marinwater.org

Sincerely,

Marie Pollad

Carrie Pollard Water Efficiency Manager



220 Nellen Avenue Corte Madera CA 94925-1169 marinwater.org

November 9, 2020

Town of Ross Mr. Joe Chinn Town Manager P.O. Box 320 Ross, CA 94957

#### Re: Notice of Preparation of Urban Water Management Plan - 2020 Update

The Urban Water Management Planning Act (California Water Code §10608–10656) requires the Marin Municipal Water District ("MMWD") to update its Urban Water Management Plan ("UWMP") every 5 years. MMWD is currently reviewing its existing UWMP, which was updated in 2016, and considering revisions to the document. The updated UWMP is due by July 1, 2021. We invite your agency's participation in this revision process.

A draft of the 2020 UWMP will be made available for public review and a public hearing will be scheduled for spring of next year. In the meantime, if you would like more information regarding the District's 2015 UWMP and the schedule for preparing the 2020 UWMP, or if you would like to participate in the preparation of the 2020 UWMP, please contact Carrie Pollard at:

Marin Municipal Water District 220 Nellen Avenue Corte Madera, CA 94925 Phone: (415) 945-1522 capollard@marinwater.org

Sincerely,

Marie Bellal

Carrie Pollard Water Efficiency Manager

Appendices 2020 Urban Water Management Plan Marin Municipal Water District



## Appendix E

### **UWMP Public Notification Notices**

Pending Appendix

Appendices 2020 Urban Water Management Plan Marin Municipal Water District



# Appendix F

### SB X7-7 Compliance Tables

SB X7-7 Regional Alliance - 2020 GPCD (Actual)					
Participating Member Agency Name Add rows as needed	2020 Actual GPCD <sup>1</sup>	2020 Population	(2020 GPCD) X (2020 Population)	Regional Alliance GPCD (Actual)	2020
City of Cotati	113	7,533	851,229		
Marin Municipal Water District	128	191,269	24,482,432		
North Marin Water District	119	61,658	7,337,302		
City of Petaluma	107	64,251	6,900,557		
City of Rohnert Park	96	43,069	4,126,872		
City of Santa Rosa	99	173,628	17,189,172		
City of Sonoma	166	11,725	1,946,350		
Valley of the Moon Water Distict	102	23,077	2,353,854		
Town of Windsor	119	28,397	3,379,243		
Regional Alliance Totals	1,049	604,607	68,567,011		113

<sup>\*</sup> All participating agencies must submit individual SB X7-7 Tables, as applicable, showing the individual agency's calculations. These tables are: SB X7-7 Tables 0 through 6, Table 7, any required supporting tables (as stated in SB X7-7 Table 7), and SB X7-7 Table 9, as applicable.These individual agency tables will be submitted with the individual or Regional Urban Water Management Plan.

NOTES

SB X7-7 Regional Alliance - 2020 Compliance					
2020 Actual GPCD	Optional Adjustment for Economic Growth <sup>1</sup>	Adjusted 2020 Actual GPCD	2020 Target GPCD <sup>2</sup>	Did Alliance Achieve Targeted Reduction for 2020?	
113	-	113	129	YES	
<sup>1</sup> Adjustments for economic growth can be applied to either the individual supplier's data or to the aggregate regional alliance data (but not both), depending upon availability of suitable data and methods. GPCD will be taken from the Regional Alliance's SB X7-7 Verification Form, Weighted Target Table.					
NOTES					

SB X7-7 RA1 - Weighted Baseline				
Participating Member Agency Name	10-15 year Baseline GPCD*	Average Population During 10-15 Year Baseline Period	(Baseline GPCD) X (Population)	Regional Alliance Weighted Average 10-15 Year Baseline GPCD
City of Cotati	159	6,559	1,043,146	
Marin Municipal Water District	149	178,670	26,690,318	
North Marin Water District	173	54,061	9,370,435	
City of Petaluma	180	52,622	9,491,997	
City of Rohnert Park	161	40,811	6,582,847	
City of Santa Rosa	145	143,109	20,806,963	
City of Sonoma	225	9,679	2,173,212	
Valley of the Moon Water Distict	146	20,969	3,058,648	
Town of Windsor	156	24,572	3,834,809	
Regional Alliance Total	1,495	531,051	83,052,375	156

\*All participating agencies must submit individual SB X7-7 Tables, as applicable, showing the individual agency's calculations. These tables are: SB X7-7 Tables 0 through 6, Table 7, any required supporting tables (as stated in SB X7-7 Table 7), and SB X7-7 Table 9, as applicable. These individual agency tables will be submitted with the individual or Regional Urban Water Management Plan.

NOTES

SB	SB X7-7 RA1 - Weighted 2020 Target					
Participating Member Agency Name	2020 Target GPCD*	2015 Population	(Target) X (Population)	Regional Alliance Weighted Average 2020 Target		
City of Cotati	130	7,288	947,440			
Marin Municipal Water District	124	189,000	23,436,000			
North Marin Water District	139	61,381	8,531,959			
City of Petaluma	141	61,798	8,713,518			
City of Rohnert Park	119	41,675	4,959,325			
City of Santa Rosa	126	173,071	21,806,946			
City of Sonoma	180	11,147	2,006,460			
Valley of the Moon Water Distict	124	23,478	2,911,272			
Town of Windsor	130	27,486	3,573,180			
Regional Alliance Total	1,213	596,324	76,886,100	129		

\*All participating agencies must submit individual SB X7-7 Tables, as applicable, showing the individual agency's calculations. These tables are: SB X7-7 Tables 0 through 6 , Table 7, any required supporting tables (as stated in SB X7-7 Table 7), and SB X7-7 Table 9, as applicable.These individual agency tables will be submitted with the individual or Regional Urban Water Management Plan.

NOTES

### SB X7-7 2020 Compliance Form

The SB X7-7 2020 Compliance Form is for the calculation of 2020 compliance only. All retail suppliers must complete the SB X7-7 Compliance Form. Baseline and target calculations are done in the SB X 7-7 Verification Form.

The SB X7-7 Verification Form is for the calculation of baselines and targets and is a separateworkbook from the SB X7-7 2020 Compliance Form.Most Suppliers willhave completed the SB X7-7 Verification Form with their 2015 UWMP and do not need tocomplete this form again in 2020. See Chapter 5 Section 5.3 of the UWMP Guidebook for moreinformation regarding which Suppliers must, or may, complete the SB X7-7 Verification Form fortheir 2020 UWMP.2020 compliance calculations are done in the SB X7-7 2020 Compliance Form.

### **WUE Data Portal Entry Exceptions**

The data from the tables below will not be entered into WUE Data Portal tables. These tables will be submitted as separate uploads, in Excel, to WUE Data Portal.

### **Process Water Deduction**

SB X7-7 tables 4-C, 4-C.1, 4-C.2, 4-C.3, 4-C.4 and 4-D A supplier that will use the process water deduction will complete the appropriate tables in Excel, submit them as a separate upload to the WUE Data Portal, and include them in its UWMP. **SB X7-7 Table 0: Units of Measure Used in 2020 UWMP\*** *(select one from the drop down list)* 

Acre Feet

\*The unit of measure must be consistent throughout the UWMP, as reported in Submittal Table 2-3.

NOTES:

SB X7-7 T	SB X7-7 Table 2: Method for 2020 Population Estimate			
	Method Used to Determine 2020 Population (may check more than one)			
	1. Department of Finance (DOF) or American Community Survey (ACS)			
	2. Persons-per-Connection Method			
	3. DWR Population Tool			
	<b>4. Other</b> DWR recommends pre-review			
NOTES:				

SB X7-7 Table 3: 2020 Service Area Population			
2020 Compliance Year Population			
<b>2020</b> 191,26			
NOTES:			

SB X7-7 Table 4: 2020 Gross Water Use							
			-	2020 Deducti	ons	_	
Compliance Year 2020	2020 Volume Into Distribution System This column will remain blank until SB X7-7 Table 4-A is completed.	Exported Water *	Change in Dist. System Storage* (+/-)	Indirect Recycled Water This column will remain blank until SB X7-7 Table 4-B is completed.	Water Delivered for Agricultural Use*	Process Water This column will remain blank until SB X7-7 Table 4-D is completed.	2020 Gross Water Use
	27,451	-	-	-	-	-	27,451
* Units of measure (AF, MG, or CCF) must remain consistent throughout the UWMP, as reported in SB X7-7 Table 0 and							
Submittal Table 2	Submittal Table 2-3.						
NOTES:							

SB X7-7 Table 4-A: 2020 Volume Entering the Distribution System(s), Meter Error Adjustment Complete one table for each source.				
Name of S	ource	Local Lakes		
This water	<b>source is</b> (a	check one) :		
$\checkmark$	The supplie	er's own water source		
	A purchase	d or imported source		
Compliance Year 2020		Volume Entering Distribution System <sup>1</sup>	Meter Error Adjustment <sup>2</sup> <i>Optional</i> (+/-)	Corrected Volume Entering Distribution System
		20,629	-	20,629
<sup>1</sup> Units of measure (AF, MG, or CCF) must remain consistent throughout the UWMP, as reported in SB X7-7 Table 0 and Submittal Table 2-3. Adjustment - See guidance in Methodology 1, Step 3 of Methodologies Document				
NOTES				

SB X7-7 T	able 4-A: 2	2020 Volume Entering	the Distributio	n System(s) Meter
Error Adju	ustment			
Complete	one table fo	or each source.		
Name of S	ource	Sonoma County Water Age	ency	
This water	<b>source is</b> (a	check one):		
	The supplie	er's own water source		
$\checkmark$	A purchase	ed or imported source		
Compliance Year 2020		Volume Entering Distribution System <sup>1</sup>	Meter Error Adjustment <sup>2</sup> <i>Optional</i> (+/-)	Corrected Volume Entering Distribution System
		6,822		6,822
<sup>1</sup> Units of measure (AF, MG , or CCF) must remain consistent throughout the UWMP, as reported in SB X7-7 Table 0 and Submittal Table 2-3. Adjustment - See guidance in Methodology 1, Step 3 of Methodologies Document				
NOTES:				

SB X7-7 Table 5: 2020 Gallons Per Capita Per Day (GPCD)				
2020 Gross Water Fm SB X7-7 Table 4	2020 Population Fm SB X7-7 Table 3	2020 GPCD		
27,451	191,269	128		
NOTES:				

SB X7-7 Table 9: 2020 Compliance							
		Optional Ad	justments to 20	20 GPCD			Did Gundling
	Enter "C	)" if Adjustment No	ot Used				Did Supplier
Actual 2020 GPCD <sup>1</sup>	Extraordinary Events <sup>1</sup>	Weather Normalization <sup>1</sup>	Economic Adjustment <sup>1</sup>	TOTAL Adjustments <sup>1</sup>	GPCD <sup>1</sup> (Adjusted if applicable)	2020 Confirmed Target GPCD <sup>1, 2</sup>	Targeted Reduction for 2020?
128	-	-	-	-	128	124	NO
<sup>1</sup> All values are	<sup>1</sup> All values are reported in GPCD						
<sup>2</sup> 2020 Confirm	ned Target GPCD	is taken from the	Supplier's SB X7	-7 Verification F	orm Table SB X7	′-7, 7-F.	
NOTES:							

SB X7-7 Table 7: 2020 Target Method Select Only One			
Targe	t Method	Supporting Documentation	
	Method 1	SB X7-7 Table 7A	
	Method 2	SB X7-7 Tables 7B, 7C, and 7D <i>Contact DWR for these tables</i>	
$\checkmark$	Method 3	SB X7-7 Table 7-E	
	Method 4	Method 4 Calculator	
NOTES:			

SB X7-7 Table 7-A: Target Method 1 20% Reduction			
10-15 Year Baseline GPCE	2020 Target GPCD		
149	120		
NOTES:			

Target Landscape

Tables for Target Method 2 (SB X7-7 Tables 7-B, 7-C, and 7-D) are not included in the SB X7-7 Verification Form, but are still required for water suppliers using Target Method 2. These water suppliers should contact Gwen Huff at (916) 651-9672 or gwen.huff@water.ca.gov

### SB X7-7 Table 7-C: Target Method 2 Target CII Water Use

Tables for Target Method 2 (SB X7-7 Tables 7-B, 7-C, and 7-D) are not included in the SB X7-7 Verification Form, but are still required for water suppliers using Target Method 2. These water suppliers should contact Gwen Huff at (916) 651-9672 or gwen.huff@water.ca.gov

### SB X7-7 Table 7-D: Target Method 2 Summary

Tables for Target Method 2 (SB X7-7 Tables 7-B, 7-C, and 7-D) are not included in the SB X7-7 Verification Form, but are still required for water suppliers using Target Method 2. These water suppliers should contact Gwen Huff at (916) 651-9672 or gwen.huff@water.ca.gov

SB X7-7 Table 7-E: Target Method 3					
Agency May Select More Than One as Applicable	Percentage of Service Area in This Hydrological Region	Hydrologic Region	"2020 Plan" Regional Targets	Method 3 Regional Targets (95%)	
		North Coast	137	130	
		North Lahontan	173	164	
		Sacramento River	176	167	
$\checkmark$	100%	San Francisco Bay	131	124	
		San Joaquin River	174	165	
		Central Coast	123	117	
		Tulare Lake	188	179	
		South Lahontan	170	162	
		South Coast	149	142	
		Colorado River	211	200	
<b>Target</b> (If more than one region is selected, this value is calculated.)				124	
NOTES:					

SB X7-7 Table 7-F: Confirm Minimum Reduction for 2020 Target					
5 Year Baseline GPCD From SB X7-7 Table 5	Maximum 2020 Target*	Calculated 2020 Target Fm Appropriate Target Table	Confirmed 2020 Target		
148	140	124	124		
* Maximum 2020 Target is 95% of the 5 Year Baseline GPCD					
NOTES:					

Appendices 2020 Urban Water Management Plan Marin Municipal Water District



## Appendix G

### Board Policy No. 2 Recycled Water



BOARD POLICY

No.: \_\_\_\_

Date: 4/26/78

Revised 10/31/90 Revised 2/23/94 Revised 5/21/97

Subject:

### RECYCLED WATER

The Marin Municipal Water District wishes to encourage the maximum water reclamation of sanitary effluent for all beneficial uses. The development and operation of water reclamation facilities is consistent with the goals of the Marin Countywide Plan, and is determined to be compatible with the development of other water sources and the operation of a potable water system.

The following are offered to show the extent to which the District supports the development of water reclamation:

- The Marin Municipal Water District recognizes water reclamation as an additional water source and an integral part of its water supply.
- 2. The District will from time to time determine the points, quantities, and rates of flow at which it will deliver recycled water for use by its customers. These determinations shall be made solely by the District on the basis of availability of dependable supply of recycled water, the feasibility of the distribution thereof to the point of delivery, and the water requirements of the customer.
- 3. In the discretion of the District, certain areas in and around a water reclamation facility may be designated as "recycled water use areas" which may require, as a condition of water service, the use of recycled water for irrigation, indoor water use, and other types of non-potable use.
- 4. The Water District shall pay all costs associated with the financing of reclamation treatment plants, transmission and distribution systems, excepting service installation charges and connection fees, which shall be determined and levied as follows:

Existing Consumers. Consumers within a "recycled water use area" who have installed an irrigation service prior to designation of said use area, or who have a possible recycled water use which can be separated from any potable use, may, at the discretion of the District, be required to connect to the recycled water system by assuring their onsite system is properly plumbed, but without charge or fee of any nature from the District.



BOARD POLICY			
No.:	2		
Page:	2		

<u>New Consumers</u>. New consumers requesting service within a "recycled water use area" may be required, at the discretion of the District, to connect an irrigation service, separate indoor water use service, or other type of nonpotable use service to the recycled water system. In such case, the District may charge said consumer the actual costs of service installation, together with the connection fee established for new uses.

5. In order to encourage the use of recycled water for appropriate and beneficial purposes, the rates to be charged for recycled water are less than the potable rates and are set forth in District Code Section 6.01.070. Appendices 2020 Urban Water Management Plan Marin Municipal Water District



# Appendix H

## Water Shortage Contingency Plan



# Water Shortage Contingency Plan 2020 Update

## Marin Municipal Water District

June 2021

DRAFT - May 2021



### TABLE OF CONTENTS

1.	INTRO	DUCTION	4	
1	WATE	R SUPPLY RELIABILITY ANALYSIS	5	
2	PRIOR	DROUGHT ACTIONS	6	
3	ANNU	AL WATER SUPPLY AND DEMAND ASSESSMENT PROCEDURES	7	
4	WATE	R SHORTAGE LEVELS	12	
5	SHORTAGE RESPONSE ACTIONS			
	5.1	Prohibitions on End Users	14	
	5.2	Variances to Dry Period Regulations	24	
	5.3	District Actions	25	
	5.4	Penalties, Charges, and Other Enforcement of Prohibitions	26	
	5.5	Defining Water Features	27	
	5.6	Supply Augmentation	27	
	5.7	Shortage Response Action Effectiveness	31	
	5.8	Catastrophic Supply Interruption	36	
6	SEISM	IC RISK ASSESSMENT	37	
7	COMN	/UNICATION PROTOCOLS	39	
8	COMP	LIANCE AND ENFORCEMENT	41	
9	LEGAL	AUTHORITIES	42	
10	FINAN	ICIAL CONSEQUENCES OF WSCP	44	
11	MONI	TORING AND REPORTING	45	
12	WSCP	REFINEMENT PROCEDURES	46	
13	PLAN ADOPTION, SUBMITTAL, AND AVAILABILITY			



### TABLES

- Table 3-1 Annual Supply-Demand Assessment Procedures Decision-Making Timeline
- Table 4-1
   Water Shortage Contingency Plan Levels (DWR Table 8-1)
- Table 5-1 Demand Reduction Actions (DWR Table 8-2)
- Table 5-2 Supply Augmentation and Other Actions (DWR Table 8-3)
- Table 5-3 Baseline Residential Per Capita Water Demand
- Table 5-4 Baseline Water Use Profile

### ATTACHMENTS

- Attachment 1. Sonoma County Water Agency Annual Water Supply and Demand Assessment Procedures
- Attachment 2. Drought Response Tool Quantitative Assessment
- Attachment 3. Water Shortage Contingency Plan Resolutions



### **1** INTRODUCTION

#### ☑ CWC § 10640

(a) Every urban water supplier required to prepare a plan pursuant to this part shall prepare its plan pursuant to Article 2 (commencing with Section 10630). The supplier shall likewise periodically review the plan as required by Section 10621, and any amendments or changes required as a result of that review shall be adopted pursuant to this article.

(b) Every urban water supplier required to prepare a water shortage contingency plan shall prepare a water shortage contingency plan pursuant to Section 10632. The supplier shall likewise periodically review the water shortage contingency plan as required by paragraph (10) of subdivision (a) of Section 10632 and any amendments or changes required as a result of that review shall be adopted pursuant to this article.

Marin Municipal Water District's (MMWD's or District's) Water Shortage Contingency Plan (WSCP) has been developed to serve as a flexible framework of planned response measures to mitigate future water supply shortages. This WSCP builds upon and supersedes the WSCP that was presented in the 2015 Urban Water Management Plan (UWMP). The WSCP includes the stages of response to a water shortage caused by drought or by supply interruptions caused by infrastructure failure, regulatory mandate, or catastrophic human-caused or natural events. The primary objective of the WSCP is to ensure that the District has in place the necessary resources and management responses needed to protect health and human safety, minimize economic disruption, and preserve environmental and community assets during water supply shortages and interruptions. The WSCP also includes procedures to conduct an annual assessment of water supply and demand in order to determine whether water shortage conditions are likely to exist in the forthcoming year, and to proactively begin the process of implementing WSCP stages of action, as appropriate.

This WSCP has been prepared in accordance with California Water Code (CWC) § 10640 and CWC § 10632 of the UWMP Act. Text from the UWMP Act has been included in grey text boxes with italicized font at beginning of relevant sections of this WSCP. The information presented in the respective WSCP sections and the associated text and tables are collectively intended to fulfill the requirements of that sub-section of the UWMP Act.



### 2 WATER SUPPLY RELIABILITY ANALYSIS

### *CWC* § 10632 (a) (1) The analysis of water supply reliability conducted pursuant to Section 10635.

Assessment of water supply reliability is dependent upon a number of factors, such as: the sources of water, regulatory and legal constraints, hydrological and environmental conditions, projected climate change impacts, and expected growth, among others. Based on historical data, water supply modeling, projections of future water uses, imported water, and recycled water availability, an assessment was conducted for determining the reliability of future water supply. Based on the service reliability analysis, the District is expected to have adequate water supplies during normal years, single dry years, and multiple dry years to meet projected demands through 2045.

A Drought Risk Assessment was also conducted during the water supply reliability assessment, which evaluates the effects on available water supply sources of an assumed five-year drought commencing the year after the assessment is completed (i.e., from 2021 through 2025). Based on the Drought Risk Assessment, the District is expected to have sufficient water supply from 2021 to 2025 in this multi-year drought scenario, although, as described in this WSCP, there are a number of actions that the District will implement to reduce demands and further ensure supply reliability at various levels of water shortage.



### **3** PRIOR DROUGHT ACTIONS

The District has historically developed different strategies for reducing water demand during water shortages. The District's actions in response to the recent severe drought that occurred in California between 2014 and 2017 are discussed below.

On 1 April 2015, Governor Brown issued the fourth in a series of Executive Orders regarding actions necessary to address California's severe drought conditions. Executive Order B-29-15 directed the State Water Resources Control Board (SWRCB) to impose the first ever mandatory restrictions on urban water suppliers to achieve a statewide 25% reduction in potable urban water usage through February 2016. The Executive Order also required commercial, industrial, and institutional (CII) users to implement water efficiency measures, prohibited irrigation with potable water of ornamental turf in public street medians, and prohibited irrigation with potable water outside newly constructed homes and buildings that is not delivered by drip or microspray systems, along with numerous other directives.

On 5 May 2015, the SWRCB adopted Resolution 2015-0032 that mandated minimum actions by water suppliers and their customers to conserve water supplies into 2016 and assigned a mandatory water conservation savings goal to each water supplier based on a measurement of their residential water use in gallons per capita per day (R-GPCD). The Office of Administrative Law approved the regulations and modified the CWC on 18 May 2015. On 2 February 2016, the SWRCB voted to extend the emergency regulations until October 2016 with some modifications. On 9 May 2016, the Governor issued Executive Order B-37-16, which directed the SWRCB to extend the emergency regulations through the end of January 2017 as well as make certain water use restrictions permanent. On 18 May 2016, the SWRCB adopted Resolution 2016-0029 that adjusted the water conservation savings goal and replaced the February 2016 emergency regulation. The SWRCB may take separate action to make some of the requirements of the regulations permanent in response to the Executive Order.

The mandatory conservation standards included in CWC § 865(c) ranged from 8% for suppliers with an R-GPCD below 65 R-GPCD, up to 36% for suppliers with an R-GPCD of greater than 215 GPCD. As with previous emergency drought regulations adopted by the SWRCB in 2014, the new water conservation regulation was primarily intended to reduce outdoor urban water use. Based on their R-GPCD, the District was required to reduce water use by 20% relative to its 2013 water use.

Through enactment of its WSCP, the District surpassed these reduction targets. During the June 2015 through May 2016 compliance period, the District surpassed its water use reduction target with a cumulative savings of 21% relative to its 2013 use.

In June 2016, the District adopted its 2015 UWMP and associated WSCP update. In April 2017, Governor Brown ended the drought State of Emergency.



### 4 ANNUAL WATER SUPPLY AND DEMAND ASSESSMENT PROCEDURES

#### ☑ CWC § 10632 (a) (2)

The procedures used in conducting an annual water supply and demand assessment that include, at a minimum, both of the following:

(A) The written decision making process that an urban water supplier will use each year to determine its water supply reliability.

(B) The key data inputs and assessment methodology used to evaluate the urban water supplier's water supply reliability for the current year and one dry year, including all of the following:

(i) Current year unconstrained demand, considering weather, growth, and other influencing factors, such as policies to manage current supplies to meet demand objectives in future years, as applicable.

(ii) Current year available supply, considering hydrological and regulatory conditions in the current year and one dry year. The annual supply and demand assessment may consider more than one dry year solely at the discretion of the urban water supplier.

(iii) Existing infrastructure capabilities and plausible constraints.

(iv) A defined set of locally applicable evaluation criteria that are consistently relied upon for each annual water supply and demand assessment.

(v) A description and quantification of each source of water supply.

#### ☑ CWC § 10632.1

An urban water supplier shall conduct an annual water supply and demand assessment pursuant to subdivision (a) of Section 10632 and, on or before July 1 of each year, submit an annual water shortage assessment report to the department with information for anticipated shortage, triggered shortage response actions, compliance and enforcement actions, and communication actions consistent with the supplier's water shortage contingency plan. An urban water supplier that relies on imported water from the State Water Project or the Bureau of Reclamation shall submit its annual water supply and demand assessment within 14 days of receiving its final allocations, or by July 1 of each year, whichever is later.

#### **☑** CWC § 10632.2

An urban water supplier shall follow, where feasible and appropriate, the prescribed procedures and implement determined shortage response actions in its water shortage contingency plan, as identified in subdivision (a) of Section 10632, or reasonable alternative actions, provided that descriptions of the alternative actions are submitted with the annual water shortage assessment report pursuant to Section 10632.1. Nothing in this section prohibits an urban water supplier from taking actions not specified in its water shortage contingency plan, if needed, without having to formally amend its urban water management plan or water shortage contingency plan.

On an annual basis, the District will conduct a Supply-Demand Assessment (SDA) to identify whether there is likely to be a water shortage condition in the following year. Because a substantial portion of the District's potable water supply is from SCWA, the evaluation of SCWA supplies for a particular year will be based on information provided by SCWA.

For purposes of this assessment, a water shortage condition is defined as an anticipated shortfall of 20%, corresponding to Water Shortage Level 2. Each element of the SDA is described below, along with the key

### 2020 Updates Water Shortage Contingency Plan Marin Municipal Water District



data inputs and methodologies for determining these elements, and expected timing of the decision process.

1. Evaluation Criteria

The evaluation criteria that will be used to identify whether the District is likely to experience a water shortage in the coming year include:

SCWA Available Supply – SCWA will develop and present the draft annual assessment to the Technical Advisory Committee (TAC) at the April meeting. The final annual assessment will be presented in the June TAC meeting. The District is a member of the TAC and the Water Advisory Committee (WAC) that represents the major cities and water districts that receive water delivered by the Sonoma Water aqueduct system. The District will conduct the Annual Assessment regarding the SCWA available supply as part of a coordinated effort led by SCWA.

Further details about the evaluation criteria and procedure used by SCWA in conducting an Annual Assessment could be found in **Attachment 1** of this WSCP. As discussed in **Attachment 1**, evaluation criteria used by SCWA include:

- Unconstrained customer demand for each of SCWA's wholesale customers, considering weather, growth, and other influencing factors;
- Russian River operations, including current reservoir releases from Lake Sonoma and Mendocino and anticipated releases to meet in-stream flow requirements and water demand;
- Hydrology and watershed conditions, including Lake Sonoma and Lake Mendocino cumulative inflows and storage levels, soil moisture, and snowpack; and
- Potter Valley Project inflows, including Lake Pillsbury storage levels and observed and projected project transfers.
- Local Surface Water Available Supply The District keeps real time data on all local reservoir water levels with periodic evaluations of reservoir capacity through bathymetric studies. The last bathymetric study was completed in April 2016.
- Rainfall Data Rainfall data is maintained by the District through calibrated rain gauges located on District's watershed lands. There are two principal rain gauge locations: one located at Lake Lagunitas for which the District has 142 years of rainfall data, and the other located at the Kent Lake stream release that is used for determination of dry or normal year conditions for stream releases to Lagunitas Creek. The gauges are manually read by District staff. Unofficially, the District currently has the ability to record these rain gauge information via an installed SCADA system. After a couple years of comparing the manual reads to the SCADA data, if the two results are close enough then the District will solely collect the rain data via SCADA.



- Local Regulatory Conditions The District and North Marin Water District entered into the Interconnection Agreement in 2014. The purpose of the agreement is to provide the basis for and mechanism whereby Marin Municipal and North Marin can each, as a first priority, continue to meet the needs of their respective customers, and simultaneously, to the extent feasible, utilize their respective water systems and surplus system capacity in a coordinated manner for the benefit of the customers of both Districts. Section 3 C of the agreement governs the release by Marin Municipal for the benefit of North Marin as follows: "On North Marin's request, when flow in Lagunitas Creek is determined by North Marin to be low or when the dry year condition described in State Water Resources Control Board Order 95-17 occurs, Marin Municipal shall release to Nicasio Creek or Lagunitas Creek for North Marin's benefit up to 250 acre-feet per year. Said water shall be used by North Marin within its Inverness Point Reyes Annexation Territory." Since the agreement was signed North Marin has not requested release volumes of any significance.
- State Regulatory Conditions Evaluation of any state-mandated drought or water use restrictions known during preparation of the SDA. Water Rights Order 95-17 defines dry year conditions which impact the regulated volume of stream releases from Kent Lake.

These criteria will be assessed by District staff with detailed knowledge of District operations. The data used to support these assessments may include, but are not limited to, SCWA's Annual Water Supply and Demand Assessment, and system demand.

2. Water Supply

On the basis of the evaluation criteria above and available supporting information, the District will quantify the projected available supply over the forthcoming year. This quantification will likely be a range, and subject to revision as new data are available and as conditions evolve.

#### 3. Unconstrained Customer Demand

Unconstrained customer demands (i.e., the expected water use in the absence of shortage-caused reductions in water use) will be evaluated and estimated for the forthcoming year based on:

- A comparison of monthly customer demands relative to prior years (e.g., last 3 years),
- Evaluation of current and anticipated weather conditions,
- New demands anticipated during the coming year (e.g., new accounts coming online), and
- Any other potentially pertinent factors identified by the District (e.g., pandemic-related stay-at-home orders).
- 4. Planned Water Use for Current Year Considering Dry Subsequent Year

The District will compare the estimated unconstrained demands to the anticipated supplies for the current year, assuming that the following year will be dry (i.e., a 20% supply shortfall), using the Evaluation Criteria identified above.

5. Infrastructure Considerations


The District will evaluate how infrastructure capabilities and constraints may affect its ability to deliver supplies to meet expected customer water demands in the coming year. The constraints and capabilities are expected to include, among other things:

- Anticipated capital projects and upgrades, and
- Anticipated maintenance and repairs.
- 6. Team Members and Decision Makers

Team members include for the assessment include:

- District's Operations Director,
- District's Operations Department,
- District's Conservation Department,
- District's Watershed Department,
- District's Engineering Planning Department,
- District's Public Outreach Department,
- District's Finance Department, and
- District's Legal Counsel.
- 7. <u>Timeline</u>

## Table 4-1 Annual Supply-Demand Assessment Procedures Decision-Making Timeline

Decision-Making Process	Responsible Parties	Start Date	End Date
Convening the team members if shortage level trigger is hit	Operations Director convenes the Drought Task Force including Operations, Engineering, Conservation, Watershed, Public Outreach, Finance, and Legal Counsel	1 Jan	
Using WSCP to activate the appropriate protocols	Drought Task Force	1 Jan	
Preparing annual water shortage assessment report	Drought Task Force	1 Jan	1 June
Implementing WSCP actions as approved	Drought Task Force	1 Jan	1 June

## 2020 Updates

# Water Shortage Contingency Plan



# **Marin Municipal Water District**

Decision-Making Process	Responsible Parties	Start Date	End Date
Sending final annual water shortage assessment report to the State	Operations and Planning Departments	1 Jan	No later than July 1 <sup>st</sup> of each year beginning in 2022
Determining water supplies by source for the current year	Operations and Planning Departments	1 February	
Obtain Draft Assessment from SCWA	SCWA	April	April
Provide Comments on SCWA Draft Assessment	MMWD	April	April
Calculating the water supply reliability using spreadsheet, computer model, or other method	Operations and Planning Departments	1 April	
Determining shortages and response actions	Drought Task Force	1 April	
Preparing and presenting preliminary report to District Board	Drought Task Force	30 April	
Preparing decision- making documents for approval	Drought Task Force	1 Jan	30 April
Obtain Final Annual Assessment from SCWA	SCWA	May/June	May/June
Updating assessment based on final water supplies	Drought Task Force	1 May	
NOTES:			

Consistent with California Water Code (CWC) § 10632.1, the District will perform and submit an SDA to DWR by July 1<sup>st</sup> of each year beginning in 2022.



## 5 WATER SHORTAGE LEVELS

#### ☑ CWC § 10632 (a) (3)

(A) Six standard water shortage levels corresponding to progressive ranges of up to 10, 20, 30, 40, and 50 percent shortages and greater than 50 percent shortage. Urban water suppliers shall define these shortage levels based on the suppliers' water supply conditions, including percentage reductions in water supply, changes in groundwater levels, changes in surface elevation or level of subsidence, or other changes in hydrological or other local conditions indicative of the water supply available for use. Shortage levels shall also apply to catastrophic interruption of water supplies, including, but not limited to, a regional power outage, an earthquake, and other potential emergency events.

(B) An urban water supplier with an existing water shortage contingency plan that uses different water shortage levels may comply with the requirement in subparagraph (A) by developing and including a cross-reference relating its existing categories to the six standard water shortage levels.

In 1999, the District developed a rationing plan (Title 13 sections 13.020.30-13.02.040), with updates in 2011, 2014, and 2015. The District's prior WSCP included three triggers at 10%, 25%, and 50% rationing levels. This new WSCP includes six triggers selected because they provide the District more flexibility in addressing dry periods earlier. Developed through the GoldSim modeling effort, the triggers allow the District to successfully manage supplies through a Six-Year Severe Drought and are designed to reduce the likelihood of a water shortage that will negatively affect customers. Table 5-1 shows the six stages of water shortage currently used by the District, including a 10 and 20% voluntary rationing and mandatory rationing levels set at 30%, 40%, and 50%. Mandatory rationing limited to human health and sanitation is included for extreme Stage 6 water shortage emergencies. The water rationing stages are linked to accumulated rainfall and the amount of water in the District's reservoirs as shown in Table 5-1. The 10 and 20% voluntary rationing stage are triggered when rainfall is 30% below average for the water year<sup>1</sup> as of April 1<sup>st</sup> and storage is in the vicinity of 60,000AF (Stage 1) and/or total reservoir storage is in the vicinity of 45,000 AF on January 1<sup>st</sup> (Stage 2). The 30% mandatory rationing stage (Stage 3) is triggered when total reservoir storage is in the vicinity of 50,000 AF on February 1<sup>st</sup>. The 40% mandatory rationing stage (Stage 4) is triggered when total reservoir storage is in the vicinity of 55,000 AF on April 30<sup>th</sup> and/or storage level projections using average rainfall indicate December 1<sup>st</sup> storage in vicinity of 30,000 AF. The 50% mandatory rationing stage (Stage 5) is triggered when total reservoir storage on December 1 is less than of 30,000 AF. Mandatory rationing limited to strictly human health and sanitation (Stage 6) is triggered when drought conditions persist beyond the Critical Stage. Each stage of action is described in further detail in Sections 6.1 through 6.8 below.

## 2020 Updates

Water Shortage Contingency Plan





Table 5-1	Water Shortage Contingency Plan Levels (DWR Table 8-1)
Table J-1	water Shortage Contingency Flan Levels (Dwit Table 6-1)

Shortage Level	Percent Shortage Range	Shortage Response Actions
0	0%	<ul> <li>Includes water waste prohibitions effective at all times.</li> </ul>
1	Up to 10%	<ul> <li>If rainfall is 30% below average for the water year<sup>1</sup> as of April 1<sup>st</sup></li> <li>Includes implementation of mandatory restrictions on end uses (see Table 6-1) as well as agency actions (see Table 6-2).</li> </ul>
2	Up to 20%	<ul> <li>Total reservoir storage is in the vicinity of 45,000 acre-feet on January 1<sup>st</sup>.</li> <li>Includes implementation of mandatory restrictions on end uses (see Table 6-1) as well as agency actions (see Table 6-2).</li> </ul>
3	Up to 30%	<ul> <li>Total reservoir storage is in the vicinity of 50,000 AF on February 1<sup>st</sup></li> <li>Includes implementation of mandatory restrictions on end uses (see Table 6-1) as well as agency actions (see Table 6-2).</li> </ul>
4	Up to 40%	<ul> <li>Total reservoir storage is in the vicinity of 55,000 AF on April 30<sup>th</sup> and/or storage level projections using average rainfall indicate December 1<sup>st</sup> storage in vicinity of 30,000 AF.</li> <li>Includes implementation of mandatory restrictions on end uses (see <b>Table 6-1</b>) as well as agency actions (see <b>Table 6-2</b>).</li> </ul>
5	Up to 50%	<ul> <li>Total reservoir storage on December 1 is less than 30,000 AF.</li> <li>Includes implementation of mandatory restrictions on end uses (see Table 6-1) as well as agency actions (see Table 6-2).</li> </ul>
6	>50%	<ul> <li>Continued drought and forecasted decreasing storage levels or catastrophic event beyond Stage 5, or total reservoir storage is projected to be in the vicinity of 30,000 AF on April 1st.</li> <li>Includes implementation of mandatory restrictions on end uses (see Table 6-1) as well as agency actions (see Table 6-2).</li> </ul>
NOTES: 1. 30	% below average range for the	ainfall is defined as less than 32 inches of rainfall at the
Labarnitas i ai	man gauge for the	



## 6 SHORTAGE RESPONSE ACTIONS

#### ☑ CWC § 10632 (a) (4)

Shortage response actions that align with the defined shortage levels and include, at a minimum, all of the following:

(A) Locally appropriate supply augmentation actions.

(B) Locally appropriate demand reduction actions to adequately respond to shortages.

(C) Locally appropriate operational changes.

(D) Additional, mandatory prohibitions against specific water use practices that are in addition to state-mandated prohibitions and appropriate to the local conditions.

(E) For each action, an estimate of the extent to which the gap between supplies and demand will be reduced by implementation of the action.

#### ☑ CWC § 10632 (b)

For purposes of developing the water shortage contingency plan pursuant to subdivision (a), an urban water supplier shall analyze and define water features that are artificially supplied with water, including ponds, lakes, waterfalls, and fountains, separately from swimming pools and spas, as defined in subdivision (a) of Section 115921 of the Health and Safety Code.

This section describes the response actions the District will take to deal with the shortages associated with each of the six stages enumerated in Section 5.

#### 6.1 **Prohibitions on End Users**

The District has a number of restrictions and prohibitions that it implements during periods of rationing (i.e., water shortage stages). Additionally, the District implements on-going prohibitions to reduce baseline water waste. **Table 6-1** below identifies these prohibitions, the water shortage stage(s) at which they are implemented, and whether a penalty, charge, or other enforcement mechanism is applied for violations of these prohibitions. The water shortage stages of action are discussed further in Section 5 and the penalties, charges, and enforcement are discussed in Section 6.4. These prohibitions and enforcement actions have been adopted and are codified in the District's Title 13 Sections 13.02, 13.03, and 13.04.



Shortage Level	Demand Reduction Actions	How much is this going to reduce the shortage gap? (a)	Additional Explanation or Reference (optional)	Penalty, Charge, or Other Enforcement?
Landscape Irri	gation			
1, 2, 3, 4, 5, 6	Other landscape restriction or prohibition	5%	The application of potable water to outdoor landscapes during and within 48 hours after measurable rainfall is prohibited.	Y
1, 2, 3, 4, 5, 6	Restrict or prohibit runoff from landscape irrigation	5%	Irrigation shall not be conducted in a manner or to an extent that allows water to run off or overspray the areas being watered. Every customer is required to have his or her water distribution lines and facilities under control at all times to avoid water waste.	Y



Shortage Level	Demand Reduction Actions	How much is this going to reduce the shortage gap? (a)	Additional Explanation or Reference (optional)	Penalty, Charge, or Other Enforcement?
1, 2, 3, 4, 5, 6	Limit landscape irrigation to specific times	5%	Any landscape irrigation between the hours of 9:00 a.m. and 7:00 p.m. is prohibited. Necessary testing and repair of irrigation systems for the purpose of eliminating water waste is permitted during the hours of 9:00 a.m. and 7:00 p.m. Customers shall maintain appropriate documentation of any necessary testing and repairs for these purposes. For example, this documentation may include, but not be limited to, any applicable reports, invoices, photos, videos, and/or receipts for materials and labor related to the testing and repairs.	Υ
1, 2, 3	Limit landscape irrigation to specific days	5%	Operating outdoor irrigation systems using potable water for the purpose of irrigating ornamental landscape areas or turf areas more than three days, within any calendar week is prohibited. For the purposes of this section, "calendar week" shall mean a period running from Monday- Sunday.	Y



Shortage Level	Demand Reduction Actions	How much is this going to reduce the shortage gap? (a)	Additional Explanation or Reference (optional)	Penalty, Charge, or Other Enforcement?
3, 4, 5, 6	Limit landscape irrigation to specific days	25%	Increase restrictions on irrigation. For example, operating outdoor irrigation systems using potable water for the purpose of irrigating with overhead spray more than two days, as assigned by the District, within any calendar week may be prohibited. For the purposes of this section, "calendar week" shall mean a period running from Monday- Sunday.	Y
1, 2, 3, 4, 5, 6	Other landscape restriction or prohibition	5%	Irrigating ornamental turf on public street medians is prohibited.	Υ
4, 5, 6	Prohibit certain types of landscape irrigation	35%	Golf course irrigation with potable or raw water of any areas beyond the greens and tee areas is prohibited	Y
4	Limit landscape irrigation to specific days	35%	Further increase restrictions on irrigation. For example, irrigation may be limited to one day per week, except for minimal irrigation to maintain tree health.	Y



Shortage Level	Demand Reduction Actions	How much is this going to reduce the shortage gap? (a)	Additional Explanation or Reference (optional)	Penalty, Charge, or Other Enforcement?	
5	Other landscape restriction or prohibition	45%	Further increase restrictions on irrigation. For example, all landscape irrigation may be prohibited, except minimal irrigation to maintain tree health.	Y	
6	Other	55%	Water allotments go into effect (60% of last three-year average January usage)	Υ	
6	Other	55%	The use of potable water for any purpose other than human health and sanitation is prohibited.	Y	
Restrictions on Consumer Products					
1, 2	Other	5%	The installation of reverse osmosis water purifying systems not equipped with an automatic shutoff unit is prohibited.	Y	
3, 4, 5, 6	Pools and Spas - Require covers for pools and spas	25%	Pool covers are required for all outdoor swimming pools.	Y	
Restrictions o	Restrictions on New Connections and Landscaping				
1, 2	Other	5%	(A) Single pass cooling systems for air conditioning or other cooling system applications are prohibited, unless required for health or safety reasons.	Y	



Shortage Level	Demand Reduction Actions	How much is this going to reduce the shortage gap? (a)	Additional Explanation or Reference (optional)	Penalty, Charge, or Other Enforcement?
4, 5, 6	Other	35%	No installation of new landscapes including no expansion of existing landscapes.	Y
5	Other	45%	Consider restrictions on new water service applications such that they will be granted upon the condition that water shall be used for interior purposes and landscaping shall be delayed until the District determines that Stage 5 rationing levels are no longer needed.	Y
5	Other	45%	New non-recirculating systems for conveyer car wash applications is prohibited.	Y
6	Other	55%	Potable water shall only be used for health and safety.	Y
6	Moratorium or Net Zero Demand Increase on New Connections	55%	New water service applications will not be granted.	Y
<b>Restrictions o</b>	n Commercial Operations			
1, 2, 3, 4, 5, 6	Lodging establishment must offer opt out of linen service	5%	Lodging establishments must provide patrons the option of not having towels and linen laundered daily.	Y



Shortage Level	Demand Reduction Actions	How much is this going to reduce the shortage gap? (a)	Additional Explanation or Reference (optional)	Penalty, Charge, or Other Enforcement?
1, 2, 3, 4, 5, 6	Restaurants may only serve water upon request	5%	Drinking water served upon request only.	Y
Other				
1, 2, 3, 4, 5, 6	Prohibit use of potable water for washing hard surfaces	5%	The washing of sidewalks, walkways, driveways, parking lots and all other hard surfaced areas by direct hosing Is prohibited, except as may be permitted by current regulations pertaining to urban water runoff pollution prevention as defined by the Marin County Stormwater Pollution Prevention Program and other controlling agencies, or as determined necessary by any public agency for the health and safety of the public.	Υ



Shortage Level	Demand Reduction Actions	How much is this going to reduce the shortage gap? (a)	Additional Explanation or Reference (optional)	Penalty, Charge, or Other Enforcement?
1, 2, 3, 4, 5, 6	Customers must repair leaks, breaks, and malfunctions in a timely manner	5%	The escape of water through breaks or leaks within the customer's plumbing or private distribution system for any substantial period of time within which such break or leak should reasonably have been discovered and corrected is prohibited. A period of forty-eight hours after the customer discovers such a leak or break, or receives notice from the District of such leak or break, whichever occurs first, shall constitute a "reasonable time" within which to correct such leak or break for the purposes of this section.	γ
3, 4, 5, 6	Other	25%	Gutter flooding is prohibited.	Y
4, 5, 6	Other	35%	Decorative water fountains or pools, including the refilling or make-up of any decorative fountain or pool (excludes recreational pools) is prohibited.	Y
4	Other	35%	Consider implementation of drought rates and water surcharges	N



Shortage Level	Demand Reduction Actions	How much is this going to reduce the shortage gap? (a)	Additional Explanation or Reference (optional)	Penalty, Charge, or Other Enforcement?
4	Other	35%	Begin discussion on limiting or excluding new service connections during drought	Ν
5, 6	Other landscape restriction or prohibition	45%	Filling recreational pools with potable water is prohibited	Y
5	Other	45%	Enact Drought Rates and Water Surcharges	
1, 2, 3, 4, 5, 6	Require automatic shut-off hoses	5%	Using a garden hose without a shut- off nozzle is prohibited.	Ν
1, 2, 3, 4, 5, 6	Other	5%	Any excess water runoff flowing onto the public right-of-way at a rate of one gallon per minute or greater not caused by storm water or naturally occurring groundwater is prohibited.	Y
1, 2, 3, 4, 5, 6	Other	5%	Use of private fire lines or private fire taps for any purposes other than fire suppression and necessary testing is prohibited.	Y
3, 4, 5, 6	Other	25%	Using potable water to wash vehicles except at commercial carwash facilities that use recycled water, is prohibited.	Υ
4, 5, 6	Other	35%	Powerwashing of any structure is prohibited.	Y

|--|



Shortage Level	Demand Reduction Actions	How much is this going to reduce the shortage gap? (a)	Additional Explanation or Reference (optional)	Penalty, Charge, or Other Enforcement?
4, 5, 6	Limit use of potable water for construction and dust control	35%	Dust control, compaction, sewer flushing, street cleaning, or any other use, as determined by the District, which can be met with disinfected tertiary recycled water is prohibited.	Y
4,5,6	Other	35%	Request that local fire departments limit training exercises that use potable water and cease hydrant testing.	Y
Notes: (a) The reduct	ions shown are the combined redu	uctions for all the actions ass	ociated with the particular shortage leve	I.

## Table 6-1Demand Reduction Actions (DWR Table 8-2)



## 6.2 Variances to Dry Period Regulations

The District does allow for certain variances to the water shortage stage prohibitions discussed in Section 6.1. All variance requests must be submitted in writing to the District and include the account name, service number, and service address. Per the District's Water Conservation Ordinance (Title 13) Section 13.04.030 variances may be granted to District customers as follows:

The District may grant variances for use of water otherwise prohibited under mandatory water use prohibitions if it is found and determined that:

(1) Failure to do so would cause an unnecessary and undue hardship on applicant or the public, including but not limited to, adverse economic impacts;

(2) Failure to do so would cause an emergency condition affecting the health, sanitation, fire protection or safety of the applicant or the public; or

(3) Customer is able and agrees to provide an alternative means of providing comparable water conservation.

Any request for a variance shall be submitted to the District in a writing providing sufficient detail regarding the request and the reasons therefore. After consideration of the variance request, a written decision shall be provided to the customer rejecting, partially approving or approving the variance request. If the customer disagrees with the initial determination, the customer may avail themselves of the appeal process set below.

(1) Customers may appeal a decision regarding a variance or an enforcement action by following the procedures set forth below:

(A) Within thirty (30) calendar days of the variance denial or partial denial or a notice of violation, customer shall mail a written appeal containing all applicable evidence supporting their position to the Water Efficiency Department at 220 Nellen Avenue, Corte Madera, CA 94925. For purposes of this section an appeal shall be deemed received by the District on the day of post-mark by the U.S. Postal Service.

(B) The District shall respond to the appeal in writing either denying, granting or partially granting the appeal. If customer disputes the initial written determination of his/her appeal, then customer may request a further appeal by submitting a further writing to the District within fifteen (15) calendar days from the date of the initial written response to the appeal.

(C) Upon receipt of a timely further appeal, a hearing on the appeal will be scheduled and the District will mail notice of this date to the customer at least ten (10) calendar days before the hearing.

(D)The General Manager or designee shall conduct a hearing on the appeal considering all applicable facts and issue a written decision containing his or her decision on the appeal. The General Manager's or designee's decision shall be final.



(E) Any action not timely appealed shall be deemed final.

(F) Pending receipt of a written appeal or pending hearing pursuant to an appeal, the District may take appropriate steps to prevent unauthorized use of water as appropriate to prevent waste.

(G) This notice and hearing procedure shall not apply to those water waste situations charged as misdemeanors.

## 6.3 District Actions

In addition to implementing and enforcing the prohibitions on end users discussed in 6.1, the District has identified a suite of actions and operational changes it will enact at the various stages of water shortage. These actions are meant to reduce water demands on the distribution system itself, facilitate implementation of the identified prohibitions, provide educational resources to customers, and to continue to offer customers opportunities to reduce their water use. The District actions identified for each stage of action are listed below.

#### Stage 1: Low Rainfall

- Initiate public information campaign
  - Education of on ongoing prohibitions
  - Water waste reporting (website and phone)
- Provide a list of simple things that can be done to save water (e.g., use a broom instead of hosing down a sidewalk, run full loads of dishes and laundry, etc.)
- Explain drought situation to the public and governmental bodies via "Drought Watch" to provide a snapshot of the water supply picture, restrictions, and water conservation tips
- Explain other stages and forecast future actions
- Prepare and disseminate educational brochures, bills inserts, etc.
- Notify media
- Begin advertising campaign
- Market Weekly Watering Schedule, watering day restrictions

## Stage 2: Advisory Stage

- Accelerate public information program
- Advertise 20% voluntary reduction goals
- Develop (or use existing) brief and simple educational videos that cover topics normally touched on during phone consultations
- Notify cities in service area of drought conditions to ensure collaboration
- Encourage use of graywater and weather based landscape watering
- Increase advertising
- Explain other stages and forecast future actions
- Minimize system flushing
- Initiate water waste patrols
- Publish Weekly Watering Schedule watering day restrictions



#### Stage 3: Alert Stage

- Enact 30% mandatory water use reductions
- Accelerate public information program
- Notify cities in service area of drought conditions to ensure collaboration
- Increase advertising
- Explain other stages and forecast future actions
- Increase system leak repair to prioritize class 2 leaks
- Reduce Weekly Watering Schedule to deficit irrigation
- Increase water waste patrols
- Restrict line flushing to include only regulatory compliance actions

#### Stage 4: Severe Stage

- Declare Water Shortage Emergency
- Enact 40% mandatory water use reductions
- Consider drought water rates
- Accelerate public information program
- Notify cities in service area of drought conditions to ensure collaboration
- Increase advertising
- Explain other stages and forecast future actions
- Increase water waste patrols
- Increase system leak repair to prioritize class 1-3 leaks
- Provide information about how to water trees in a drought recognizing that their preservation should be prioritized over more easily replaced plants in a landscape

#### Stage 5: Critical Stage

- Accelerate public information program.
- Enact 50% mandatory water use reduction prohibitions
- Enact drought water rates
- Notify cities in service area of drought conditions to ensure collaboration
- Increase advertising
- Explain other stages and forecast future actions
- Increase water waste patrols

#### Stage 6: Emergency Stage

- Potable water shall only be used for human health and sanitation
- New water service applications will not be granted

## 6.4 Penalties, Charges, and Other Enforcement of Prohibitions

**Table 6-1** in Section 6.1 identified the restrictions on water use by customers to be implemented during various stages of water shortage, as well as which of these items are enforceable by penalty. Any customer violating the restrictions on water use identified in **Table 6-1** shall receive a penalty as described below:



- <u>First Notice Warning Letter</u>: Any customer violating the regulations and restrictions on water use shall receive a written warning informing them of the violation for the first such violation and warning that a second such violation will result in a penalty.
- <u>Notice of Violation</u>: If, after receiving a written warning of violation for the same category of violation within a two-week period, the District shall issue a Notice of Violation imposing a \$25 fine on the customer's next water bill.
- <u>Repeat Violations</u>: For customers found by the District to incur a further violation within the same category for which customer has already received a Notice of Violation within the past sixty (60) calendar days, customer shall be charged a fine of \$250 for each successive violation noticed by the District.
- Additional Enforcement Procedures:
  - (A) Failure by the customer to correct the violation and pay the applicable fine, may cause the District to install a flow restrictor on the service. If a flow restrictor is placed, a charge of \$150 for cost of installation and an additional \$150 cost for removal shall be paid by the violator.
  - (B) Any willful violation occurring subsequent to the issuance of the third written notice of violation may constitute a misdemeanor and may be referred to the Marin County district attorney's office for prosecution.
  - (C) The District may also disconnect the water service pursuant its code. If water service is disconnected, it shall be restored only upon payment of the turn-on charge fixed by the board of directors.

## 6.5 Defining Water Features

## ☑ CWC § 10632 (b)

For purposes of developing the water shortage contingency plan pursuant to subdivision (a), an urban water supplier shall analyze and define water features that are artificially supplied with water, including ponds, lakes, waterfalls, and fountains, separately from swimming pools and spas, as defined in subdivision (a) of Section 115921 of the Health and Safety Code.

As required by CWC §10632, the District distinguishes between "decorative water features" such as ponds, lakes, and fountains that are artificially supplied with water and "recreational water features" such as swimming pools and spas. Prohibitions on water use for decorative water features are listed separately from those for recreational water features (see **Table 6-1**).

## 6.6 Supply Augmentation

**Table 6-2** below includes a list of potential water supply augmentation actions that may be implemented in the event of a water shortage. While shortage levels are indicated in the table below, prior to enacting these actions, the District will evaluate on a case-by-case basis:



- (1) the sufficiency of demand reduction actions (Table 6-1) to supply shortage conditions,
- (2) the feasibility of the implementing the action in light of regulatory, operational, and other constraints, and
- (3) the costs of implementing the action in context with the severity of the water shortage condition.



Shortage Level	Supply Augmentation Methods and Other Actions by Water Supplier	How much is this going to reduce the shortage gap?	Additional Explanation or Reference (optional)
1	Increase supplemental water imports and closely monitor storage levels and weather conditions	Up to 5,000 AF	Range dependent on hydraulic capabilities
1	Enact dry year stream release flow reductions	Up to 100 AF	As defined under Water Right Order 95-17 and agreements with NMWD
2	Minimize system flushing	Unknown	Focus on water quality improvements
3	Increase water waste patrols	Unknown	
3	Restrict line flushing to include only regulatory compliance actions	Unknown	Flush for violations of water quality regulations or required disinfection for new construction and repairs.
3	Increase system leak repair to prioritize class 2 leaks	Unknown	
3	Access Stored Emergency Supply	Unknown	Rental of generator for Soulajule Reservoir to transfer water to Nicasio Reservoir
3	Access Stored Emergency Supply	Unknown	Initiate adjustments to pipe configuration to transfer water from Phoenix Lake to Bon Tempe Lake
4	Initiate Study/Application for Temporary Urgency Change Petition	Up to 2,500 AF	Change to environmental releases – possible reduction of migration flow volumes.
4	Increase water waste patrols	Unknown	
4	Increase system leak repair to prioritize class 1-3 leaks	Unknown	
5	Access Stored Emergency Supply	500 AF	Develop access to unusable water in reservoir
5	Initiate water transfers	Unknown	Pending investigation
5	Initiate groundwater banking agreement for increased imports	Unknown	Long term

## Table 6-2Supply Augmentation and Other Actions (DWR Table 8-3)

# 2020 Updates

# Water Shortage Contingency Plan





Shortage Level	Supply Augmentation Methods and Other Actions by Water Supplier	How much is this going to reduce the shortage gap?	Additional Explanation or Reference (optional)
5	Access Emergency Water Treatment skids for desalination	Unknown	Conducting feasibility analysis
NOTES:			



## 6.7 Shortage Response Action Effectiveness

In order to evaluate and ensure that effective actions will be implemented with the proper level of intensity, the District employed the Drought Response Tool (DRT), an Excel spreadsheet model developed by EKI Environment and Water, Inc. The DRT model calculates monthly savings anticipated by implementing each stage of action as detailed below.

#### 6.7.1 Baseline Water Use Profile

Using the DRT, the District developed a pre-drought baseline water use profile that reflected usage patterns within the District's service area by major water use sector during 2019 that was used to guide development of the WSCP. Key findings from this analysis are presented below.

#### Residential Per Capita Demand

The District's baseline residential gallons per capita per day (R-GPCD) demand in 2019 was approximately 77 R-GPCD. As shown in **Table 6-3** and its associated chart, this R-GPCD is lower than the statewide average of 85 R-GPCD.

#### Proportion of Outdoor Water Use

As shown on **Table 6-4** and associated charts, outdoor water use, which can generally be considered as a "discretionary water use", was estimated to be approximately 34% of the District's potable consumption during this pre-drought time period. Dedicated irrigation meters for potable water accounted for 16% of the total potable irrigation demand. The remaining irrigation water uses within the District's service area are supplied by recycled water.

The DRT estimates indoor water use to be equivalent to the lowest monthly water use for each sector, accounting for the number of days in each month. Outdoor water use for each sector was estimated to be the difference between the total water use and the estimated indoor water use. If District customers tend to irrigate more heavily during winter months, an underestimation of the proportion of outdoor water use would occur.

The proportion of outdoor water use within the residential and commercial sectors is estimated to be 38% and 18%, respectively. This indicates that there is the potential to achieve significant potable water savings across these sectors, particularly within the residential sector, simply by focusing on outdoor uses. If the proportion of outdoor water use is being underestimated by the DRT method, then even more substantial savings may be achieved through targeting outdoor water use. As further shown in **Table 6-4** and its associated charts, the seasonal variation in baseline potable water use reflects increased irrigation demands during the summer and fall months. Therefore, the greatest potential for reductions in non-essential water use are expected during these months.



	Baseline Residential Per Capita Water Demand (R-GPCD)								
MMWD (a) 77									
Statewide Average (b) 85									
NOTES: (a) District R-GPCD calculated using 2019 metered data. (b) State-wide R-GPCD for 2019 obtained from data provided at California State Water Resources Control Board Water Conservation Portal - Conservation Reporting, http://www.waterboards.ca.gov/water_issues/programs/cons ervation_portal/conservation_reporting.shtml, accessed									

# Table 6-3 Baseline Residential Per Capita Water Demand







		Baseline (2019) Water Use													
Sector	End-Use	January	February	March	April	May	June	July	August	September	October	November	December	Annual	Annual % of Total by Sector
	Indoor	870	786	870	842	870	842	870	870	842	870	842	870	10,246	62%
Residential	Outdoor	153	138	0	0	421	407	867	867	1,038	1,072	672	694	6,327	38%
Residential	Subtotal Residential	1,023	924	870	842	1,291	1,249	1,737	1,737	1,880	1,942	1,514	1,564	16,573	
	Indoor	279	252	279	270	279	270	279	279	270	279	270	279	3,286	82%
CII	Outdoor	21	19	0	0	60	58	101	101	114	118	66	68	724	18%
CII	Subtotal CII	300	271	279	270	339	328	380	380	384	397	336	347	4,010	
Dedicated Irrigation	Outdoor	20	18	16	15	110	106	211	211	204	211	112	115	1,348	100%
Non- Revenue	Non- Revenue	165	149	143	138	213	206	285	285	303	313	240	248	2,689	100%
Total	Indoor	1,149	1,038	1,149	1,112	1,149	1,112	1,149	1,149	1,112	1,149	1,112	1,149	13,532	55%
	Outdoor	193	175	16	15	590	571	1,178	1,178	1,355	1,401	849	877	8,399	34%
	Non- Revenue	165	149	143	138	213	206	285	285	303	313	240	248	2,689	11%
	Total	1,507	1,361	1,308	1,266	1,953	1,890	2,613	2,613	2,770	2,863	2,201	2,275	24,620	
NOTES:	NOTES:														
(a) Volumes	are in units of	FAF.													

#### Table 6-4Baseline Water Use Profile

(b) Only potable water use is included herein.









## 6.7.2 Shortage Response Action Effectiveness

The DRT provides a quantitative framework that allows the District to systematically estimate the monthly and cumulative annual demand reductions expected to result from particular combinations of drought response actions and associated implementation rates. Data inputs to the DRT include total production, class-specific water use, population, and assumptions regarding the split between indoor and outdoor water use for each customer class.

For each drought response action, the user specifies:

- The customer class(es) and end use(s) that are affected;
- The percent savings for that end use for each account that implements the action. These are based on evaluations reported in the literature, or where such studies are not available, on best estimates based on the District's experience; and
- The percentage of accounts assumed to implement the action, which is presumed to be the result of the intensity level of the District's program implementation, including but not limited to, marketing and enforcement activities.

An additional critical DRT user input is a set of constraints on demand reductions to ensure that usage levels do not endanger health and safety or result in unacceptable economic impacts. The DRT will not permit estimated usage reductions to violate these constraints, regardless of the demand reduction actions selected. The constraints are:

- A minimum residential indoor per capita daily usage of 25 gallons,
- A maximum residential outdoor usage reduction of 100%,
- A maximum Commercial, industrial, and institutional (CII) indoor usage reduction of 30%, and
- A maximum CII outdoor usage reduction of 100%.

Based on the foregoing data, the DRT model calculates the resulting monthly savings. The District adjusted the combination of actions and implementation levels to achieve the targeted savings levels at each of the six stages of action.

For each stage of action, the modeling targeted the mid-range of the required demand reduction range, ergo:

- 5% for Stage 1,
- 15% for Stage 2,
- 25% for Stage 3,
- 35% for Stage 4,
- 45% for Stage 5, and
- 55% for Stage 6.

MMWD's shortage response actions are summarized in **Table 6-1** and **Table 6-2**.

The key DRT inputs and outputs for each of the stages of action are reproduced in **Attachment 2**, including the water shortage reduction actions, savings assumptions, and implementation rates that are required for the District to achieve the required annual demand reductions for each of the six stages of action. At each stage, there are two types of demand-reduction actions identified:

• Restrictions on customer water usage; and



• Consumption reduction actions by the District to encourage decreased water usage.

Many actions are implemented across a number of stages, some at increasing implementation levels. Therefore, the actions in **Table 6-1** and **Table 6-2** are listed as a row under the first stage at which they are implemented. The percentages shown in the tables represent end user savings.

## 6.8 Catastrophic Supply Interruption

Catastrophic supply interruptions may be caused by a regional power outage, an earthquake, or other disaster. In accordance with the Emergency Services Act, the District has developed an Emergency Operation Plan (EOP). This EOP guides response to unpredicted catastrophic events that might impact water delivery including regional power outages, earthquakes, or other disasters. The EOP outlines standard operating procedures for all levels of emergency, from minor accidents to major disasters. The District's EOP has been coordinated with the SCWA and neighboring water purveyors.

In addition to the EOP, the City has prepared Emergency Action Plans for each of their dams that have been filed with the Department of Safety of Dams and the Sonoma County Office of Emergency Services. These Emergency Action Plans include inundation maps as well as notification procedures and contacts with the California Office of Emergency Services to assist first responders in case of an emergency.

The District also has an Emergency Response Plan that is in the process of being updated.



## 7 SEISMIC RISK ASSESSMENT

#### ☑ CWC § 10632.5

(a) In addition to the requirements of paragraph (3) of subdivision (a) of Section 10632, beginning January 1, 2020, the plan shall include a seismic risk assessment and mitigation plan to assess the vulnerability of each of the various facilities of a water system and mitigate those vulnerabilities.

(b) An urban water supplier shall update the seismic risk assessment and mitigation plan when updating its urban water management plan as required by Section 10621.

(c) An urban water supplier may comply with this section by submitting, pursuant to Section 10644, a copy of the most recent adopted local hazard mitigation plan or multihazard mitigation plan under the federal Disaster Mitigation Act of 2000 (Public Law 106-390) if the local hazard mitigation plan or multihazard mitigation plan addresses seismic risk.

Impacts associated with earthquakes and liquefaction are discussed in the 2018 Marin County Multi-Jurisdictional Local Hazard Mitigation Plan (County LHMP; Marin County, 2018).<sup>1</sup> The MCM LHMP assesses Marin County's vulnerabilities to various hazards, including seismic hazards, and presents mitigation strategies that are planned over the next five years. While the District was not a participating member of this plan, Marin County is currently in the process updating its LHMP, using a multijurisdictional planning approach overseen by a steering committee made up of various stakeholders, including the District.

The County LHMP includes a discussion of the probability of a seismic event affecting Marin County, citing an ABAG projection of a 52% chance of an earthquake of magnitude 6.7 of greater on one of the faults affecting Marin County between now and 2036. The County LHMP notes that much of the Marin County infrastructure is located in areas of Bay Mud, as well as in current and former marshlands that have been artificially filled. These areas are vulnerable to liquefaction during seismic events. The County LHMP includes an assessment of the County's vulnerability in the event of a major seismic event, and estimates that an earthquake on the San Andreas Fault of magnitude 7.8 would result in a total building damage of approximately \$1.26 trillion.

Measures to mitigate seismic risk identified in the County LHMP include:

- Using incentives for mitigation of privately owned-at risk or seismically deficient structures;
- Adopting applicable standards and identification/assessment protocols for voluntary or mandatory retrofits of seismically vulnerable buildings;
- Upgrading deficient ramps and bridges; and
- Requiring engineered plan sets for seismic retrofits of homes on steep hillsides, soft-story homes, and other split-level or vulnerable homes.

<sup>&</sup>lt;sup>1</sup> The County LHMP could be found in the following website: <u>https://www.marinwatersheds.org/sites/default/files/2020-07/Marin%20County%20Multi-</u> <u>Jurisdictional%20Local%20Hazard%20Mitigation%20Plan%202018.pdf</u>



Further discussion of seismic risks specific to the SCWA water system is provided in the *Sonoma County Water Agency Local Hazard Mitigation Plan*, dated 16 October 2018 (SCWA LHMP; SCWA, 2018).<sup>2</sup> The SCWA LHMP specifically assesses SCWA's natural hazard risks and vulnerabilities facing the SCWA infrastructure and provides a plan of action to address these vulnerabilities. The SCWA LHMP identifies a series of mitigation measures to address seismic risk, including seismic retrofits of distribution system components to protect against damage due to liquefaction and lateral spread hazard and installation of automated throttling valves at aqueducts and interties to minimize uncontrolled releases out of SCWA facilities. For more detail regarding planned mitigation measures to address seismic risks, please refer to the SCWA LHMP.

The District also assessed the risks of an earthquake to the District's water supply system and infrastructure in the Marin Municipal Water District Water Resources Plan 2040 (MMWD, 2017).<sup>3</sup> Through use of the Marin WaterSim model, it was determined that, while several treatment plants could become un-operational for up to three months following an earthquake event, District water demands could still be met by increasing production from unaffected treatment plants. More detail regarding the earthquake risk assessment and modelling can be found in District's Water Resources Plan 2040.

<sup>&</sup>lt;sup>2</sup> The SCWA LHMP could be found in the following website: https://evogov.s3.amazonaws.com/185/media/186587.pdf

<sup>&</sup>lt;sup>3</sup> The Marin Municipal Water District Water Resources Plan 2040 could be found in the District website: <u>https://www.marinwater.org/sites/default/files/2020-09/Water%20Resources%20Plan%202040.pdf</u>



## 8 COMMUNICATION PROTOCOLS

#### **CWC** § 10632 (a) (5)

*Communication protocols and procedures to inform customers, the public, interested parties, and local, regional, and state governments, regarding, at a minimum, all of the following:* 

(A) Any current or predicted shortages as determined by the annual water supply and demand assessment described pursuant to Section 10632.1.

(B) Any shortage response actions triggered or anticipated to be triggered by the annual water supply and demand assessment described pursuant to Section 10632.1.

(C) Any other relevant communications.

Each stage of the WSCP is implemented with a formal declaration by the District Board of Directors upon the determination that SCWA or another governing authority (e.g., the SWRCB) has required a voluntary or mandatory reduction in water use due to a water supply shortage or emergency. Procedures for water shortage declaration and termination are detailed below in Section 10.

Even before formal declaration of a water shortage, a public information program will be activated to provide customers with as much advance notice as possible. Following declaration of a shortage, District customers would need to be provided notice of water shortage rules and regulations via a variety of media and communications methods. Some of these communication methods will include the following:

#### Public Outreach:

- Social media awareness campaign
- Customer bill inserts
- Customer newsletter On the Waterfront
- Drought website resources
- Yard sign distribution
- District E-newsletter
- Continued media coverage: Op-Ed and Editorial
- Public Awareness Conservation Advertising Campaign
  - Phase I: newspaper print and digital, digital display ads, transit shelters, and buses
  - o Phase II: all Phase I actions plus radio, banners, outdoor kiosks, and video PSAs

#### Targeted Outreach:

- Sector-specific call-to-action letters
- Scheduling Community Presentations
  - Homeowner/Neighborhood Associations



- City/Town Councils
- Rotary/Chamber of Commerce
- Water Waste Notification System Website and Phone
- Staff Training on drought messaging and water waste prohibitions
- Stakeholder presentations begin
- Gardening articles
- FireSafe Marin Newsletter

Coordination between the District and with other public agencies can begin prior to formal declaration of a water shortage and can be accomplished through regular meetings, e-mail group updates, and presentations. In a regional water shortage scenario, the District would use public outreach resources and materials provided by SCWA, as well as its own resources as described above.

As discussed in Chapter 9 of the District's 2020 Urban Water Management Plan (UWMP), several District staff members jointly share the responsibility for water conservation. Staff time dedicated to water conservation and enforcement action will increase with the severity of a supply shortage. Additional duties may be assigned to current employees or hiring of temporary staff may be considered to meet staffing needs during extreme water shortages.



## 9 COMPLIANCE AND ENFORCEMENT

**CWC § 10632 (a) (6)** For an urban retail water supplier, customer compliance, enforcement, appeal, and exemption procedures for triggered shortage response actions as determined pursuant to Section 10632.2.

As discussed in Section 6.4, any customer violating the regulations and restrictions on water use set forth above in **Table 6-1** shall receive a written warning for the first such violation, and then a notice of violation and a \$25 fine for a second violation, and additional penalties for repeat violations as noted in Section 6.4. Additional enforcement procedures include the following:

- Any willful violation occurring subsequent to the issuance of the third written notice of violation may constitute a misdemeanor and may be referred to the Marin County district attorney's office for prosecution.
- The District may also disconnect the water service pursuant its code. If water service is disconnected, it shall be restored only upon payment of the turn-on charge fixed by the board of directors.



#### **10 LEGAL AUTHORITIES**

#### ☑ CWC § 10632 (a) (7)

(A) A description of the legal authorities that empower the urban water supplier to implement and enforce its shortage response actions specified in paragraph (4) that may include, but are not limited to, statutory authorities, ordinances, resolutions, and contract provisions.

(B) A statement that an urban water supplier shall declare a water shortage emergency in accordance with Chapter 3 (commencing with Section 350) of Division 1.

(C) A statement that an urban water supplier shall coordinate with any city or county within which it provides water supply services for the possible proclamation of a local emergency, as defined in Section 8558 of the Government Code.

The District has the ability to declare a Water Shortage Emergency under the following authorities and conditions as described under the WSCP shortage levels:

- Article X, Section 2 of the California Constitution mandates that the water resources of the State be put to beneficial use to the fullest extent and that waste or unreasonable use or method of use of water be prevented.
- Pursuant to District Code Section 13.02.015, "Declaration of Water Shortage Emergency," when the District's reservoir storage on December 1st is projected to be in the vicinity of, or less than, 30,000 AF, the District's Board may declare by resolution a Water Shortage Emergency to indicate that it is in accordance with Water Code Chapter 3 (commencing with Section 350) of Division 1 general provision regarding water shortage emergencies.
- California Water Code Sections 350 and 71640 authorize the governing body of a municipal
  water district to find the existence or threat of a drought emergency or other threatened or
  existing water shortage, and that finding is prima facie evidence of the fact or matter so found,
  and such fact or matter shall be presumed to continue unchanged unless and until a contrary
  finding is made by the board by resolution or ordinance.
- Pursuant to California Water Code Sections 353 and 71641, the District may restrict the use of district water during the drought emergency or other water shortage condition and may prohibit the wastage of district water or the use of district water during such periods for any purpose other than household uses or other restricted uses as the District determines to be necessary.
- Pursuant to California Water Code Sections 376 and 71641 and Government Code Section 6061, the District must publish in a newspaper of general circulation any ordinance setting forth the restrictions, prohibitions, and exclusions determined to be necessary under Water Code Sections 353 and 71640 within 10 days after its adoption.
- Pursuant to Water Code sections 350 and 71640, and for per the annual water supply and demand assessment, the Board may find the existence or threat of a drought emergency or other water shortage condition.



- The District's Board may adopt mandatory restriction and prohibitions on the consumption and use of water within the service area so that the water supply can be conserved for the greater public benefit.
- Pursuant to Water Code sections 376 and 71641 and Government Code section 6061, the Board shall direct staff to publish in a newspaper of general circulation any ordinance, or a summary thereof, adopted by the Board setting forth the restrictions, prohibitions, and exclusions determined to be necessary under Water Code sections 353 and 71640 within 10 days of adoption.
- The District shall coordinate with any city or county within which it provides water supply services for the possible proclamation of a local emergency.



#### 11 FINANCIAL CONSEQUENCES OF WSCP

#### ☑ CWC § 10632 (a) (8)

A description of the financial consequences of, and responses for, drought conditions, including, but not limited to, all of the following:

(A) A description of potential revenue reductions and expense increases associated with activated shortage response actions described in paragraph (4).

(B) A description of mitigation actions needed to address revenue reductions and expense increases associated with activated shortage response actions described in paragraph (4).

(C) A description of the cost of compliance with Chapter 3.3 (commencing with Section 365) of Division 1.

Since the District bills its customers per unit volume of water consumed, the District would experience a reduction in revenue upon implementation of water conservation measure pursuant to the WSCP. The District may also experience expense increases related to increased public outreach, water waste patrols, emergency water transfers, and generators and fuel. To compensate for the expected revenue reduction caused by water conservation, the District may utilize reserves to the extent possible and has the authority to implement temporary drought rates, increasing water rates. Additionally, the District's Board of Directors may establish a water rate structure, including excess water use surcharges, that provides incentives to conserve water. Individual customers may seek a waiver of excess water use surcharges through the variance process as described in Section 6.2.

The implementation of the WSCP also results in an increase in operational costs due to costs associated with emergency transfers, emergency generators and fuel, water waste patrols, water conservation incentives, and public outreach.



# **12 MONITORING AND REPORTING**

CWC § 10632 (a) (9) For an urban retail water supplier, monitoring and reporting requirements and procedures that ensure appropriate data is collected, tracked, and analyzed for purposes of monitoring customer compliance and to meet state reporting requirements.

The District's local surface water supply and SCWA supply turnouts are all equipped with water meters. In addition, each potable water customer is metered. Non-residential landscape irrigation is metered separately from indoor use at most non-residential sites. The District reads meters on a monthly basis and is able to document both demand reductions and a typically high water use. The District contacts individual customers to resolve issues related to a typically high water use.

Pursuant to California Code of Regulations (CCR) Title 23 §991, the District reports monthly water use and production to the SWRCB.<sup>4</sup> Effective October 1, 2020, during a governor declared drought emergency or when an urban water supplier invokes a water shortage level to respond to a drought greater than 10%, each supplier is required to submit an expanded report that contains the supplier's actions and statistics in achieving planning reductions.

<sup>&</sup>lt;sup>4</sup> Water supplier monthly reports can be accessed at <u>https://www.waterboards.ca.gov/water\_issues/programs/conservation\_portal/conservation\_reporting.html</u>


### **13 WSCP REFINEMENT PROCEDURES**

**CWC § 10632 (a) (10)** Reevaluation and improvement procedures for systematically monitoring and evaluating the functionality of the water shortage contingency plan in order to ensure shortage risk tolerance is adequate and appropriate water shortage mitigation strategies are implemented as needed.

The WSCP is implemented as an adaptive management plan. The District will evaluate the need for revise its WSCP every year after performing its Annual Assessment. The evaluation will consider effective of WSCP actions and any anticipated water supply shortages assessed by the Annual Assessment. If the WSCP is revised, the District Board of Directors will adopt a new resolution adopting the revised WSCP, and if necessary, declare a water shortage level to implement.

46



### 14 PLAN ADOPTION, SUBMITTAL, AND AVAILABILITY

CWC § 10632 (c) The urban water supplier shall make available the water shortage contingency plan prepared pursuant to this article to its customers and any city or county within which it provides water supplies no later than 30 days after adoption of the water shortage contingency plan.

As described in Chapter 10 of the District's 2020 Urban Water Management Plan (UWMP), the District informed the public and the appropriate agencies of: (1) its intent to prepare a WSCP, (2) where the WSCP was available for public review, and (3) when the public hearing regarding the WSCP would be held. All notifications were completed in compliance with the stipulations of Section 6066 of the Government Code.

A copy of the adopted 2020 WSCP including any amendments will be provided to the Department of Water Resources (DWR), the California State Library, Marin County, and SCWA within 30 days of the adoption (**Attachment 3**). An electronic copy of the adopted 2020 WSCP will be submitted to the DWR using the DWR online submittal tool.

A copy of the adopted 2020 WSCP will be available for public review on the District's website within 30 days after filing the plan with DWR.



# **ATTACHMENT 1**

# SONOMA COUNTY WATER AGENCY ANNUAL WATER SUPPLY AND DEMAND ASSESSMENT PROCEDURES

### **Annual Water Supply and Demand Assessment Procedures**

This section presents the procedures that will be used by Sonoma Water to conduct an annual water supply and demand assessment (annual assessment). The annual assessment is required to be submitted annually to DWR beginning on July 1, 2022. The assessment forecasts near-term water supply conditions (12 months) to ensure shortage response actions are triggered in a timely manner. The annual assessment will provide a description and quantification of each source of Sonoma Water's water supply compared to water demands for the current year and one subsequent dry year. The following subsections describe the decision-making process and data and methodologies. Sonoma Water may modify this procedure based on its experiences that it will gain from the development of the annual assessment.

### **Decision-Making Process**

This section presents the decision-making process and timeline (see Table 1) that Sonoma Water will use each year to determine its water supply reliability. The assessment will be conducted annually and completed by July 1. Sonoma Water will conduct an annual assessment that follows the steps described below.

- Develop draft annual assessment. Sonoma Water staff will compile the draft annual assessment. The draft annual assessment will document the evaluation of water supply conditions, considering projections of the demand for Sonoma Water provided by the customers by February 1. Sonoma Water staff will start conducting the assessment prior to the January Decision 1610 trigger point and then thereafter at the middle of each month prior to the trigger point at the beginning of each month through June. Decision 1610 is described later in this assessment procedure description.
- 2. **Submit draft annual assessment report to the customers.** The draft annual assessment will be submitted and presented to the TAC ad-hoc committee at the April meeting. An initial determination will be made regarding the potential for a water shortage condition to occur.
- 3. **Receive review comments.** The customers will present their review comments including their updated demands and local supply projections at the May TAC meeting.
- 4. **Submit final annual assessment to the TAC.** The final annual assessment will be submitted and presented at the June TAC meeting. The annual assessment may be presented to the WAC. Sonoma Water will coordinate with the customers to identify if any water supply gaps exist for each customer when considering both Sonoma Water supplies and local supplies.
- 5. **Optional presentation of the annual assessment to the Board of Directors.** The annual assessment may be included in the agenda for Sonoma Water's Board of Directors regular meeting, particularly if a shortage is anticipated or if an existing shortage condition is to be ended.
- 6. **Submit annual assessment to DWR.** Sonoma Water will submit the annual assessment report to DWR by July 1 of each year.

Table 1. Annual Assessment Timeline											
Task     Aug     Sep     Oct     Nov     Dec     Jan     Feb     Mar     Apr     May     Jun										Jul	
Develop draft assessment											
Submit draft annual assessment to the customers											
Receive review comments											
Submit final annual assessment to the TAC and WAC											
Present annual assessment to the Board of Directors											
Submit annual assessment to DWR											

### **Data and Methodologies**

This section presents the key data inputs and assessment methodology that will be used to evaluate Sonoma Water's water supply. The evaluation criteria, water supply, unconstrained demand, water supply, planned water use, infrastructure considerations, and other factors are described.

**Evaluation Criteria.** Evaluation criteria are determined by the supply source conditions and factors that impact the condition of each supply source. The criteria include the key data inputs and the constraints that are imposed on the water supplies.

The key data inputs that are used by Sonoma Water staff to forecast water supply for the remainder of the current year and a subsequent dry year include the items described below.

- **Unconstrained customer demand.** Current and subsequent year unconstrained demand for each of Sonoma Water's wholesale customers considering weather, growth, and other influencing factors.
- **Russian River operations.** Current reservoir releases from Lake Sonoma and Lake Mendocino, including anticipated releases to meet in-stream flow requirements and water demands and based on reservoir curves and forecast informed reservoir operations (FIRO) decision support tools.
- **Hydrology and watershed conditions.** Lake Sonoma and Lake Mendocino cumulative inflows and storage levels, and soil moisture and snowpack.
- **Potter Valley Project inflows.** Lake Pillsbury storage levels and observed and projected project transfers. Decision 1610 contains trigger points at the first of each month from January to June to establish a hydrologic index based on cumulative inflows into Lake Pillsbury on the Eel River.

Sonoma Water's Russian River water supply is controlled and influenced by a variety of agreements and decisions. There are several constraints, requirements, and restrictions on water supply that will be considered as part of the assessment of the available water supplies, as follows.

• Lake Sonoma storage level. Minimum 100,000 ac-ft Lake Sonoma storage level and 30 percent delivery deficiency. This key constraint is described later in this section.

- Lake Mendocino storage level. Having a sufficient supply of water in Lake Mendocino in the fall is of critical importance to the salmonid species in the Russian River.
- Minimum instream flow requirements. The minimum instream flow schedule varies based on the hydrologic classifications of *Normal, Dry,* and *Critical* water supply conditions as defined in Decision 1610. Minimum instream flow requirements for the Russian River and Dry Creek are met by releases from Coyote Valley Dam and Warm Springs Dam.
- Maximum flow releases from Warm Springs and Coyote Valley Dams.
- US Army Corps of Engineers' flood control operations criteria.
- **The Russian River Biological Opinion.** The Russian River Biological Opinion places certain terms and conditions on the Sonoma Water with respect to its water supply operations.

Sonoma Water's water rights permits include a provision that requires Sonoma Water to impose a 30 percent deficiency in deliveries from the Russian River to its service area when Lake Sonoma storage levels drop below 100,000 ac-ft before July 15 of any year. This deficiency must remain in effect until "(1) storage in Lake Sonoma rises to greater than 70,000 ac-ft subsequent to December 31 after having fallen below that level, or (2) permittee has projected, to the satisfaction of the Chief, Division of Water Rights, that storage at Lake Sonoma will not fall below 70,000 ac-ft, or (3) hydrologic conditions result in sufficient flow to satisfy permittee's demands at Wohler and Mirabel Park and minimum flow requirements in the Russian River at Guerneville."

**Water Supply.** This subsection provides a brief overview of Sonoma Water's supply sources. These water supply sources will be described, and estimates made of the availability of supplies in the annual assessment. Sonoma Water's most recent Urban Water Management Plan provides a more detailed description of the water supplies.

The Russian River provides most of Sonoma Water's water supply with groundwater supply from the Santa Rosa Plain as a secondary source. Sonoma Water diverts water from the Russian River near Forestville and conveys the water via its transmission system to its customers. The surface water is subject to varying quantities of available supply based on hydrologic conditions and sometimes regulatory restrictions.

Almost all of Sonoma Water's customers have other water supplies, in addition to those provided by Sonoma Water, which include local surface water, local groundwater, and recycled water. These local supplies will not be included in the assessment. Each customer will develop its own assessment of their available supplies.

Two federal projects impound water in the Russian River watershed: the Coyote Valley Dam on the Russian River east of the City of Ukiah in Mendocino County (forming Lake Mendocino), and the Warm Springs Dam on Dry Creek (a tributary of the Russian River). The Potter Valley Project diverts water from the Eel River into the Russian River watershed.

**Unconstrained Customer Demand.** The assessment will present the current year unconstrained demands from Sonoma Water's customers, considering weather, growth, and other influencing factors. The unconstrained water demands will be provided by the customers.

**Planned Water Use for Current Year Considering Dry Subsequent Year.** The assessment will present an evaluation of the amount of anticipated water supplies for the current year as well as how the supplies will be used, while anticipating that the following year will be dry.

The annual assessment will be based on evaluating the key data inputs to determine the water supply reliability. The methodology to develop the annual assessment will follow the general approach described below.

- 1. Quantify current year water supply. The available water supply from all water supply sources will be estimated for the current year based on the data inputs, evaluation criteria, and hydrological and regulatory conditions in the current year. Sonoma Water staff will evaluate water supply conditions beginning at least mid-month prior to each of the January to June Decision 1610 trigger dates to determine whether anticipated conditions at the trigger dates warrant any actions by Sonoma Water. The projections of the water supply will be expressed as a range and based on the results of operations modeling of the Russian River system consisting of the statistical evaluation of multiple scenarios. The model is described later in this subsection. Figure 1 presents the key considerations for the assessment of Russian River supply conditions.
- 2. **Quantify subsequent year supply.** The subsequent year water supplies will be estimated by assuming a dry year. Sonoma Water may base the estimate of dry year water supplies on the historical hydrologic record or some other approach.
- 3. **Identify infrastructure constraints.** The existing infrastructure capabilities and plausible constraints as they impact Sonoma Water's ability to deliver supplies to meet expected customer water use needs in the coming year will be considered.
- 4. **Quantify unconstrained water demand.** The unconstrained water demands for all the customers will be provided by the customers.
- 5. **Compare projected water supplies to demands.** The water supplies identified in the annual assessment will represent the water demand that can be met while maintaining adequate storage in Lakes Mendocino and Sonoma.
- 6. Identify and quantify anticipated water supply shortages, if any. The forecast of water supplies in comparison to water demands will identify and quantify any anticipated water shortages. The forecast will be coordinated with the customers. Depending on the extent of the forecast shortage, the appropriate shortage stage will be selected. If the forecast is for a wet season, there would be no concerns. If the season was dry in the early wet season, there would be a potential concern and river flows and reservoir levels would be monitored more closely. Depending on the extent of precipitation in the latter portion of the wet season, the forecast could be changed to no concern or to an anticipated shortage.
- 7. Extent of water shortage. The water shortage may be caused by the requirement to reduce supplies by 30 percent based on the Lake Sonoma level. Sonoma Water may request voluntary reductions and perhaps mandatory reductions before Lake Sonoma levels reached 100,000 ac-ft by July 15in accordance with the applicable provisions of the Restructured Agreement and consistent with the defined shortage stages. If a shortage is identified, the water shortage allocation methodology will be used to allocate the reduced supply to each customer. Each of Sonoma Water's customers will develop their own annual assessments that will include estimates of their projected quantity of local water supplies.

The forecast of the amount of available water supplies will be developed by Sonoma Water using the Russian River System Model (RR ResSim). The model is used as a planning tool to simulate the effects of various climatic conditions, levels of demand, and operational criteria on the water supply available for use by Sonoma Water and others.

**Infrastructure Considerations.** The annual assessment will include an evaluation of how infrastructure capabilities and constraints may affect Sonoma Water's ability to deliver supplies to meet expected customer water use needs in the current year.

**Other Factors.** The annual assessment will describe any other locally applicable factors that could influence the amount of available water supplies.

**Summary:** D1610 contains trigger points at first of month (January – June) to establish Hydrologic Index (HI) based on cumulative inflows into Lake Pillsbury (Eel River). Sonoma Water staff evaluate water supply conditions (see below) beginning at least mid-month prior to each of the D1610 trigger dates to determine whether anticipated conditions at trigger date warrant any actions by SW.

**Process:** Mid-month evaluate water supply conditions relative to D1610 triggers to set HI at first of following month to determine which scenario applies:

No Concerns -Re-evaluate middle of next month **Potential Concerns** – Close monitoring. Consider water conservation messaging program Anticipated Shortages – Submit TUCP to SWRCB & initiate water conservation messaging program

**Evaluation of Water Supply Conditions:** 

• Potter Valley Project Operations: Lake Pillsbury storage levels, observed & projected project transfers

- Russian River Operations: Current release & minimum in-stream flows, water demands
- Hydrology & Watershed Conditions: Cumulative inflows, storage levels, soil moisture, snowpack
- Meteorology: Cumulative rainfall, near-term and long-term forecasts

Figure 1. Assessment of Russian River Water Supply Conditions



# **ATTACHMENT 2**

# DROUGHT RESPONSE TOOL QUANTITATIVE ASSESSMENT



### 1 - Home Marin Municipal Water District

Enter Agency	Information
Agency Name	Marin Municipal Water District
Total Population Served	192,138
Conservation Goal (%)	5%
Drought Stage	Stage 1
Number of Residential Accounts	55,390
Number of Commercial, Industrial, and Institutional (CII) Accounts	3,461
Number of Dedicated Irrigation Accounts	858
Baseline Year(s)	2019
Percentage of Residential Indoor Use During Minimum Month (%)	100%
Percentage of CII Indoor Use During Minimum Month (%)	100%
Comments	

Navigation						
USER'S GUIDE	Download and read the guide before using this Tool					
1 - HOME	Enter agency information					
2 - INPUT BASELINE YEAR WATER USE	Enter Baseline Year production and use					
3 - BASELINE YEAR WATER USE	Review and confirm entered information					
4 - DROUGHT RESPONSE ACTIONS	Select Drought Response Actions and input estimated water savings and implementation rates.					
5 - ESTIMATED WATER SAVINGS	Review estimated water production and compare estimated savings to conservation target.					
6 - DROUGHT RESPONSE TRACKING	Track production and water savings against the conservation target.					



1 - Home Marin Municipal Water District

For questions about this tool or for additional information, contact:

Anona Dutton, P.G., C.Hg. <u>adutton@ekiconsult.com</u> (650) 292-9100

Disclaimer: This electronic file is being provided by EKI Environment & Water Inc. (EKI; fomerly Erler & Kalinowski, Inc.) at the request of (CLIENT). The Drought Response Tool was transmitted to CLIENT in electronic format, on a CD dated [DATE] (Original Document). Only the Original Document, provided to, and for the sole benefit of, CLIENT constitutes EKI's professional work product. An electronic copy of the Drought Response Tool is provided to CLIENT's Customer Agencies, for use only by CLIENT-designated Customer Agencies. The Drought Response Tool is copyrighted by EKI. All rights are reserved by EKI, and content may not be reproduced, downloaded, disseminated, published, or transferred in any form or by any means, except with the prior written permission of EKI. Customer Agencies may use the Drought Response Tool for reviewing potential drought response alternatives. The delivery to, or use by, Customer Agencies of the Drought Response Tool does not provide rights of reliance by Client Agencies or other third parties without the express written consent of EKI and subject to the execution of an agreement between such Customer Agency or other third party and EKI. EKI makes no warranties, either express or implied, of the electronic media or regarding its merchantability, applicability, compatibility with the recipients' computer equipment or software; of the fitness for any particular purpose; or that the electronic media contains no defect or is virus free. Use of EKI's Drought Response Tool, other electronic media, or other work product by Client Agency or others shall be at the party's sole risk. Further, by use of this electronic media, the user agrees, to the fullest extent permitted by law, to defend, indemnify and hold harmless EKI, CLIENT, and their officers, directors, employees, and subconsultants against all damages, liabilities or costs, including reasonable attorneys' fees and defense costs, arising from any use, modification or changes made to the electronic files by anyone other than EKI or from any unauthorized distribution or reuse of the electronic files without the prior written consent of EKI.

© 2015 Erler & Kalinowski, Inc.

*Worksheet 1 - Home* Page 2 of 11 Date Printed: 5/25/2021

KI Drought Response Tool									
lome	Input Baseline Yea Water Use	r Baselin Us	e Year Water e Profile	Drought Re Actior	sponse is	Estimated Wate Savings	er Drought Response Tracking		
2 - Input Baseline Year (2019) Water Use Marin Municipal Water District									
Units: (af) Select the units to input monthly production and use data. Enter the total monthly potable water production for the Baseline Year. Next, enter monthly water use data by sector for the Baseline Year. If you bill on a bi-monthly basis, divide your billing data between the months that the billing cycle includes. If your single-family and multi-family accounts are tracked separately, enter the combined water use for both sectors in the Residential Water Use column. If your commercial, industrial, and institutional (CI) accounts are tracked separately, enter the combined water use for each sector in the CII Water Use column. Your non-revenue water use is calculated by subtracting your monthly residential, CII, and dedicated irrigation water uses from your monthly production. Your monthly residential gallons per capita per day (R-GPCD) is calculated by dividing your monthly residential water use by your population entered in Worksheet 1 - Home.									
	Total Production	Residential Water Use	CII Water Use	Irrigation Water Use	Non-Revenue Water Use				
Date	(af)	(af)	(af)	(af)	(af)	Total R-GPCD	Comments		
January	1,507	1,023	300	20	165	56	NRW is assumed to be 11%.		
February	1,361	924	271	18	149	56			
March	1,308	870	279	16	143	48			
April	1,266	842	270	15	138	48			
Мау	1,953	1,291	339	110	213	71			
June	1,890	1,249	328	106	206	71			
July	2,613	1,737	380	211	285	95			
August	2,613	1,737	380	211	285	95			
September	2,770	1,880	384	204	303	106			
October	2,863	1,942	397	211	313	106			
November	2,201	1,514	336	112	240	86			
December	2 275	1 564	347	115	248	86			

ekı		Drought Response Tool								
Но	ome Input Base Water	line Year Use	Baseline Year Water Use Profile	Drought Ac	Response tions	Estimated Wate Savings	r			
	3 - Baseline Year (2019) Water Use Profile Marin Municipal Water District									
	Baseline Year (2019) Annual Water Use Summary									
	Units	(af)								
	A summary of your Baseline	Year water use by sector and mai	ior end use categorv is shown b	elow. Select the units in which v	our production and use data are	displaved.				
				Water	lse (af)					
	Water Use	Total Production (af)	Residential	CII	Dedicated Irrigation	Non-Revenue				
	Total	24,620	16,573	4,010	1,348	2,689				
	Total Indoor	13,532	10,246	3,286						
	Total Outdoor	8,399	6,327	724	1,348					
	Total Non-Revenue	2,689				2,689				
	Total Indoor %	55%	62%	82%	0%					
	Total Outdoor %	34%	38%	18%	100%					





*Worksheet 3 - Baseline Year Water Use Profile* Page 4 of 11 Date Printed: 5/25/2021

## Drought Response Tracking



3 - Baseline Year (2019) Water Use Profile Marin Municipal Water District





## **Drought Response** Tracking

		Drought Response Tool					
Home Input Ba	seline Year	Baseline Year	Drought Response	Estimated Water			
Wat	ter Use	Water Use Profile	Actions	Savings			

## 4 - Drought Response Actions - Stage 1 Marin Municipal Water District

Maximum Savings Potential           Use the default values or enter your own criteria for the maximum savings potential. Estimated water savings within each sector will not exceed the maximum savings criteria.							
Minimum Residential Indoor GPCD	25	R-GPCD					
Maximum Residential Outdoor Savings	100%	of Baseline Residential Outdoor Water Use					
Maximum CII Indoor Savings	30%	of Baseline CII Indoor Water Use					
Maximum CII Outdoor Savings	100%	of Baseline CII Outdoor Water Use					
Maximum Dedicated Irrigation Account Savings	100%	of Baseline Dedicated Irrigation Water Use					
Maximum Non-Revenue Water Savings	75%	of Baseline Non-Revenue Water Use					
Resulting Total Maximum Annual Savings Potential	66%	of Total Baseline Production					



# Drought Response Tracking





ekı		Drought Response Tool					
Home Input Basel Water	line Year Baseline Yea Use Water Use Pro	ar Drought Respons ofile Actions	Estimated Water Savings				
			Cavingo				

## 4 - Drought Response Actions - Stage 1 Marin Municipal Water District

## Drought Response Actions

Select the Drought Response Actions you would like to include in your estimated savings calculations. For each selected action, use the default end use savings estimates and implementation rates or input your own values. The "End Use Savings" estimates the percent water use reduction that could occur at a particular end use as a result of a specific action. The "Implementation Rate" refers to the estimated percentage of accounts that will implement a specific action. The water savings potential at each end use is capped based on the assumed distribution of end use water demands shown in the pie charts above. A dash (--) indicates that professional judgement was used to establish the default value, or that savings are expected to be accounted for as part of a Public Information Program; additional basis for the default values are included in the User Manual.

Action Description	End Use(s)	Implement Program	End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default
Possible Mandatory Prohibitions	All Outdoor	$\checkmark$	14%	75%		
Prohibit Irrigation with Potable Water Outside of Newly Constructed Homes and Buildings that is not Delivered by Drip or Microspray Systems	Irrigation	✓				
Require Shut-Off Nozzles on Hoses for Vehicle Washing	Misc. Outdoor	<b>v</b>	17%	50%		
Prohibit Use of Potable Water to Wash Sidewalks and Driveways	Misc. Outdoor	✓	17%	50%	See Appendix D of the DRP	
Prohibit the Use of Potable Water for Street Washing	Misc. Outdoor	✓	17%	50%		
Prohibit Irrigation with Potable Water in a Manner that causes Runoff	Irrigation	<b>v</b>	3%	50%	DeOreo et al., 2011	
Prohibit Irrigation with Potable Water within 48 Hours following Measurable Rainfall	Irrigation	✓				
Prohibit Irrigation of Ornamental Turf with Potable Water on Street Medians	Irrigation	<ul> <li>✓</li> </ul>				
Prohibit Potable Water Use for Decorative Water Features that do not Recirculate Water	Misc. Outdoor	<ul> <li>Image: A start of the start of</li></ul>	50%	50%	EBMUD, 2008	
Provide Linen Service Opt Out Options	Fixtures & Appliances	✓	0.5%	50%	EBMUD, 2011	-
Prohibit Serving Drinking Water other than upon Request in Eating or Drinking Establishments	Fixtures & Appliances	<ul> <li>✓</li> </ul>	0.5%	50%	EBMUD, 2011	-

## Drought Response Tracking

	Drought	Respons	e Tool			
Home Input Baseline Year Ba Water Use Water	seline Year er Use Profile	Drought Res Actions	sponse s	Estimated Saving	Water Dr gs	ought Response Tracking
	4 - Drought Res Marin Mu	sponse Actio Inicipal Water Dis	ns - Stage 1 <sup>strict</sup>			
Action Description	End Use(s)	I Response Action Implement Program	End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default
Agency Drought Actions / Restrictions						
Agency Actions						
Media Campaign, Newspaper Articles, Website	All	✓	0.5%	50%	EBMUD, 2011	
Promote Water Conservation / Rebate Programs	All			50%		
Water Efficiency Workshops, Public Events	All		0.5%	25%	EBMUD, 2011	
Water Bill Inserts	All	 ✓	0.5%	100%	EBMUD, 2011	
Promote / Expand Use of Recycled Water	Irrigation		100%			
Home or Mobile Water Use Reports	All		5%	10%	WaterSmart Software, 2015	
Decrease Frequency and Length of Line Flushing	Non Revenue Water		25%	50%	See Appendix D of the DRP	Reduced flushing by 50%.
Audit and Reduce System Water Loss	Non Revenue Water		45%	50%	DWR, 2015	Target 50% of leakage.
Implement Drought Rate Structure / Water Budgets	All		5%	100%	CUWCC, 2015	
Establish Retrofit on Resale Ordinance	All Residential Indoor		21%	6%	SFPUC, 2004	First Tuesday, 2015
Require Net Zero Demand Increase on New Connections	All					
Moratorium on New Connections	All					
Move to Monthly Metering / Billing	All		5%	10%	See Appendix D of the DRP	
Increase Water Waste Patrols / Enforcement	All					
Establish Drought Hotline	All					
Reduce Distribution System Pressures	Non Revenue Water		4.5%	100%	CUWCC, 2010; DWR, 2015	
Dedicated Irrigation						
Conduct Irrigation Account Surveys			30%	10%	FBMUD 2011	
Limit Irrigation Days, Time and Duration (Select One)			00,0	1070		
Limit Irrigation to 3 Days/Week, 15 Minutes/Day, Between 8PM and 8AM	Irrigation	<b>v</b>	19%	75%		
Limit Irrigation to 1 Day/Week, 10 Minutes/Day, Between 9PM and 6AM	Irrigation		79%	50%	UC IPM, 2014	
Prohibit use of Potable Water for Irrigation	Irrigation		100%	50%		
Require Repair of all Leaks within 24 hours	External Leaks	✓	100%	5%		-
Customer Water Budgets						
Establish Water Budget - 25% Reduction	Irrigation		25%	50%		
Establish Water Budget - 50% Reduction	Irrigation		50%	50%		
Establish Water Budget - 75% Reduction	Irrigation		75%	50%		

Drought Response Tool							
Home Input Baseline Year E Water Use Wa	Baseline Year ater Use Profile	Drought Res Actions	ponse s	Estimated Saving	Water Dro Js	ought Response Tracking	
4 - Drought Response Actions - Stage 1 Marin Municipal Water District							
Action Description	Drough End Use(s)	t Response Acti Implement Program	ons End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default Implementation Rate	
Agency Drought Actions / Restrictions							
Residential							
Conduct Water Use Surveys Targeting High Water Users	All Residential Uses		10%	10%	EBMUD. 2011		
Limit Irrigation Days, Time and Duration (Select One)							
Limit Irrigation to 3 Days/Week, 15 Minutes/Day,	Irrigation						
Between 9PM and 6AM		✓	19%	75%			
Limit Irrigation to 1 Day/Week, 10 Minutes/Day, Between 9PM and 6AM	Irrigation		79%	50%	UC IPM, 2014		
Prohibit use of Potable Water for Irrigation	Irrigation		100%	50%			
Prohibit Vehicle Washing Except with Recycled Water	Misc. Outdoor		50%	50%	EBMUD, 2008		
Require Repair of all Leaks within 24 hours	Leaks	✓	100%	35%		-	
Require Pool Covers	Misc. Outdoor		28%	25%	Maddaus & Mayer, 2001	-	
Prohibit Filling of Pools	Misc. Outdoor		55%	25%	DeOreo et al., 2011		
Customer Water Budgets							
Establish Water Budget - 10% Reduction	All Residential Uses		10%	50%			
Establish Water Budget - 20% Reduction	All Residential Uses		20%	50%			
Conduct CII Surveys Targeting High Water Users	All CII uses		10%	10%	EBMUD 2011		
Limit Irrigation Days, Time and Duration (Select One)			1070	1070			
Limit Irrigation to 3 Days/Week, 15 Minutes/Day, Between 9PM and 6AM	Irrigation	Image: A state of the state	19%	75%			
Limit Irrigation to 1 Day/Week, 10 Minutes/Day, Between 9PM and 6AM	Irrigation		79%	50%	UC IFM, 2014	_	
Prohibit Use of Potable Water for Construction and Dust Control	Misc. Outdoor			100%	-	-	
Prohibit Single-Pass Cooling Systems	Cooling		80%	1%	Vickers, 2001		
Require Repair of all Leaks within 24 hours	Leaks		100%	5%	-		
Prohibit Vehicle Washing Except with Recycled Water	Misc. Outdoor		50%	50%	EBMUD, 2008		
Require Water-Efficient Pre-Rinse Spray Valves	Fixtures & Appliances		0.8%	50%	EPA, 2015; Pacific Institute, 2003		
Customer Water Budgets						1	
Establish Water Budget - 10% Reduction	All CII uses		10%	50%	-		
Establish Water Budget - 20% Reduction	All CII uses		20%	50%	-		
Establish Water Budget - 30% Reduction	All CII uses		30%	50%			

ekı	Drought Response Tool						
Home Input Baseline Year Water Use	Baseline Year Water Use Profile	Drought Response Actions	Estimated Water Savings				
4 - Drought Response Actions - Stage 1							

### 4 - Drought Response Actions - Stage 1 Marin Municipal Water District

Drought Response Actions								
Action Description	End Use(s)	Implement Program	End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default Implementation Rate		
Residential Customer Actions to Encourage								
Install Bathroom Faucet Aerators	Faucets and Dishwashers							
Install a Water-Efficient Showerhead	Showers/Baths							
Turn Off Water when Brushing Teeth, Shaving, Washing Dishes, or Cooking	Faucets and Dishwashers							
Fill the Bathtub Halfway	Showers/Baths							
Wash Only Full Loads of Clothes	Clothes Washers							
Install a High-Efficiency Toilet	Toilets				-			
Take Shorter Showers	Showers/Baths							
Run Dishwasher Only When Full	Faucets and Dishwashers				-			
Reduce Outdoor Irrigation	Irrigation				-			
Install Drip-Irrigation	Irrigation							
Use Mulch	Irrigation				-			
Plant Drought Resistant Trees and Plants	Irrigation				-			
Use a Broom to Clean Outdoor Areas	Misc. Outdoor							
Flush Less Frequently	Toilets				-			
Re-Use Shower or Bath Water for Irrigation	Irrigation							
Wash Car at Facility that Recycles the Water	Misc. Outdoor							

# Drought Response Tracking



Estimated Monthly Water Use and Savings Summary											
Units:	(af)										
This provides a summary of the estimated production relative to Baseline Year production and potential water savings, assuming implementation of selected actions at the water savings and implementation rates indicated in the Drought Response Actions worksheet. Select the units that your production data are displayed in.											
	Baseline Year	Estimated Drought		Potential							
	(2019) Production	Year Production	Estimated Potential	Cumulative							
Month	(af)	(af)	Monthly Savings	Savings	Conservation Goal	Comments					
January	1,507	1,385	8%	8%	5%						
February	1,361	1,251	8%	8%	5%						
March	1,308	1,229	6%	7%	5%						
April	1,266	1,190	6%	7%	5%						
Мау	1,953	1,727	12%	8%	5%						
June	1,890	1,671	12%	9%	5%						
July	2,613	2,235	14%	10%	5%						
August	2,613	2,235	14%	11%	5%						
September	2,770	2,351	15%	12%	5%						
October	2,863	2,430	15%	12%	5%						
November	2,201	1,913	13%	12%	5%						
December	2,275	1,977	13%	12%	5%						





*Worksheet 5 - Estimated Water Savings* Page 11 of 11 Date Printed: 5/25/2021





### 1 - Home Marin Municipal Water District

Enter Agency	Information
Agency Name	Marin Municipal Water District
Total Population Served	192,138
Conservation Goal (%)	15%
Drought Stage	Stage 2
Number of Residential Accounts	55,390
Number of Commercial, Industrial, and Institutional (CII) Accounts	3,461
Number of Dedicated Irrigation Accounts	858
Baseline Year(s)	2019
Percentage of Residential Indoor Use During Minimum Month (%)	100%
Percentage of CII Indoor Use During Minimum Month (%)	100%
Comments	

Navigation							
USER'S GUIDE	Download and read the guide before using this Tool						
1 - HOME	Enter agency information						
2 - INPUT BASELINE YEAR WATER USE	Enter Baseline Year production and use						
3 - BASELINE YEAR WATER USE	Review and confirm entered information						
4 - DROUGHT RESPONSE ACTIONS	Select Drought Response Actions and input estimated water savings and implementation rates.						
5 - ESTIMATED WATER SAVINGS	Review estimated water production and compare estimated savings to conservation target.						
6 - DROUGHT RESPONSE TRACKING	Track production and water savings against the conservation target.						



1 - Home Marin Municipal Water District

For questions about this tool or for additional information, contact:

Anona Dutton, P.G., C.Hg. <u>adutton@ekiconsult.com</u> (650) 292-9100

Disclaimer: This electronic file is being provided by EKI Environment & Water Inc. (EKI; fomerly Erler & Kalinowski, Inc.) at the request of (CLIENT). The Drought Response Tool was transmitted to CLIENT in electronic format, on a CD dated [DATE] (Original Document). Only the Original Document, provided to, and for the sole benefit of, CLIENT constitutes EKI's professional work product. An electronic copy of the Drought Response Tool is provided to CLIENT's Customer Agencies, for use only by CLIENT-designated Customer Agencies. The Drought Response Tool is copyrighted by EKI. All rights are reserved by EKI, and content may not be reproduced, downloaded, disseminated, published, or transferred in any form or by any means, except with the prior written permission of EKI. Customer Agencies may use the Drought Response Tool for reviewing potential drought response alternatives. The delivery to, or use by, Customer Agencies of the Drought Response Tool does not provide rights of reliance by Client Agencies or other third parties without the express written consent of EKI and subject to the execution of an agreement between such Customer Agency or other third party and EKI. EKI makes no warranties, either express or implied, of the electronic media or regarding its merchantability, applicability, compatibility with the recipients' computer equipment or software; of the fitness for any particular purpose; or that the electronic media contains no defect or is virus free. Use of EKI's Drought Response Tool, other electronic media, or other work product by Client Agency or others shall be at the party's sole risk. Further, by use of this electronic media, the user agrees, to the fullest extent permitted by law, to defend, indemnify and hold harmless EKI, CLIENT, and their officers, directors, employees, and subconsultants against all damages, liabilities or costs, including reasonable attorneys' fees and defense costs, arising from any use, modification or changes made to the electronic files by anyone other than EKI or from any unauthorized distribution or reuse of the electronic files without the prior written consent of EKI.

© 2015 Erler & Kalinowski, Inc.

*Worksheet 1 - Home* Page 2 of 11 Date Printed: 5/25/2021

Cl Drought Response Tool										
lome	Input Baseline Yea Water Use	Baselin Us	e Year Water e Profile	Drought Re Actior	sponse is	Estimated Wate Savings	er Drought Response Tracking			
2 - Input Baseline Year (2019) Water Use Marin Municipal Water District										
Input Baseline Year (2019) Production and Water Use										
	Total Production	Residential Water Use	CII Water Use	Dedicated Irrigation Water Use	Non-Revenue Water Use					
Date	(af)	(af)	(af)	(af)	(af)	Total R-GPCD	Comments			
January	1,507	1,023	300	20	165	56	NRW is assumed to be 11%.			
February	1,361	924	271	18	149	56				
March	1,308	870	279	16	143	48				
April	1,266	842	270	15	138	48				
Мау	1,953	1,291	339	110	213	71				
June	1,890	1,249	328	106	206	71				
July	2,613	1,737	380	211	285	95				
August	2,613	1,737	380	211	285	95				
September	2,770	1,880	384	204	303	106				
October	2 863	1.942	397	211	313	106				
	2,000	1-								
November	2,201	1,514	336	112	240	86				

ekı			Dr	ought Respo	onse Tool		
Но	ome Input Base Water	line Year Use	Baseline Year Water Use Profile	Drought Ac	Response tions	Estimated Wate Savings	r
			3 - Bas	Seline Year (2019) V Marin Municipal Wat	Vater Use Profile er District		
			Baselin	e Year (2019) Annual \	Nater Use Summary		
	Units	(af)					
	A summary of your Baseline	Year water use by sector and mai	ior end use categorv is shown b	elow. Select the units in which v	our production and use data are	displaved.	
				Water	lse (af)		
	Water Use	Total Production (af)	Residential	CII	Dedicated Irrigation	Non-Revenue	
	Total	24,620	16,573	4,010	1,348	2,689	
	Total Indoor	13,532	10,246	3,286			
	Total Outdoor	8,399	6,327	724	1,348		
	Total Non-Revenue	2,689				2,689	
	Total Indoor %	55%	62%	82%	0%		
	Total Outdoor %	34%	38%	18%	100%		





*Worksheet 3 - Baseline Year Water Use Profile* Page 4 of 11 Date Printed: 5/25/2021

## Drought Response Tracking



3 - Baseline Year (2019) Water Use Profile Marin Municipal Water District





## **Drought Response** Tracking

ekı		Drought Response Tool						
Home	Input Baseline Year Water Use		Baseline Year Water Use Profile		Drought Response Actions		Estimated Water Savings	
				_				

## 4 - Drought Response Actions - Stage 2 Marin Municipal Water District

Maximum Savings Potential           Use the default values or enter your own criteria for the maximum savings potential. Estimated water savings within each sector will not exceed the maximum savings criteria.							
Minimum Residential Indoor GPCD	25	R-GPCD					
Maximum Residential Outdoor Savings	100%	of Baseline Residential Outdoor Water Use					
Maximum CII Indoor Savings	30%	of Baseline CII Indoor Water Use					
Maximum CII Outdoor Savings	100%	of Baseline CII Outdoor Water Use					
Maximum Dedicated Irrigation Account Savings	100%	of Baseline Dedicated Irrigation Water Use					
Maximum Non-Revenue Water Savings	75%	of Baseline Non-Revenue Water Use					
Resulting Total Maximum Annual Savings Potential	66%	of Total Baseline Production					



# Drought Response Tracking





ekı		Drought Response Tool						
Home Input B Wa	aseline Year ater Use	Baseline Year Vater Use Profile	Drought Response Actions	Estimated Water Savings				

## 4 - Drought Response Actions - Stage 2 Marin Municipal Water District

### **Drought Response Actions**

Select the Drought Response Actions you would like to include in your estimated savings calculations. For each selected action, use the default end use savings estimates and implementation rates or input your own values. The "End Use Savings" estimates the percent water use reduction that could occur at a particular end use as a result of a specific action. The "Implementation Rate" refers to the estimated percentage of accounts that will implement a specific action. The water savings potential at each end use is capped based on the assumed distribution of end use water demands shown in the pie charts above. A dash (--) indicates that professional judgement was used to establish the default value, or that savings are expected to be accounted for as part of a Public Information Program; additional basis for the default values are included in the User Manual.

Action Description	End Use(s)	Implement Program	End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default Implementation Rate
Possible Mandatory Prohibitions	All Outdoor	- 	14%	75%		
Prohibit Irrigation with Potable Water Outside of Newly Constructed Homes and Buildings that is not Delivered by Drip or Microspray Systems	Irrigation	Image: A state of the state	1170			
Require Shut-Off Nozzles on Hoses for Vehicle Washing	Misc. Outdoor	<b>_</b>	17%	50%		
Prohibit Use of Potable Water to Wash Sidewalks and Driveways	Misc. Outdoor	✓	17%	50%	See Appendix D of the DRP	
Prohibit the Use of Potable Water for Street Washing	Misc. Outdoor	✓	17%	50%		
Prohibit Irrigation with Potable Water in a Manner that causes Runoff	Irrigation	✓	3%	50%	DeOreo et al., 2011	
Prohibit Irrigation with Potable Water within 48 Hours following Measurable Rainfall	Irrigation					
Prohibit Irrigation of Ornamental Turf with Potable Water on Street Medians	Irrigation	<b>v</b>				
Prohibit Potable Water Use for Decorative Water Features that do not Recirculate Water	Misc. Outdoor	<ul> <li>Image: A start of the start of</li></ul>	50%	50%	EBMUD, 2008	
Provide Linen Service Opt Out Options	Fixtures & Appliances	<b>_</b>	0.5%	50%	EBMUD, 2011	
Prohibit Serving Drinking Water other than upon Request in Eating or Drinking Establishments	Fixtures & Appliances	<b>_</b>	0.5%	50%	EBMUD, 2011	-

## Drought Response Tracking

<	Drought	Respons	e Tool			
Home Input Baseline Year Baseline Year Water Use Water	seline Year er Use Profile	Drought Res Actions	sponse s	Estimated Saving	Water Dr gs	ought Response Tracking
	4 - Drought Res Marin Mu	sponse Actio nicipal Water Dis	ns - Stage 2 <sup>strict</sup>			
Action Description	Drought End Use(s)	Response Acti Implement Program	ions End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default
Agency Drought Actions / Restrictions						
Agency Actions						
Media Campaign, Newspaper Articles, Website	All	✓	1.5%	80%	EBMUD, 2011	
Promote Water Conservation / Rebate Programs	All			50%		
Water Efficiency Workshops, Public Events	All		1.5%	80%	EBMUD, 2011	
Water Bill Inserts	All		1.5%	100%	EBMUD, 2011	
Promote / Expand Use of Recycled Water	Irrigation		100%			
Home or Mobile Water Use Reports	All		5%	10%	WaterSmart Software, 2015	
Decrease Frequency and Length of Line Flushing	Non Revenue Water	✓	25%	50%	See Appendix D of the DRP	Reduced flushing by 50%.
Audit and Reduce System Water Loss	Non Revenue Water		45%	50%	DWR, 2015	Target 50% of leakage.
Implement Drought Rate Structure / Water Budgets	All		5%	100%	CUWCC, 2015	-
Establish Retrofit on Resale Ordinance	All Residential Indoor		21%	6%	SFPUC, 2004	First Tuesday, 2015
Require Net Zero Demand Increase on New Connections	All				-	
Moratorium on New Connections	All				-	
Move to Monthly Metering / Billing	All		5%	10%	See Appendix D of the DRP	
Increase Water Waste Patrols / Enforcement	All					
Establish Drought Hotline	All				-	
Reduce Distribution System Pressures	Non Revenue Water		4.5%	100%	CUWCC, 2010; DWR, 2015	
Dedicated Irrigation						
Conduct Irrigation Account Surveys	Irrigation		30%	10%	EBMUD, 2011	
Limit Irrigation Days, Time and Duration (Select One)						
Limit Irrigation to 3 Days/Week, 15 Minutes/Day, Between 8PM and 8AM	Irrigation		19%	85%		
Limit Irrigation to 1 Day/Week, 10 Minutes/Day, Between 9PM and 6AM	Irrigation		79%	50%	UC IPM, 2014	
Prohibit use of Potable Water for Irrigation	Irrigation		100%	50%		
Require Repair of all Leaks within 24 hours	External Leaks	✓	100%	5%		
Customer Water Budgets						
Establish Water Budget - 25% Reduction	Irrigation		25%	50%	-	
Establish Water Budget - 50% Reduction	Irrigation		50%	50%		
Establish Water Budget - 75% Reduction	Irrigation		75%	50%		

ekı	Drought	Respons	e Tool			
Home Input Baseline Year Base Water Use Water U	line Year Jse Profile	Drought Res Actions	ponse s	Estimated Saving	Water Dro js	ought Response Tracking
	4 - Drought Re Marin Mu	sponse Action unicipal Water Dis	ns - Stage 2 strict			
Action Description	Drough End Use(s)	t Response Acti Implement Program	ons End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default Implementation Rate
Agency Drought Actions / Restrictions						
Residential						
Conduct Water Use Surveys Targeting High Water Users	All Residential Uses		10%	10%	EBMUD. 2011	
Limit Irrigation Days, Time and Duration (Select One)						
Limit Irrigation to 3 Days/Week, 15 Minutes/Day,	Irrigation					
Between 9PM and 6AM	Inigation		19%	85%		
Limit Irrigation to 1 Day/Week, 10 Minutes/Day, Between 9PM and 6AM	Irrigation		79%	50%	UC IPM, 2014	-
Prohibit use of Potable Water for Irrigation	Irrigation		100%	50%		
Prohibit Vehicle Washing Except with Recycled Water	Misc. Outdoor		50%	50%	EBMUD, 2008	
Require Repair of all Leaks within 24 hours	Leaks	✓	100%	35%		
Require Pool Covers	Misc. Outdoor		28%	25%	Maddaus & Mayer, 2001	
Prohibit Filling of Pools	Misc. Outdoor		55%	25%	DeOreo et al., 2011	
Customer Water Budgets						1
Establish Water Budget - 10% Reduction	All Residential Uses		10%	50%		-
Establish Water Budget - 20% Reduction	All Residential Uses		20%	50%		
► CII						
Conduct CII Surveys Targeting High Water Users	All CII uses		10%	10%	EBMUD, 2011	
Limit Irrigation Days, Time and Duration (Select One)			1	1		1
Limit Irrigation to 3 Days/Week, 15 Minutes/Day, Between 9PM and 6AM	Irrigation		19%	75%		
Limit Irrigation to 1 Day/Week, 10 Minutes/Day, Between 9PM and 6AM	Irrigation		79%	50%	UC IPM, 2014	
Prohibit Use of Potable Water for Construction and Dust Control	Misc. Outdoor			100%		
Prohibit Single-Pass Cooling Systems	Cooling		80%	1%	Vickers, 2001	
Require Repair of all Leaks within 24 hours	Leaks	✓	100%	5%		
Prohibit Vehicle Washing Except with Recycled Water	Misc. Outdoor		50%	50%	EBMUD, 2008	
Require Water-Efficient Pre-Rinse Spray Valves	Fixtures & Appliances		0.8%	50%	EPA, 2015; Pacific Institute, 2003	
Customer Water Budgets						
Establish Water Budget - 10% Reduction	All CII uses		10%	50%		-
Establish Water Budget - 20% Reduction	All CII uses		20%	50%		
Establish Water Budget - 30% Reduction	All CII uses		30%	50%		-

ekı		Drought Response Tool						
Home	Input Baseline Year Water Use	Baseline Year Water Use Profile	Drought Response Actions	Estimated Water Savings				
	4 - Drought Response Actions - Stage 2							

### 4 - Drought Response Actions - Stage 2 Marin Municipal Water District

Drought Response Actions								
Action Description	End Use(s)	Implement Program	End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default Implementation Rate		
Residential Customer Actions to Encourage								
Install Bathroom Faucet Aerators	Faucets and Dishwashers							
Install a Water-Efficient Showerhead	Showers/Baths							
Turn Off Water when Brushing Teeth, Shaving, Washing Dishes, or Cooking	Faucets and Dishwashers							
Fill the Bathtub Halfway	Showers/Baths							
Wash Only Full Loads of Clothes	Clothes Washers							
Install a High-Efficiency Toilet	Toilets							
Take Shorter Showers	Showers/Baths							
Run Dishwasher Only When Full	Faucets and Dishwashers							
Reduce Outdoor Irrigation	Irrigation							
Install Drip-Irrigation	Irrigation							
Use Mulch	Irrigation							
Plant Drought Resistant Trees and Plants	Irrigation							
Use a Broom to Clean Outdoor Areas	Misc. Outdoor							
Flush Less Frequently	Toilets							
Re-Use Shower or Bath Water for Irrigation	Irrigation					-		
Wash Car at Facility that Recycles the Water	Misc. Outdoor							

# Drought Response Tracking



Estimated Monthly Water Use and Savings Summary									
Units:	(af)								
This provides a summary of the estimated production relative to Baseline Year production and potential water savings, assuming implementation of selected actions at the water savings and implementation rates indicated									
Baseline Year Estimated Drought Polyce in Poly									
	(2019) Production	Year Production	Estimated Potential	Cumulative					
Month	(af)	(af)	Monthly Savings	Savings	Conservation Goal	Comments			
January	1,507	1,321	12%	12%	15%				
February	1,361	1,193	12%	12%	15%				
March	1,308	1,176	10%	12%	15%				
April	1,266	1,138	10%	11%	15%				
Мау	1,953	1,639	16%	13%	15%				
June	1,890	1,586	16%	13%	15%				
July	2,613	2,111	19%	15%	15%				
August	2,613	2,111	19%	15%	15%				
September	2,770	2,219	20%	16%	15%				
October	2,863	2,293	20%	17%	15%				
November	2,201	1,811	18%	17%	15%				
December	2,275	1,872	18%	17%	15%				





*Worksheet 5 - Estimated Water Savings* Page 11 of 11 Date Printed: 5/25/2021





### 1 - Home Marin Municipal Water District

Enter Agency	Information
Agency Name	Marin Municipal Water District
Total Population Served	192,138
Conservation Goal (%)	25%
Drought Stage	Stage 3
Number of Residential Accounts	55,390
Number of Commercial, Industrial, and Institutional (CII) Accounts	3,461
Number of Dedicated Irrigation Accounts	858
Baseline Year(s)	2019
Percentage of Residential Indoor Use During Minimum Month (%)	100%
Percentage of CII Indoor Use During Minimum Month (%)	100%
Comments	

Navigation					
USER'S GUIDE	Download and read the guide before using this Tool				
1 - HOME	Enter agency information				
2 - INPUT BASELINE YEAR WATER USE	Enter Baseline Year production and use				
3 - BASELINE YEAR WATER USE	Review and confirm entered information				
4 - DROUGHT RESPONSE ACTIONS	Select Drought Response Actions and input estimated water savings and implementation rates.				
5 - ESTIMATED WATER SAVINGS	Review estimated water production and compare estimated savings to conservation target.				
6 - DROUGHT RESPONSE TRACKING	Track production and water savings against the conservation target.				



1 - Home Marin Municipal Water District

For questions about this tool or for additional information, contact:

Anona Dutton, P.G., C.Hg. <u>adutton@ekiconsult.com</u> (650) 292-9100

Disclaimer: This electronic file is being provided by EKI Environment & Water Inc. (EKI; fomerly Erler & Kalinowski, Inc.) at the request of (CLIENT). The Drought Response Tool was transmitted to CLIENT in electronic format, on a CD dated [DATE] (Original Document). Only the Original Document, provided to, and for the sole benefit of, CLIENT constitutes EKI's professional work product. An electronic copy of the Drought Response Tool is provided to CLIENT's Customer Agencies, for use only by CLIENT-designated Customer Agencies. The Drought Response Tool is copyrighted by EKI. All rights are reserved by EKI, and content may not be reproduced, downloaded, disseminated, published, or transferred in any form or by any means, except with the prior written permission of EKI. Customer Agencies may use the Drought Response Tool for reviewing potential drought response alternatives. The delivery to, or use by, Customer Agencies of the Drought Response Tool does not provide rights of reliance by Client Agencies or other third parties without the express written consent of EKI and subject to the execution of an agreement between such Customer Agency or other third party and EKI. EKI makes no warranties, either express or implied, of the electronic media or regarding its merchantability, applicability, compatibility with the recipients' computer equipment or software; of the fitness for any particular purpose; or that the electronic media contains no defect or is virus free. Use of EKI's Drought Response Tool, other electronic media, or other work product by Client Agency or others shall be at the party's sole risk. Further, by use of this electronic media, the user agrees, to the fullest extent permitted by law, to defend, indemnify and hold harmless EKI, CLIENT, and their officers, directors, employees, and subconsultants against all damages, liabilities or costs, including reasonable attorneys' fees and defense costs, arising from any use, modification or changes made to the electronic files by anyone other than EKI or from any unauthorized distribution or reuse of the electronic files without the prior written consent of EKI.

© 2015 Erler & Kalinowski, Inc.

*Worksheet 1 - Home* Page 2 of 11 Date Printed: 5/25/2021

<	Drought Response Tool							
lome	Input Baseline Yea Water Use	r Baselin Us	ne Year Water e Profile	Drought Re Actior	sponse is	Estimated Wate Savings	er Drought Response Tracking	
			2 - Input Ba <sub>Mar</sub>	seline Year (201 in Municipal Water D	9) Water Use			
Un Select th basis, di Use colu subtract water us	its: (af) he units to input monthly productivide your billing data between the wim. If your commercial, industri ing your monthly residential, CII, he by your population entered in the	ion and use data. Enter t e months that the billing al, and institutional (Cll) and dedicated irrigation Vorksheet 1 - Home.	the total monthly potable w cycle includes. If your sin accounts are tracked sepa water uses from your mon	vater production for the Bas Igle-family and multi-family arately, enter the combined nthly production. Your mont	eline Year. Next, enter mo accounts are tracked sep water use for each sector hly residential gallons per	nthly water use data by sec arately, enter the combined in the Cll Water Use colum r capita per day (R-GPCD) is	ctor for the Baseline Year. If you bill on a bi-monthly water use for both sectors in the Residential Wate n. Your non-revenue water use is calculated by s calculated by dividing your monthly residential	
	Total Production	Residential Water Use	CII Water Use	Dedicated Irrigation Water Use	Non-Revenue Water Use			
Date	(af)	(af)	(af)	(af)	(af)	Total R-GPCD	Comments	
January	1,507	1,023	300	20	165	56	NRW is assumed to be 11%.	
February	1,361	924	271	18	149	56		
March	1,308	870	279	16	143	48		
April	1,266	842	270	15	138	48		
Мау	1,953	1,291	339	110	213	71		
June	1,890	1,249	328	106	206	71		
July	2,613	1,737	380	211	285	95		
August	2,613	1,737	380	211	285	95		
September	2,770	1,880	384	204	303	106		
October	2.863	1.942	397	211	313	106		
	_,	.,						
November	2,201	1,514	336	112	240	86		

ekı			Drought Response Tool								
Но	ome Input Base Water	Baseline Year Water Use Profile	Drought Ac	Response tions	Estimated Wate Savings	r					
	3 - Baseline Year (2019) Water Use Profile Marin Municipal Water District										
			Baselin	e Year (2019) Annual \	Nater Use Summary						
	Units	(af)									
	A summary of your Baseline	Year water use by sector and mai	ior end use categorv is shown b	elow. Select the units in which v	our production and use data are	displaved.					
		Mator Use (af)									
	Water Use	Total Production (af)	Residential	CII	Dedicated Irrigation	Non-Revenue					
	Total	24,620	16,573	4,010	1,348	2,689					
	Total Indoor	13,532	10,246	3,286							
	Total Outdoor	8,399	6,327	724	1,348						
	Total Non-Revenue	2,689				2,689					
	Total Indoor %	55%	62%	82%	0%						
	Total Outdoor %	34%	38%	18%	100%						





*Worksheet 3 - Baseline Year Water Use Profile* Page 4 of 11 Date Printed: 5/25/2021

## Drought Response Tracking



3 - Baseline Year (2019) Water Use Profile Marin Municipal Water District





## **Drought Response** Tracking
ekı		Dro	ought	: Response Too	
Home	Input Baseline Year Water Use	Baseline Year Water Use Profile		Drought Response Actions	Estimated Water Savings

### 4 - Drought Response Actions - Stage 3 Marin Municipal Water District

Maximum Use the default values or enter your own criteria for the maximum savings potential. Estimated water savings within eac	Savings Pot	ential ceed the maximum savings criteria.
Minimum Residential Indoor GPCD	25	R-GPCD
Maximum Residential Outdoor Savings	100%	of Baseline Residential Outdoor Water Use
Maximum CII Indoor Savings	30%	of Baseline CII Indoor Water Use
Maximum CII Outdoor Savings	100%	of Baseline CII Outdoor Water Use
Maximum Dedicated Irrigation Account Savings	100%	of Baseline Dedicated Irrigation Water Use
Maximum Non-Revenue Water Savings	75%	of Baseline Non-Revenue Water Use
Resulting Total Maximum Annual Savings Potential	66%	of Total Baseline Production



## Drought Response Tracking





ekı	Droug	ht Response Tool	
Home Input Baseline Year Water Use	Baseline Year Water Use Profile	Drought Response Actions	Estimated Water Savings

### 4 - Drought Response Actions - Stage 3 Marin Municipal Water District

### Drought Response Actions

Select the Drought Response Actions you would like to include in your estimated savings calculations. For each selected action, use the default end use savings estimates and implementation rates or input your own values. The "End Use Savings" estimates the percent water use reduction that could occur at a particular end use as a result of a specific action. The "Implementation Rate" refers to the estimated percentage of accounts that will implement a specific action. The water savings potential at each end use is capped based on the assumed distribution of end use water demands shown in the pie charts above. A dash (--) indicates that professional judgement was used to establish the default value, or that savings are expected to be accounted for as part of a Public Information Program; additional basis for the default values are included in the User Manual.

Action Description	End Use(s)	Implement Program	End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default Implementation Rate
Possible Mandatory Prohibitions	All Outdoor	- 	14%	75%		
Prohibit Irrigation with Potable Water Outside of Newly Constructed Homes and Buildings that is not Delivered by Drip or Microspray Systems	Irrigation	Image: A state of the state	1170			
Require Shut-Off Nozzles on Hoses for Vehicle Washing	Misc. Outdoor	<b>√</b>	17%	50%		
Prohibit Use of Potable Water to Wash Sidewalks and Driveways	Misc. Outdoor	✓	17%	50%	See Appendix D of the DRP	
Prohibit the Use of Potable Water for Street Washing	Misc. Outdoor	✓	17%	50%		
Prohibit Irrigation with Potable Water in a Manner that causes Runoff	Irrigation	✓	3%	50%	DeOreo et al., 2011	
Prohibit Irrigation with Potable Water within 48 Hours following Measurable Rainfall	Irrigation	<ul> <li>Image: A start of the start of</li></ul>				
Prohibit Irrigation of Ornamental Turf with Potable Water on Street Medians	Irrigation	<b>v</b>				
Prohibit Potable Water Use for Decorative Water Features that do not Recirculate Water	Misc. Outdoor	<ul> <li>Image: A start of the start of</li></ul>	50%	50%	EBMUD, 2008	
Provide Linen Service Opt Out Options	Fixtures & Appliances	<b>_</b>	0.5%	50%	EBMUD, 2011	
Prohibit Serving Drinking Water other than upon Request in Eating or Drinking Establishments	Fixtures & Appliances	<b>_</b>	0.5%	50%	EBMUD, 2011	-

## Drought Response Tracking

<i< th=""><th>Drought</th><th>Respons</th><th>e Tool</th><th></th><th></th><th></th></i<>	Drought	Respons	e Tool			
Home Input Baseline Year Base Water Use Water	seline Year r Use Profile	Drought Res Actions	sponse s	Estimated Saving	Water Dr gs	ought Response Tracking
	4 - Drought Res Marin Mu	sponse Actio Inicipal Water Dis	ns - Stage 3 strict			
Action Description	End Use(s)	Implement Program	End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default
Agency Drought Actions / Restrictions						
Agency Actions						
Media Campaign, Newspaper Articles, Website	All	✓	1.5%	80%	EBMUD, 2011	-
Promote Water Conservation / Rebate Programs	All			50%		
Water Efficiency Workshops, Public Events	All		1.5%	80%	EBMUD, 2011	
Water Bill Inserts	All	 ✓	1.5%	100%	EBMUD, 2011	
Promote / Expand Use of Recycled Water	Irrigation		100%			
Home or Mobile Water Use Reports	All		5%	10%	WaterSmart Software, 2015	
Decrease Frequency and Length of Line Flushing	Non Revenue Water		50%	80%	See Appendix D of the DRP	Reduced flushing by 50%.
Audit and Reduce System Water Loss	Non Revenue Water		45%	50%	DWR, 2015	Target 50% of leakage.
Implement Drought Rate Structure / Water Budgets	All		5%	100%	CUWCC, 2015	
Establish Retrofit on Resale Ordinance	All Residential Indoor		21%	6%	SFPUC, 2004	First Tuesday, 2015
Require Net Zero Demand Increase on New Connections	All					
Moratorium on New Connections	All					
Move to Monthly Metering / Billing	All		5%	10%	See Appendix D of the DRP	
Increase Water Waste Patrols / Enforcement	All					
Establish Drought Hotline	All	✓				
Reduce Distribution System Pressures	Non Revenue Water		4.5%	100%	CUWCC, 2010; DWR, 2015	
Dedicated Irrigation						
Conduct Irrigation	Irrigation		20%	10%		
Limit Irrigation Days, Time and Duration (Select One)	ingation		5070	1070	LDW0D, 2011	-
Limit Irrigation to 2 Days/Week, 15 Minutes/Day, Between 8PM and 8AM	Irrigation		38%	85%		
Limit Irrigation to 1 Day/Week, 10 Minutes/Day, Between 9PM and 6AM	Irrigation		79%	50%	UC IPM, 2014	-
Prohibit use of Potable Water for Irrigation	Irrigation		100%	50%		
Require Repair of all Leaks within 24 hours	External Leaks	✓	100%	5%		
Customer Water Budgets						
Establish Water Budget - 25% Reduction	Irrigation		25%	50%		-
Establish Water Budget - 50% Reduction	Irrigation		50%	50%		-
Establish Water Budget - 75% Reduction	Irrigation		75%	50%		

ekı	Drought	Respons	e Tool			
Home Input Baseline Year Bas Water Use Wate	seline Year r Use Profile	Drought Res Actions	sponse s	Estimated Saving	Water Dro js	ought Response Tracking
	4 - Drought Re Marin M	esponse Action unicipal Water Dis	ns - Stage 3 <sub>strict</sub>			
Action Description	Drough End Use(s)	it Response Acti Implement Program	ions End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default Implementation Rate
Agency Drought Actions / Restrictions						
Residential						
Conduct Water Use Surveys Targeting High Water Users	All Residential Uses		10%	10%	EBMUD. 2011	
Limit Irrigation Days, Time and Duration (Select One)						
Limit Irrigation to 2 Days/Week, 15 Minutes/Day,	Irrigation					
Between 9PM and 6AM	ingation		38%	85%		
Limit Irrigation to 1 Day/Week, 10 Minutes/Day, Between 9PM and 6AM	Irrigation		79%	50%	UC IPM, 2014	-
Prohibit use of Potable Water for Irrigation	Irrigation		100%	50%		
Prohibit Vehicle Washing Except with Recycled Water	Misc. Outdoor		50%	50%	EBMUD, 2008	
Require Repair of all Leaks within 24 hours	Leaks	✓	100%	35%		
Require Pool Covers	Misc. Outdoor		28%	25%	Maddaus & Mayer, 2001	
Prohibit Filling of Pools	Misc. Outdoor		55%	25%	DeOreo et al., 2011	
Customer Water Budgets						1
Establish Water Budget - 10% Reduction	All Residential Uses		10%	50%		
Establish Water Budget - 20% Reduction	All Residential Uses		20%	50%		-
Conduct CII Surveys Targeting High Water Users	All CII uses		10%	10%	EBMUD, 2011	
Limit Irrigation Days, Time and Duration (Select One)				1		
Limit Irrigation to 2 Days/Week, 15 Minutes/Day, Between 9PM and 6AM	Irrigation	<b>_</b>	38%	85%		
Limit Irrigation to 1 Day/Week, 10 Minutes/Day, Between 9PM and 6AM	Irrigation		79%	50%	UC IPM, 2014	
Prohibit Use of Potable Water for Construction and Dust Control	Misc. Outdoor			100%		
Prohibit Single-Pass Cooling Systems	Cooling		80%	1%	Vickers, 2001	
Require Repair of all Leaks within 24 hours	Leaks	Image: A state of the state	100%	5%		
Prohibit Vehicle Washing Except with Recycled Water	Misc. Outdoor		50%	50%	EBMUD, 2008	
Require Water-Efficient Pre-Rinse Spray Valves	Fixtures & Appliances		0.8%	50%	EPA, 2015; Pacific Institute, 2003	
Customer Water Budgets						
Establish Water Budget - 10% Reduction	All CII uses		10%	50%		-
Establish Water Budget - 20% Reduction	All CII uses		20%	50%		
Establish Water Budget - 30% Reduction	All CII uses		30%	50%		-

ekı		Droug	ht Response Tool	
Home	Input Baseline Year Water Use	Baseline Year Water Use Profile	Drought Response Actions	Estimated Water Savings
		4 - Drought	Response Actions - Stage 3	

### 4 - Drought Response Actions - Stage 3 Marin Municipal Water District

	Drought	Response Acti	ions			
Action Description	End Use(s)	Implement Program	End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default Implementation Rate
Residential Customer Actions to Encourage						
Install Bathroom Faucet Aerators	Faucets and Dishwashers					
Install a Water-Efficient Showerhead	Showers/Baths					
Turn Off Water when Brushing Teeth, Shaving, Washing Dishes, or Cooking	Faucets and Dishwashers					
Fill the Bathtub Halfway	Showers/Baths					
Wash Only Full Loads of Clothes	Clothes Washers					
Install a High-Efficiency Toilet	Toilets				-	
Take Shorter Showers	Showers/Baths					
Run Dishwasher Only When Full	Faucets and Dishwashers					
Reduce Outdoor Irrigation	Irrigation				-	
Install Drip-Irrigation	Irrigation					
Use Mulch	Irrigation				-	
Plant Drought Resistant Trees and Plants	Irrigation					
Use a Broom to Clean Outdoor Areas	Misc. Outdoor					
Flush Less Frequently	Toilets					-
Re-Use Shower or Bath Water for Irrigation	Irrigation					-
Wash Car at Facility that Recycles the Water	Misc. Outdoor					

# Drought Response Tracking



		Estimate	ed Monthly Water Use	e and Savings Sumr	nary	
Units:	(af)					
This provides a sun	nmary of the estimated produc	ction relative to Baseline Year p	production and potential water s	avings, assuming implementa	ation of selected actions at the w	ater savings and implementation rates indicated
in the Drought Resp	oonse Actions worksheet. Sele	Estimated Drought	on data are displayed in.	Detential		
	Daseline rear	Estimated Drought	Estimated Detential	Potential		
	(2019) Production	fear Production	Estimated Potential	Cumulative		
Month	(af)	(af)	Monthly Savings	Savings	Conservation Goal	Comments
January	1,507	1,243	18%	18%	25%	
February	1,361	1,122	18%	18%	25%	
March	1,308	1,134	13%	16%	25%	
April	1,266	1,097	13%	16%	25%	
Мау	1,953	1,481	24%	18%	25%	
June	1,890	1,433	24%	19%	25%	
July	2,613	1,835	30%	21%	25%	
August	2,613	1,835	30%	23%	25%	
September	2,770	1,907	31%	24%	25%	
October	2,863	1,971	31%	25%	25%	
November	2,201	1,602	27%	25%	25%	
December	2,275	1,656	27%	26%	25%	





*Worksheet 5 - Estimated Water Savings* Page 11 of 11 Date Printed: 5/25/2021





### 1 - Home Marin Municipal Water District

Enter Agency	Information
Agency Name	Marin Municipal Water District
Total Population Served	192,138
Conservation Goal (%)	35%
Drought Stage	Stage 4
Number of Residential Accounts	55,390
Number of Commercial, Industrial, and Institutional (CII) Accounts	3,461
Number of Dedicated Irrigation Accounts	858
Baseline Year(s)	2019
Percentage of Residential Indoor Use During Minimum Month (%)	100%
Percentage of CII Indoor Use During Minimum Month (%)	100%
Comments	

	Navigation
USER'S GUIDE	Download and read the guide before using this Tool
1 - HOME	Enter agency information
2 - INPUT BASELINE YEAR WATER USE	Enter Baseline Year production and use
3 - BASELINE YEAR WATER USE	Review and confirm entered information
4 - DROUGHT RESPONSE ACTIONS	Select Drought Response Actions and input estimated water savings and implementation rates.
5 - ESTIMATED WATER SAVINGS	Review estimated water production and compare estimated savings to conservation target.
6 - DROUGHT RESPONSE TRACKING	Track production and water savings against the conservation target.



1 - Home Marin Municipal Water District

For questions about this tool or for additional information, contact:

Anona Dutton, P.G., C.Hg. <u>adutton@ekiconsult.com</u> (650) 292-9100

Disclaimer: This electronic file is being provided by EKI Environment & Water Inc. (EKI; fomerly Erler & Kalinowski, Inc.) at the request of (CLIENT). The Drought Response Tool was transmitted to CLIENT in electronic format, on a CD dated [DATE] (Original Document). Only the Original Document, provided to, and for the sole benefit of, CLIENT constitutes EKI's professional work product. An electronic copy of the Drought Response Tool is provided to CLIENT's Customer Agencies, for use only by CLIENT-designated Customer Agencies. The Drought Response Tool is copyrighted by EKI. All rights are reserved by EKI, and content may not be reproduced, downloaded, disseminated, published, or transferred in any form or by any means, except with the prior written permission of EKI. Customer Agencies may use the Drought Response Tool for reviewing potential drought response alternatives. The delivery to, or use by, Customer Agencies of the Drought Response Tool does not provide rights of reliance by Client Agencies or other third parties without the express written consent of EKI and subject to the execution of an agreement between such Customer Agency or other third party and EKI. EKI makes no warranties, either express or implied, of the electronic media or regarding its merchantability, applicability, compatibility with the recipients' computer equipment or software; of the fitness for any particular purpose; or that the electronic media contains no defect or is virus free. Use of EKI's Drought Response Tool, other electronic media, or other work product by Client Agency or others shall be at the party's sole risk. Further, by use of this electronic media, the user agrees, to the fullest extent permitted by law, to defend, indemnify and hold harmless EKI, CLIENT, and their officers, directors, employees, and subconsultants against all damages, liabilities or costs, including reasonable attorneys' fees and defense costs, arising from any use, modification or changes made to the electronic files by anyone other than EKI or from any unauthorized distribution or reuse of the electronic files without the prior written consent of EKI.

© 2015 Erler & Kalinowski, Inc.

*Worksheet 1 - Home* Page 2 of 11 Date Printed: 5/25/2021

Drought Response Tool								
lome	Input Baseline Yea Water Use	Baselin Us	e Year Water e Profile	Drought Re Actior	sponse is	Estimated Wate Savings	er Drought Response Tracking	
			2 - Input Ba <sub>Mar</sub>	seline Year (201 in Municipal Water D	9) Water Use			
Un Select th basis, di Use colu subtract water us	its: (af) e units to input monthly producti vide your billing data between the mn. If your commercial, industri ing your monthly residential, CII, e by your population entered in V	on and use data. Enter t e months that the billing al, and institutional (Cil) and dedicated irrigation Vorksheet 1 - Home.	he totai monthiy potable w cycle includes. If your sin accounts are tracked seps water uses from your mon	vater production for the Bas Igle-family and multi-family arately, enter the combined nthly production. Your mont	eline Year. Next, enter mo accounts are tracked sep water use for each sector hly residential gallons per	nthly water use data by sec arately, enter the combined in the Cll Water Use colum r capita per day (R-GPCD) is	tor for the Baseline Year. If you bill on a bi-monthly water use for both sectors in the Residential Wate n. Your non-revenue water use is calculated by s calculated by dividing your monthly residential	
	Total Production	Residential Water Use	CII Water Use	Dedicated Irrigation Water Use	Non-Revenue Water Use			
Date	(af)	(af)	(af)	(af)	(af)	Total R-GPCD	Comments	
January	1,507	1,023	300	20	165	56	NRW is assumed to be 11%.	
February	1,361	924	271	18	149	56		
March	1,308	870	279	16	143	48		
April	1,266	842	270	15	138	48		
Мау	1,953	1,291	339	110	213	71		
June	1,890	1,249	328	106	206	71		
July	2,613	1,737	380	211	285	95		
August	2,613	1,737	380	211	285	95		
September	2,770	1,880	384	204	303	106		
October	2 863	1.942	397	211	313	106		
	2,000	1-						
November	2,201	1,514	336	112	240	86		

ekı			Drought Response Tool					
Но	ome Input Base Water	line Year Use	Baseline Year Water Use Profile	Drought Ac	Response tions	Estimated Wate Savings	r	
			3 - Bas	Seline Year (2019) V Marin Municipal Wat	Vater Use Profile er District			
			Baselin	e Year (2019) Annual \	Nater Use Summary			
	Units	(af)						
	A summary of your Baseline	Year water use by sector and mai	l ior end use categorv is shown b	elow. Select the units in which v	our production and use data are	displaved.		
				Water	lse (af)			
	Water Use	Total Production (af)	Residential	CII	Dedicated Irrigation	Non-Revenue		
	Total	24,620	16,573	4,010	1,348	2,689		
	Total Indoor	13,532	10,246	3,286				
	Total Outdoor	8,399	6,327	724	1,348			
	Total Non-Revenue	2,689				2,689		
	Total Indoor %	55%	62%	82%	0%			
	Total Outdoor %	34%	38%	18%	100%			





*Worksheet 3 - Baseline Year Water Use Profile* Page 4 of 11 Date Printed: 5/25/2021

## Drought Response Tracking



3 - Baseline Year (2019) Water Use Profile Marin Municipal Water District





## **Drought Response** Tracking

ekı		Dro	ought	Response Too	
Home	Input Baseline Year Water Use	Baseline Year Water Use Profile		Drought Response Actions	Estimated Water Savings

## 4 - Drought Response Actions - Stage 4 Marin Municipal Water District

Maximum Savings Potential           Use the default values or enter your own criteria for the maximum savings potential. Estimated water savings within each sector will not exceed the maximum savings criteria.						
Minimum Residential Indoor GPCD	25	R-GPCD				
Maximum Residential Outdoor Savings	100%	of Baseline Residential Outdoor Water Use				
Maximum CII Indoor Savings	30%	of Baseline CII Indoor Water Use				
Maximum CII Outdoor Savings	100%	of Baseline CII Outdoor Water Use				
Maximum Dedicated Irrigation Account Savings	100%	of Baseline Dedicated Irrigation Water Use				
Maximum Non-Revenue Water Savings	75%	of Baseline Non-Revenue Water Use				
Resulting Total Maximum Annual Savings Potential	66%	of Total Baseline Production				



## Drought Response Tracking





Input Baseline Year Baseline Year Drought Response Estimated Wate	ekı	Droug	ht Response Tool	
Water Use         Water Use         Profile         Actions         Savings	Home Input Baseline Year	Baseline Year	Drought Response	Estimated Water
	Water Use	Water Use Profile	Actions	Savings

### 4 - Drought Response Actions - Stage 4 Marin Municipal Water District

### **Drought Response Actions**

Select the Drought Response Actions you would like to include in your estimated savings calculations. For each selected action, use the default end use savings estimates and implementation rates or input your own values. The "End Use Savings" estimates the percent water use reduction that could occur at a particular end use as a result of a specific action. The "Implementation Rate" refers to the estimated percentage of accounts that will implement a specific action. The water savings potential at each end use is capped based on the assumed distribution of end use water demands shown in the pie charts above. A dash (--) indicates that professional judgement was used to establish the default value, or that savings are expected to be accounted for as part of a Public Information Program; additional basis for the default values are included in the User Manual.

Action Description	End Use(s)	Implement Program	End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default
Possible Mandatory Prohibitions	All Outdoor	$\checkmark$	14%	75%		
Prohibit Irrigation with Potable Water Outside of Newly Constructed Homes and Buildings that is not Delivered by Drip or Microspray Systems	Irrigation	✓				
Require Shut-Off Nozzles on Hoses for Vehicle Washing	Misc. Outdoor	<b>v</b>	17%	50%		
Prohibit Use of Potable Water to Wash Sidewalks and Driveways	Misc. Outdoor	✓	17%	50%	See Appendix D of the DRP	
Prohibit the Use of Potable Water for Street Washing	Misc. Outdoor	<ul> <li>✓</li> </ul>	17%	50%		
Prohibit Irrigation with Potable Water in a Manner that causes Runoff	Irrigation	<b>v</b>	3%	50%	DeOreo et al., 2011	
Prohibit Irrigation with Potable Water within 48 Hours following Measurable Rainfall	Irrigation	<b>v</b>				
Prohibit Irrigation of Ornamental Turf with Potable Water on Street Medians	Irrigation	<ul> <li>✓</li> </ul>				
Prohibit Potable Water Use for Decorative Water Features that do not Recirculate Water	Misc. Outdoor	<ul> <li>Image: A start of the start of</li></ul>	50%	50%	EBMUD, 2008	
Provide Linen Service Opt Out Options	Fixtures & Appliances	✓	0.5%	50%	EBMUD, 2011	-
Prohibit Serving Drinking Water other than upon Request in Eating or Drinking Establishments	Fixtures & Appliances	<ul> <li>✓</li> </ul>	0.5%	50%	EBMUD, 2011	-

## Drought Response Tracking

CI Drought Response Tool						
Home Input Baseline Year Ba Water Use Water	er Use Profile	Drought Res Action	sponse s	Estimated Saving	Water Dr gs	ought Response Tracking
	4 - Drought Res Marin Mu	sponse Actio	ns - Stage 4 strict			
Action Description	End Use(s)	Implement Program	End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default Implementation Rat
Agency Drought Actions / Restrictions						
Agency Actions						
Media Campaign, Newspaper Articles, Website	All	✓	1.5%	80%	EBMUD, 2011	
Promote Water Conservation / Rebate Programs	All	✓		50%		
Water Efficiency Workshops, Public Events	All	✓	1.5%	80%	EBMUD, 2011	
Water Bill Inserts	All	✓	1.5%	100%	EBMUD, 2011	
Promote / Expand Use of Recycled Water	Irrigation		100%			
Home or Mobile Water Use Reports	All		5%	10%	WaterSmart Software, 2015	
Decrease Frequency and Length of Line Flushing	Non Revenue Water	✓	50%	80%	See Appendix D of the DRP	Reduced flushing by 50%.
Audit and Reduce System Water Loss	Non Revenue Water		45%	50%	DWR, 2015	Target 50% of leakage.
Implement Drought Rate Structure / Water Budgets	All		5%	100%	CUWCC, 2015	
Establish Retrofit on Resale Ordinance	All Residential Indoor		21%	6%	SFPUC, 2004	First Tuesday, 2015
Require Net Zero Demand Increase on New Connections	All					
Moratorium on New Connections	All					
Move to Monthly Metering / Billing	All		5%	10%	See Appendix D of the DRP	
Increase Water Waste Patrols / Enforcement	All					
Establish Drought Hotline	All	✓				
Reduce Distribution System Pressures	Non Revenue Water		4.5%	100%	CUWCC, 2010; DWR, 2015	
Dedicated Irrigation						
Conduct Irrigation Account Surveys	Irrigation		30%	10%	FBMUD 2011	
Limit Irrigation Days, Time and Duration (Select One)			0070	10,0		
Limit Irrigation to 2 Days/Week, 15 Minutes/Day, Between 8PM and 8AM	Irrigation		38%	85%		
Limit Irrigation to 1 Day/Week, 10 Minutes/Day, Between 9PM and 6AM	Irrigation	<b>V</b>	79%	85%	UC IPM, 2014	-
Prohibit use of Potable Water for Irrigation	Irrigation		100%	50%		
Require Repair of all Leaks within 24 hours	External Leaks	✓	100%	5%		
Customer Water Budgets						
Establish Water Budget - 25% Reduction	Irrigation		25%	50%	-	-
Establish Water Budget - 50% Reduction	Irrigation		50%	50%		-
Establish Water Budget - 75% Reduction	Irrigation		75%	50%		

Drought Response Tool							
Home Input Baseline Year Bas Water Use Water	eline Year Use Profile	Drought Res Actions	sponse s	Estimated Saving	Water Dro	ought Response Tracking	
	4 - Drought Re Marin M	sponse Actio unicipal Water Dis	ns - Stage 4 <sup>strict</sup>				
Action Description	Drough End Use(s)	It Response Acti Implement Program	ions End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default Implementation Rate	
Agency Drought Actions / Restrictions							
Residential							
Conduct Water Use Surveys Targeting High Water Users	All Residential Uses		10%	10%	FBMUD 2011		
Limit Irrigation Days. Time and Duration (Select One)			1070	1070	251105,2011		
Limit Irrigation to 2 Days/Week, 15 Minutes/Day, Between 9PM and 6AM	Irrigation		38%	85%			
Limit Irrigation to 1 Day/Week, 10 Minutes/Day, Between 9PM and 6AM	Irrigation		79%	85%	UC IPM, 2014		
Prohibit use of Potable Water for Irrigation	Irrigation		100%	50%			
Prohibit Vehicle Washing Except with Recycled Water	Misc. Outdoor	✓ ✓	50%	50%	EBMUD, 2008	-	
Require Repair of all Leaks within 24 hours	Leaks	✓ ✓	100%	35%		-	
Require Pool Covers	Misc. Outdoor	✓	28%	25%	Maddaus & Mayer, 2001	-	
Prohibit Filling of Pools	Misc. Outdoor		55%	25%	DeOreo et al., 2011	-	
Customer Water Budgets							
Establish Water Budget - 10% Reduction	All Residential Uses		10%	50%		-	
Establish Water Budget - 20% Reduction	All Residential Uses		20%	50%			
► CII							
Conduct CII Surveys Targeting High Water Users	All CII uses		10%	10%	EBMUD, 2011		
Limit Irrigation Days, Time and Duration (Select One)					1		
Limit Irrigation to 2 Days/Week, 15 Minutes/Day, Between 9PM and 6AM	Irrigation		38%	85%	LIC IPM 2014	_	
Limit Irrigation to 1 Day/Week, 10 Minutes/Day, Between 9PM and 6AM	Irrigation	Image: A state of the state	79%	85%	00 II M, 2014		
Prohibit Use of Potable Water for Construction and Dust Control	Misc. Outdoor	✓		100%			
Prohibit Single-Pass Cooling Systems	Cooling		80%	1%	Vickers, 2001		
Require Repair of all Leaks within 24 hours	Leaks		100%	5%			
Prohibit Vehicle Washing Except with Recycled Water	Misc. Outdoor	✓	50%	50%	EBMUD, 2008		
Require Water-Efficient Pre-Rinse Spray Valves	Fixtures & Appliances		0.8%	50%	EPA, 2015; Pacific Institute, 2003		
Customer Water Budgets						1	
Establish Water Budget - 10% Reduction	All CII uses		10%	50%		-	
Establish Water Budget - 20% Reduction	All CII uses		20%	50%		-	
Establish Water Budget - 30% Reduction	All CII uses		30%	50%		-	

ekı		Droug	nt Response Tool	
Home	Input Baseline Year Water Use	Baseline Year Water Use Profile	Drought Response Actions	Estimated Water Savings
		4 - Drought	Response Actions - Stage 4	

### 4 - Drought Response Actions - Stage 4 Marin Municipal Water District

Drought Response Actions						
Action Description	End Use(s)	Implement Program	End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default Implementation Rate
Residential Customer Actions to Encourage						
Install Bathroom Faucet Aerators	Faucets and Dishwashers					
Install a Water-Efficient Showerhead	Showers/Baths					-
Turn Off Water when Brushing Teeth, Shaving, Washing Dishes, or Cooking	Faucets and Dishwashers					
Fill the Bathtub Halfway	Showers/Baths					
Wash Only Full Loads of Clothes	Clothes Washers					
Install a High-Efficiency Toilet	Toilets					
Take Shorter Showers	Showers/Baths					
Run Dishwasher Only When Full	Faucets and Dishwashers					
Reduce Outdoor Irrigation	Irrigation					
Install Drip-Irrigation	Irrigation					
Use Mulch	Irrigation					
Plant Drought Resistant Trees and Plants	Irrigation					
Use a Broom to Clean Outdoor Areas	Misc. Outdoor					
Flush Less Frequently	Toilets					
Re-Use Shower or Bath Water for Irrigation	Irrigation					
Wash Car at Facility that Recycles the Water	Misc. Outdoor					

# Drought Response Tracking



	Estimated Monthly Water Use and Savings Summary							
Units:	(af)							
This provides a sum	This provides a summary of the estimated production relative to Baseline Year production and potential water savings, assuming implementation of selected actions at the water savings and implementation rates indicated							
In the Drought Resp	Racolino Voar	Estimated Drought	on data are displayed in.	Potential				
	(2019) Production	Vear Production	Estimated Potential	Cumulativo				
Month	(2013) Production	(af)	Monthly Savinge	Savinge	Conservation Goal	Comments		
Month			wontiny Savings	Savings	Conservation Goar	Comments		
January	1,507	1,184	21%	21%	35%			
February	1,361	1,069	21%	21%	35%			
March	1,308	1,129	14%	19%	35%			
April	1,266	1,092	14%	18%	35%			
Мау	1,953	1,299	33%	22%	35%			
June	1,890	1,257	33%	24%	35%			
July	2,613	1,473	44%	29%	35%			
August	2,613	1,473	44%	31%	35%			
September	2,770	1,493	46%	34%	35%			
October	2,863	1,542	46%	35%	35%			
November	2,201	1,343	39%	36%	35%			
December	2,275	1,388	39%	36%	35%			





*Worksheet 5 - Estimated Water Savings* Page 11 of 11 Date Printed: 5/25/2021





### 1 - Home Marin Municipal Water District

Enter Agency	Information
Agency Name	Marin Municipal Water District
Total Population Served	192,138
Conservation Goal (%)	45%
Drought Stage	Stage 5
Number of Residential Accounts	55,390
Number of Commercial, Industrial, and Institutional (CII) Accounts	3,461
Number of Dedicated Irrigation Accounts	858
Baseline Year(s)	2019
Percentage of Residential Indoor Use During Minimum Month (%)	100%
Percentage of CII Indoor Use During Minimum Month (%)	100%
Comments	

	Navigation
USER'S GUIDE	Download and read the guide before using this Tool
1 - HOME	Enter agency information
2 - INPUT BASELINE YEAR WATER USE	Enter Baseline Year production and use
3 - BASELINE YEAR WATER USE	Review and confirm entered information
4 - DROUGHT RESPONSE ACTIONS	Select Drought Response Actions and input estimated water savings and implementation rates.
5 - ESTIMATED WATER SAVINGS	Review estimated water production and compare estimated savings to conservation target.
6 - DROUGHT RESPONSE TRACKING	Track production and water savings against the conservation target.



1 - Home Marin Municipal Water District

For questions about this tool or for additional information, contact:

Anona Dutton, P.G., C.Hg. <u>adutton@ekiconsult.com</u> (650) 292-9100

Disclaimer: This electronic file is being provided by EKI Environment & Water Inc. (EKI; fomerly Erler & Kalinowski, Inc.) at the request of (CLIENT). The Drought Response Tool was transmitted to CLIENT in electronic format, on a CD dated [DATE] (Original Document). Only the Original Document, provided to, and for the sole benefit of, CLIENT constitutes EKI's professional work product. An electronic copy of the Drought Response Tool is provided to CLIENT's Customer Agencies, for use only by CLIENT-designated Customer Agencies. The Drought Response Tool is copyrighted by EKI. All rights are reserved by EKI, and content may not be reproduced, downloaded, disseminated, published, or transferred in any form or by any means, except with the prior written permission of EKI. Customer Agencies may use the Drought Response Tool for reviewing potential drought response alternatives. The delivery to, or use by, Customer Agencies of the Drought Response Tool does not provide rights of reliance by Client Agencies or other third parties without the express written consent of EKI and subject to the execution of an agreement between such Customer Agency or other third party and EKI. EKI makes no warranties, either express or implied, of the electronic media or regarding its merchantability, applicability, compatibility with the recipients' computer equipment or software; of the fitness for any particular purpose; or that the electronic media contains no defect or is virus free. Use of EKI's Drought Response Tool, other electronic media, or other work product by Client Agency or others shall be at the party's sole risk. Further, by use of this electronic media, the user agrees, to the fullest extent permitted by law, to defend, indemnify and hold harmless EKI, CLIENT, and their officers, directors, employees, and subconsultants against all damages, liabilities or costs, including reasonable attorneys' fees and defense costs, arising from any use, modification or changes made to the electronic files by anyone other than EKI or from any unauthorized distribution or reuse of the electronic files without the prior written consent of EKI.

© 2015 Erler & Kalinowski, Inc.

*Worksheet 1 - Home* Page 2 of 11 Date Printed: 5/25/2021

<	Drought Response Tool							
lome	Input Baseline Yea Water Use	Baselin Us	e Year Water e Profile	Drought Re Actior	sponse is	Estimated Wate Savings	er Drought Response Tracking	
2 - Input Baseline Year (2019) Water Use Marin Municipal Water District								
Un Select th basis, di Use colu subtract water us	its: (af) e units to input monthly producti vide your billing data between the mn. If your commercial, industri ing your monthly residential, CII, e by your population entered in V	on and use data. Enter t e months that the billing al, and institutional (Cil) and dedicated irrigation Vorksheet 1 - Home.	he totai monthiy potable w cycle includes. If your sin accounts are tracked seps water uses from your mon	vater production for the Bas Igle-family and multi-family arately, enter the combined nthly production. Your mont	eline Year. Next, enter mo accounts are tracked sep water use for each sector hly residential gallons per	nthly water use data by sec arately, enter the combined in the Cll Water Use colum r capita per day (R-GPCD) is	tor for the Baseline Year. If you bill on a bi-monthly water use for both sectors in the Residential Wate n. Your non-revenue water use is calculated by s calculated by dividing your monthly residential	
	Total Production	Residential Water Use	CII Water Use	Dedicated Irrigation Water Use	Non-Revenue Water Use			
Date	(af)	(af)	(af)	(af)	(af)	Total R-GPCD	Comments	
January	1,507	1,023	300	20	165	56	NRW is assumed to be 11%.	
February	1,361	924	271	18	149	56		
March	1,308	870	279	16	143	48		
April	1,266	842	270	15	138	48		
Мау	1,953	1,291	339	110	213	71		
June	1,890	1,249	328	106	206	71		
July	2,613	1,737	380	211	285	95		
August	2,613	1,737	380	211	285	95		
September	2,770	1,880	384	204	303	106		
October	2 863	1.942	397	211	313	106		
	2,000	1-						
November	2,201	1,514	336	112	240	86		

ekı		Drought Response Tool								
Но	ome Input Base Water	line Year Use	Baseline Year Water Use Profile	Drought Ac	Response tions	Estimated Wate Savings	r			
	3 - Baseline Year (2019) Water Use Profile Marin Municipal Water District									
			Baselin	e Year (2019) Annual \	Nater Use Summary					
	Units	(af)								
	A summary of your Baseline	Year water use by sector and mai	ior end use categorv is shown b	elow. Select the units in which v	our production and use data are	displaved.				
				Water	lse (af)					
	Water Use	Total Production (af)	Residential	CII	Dedicated Irrigation	Non-Revenue				
	Total	24,620	16,573	4,010	1,348	2,689				
	Total Indoor	13,532	10,246	3,286						
	Total Outdoor	8,399	6,327	724	1,348					
	Total Non-Revenue	2,689				2,689				
	Total Indoor %	55%	62%	82%	0%					
	Total Outdoor %	34%	38%	18%	100%					





*Worksheet 3 - Baseline Year Water Use Profile* Page 4 of 11 Date Printed: 5/25/2021

## Drought Response Tracking



3 - Baseline Year (2019) Water Use Profile Marin Municipal Water District





## **Drought Response** Tracking

ekı		Drought Response Tool					
Home	Input Baseline Year Water Use		Baseline Year Water Use Profile		Drought Response Actions		Estimated Water Savings
						-	

### 4 - Drought Response Actions - Stage 5 Marin Municipal Water District

Maximum Savings Potential Use the default values or enter your own criteria for the maximum savings potential. Estimated water savings within each sector will not exceed the maximum savings criteria.								
Minimum Residential Indoor GPCD	25	R-GPCD						
Maximum Residential Outdoor Savings	100%	of Baseline Residential Outdoor Water Use						
Maximum CII Indoor Savings	30%	of Baseline CII Indoor Water Use						
Maximum CII Outdoor Savings	100%	of Baseline CII Outdoor Water Use						
Maximum Dedicated Irrigation Account Savings	100%	of Baseline Dedicated Irrigation Water Use						
Maximum Non-Revenue Water Savings	75%	of Baseline Non-Revenue Water Use						
Resulting Total Maximum Annual Savings Potential	66%	of Total Baseline Production						



## Drought Response Tracking





ekı	Dr	ought Response Tool	
Home Input Baselin Water U	ne Year Ise Baseline Year Water Use Profile	Drought Response Actions	Estimated Water Savings
			_

### 4 - Drought Response Actions - Stage 5 Marin Municipal Water District

### Drought Response Actions

Select the Drought Response Actions you would like to include in your estimated savings calculations. For each selected action, use the default end use savings estimates and implementation rates or input your own values. The "End Use Savings" estimates the percent water use reduction that could occur at a particular end use as a result of a specific action. The "Implementation Rate" refers to the estimated percentage of accounts that will implement a specific action. The water savings potential at each end use is capped based on the assumed distribution of end use water demands shown in the pie charts above. A dash (--) indicates that professional judgement was used to establish the default value, or that savings are expected to be accounted for as part of a Public Information Program; additional basis for the default values are included in the User Manual.

Action Description	End Use(s)	Implement Program	End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default Implementation Rate
Possible Mandatory Prohibitions	All Outdoor	- 	14%	75%		
Prohibit Irrigation with Potable Water Outside of Newly Constructed Homes and Buildings that is not Delivered by Drip or Microspray Systems	Irrigation	Image: A state of the state	1170			
Require Shut-Off Nozzles on Hoses for Vehicle Washing	Misc. Outdoor	<b>_</b>	17%	50%		
Prohibit Use of Potable Water to Wash Sidewalks and Driveways	Misc. Outdoor	✓	17%	50%	See Appendix D of the DRP	
Prohibit the Use of Potable Water for Street Washing	Misc. Outdoor	✓	17%	50%		
Prohibit Irrigation with Potable Water in a Manner that causes Runoff	Irrigation	✓	3%	50%	DeOreo et al., 2011	
Prohibit Irrigation with Potable Water within 48 Hours following Measurable Rainfall	Irrigation	<ul> <li>Image: A start of the start of</li></ul>				
Prohibit Irrigation of Ornamental Turf with Potable Water on Street Medians	Irrigation	<b>v</b>				
Prohibit Potable Water Use for Decorative Water Features that do not Recirculate Water	Misc. Outdoor	<ul> <li>Image: A start of the start of</li></ul>	50%	50%	EBMUD, 2008	
Provide Linen Service Opt Out Options	Fixtures & Appliances	<b>_</b>	0.5%	50%	EBMUD, 2011	
Prohibit Serving Drinking Water other than upon Request in Eating or Drinking Establishments	Fixtures & Appliances	<b>_</b>	0.5%	50%	EBMUD, 2011	-

## Drought Response Tracking

KI	Drought	Respons	e Tool			
Home Input Baseline Year Bas Water Use Wate	seline Year r Use Profile	Drought Res Actions	sponse s	Estimated Saving	Water Dr gs	ought Response Tracking
	4 - Drought Res Marin Mu	sponse Actio inicipal Water Dis	ns - Stage 5 <sub>strict</sub>			
Action Description	Drought End Use(s)	Response Acti Implement Program	ions End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default
Agency Drought Actions / Restrictions						
Agency Actions						
Media Campaign, Newspaper Articles, Website	All	✓	2.0%	90%	EBMUD, 2011	
Promote Water Conservation / Rebate Programs	All			50%		
Water Efficiency Workshops, Public Events	All		2.0%	90%	EBMUD, 2011	
Water Bill Inserts	All		2.0%	100%	EBMUD, 2011	
Promote / Expand Use of Recycled Water	Irrigation		100%			
Home or Mobile Water Use Reports	All		5%	10%	WaterSmart Software, 2015	
Decrease Frequency and Length of Line Flushing	Non Revenue Water		50%	80%	See Appendix D of the DRP	Reduced flushing by 50%.
Audit and Reduce System Water Loss	Non Revenue Water		45%	50%	DWR, 2015	Target 50% of leakage.
Implement Drought Rate Structure / Water Budgets	All		5%	100%	CUWCC, 2015	
Establish Retrofit on Resale Ordinance	All Residential Indoor		21%	6%	SFPUC, 2004	First Tuesday, 2015
Require Net Zero Demand Increase on New Connections	All					
Moratorium on New Connections	All					
Move to Monthly Metering / Billing	All		5%	10%	See Appendix D of the DRP	
Increase Water Waste Patrols / Enforcement	All					
Establish Drought Hotline	All	✓				
Reduce Distribution System Pressures	Non Revenue Water		4.5%	100%	CUWCC, 2010; DWR, 2015	
Dedicated Irrigation						
Conduct Irrigation Account Surveys	Irrigation		30%	10%	FBMUD 2011	
Limit Irrigation Days, Time and Duration (Select One)	inigation		0070	1070		
Limit Irrigation to 2 Days/Week, 15 Minutes/Day, Between 8PM and 8AM	Irrigation		38%	85%		
Limit Irrigation to 1 Day/Week, 10 Minutes/Day, Between 9PM and 6AM	Irrigation		79%	85%	UC IPM, 2014	
Prohibit use of Potable Water for Irrigation	Irrigation		100%	90%		
Require Repair of all Leaks within 24 hours	External Leaks		100%	5%	-	
Customer Water Budgets						
Establish Water Budget - 25% Reduction	Irrigation		25%	50%		
Establish Water Budget - 50% Reduction	Irrigation		50%	50%		-
Establish Water Budget - 75% Reduction	Irrigation		75%	50%		

CI Drought Response Tool							
Home Input Baseline Year Bas Water Use Wate	seline Year r Use Profile	Drought Res Actions	ponse s	Estimated Saving	Water Dro	ought Response Tracking	
	4 - Drought Re Marin M	sponse Action unicipal Water Dis	ns - Stage 5 strict				
Action Description	Drough End Use(s)	t Response Acti Implement Program	ons End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default Implementation Rate	
Agency Drought Actions / Restrictions							
► Residential							
Conduct Water Use Surveys Targeting High Water Users	All Residential Uses		10%	10%	EBMUD. 2011		
Limit Irrigation Days, Time and Duration (Select One)							
Limit Irrigation to 2 Days/Week, 15 Minutes/Day, Between 9PM and 6AM	Irrigation		38%	85%			
Limit Irrigation to 1 Day/Week, 10 Minutes/Day, Between 9PM and 6AM	Irrigation		79%	85%	UC IPM, 2014	-	
Prohibit use of Potable Water for Irrigation	Irrigation		100%	90%			
Prohibit Vehicle Washing Except with Recycled Water	Misc. Outdoor		50%	50%	EBMUD, 2008	-	
Require Repair of all Leaks within 24 hours	Leaks	✓ 	100%	35%		-	
Require Pool Covers	Misc. Outdoor		28%	25%	Maddaus & Mayer, 2001	-	
Prohibit Filling of Pools	Misc. Outdoor	✓	55%	25%	DeOreo et al., 2011		
Customer Water Budgets			( (				
Establish Water Budget - 10% Reduction	All Residential Uses		10%	50%			
Establish Water Budget - 20% Reduction	All Residential Uses		20%	50%		-	
► CII							
Conduct CII Surveys Targeting High Water Users	All CII uses		10%	10%	EBMUD, 2011		
Limit Irrigation Days, Time and Duration (Select One)							
Limit Irrigation to 2 Days/Week, 15 Minutes/Day, Between 9PM and 6AM	Irrigation		38%	85%	UC IPM, 2014	_	
Prohibit use of Potable Water for Irrigation	Irrigation		100%	90%			
Prohibit Use of Potable Water for Construction and Dust Control	Misc. Outdoor			100%			
Prohibit Single-Pass Cooling Systems	Cooling		80%	1%	Vickers, 2001		
Require Repair of all Leaks within 24 hours	Leaks	✓	100%	5%			
Prohibit Vehicle Washing Except with Recycled Water	Misc. Outdoor	Image: A state of the state	50%	50%	EBMUD, 2008		
Require Water-Efficient Pre-Rinse Spray Valves	Fixtures & Appliances		0.8%	50%	EPA, 2015; Pacific Institute, 2003		
Customer Water Budgets	Customer Water Budgets						
Establish Water Budget - 10% Reduction	All CII uses		10%	50%		-	
Establish Water Budget - 20% Reduction	All CII uses		20%	50%		-	
Establish Water Budget - 30% Reduction	All CII uses		30%	50%			

ekı		Drought Response Tool					
Home	Input Baseline Year Water Use		Baseline Year Water Use Profile		Drought Response Actions		Estimated Water Savings
			4 - Droug	aht Re	sponse Actions - Stag	ə 5	

### 4 - Drought Response Actions - Stage 5 Marin Municipal Water District

	Drought	Response Acti	ons			
Action Description	End Use(s)	Implement Program	End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default Implementation Rate
Residential Customer Actions to Encourage						
Install Bathroom Faucet Aerators	Faucets and Dishwashers					
Install a Water-Efficient Showerhead	Showers/Baths					
Turn Off Water when Brushing Teeth, Shaving, Washing Dishes, or Cooking	Faucets and Dishwashers					
Fill the Bathtub Halfway	Showers/Baths					
Wash Only Full Loads of Clothes	Clothes Washers					
Install a High-Efficiency Toilet	Toilets					
Take Shorter Showers	Showers/Baths					
Run Dishwasher Only When Full	Faucets and Dishwashers					
Reduce Outdoor Irrigation	Irrigation					
Install Drip-Irrigation	Irrigation					
Use Mulch	Irrigation					
Plant Drought Resistant Trees and Plants	Irrigation					
Use a Broom to Clean Outdoor Areas	Misc. Outdoor					
Flush Less Frequently	Toilets					
Re-Use Shower or Bath Water for Irrigation	Irrigation					
Wash Car at Facility that Recycles the Water	Misc. Outdoor					

# Drought Response Tracking



	Estimated Monthly Water Use and Savings Summary							
Units:	(af)							
This provides a sum	mary of the estimated produc	ction relative to Baseline Year p	production and potential water s	avings, assuming implementa	ation of selected actions at the w	ater savings and implementation rates indicated		
in the Drought Resp	Populing Voor	Ectimated Drought	on data are displayed in.	Detential				
		Estimated Drought	Estimated Detential	Potential				
	(2019) Production	Year Production	Estimated Potential	Cumulative				
Month	(af)	(af)	Monthly Savings	Savings	Conservation Goal	Comments		
January	1,507	1,095	27%	27%	45%			
February	1,361	989	27%	27%	45%			
March	1,308	1,075	18%	24%	45%			
April	1,266	1,040	18%	23%	45%			
Мау	1,953	1,135	42%	28%	45%			
June	1,890	1,099	42%	31%	45%			
July	2,613	1,196	54%	36%	45%			
August	2,613	1,196	54%	39%	45%			
September	2,770	1,182	57%	42%	45%			
October	2,863	1,222	57%	44%	45%			
November	2,201	1,130	49%	45%	45%			
December	2,275	1,168	49%	45%	45%			





*Worksheet 5 - Estimated Water Savings* Page 11 of 11 Date Printed: 5/25/2021





### 1 - Home Marin Municipal Water District

Enter Agency	Information
Agency Name	Marin Municipal Water District
Total Population Served	192,138
Conservation Goal (%)	55%
Drought Stage	Stage 6
Number of Residential Accounts	55,390
Number of Commercial, Industrial, and Institutional (CII) Accounts	3,461
Number of Dedicated Irrigation Accounts	858
Baseline Year(s)	2019
Percentage of Residential Indoor Use During Minimum Month (%)	100%
Percentage of CII Indoor Use During Minimum Month (%)	100%
Comments	

Navigation					
USER'S GUIDE	Download and read the guide before using this Tool				
1 - HOME	Enter agency information				
2 - INPUT BASELINE YEAR WATER USE	Enter Baseline Year production and use				
3 - BASELINE YEAR WATER USE	Review and confirm entered information				
4 - DROUGHT RESPONSE ACTIONS	Select Drought Response Actions and input estimated water savings and implementation rates.				
5 - ESTIMATED WATER SAVINGS	Review estimated water production and compare estimated savings to conservation target.				
6 - DROUGHT RESPONSE TRACKING	Track production and water savings against the conservation target.				



1 - Home Marin Municipal Water District

For questions about this tool or for additional information, contact:

Anona Dutton, P.G., C.Hg. <u>adutton@ekiconsult.com</u> (650) 292-9100

Disclaimer: This electronic file is being provided by EKI Environment & Water Inc. (EKI; fomerly Erler & Kalinowski, Inc.) at the request of (CLIENT). The Drought Response Tool was transmitted to CLIENT in electronic format, on a CD dated [DATE] (Original Document). Only the Original Document, provided to, and for the sole benefit of, CLIENT constitutes EKI's professional work product. An electronic copy of the Drought Response Tool is provided to CLIENT's Customer Agencies, for use only by CLIENT-designated Customer Agencies. The Drought Response Tool is copyrighted by EKI. All rights are reserved by EKI, and content may not be reproduced, downloaded, disseminated, published, or transferred in any form or by any means, except with the prior written permission of EKI. Customer Agencies may use the Drought Response Tool for reviewing potential drought response alternatives. The delivery to, or use by, Customer Agencies of the Drought Response Tool does not provide rights of reliance by Client Agencies or other third parties without the express written consent of EKI and subject to the execution of an agreement between such Customer Agency or other third party and EKI. EKI makes no warranties, either express or implied, of the electronic media or regarding its merchantability, applicability, compatibility with the recipients' computer equipment or software; of the fitness for any particular purpose; or that the electronic media contains no defect or is virus free. Use of EKI's Drought Response Tool, other electronic media, or other work product by Client Agency or others shall be at the party's sole risk. Further, by use of this electronic media, the user agrees, to the fullest extent permitted by law, to defend, indemnify and hold harmless EKI, CLIENT, and their officers, directors, employees, and subconsultants against all damages, liabilities or costs, including reasonable attorneys' fees and defense costs, arising from any use, modification or changes made to the electronic files by anyone other than EKI or from any unauthorized distribution or reuse of the electronic files without the prior written consent of EKI.

© 2015 Erler & Kalinowski, Inc.

*Worksheet 1 - Home* Page 2 of 11 Date Printed: 5/25/2021

<i drought="" response="" th="" tool<=""></i>									
lome	Input Baseline Yea Water Use	r Baselin Us	ne Year Water e Profile	Drought Re Actior	sponse is	Estimated Wate Savings	er Drought Response Tracking		
2 - Input Baseline Year (2019) Water Use Marin Municipal Water District									
Units: (af)  Select the units to input monthly production and use data. Enter the total monthly potable water production for the Baseline Year. Next, enter monthly water use data by sector for the Baseline Year. If you bill on a bi-monthly basis, divide your billing data between the months that the billing cycle includes. If your single-family and multi-family accounts are tracked separately, enter the combined water use for both sectors in the Residential Water Use column. If your commercial, industrial, and institutional (CII) accounts are tracked separately, enter the combined water use for each sector in the CII Water Use column. Your non-revenue water uses is calculated by subtracting your monthly residential, CII, and dedicated irrigation water uses from your monthly production. Your monthly residential gallons per capita per day (R-GPCD) is calculated by dividing your monthly residential water use by your population entered in Worksheet 1 - Home.									
	Total Production	Residential Water Use	CII Water Use	Dedicated Irrigation Water Use	Non-Revenue Water Use				
Date	(af)	(af)	(af)	(af)	(af)	Total R-GPCD	Comments		
January	1,507	1,023	300	20	165	56	NRW is assumed to be 11%.		
February	1,361	924	271	18	149	56			
March	1,308	870	279	16	143	48			
April	1,266	842	270	15	138	48			
Мау	1,953	1,291	339	110	213	71			
June	1,890	1,249	328	106	206	71			
July	2,613	1,737	380	211	285	95			
August	2,613	1,737	380	211	285	95			
September	2,770	1,880	384	204	303	106			
October	2.863	1.942	397	211	313	106			
	_,	.,							
November	2,201	1,514	336	112	240	86			

ekı	Drought Response Tool								
Но	ome Input Base Water	line Year Use	Baseline Year Water Use Profile	Drought Ac	Response tions	Estimated Wate Savings	r		
			3 - Bas	Seline Year (2019) V Marin Municipal Wat	Vater Use Profile er District				
			Baselin	e Year (2019) Annual \	Vater Use Summary				
	Units	(af)							
	A summary of your Baseline	Year water use by sector and mai	ior end use categorv is shown b	elow. Select the units in which v	our production and use data are	displaved.			
				Water	lse (af)				
	Water Use	Total Production (af)	Residential	CII	Dedicated Irrigation	Non-Revenue			
	Total	24,620	16,573	4,010	1,348	2,689			
	Total Indoor	13,532	10,246	3,286					
	Total Outdoor	8,399	6,327	724	1,348				
	Total Non-Revenue	2,689				2,689			
	Total Indoor %	55%	62%	82%	0%				
	Total Outdoor %	34%	38%	18%	100%				





*Worksheet 3 - Baseline Year Water Use Profile* Page 4 of 11 Date Printed: 5/25/2021

## Drought Response Tracking



3 - Baseline Year (2019) Water Use Profile Marin Municipal Water District





## **Drought Response** Tracking

ekı		Drought Response Tool						
Home	Input Baseline Year Water Use		Baseline Year Water Use Profile		Drought Response Actions		Estimated Water Savings	

### 4 - Drought Response Actions - Stage 6 Marin Municipal Water District

Maximum Savings Potential           Use the default values or enter your own criteria for the maximum savings potential. Estimated water savings within each sector will not exceed the maximum savings criteria.							
Minimum Residential Indoor GPCD	25	R-GPCD					
Maximum Residential Outdoor Savings	100%	of Baseline Residential Outdoor Water Use					
Maximum CII Indoor Savings	30%	of Baseline CII Indoor Water Use					
Maximum CII Outdoor Savings	100%	of Baseline CII Outdoor Water Use					
Maximum Dedicated Irrigation Account Savings	100%	of Baseline Dedicated Irrigation Water Use					
Maximum Non-Revenue Water Savings	75%	of Baseline Non-Revenue Water Use					
Resulting Total Maximum Annual Savings Potential	66%	of Total Baseline Production					



## Drought Response Tracking





ekı		Drought Response Tool						
Home Input I	Baseline Year Vater Use	Baseline Year Water Use Profile	Drought Response Actions	Estimated Water Savings				

### 4 - Drought Response Actions - Stage 6 Marin Municipal Water District

### **Drought Response Actions**

Select the Drought Response Actions you would like to include in your estimated savings calculations. For each selected action, use the default end use savings estimates and implementation rates or input your own values. The "End Use Savings" estimates the percent water use reduction that could occur at a particular end use as a result of a specific action. The "Implementation Rate" refers to the estimated percentage of accounts that will implement a specific action. The water savings potential at each end use is capped based on the assumed distribution of end use water demands shown in the pie charts above. A dash (--) indicates that professional judgement was used to establish the default value, or that savings are expected to be accounted for as part of a Public Information Program; additional basis for the default values are included in the User Manual.

Action Description	End Use(s)	Implement Program	End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default Implementation Rate
Possible Mandatory Prohibitions	All Outdoor		14%	75%		
Prohibit Irrigation with Potable Water Outside of Newly Constructed Homes and Buildings that is not Delivered by Drip or Microspray Systems	Irrigation	V	1170		-	
Require Shut-Off Nozzles on Hoses for Vehicle Washing	Misc. Outdoor	✓	17%	50%		
Prohibit Use of Potable Water to Wash Sidewalks and Driveways	Misc. Outdoor	✓	17%	50%	See Appendix D of the DRP	
Prohibit the Use of Potable Water for Street Washing	Misc. Outdoor	✓	17%	50%		
Prohibit Irrigation with Potable Water in a Manner that causes Runoff	Irrigation	$\checkmark$	3%	50%	DeOreo et al., 2011	
Prohibit Irrigation with Potable Water within 48 Hours following Measurable Rainfall	Irrigation	<b>I</b>				
Prohibit Irrigation of Ornamental Turf with Potable Water on Street Medians	Irrigation	~				
Prohibit Potable Water Use for Decorative Water Features that do not Recirculate Water	Misc. Outdoor	<b>v</b>	50%	50%	EBMUD, 2008	
Provide Linen Service Opt Out Options	Fixtures & Appliances	✓	0.5%	50%	EBMUD, 2011	
Prohibit Serving Drinking Water other than upon Request in Eating or Drinking Establishments	Fixtures & Appliances	<b>v</b>	0.5%	50%	EBMUD, 2011	-

## Drought Response Tracking

KI Drought Response Tool						
Home Input Baseline Year Ba Water Use Water	seline Year er Use Profile	Drought Res Actions	sponse s	Estimated Saving	Water Dr js	ought Response Tracking
	4 - Drought Res Marin Mu	sponse Action Inicipal Water Dis	ns - Stage 6 <sup>strict</sup>			
Action Description	Drought End Use(s)	t Response Acti Implement Program	ions End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default
Agency Drought Actions / Restrictions						
Agency Actions						
Media Campaign, Newspaper Articles, Website	All	✓	2.0%	90%	EBMUD, 2011	
Promote Water Conservation / Rebate Programs	All			50%		
Water Efficiency Workshops, Public Events	All	✓	2.0%	90%	EBMUD, 2011	
Water Bill Inserts	All		2.0%	100%	EBMUD, 2011	
Promote / Expand Use of Recycled Water	Irrigation		100%			
Home or Mobile Water Use Reports	All		5%	10%	WaterSmart Software, 2015	
Decrease Frequency and Length of Line Flushing	Non Revenue Water		50%	80%	See Appendix D of the DRP	Reduced flushing by 50%.
Audit and Reduce System Water Loss	Non Revenue Water		45%	50%	DWR, 2015	Target 50% of leakage.
Implement Drought Rate Structure / Water Budgets	All		5%	100%	CUWCC, 2015	
Establish Retrofit on Resale Ordinance	All Residential Indoor		21%	6%	SFPUC, 2004	First Tuesday, 2015
Require Net Zero Demand Increase on New Connections	All					
Moratorium on New Connections	All	✓				
Move to Monthly Metering / Billing	All		5%	10%	See Appendix D of the DRP	
Increase Water Waste Patrols / Enforcement	All	✓				
Establish Drought Hotline	All	✓				
Reduce Distribution System Pressures	Non Revenue Water		4.5%	100%	CUWCC, 2010; DWR, 2015	
Dedicated Irrigation						
Conduct Irrigation Account Surveys	Irrigation		30%	10%	EBMUD 2011	
Limit Irrigation Days, Time and Duration (Select One)	inigation		5070	1070	LDM0D, 2011	
Limit Irrigation to 2 Days/Week, 15 Minutes/Day, Between 8PM and 8AM	Irrigation		38%	85%		
Limit Irrigation to 1 Day/Week, 10 Minutes/Day, Between 9PM and 6AM	Irrigation		79%	85%	UC IPM, 2014	-
Prohibit use of Potable Water for Irrigation	Irrigation	✓ 	100%	85%		
Require Repair of all Leaks within 24 hours	External Leaks	✓	100%	5%		
Customer Water Budgets						
Establish Water Budget - 56% Reduction	Irrigation		56%	90%	-	-
Establish Water Budget - 50% Reduction	Irrigation		50%	50%	-	
Establish Water Budget - 75% Reduction	Irrigation		75%	50%		
Drought Response Tool						
--	--------------------------------	---	-------------------------------	------------------------	---------------------------------------	--
Home Input Baseline Year Ba Water Use Water	aseline Year er Use Profile	Drought Res Actions	ponse s	Estimated Saving	Water Dr js	ought Response Tracking
4 - Drought Response Actions - Stage 6 Marin Municipal Water District						
Action Description	Drough End Use(s)	t Response Acti Implement Program	ons End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default Implementation Rate
Agency Drought Actions / Restrictions						
Residential						
Conduct Water Use Surveys Targeting High Water Users	All Residential Uses		10%	10%	EBMUD. 2011	
Limit Irrigation Days, Time and Duration (Select One)						
Limit Irrigation to 2 Days/Week, 15 Minutes/Day, Between 9PM and 6AM	Irrigation		38%	85%		
Limit Irrigation to 1 Day/Week, 10 Minutes/Day, Between 9PM and 6AM	Irrigation		79%	85%	UC IPM, 2014	
Prohibit use of Potable Water for Irrigation	Irrigation		100%	90%		
Prohibit Vehicle Washing Except with Recycled Water	Misc. Outdoor	✓	50%	50%	EBMUD, 2008	
Require Repair of all Leaks within 24 hours	Leaks		100%	35%		-
Require Pool Covers	Misc. Outdoor		28%	25%	Maddaus & Mayer, 2001	-
Prohibit Filling of Pools	Misc. Outdoor	<i>✓</i>	55%	25%	DeOreo et al., 2011	
Customer Water Budgets						
Establish Water Budget - 56% Reduction	All Residential Uses		56%	85%		
Establish Water Budget - 20% Reduction	All Residential Uses		20%	50%		
► CII						
Conduct CII Surveys Targeting High Water Users	All CII uses		10%	10%	EBMUD, 2011	
Limit Irrigation Days, Time and Duration (Select One)						
Limit Irrigation to 2 Days/Week, 15 Minutes/Day, Between 9PM and 6AM	Irrigation		38%	85%	UC IPM, 2014	
Prohibit use of Potable Water for Irrigation	Irrigation		100%	90%		
Prohibit Use of Potable Water for Construction and Dust Control	Misc. Outdoor			100%		
Prohibit Single-Pass Cooling Systems	Cooling		80%	1%	Vickers, 2001	
Require Repair of all Leaks within 24 hours	Leaks		100%	5%		
Prohibit Vehicle Washing Except with Recycled Water	Misc. Outdoor		50%	50%	EBMUD, 2008	
Require Water-Efficient Pre-Rinse Spray Valves	Fixtures & Appliances		0.8%	50%	EPA, 2015; Pacific Institute, 200	3
Customer Water Budgets						
Establish Water Budget - 56% Reduction	All CII uses	<ul> <li>✓</li> </ul>	56%	85%		-
Establish Water Budget - 20% Reduction	All CII uses		20%	50%	-	
Establish Water Budget - 30% Reduction	All CII uses		30%	50%		

ekı		Drought Response Tool					
Home	Input Baseline Year Water Use	)	Baseline Year Water Use Profile		Drought Response Actions	)	Estimated Water Savings
			4 - Droug	ght R	esponse Actions - Stage 6		

#### 4 - Drought Response Actions - Stage 6 Marin Municipal Water District

Drought Response Actions							
Action Description	End Use(s)	Implement Program	End Use Savings (%)	Implementation Rate	Source of Default Savings Estimate	Source of Default Implementation Rate	
Residential Customer Actions to Encourage							
Install Bathroom Faucet Aerators	Faucets and Dishwashers						
Install a Water-Efficient Showerhead	Showers/Baths						
Turn Off Water when Brushing Teeth, Shaving, Washing Dishes, or Cooking	Faucets and Dishwashers						
Fill the Bathtub Halfway	Showers/Baths						
Wash Only Full Loads of Clothes	Clothes Washers						
Install a High-Efficiency Toilet	Toilets						
Take Shorter Showers	Showers/Baths						
Run Dishwasher Only When Full	Faucets and Dishwashers						
Reduce Outdoor Irrigation	Irrigation						
Install Drip-Irrigation	Irrigation						
Use Mulch	Irrigation						
Plant Drought Resistant Trees and Plants	Irrigation						
Use a Broom to Clean Outdoor Areas	Misc. Outdoor						
Flush Less Frequently	Toilets						
Re-Use Shower or Bath Water for Irrigation	Irrigation						
Wash Car at Facility that Recycles the Water	Misc. Outdoor						

#### Drought Response Tracking

Drought Response Tool © 2015 Erler & Kalinowski, Inc.



	Estimated Monthly Water Use and Savings Summary							
Units:	Units: (af)							
This provides a sum	This provides a summary of the estimated production relative to Baseline Year production and potential water savings, assuming implementation of selected actions at the water savings and implementation rates indicated							
in the Drought Resp	In the Drought Response Actions worksheet. Select the units that your production data are displayed in.							
	(2019) Production	Year Production	Estimated Potential	Cumulative				
Month	(af)	(af)	Monthly Savings	Savings	Conservation Goal	Comments		
January	1,507	768	49%	49%	55%			
February	1,361	693	49%	49%	55%			
March	1,308	706	46%	48%	55%			
April	1,266	683	46%	48%	55%			
Мау	1,953	881	55%	50%	55%			
June	1,890	852	55%	51%	55%			
July	2,613	1,058	60%	53%	55%			
August	2,613	1,058	60%	54%	55%			
September	2,770	1,100	60%	55%	55%			
October	2,863	1,136	60%	56%	55%			
November	2,201	949	57%	56%	55%			
December	2,275	980	57%	56%	55%			





*Worksheet 5 - Estimated Water Savings* Page 11 of 11 Date Printed: 5/25/2021



Drought Response Tool © 2015 Erler & Kalinowski, Inc.



### **ATTACHMENT 3**

#### WATER SHORTAGE CONTINGENCY PLAN RESOLUTIONS

Pending Appendix

Appendices 2020 Urban Water Management Plan Marin Municipal Water District



### Appendix I

#### 2007 WATER CONSERVATION MASTER PLAN



### MARIN MUNICIPAL WATER DISTRICT

# 2007 Water Conservation Master Plan



### Adopted June 20, 2007

This page intentionally left blank.

### Acknowledgments

#### Marin Municipal Water District 2007 Board of Directors

Cynthia Koehler, President Larry Russell, Vice President David Behar, Director Alex Forman, Director Jack Gibson, Director

> Paul Helliker General Manager

#### Daniel Carney Water Conservation Manager

MMWD has made every effort to be accurate and complete in the presentation of this report. Thanks to the many individuals who have contributed the ideas, questions, and comments that have made the preparation of this Master Plan possible, some of whom are recognized as follows:

#### Members of the Conservation Action Committee

#### Maddaus Water Management

Charlton Research

Descomstudios

MMWD Staff:

Environmental Review

Eric McGuire

Public Information

Neil Dukas Libby Pischel <u>Finance</u>

Cheryl Howlett Terry Stigall

#### Water Conservation

Keith Bancroft Charlene Burgi Corinne Dare Elena Freeman Mark Guthrie Robin McKillop Nancy McQuilkin Wendy Menara

#### Facilities & Watershed Information Technology

Steve Phelps Greg Renfort Carl Sanders Bill Yungert Debby Barrett Bob Fairchild Nancy Gibbs Gavin McGhie Nick Salcedo Mariette Shin

This page intentionally left blank.



#### **TABLE OF CONTENTS**

	ACKNOWLEDGMENTS	i
►		1
►	EXECUTIVE SUMMARY	7

#### ► <u>SECTIONS</u>

•	MMWD CONSERVATION PROGRAMS	15
	• STATEWIDE CONTEXT	15
	• FUTURE MMWD PROGRAM GOALS AND OBJECTIVES	18
	• AUTOMATED METER READING (AMR) & ADVANCED METER INFRASTRUCTURE	
	(AMI) PROGRAM	19
	LEAK DETECTION PROGRAM	21
	CONSERVATION PROGRAM #1	24
	CONSERVATION PROGRAM #2	26
	CONSERVATION PROGRAM #3	28
	CURRENT MMWD PROGRAM STATUS	30
	• PROGRAM ACTIVITY LEVELS	31
	• Сомминту Input	32
	• CONSERVATION SCRAPBOOK – 2006/2007	33
•	POLICY RECOMMENDATIONS	37
•	CONSERVATION MARKETING PLAN	41
•	CONCLUSIONS	51
•	Appendices	
	MADDAUS CONSERVATION TECHNICAL ANALYSIS	ENDIX A
	STAFF REPORTSAPPE	ENDIX B
	MMWD 2006 WATER MANAGEMENT REPORT APPE	ENDIX C
	<ul> <li>MMWD SURVEY (CHARLTON RESEARCH CO., 2005) APPE</li> </ul>	ENDIX D

#### ► LIST OF FIGURES

•	FIGURE 1: RELATIONAL MODEL OF AN URBAN WATER CONSERVATION PROGRAM	6
•	FIGURE 2: MADDAUS PRESENT VALUE OF UTILITY COSTS VERSUS CUMULATIVE	
	WATER SAVED IN 2030	10
•	FIGURE 3: STATEWIDE BMP COMPLIANCE	16
•	FIGURE 4: PROGRAM #1 ORGANIZATIONAL CHART	24
•	FIGURE 5: PROGRAM #2 ORGANIZATIONAL CHART	26
•	FIGURE 6: PROGRAM #3 ORGANIZATIONAL CHART	28

#### ► LIST OF TABLES

•	TABLE 1:	COMPARISON OF MMWD AND MADDAUS CONSERVATION ANALYSES	8
•	TABLE 2:	MADDAUS PROGRAM DESCRIPTIONS AND FUTURE WATER SAVINGS	9
•	TABLE 3:	Maddaus Program Costs (2008-2030)	9
•	TABLE 4:	MADDAUS COST PER ACRE FOOT & WATER SAVINGS POTENTIAL	10
•	TABLE 5:	MMWD CONSERVATION PROGRAM ALTERNATIVES	11
•	TABLE 6:	COMPARISON OF BMP PROGRAM IMPLEMENTATION	17
•	TABLE 7:	PROGRAM #1 BUDGETED ACTIVITIES FY '08/'09	25
•	TABLE 8:	PROGRAM #2 BUDGETED ACTIVITIES FY '08/'09	27
•	TABLE 9:	PROGRAM #3 BUDGETED ACTIVITIES FY '08/'09	29

#### INTRODUCTION

In 2007, the relationship between human activities and the resulting environmental impacts is a fact accepted by most people. We *know* we must act now in order to protect natural resources for future generations. Water is no exception. Worldwide, overuse and mismanagement of water supplies lead to over-drafted aquifers, depleted fisheries, habitat degradation, water-borne diseases, war, and economic hardship.

Marin is an oasis. Sweet, clean water runs off local mountains and flows abundantly in most years. If it were not for the problems of limited storage, recurring droughts, a growing population, and the financial and environmental costs of developing new water supplies, the need to increase efficiency would be less urgent. In light of these problems, it is appropriate for the District to redouble conservation efforts and sustain them in the future.

Conservation, however, is not intended to be a stand-alone strategy for managing the District's water portfolio. Efficiency measures have a point of diminishing returns, just as reservoirs have a limited storage capacity. Although there is a significant amount of water to be conserved in Marin, at a certain point it becomes too difficult, expensive, or inconvenient to use less water and still maintain a customary standard of living.

Ongoing capital improvements will always be necessary to increase system operating efficiencies and provide a hedge against catastrophic drought emergencies. To this end, the District is analyzing the benefits of detecting and repairing leaky pipes, augmenting delivery capacity, modernizing outdated equipment, entering into cooperative supply agreements with neighboring water agencies, and exploring new filtration technologies.

How much water can ultimately be conserved on a sustained, normal year basis? What will be the water demand in 2025? Will we get more or less rain due to global climate change? There are no certain answers to these questions. History shows us that conservation programs reliably and permanently reduce water demand. We do know that new water-saving products are in the stores today and more are being developed, and that the cost of these products will drop over time as more manufacturers enter the market in response to greater demand. We also know that public education is the golden key that can unlock and sustain the greatest water conservation potential.

Public opinion surveys indicate that the citizens of Marin value a healthy environment and that they value water conservation as one of the most important solutions to the ongoing specter of inadequate water supply. These surveys also indicate that most MMWD customers believe they are doing all they can to conserve water. Clearly, there are opportunities to educate our customers on new, cost-effective ways to use our water supplies as efficiently as possible. In this Master Plan, the terms *water conservation* and *water efficiency* are used interchangeably to mean the practice of making the *best use of water resources*. The basic goal of the Plan is to draw a conservation trail map for the District. Several program options are presented, and all trails lead to the same destination—a future where water waste is reduced, the environment is protected, and water rates are based on the efficient use of available resources.

#### <u>Reliability</u>

In order to evaluate the reliability of water efficiency programs and the savings they can provide, two fundamental concepts come into play:

- **Technical Savings Potential:** savings resulting from the installation of efficient plumbing fixtures such as high-efficiency toilets, clothes washers and irrigation systems; and from maintaining the water system infrastructure.
- **Behavioral Savings Potential**: savings resulting from activities such as taking shorter showers, turning off faucets, and reducing landscape watering.

The results the District can achieve by targeting both of these savings potentials are more than twice as great as the technical savings alone. This *elasticity* is a key concept that must be understood when determining the investment value of technical and behavioral programs such as plumbing fixtures and customer education.

*Technical Savings* comprise more than 90% of the potential water savings, costs, and benefits associated with the various conservation program options in this Plan. This is the 'standard method' used in the conservation industry because it is the most objective method to forecast future savings. It is more difficult to create a formula that predicts how people will behave. Fortunately, a method has been devised to quantify behavioral savings. Bill Maddaus, a recognized expert in the field of water efficiency, has calculated the behavioral savings that can be gained when people take shorter showers, fix leaks, and turn off running faucets. These detailed calculations are included in Appendix A, starting on page 37 of the *Maddaus Conservation Technical Analysis*.

The calculation below estimates the percentage of savings from technical and behavioral actions based on the calculations in Table 10 of the Maddaus Analysis. (Note: The behavioral savings of 500 acre feet represents 10% of the total conservation savings shown for Program D in Appendix A, Table 10, page 41 of the Maddaus Analysis.)

Plumbing Code	=	1,600 acre feet per year
Conservation Savings (technical)	=	2,900 acre feet per year
Technical Savings Potential	=	4,500 acre feet per year
25% Drought Ordinance	=	7,000 acre feet per year
Conservation Savings (behavioral)	=	500 acre feet per year
Behavioral Savings Potential	=	7,500 acre feet per year

**Total District Potential in 2025** = **12,000 acre feet per year** (Plumbing Code + Technical + Behavioral) The savings potential of daily activities can be illustrated with a simple 'shower scenario':

If a 2.5 gallon-per-minute shower head is replaced with a 2.0 gallon-perminute shower head, the savings can easily be formulated: i.e., **0.5 gallon**per-minute is the *technical savings* potential for that shower head replacement—a **20%** savings per minute. If the person takes a 10-minute shower, then the technical savings would be **5 gallons** (10 minutes x 0.5 gallon-per-minute).

Plus, there is a *behavioral savings potential*. How much more water would be saved if this person reduced their shower time from 10 minutes to 6 minutes? A 4-minute shorter shower would result in *behavioral savings* of **8 gallons** (4-minutes x 2.0 gallons-per-minute).

To recap:

Savings for each shower	=	13	gallons
New 2.0 gpm shower head x 6 minutes	=	<u>12 g</u>	gallons
Original 2.5 gpm shower head x 10 minutes	=	25 g	gallons

Savings elasticity can work in both directions, however, as customers can reduce or increase their water use. Increases in average water use are common whenever water agencies reduce conservation activities in the years following drought emergencies. This phenomenon is seen at MMWD today, where average water use is gradually increasing 15-years after the last drought and following a period of significant decreases in conservation program activity.

On the other hand, increases in conservation activities typically result in reductions in average consumption. This has been observed since 1991 with the enactment of conservation legislation and the implementation of sustained conservation programs in California. Every major water purveyor in California reports that total water consumption has remained virtually unchanged due to conservation—despite significant increases in population:

"While the District's customer base has increased by more than 20% and the population increased by more than 10% between 1970 and 2004, use in 2004 was essentially the same as use in 1970 due to a variety of water conservation efforts now in place."

Source: MMWD UWMP 2005

Page 4

If the District was able to achieve the deployment rates of water-efficient technologies projected in the Maddaus analysis, and if we succeed in securing ongoing behavioral changes among our customers, or when required during infrequent 25% demand reduction rationing events, it could be possible to achieve water use savings of approximately **12,000** acre feet per year by 2025.

It is important to remember, however, that behavioral savings require ongoing effort to sustain them, while technical savings are less dependent on behavior and therefore more reliable. Additionally, it is important to note that the calculations in this Plan describe changes in customer's water *Demand*, and are completely separate from, and different than, MMWD's *supply* or *operational yield* calculations.

The most efficient and cost-effective conservation programs target both behavioral and technical savings. Providing thousands of rebates for toilets and clothes washers is expensive. Instead, if we can inspire (or require) people to buy these fixtures without a rebate, it can cost the District less money and achieve the same water-saving results.

The strategy recommended in this Master Plan places a greater emphasis on 'human resources' than on plumbing fixtures, per se, although both are essential components of each program alternative. Incentive programs, new development ordinances, and building codes generate the technical savings, while school education, public outreach, and customer survey programs create the social awareness needed to harvest the even greater behavioral savings.

There are financial impacts to the District and its customers when normal year water use is decreased *and* when it is increased. This Plan calculates costs and potential water savings associated with decreasing normal year water use in the District. A growing body of research indicates that combined "**triple bottom-line**" benefits—i.e., benefits to the *District* + *Customer* + *Environment*—exceed the relatively simple cost/benefit calculations included in this Plan. Numerous environmental impacts (and costs) are reduced when communities become more water efficient. Staff recommends that the District conduct advanced research to quantify the regional costs and benefits associated with environmental changes resulting from conservation activities. Areas of interest may include climate change, greenhouse gases, energy consumption, waste water, urban runoff pollution, air quality, water quality, invasive plants, fire safety, and changes in native habitats.

Modern water efficiency programs have been operating in California for almost two decades. Since 1991, research and field experience have demonstrated that these programs are the most cost-effective and reliable first line of defense for worsening water problems. Less waste means lower energy use, less urban runoff pollution, extending available storage supplies, more water for fisheries, reduced waste water treatment and discharge, downsizing or eliminating costly new water projects, and being better prepared to weather the next drought.

This Plan examines District conservation programs in the past, present, and future from the 1970s through 2030. It incorporates the history of water shortages, current and future demands, marketing strategies, and the costs and benefits of conservation programs. Information is taken from a variety of sources including the general public, consultants, water agencies, Conservation Action Committee meetings, the Board of Directors, and District staff. The primary source materials referenced in the Plan include:

- Water Conservation Baseline Study Final Report (DMC 1994)
- Water Efficient Conservation Master Plan (Barakat & Chamberlin 1994)
- Water Entitlement and Water Budget Program Report (MMWD Staff 1995)
- Water Conservation Action Plan for 1997 (MMWD Staff 1997)
- Review of Conservation Activities Final Report (Fiske, Stout, Nelson 2001)
- MMWD Urban Water Management Plan 2005 (MMWD Staff 2005)
- 2006 Water Management Report (MMWD Staff 2006)
- Conservation Technical Analysis (Maddaus 2007)
- Memorandum of Understanding Regarding Water Conservation in California (CUWCC, Amended 2007)

A well-structured conservation program is complex and is woven into the social fabric of the community. **Figure 1** shows relationships among elements in a model conservation program:

#### FIGURE 1

#### **RELATIONAL MODEL OF AN URBAN WATER CONSERVATION PROGRAM**



#### **EXECUTIVE SUMMARY**

Today, MMWD is increasing conservation activities in response to rising water demand caused by increasing population and 15 years of relatively drought-free conditions. In 2006 – 2007, the conservation department completed a major reorganization and embarked on a long-term strategy to strengthen relationships with MMWD customers and resource agencies in the region. Every program element is under review and is being revised as needed.

This strategy is based on several basic principles:

- Incentives are the backbone of effective conservation programs.
- Positive conservation messages build customer confidence and goodwill in the District.
- Customer participation increases when programs emphasize services rather than penalties and enforcement of ordinances.
- High-visibility public outreach and education programs are essential in order for conservation programs to achieve sustainable long-term reductions in demand.
- Water pricing is a powerful conservation tool that provides reliable improvements in efficiency.
- A conservation-focused analysis of the District's tiered rate structure, entitlements, connection fees, and water budgets is required.
- A revision of all conservation ordinances is required.

#### Program Analysis

The District has completed two studies of the District's water conservation potential in the past year:

- 2006 Water Management Report (MMWD)
- Conservation Technical Analysis (Maddaus)

In April 2006, the Water Management Report was completed by District staff to accomplish three goals:

- 1. Calculate water savings resulting from conservation activities in the District from 1991 through 2006.
- 2. Reconcile the number of water-efficient plumbing fixtures that have been installed in the District and the number of inefficient fixtures remaining.
- 3. Calculate the maximum *technical* water savings remaining in the District.

Staff calculated current savings in the District to be approximately **3,000** acre feet per year, resulting from program activities and building code changes since 1991. The future *maximum technical savings* was determined to be approximately **7,000** acre feet per year in a normal rainfall year. Therefore, according to these calculations, the total normal year conservation potential in the District is approximately **10,000** acre feet per year (current + maximum technical savings). The Report concluded that the District would need to implement a model conservation program over at least the next ten years in order to obtain these savings (Appendix C, 2006 Water Management Report, pages 1-6).

In October 2006, the District engaged the Maddaus Water Management consulting group to analyze the District's conservation potential, using the Least Cost Planning Decision Support System (DSS) Model they developed. This analysis was completed in April 2007 and provides a detailed account of the savings potential of four program alternatives (Programs A - D) through the year 2030, as well as a drought and demand hardening assessment.

The Maddaus Analysis is used as the primary reference source because of the accuracy of the DSS model and expertise of Maddaus Water Management.

STUDY	CURRENT DEMAND REDUCTION (ac. ft./year)	POTENTIAL DEMAND REDUCTION (ac. ft./year)	PROGRAM COST PER ACRE FOOT SAVED	Average % Customer Saturation
<b>Maddaus*</b> April 2007	Not Calculated	<b>3,000 – 5,400</b> Programs A – D by 2030	\$1,000	10 – 50%
<b>MMWD</b> ** April 2006	2,800 - 3,400	<b>5,400 – 8,600</b> Maximum Savings by 2018	\$1,400	90 – 100%

**Table 1** summarizes the results of both analyses:

## TABLE 1 COMPARISON OF MMWD AND MADDAUS CONSERVATION ANALYSES

\* Appendix A \*\* Appendix C

The Maddaus program alternatives include savings that will be obtained through the effects of state plumbing codes and new development ordinances. These savings will accrue as property owners replace older, less efficient appliances and in new building developments. Except for some minor costs associated with the administration of District ordinances related to plumbing retrofits and plan review, these savings are virtually 'free' from the utility cost perspective.

**Table 2** describes the general elements and savings potential associated with each program option:

Program	Description	2030 Water Savings, AF/Yr	2030 Water Savings with Plumbing Code, AF/Yr
Plumbing Code	No Conservation Program	2,000	2,000
A	Continue Current MMWD Program (Mostly BMPs) Market Penetration 6-12%	1,000	3,000
В	Add New Measures to Current Program Market penetration 10-20%	2,000	4,000
С	Add New Measures to Current Program Market penetration 20-30%	3,000	5,000
D	Add New Measures to Current Program Market penetration 30-50%	3,500	5,500

 TABLE 2

 Maddaus Program Descriptions and Future Water Savings\*

\* Adapted from Maddaus Analysis, Appendix A, Table 8, page 35.

**Table 3** lists the costs for the Maddaus program alternatives (B - D) in years 2008 – 2030. (Note: These three alternatives were selected for comparison based on staff's budget analysis and final program recommendations.)

Program	Average Annual District Cost* (\$1,000s)	Average Annual Customer Cost* (\$1,000s)	Average Annual Community Cost (District + Customer)* (\$1,000s)	Average Annual Community Cost Per Service Connection**	Total District Cost*** (\$1,000s)	Total Customer Cost*** (\$1,000s)	Total Community Cost*** (\$1,000s)
В	\$1,117	\$2,300	\$3,679	\$61.00	\$22,244	\$41,136	\$63,380
С	\$1,316	\$2,709	\$4,333	\$71.00	\$29,269	\$48,454	\$77,723
D	\$1,826	\$3,262	\$5,088	\$84.00	\$36,255	\$60,031	\$96,286

 TABLE 3

 MADDAUS PROGRAM COSTS (2008-2030)

\* Adapted from Maddaus Analysis, Appendix A, Attachment 2, page 51; costs are undiscounted.

\*\* Based on 60,670 accounts. Source: CUWCC 2004 BMP Report.

\*\*\* Adapted from Maddaus Analysis, Appendix A, Table 8, page 35.

Figure 2 graphs the utility costs versus annual water savings in 2030.



FIGURE 2 <u>Maddaus Present Value of Utility Costs versus Cumulative Water Saved in 2030\*</u>

\* From Maddaus Analysis, Appendix A, Figure 11, page 36.

**Table 4** lists the cost per acre foot and potential water savings during this same time period:

Program	District Cost per ac. ft.*	Customer Cost per ac. ft.*	Community Cost per ac. ft. (District + Customer)*	Annual Water Savings Potential in 2030 (ac. ft.)*	Total Water Savings as a % of Total Production in 2030**		
B, C, D	\$400	\$700	\$1,100	4,300 - 5,400	13% - 17%		

 TABLE 4

 MADDAUS COST PER ACRE FOOT & WATER SAVINGS POTENTIAL

\* Adapted from Maddaus Analysis, Appendix A, Table 8, page 35. Savings includes plumbing code.
 \*\* Adapted from Maddaus Analysis, Appendix A, Table ES-2, page 3. Percentage includes plumbing code.

#### MMWD Program Recommendations

By using the Maddaus Analysis as a reference model, and adapting Maddaus Programs B, C, and D to the organizational framework of the District, staff has prepared customized program alternatives for MMWD. The MMWD alternatives are roughly equivalent to Maddaus programs in terms of overall costs and savings targets, and Tables 1 - 4 can be used for purposes of direct comparison.

The MMWD Programs include an Automated Meter Reading (AMR)/Advanced Meter Infrastructure (AMI) and Leak Detection Program that will be operated by the Facilities and Watershed Division, and Conservation Programs 1 - 3 to be operated by the Water Conservation Department. **Table 5** summarizes the costs and savings targets associated with each MMWD program alternative and provides a description of program activities that would be accomplished during the next two-year budget cycle:

Program	AMR/AMI & Leak Detection	Conservation Program #1 (compare to Maddaus Program B)	<b>Conservation Program #2</b> (compare to Maddaus Program C)	Conservation Program #3 (compare to Maddaus Program D)
Description	The District has not had an individual dedicated solely to leak detection in more than five years. Staff recommends that two positions be dedicated to leak detection. These positions would be classified as Utility Workers III, reporting to the Area Field Supervisor in the Facilities and Watershed Division. Based on past experience, it requires five years for one worker to survey all nine hundred miles of pipeline and sixty thousand meters in the District. With two Staff dedicated to leak detection, the task should be accomplished in three years. Automated Meter Reading (AMR) and Advanced Meter Infrastructure (AMI) technologies would replace existing meters and provide numerous technological opportunities for revenue recovery and conservation.	MMWD Program #1 is the "status quo" option – with program expenditures in FY '08/'09 similar to FY '07 levels. Program costs will change in response to inflation, staff salaries and benefits, and program implementation for all programs through 2030. New rebate programs initiated in FY '07 will be funded at partial levels, and customer saturation targets would remain at current levels. The Customer Assistance Program (CAP), water waste, and Time of Service programs will continue at minimum levels; and a new school education program will be initiated (Appendix B, Staff Reports). Some website and database tools will be developed.	Program #2 includes increases in spending for rebates, staffing, and customer services beyond FY '07 levels. New rebate programs initiated in 2007 will be funded at higher levels, and CAP services will increase significantly for all customer types. Three new Conservation Specialist I positions would be added to perform field surveys and provide program support. One existing Conservation Specialist III position would be upgraded to an existing Conservation Coordinator position, and the Mid-Manager position will be upgraded to a Senior Manager position. A new School Education Program will be initiated and funded at increased levels. Improvements to public outreach and marketing, conservation databases, website development, contractor education, and regional development projects will also be accelerated.	Five new Conservation Specialist I positions will be added to perform field surveys and provide program support. One new Conservation Specialist II will be added to support School Education Programs. One existing Conservation Specialist III position will be upgraded to an existing Conservation Coordinator position, and the Mid-Manager position will be upgraded to a Senior Manager. Rebate programs will be funded at higher levels, and very significant improvements in the CAP programs will occur for all customer types. The new School Education Program will be funded at above average levels. Significant improvements will be made in public outreach and marketing, conservation database and website development, contractor education, research, and regional development projects.
FY '08/'09 District Costs (per year)	AMR/AMI – \$1,700,000 Leak Detection – \$272,000	\$1,800,000	\$2,700,000	\$3,300,000
Staff Requirements	AMR/AMI – to be determined Leak Detection – Add 2 positions	No new positions	Add 3 and upgrade 2 positions	Add 6 and upgrade 2 positions
2015 Demand Reduction (ac.ft./year)	AMR/AMI – to be determined Leak Detection – 400*	2,400**	2,800**	3,400**
2030 Demand Reduction (ac.ft./vear)	AMR/AMI – to be determined Leak Detection – 800*	3,500**	3,900**	4,600**

 TABLE 5

 MMWD CONSERVATION PROGRAM ALTERNATIVES

\* Adapted from Maddaus Analysis, Appendix A, Table 4, page 20. Savings are calculated as a "30-year Average".

\*\* Adapted from Maddaus Analysis, Appendix A, Table 8, page 35. Savings includes plumbing code minus Leak Detection Program savings.

#### Program Implementation

#### <u>Strategy</u>

In order to cost-effectively operate MMWD's conservation programs, staff will work collaboratively with private and public partners to the greatest extent possible. This strategy is designed to produce the following results:

- Administrative costs can be reduced or deferred to other service providers.
- Program participation will increase via networking among partners.
- Communication and marketing messages will be more effective as they reach more customers more often.
- Rebate and incentive costs can be reduced more quickly if market transformation occurs at a faster pace—water-saving devices become more available and affordable.
- Public events will cost less to produce and will reach more customers.
- Water savings resulting from behavioral changes will increase with growing public awareness.
- Public support for rate increases may increase if more customers realize direct benefits from District programs.
- Funding opportunities for research and new technologies will increase with greater program visibility and wider regional benefits.

#### Communications and Marketing

MMWD's audience includes all of its 190,000 customers. Reminding them of the need to be mindful about water—use well, not waste—is an ongoing and overarching goal of the District's communications plan. MMWD has identified audiences where it expects to have the greatest impact on water use:

- 1. Residential customers
  - Homeowners with irrigation systems
  - Homeowners' Associations
  - Professional landscapers, plant nurseries, garden supply centers
  - Civic, health, environmental, and conservation organizations
  - Highest water users

#### 2. Commercial customers

- City and county public works departments
- Schools (maintenance supervisors)
- Hospitals
- Private businesses
- Golf courses
- Owners of multi-family residential housing
- 3. Teachers, school-age children, and their families

#### **District Policy**

#### <u>Funding</u>

To provide long-term funding for conservation programs and revenue for District operations, the following funding mechanisms are proposed:

- Regular tier-break adjustments
- Service and/or connection fees dedicated to conservation programs
- On-bill conservation surcharges based on current consumption
- Bond financing secured with conservation surcharges
- On-bill financing (e.g., "Pay-As-You-Save")

#### <u>Ordinances</u>

Revising the existing conservation ordinances, and adding new development ordinances, should be a high priority for the District over the next year. Although overall savings resulting from ordinances will be relatively small, they are highly cost effective. Opportunities exist to increase the effectiveness of ordinances by building enforcement networks with local municipalities and other special districts.

#### **Conclusions**

The District is at a critical turning point in its history and must make significant investments that balance water supply and demand, and fund infrastructure maintenance and operations.

Investments in water conservation programs have reliably reduced water demand in the District for over 30 years and continue to yield consistent results. The conditions now exist for the District to launch an aggressive, sustained conservation campaign and set a new achievement standard for water efficiency in California: Water supplies are limited, new supplies are expensive to develop and impact the environment, the citizens of Marin are knowledgeable and motivated, the District has the in-house expertise to develop and operate programs successfully, and the technologies exist that can reliably reduce consumption in the future.

Although conservation is a cost-effective method that extends available water supplies and provides environmental benefits, under the current District pricing structure, lowered water sales also reduce District revenue. Solving this financial catch-22 is the most critical challenge limiting conservation today.

Finally, it is a political decision that will determine the level of conservation savings the District attains. Only by making a commitment to work together as an agency and a community, to overcome financial and social barriers, and to invest in public education and technology, can the conservation potential in Marin be achieved.

This page intentionally left blank.

#### **MMWD CONSERVATION PROGRAMS**

#### STATEWIDE CONTEXT

California communities are under increasing pressure to meet future water demands. Increasing population, environmental conditions, rising capital and operational costs, recreational needs, agricultural demands, and changing weather patterns are all contributing factors. Statewide, modern conservation programs began in 1991 when MMWD and most other water agencies signed the first *Memorandum of Understanding Regarding Water Conservation in California*. This historic event transformed conservation programs among individual agencies into a statewide coalition dedicated to sustained water use efficiency practices. The *Memorandum* describes 14 Best Management Practices (BMPs) that include everything from toilets to irrigation systems and rate structures. An agency is considered "on track" for a given BMP if it is meeting the implementation schedule. For example, MMWD has completed the implementation of BMP #14, the "Residential Ultra-low Flow Toilet BMP", because an estimated 48,000 ULFTs have been installed in the District since 1994.

For an agency to be 'on-track' with the BMPs is comparable to constructing a building to meet minimum building codes. BMPs were born out of statewide political compromise, and as such, represent a fairly low standard of achievement. Nevertheless, it is a good standard and serves as a starting point for future conservation efforts. If all the BMPs were fully implemented by all water agencies, plus savings from building codes, demand in California would be reduced by approximately 1 million acre feet each year by 2030. This represents approximately 33% of the maximum savings technically available through improved water efficiency practices.

Currently, however, no water agencies have fully implemented the BMPs. As a result, actual water savings equal only one-third of the BMP goal. Lack of resources dedicated to conservation programs is the primary reason for not achieving anticipated savings.

**Figure 3** below is extracted from a presentation of the CUWCC *CALFED Year 4 Comprehensive Report*, and it shows the relative levels of water agency BMP compliance as of 2002.



FIGURE 3 Statewide BMP Compliance

**Table 6** on the following page compares BMP program statistics of seven California water agencies, including MMWD.

Comparison of BMP Program Implementation *							
Water Agency (%) indicates customer saturation	Alameda County Water District	Redwood City	East Bay Municipal Utility District	Irvine Ranch Water District	Long Beach Water Department	San Francisco PUC	Marin Municipal Water District
BMP 1 Residential Surveys	8,014 (11.3%)	368 (1.8%)	67,056 (19.2%)	3,052 (4.0%)	32,697 (40.8%)	89,379 (60.6%)	4,653 (8.4%)
BMP 3 Leak Detection	System audit in 2004	No program reported	2 – Staff 1 – Vehicle	No audit reported	System audit in 2000	Program to begin in 2005	System audit in 2001
BMP 5 Landscape Surveys	842 (16.0%)	22 (1.1%)	2,025 (7.2%)	0 reported	364 (4.0%)	3,165 (14.0%)	97 (9.5%)
BMP 6 2006 Data Clothes Washers	13,324 (18.8%)	1,080 (5.4%)	30,368 (8.7%)	8,435 (9.9%)	6,703 (8.4%)	7,811 (5.3%)	6619 (11.9%)
BMP 9 CII Audits	370 (10.6%)	0 reported	3,393 (14.8%)	50 (1.0%)	2,336 (28.5%)	12,885 (60%)	253 (7.1%)
BMP 14 ULF Toilets	0 reported	0 reported	24,443	26,707	43,419	37,034	48,857
Total Accounts	77,907	22,931	377,318	85,728	89,381	170,670	60,670
Dollars Spent on BMP Programs*	\$683,444	\$660,486	\$4,848,518 ('03/'04 avg.)	\$633,752	\$1,489,364	\$1,016,952	\$973,312 (source:MMWD)
Dollars Spent per Person	\$2.11	\$7.96	\$7.96	\$2.06	\$3.04	\$1.28	\$5.23
Dollars Spent per Ac. Ft. Delivered	\$12.11	\$48.36	\$18.54	\$7.05	\$17.93	\$9.98	\$29.97
Population*	323,250	83,000	1,325,000	308,400	490,000	792,700	186,000
Water Deliveries (Ac. Ft.)*	56,427	13,659	261,560	89,916	83,086	101,870	32,478
Ac. Ft. Saved per Year		563 ac.ft @ \$1,173/ac. ft. in 2005 <sup>1</sup>	18,412 ac.ft./year @ \$263/ac. ft. <sup>2</sup>				3,000 ac. ft./year @ \$324/ac. ft. <sup>3</sup>

TABLE 6

\* Source: <u>http://bmp.cuwcc.org/bmp/read\_only/list.lasso</u>. Important Note: All data is from CUWCC 2004 BMP reports, except where indicated.
1. Source: 2005 UWMP, Table 6.2, City of Redwood City.
2. Source: EBMUD 2004 Water Conservation & Water Recycling Annual Report page E-2.
3. Source: MMWD 2006 Water Management Report, Appendix C, page 6.

#### Page 18

#### FUTURE MMWD PROGRAM GOALS AND OBJECTIVES

#### Goals:

- 1. Produce reliable water savings at the lowest cost
- 2. Protect the natural environment
- 3. Foster positive customer experiences
- 4. Enhance the health of urban forests
- 5. Cultivate the conservation ethic in Marin

#### **Objectives:**

- Promote behavioral conservation actions to the greatest extent possible through sustained public outreach programs
- Develop programs with manufacturers and retailers that leverage bulk discounts on plumbing fixtures
- Enact new development ordinances in coordination with all municipal agencies and special districts
- Enlist local community organizations as program partners to increase the effectiveness and number of customer contacts
- Provide service-oriented programs that enhance customers' lifestyles and property values
- Offer financial incentives that promote market transformation
- Increase public education activities
- Develop technologies that lower program transaction costs and barriers to customer participation
- Participate in regional partnerships related to funding, marketing, and customer education
- Promote programs that protect environmental resources

#### FUTURE MMWD PROGRAM DESCRIPTIONS

Each of the program options listed below is based on a combination of information from the Maddaus Analysis, MMWD's budget projections, and staff's professional judgment. In general, the programs are designed to maximize staff resources by leveraging community resources and to minimize costs through volume purchases and technology.

- Automated Meter Reading (AMR) & Advanced Meter Infrastructure (AMI)
- Leak Detection Program
- Conservation Program #1
- Conservation Program #2
- Conservation Program #3

#### Page 19

#### <u>Automated Meter Reading (AMR) &</u> <u>Advanced Meter Infrastructure (AMI)</u>

- Total FY '08/'09 Utility Costs:
- Average Annual Cost (10 years):
- Total 10-year cost: (including replacement of 60,000 meters)
- Staff Requirements:
- Demand Reduction potential 2015:
- Demand Reduction potential 2035:
- \* Based on an estimated total project cost of \$15 million at a nominal annual interest rate of 4.160%

#### **Background**

District staff has conducted research on various new meter reading technologies over the past several years. Many unanswered questions remain regarding potential costs for additional staff resources, demand reduction potential, and changes in revenue that may result from more accurate meter readings. In order to answer these questions, MMWD has engaged an industry consultant that is currently conducting a detailed assessment that will be completed by August 2007. Once this assessment is completed staff will have the information needed to make a more informed recommendation on how the District should proceed and what the costs associated with such a program would be.

While new technologies continue to be developed at a rapid pace, AMR and AMI systems are becoming increasingly cost-effective tools offering a variety of benefits. Automated Meter Reading (AMR) refers to the ability to collect data from a meter remotely and automatically via various different methods, RF (radio frequency) wireless, power line, and telephone lines. Advanced Meter Infrastructure (AMI) is the next, newest step in meter technology. AMI represents an infrastructure that utilizes smart meters with advanced two-way communications that enable utilities to meet their business needs for meter data collection. Utilities with AMI systems have the ability to collect data on an hour-by-hour basis, if needed. And, it enables consumers to actively and frequently collect their own consumption data.

The District has approximately 60,000 meters in the system. Of these, approximately 55,000 are residential meters; 2,000 are irrigation meters; and the remainder serves businesses and/or office complexes. The District's current Meter Change Program (MCP) is based on a 20-year meter life. Meters are warranted by the manufacturer for 15 years or 1.5 million gallons of water/2,000 ccfs. Within the next six years, 32,000 of the District's meters are scheduled for replacement, representing more than 50% of the District's meters.

The cost of the basic meter currently being installed by the District is approximately \$48.00, while the cost of an AMR/AMI meter is approximately \$180.00, not including the cost of installation.

\$1,700,000 per year \$1,700,000 per year\* \$14 – 18 million

To be determined To be determined To be determined

#### **Potential Benefits**

AMR/AMI meters can provide the District and its customers with a wide range of benefits that are unavailable with manually read meters. Some of these benefits are summarized below:

#### Meter Operations

- Reduces stuck meter maintenance calls
- Decreases vehicle traffic
- Decreases vehicle carbon gas emissions
- Decreases vehicle maintenance and fuel costs

#### Customer Service/Finance

- Improves service by providing in-home water usage monitoring by consumers
- Facilitates monthly billing
- Decreases re-reads/mis-read trips
- Decreases move-in/move-out trips
- Decreases vehicle traffic on the road
- Decreases vehicle carbon gas emissions
- Decreases vehicle maintenance and fuel costs

#### Engineering

• Decreases unaccounted for water loss

#### Water Conservation

- Provides leak detection, data logging, and usage monitoring
- Allows time-of-day use monitoring and reporting
- Facilitates accurate water savings calculations

#### Safety

• Reduces Workmen's Compensation liability claims

#### Leak Detection Program

<ul> <li>Total FY '08/'09 Utility Costs*:</li> </ul>	\$230,000 per year
Average Annual Cost (30 years):	\$190,000
Staff Requirements:	Add 2 positions
Leak Reduction Potential	
(30-year average, ac. ft./year)**:	1.5% = 400 acre feet
	2.0% = 500 acre feet
	3.0% = 800 acre feet

- \* Note: The leak detection program is included in each of Maddaus Programs B, C, and D—see Appendix A on page 44 of the Maddaus Analysis for details on leak detection program costs. However, since the leak detection program will be operated separately from the MMWD Conservation Programs, staff has calculated the costs for this program separately.
- \*\* Adapted from Maddaus Analysis, Table 4, page 20.

#### **Background**

The District measures annual "unaccounted for water losses" due to leaks and other unknown sources. There has been a marked improvement in loss control since the 1970s, when the average system losses were around 14%. Today, due to pipeline replacements and leak repairs, the average system loss is 10%. This 4% reduction in losses equates to a savings of about 1,200 acre-feet of water each year.

The District has not had an individual dedicated solely to leak detection in more than five years. Leak detection/survey is now handled by the Valve Technicians on an incident by incident basis. The Valve Technicians are first responders to all mainline leaks, mainline shut downs for the crews and contractors, consumer calls, and meter turn-on/offs. Under this scenario, it is difficult to dedicate the required blocks of time to complete leak survey work in a systematic manner.

The process of leak detection involves pinpointing a specific leak location. This process assists staff in completing repairs with a minimal amount of excavation. Leak surveying is accomplished by canvassing the entire water distribution system. Using sonic leak detection equipment, technicians are able to locate leaks that may be hidden from view, because they are not yet surfacing. The leaks typically found during a survey include District leaks—mainline and service line—and consumer leaks.

#### **Benefits**

Detecting and repairing leaks in the District's delivery system and on a customer's property is a highly cost-effective way to conserve water that the District has already purchased, treated, and delivered.

For example: Every 1% decrease in system water loss equals a savings of about 300 acre-feet per year. At Tier I prices, the retail value of this water is over \$300,000 every year. By factoring in the additional benefits derived from avoided costs for potential property damages, and treating and pumping this water, the value of leak detection increases. Leak detection programs can provide other substantial benefits, including:

- Safeguarding public health and property. Discovery of leaks before they turn into larger main breaks and reduce the likelihood of property damage and public safety.
- Improved public relations. Consumers appreciate maintenance of the water system and help with consumer problems. Field personnel doing leak detection provide visual assurance that the system is being maintained.
- Reduced legal liability. By early detection, we are able to repair leaks before they become large main breaks. This helps protect the District from expensive lawsuits.
- Reduced disruption to consumers. More leaks are repaired on a planned basis rather than developing into major breaks that disrupt service.

#### **Recommendations**

Staff recommends that two positions be dedicated to leak detection. These positions would be would be filled by in-house candidates, and classified as Utility Workers III; based on the job skills and experience the position requires. They would report to the Area Field Supervisor in the Facilities and Watershed Division.

Based on past experience, it requires five years for one worker to survey all nine hundred miles of pipeline and sixty thousand meters in the District. With two staff dedicated to leak detection, the task should be accomplished in three years. New technology is available for leak detection, including data loggers that are placed at various locations within the District's system, such as service laterals, system mainline valves and hydrants. These data loggers are programmed to turn on automatically at different intervals during the night and 'listen' for leaks. This information is downloaded to a computer and then used to pinpoint leaks. This is a great tool and will be used to conduct a thorough and comprehensive leak survey, especially at program start up. Based on the fact that it has been over five years since the last leak survey, staff is recommending the following approach be taken to initiate this program:

- Survey the District's distribution system by walking the entire service area, listening on all service laterals and other points of contact. This will enable the surveyor to find the smaller meter leaks, service leaks, and also consumer leaks that would not be picked up by the data loggers.
- A large portion of the work would be done at night. Typically, the best hours for leak detection are between 10 pm and 5 am. These are the hours of the least amount of water consumption and there is not the noise from traffic that you would find during daylight hours.

#### First-Year Startup Costs

(2) Utility Worker III	\$140,000 (@ step 5)
Benefits	50,000
(2) Trucks	60,000
Sonic Leak Detectors & Pipe Locators	10,000
Total First-Year Costs	\$260,000

Staff will reevaluate the leak detection program after approximately three years, following the initial survey of the entire water distribution system. Recommendations about future staffing and funding levels will be provided to the District based on the volume of survey activity.

Costs for mainline, service, and meter leaks would be borne by the respective cost centers already in place. Although additional staff and equipment for leak repairs may be needed, depending on the volume of leaks discovered, staff does not anticipate this occurring at this time.

The Leak Detection Program could start as soon as the positions are advertised by Human Resources and filled.

#### Page 24

#### Conservation Program #1

(compare to Maddaus Program B)

- Total FY '08/'09 Utility Costs: •
- Total Utility Costs FY 2008 2022: •
- **Staff Requirements:** •
- **Market Penetration:** •
- Demand Reduction with plumbing code 2015: .
- Demand Reduction with plumbing code 2030: •
  - MMWD calculated the "Total FY '08/'09 Utility Costs" based on staff's budget analysis using actual full-time equivalent costs and program expenditures.
  - \*\* "Total Utility Costs FY 2008 2022" are based on the Maddaus Program B cost projections listed on page 51 of Appendix A and have been adjusted by MMWD to include a 3% inflation rate.
  - \*\*\* Demand Reductions are derived by subtracting leak detection program savings of 1.5% from the Conservation Program B + plumbing code savings listed in Table 8 on page 35 of the Maddaus Analysis.



The MMWD conservation programs are currently operated using an interactive team approach. Figure 4 illustrates this interactivity, using multiple lines that connect staff positions. These lines indicate that a staff member coordinates job tasks with multiple teams. MMWD Program #1 is the "status quo" option-with program expenditures in FY '08/'09 similar to FY '07 levels. Program costs will change in response to inflation, staff salaries and benefits, and program implementation for all programs through 2030. New rebate programs initiated in FY '07 will be funded at partial levels, and customer saturation targets would remain at current levels. The Customer Assistance Program (CAP), water waste, and Time of Service programs will continue at minimum levels; and a new school education program will be initiated (Appendix B, Staff Reports). Some website and database tools will be developed.

- \$1.8 million per year\* \$25.000.000\*\*

  - 10 20%
  - 2,400 acre feet per year\*\*\*
  - 3,900 acre feet per year\*\*\*
  - No change

**Table 7** compares **MMWD Conservation Program #1** budgeted activities in FY'08/'09

 with the number of activities listed in the Maddaus Program B analysis:

## TABLE 7 Program #1 Budgeted Activities FY '08/'09

Activity Description	MMWD Budgeted Activities per year in FY '08/'09	Maddaus Program B Activities per year	Maddaus Water Use Reduction Target	Maddaus Market Penetration Goal
Indoor Residential Survey BMP 1	440	588	5%	30%
Outdoor Residential Survey BMP 1	440	588	10%	30%
Landscape Budgets BMP 5	300	116	15%	90%
Landscape Survey BMP 5	20	7	10%	1.8%
Clothes Washers BMP 6	1,200	1,200	34%	26%
Public Education BMP 7	51,000	51,000	1%	100%
Commercial Audits BMP 9	12	49	12%	12%
ULFT Ordinance BMP 14	2,000	575	Equals service change rate	60%
Rain Sensor Retrofit	1,500	3,259	30%	9%
San Quentin Toilets	300	200	NA	54%
HET Rebates	1,000	1,085	20%	57%
Homeowner Classes	54	272	5%	5%
Coin-Op Washers	71	22	15%	34%
Irrigation Incentives	275	212	5% Residential 15% Commercial	15%
Hotel Retrofit	50	48	40%	20%
Commercial Rebate	150	24	10%	10%
HE Urinal Rebate	100	24	10%	71%
New Development Ordinance	Yes	Yes	NA	100%
#### Conservation Program #2

(compare to Maddaus Program C)

- Total FY '08/'09 Utility Costs:
- Total Utility Costs FY 2008 2022:
- Staff Requirements:
- Market Penetration:
- Demand Reduction with plumbing code 2015:
- Demand Reduction with plumbing code 2030:
- \$2.7 million per year\* \$36,000,000\*\* Add 3 positions/Upgrade 2 20 – 30% 15: 2,700 acre feet per year\*\*\*
- 030: 4,200 acre feet per year\*\*\*
  - \* MMWD calculated the "Total FY '08/'09 Utility Costs" based on staff's budget analysis using actual full-time equivalent costs and program expenditures.
  - \*\* "Total Utility Costs FY 2008 2022" for program C are based on 81% of the Maddaus Program D cost projections listed on page 51 of Appendix A and have been adjusted by MMWD to include a 3% inflation rate.
  - \*\*\* Demand Reductions are derived by subtracting leak detection program savings of 2.0% from the Conservation Program C + plumbing code savings listed in Table 8 on page 35 of the Maddaus Analysis.





Program #2 includes increases in spending for rebates, staffing, and customer services beyond FY '07 levels. New rebate programs initiated in 2007 will be funded at higher levels, and CAP services will increase significantly for all customer types. Three new Conservation Specialist I positions would be added to perform field surveys and provide program support. One existing Conservation Specialist III position would be upgraded to an existing Conservation Coordinator position, and the Mid-Manager position will be upgraded to a Senior Manager position. A new School Education Program will be initiated and funded at increased levels. Improvements to public outreach and marketing, conservation databases, website development, contractor education, and regional development projects will also be accelerated.

**Table 8** compares **MMWD Conservation Program #2** budgeted activities in FY '08/'09with the number of activities listed in the Maddaus Program C analysis:

TABLE 8	
Program #2 Budgeted Activities FY	<u>'08/'09</u>

Activity Description	iption MMWD Budgeted Maddaus Activities per year Acti in FY '08/'09 per		Maddaus Water Use Reduction Target	Maddaus Market Penetration Goal				
Indoor Residential Survey BMP 1	575	588	5%	30%				
Outdoor Residential Survey BMP 1	575	588	10%	30%				
Landscape Budgets BMP 5	360	116	15%	90%				
Landscape Survey BMP 5	e Survey 60 7		10%	1.8%				
Clothes Washers BMP 6	1,440	1,200	34%	26%				
Public Education BMP 7	blic Education 100,000 51,00		1%	100%				
Commercial Audits BMP 9	rcial Audits 40		26%	12%				
ULFT Ordinance BMP 14	2,000	575	Equals service change rate	60%				
Rain Sensor Retrofit	ain Sensor 4,483		30%	9%				
San Quentin Toilets	aentin 300 200		NA	54%				
HET Rebates	1,800	00 1,671 30%		57%				
Homeowner Classes	297	542	5%	10%				
Coin-Op Washers	150	37	25%	34%				
Irrigation Incentives	750 518 1		10% Residential 25% Commercial	15%				
Hotel Retrofit	137 Rooms	72 Rooms	60%	20%				
Commercial Rebate	ercial 150 24		10%	10%				
HE Urinal Rebate	250	50	20%	71%				
New Development Ordinance	Yes	Yes	NA	100%				

#### **Conservation Program #3**

(compare to Maddaus Program D)

- Total FY '08/'09 Utility Costs:
- Total Utility Costs FY 2008 2022:
- Staff Requirements:
- Market Penetration:

•

• Demand Reduction with plumbing code 2015:

Demand Reduction with plumbing code 2030:

3,000 acre feet per year\*\*\* 4,600 acre feet per year\*\*\*

Add 6 positions/Upgrade 2

\$3.3 million per year\*

\$44,000,000\*\*

30 - 50%

- \* MMWD calculated the "Total FY '08/'09 Utility Costs" based on staff's budget analysis using actual full-time equivalent costs and program expenditures.
- \*\* "Total Utility Costs FY 2008 2022" are based on the Maddaus Program D cost projections listed on page 51 of Appendix A and have been adjusted by MMWD to include a 3% inflation rate.
- \*\*\* Demand Reductions are derived by subtracting leak detection program savings of 3.0% from the Conservation Program D + plumbing code savings listed in Table 8 on page 35 of the Maddaus Analysis.



Program #3 increases program expenditures for staffing and customer services beyond Program #2 levels. Five new Conservation Specialist I positions will be added to perform field surveys and provide program support. One new Conservation Specialist II will be added to support School Education Programs. One existing Conservation Specialist III position will be upgraded to an existing Conservation Coordinator position, and the Mid-Manager position will be upgraded to a Senior Manager. New rebate programs initiated in 2007 will be funded at higher levels, and very significant improvements in the CAP programs will occur for all customer types. A new School Education Program will be initiated and funded at above average levels. Significant improvements will be made in public outreach and marketing, conservation database and website development, contractor education, research, and regional development projects.

FIGURE 6 Program #3 Organizational Chart

 Table 9 compares MMWD Conservation Program #3 budgeted activities in FY '08/'09

 with the number of activities listed in the Maddaus Program D analysis:

Activity Description	MMWD Budgeted Activities per year in FY '08/'09	Maddaus Program D Activities per year	Maddaus Water Use Reduction Target	Maddaus Market Penetration Goal			
Indoor Residential Survey BMP 1	1200	588	5%	30%			
Outdoor Residential Survey BMP 1	1200	588	10%	30%			
Landscape Budgets BMP 5	750	116	15%	90%			
Landscape Survey BMP 5	120	7	10%	1.8%			
Clothes Washers BMP 6	1,750	2,700	34%	26%			
Public Education BMP 7	120,000	51,000	1%	100%			
Commercial Audits BMP 9	165	123	26%	12%			
ULFT Ordinance BMP 14	2,500	575	Equals service change rate	60%			
Rain Sensor Retrofit	4,900	6,085	30%	9%			
San Quentin Toilets	400	200	NA 54%				
HET Rebates	1,980	1,671	30%	57%			
Homeowner Classes	ner 325 542		5%	10%			
Coin-Op Washers	150	37	25%	34%			
Irrigation Incentives	1,000	57         25%         34%           00         1,036         10% Residential 25% Commercial         15%		15%			
Hotel Retrofit	150 Rooms	72 Rooms 60%		20%			
Commercial Rebate	165	24	10%	10%			
HE Urinal Rebate	275	50	20%	71%			
New Development Ordinance	Yes	Yes	NA	100%			

# TABLE 9 Program #3 Budgeted Activities FY '08/'09

#### CURRENT MMWD PROGRAM STATUS

Since 1991, MMWD has completed, or is on track to complete, 11 out of 14 BMPs. This is a very good track record compared to many other water agencies in California. The District has excelled in programs to install ULFTs, residential showerheads and high-efficiency clothes washers; promote public information; and implement conservation pricing. Areas needing improvement include customer site surveys, landscape equipment retrofits, and school education programs. A complete history of the District's conservation achievements is included in Appendix C, the *2006 Water Management Report*.

Four new conservation programs have been approved and implemented in 2006 – 2007: the Conservation Action Committee; High-Efficiency Toilet Rebate Program; Landscape Equipment Rebate; and Conservation Outreach and Marketing Programs. A new School Education Program proposal (Appendix B) has also been prepared for Board review.

In order to track and report on the progress of conservation programs at the District, new database tools are currently under development. These tools will be integrated with the existing SAP system, Access databases, and GIS programs. When completed, the District will have the ability to track conservation activities for each service address and report on changes in water consumption before and after activity. Additionally, landscape area measurements and the water budget process will be automated and allow significant increases in staff productivity.

#### Collaborations are under way with the following community partners:

School Education Programs

- Next Generation
- California Regional Environmental Education Community (CREEC) Network
- Environmental Education Council of Marin
- The Bay Institute
- School Environment Education Docents (SEED)
- PlayFair

Landscape Programs

- Marin Master Gardeners
- Bay-Friendly Landscaping and Gardening
- Irrigation manufacturers and retailers
- California Landscape Contractors Association
- U.C. Cooperative Extension

Commercial and Residential Programs

- Bay Area Public Agencies
- Intergy
- Electric & Gas Industries Association (EGIA)
- California Youth Energy Services

#### PROGRAM ACTIVITY LEVELS

Customers are responding positively to MMWD's customer-friendly conservation strategy. For example, the new high-efficiency toilet rebate program (started in January 2007) has issued more than 300 toilet rebates over the first 3-month period, and interest in this program continues to grow. To put this number of rebates into perspective, other larger water agencies in the Bay Area completed only 150 – 300 HET rebates in all of 2006, while Redwood City installed 4,800 residential toilets between 2005 and 2006 with their direct-installation program.

MMWD's high-efficiency clothes washer program also continues to rebate an average of 1,200 washers each year, after 12 years in operation and more than 9,000 rebates. Staff calculates that as many as 40,000 washers are eligible for rebates in the next 10 years (Appendix C, 2006 Water Management Report, page 14).

This positive response to District programs can be attributed to several demographic characteristics—which are indicators that conservation programs can maintain high activity levels in Marin:

- Residential accounts represent more than 90% of the customer base.
- Residential remodeling projects are very common in Marin and provide opportunities to upgrade plumbing fixtures.
- The average construction age of houses is 1968, prior to enactment of the National Energy Policy Act in 1992 that mandated water efficient plumbing fixtures, indicating a larger than average pool of older, less efficient fixtures.
- Marin is an affluent and highly educated community, with fewer economic barriers and a strong history of participation in conservation activities.
- Public opinion surveys and comments indicate that a majority of residents see conservation as the preferred water supply option.
- Conservation is a cost-effective method to extend limited water supplies and provide a host of environmental benefits to the community.

#### **COMMUNITY INPUT**

Marin residents are extraordinarily interested in environmental issues, including water conservation. District customers have contributed many outstanding ideas intended to improve the environment through natural resource conservation. In 2006 – 2007, Staff conducted two Conservation Action Committee meetings and a "HydroForce Workshop" at College of Marin, and also published a notice about the Master Plan in the March/April edition of *On the Water Front* to solicit input on the future of conservation programs in the District. Staff has recorded the following ideas voiced at Board meetings, public events, during informal conversations, and from e-mails and letters:

- Expedite installation of water-saving devices for large municipal water users by providing direct-installation programs that remove financial and institutional barriers.
- Encourage the use of rain water catchments, grey water, and wells to offset irrigation requirements.
- Increase public awareness through highly visible logos, messages, and product branding.
- Institute a District conservation fee on each water bill that would be removed during any billing cycle that the customer stayed within a set water budget.
- Improve the District's phone system and website so it is easier to contact Staff, participate in incentive programs, and get information.
- Create demonstration projects (gardens, art, retrofit public facilities) to increase awareness.
- Proactively contact customers who are over their water budget to offer program assistance.
- Encourage the use of sub-meters to help customers self-monitor their water use throughout the month.
- Work with manufacturers to provide local "time of purchase" rebates to simplify and enhance incentive programs.
- To highlight the relationship between water and power, a special brochure— "101 Ways to Save Water and Power"— was created.

- Coordinate with relevant programs and ordinances in the County, Cities, wastewater districts, and other water agencies.
- Expand recycling and storm water capture systems throughout the District.
- Increase education programs and internships for students.
- Work with retailers and manufacturers to add labels to plants and irrigation equipment to help consumers identify waterconserving products.
- Implement on-bill financing opportunities like the "Pay-As-You -Save" program.
- Conduct pilot studies on a community or neighborhood scale.
- Study the potential to reduce evaporation from surface reservoirs.
- Require auto dealerships and rental agencies to use recycling car washes.
- Study the potential for composting toilets.
- Sponsor friendly community competitions offering prizes and recognition to the most efficient participants.

Page 33

The following section titled the 'Conservation Scrapbook' is provided in order to illustrate many of the activities currently underway or completed in the past year by the conservation department:

#### CONSERVATION SCRAPBOOK - 2006/2007

Analysis	Commissioned the <b>Conservation</b> <b>Technical Analysis</b> by Maddaus Water Management (Appendix A). A detailed technical analysis of the conservation savings potential through 2030.
	District staff completed the <i>MMWD 2006 Water</i> <i>Management Report</i> (Appendix C) in April 2006. This history of conservation program accomplishments from the 1970s through April 2006 provides a detailed account of the water savings achieved in the past and calculates the maximum technical savings in the future.
Education	Image: Note of the image of
	Qualified Water Efficient Landscape Training Program – MMWD partnered with the Sonoma County Water Agencies, College of Marin, the California Landscape Contractors Association and numerous landscape professionals to develop a comprehensive certification program. Over a period of weeks, the program trains landscape workers in the principles of water conservation, integrated pest management, and irrigation systems.

Page 34



The Northern California Water Management & Technology Education Center



This page intentionally left blank.

## POLICY RECOMMENDATIONS

Water supply and water demand are two sides of same decision making equation. If either side is over- or under-estimated, serious financial and social consequences can occur. Policies that encourage water efficiency through rate structures, incentives, and public awareness messaging make the best use of public dollars and, at the same time, reduce impacts to the environment and the cost of new water supply projects.

The job of conservation programs is to stabilize long-term consumption patterns at the lowest level possible. This task must be accomplished cost-effectively, and the programs must be embraced by a majority of customers in order to succeed.

Staff recommends an overall policy strategy that encourages participation by the greatest number of customers. This strategy differs from recommendations in the District's 1994 Master Plan, which contained programs to target customers using the greatest amount of water. Facts that support this recommendation include:

- Data derived from site survey experience shows that customers in the "top 20%" include parks, golf courses, and estates with extensive landscaping. When these sites have quality irrigation systems and are professionally managed, potential water savings can be relatively small. Residential customers in the "top 20%" have proven to be particularly resistant to changing water use behaviors. Resources spent to influence this small group of customers are often ineffective.
- Approximately 70% 95% of residential accounts are billed at the Tiers 1 and 2 levels. Although each of these accounts uses less water than customers in Tiers 3 and 4, the water use efficiency is often low. By casting a wider program "net", the cumulative savings potential is increased.
- Conservation programs that reach a larger customer base increase the probability for behavior-based savings. Experience during drought emergencies shows that short-term savings can be as high as 65%. Long-term savings would be substantial if even 10% of customers implemented behavioral changes (e.g., turning off faucets, taking shorter showers, or reducing minutes on sprinkler timers).
- Customer surveys done in 2003 and 2005 indicate that more than 80% of customers support increased conservation, while 57% of those asked were not aware of any programs at the District. Thirty-seven percent had done nothing to conserve water.

The most successful conservation programs provide financial support, services, and information to help customers reduce costs and enhance their lifestyle. We have learned that:

- Incentives are the backbone of effective conservation programs. Incentives reinforce the District's role as a good steward of public resources.
- Positive messages about conservation build customer confidence and goodwill in the District. For example, in the past year, MMWD has shifted the conservation message away from penalties and enforcement action to focus on services and rebates. This shift has resulted in greater customer participation in programs, and a significant reduction in antagonistic phone calls.
- High visibility public outreach and education programs are essential for conservation programs to achieve sustainable long-term demand reductions. When asked, most customers want to help take care of Marin's natural resources.
- Water pricing is a powerful conservation tool that can provide reliable improvements in efficiency.
- Staff recommends a comprehensive study of the District's tiered rate structure, entitlements, connection fees, and water budgets be performed. As many as one-half of the water budgets in the District are inaccurate and based on inefficient historical use patterns.
- Conservation ordinances and penalties can be effective if they are enforceable. New development ordinances and conservation requirements attached to re-development permits are enforceable and should be a primary goal for MMWD during the next year. Widespread adoption of conservation requirements by local permitting agencies would yield significant savings at a very low cost to the District.

#### <u>Funding</u>

To provide long-term funding for conservation programs and revenue for District operations, the following funding mechanisms are proposed:

- Regular tier-break reviews and adjustments, as warranted, to monitor consumption trends and maintain customer awareness about the importance of conservation.
- Service and/or connection fees that fund conservation programs and offset revenue losses resulting from lowered consumption.
- On-bill conservation surcharges based on current consumption. Surcharges would be waived in any billing period where consumption was within a pre-determined water budget. Surcharges would be added in any billing period where use exceeded budgeted amounts. Budgets would be determined based on number of residents and landscape area.
- Bond financing to provide capital for conservation plumbing fixtures, secured with conservation surcharges.
- On-bill financing, such as the "Pay-As-You-Save" system or other zerointerest loan option.

#### COST SAVING STRATEGIES

Staff will work collaboratively with agencies, organizations, retailers, and manufacturers that can provide co-sponsoring, or in-kind support for MMWD's programs. Cost savings will be realized in several key program areas:

- Administrative costs can be reduced or deferred to other service providers.
- Program participation will increase via networking among customers.
- Communication and marketing messages will be more effective as they reach more customers more often.
- Rebate and incentive costs can be reduced more quickly if market transformation occurs at a faster pace—water-saving devices become more available and affordable.
- Public events will cost less to produce and reach more customers.
- Water savings from devices and behavioral changes will increase more rapidly with growing public awareness.
- Political support for rate adjustments may increase if more customers realize direct benefits from District programs.
- Funding opportunities will be increased due to greater program visibility and regional impacts.

This page intentionally left blank.

## **CONSERVATION MARKETING PLAN**

#### <u>Background</u>

Historically, Marin Municipal Water District has been at the forefront of promoting water conservation. But record rainfall in recent years has lulled the public's concern about water conservation. The District has not communicated about the need for conservation as aggressively as in the past. Marin's water use is currently at its highest levels for the quarter since 1987.

Marin's water supplies are limited and precariously tied to annual rainfall. Marin is never more than one year away from a drought.

While drought conditions do not currently exist, MMWD recognizes the need to reinvigorate awareness of the need to use water wisely among customers. This has recently become even more urgent, as Marin's access to Sonoma County water may be curtailed because of last winter's scant rainfall.

Marin residents have a strong connection to the natural environment, are highly informed about environmental issues, and have rallied to conserve water in past drought years. Research and history indicate they are willing to "do the right thing" if given the information and support to do so. With environmental concerns at an all-time high, the public is highly receptive to taking action that can advance environmental solutions.

Water-saving technologies have evolved since Marin was last hard hit by drought, and MMWD is stepping up efforts to replace toilets, urinals, washing machines, and sprinkler systems with models offering greater efficiency. But MMWD's conservation messages cannot be limited to the installation of water-saving devices, "tips," rebates, or telling people that saving water is a good idea. As a public utility capable of reaching a vast customer base, and as the county's largest user of electricity, MMWD has an opportunity—and a responsibility—to assume a leadership role in conservation.

We recommend that MMWD position itself as a responsive, well-run public agency committed to conservation. This can be addressed by reinforcing conservation messaging throughout our communications with our customers, and by modeling conservation, leadership, and customer care throughout our operations.

#### Messaging and Positioning

Connecting MMWD to what our customers believe in and care about is essential to garnering their trust and buy-in.

Currently, the District isn't communicating a coherent conservation message. There is an abundance of collateral material offering conservation tips, technical fixes, and rebates. This information is by and large technical, not clearly targeted to intended audiences, and does not connect to the conservation values that are deeply held among people living in Marin. The goal isn't to inform customers about MMWD or what it does. The goal is to build a reservoir of confidence and trust, which will be critical should conditions require MMWD to ration water use. MMWD needs to focus less on what it does than on the benefits provided.

The core strategies of MMWD's conservation campaign are to:

- 1. Connect the District's values to community values;
- 2. Make MMWD user-friendly and a "partner" in its interactions with customers;
- 3. Provide customers with easy-to-adopt programs and behaviors for using less water;
- 4. Demonstrate that the District is itself taking steps to conserve and sustain Marin's environment.

#### <u>Audience</u>

MMWD's audience includes all of its 190,000 customers. Reminding them of the need to be mindful about water—use well, not waste—is an ongoing and overarching goal of the District's communications.

MMWD has identified audiences where it expects to have the greatest impact on water use as well as awareness of water conservation ethic and education (see accompanying grid for breakout of targets and potential actions):

- 1. Residential customers
  - Homeowners with irrigation systems
  - HOAs
  - Professional landscapers, plant nurseries, garden supply centers
  - Civic, health, environmental, and conservation organizations
  - Highest water users
- 2. Commercial customers
  - City and county public works departments
  - Open space and parks departments
  - Schools (maintenance supervisors)
  - Colleges (maintenance, programs)
  - Hospitals
  - Private businesses
  - Golf courses
  - · Owners of multi-family residential housing
  - Visitors/Hotels/Restaurants
- 3. Teachers, school-age children, college students and their families

(			<b>CONSERVATION MARKETING</b> target audiences, programs, strategies												
MARIN MUNICIPAL WATER DISTRICT		REB	BATES		INCENTIVES			PUBLIC EDUCATION			PARTNERSHIPS C			HANGE /	AGENTS
desco	omstudios								nce						
	ACTIONS						bags, etc)	em	asion, Ordina	ps, Training		omotion			S
	AUDIENCES	Urinal/Toilet	Washer	Irrigation	Bill Savings %	Staff visit	Premiums (Bottles,	Signage, icon, embl	Consultation, Persu	Education, Worksho	Certification	CoBranding & CoPro	Sponsorship	Pilot Projects	Front Line Markete
_	Homeowner/tenant	•	•	•	•	•	•	•	•	•					•
NTIA	Neighborhood Assn					•	•	•	•	•		•	•.	•	•
SIDE	HOAs			•	•	•	•	•	•	•	•	•	•	•	•
RE	MultiFamily Bldg	•	•	•	•	•	•	•	•					•	•
	Building/Grounds Manager	_	-	•	•	•	•	•	•	•	•				•
_	Business Conden/complex		•		•		•					•	•	•	
RCIA	Garden/supply									•	•	•	•	•	
WW	Rotail													****	
8	Restaurant			•	•	•	•								•
	Landscapers			•	•	•	•	•	•	•	•	•	•	•	•
_	County	•	-	•	•	•	•		_		•		•		•
IPAI	Cities	•		•	•	•	•	•	PING		•	•	•	•	•
NIN	Special Districts	•		•	•	•	•	•	VELO		•	•	•	•	•
×	Hospitals	•	•	•	•	•	•	•	MO		•	•	•	•	٠
-	K12	•		•	•	•	•	•		•	•		•	•	•
TION	College of Marin (2)			•	•	•	٠	•		•	•		•	•	•
DUCA	Dominican University	•	•	•	٠	٠	٠	•			٠	•	•	•	•
=	PTAs							•					•		•
S	Haalah						-				-		-		-
ROFI	Sustainables						•	•		•					•
ONP	Social Services (assist)	•					•	•				•	•		•
& NC	Social Service (ie, Rotary)						•	•				•	•	•	•
ONS	Trade Associations	•		•	•	•	٠	•			•	•	•		•
ZATI	<b>Chambers of Commerce</b>	•		•	•		•	•				•	•		•
SANI	Environmental & Energy						•	•		•	٠		•	•	•
ORG															
IONALS	Developers	•	•		•	•	•	•	•	•	•	•	•	•	•
	Inspectors							•		•	•	•			•
ESS	Architects	•		•	•	•	•	•		•	•	•	•	•	•
PRO	Landscape Architects			٠	•	•	•	•	•		•	•	•	•	•
ATE	Agents & Brokers							•	•	•	•	•		•	•
LEST	Contractors		•		•	•	•	•	•	•	•	•	•	•	•
REAL	Roard of Realtons	•	•	•	•		•	•	•	•		•	•		•
	board of healtons	•						•		•		•			•

#### <u>Trends</u>

In addition to developing conservation programs and incentives—some of which are included below—MMWD can have a greater impact by connecting its messages with issues the media is currently covering, as well as those of concern to customers.

Some examples of current media trends that MMWD can tie to are:

- Alternative energy/energy savings
- Bottled water
- Carbon trading/offsets
- Clean Technologies
- Climate change
- Dams
- Ecological footprint
- Endangered species
- Environment and health
- Fitness
- Fuel efficiency
- Gas prices
- Global warming
- Green building
- Green business
- Greenhouse gases
- High efficiency light bulbs
- Land use
- Organics and local food
- Outdoor recreation
- Recycling
- Salmon
- Sustainability
- Transportation
- Voluntarism
- Waste reduction

#### **Recommendations**

- MMWD establish and publicize a measurable, achievable, short-term target to rally around and demonstrate the water savings that Marin can achieve with small changes in water-use habits.
- MMWD adopt ongoing and consistent messaging that responsible water use is a way of life, whether or not drought conditions are present.
- MMWD audit, set targets, and create a plan to reduce its own ecological footprint and set an example to other Marin agencies, businesses, and customers.
- MMWD associate itself with the mountain, not the faucet (emphasizing values, not plumbing).

#### Communication and Customer Support

- \* suggests high priority
- \*1. Institute semi-annual personal letters from Board members to their constituents, explaining MMWD's current water situation, commitment to conservation, conservation efforts, drought protocols, and what customers can do to be more conscious consumers.
- \*2. Institute a monthly email from MMWD Board to Marin's elected officials/staffs on status of water supply, to build awareness and support for MMWD's conservation efforts.
- \*3. Cultivate a network of elected and agency leaders, educators, conservation leaders, and green businesses. Enlist public agencies and Marin businesses as co-sponsors/ endorsers of MMWD water-saving campaign.
- \*4. Redesign and produce a series of billing inserts and billing envelopes tied to the conservation campaign and designed to regain customers' attention and inspire them to take water-saving actions.
- \*5. Give customers an opportunity to opt-in to a "conservation education fund" or add small assessment to bills that would be dedicated to conservation education and restoration activities, including school gardens, community gardens, field trips, volunteer programs, and outreach.

- \*6. Upgrade Web site to be interactive and customer-friendly, including features such as:
  - Calculators for water use and reduction
  - Plant selection database
  - "How-tos" (video/multimedia format?) such as correcting irrigation problems, resetting sprinklers, installing a pressure valve, finding a leak, testing your toilet, etc.
  - Downloadable rebate forms with online submission, online scheduling for a water audit, etc.
  - Enable email from Web site to permit MMWD to answer customers' conservation questions
  - Enable one-click email contact from Web site to report a leak or misuse
  - Include ETo information, standardized run times, when to start watering, etc., on home page
- \*7. Collect email addresses. Develop an email news format with visually interesting and timely information and customer incentives.
- \*8. Upgrade phone system and install a "Conservation Hot Line," with recorded messages to answer common questions, as well as the ability to talk to an "expert" or leave a message with assurance of a reply via phone or email within a specific time period.
- \*9. Develop a graphically unified family of brochures to replace the many now in use, to create a visual identity for MMWD's conservation programs.
- 10. Develop a "conservation pledge" that customers can take. Display pledge at public events, in stores, and on Web site. Offer incentives (such as coupons for water-saving devices, plants) or premiums (such as non-plastic water bottle, watershed recreation map, etc.) for those who sign the pledge.
- 11. Host a workshop of water agencies to share best practices.
- 12. Create a branded "certification" program, modeled on Marin's Green Business Certification program, for organizations that take actions to reduce water use and promote resource efficiency to their colleagues, customers, and employees.
- 13. Display outdoor banners and install attractive display in MMWD lobby. Identify opportunities for public displays.
- 14. Participate in and create tie-ins to community events—town parades, picnics, Farmers' Markets, art festivals, and fairs (such as MMWD's Eco-Sperience environment at the Marin County Fair that focuses on water and healthy living in a healthy environment).

#### Media Outreach

- \*15. Develop a regular graphic feature (for Marin IJ) that illustrates progress toward a water-saving goal. (Current graphic is on existing supply).
- \*16. Proactively place stories on topics tied to current trends:
  - What's on the horizon in ways to save water (new water meter technologies, smart irrigation controllers, GPS to pinpoint progress of invasive plants in the watershed)?
  - Who's leading the way in water conservation?
  - How does Marin compare to other districts, states, and nations in innovations and achievements in water conservation?
  - What are the economic benefits/impacts of water conservation?
- \*17. Develop and run advertising on water conservation themes.
- 18. Develop a graphic or illustration tracing where Marin's water comes from. Highlight non-human dependence on water.
- 19. Develop outreach materials for publication in newsletters of nonprofits, agencies, and large employers, and posters for display in schools, stores, and businesses.

#### <u>Programs</u>

- \*20. Simplify rebate programs. Make rebates seamless for customers— i.e., rebate submission at point of purchase.
- \*21. Offer cash rebates and incentives to customers who voluntarily cut their water use 10% (note: PG&E's Flex Your Power campaign). Incentives might include coupon for drought-resistant plants, non-plastic water bottle, calendar, watershed recreational map, etc.)
- \*22. Institute amnesty and incentive campaign for multi-unit dwellings. (Despite free install and rebates, many residential multi-dwelling units are out of compliance with plumbing efficiency standards.)
- \*23. Prepare water conservation outreach targeted to renters, both residential and commercial.
- \*24. Mount "do-it-yourself" leak-detection awareness campaign.
- \*25. Work with schools and existing school programs (i.e., Next Generation, River of Words, school and community gardens) to make nature/conservation ethic message congruent with MMWD messaging.
- \*26. Provide Spanish-language programs and collateral materials for district communications. Market to Spanish-speaking landscape workers.

- 27. Capitalize on current high level of interest in the environment by offering direct install of residential toilets, with charge to customers' water bill (on-bill financing).
- 28. Develop field trip programs for schools, families, scouts, church, and other groups.
- 29. Explore creating a "junior watershed ranger" program.

#### Partnerships and Sponsorships

\*30. Leverage broad awareness and interest in climate change and "being green" with partnerships and sponsorships with nonprofits, organizations, businesses, and agencies who share MMWD's environmental and public service goals.

#### Sample Campaign: NURSERY/GARDEN SUPPLY CENTER

District customers are purchasing new plants and irrigation equipment for summer; below is a menu of ideas. Implementing such a campaign typically requires months of development and coordination with all parties and partners.

- 1. Signage promoting low water use and native plants, zone designations, and proper irrigation equipment, timing, rain shut-off, and irrigation best practices at nurseries and garden centers.
- 2. Advertising with the same message in the IJ and Pacific Sun. If possible, tie in a special discount or sale at the store for co-branding and cooperation to bring in lots of customers at one time.
- 3. Collection of email address at time of purchase to send irrigation/conservation tips via email: when to start watering, when to add or decrease time on automatic systems, when new information or interactive features are posted at <u>www.marinwater.org</u>.
- 4. Co-branded incentive with retailer, offering a discount coupon on future purchase of water saving plants and products.
- 5. Garden signs for home use displaying an icon with strategic message: "doing our part to save water for next year" (language to be developed) along the lines of the ladybug "pesticide free zone" signs or the "Certified Green Business" logo.
- 6. Eye-catching and thematically unified tips card showing the top 5 10 things a water conscious gardener does.
- 7. Display of appropriate plants and irrigation equipment.
- 8. Discount tickets to Eco-Sperience/County Fair with minimum purchase; perhaps tie in visits to the various exhibits or workshops.
- 9. Offer a simplified rebate application process (perhaps an MMWD staff person at a table to help customers complete) as well as a coupon for a discount at that store for suitable devices or plants at the next visit if rebate form is filled out in store.
- 10. Contact and coordinate with environmental and sustainability groups for assistance in "getting the word out" through their email lists, communications, and meetings. Craft a simple message and identifying graphic to supply for their use. Investigate opportunities for MMWD spokespersons to make short presentations.

This page intentionally left blank.

### CONCLUSIONS

- 1. The District is at a critical turning point in its history and must make significant investments that balance water supply and demand, and fund infrastructure maintenance and operations.
- 2. Investments in water conservation programs have reliably reduced water demand in the District for over 30 years and continue to yield consistent results.
- 3. The conditions now exist for the District to launch an aggressive, sustained conservation campaign and set a new achievement standard for water efficiency in California:
  - Water supplies are limited; new supplies are expensive to develop and impact the environment.
  - The pubic is knowledgeable and motivated.
  - The District has the in-house expertise to develop and operate successful conservation programs.
  - New technologies are available to reliably reduce consumption into the future.
  - Although conservation is a cost-effective method to extend available water supplies and provides environmental benefits, under the current District pricing structure, lowered water sales also reduce District revenue. Solving this financial catch-22 is the most critical challenge limiting conservation today.
- 4. Only by making a long-term commitment to overcome financial and social barriers, and to invest in public education and technology, can the District secure reliable conservation savings.

This page intentionally left blank.

Appendices 2020 Urban Water Management Plan Marin Municipal Water District



# **Appendix J**

# Marin Municipal Water District Title 13 Water Service Conditions and Water Conservation Measures

#### <u>Title 13</u>

#### WATER SERVICE CONDITIONS AND WATER CONSERVATION MEASURES

Chapters:

- 13.01 Eliminated (Ord. 346, 1993)
- 13.02 Water Conservation and Dry Year Water Use Reduction Program
- 13.03 Water Budgets and Related Conservation Measures

#### Chapter 13.02

#### WATER CONSERVATION AND DRY YEAR WATER USE REDUCTION PROGRAM\*

#### Sections:

- 13.02.010 Declaration of purpose.
- 13.02.015 Declaration of Water Shortage Emergency.
- 13.02.020 Water waste prohibitions.
- 13.02.021 Water Conservation: Normal Year Water Conservation
- 13.02.030 Water use reduction in dry periods.
- 13.02.040 Calculation of allowable water use.
- 13.02.050 Variances.
- 13.02.060 Enforcement.
- 13.02.065 Unauthorized water use.
- 13.02.070 Further prohibitions.
- 13.02.080 Penalty for violations.
- 13.02.090 Appeals.
- 13.02.100 Remedies/cumulative.
- 13.02.110 Chapter controlling.

<u>13.02.010</u> Declaration of purpose. The purpose of this chapter is to provide a water conservation plan to minimize the effect of a shortage of water on the district's consumers and to adopt provisions that will significantly reduce the consumption of water during an extended dry weather period (drought), thereby extending the available water for the district's consumers while reducing the hardship on the general public to the greatest extent possible, voluntary conservation efforts having proved insufficient to achieve these ends. The programs developed in this chapter are triggered based on lake storages developed by computer simulations performed utilizing the district's seven reservoirs with approximately eighty thousand acre-feet of total capacity and up to nine thousand acre-feet per year of imported water. (Ord. 387 §1, 1999; Ord. 316 §2 (part), 1991).

<sup>\*</sup> Prior ordinance history: Ords. 279, 286, 290 and 314.

<u>13.02.015</u> Declaration of Water Shortage Emergency. When the district's lake storage on December 1 is projected to be in the vicinity of, or less than, thirty thousand acre-feet, the board may declare by resolution a Water Shortage Emergency as defined in the Water Code and then advise the State Water Resources Control Board of the need to conserve local storage. (Ord. 387  $\S1$ , 1999)

<u>13.02.020 Water waste prohibitions</u>. No customer of the district shall make, cause, use or permit the use of potable water from the district for residential, commercial, industrial, agricultural, governmental or any other purpose in a manner contrary to any provision of this section.

(1) Prohibited Nonessential Uses Applicable to All Consumers. It is unlawful for any person, firm, partnership, association, corporation, or political entity to use water from the district for the following nonessential uses:

(A) The washing of sidewalks, walkways, driveways, parking lots and all other hard-surfaced areas by direct hosing, except as may be permitted by current regulations pertaining to urban water runoff pollution prevention as defined by the Marin County Stormwater Pollution Prevention Program and other controlling agencies;

(B) The escape of water through breaks or leaks within the consumer's plumbing or private distribution system for any substantial period of time within which such break or leak should reasonably have been discovered and corrected. It shall be presumed that a period of forty-eight hours after the consumer discovers such a leak or break, or receives notice from the district of such leak or break, whichever occurs first, is a reasonable time within which to correct such leak or break;

(C) Non-recycling decorative water fountains.

(D) Restrictions on Irrigation. Irrigation shall not be conducted in a manner or to an extent that allows water to run off or overspray the areas being watered. Every consumer is required to have his/her water distribution lines and facilities under control at all times to avoid water waste.

(E) Any excess water runoff flowing onto the public right of way at a rate of 1 gallon per minute or greater not caused by storm water or naturally occurring groundwater, is prohibited.

(F) Using a garden hose without a shut-off nozzle.

(G) Landscape irrigation between the hours of 9:00 a.m. and 7:00 p.m.

(H) Operating outdoor irrigation systems using potable water for the purpose of irrigating ornamental landscape areas or turf areas on more than three days within any week of the calendar year. Customers are strongly encouraged to irrigate ornamental landscapes or turf on fewer days and only as required to maintain plant health and replace evapotranspiration loss as defined by the California Irrigation Management Information System.

(I) The application of potable water to outdoor landscapes during and within 48 hours after measurable rainfall.

(2) Restrictions on Reverse-Osmosis Units. The installation of reverse-osmosis water purifying systems not equipped with an automatic shutoff unit is prohibited.

(3) The following are prohibited for new connections:

(A) Single-pass cooling systems for air conditioning or other cooling system applications unless required for health or safety reasons;

(B) Non-recirculating systems for conveyer carwash applications.

(4) Exemption from daytime water prohibition: Notwithstanding anything contained in this Title 13, testing and repairing irrigation systems for the purpose of eliminating water waste is permitted during the hours of 9:00am and 7:00pm.

(Ord. 428 §2, 2015; Ord. 427 §2, 2014; Ord. 421 §2, 2011; Ord. 387 §1, 1999; Ord. 332 §1, 1992; Ord. 316 §2 (part), 1991).

13.02.021 Water Conservation: Normal Year Water Conservation.

(1) Declaration of Purpose. The purpose of this chapter is to provide a water conservation plan to maximize the water supply during periods of relatively normal rainfall and to minimize the effect of a shortage of water on the district's consumers during an extended dry weather period (drought). The normal year conservation programs in this chapter are based on industry standards promulgated by the American Rainwater Catchment Systems Association (ARCSA), Bay-Friendly Landscape and Gardening Practices (Bay-Friendly), Best Management Practices developed by the California Urban Water Conservation Council (CUWCC), California Department of Water Resources (DWR), California Invasive Plant Council (Cal-IPC), California Irrigation Management Information System (CIMIS), Consortium for Energy Efficiency (CEE), University of California Cooperative Extension (U.C. Extension), USEPA WaterSense Program (WaterSense), Water Use Classification of Landscape Species (WUCOLS), and other recognized conservation industry standards. In every case, the intent of this chapter is to remain a living document, incorporating the most restrictive industry standards in practice at the time in question. In the event that there is a conflict in regulations, the default shall be determined by the District, or as required by law.

Section 2 of Article X of the California Constitution specifies that the right to use water is limited to the amount reasonably required for the beneficial use to be served and the right does not and shall not extend to waste or unreasonable method of use. This policy protects local water supplies through the implementation of a whole systems approach to design, construction, installation and maintenance of the landscape resulting in water conserving climate-appropriate landscapes, improved water quality and the minimization of natural resource inputs.

- (2) <u>Definitions</u>. Definitions used in this chapter are as follows:
  - **A.** Application for Service from an Existing Connection: The application for service from an existing connection, whether it is a new, increased, or modified water service, in a customer's name for a property.
  - **B.** Backflow Prevention Device: means an approved device installed to District standards which will prevent backflow or back-siphonage into the potable water system.
  - **C.** Booster Pumps: used where the normal water system pressure is low and needs to be increased.
  - **D.** Bubblers: Irrigation heads that produce a large volume of output, measured in gallons per minute (gpm) that flood the soil area surrounding the bubbler head.

- **E.** Check Valve: a valve located under a sprinkler head or other location in the irrigation system, to hold water in the system to prevent drainage from sprinkler heads when the sprinkler is off.
- **F.** Common Interest Development: community apartment projects, condominium projects, planned developments, and stock cooperatives per Civil Code Section 1351.
- **G.** Compost: the decayed remains of organic matter that has rotted into a natural fertilizer suitable as a soil amendment to enhance plant growth.
- **H.** Developed landscape area: All outdoor areas under irrigation, swimming pools, and water features, but excluding hardscape areas.
- **I.** Ecological Restoration Project: a project where the site is intentionally altered to establish a defined, indigenous, historic ecosystem.
- **J.** Effective Rainfall: the portion of total rainfall which becomes available for plant growth and that is used by the plants, defined as an average of 25% of total rainfall.
- **K.** Emitter: a drip irrigation device that delivers water slowly from the system to the soil.
- L. Estimated Total Water Use (ETWU): a calculated amount of water needed to irrigate a given landscape, and used as the basis for assigning water budgets at a site.
- **M.** ET Adjustment Factor: a factor of 0.6, that, when applied to reference evapotranspiration as measured by a CIMIS weather station, or equivalent, adjusts for plant factors and irrigation efficiency, two major influences upon the amount of water that needs to be applied to the landscape.
- **N.** Evapotranspiration rate: the quantity of water evaporated from adjacent soil and other surfaces and transpired by plants during a specific specified time.
- **O.** Flow Rate: the rate at which water flows through pipes, and valves and emission devices, measured in (gallons per minute, gallons per hour, or cubic feet per second).
- **P.** Hardscape: Impermeable areas including patios, decks and paths, driveways and sidewalks.
- **Q.** Head-to-Head Coverage: a high-flow irrigation system designed to provide an irrigation spray pattern that delivers water from one sprinkler head to the next.

- **R.** High-efficiency Fixture(s): High efficiency fixtures shall, at a minimum, meet the current requirements of the Water Sense labeling program and those of the California Department of Water Resources and the District.
  - 1. High-efficiency Irrigation Controller: An electronic device that controls the amount of time and frequency of operation for an irrigation system and adjusts automatically to compensate for the seasonal plant water requirements at the site (commonly referred to as weather-based irrigation controllers).
  - 2. High-efficiency Irrigation System: An irrigation system connected to a water service where the overall distribution uniformity (how evenly water is distributed across the irrigated landscape area) is a minimum of 71% and the volume of water used is consistent with seasonal plant requirements as defined by the District.
- **S.** High volume irrigation: An irrigation device or system that delivers water to the landscape in a spray, stream-like, or flooding manner from above-ground irrigation nozzles with output expressed in gallons per minute (include many bubblers and micro-spray devices).
- **T.** High-Flow Sensor: a device for sensing the rate of flow in the irrigation system.
- U. High-water-use plants: Annuals, plants in containers, and plants identified as high-water-use in the current edition of the WUCOLS list published by the U.C. Extension. High-water-using plants are characterized by high transpiration rates, shallow rooting, the need for frequent watering during summer months or with exposure to hot and drying climatic conditions.
- V. Hydrozones: A distinct grouping of plants with similar water needs and climatic requirements. Hydrozone types include, but are not limited to turf, high-water-use plants, low-water-use plants, microclimates (i.e., sun or shade, southern or northern exposures, surrounded by highly reflective surfaces), and partially hardscaped areas with plants, pool areas and water-use features.
- **W.** Infiltration Rate: the rate of water entry into the soil expressed as a depth of water per unit of time (e.g., inches per hour).
- X. Invasive Plant Species: species of plants not historically found in California and/or that spread outside cultivated areas and can damage environmental or economic resources as determined by Cal-IPC (www.cal-ipc.org) and the District.
- Y. Irrigation Design Capacity: The maximum amount of water calculated to flow through an irrigation system, or section of a system, based on pipe size, pipe material, and operating pressure.

- **Z.** Irrigation Efficiency (IE): a calculated measurement of the amount of water beneficially used divided by the amount of water applied. Irrigation efficiency is derived from measurements and estimates of irrigation system characteristics and management practices. The minimum average irrigation efficiency for purposes of this chapter is 0.71.
- **AA.** Irrigation Meter: a separate water meter that measures the amount of water used exclusively for landscape uses, such as lawns, washing exterior surfaces, washing vehicles, filling pools, etc.
- **BB.** Isolation Valves: used to isolate and shut-off water to a portion of the piping system.
- **CC.** Landscape Agent: The consumer's designated representative for interacting with the District on landscape plan reviews.
- **DD.** Landscape Plans: This includes a planting plan, an irrigation plan, and a grading plan drawn at the same scale and that clearly and accurately identify specified plants, irrigation layout, equipment, finish grades and drainage, specifications and construction details, plan sheet numbers, and drawing date of plans.
- **EE.** Landscape Water Budget: The amount of water allowed for landscape water use at a site, adjusted on a seasonal basis, as determined by the District.
- **FF.** Landscaped Area: the entire parcel, less the building footprint, driveways, and non-irrigated portions of parking lots and hardscapes. Water features, areas dedicated to edible plants, such as orchards or vegetable gardens are included in the calculation of the landscaped area. The landscape area does not include footprints of buildings or structures, sidewalks, driveways, parking lots, decks, patios, gravel or stone walks, other pervious or non-pervious hardscapes, and other nonirrigated areas designated for non-development (e.g., open spaces and existing native vegetation).
- **GG.** Lateral Line: Non-pressurized pipe that is located downstream of an irrigation valve.
- **HH.** Low Volume Irrigation: Irrigation devices, commonly called drip or pointsource irrigation, with output measured and typically expressed in gallons per hour (gph), that apply water directly to soil in the plants root zone.
- **II.** Low-Head Drainage: water that flows out of the system after the valve turns off due to elevation changes within the system.

- JJ. Low-water-use plants: Plants identified as low-water-use in the current edition of the Water Use Classification of Species list published by the U.C. Extension. (generally, plants that once established, can survive on two irrigations per month during the summer months).
- **KK.** Main Line: the pressurized pipeline that delivers water from the water source to the valve or outlet.
- **LL.** Maximum Applied Water Allowance (MAWA): for design purposes, the upper limit of annual applied water for the established landscape as determined by the District.
- **MM.** Microclimate: The climate of a specific area in the landscape that has substantially differing sun exposure, temperature, or wind, than adjacent areas or the area as a whole.
- **NN.** Moderate Water Use Plants: ornamental trees, shrubs ground covers, and perennials and other plants recognized as moderate-water-use by WUCOLS.
- **OO.** Mulch: any organic material such as leaves, bark, straw, compost or other inorganic mineral materials such as rocks, gravel, and decomposed granite left loose and applied to the soil surface for the beneficial purposes of reducing evaporation, suppressing weeds, moderating soil temperature and preventing soil erosion.
- **PP.** Operating Pressure: the pressure when water is flowing through the irrigation system.
- **QQ.** Overhead Irrigation: those systems that deliver water through the air (e.g., pop-ups, impulse sprinklers, spray heads, rotors, micro-sprays, etc).
- **RR.** Overspray: Water delivered by an irrigation system outside the targeted landscape area during average operating conditions onto any adjacent hardscapes or other non-landscaped areas during an irrigation cycle, and specifically, for purposes of this Code, limited to maximum of 5% of spray radius area for each nozzle.
- **SS.** Pervious: any surface or material that allows the passage of water through the material and into the underlying soil.
- **TT.** Plant Factor: a factor specified in WUCOLS that, when multiplied by reference evapotranspiration (ETo), estimates the amount of water used by specified plants.
- **UU.** Point of Connection (POC): The location where an irrigation system is connected to water supply.

- **VV.** Precipitation Rate: the rate of application of water measured in inches per hour.
- **WW.** Pressure Regulating Valve: a valve that automatically reduces the pressure in a pipe.
- XX. Project Applicant: the individual or entity submitting a Landscape Documentation Package, to request a permit, plan check or design review from the local agency. A project applicant may be the property owner or his or her designee.
- **YY.** Property: Any structure, including but not limited to single family residential, multi-family residential and floating homes, built and/or intended primarily for sheltering or housing of any person and ancillary structures thereto.
- **ZZ.** Property Owner: A person or entity that owns or has the financial authority or control over the property to comply with the requirements set forth in this chapter.
- **AAA.** Rain Sensor: a system component which automatically shuts off and suspends the irrigation system when it rains.
- **BBB.** Recreational Area: areas dedicated to active play or recreation such as sports fields, school yards, picnic grounds, or other areas with intense foot traffic parks, sports fields and golf courses where turf provides a playing surface.
- **CCC.** Recycled Water: means tertiary treated water which results from the treatment of wastewater, is suitable for direct beneficial use, and conforms to the definition of disinfected tertiary recycled water in accordance with state law.
- **DDD.** Reference Evapotranspiration or ETo: a standard measurement of environmental parameters which affect the water use of plants and are an estimate of the evapotranspiration of a large field of four to seven-inch tall, cool-season grass that is well watered.
- **EEE.** Rehabilitated Landscape: any re-landscaping project that requires a building or grading permit, plan check or design review.
- **FFF.** Residential Customer: The person(s) or entity with an existing water service connection for a residential property.
- **GGG.** Runoff: Irrigation water that is not absorbed by the soil or landscape area to which it is applied and which flows onto other non-targeted areas, including runoff into storm drain systems.
- **HHH.** Soils Laboratory Report: the analysis of a soil sample to determine nutrient content, composition and other characteristics, including contaminants, for horticultural purposes.
- **III.** Special Landscape Area (SLA): an area of the landscape dedicated solely to edible plants, areas irrigated with recycled water, water features using recycled water and areas dedicated to active play such as parks, sports fields, golf courses, and where turf provides a playing surface.
- **JJJ.** Sprinkler Head: a high volume irrigation device that delivers water to the landscape through a spray nozzle.
- **KKK.** Static Water Pressure: the pipeline or municipal water supply pressure when water is not flowing.
- **LLL.** Station: an area served by one valve or by a set of valves that operate simultaneously.
- **MMM.** Submeter: a separate meter that is located on the private side of the water system and is plumbed to measure all water that flows only through the irrigation system. This meter is to be used by the owner to monitor irrigation water use and will not be read or maintained by the District.
- **NNN.** Swing Joint: an irrigation component that provides a flexible, leak-free connection between the emission device and lateral pipeline to allow movement in any direction and to prevent equipment damage.
- **OOO.** Turf: A mat layer of monocotyledonous plants with shallow rooting structures requiring frequent watering during the growing season; i.e., cool or warm season grass consisting, but not limited to Blue, Rye, Fescue, Bent, Bermuda, Kikuyu, St. Augustine, Zoysia, and Buffalo.
- **PPP.** Valve: a device used to control the flow of water in the irrigation system.
- **QQQ.** Valve Manifold: a one-piece manifold for use in a sprinkler valve assembly that includes an intake pipe having a water inlet and a plurality of ports adapted for fluid connection to inlets.
- **RRR.** Water Budget: an allocation of water based on plant water needs, used to determine the billing tiers for customers with dedicated landscape irrigation meters, for example.
- **SSS.** Water Feature: a design element where open water performs an aesthetic or recreational function. Water features include ponds, lakes, waterfalls, fountains, artificial streams, spas and swimming pools (where water is artificially supplied). The surface area of water features is included in the high water use hydrozone of the landscape area.

- **TTT.** Weather Based or Sensor Based Irrigation Control Technology: uses local weather and landscape conditions to tailor irrigation schedules to actual conditions on the site or uses historical weather data.
- **UUU.** WUCOLS: the Water Use Classification of Landscape Species published by the University of California Cooperative Extension, the Department of Water Resources and the Bureau of Reclamation, 2000.

(Ord. 421 §3(part), 2011: Ord. 403 §4, 2004: Ord. 394 §1, 2001: Ord. 385 §1(part), 1999): Ord. 326 §1(part), 1991).

- (3) Requirements for All Services.
  - A. <u>Pressure Regulation</u>. A pressure-regulating valve shall be installed and maintained by the consumer if static service pressure exceeds 80 pounds per square inch (psi), and be set at a maximum operating pressure of 60 psi at the regulator outlet. The pressure-regulating valve shall be located between the meter and the first point of water use, or first point of division in the pipe, and pressure-relief valves and other plumbing safety devices shall be installed as required by local codes. The operating pressure requirement may be waived if the consumer presents evidence satisfactory to the District that high pressure is necessary in the design and that no water will be wasted as a result of higher pressure operation.
  - **B.** <u>Interior Plumbing Fixtures</u>. All plumbing installed, replaced or moved in any new or existing service shall be high-efficiency fixtures and shall meet the following minimum requirements:
    - 1. High-efficiency Clothes Washers: Residential or commercial clothes washers that meet the current highest water efficiency standards as defined by the District. The General Manager shall have authority to grant a variance from the requirements of this section based upon financial hardship.
    - 2. High-efficiency Lavatory Faucet: The maximum flow rate shall not exceed 1.5 gallons per minute (gpm) at a pressure of 60 pounds per square inch (psi) at the inlet, when water is flowing.
    - 3. High-efficiency Shower Head: The manufacturer shall specify a maximum flow rate equal to or less than 2.0 gallons per minute (gpm), at a pressure of 60 pounds per square inch (psi) at the inlet, when water is flowing.
    - 4. High-efficiency Toilet: Any WaterSense listed toilet rated at an effective flush volume of no greater than 1.28 gallons.
    - 5. High-efficiency Urinal: The average water consumption shall not exceed 0.25 gallons per flush (gpf).

- C. <u>Pool Covers</u>. Pool covers are required for all new outdoor swimming pools. (Ord. 421 §3(part), 2011; Ord. 385 §1(part), 1999); Ord. 326 §1(part), 1991).
- (4) <u>Non-Residential Interior Plumbing Fixtures</u>. All plumbing installed, moved or replaced in any new or existing service shall be high-efficiency fixtures and shall meet the following minimum requirements: (Ord. 421 §3(part), 2011).
  - **A.** <u>Faucets</u>. Lavatory faucets, other than public lavatory or metering faucets, shall deliver 1.5 gallons, or less of water per minute.
    - 1. **Metered Faucets** Self-closing or self-closing metering faucets shall be installed on lavatories intended to serve the transient public, such as those in, but not limited to, service stations, train stations, airports, restaurants, and convention halls. Metered faucets shall deliver no more than .25 gallons of water per use. Self-closing faucets shall deliver no more than .5 gallon per minute.
    - 2. **Public Lavatory** (other than metering) faucets shall deliver 0.5 gallons, or less, of water per minute.
    - 3. **Kitchen, Bar and Utility/Service** (other than hand-washing sinks) faucets shall deliver 2.2 gallons, or less, of water per minute.
  - **B.** <u>Private Use, Public Use</u>. *Pursuant to the International Plumbing Code (IPC):* "In the classification of plumbing fixtures, "private" applies to fixtures in residences and apartments, and to fixtures in nonpublic toilet rooms of hotels and motels and similar installations in buildings where the plumbing fixtures are intended for utilization by a family or an individual..."public" applies to fixtures in general toilet rooms of schools, gymnasiums, hotels, airports, bus and railroad stations, public buildings, bars public comfort stations, office buildings, stadiums, stores, restaurants and other installations where a number of fixtures are installed so that their utilization is similarly unrestricted".
  - C. <u>Commercial Equipment Specifications</u>.
    - 1. **Dishwashers.** Dishwashers are machines designed to clean and sanitize plates, glasses, cups, bowls, utensils, and trays by applying sprays of detergent solution (with or without blasting media granules) and a sanitizing final rinse. Dishwashers shall meet the current specifications set by the Consortium for Energy Efficiency's (CEE) "High Efficiency Specifications for Commercial Dishwashers and any and all amendments thereto".
    - 2. **Steamers.** A "steamer" or "steam cooker" is a device with one or more food steaming compartments in which the energy in the steam is transferred to the food by direct contact. Steamers shall meet the current specifications set by the CEE's "High Efficiency Specifications for Commercial Steamers and any and all amendments thereto".

- 3. **Pre-Rinse Spray Valves.** Pre-rinse valves use a spray of water to remove food waste from dishes prior to cleaning in a dishwasher. Pre-rinse spray valves shall (1) deliver 1.3 gallons, or less, of water per minute based on tested performance by the FSTC and (2) meets the cleaning performance standard of 26 seconds per plate or less, based on the ASTM *Standard Test Method for Performance of Pre-Rinse Spray Valves and any and all amendment thereto*.
- 4. **Dipper Wells.** A "dipper well" is a basin into which clean tap water flows constantly to provide a fresh supply of water for soaking utensils. The runoff goes down the drain. Dipper well flow rate shall be .3 gallon, or less, per minute.
- 5. Ice Machines. Ice machine are a factory-made assembly (not necessarily shipped in one package) consisting of a condensing unit and ice-making section operating as an integrated unit, with means for making and harvesting ice. It is an assembly that makes up to 4,000 lbs of ice per day at Standard Ratings Conditions, as defined in Section 5.2.1 of ARI Standard 810-2006, and may also include means for storing or dispensing ice, or both. Ice machines shall (1) be Energy Star qualified and (2) meet the current highest Tier specification set by the CEE's "High Efficiency Specifications for Air-Cooled Ice Machines and any and all amendments thereto".
- 6. **Clothes Washers.** "Commercial clothes washer" means a soft mount frontloading or soft mount top loading clothes washer with clothes container compartment no greater than 3.5 ft<sup>3</sup> for horizontal axis clothes washers, or nor greater than 4.0 ft<sup>3</sup> for vertical axis clothes washers, that is designed for use in (1) applications where the occupants of more than one household will be using it, such as multi-family housing common areas and coin laundries, or (2) other commercial applications. Commercial clothes washers shall meet the minimum Modified Energy Factor (MEF) and maximum Water Factor (WF) corresponding to the highest efficiency machines on the most recent CEE "High Efficiency Specification for Commercial, Family-Sized Clothes Washers and any and all amendments thereto". As of January 1, 2007, the highest efficiency machines have a minimum MEF of 2.20 and a maximum WF of 4.5.
- 7. **Heating, Ventilation and Air Conditioning (HVAC) Equipment.** HVAC Equipment shall eliminate all once-through cooling, replacing with an air-cooled system or a cooling tower. For cooling towers, the following are recommended:
  - (a) flow submeters on make-up and bleed-off lines; submeters should, at a minimum, be capable of totaling the flow.

- (b) conductivity controllers that activate the blowdown valve for dissolved solids control.
- (c) overflow sensors on the overflow pipes.
- (d) baffles or drift eliminators.

All cooling towers shall be monitored and maintained in a manner consistent with applicable regulatory guidelines and manufacturers recommendations.

#### (5) Water Efficient Landscaping

- A. After January 1, 2011, this chapter shall apply to all of the following:
  - 1. New construction and rehabilitated landscapes for public agency projects and private development projects with a landscape area equal to or greater than 1,000 square feet requiring a building or landscape permit, plan check or design review;
  - 2. New construction and rehabilitated landscapes which are developer or contractor-installed in single-family and multi-family projects with a landscape area equal to or greater than 1,000 square feet requiring a building or landscape permit, plan check, or design review;
  - 3. New construction and rehabilitated landscapes which are homeowner-provided in single family and multi-family residential projects with a total project landscape area equal to or greater than 2,500 square feet requiring a building or landscape permit, plan check or design review.
- B. This chapter shall not apply to:
  - 1. Registered local, state or federal historical sites;
  - 2. Ecological restoration projects that do not require a permanent irrigation system;
  - 3. Mined-land reclamation projects that do not require a permanent irrigation system; or
  - 4. Plant collections, as part of botanical gardens and arboretums open to the public.

Note: Authority Cited: Section 65595, Government Code. Reference: Section 65596, Government Code.

- C. <u>Landscape Design Plan</u>. For each landscape project subject to this chapter applicants shall submit a landscape design plan in accordance with the following:
  - 1. Amendments, Mulching and Soil Conditioning: A minimum of 8" of nonmechanically compacted soil shall be available for water absorption and root growth in planted areas.
  - 2. Incorporate compost or natural fertilizer into the soil to a minimum depth of 8" at a minimum rate of 6 cubic yards per 1000 square feet or per specific amendment recommendations from a soils laboratory report.
  - 3. A minimum 3" layer of mulch shall be applied on all exposed soil surfaces of planting areas except in turf areas, creeping or rooting groundcovers or direct seeding applications.
  - 4. Plants.
    - (a) Selected plants shall not cause the Estimated Total Water Use to exceed the Maximum Applied Water Allowance (see calculation in Appendix A).
    - (b) Plants with similar water use needs shall be grouped together in distinct hydrozones and where irrigation is required, the distinct hydrozones shall be irrigated with separate valves.
    - (c) Low and moderate water use plants can be mixed, but the entire hydrozone will be classified as moderate water use for MAWA calculations.
    - (d) High water use plants shall not be mixed with low or moderate water use plants.
    - (e) All non-turf plants shall be selected, spaced, and planted appropriately based upon their adaptability to the climatic, geologic, and topographical conditions of the project site.
    - (f) Turf shall not be allowed in the following conditions: Slopes exceeding 10%, planting areas 8 feet wide or less, street medians, traffic islands, planter strips adjacent to hardscape, or bulbouts of any size.
    - (g) Invasive plants as listed by the Cal-IPC are prohibited. Weedy species, listed as invasive in California at (www.calipc.org/ip/inventory/index.php) shall not be planted. Please check the species you might be thinking of planting against these lists, broken out by plant type. Exemptions may be granted on a case by case basis if District staff determine that the proposed location, species, size, number

of plants, and other cultural methods are not likely to cause harm to the watershed ecosystem.

- (h) Fire Safe Landscape Practices. The requirements in this chapter are intended to support, and be in compliance with, all local and State requirements related to Fire Safe Landscaping practices, including, but not limited to, requirements for Wildlife Urban Interface zones as specified by the local authority.
- (i) Identify any applicable rain harvesting, graywater, or catchment technologies (e.g. rain gardens, cisterns, etc.). Applicants are encouraged to employ alternative irrigation techniques as appropriate, and where permitted by law.
- (j) Identify location and installation details of any applicable stormwater best management practices that encourage on-site retention and infiltration of stormwater. Appropriate stormwater best management practices are encouraged in the landscape design.
- 5. Water Features.
  - (a) Re-circulating water systems shall be used for water features.
  - (b) Recycled water shall be used when available and approved for use onsite.

#### D. Irrigation Design Plan.

- 1. For each landscape project subject to this chapter applicants shall submit an irrigation design plan that is designed, and installed, to meet irrigation efficiency criteria as described in Appendix A (MAWA) and in accordance with the following:
  - (a) Irrigation systems with meters 1 <sup>1</sup>/<sub>2</sub>" or greater require a high-flow sensor that can detect high flow conditions and have the capabilities to shut off the irrigation system.
  - (b) Isolation valves shall be installed at the point of connection and before each valve or valve manifold.
  - (c) High-efficiency controllers, weather-based, or other sensor based selfadjusting irrigation controllers shall be required.
  - (d) Rain sensors shall be installed for each irrigation controller.
  - (e) Pressure regulation and/or booster pumps shall be installed so that all components of the irrigation system operate at the manufacturer's recommended optimal pressure.

- (f) Irrigation systems shall be designed to prevent runoff or overspray onto non-targeted areas, and wherever overhead irrigation is located directly adjacent to hardscape areas, where runoff water flows into the curb and gutter; all spray heads shall be setback a minimum of 24" from hardscape edges.
- (g) Low volume or bubbler irrigation is required where plant height at maturity will affect the uniformity of a high volume overhead spray system.
- (h) Minimum 24" setback of overhead spray irrigation is required where turf is directly adjacent to a continuous hardscape area where runoff water flows into the curb and gutter.
- (i) Slopes greater than 15% shall not be irrigated with an irrigation system with a precipitation rate exceeding 0.75 inches per hour (or lower if appropriate for site conditions as determined the District). This restriction may be modified if the irrigation designer specifies an alternative design or technology, as part of the Landscape Documentation Package, and clearly demonstrates no runoff or erosion will occur. Prevention of runoff and erosion must be confirmed during the irrigation audit.
- (j) A single valve shall not irrigate hydrozones that mix high water use plants with moderate or low water use plants.
- (k) Trees shall be placed on separate irrigation valves except when planted in turf areas.
- (1) Sprinkler heads, rotors and other emission devices on a valve shall have matched precipitation rates.
- (m) For all irrigation systems, coverage to sustain plant material in a healthy condition and provide irrigation efficiency shall be required. Head-tohead coverage is required for high volume spray systems unless otherwise directed by the manufacturer's specifications.
- (n) Swing joints or other pipe protection components are required on aboveground irrigation piping.
- (o) Check valves shall be installed to prevent low-head drainage.
- E. Separate District landscape water service meters shall be required for all new landscapes, other than single-family and two-unit residential landscapes, for which the irrigated area is equal to or greater than 1,000 square feet.

- 1. A private submeter shall be required for all rehabilitated landscapes, other than single-family and two-unit residential landscapes, for which the irrigated landscape area is equal to or greater than 1,000 square feet.
- 2. A private submeter shall be required for all points of connection on singlefamily and two-unit residential sites for which the irrigated landscape area is equal to or greater than 2,500 square feet.
- F. Documentation for Compliance. The following documentation is to be presented to the District at each of the three steps of review defined below. This documentation shall be required for compliance with this chapter. 13.02.021
  - 1. STEP 1: DESIGN REVIEW. For those landscape projects that require Design Review, applicants shall submit the following documentation to the District:
    - (a) Completed Appendix A, Maximum Applied Water Allowance (MAWA)
    - (b) A landscape planting design plan that accurately and clearly identifies and depicts new and existing trees, shrubs, groundcovers, turf, and any other planting areas; plants by botanical name, common name, and plant factor; plant sizes and quantities; property lines, new and existing building footprints, streets, driveways, sidewalks and other hardscape features; pools, fountains, water features.
    - (c) An irrigation design plan drawn at the same scale as the planting plan that:
      - (i) Accurately and clearly identifies and depicts irrigation system point(s) of connection;
      - (ii) Accurately and clearly identifies and depicts irrigation system components, e.g. controller, pipe, remote-control valves, sprinklers, rain shut-off device, check valves, pressure regulating devices, backflow prevention devices, and other required devices
      - (iii) Includes the Hydrozone Table (See Appendix B).
      - (iv) For the efficient use of water, grading of a project site shall be designed to minimize soil erosion, runoff, and water waste. A grading design plan drawn at the same scale as the planting design plan shall be submitted as part of the Landscape Documentation Package. Items 1(a-e) below are required for all projects.

- (1) The project applicant shall submit a landscape grading plan that indicates finished configurations and elevations of the landscape area including:
  - (a) height of graded slopes:
  - (b) drainage patterns:
  - (c) pad elevations;
  - (d) finish grade: and
  - (e) stormwater retention improvements, if applicable.
- (2) It is highly recommended that, when site conditions allow, project applicants consider grading so that all irrigation and normal rainfall remains within property lines and does not drain on to non-permeable hardscapes.

The grading design plan shall contain the following statement: "I have complied with the criteria of this chapter and applied them accordingly for the efficient use of water in the grading design plan" and shall bear the signature of a licensed professional or contractor as authorized by law.

- (d) For homeowner-provided projects, a completed Homeowner's Irrigation Design Statement, Appendix B, which describes irrigation methods and design actions that will be employed to meet the irrigation specifications of this chapter, may be submitted in lieu of the irrigation design plan.
- 2. STEP 2: COMPLETION OF INSTALLATION. Upon installation and completion of the landscape, applicant shall submit Appendix D, the Certificate of Completion.

The certificate shall be accompanied by an irrigation audit that contains the following:

- (a) Operating pressure of the irrigation system.
- (b) Distribution uniformity of overhead irrigation.
- (c) Precipitation rate of overhead irrigation.
- (d) Report of any overspray or broken irrigation equipment.
- (e) Irrigation schedule including:
- (f) Plant establishment irrigation schedule.

- Regular irrigation schedule by month including: plant type, root depth, soil type, slope factor, shade factor, irrigation interval (days per week), irrigation runtimes, number of start times per irrigation day, gallons per minute for each valve, precipitation rate, distribution uniformity and monthly estimated water use calculations.
- (ii) An irrigation maintenance schedule timeline shall be attached to the certificate of completion that includes: Routine inspections, adjustment and repairs to the irrigation system, aerating and dethatching turf areas, replenishing mulch, fertilizing, pruning and weeding.
- 3. STEP 3: FINAL INSPECTION. A final inspection shall be performed by District staff to verify compliance with this chapter. Once the completion form is received, the District will conduct an inspection to check for proper installation and operation of all landscape and irrigation elements per the approved plan; however, the District reserves the right to perform site inspections at any time before, during, or after irrigation system and landscape installation and to require corrective measures if requirements of this chapter are not satisfied. If corrective measures are necessary, the District will set the water budget to zero until corrective measures are completed.

Advanced notice is required for all inspections. Inspections can be requested for either morning or afternoon during regular business hours. Final approval shall not be completed until the landscape inspection is approved. An extension of the approval process, to complete landscape and irrigation installation, shall be requested and shall be approved District staff.

(Ord. 421 §3(part), 2011).

(6) Drinking Water Served Upon Request Only.

By January 1, 2011, eating or drinking establishments, including but not limited to a restaurant, hotel, café, cafeteria, bar, or other public place where food or drinks are sold, served, or offered for sale, are prohibited from providing drinking water to any person unless expressly requested.

(7) <u>Commercial Lodging Establishments Must Provide Guests Option to Decline Daily Linen</u> <u>Services</u>.

By January 1, 2011, hotels, motels and other commercial lodging establishments shall provide customers the option of not having towels and linen laundered daily. Commercial lodging establishments shall prominently display notice of this option in each bathroom using clear and easily understood language.

(8) <u>Grey Water Systems</u>. This section is reserved for future provisions regarding grey water systems.

- (9) <u>Rain Water Harvesting Systems</u>. This section is reserved for future provisions regarding rain water harvesting systems.
- (10) <u>Other Provisions</u>. The General Manager will consider and may allow the substitution of design alternatives and innovation which may equally reduce water consumption for any of these requirements. The General Manager may accept documentation methods, water allowance determination, and landscape and irrigation design requirements of the State of California Model Water Efficient Landscape Ordinance instead of Chapters 14-30.040 and 14-30.050 of these requirements where it can be demonstrated that the State procedure will more effectively address the design requirements of the project.
- (11) <u>Provisions For Appeal</u>. The applicant or any affected person may appeal the final decision of staff regarding plan check or final inspection to the General Manager, The decision of the General Manager shall be final. An appeal regarding plan check shall be submitted prior to the installation of the landscape or it will be deemed to have been waived.
- (12) Forms. The following forms shall be submitted as described in this chapter: Appendix A, Maximum Applied Water Allowance; Appendix B, Homeowner's Irrigation Design Statement; Appendix C, Hydrozone Table; Appendix D, Certificate of Completion. (Ord. 421 §3(part), 2011; Ord. 414 §2, 2010)

#### Appendix A

#### Maximum Applied Water Allowance

The following calculations will help you determine your site specific water budget and establish a planting mix that will allow you to meet your water budget. Your Estimated Total Water Use must be less than your Maximum Applied Water Allowance.

1.) N	Iaximum Applied Water Allowand	ce (MAWA)			
Ν	MAWA = (ETo) (0.62)[(0.6x LA)	+ (0.4 x SLA)]			
v	Whata.				
v F	vnere. To = Annual Net Reference Evan	otranspiration (inches)			
0	.6 = ET Adjustment Factor	ou dispiration (inches)			
L	A = Landscaped Area (square fee	et)			
0	.62 = Conversion factor (to gallon	is per square foot)			
S	LA = Portion of the landscape are	ea identified as Special Landscape A factor for Special Landscape Area	Area (square feet)		
0	.4 – ule auditional E1 aujustitent.	lactor for special Landscape Area	(1.0 - 0.0 - 0.4)		
	A.) Net Evapotranspiration (	Calculation			
	(Annual ETo)				
	(Annual Painfall)	x .25	= (Effective Pair	w6-/1)	
	(miniai nangan)		(Lysenve ran	gaij	
	Net Evapotranspiration (	Calculation = Annual ETo	- Effective Rainfall	=	
B.)	Adjusted Landscape Area Cal	culation			
		x 0.6		=	
	(Landscaped Area)	Adjustment Factor			
	(Created I made a mod (mar)	x 0.4		=	
	(Special Lanascapea Area)	Adjustment Factor			
			Sum of Adjusted Land	scape Area =	
	V		,	-	
MAWA =	=	x 0.62 2	(	=	
					]
2.) Estimated	1 Total Water Use (ETWU)				
A.)	Net Evapotranspiration Calculation	on			
	Not Erron stranspiration Coloniatio	an - Annual ETa I	Anativa Dainfall	_	
	Net Evapor anspiration Calculate	JII – Alinuar E10 - 1		-	
B.)	Adjusted Landscape Area Calcul	lation	1		
			1		
	(Low water use plant saft)	x 0.3		=	
	(200 0 000 000 000 000)		]		
		x 0.6	]	=	
	(Moderate water use plant sqft)				
		x 10	]	=	
	(High water use plant sqft)	X 1.0			
			1		
			Sum of Adjusted Landscape	Area =	7
	V	0.62			
EIWU =	-	x 0.62 x	*	=	
		Irrigation Efficiency Factor			
		Percent of total lands	cape Irrigated with Drip		
		0-25%	0.71		
		51-75%	0.75		
		51-7570	0.00		

### Appendix B Homeowner's Irrigation Design Statement

The intent of this statement is to provide the homeowner with an alternative method for conceptualizing the irrigation design. A signed, written statement shall be submitted to the District as part of the design review process, and shall include the following elements:

- □ Accurately and clearly describes the types and locations of all irrigation system point(s) of connection;
- □ Accurately and clearly describes the types and locations of all irrigation system components by valve zone, including high-efficiency irrigation controller, pipe, valves, high and low volume irrigation devices, rain shut-off device, check valves, pressure regulating devices, backflow prevention devices, and all other irrigation devices required by the District.
- □ A completed hydrozone table, Appendix C.
- □ A description of plant species irrigated in each valve zone by scientific name, water use of each plant species as High, Moderate, or Low water use according to WUCOLS (Water Use Classification of Landscape Species), and plant height at maturity for each plant. Plant height is not necessary where drip or bubbler will be used.
- □ A statement signed by the homeowner that includes the following certifying language: "The irrigation system will be installed as described in this statement, and in compliance with the requirements of the District".

## Appendix C Hydrozone Table

This worksheet is filled out by the project applicant and it is a required element of the Landscape Documentation Package. Please complete the hydrozone table(s) for each hydrozone. Use as many tables as necessary to provide the square footage of landscape area per hydrozone.

Hydrozone*	Zone or Valve	Irrigation Method**	Area (Sq. Ft.)	% of Landscape Area
	Total			100%

Summary Hydrozone Table				
Hydrozone*	Area (Sq. Ft.)	% of Landscape Area		
High Water Use				
Moderate Water Use				
Low Water Use				
	Total =	100%		

#### \*Hydrozone

HW = High Water Use Plants MW=Moderate Water Use Plants LW=Low Water Use Plants

# \*\*Irrigation Method

MS=Micro-spray S=Spray R=Rotor B=Bubbler D=Drip O=Other

#### Appendix D Certificate of Completion

This certificate is filled out by the project applicant, landscape architect and/or landscape contractor upon completion of the landscape project.

#### Part 1. Project Information Sheet

Date		
Project Name	Project Address	
Name of Project Applicant	Telephone No.	
	Fax No.	
Title	Email Address	
Company	Street Address	
City	State	Zip Code

#### Property Owner or his/her designee:

Name	Telephone No.	
	Fax No.	
Title	Email Address	
Company	Street Address	
City	State	Zip Code

"I/we certify that I/we have received copies of all the documents within the Landscape Documentation Package and that it is our responsibility to see that the project is maintained in accordance with the Landscape and Irrigation Maintenance Schedule."

Property Owner Signature

Date

#### Part 2. Landscape Architect and/or Landscape Contractor/Installer

Landscape Architect Name	Telephone No.	
	Fax No.	
Title	Email Address	
License No. or Certification No.	Telephone No.	
Company	Street Address	
City	State	Zip Code
Landscape Contractor/Installer Name	Telephone No.	
	Fax No.	
Title	Email Address	
License No. or Certification No.	Telephone No.	
Company	Street Address	
City	State	Zip Code

"I/we certify that the work has been completed in accordance with the ordinance and that the landscape planting and irrigation installation conform to the criteria and specifications of the approved Landscape Documentation Package. Additionally, the landscape audit and irrigation maintenance schedule have been completed as required and are attached to this certificate showing that the system meets the efficiency requirements used in the Estimated Total Water Use calculation"

Landscape Architect Signature

Date Date

Landscape Contractor Signature

Marin Municipal Water District 02/11 <u>13.02.030 Water use reduction in dry periods</u>. No customer of the district shall make, cause, use or permit the use of potable water from the district for residential, commercial, industrial, agricultural, governmental or any other purpose in a manner contrary to any provision of this section.

(1) Program to Reach a Ten Percent Water Use Reduction Goal (alert stage). When lake storage on April 1st in any year is below fifty thousand acre-feet and system modeling based on historic hydrologic data and use patterns indicates that a minimum reduction in use is required to assure a sufficient carry-over storage, the board may activate by resolution a voluntary use reduction program and should review its financial reserves to determine if a reduction in tier breaks is fiscally warranted to achieve ten percent savings in district's overall water use as follows:

(A) Consumer Percentage Curtailment. Every consumer shall eliminate water wastage in an effort to aid the district in achieving a ten percent reduction of the amount of potable water used by all consumers during the last year in which no restrictions in water use were required.

(B) Curtailment of Program. When lake storage on April 1st of any year is above fifty thousand acre-feet or when lake storage on or after January 1st is sufficiently above fifty thousand acre feet and projected demands will not reduce lake storage below fifty thousand acre-feet by the following April 1st, the requirements of this subsection shall be deactivated by resolution of the board.

(2) Twenty-five Percent Water Use Reduction Program. When lake storage on April 1st is below forty thousand acre-feet and system modeling based on historical hydrologic data and use patterns indicates a reduction in use is required to assure sufficient carry-over storage, the board may activate by resolution a mandatory use reduction program to achieve twenty-five percent savings in district's overall water use as follows:

(A) Consumer Use Reduction. Residential consumers, Billing Codes 1 through 5, shall use no more than the water use allocation assigned by the district based on an allotment per resident to aggregate a thirty-two percent reduction by all residential consumers. A resident is considered to be an individual who resides, or is expected to reside in the structure for a minimum of six months of the year.

Each nonresidential consumer, Billing Code 6 shall use no more than seventy-five percent, Billing Code 7 shall use no more than eighty percent, and Billing Code 8 shall use no more than fifty percent, of the annual water budget calculated by the district for said consumer pursuant to Chapter 13.03 of the district's code.

Every consumer shall eliminate water wastage and non-essential use of potable water from the district in an effort to aid the district in achieving a twenty-five percent reduction in the amount of water used by all consumers in the last year in which no restrictions in water use were required.

(B) Prohibited Nonessential Uses Applicable to All Consumers. It is unlawful for any person, firm, partnership, association, corporation or political entity to use potable water from the district contrary to the provisions of this subsection, or, if more restrictive, subsections (1), (2) and (3) of this section, or to use potable water for: refilling or as make-up water for decorative fountains or pools; irrigation between the hours of 11 AM and 7 PM; irrigation of new turf areas; washing of cars, boats, airplanes with hose without a shut-off nozzle; or serving water to restaurant patron except on request.

(C) Curtailment of Program. When lake storage on April 1st is above forty thousand acre-feet or when lake storage is sufficiently above forty thousand acre-feet and projected demands will not reduce lake storage below forty thousand acre-feet by the following April 1st,

the requirements of this subsection shall be deactivated by resolution of the board. (Ord. 387 §1, 1999; Ord. 376 §8, 1997; Ord. 325 §1, 1991; Ord. 323 §1, 1991; Ord. 316 §2 (part), 1991).

13.02.040 Calculation of allowable water use. When the requirements of Section 13.02.030 (2) are in effect, consumers in Billing Codes 6, 7 and 8 will reduce their use by the appropriate percentage of their water budget. (Ord. 387 §1, 1999; Ord. 376 §9, 1997; Ord. 316 §2 (part), 1991).

<u>13.02.050 Variances</u>. The general manager of the district may grant variances for use of water otherwise prohibited by this chapter if the general manager finds and determines that:

(1) The applicant, if requesting a variance for a nonresidential service (Billing Codes 6, 7 and 8), has agreed to abide by an annual water budget calculated pursuant to Chapter 13.03 of this title; and

(2) The applicant has agreed to adjust his water usage by complying with Section 11.04.080 of this code where determined to be applicable by district staff; and

(3) Failure to do so would cause an unnecessary and undue hardship on applicant or the public, including but not limited to, adverse economic impacts, such as loss of production or jobs; or

(4) Failure to do so would cause an emergency condition affecting the health, sanitation, fire protection or safety of the applicant or the public. (Ord. 387 §1, 1999; Ord. 316 §2 (part), 1991).

#### 13.02.060 Enforcement.

(1) Any customer violating the regulations and restrictions on water use set forth in this chapter shall receive a written warning for the first such violation.

(2) Upon a second violation, the following enforcement process shall be implemented by the District:

(A) The customer shall receive a written warning describing the nature of the second violation and the required corrective action(s).

(B) Upon receipt of the written warning for the second violation, the customer shall have a 15 calendar day period in which to implement the corrective action(s) and shall notify the District when the corrective actions have been completed.

(C) District staff will conduct a site visit to verify that the corrective action(s) are complete and satisfy the requirements in this chapter.

(3) If the customer fails to complete the required corrective action(s), and notify the District within the 15 calendar day period, the District shall issue a Notice of Violation by certified mail or personal delivery at least 10 calendar days before imposing a \$250 fine on the customer's water bill. The notice shall inform the customer that he/she may appeal the imposition of the fine to the District's General Manager by filing a written appeal within 7 calendar days of the date of the letter. Any Notice of Violation not timely appealed will be final. Upon receipt of a timely appeal, a hearing on the appeal will be scheduled and the District will mail notice of this date to the customer at least 10 calendar days before the hearing. The General Manager's decision is final. Pending receipt of a written appeal or pending hearing pursuant to an appeal, the District may take appropriate steps to prevent unauthorized use of water as appropriate to prevent waste. This notice and hearing procedure shall not apply to those water waste situations charged as misdemeanors.

(4) Failure by the customer to pay the fine and correct the violation, may cause the District to install a flow restrictor to be installed in the service. If a flow restrictor is placed, the cost of installation and removal specified in Section 11.32.020 of this code shall be paid by the violator.

(5) Any willful violation occurring subsequent to the issuance of the second written warning may constitute a misdemeanor and may be referred to the Marin County district attorney's office for prosecution pursuant to Section 13.02.080.

(6) The district may also disconnect the water service in accordance with Section 11.28.030 of this code. If water service is disconnected, it shall be restored only upon payment of the turn-on charge fixed by the board of directors under the provisions of Section 11.08.130(2) of this code. (Ord. 421 §4, 2011; Ord. 387 §1, 1999; Ord. 316 §2 (part), 1991).

<u>13.02.065 Unauthorized water use</u>. Use of water without having made application to the district for water service or use of any district water not metered pursuant to such application is prohibited pursuant to this chapter, and in addition to the penalties contained in Section 13.02.080, violators will be subject to the charges for use of such water set forth in Section 6.01.080 of this code. (Ord. 316 §2 (part), 1991).

<u>13.02.070 Further Prohibitions</u>. It is unlawful for any person, firm, partnership, association, corporation or political entity to remove, replace, alter or damage any water meter or components thereof, including but not limited to the meter face, its dials or other water usage indicators and any flow restricting device installed pursuant to Section 13.02.060. (Ord. 316 §2 (part), 1991).

<u>13.02.080 Penalty for violations</u>. Except as provided in Section 13.02.060, for the first and second violations any person, firm, partnership, association, corporation or political entity violating or causing or permitting the violation of any of the provisions of this chapter or providing false information to the district in response to district's requests for information needed by the district to calculate consumer water allotments shall be guilty of a misdemeanor punishable by imprisonment in the county jail for not more than thirty days or by a fine not exceeding one thousand dollars or both. Each separate day or portion thereof in which any violation occurs or continues without a good faith effort by the responsible party to correct the violation shall constitute a separate offense and, upon conviction thereof, shall be separately punishable. (Ord. 316 §2 (part), 1991).

<u>13.02.090 Appeals</u>. Variances from the requirements of this chapter may be granted by the board of directors only after denial of a variance request by the general manager. Appeals of variance request denials shall be made in writing to the secretary of the board at least two weeks prior to the meeting at which they will be heard. Upon granting any appeal, the board of directors may impose any conditions it determines to be just and proper. Variances granted by the board of directors shall be prepared in writing, the original to be kept on file with the district and a copy to be furnished to the applicant. The board of directors may require it to be recorded at applicant's expense. (Ord. 316 §2 (part), 1991).

<u>13.02.100 Remedies/cumulative</u>. The remedies available to the district to enforce this chapter are in addition to any other remedies available under the district's code, or any state statutes or regulations, and do not replace or supplant any other remedy, but are cumulative. (Ord. 316 §2 (part), 1991).

<u>13.02.110 Chapter controlling</u>. The provisions of this chapter shall prevail and control in the event of any inconsistency between this chapter and any other rule, regulation, ordinance or code of this district. (Ord. 316 §2 (part), 1991).

#### Chapter 13.03

### WATER BUDGETS AND RELATED CONSERVATION MEASURES\*

Sections:

13.03.010	Declaration of purpose.
13.03.020	Basis for determination of water budget.
13.03.030	Water budgets, when required.
13.03.031	Increasing a Service's Water Budget.
13.03.040	Efficient plumbing fixtures.
13.03.050	Variances from Section 13.13.040.
13.03.060	Appeals.
13.03.070	Penalty for violations.
13.03.080	Recordation of notice.
13.03.090	Cost of enforcement.
13.03.100	Chapter controlling.
13.03.110	Remedies/cumulative.
13 03 120	Flow restrictors.

<u>13.03.010</u> Declaration of purpose. The purpose of this chapter is to specify the terms and conditions under which water budgets will be required and to specify when consumers will be required to retrofit water using fixtures with low-flow or ultra-low-flow fixtures, all for the purpose of permanently reducing the per capita consumption of water by the district's customers, thereby reducing the hardship on the district's consumers resulting from over-subscription of the district's water supplies which has increased the susceptibility of the district's supply to shortfalls in dry years. (Ord. 316 §2 (part), 1991).

<u>13.03.020</u> Basis for determination of water budget. The initial annual water budget for each existing service which is not a single-family residential or multi-unit residential structure is determined by the district assigning the amount of the water entitlement based on the purchased water entitlement, calculated consumption or designated annual consumption as defined in Section 11.08.180 of this code. The water budget for each service may be adjusted below the water entitlement as set forth in Section 11.08.035. All customers except residential customers exceeding their annual water entitlements are subject to terms and conditions of Section 11.08.030 of this code regarding changing character and/or increasing use of water. (Ord. 376 §9, 1997; Ord. 340 §1 (part), 1992; Ord. 316 §2 (part), 1991).

<u>13.03.030</u> Water budgets when required. Services must conform to the annual water budget calculated by the district for each property receiving water service as follows:

(1) New services: immediately upon connection.

<sup>\*</sup> Prior ordinance history: Ords. 295, 301, 309, 310 and 314.

(2) Existing services: as a condition of receipt of a variance or as part of the calculation of allowable use pursuant to Section 13.02.040 of this title, or upon receipt of notification from district that an annual water budget has been prepared by district staff. (Ord. 316 §2 (part), 1991).

<u>13.03.031</u> Increasing a Service's Water Budget. Requests to increase a service's Water Budget, as a result of an entitlement purchase or transfer, or where the service's water budget is less than the service's water entitlement shall meet the following requirements:

(1) Mixed-use Non-Residential Services:

(A) Interior Fixtures. Interior Fixtures shall meet conditions as stated in District Code Section 13.02.02.

(B) Irrigated Landscape Areas. A minimum 3" layer of mulch shall be applied on all exposed soil surfaces of planting areas except in turf areas or creeping or rooting groundcovers. Rain sensors shall be installed for each irrigation controller. Irrigation system shall be adjusted to minimize runoff or overspray onto non-targeted areas.

(2) Dedicated Irrigation Services. The following items shall be completed by the applicant according to the requirements in District Code Section 13.02.02:

- A landscape water budget calculating both MAWA & ETWU requirements.
- A landscape hydrozone table.
- A minimum 3" layer of mulch shall be applied on all exposed soil surfaces of planting areas except in turf areas or creeping or rooting groundcovers.
- Rain sensors shall be installed for each irrigation controller.
- Irrigation system shall be adjusted to minimize runoff or overspray onto nontargeted areas.
- Backflow prevention equipment shall be installed and tested as required by the District.

Upon determining the service meets the requirements for both indoor and outdoor water use, and based on the current water efficiency standards established by the District, the service may be eligible to increase the annual water budget as calculated by staff, but in no case shall the budget exceed the services entitlement. Requests to increase a service's water budget may be made once per calendar year. (Ord. 421 §5, 2011).

<u>13.03.040</u> Efficient plumbing fixtures. Any plumbing fixture in any existing service which is replaced, added or moved shall conform to the criteria contained in District Code Section 13.02.021. (Ord. 414 §3, 2010; Ord. 316 §2 (part), 1991).

<u>13.03.050 Variances from Section 13.13.040</u>. The general manager may, in his discretion, exempt services from the requirements of Section 13.03.040 of this chapter, or impose reasonable conditions in lieu of compliance therewith, if he determines that any of the following conditions apply:

(1) Hardship. The general manager may grant an extension for hardship where the requirements of Section 13.03.040 cause an unnecessary and undue substantial hardship upon the owner, or purchaser of the facility or the public. Substantial hardship may include, but is not limited to:

(A) Plumbing in an existing facility which does not match the connections with efficient plumbing fixtures and would, therefore, require partial replumbing of the structures;

(B) Unavailability of efficient plumbing fixtures to match a well-defined historic architectural style in a locally, state or federally recognized building of historic significance, fitted with authentic plumbing fixtures;

(C) Any project exempted pursuant to subsection (1) of this section shall be required to have installed toilets using a maximum of 3.5 gallons per flush and three gallons per minute showerheads if not already so provided.

(2) Emergency. The general manager may grant an exemption for emergency purposes when the requirements of Section 13.03.040 would create a condition affecting the health, sanitation, fire protection or safety of the facility owner or the public.( Ord. 316 §2 (part), 1991)

<u>13.03.060</u> Appeals. Except for variances from Section 13.03.040, variances from the requirements of this chapter may be granted by the board of directors only after denial of a variance request by the general manager. Appeals of variance request denials shall be made in writing to the secretary of the board at least two weeks prior to the meeting at which they will be heard. Upon granting any appeal, the board of directors may impose any conditions it determines to be just and proper. Variances granted by the board of directors shall be prepared in writing, the original to be kept on file with the district and a copy to be furnished to the applicant. The board of directors may require it to be recorded at applicant's expense. (Ord. 316 §2 (part), 1991).

<u>13.03.070</u> Penalty for violations. Any person, firm, partnership, association, corporation or political entity violating or causing or permitting the violation of any of the provisions of this chapter shall be guilty of a misdemeanor punishable by imprisonment in the county jail for not more than thirty days or by a fine not exceeding one thousand dollars or both. Each separate day or portion thereof in which any violation occurs or continues without a good faith effort by the responsible party to correct the violation shall constitute a separate offense and, upon conviction thereof, shall be separately punishable. (Ord. 316 §2 (part), 1991).

<u>13.03.080</u> Recordation of notice. Whenever the general manager determines that low water-use plumbing fixtures required by Section 13.03.040 have not been installed or have been removed since initial installation, the general manager may record a notice of violation with the office of the county recorder. The owners of the property as revealed by the assessment roll on which the violation is situated and any other person responsible for the violation shall be notified of the recordation, if their address is known. The owner of record shall have ninety days to take corrective action. Failure to take corrective action within ninety days shall constitute a violation of this code. The general manager shall cause a notice of correction to be recorded at such time as the property owner has established full compliance with the provisions of this chapter. (Ord. 316 §2 (part), 1991).

<u>13.03.090</u> Cost of enforcement. Any person, firm or corporation or upon whose property a notice of violation has been recorded shall, if the condition creating the nuisance or constituting the violation is not corrected within thirty days, be liable for the cost of abatement and cost of correction which shall include, but not be limited to, cost of investigation, court costs, attorney fees and costs of monitoring compliance. (Ord. 316 §2 (part), 1991).

<u>13.03.100</u> Chapter controlling. The provisions of this chapter shall prevail and control in the event of any inconsistency between this chapter and any other rule, regulation, ordinance or code of this district. (Ord. 316 §2 (part), 1991).

<u>13.03.110</u> Remedies/cumulative. The remedies available to the district to enforce this chapter are in addition to any other remedies available under the district's rules and regulations, or any other state statutes or regulations, and do not replace or supplant any other remedy, but are cumulative. (Ord. 316 §2 (part), 1991).

<u>13.03.120 Flow restrictors</u>. In addition to the penalties provided for in Section 13.03.070 of this chapter, district may install a flow restrictor in the service of a customer who violates the provisions of this chapter if, after receiving notice from district that he is in violation, he does not rectify the violation within a reasonable time specified by the district not to exceed sixty days. (Ord. 316 §2 (part), 1991).

#### MARIN MUNICIPAL WATER DISTRICT

#### **ORDINANCE NO. 429**

### AN ORDINANCE AMENDING TITLE 13 OF THE MARIN MUNICIPAL WATER DISTRICT CODE ADDING ANOTHER ELEMENT OF THE DISTRICT'S WATER CONSERVATION PROGRAM PURSUANT TO WATER CODE SECTION 375 BY ADDING SECTION 13.02021(2) O-1 "DEFINITIONS" TO INCLUDE "GRAY WATER" AND AMENDING CHAPTER 13.02.021(8) ENTITLED "GRAY WATER SYSTEMS" OF THE DISTRICT CODE

# BE IT ORDAINED BY THE BOARD OF DIRECTORS OF THE MARIN MUNICIPAL WATER DISTRICT AS FOLLOWS:

**SECTION 1. Purpose:** The purpose of this ordinance is to adopt another element of the District's water conservation program pursuant to Water Code Section 375. The purpose of this ordinance is to require applicants for new water service and applicants requesting an enlarged water service for substantial residential or commercial remodels to install a gray water recycling system to reuse the maximum practicable amount of gray water on site. The Board of Directors views this conservation program as a fundamental and necessary step in its on-going efforts to reduce overall water use District-wide.

#### **SECTION 2. Ordinance Amendments:**

#### A. <u>Section 13.02.021 (2) O-1 is added to Section 13.02.021 (2) entitled "Definitions" to read</u> as follows:

O-1.Gray Water: Untreated wastewater that has not been contaminated by any toilet discharge, has not been affected by infectious, contaminated, or unhealthy bodily wastes, and does not present a threat from contamination by unhealthful processing, manufacturing, or operating wastes. "Gray water" includes wastewater from bathtubs, showers, bathroom washbasins, clothes washing machines, and laundry tubs, but does not include wastewater from kitchen sinks or dishwashers.

#### B. Section 13.02.021(8) is amended to read as follows:

(8) Gray Water Systems. All applicants requesting a water service for a new residential or commercial structure which requires the installation of a new service, and all applicants requesting an enlarged water service for a "substantial remodel" to an existing residential or commercial structure as that term is defined under Marin County Code Section 19.04.063 or any successor ordinance shall install a gray water recycling system to reuse the maximum practicable amount of gray water on site. Installation and operation of all gray water systems shall comply with all rules and regulations of the District, the local jurisdiction having zoning authority over the parcel and the California Plumbing and CALGreen Code.

**SECTION 3. Findings of Necessity:** The Board of Directors, after considering all of the information and testimony presented at its November 3, 2015 meeting and all of the comments and protests lodged in relation thereto and in the other portions of the record of this rate-making, finds as follows:

- A. Water is a finite and precious resource.
- B. Currently, California is in its fourth year of drought.
- C. On January 17, 2014, Governor Jerry Brown issued a drought state of emergency declaration in response to record-low water levels in California's rivers and reservoirs as well as an abnormally low snowpack. On April 1, 2015, Governor Brown issued an Executive Order calling for statewide mandatory water reductions of up to 25%.
- D. On May 5, 2015, the State Water Resources Control Board approved regulations, based on Governor Brown's Executive Order, mandating the District to reduce its water consumption by 20% percent for June 2015 through February 2016 as compared to the same months in 2013.
- E. California Plumbing Code Chapter 16 entitled "Alternative Water Sources for Nonpotable Application" contains the current state-wide requirements for the construction, alteration, discharge, use and repair of alternative water source systems", including gray water.
- F. The provisions of Chapter 16 of the state Plumbing Code are intended to, among other things, to conserve water by facilitating greater reuse of laundry, shower, lavatory, and similar sources of discharge for irrigation and/or indoor use.
- G. Requiring the installation of gray water reuse standards will improve the effectiveness of the District's efficiency standards for interior and exterior water use.
- H. Conditioning the approval for a new water service or enlargement of an existing service based upon a substantial remodel as defined above on an applicant's installation of gray water systems will encourage water re-use and preserve the water supply.
- I. Careful water management requires the implementation of a host of water conservation at all times, not just in times of drought.
- J. Article X Section 2 of the California Constitution declares that the general welfare requires that water resources be put to beneficial use to the fullest extent of which they are capable and that the waste, unreasonable use or unreasonable method of use of water be prevented, and that conservation of such waters is to be exercised with a view to the reasonable and beneficial use thereof in the interest of the people and the public welfare.
- K. California Water Codes Section 375 authorized water suppliers to adopt and enforce a comprehensive water conservation program to reduce water consumption and conserve supplies.
- L. The adoption and enforcement of the water conservation program contained in this

ordinance is necessary to manage and conserve the District's water supply and ensure the sustainability and reliability of the same while preventing water waste.

- M. The Board finds this ordinance is exempt from the provisions of the California Environmental Quality Act (CEQA) in accordance with Section 15061(b) (3) of the CEQA Guidelines.
- N. After more than twenty years of innovative and aggressive conservation efforts, more needs to be accomplished. The water conservation program required by this ordinance is necessary to conserve additional water for beneficial use and to preserve the District's water supply.

**SECTION 4.** Environmental Determination: This project has been reviewed for compliance with the California Environmental Quality Act and qualifies for an exemption under the General Rule section 15061(b) (3) because it can be seen with certainty that there is no possibility that the activity in question may have a significant effect on the environment.

**SECTION 5.** Severability: If any section, subsection, sentence, clause, phrase, portion or part of this ordinance is for any reason held to be invalid or unconstitutional by any court of competent jurisdiction, such section shall not affect the validity of the remaining portions of this code. The Board of Directors hereby declares that it would have adopted this ordinance and each section, subsection, sentence, clause, phrase, part or portion thereof, irrespective of the fact that any one or more sections subsections, clauses, phrases, parts or portions be declared invalid or unconstitutional.

**SECTION 6. Effective Date:** This ordinance shall be effective 90 days from the date of its adoption.

**PASSED AND ADOPTED** this 3rd day of November 2015, by the following vote of the Board:

AYES: Directors Bragman, Gibson, Quintero and Russell

NOES: None

ABSENT: Director Koehler (recusal)

President, Board of Directors

ATTEST: <u>Janua</u> Secretary, Board of Directors

Appendices 2020 Urban Water Management Plan Marin Municipal Water District



# Appendix K

## AWWA Water Loss Reports



Reporting	r Audit Software: WAS v5 American Water Works Ass Convribit © 2014 All Rights R	5.0 sociation.
Click to access definition     Water Audit Report for: Marin Municipal Water     Click to add a comment     Reporting Year: 2019     1/201	P District (2110002) 9 - 12/2019	
Please enter data in the white cells below. Where available, metered values should be used; if metered values input data by grading each component (n/a or 1-10) using the drop-down list to the left of the input cell. Ho	lues are unavailable please estimate a value. Indicate your confidence in the accuracy of the ver the mouse over the cell to obtain a description of the grades	
All volumes to be entere	d as: ACRE-FEET PER YEAR	
To select the correct data grading for each input, determine the highest the utility meets or exceeds all criteria for that grade and all gra	grade where des below it. Master Meter and Supply Error Adjustments	
WATER SUPPLIED <	Enter grading in column 'E' and 'J'> Pcnt: Value:	
Volume from own sources: + ? 4	19,984.567 acre-ft/yr + ? 81.550 acr	re-ft/yr
Water imported: + ? 9	5,502.991 acre-ft/yr + ? 10 -0.52% O () acre-ft/yr + 2	re-ft/yr
	Enter negative % or value for under-registration	on
WATER SUPPLIED:	25,517.873 acre-ft/yr Enter positive % or value for over-registration	I
AUTHORIZED CONSUMPTION	Click here: ?	
Billed metered: + 7 7 Billed unmetered: + 7 n/a	22,716.800 acre-ft/yr for help using option 0.000 acre-ft/yr buttons below	
Unbilled metered: + ? 8	5.087 acre-ft/yr Pcnt: Value:	
Unbilled unmetered: + ? 8	8.306 acre-ft/yr	re-ft/yr
AUTHORIZED CONSUMPTION:	22,730.193 acre-ft/yr Use buttons to select percentage of water	
	supplied OR	
WATER LOSSES (Water Supplied - Authorized Consumption)	2,787.679 acre-ft/yr	
Apparent Losses	Pcnt: Value:	
Unauthorized consumption:	63.795 acre-ft/yr 0.25% (●) ( ) acre-ft/yr acre-ft/yr	re-ft/yr
Systematic data handling errors: + ?	56.792         acre-ft/yr         0.25%         (Image: Contract of the second	re-ft/yr
Default option selected for Systematic data handling errors - a g	rading of 5 is applied but not displayed	
Apparent Losses:	823.325 acre-ft/yr	
Real Losses (Current Annual Real Losses or CARL)		
Real Losses = Water Losses - Apparent Losses: ?	1.964.354 acre-ft/vr	
WATER LOSSES:	2,787.679 acre-ft/yr	
WATER LOSSES:	2,787.679 acre-ft/yr	
WATER LOSSES:       NON-REVENUE WATER       ?	2,787.679 acre-ft/yr 2,801.073 acre-ft/yr	
WATER LOSSES:  NON-REVENUE WATER  Water Losses + Unbilled Metered + Unbilled Unmetered	2,787.679 acre-ft/yr 2,801.073 acre-ft/yr	
WATER LOSSES:          WATER LOSSES:         NON-REVENUE WATER:         • Water Losses + Unbilled Metered + Unbilled Unmetered         SYSTEM DATA	2,787.679 acre-ft/yr 2,801.073 acre-ft/yr	
WATER LOSSES: MON-REVENUE WATER = Water Losses + Unbilled Metered + Unbilled Unmetered SYSTEM DATA Length of mains: + 2 8 Number of active AND inactive service connections: + 2 7	2,787.679 acre-ft/yr 2,801.073 acre-ft/yr 853.1 miles 63,704	
WATER LOSSES:         WATER LOSSES:         NON-REVENUE WATER:         * Water Losses + Unbilled Metered + Unbilled Unmetered         SYSTEM DATA         Length of mains:         * ?         Number of active AND inactive service connections:         Service connection density:         ?	2,787.679         acre-ft/yr           2,801.073         acre-ft/yr           853.1         miles           63,704         conn./mile main	
WATER LOSSES:         WATER LOSSES:         NON-REVENUE WATER:         * Water Losses + Unbilled Metered + Unbilled Unmetered         SYSTEM DATA         Length of mains: + ?         Number of active AND inactive service connections: + ?         Number of active AND inactive service connections: + ?         Are customer meters typically located at the curbstop or property line?	2,787.679 acre-ft/yr 2,801.073 acre-ft/yr 853.1 miles 63,704 75 conn./mile main Yes (length of service line, beyond the property	
WATER LOSSES:         WATER LOSSES:         NON-REVENUE WATER: ?         = Water Losses + Unbilled Metered + Unbilled Unmetered         SYSTEM DATA         Length of mains: + ?         Number of active AND inactive service connections: + ?         Number of active AND inactive service connections: + ?         Are customer meters typically located at the curbstop or property line?         Average length of customer service line: + ?	2,787.679 acre-ft/yr 2,801.073 acre-ft/yr 853.1 miles 63,704 75 conn./mile main Yes (length of service line, beyond the property boundary, that is the responsibility of the utility)	
WATER LOSSES:         WATER LOSSES:         NON-REVENUE WATER       ?         = Water Losses + Unbilled Metered + Unbilled Unmetered       ?         SYSTEM DATA         Length of mains:       +       ?       8         Number of active AND inactive service connections:       +       ?       7         Service connection density:       ?       7         Are customer meters typically located at the curbstop or property line?	2,787.679 acre-ft/yr 2,801.073 acre-ft/yr 853.1 miles 63,704 75 conn./mile main Yes (length of service line, <u>beyond</u> the property boundary, that is the responsibility of the utility) prading score of 10 has been applied 109.3 psi	
WATER LOSSES:         NON-REVENUE WATER         = Water Losses + Unbilled Metered + Unbilled Unmetered         SYSTEM DATA         Length of mains:         Year Connections:         Year Connection         Year Connection </td <td>2,787.679 acre-ft/yr 2,801.073 acre-ft/yr 2,801.073 acre-ft/yr 853.1 miles 63,704 75 conn./mile main Yes (length of service line, <u>beyond</u> the property boundary, that is the responsibility of the utility) prading score of 10 has been applied 109.3 psi</td> <td></td>	2,787.679 acre-ft/yr 2,801.073 acre-ft/yr 2,801.073 acre-ft/yr 853.1 miles 63,704 75 conn./mile main Yes (length of service line, <u>beyond</u> the property boundary, that is the responsibility of the utility) prading score of 10 has been applied 109.3 psi	
WATER LOSSES:         WATER LOSSES:         NON-REVENUE WATER       ?         = Water Losses + Unbilled Metered + Unbilled Unmetered       ?         SYSTEM DATA       Length of mains: + ?         Number of active AND inactive service connections: + ?       ?         Number of active AND inactive service connections: + ?       ?         Are customer meters typically located at the curbstop or property line?       Average length of customer service line: + ?         Average length of customer service line has been set to zero and a data of Average operating pressure: + ?       ?         COST DATA       ?	2,787.679 acre-ft/yr 2,801.073 acre-ft/yr 2,801.073 acre-ft/yr 853.1 miles 63,704 75 conn./mile main Yes (length of service line, <u>beyond</u> the property boundary, that is the responsibility of the utility) rading score of 10 has been applied 109.3 psi	
WATER LOSSES:         WATER LOSSES:         NON-REVENUE WATER       ?         = Water Losses + Unbilled Metered + Unbilled Unmetered       ?         SYSTEM DATA         Length of mains:       +       ?       8         Number of active AND inactive service connections:       +       ?       7         Service connection density:       ?       7         Are customer meters typically located at the curbstop or property line?       Average length of customer service line:       +       ?         Average length of customer service line has been set to zero and a data of Average operating pressure:       +       ?       7         COST DATA         Total annual cost of operating water system:       *       ?       10	2,787.679 acre-ft/yr 2,801.073 acre-ft/yr 853.1 miles 63,704 75 conn./mile main Yes (length of service line, <u>beyond</u> the property boundary, that is the responsibility of the utility) rading score of 10 has been applied 109.3 psi \$80,237,859 \$/Year	
WATER LOSSES:         WATER LOSSES:         NON-REVENUE WATER: ?         = Water Losses + Unbilled Metered + Unbilled Unmetered         SYSTEM DATA         Length of mains: + ?         Number of active AND inactive service connections: + ?         Number of active AND inactive service connections: + ?         Number of active AND inactive service connections: + ?         Are customer meters typically located at the curbstop or property line?         Average length of customer service line: + ?         Average length of customer service line has been set to zero and a data of Average operating pressure: + ? 7         COST DATA         Total annual cost of operating water system: + ?         Average length of customer service line has been set to zero and a data of Average operating pressure: + ? 7         COST DATA         Total annual cost of operating water system: + ?         10         Customer retail unit cost (applied to Apparent Losses): + ?	2,787.679 acre-ft/yr 2,801.073 acre-ft/yr 2,801.073 acre-ft/yr 853.1 miles 63,704 75 conn./mile main Yes (length of service line, beyond the property boundary, that is the responsibility of the utility) rading score of 10 has been applied 109.3 psi \$80,237,859 \$/Year \$4.84 \$/100 cubic feet (ccf)	
WATER LOSSES:         WATER LOSSES:         NON-REVENUE WATER:         2         # Water Losses + Unbilled Metered + Unbilled Unmetered         SYSTEM DATA         Length of mains:       4       2       8         Number of active AND inactive service connections:       4       7       7         Service connection density:       7         Are customer meters typically located at the curbstop or property line?         Average length of customer service line has been set to zero and a data of Average operating pressure:         Average length of customer service line has been set to zero and a data of Average operating pressure:         Average length of customer service line has been set to zero and a data of Average operating pressure:         Average length of customer service line has been set to zero and a data of Average operating pressure:         Average length of customer service line has been set to zero and a data of Average operating pressure:         Average length of customer service line has been set to zero and a data of Average operating pressure:         Average length of customer service line has been set to zero and a data of Average operating pressure:         Average length of customer service line has been set to zero and a data of Average operating pres	2,787.679       acre-ft/yr         2,801.073       acre-ft/yr         853.1       miles         63,704       conn./mile main         Yes       (length of service line, beyond the property boundary, that is the responsibility of the utility)         rrading score of 10 has been applied       109.3         109.3       psi         \$80,237,859       \$/Year         \$414.03       \$/acre-ft         Use Customer Retail Unit Cost to value real losses	
WATER LOSSES:         WATER LOSSES:         NON-REVENUE WATER:         * Water Losses + Unbilled Metered + Unbilled Unmetered         SYSTEM DATA         Length of mains:       +       ?       8         Number of active AND inactive service connections:       +       ?       7         SYSTEM DATA         Length of mains:       +       ?       8         Number of active AND inactive service connections:       +       ?       7         Service connection density:       ?       7         Average length of customer service line has been set to zero and a data of Average operating pressure:       +       ?       7         COST DATA         Total annual cost of operating water system:       +       ?       10         Customer retail unit cost (applied to Apparent Losses):       +       ?       8         Variable production cost (applied to Real Losses):       +       ?       5         WATER AUDIT DATA VALIDITY SCORE:	2,787.679       acre-ft/yr         2,801.073       acre-ft/yr         853.1       miles         63,704       75         conn./mile main       75         Yes       (length of service line, beyond the property boundary, that is the responsibility of the utility)         rading score of 10 has been applied       109.3         109.3       psi         \$80,237,859       \$/Year         \$414.03       \$/acre-ft         Use Customer Retail Unit Cost to value real losses	
WATER LOSSES:         MON-REVENUE WATER         = Water Losses + Unbilled Metered + Unbilled Unmetered         SYSTEM DATA         Length of mains:         Number of active AND inactive service connections:         Yearage length of customer service connections:         Are customer meters typically located at the curbstop or property line?         Average length of customer service line has been set to zero and a data of Average operating pressure:         Average length of customer service line has been set to zero and a data of Average operating pressure:         COST DATA         Total annual cost of operating water system:         Yariable production cost (applied to Apparent Losses):         Yariable production cost (applied to Real Losses):         WATER AUDIT DATA VALIDITY SCORE:	2,787.679 acre-ft/yr 2,801.073 acre-ft/yr 2,801.073 acre-ft/yr 853.1 miles 63,704 75 conn./mile main Yes (length of service line, <u>beyond</u> the property boundary, that is the responsibility of the utility) trading score of 10 has been applied 109.3 psi \$80,237,859 \$/Year \$4.84 \$/100 cubic feet (ccf) \$4.84 \$/100 cubic feet (ccf) \$/4.84 \$/100 cubic feet (ccf) \$/acre-ft Use Customer Retail Unit Cost to value real losses out of 100 ***	
WATER LOSSES:         MON-REVENUE WATER         = Water Losses + Unbilled Metered + Unbilled Unmetered         SYSTEM DATA         Length of mains:         Yearage         Number of active AND inactive service connections:         Yearage         Number of active AND inactive service connections:         Yearage         Are customer meters typically located at the curbstop or property line?         Average length of customer service line has been set to zero and a data of Average operating pressure:         Average length of customer service line has been set to zero and a data of Average operating pressure:         Yearage length of customer service line has been set to zero and a data of Average operating pressure:         Yearage length of customer service line has been set to zero and a data of Average operating pressure:         Yearage length of customer service line has been set to zero and a data of Average operating pressure:         Yearage length of customer service line has been set to zero and a data of Average operating pressure:         Yearage length of customer service line has been set to zero and a data of Average operating pressure:         Yearage length of customer service line has been set to zero and a data of Average operating pressure:         Yearage length of customer service line has been set to zero and a data of Average operating pressure:         Yearage length of customer service line has been set to zero and a data of Average operating p	2,787.679       acre-ft/yr         2,801.073       acre-ft/yr         2,801.073       acre-ft/yr         853.1       miles         63,704       75         75       conn./mile main         Yes       (length of service line, beyond the property boundary, that is the responsibility of the utility)         rading score of 10 has been applied       109.3         109.3       psi         \$80,237,859       \$/Year         \$4.84       \$/100 cubic feet (ccf)         \$/acre-ft       Use Customer Retail Unit Cost to value real losses         out of 100       ***         uided in the calculation of the Water Audit Data Validity Score	
WATER LOSSES:         NON-REVENUE WATER         = Water Losses + Unbilled Metered + Unbilled Unmetered         SYSTEM DATA         Length of mains:       ?         Number of active AND inactive service connections:       ?         Yerage length of customer service connection density:       ?         Are customer meters typically located at the curbstop or property line?       Average length of customer service line:         Average length of customer service line has been set to zero and a data of Average operating pressure:       ?         COST DATA       Total annual cost of operating water system:         Yariable production cost (applied to Apparent Losses):       ?         Variable production cost (applied to Real Losses):       ?         WATER AUDIT DATA VALIDITY SCORE:       *** YOUR SCORE IS: 62         A weighted scale for the components of consumption and water loss is into the components of consumption and water loss is into the components of consumption and water loss is into the components of consumption and water loss is into the components of consumption and water loss is into the components of consumption and water loss is into the components of consumption and water loss is into the components of consumption and water loss is into the components of consumption and water loss is into the components of consumption and water loss is into the components of consumption and water loss is into the components of consumption and water loss is into the components of consumption and water loss is into the components of consumption and water loss is into the compon	2,787.679 acre-ft/yr 2,801.073 acre-ft/yr 2,801.073 acre-ft/yr 853.1 miles 63,704 75 conn./mile main Yes (length of service line, <u>beyond</u> the property boundary, that is the responsibility of the utility) rading score of 10 has been applied 109.3 psi \$80,237,859 \$/Year \$4.84 \$/100 cubic feet (ccf) \$414.03 \$/acre-ft Use Customer Retail Unit Cost to value real losses out of 100 *** studed in the calculation of the Water Audit Data Validity Score	
WATER LOSSES:         NON-REVENUE WATER         = Water Losses + Unbilled Metered + Unbilled Unmetered         SYSTEM DATA         Length of mains:         Yerage length of customer service connections:         Yerage length of customer service line:         Yerage length of customer service line has been set to zero and a data of Average operating pressure:         Yerage length of customer service line has been set to zero and a data of Average operating pressure:         Yerage length of customer service line has been set to zero and a data of Average operating pressure:         Yerage length of customer service line has been set to zero and a data of Average operating pressure:         Yerage length of customer service line has been set to zero and a data of Average operating pressure:         Yerage length of customer service line has been set to zero and a data of Average operating pressure:         Yerage length of customer service line has been set to zero and a data of Average operating pressure:         Yerage length of customer service line has been set to zero and a data of Average operating pressure:         Yerage length of customer service line has been set to zero and a data of Average operating pressure:         Yerage length of customer service line has been set to zero and a data of Average operating pressure:	2,787.679 acre-ft/yr 2,801.073 acre-ft/yr 3,853.1 miles 63,704 75 conn./mile main Yes (length of service line, <u>beyond</u> the property boundary, that is the responsibility of the utility) rading score of 10 has been applied 109.3 psi \$80,237,859 \$/Year \$4.84 \$/100 cubic feet (ccf) \$414.03 \$/acre-ft Use Customer Retail Unit Cost to value real losses out of 100 *** duded in the calculation of the Water Audit Data Validity Score	
WATER LOSSES:         NON-REVENUE WATER         = Water Losses + Unbilled Metered + Unbilled Unmetered         SYSTEM DATA         Length of mains:       2         Number of active AND inactive service connections:       7         Yerage length of customer service connection density:       7         Are customer meters typically located at the curbstop or property line?       7         Average length of customer service line has been set to zero and a data of Average operating pressure:       10         Average length of customer service line has been set to zero and a data of Average operating pressure:       7         COST DATA       Total annual cost of operating water system:       10         Customer retail unit cost (applied to Apparent Losses):       7       8         Variable production cost (applied to Real Losses):       7       5         WATER AUDIT DATA VALIDITY SCORE:       *** YOUR SCORE IS: 62         Mater Store ATTENTION:       Based on the information provided, audit accuracy can be improved by addressing the following components of the information provided, audit accuracy can be improved by addressing the following components	2,787.679       acre-ft/yr         2,801.073       acre-ft/yr         853.1       miles         63,704       75         conn./mile main       75         Yes       (length of service line, beyond the property boundary, that is the responsibility of the utility)         rading score of 10 has been applied       109.3         109.3       psi         \$80,237,859       \$/Year         \$4.84       \$/100 cubic feet (ccf)         \$414.03       \$/acre-ft         Use Customer Retail Unit Cost to value real losses         out of 100 ***         edued in the calculation of the Water Audit Data Validity Score         ants:	
WATER LOSSES:         NON-REVENUE WATER       2         = Water Losses + Unbilled Metered + Unbilled Unmetered       2         SYSTEM DATA       Length of mains:       •       2       8         Number of active AND inactive service connections:       •       2       7         SYSTEM DATA       Length of mains:       •       2       8         Number of active AND inactive service connection density:       •       2       7         Are customer meters typically located at the curbstop or property line?       Average length of customer service line has been set to zero and a data of Average length of customer service line has been set to zero and a data of Average operating pressure:       •       ?       7         COST DATA       Total annual cost of operating water system:       •       ?       7         COST DATA       Total annual cost of operating water system:       •       ?       ?       8         Variable production cost (applied to Apparent Losses):       •       ?       8       ?       ?         WATER AUDIT DATA VALIDITY SCORE:       *** YOUR SCORE IS: 62       A weighted scale for the components of consumption and water loss is into the information provided, audit accuracy can be improved by addressing the following components is not supported by addressing the following components is not supported by addressing the following components is not supported by addre	2,787.679       acre-ft/yr         2,801.073       acre-ft/yr         853.1       miles         63,704       rs         75       conn./mile main         Yes       (length of service line, beyond the property boundary, that is the responsibility of the utility)         rading score of 10 has been applied       109.3         109.3       psi         \$44.44       \$/Year         \$414.03       \$/Year         \$414.03       \$/acre-ft         Use Customer Retail Unit Cost to value real losses	
WATER LOSSES:         NON-REVENUE WATER       2         = Water Losses + Unbilled Metered + Unbilled Unmetered       2         SYSTEM DATA       Length of mains:       2       8         Number of active AND inactive service connections:       2       7         System Data       2       8       7         Number of active AND inactive service connection density:       2       7         Are customer meters typically located at the curbstop or property line?       7       7         Average length of customer service line has been set to zero and a data of Average operating pressure:       9       7         COST DATA       Total annual cost of operating water system:       +       2       10         Customer retail unit cost (applied to Apparent Losses):       +       2       8       10         Variable production cost (applied to Real Losses):       +       2       5       10         WATER AUDIT DATA VALIDITY SCORE:       *** YOUR SCORE IS: 62         A weighted scale for the components of consumption and water loss is incomponents of consumption and water loss is incom	2,787.679       acre-ft/yr         2,801.073       acre-ft/yr         853.1       miles         63,704       conn./mile main         Yes       (length of service line, beyond the property boundary, that is the responsibility of the utility)         rading score of 10 has been applied       109.3         109.3       psi         \$80,237,859       \$/Year         \$414.03       \$/100 cubic feet (ccf)         \$414.03       \$/acre-ft         Use Customer Retail Unit Cost to value real losses         out of 100 ***         duded in the calculation of the Water Audit Data Validity Score         ents:	

	AWWA Free Water Audit Software: System Attributes and Performance Indicators	WAS v5.0 American Water Works Association. Copyright © 2014, All Rights Reserved.
	Water Audit Report for: Marin Municipal Water District (2110002) Reporting Year: 2019 1/2019 - 12/2019	
	*** YOUR WATER AUDIT DATA VALIDITY SCORE IS: 62 out of 100 ***	
<u>System Attributes:</u>	Apparent Losses: 823.325 acre-ft/yr + Real Losses: 1,964.354 acre-ft/yr	
	= Water Losses: 2,787.679 acre-ft/yr	
	Unavoidable Annual Real Losses (UARL): 1,735.60 acre-ft/yr	
	Annual cost of Apparent Losses: \$1,735,820	
	Annual cost of Real Losses: \$813,301 Valued at	/ariable Production Cost
Performance Indicators	Return to Reporti	ig worksneet to change this assumption
renormance indicators.	Non-revenue water as percent by volume of Water Supplied:	
Financial:	Non-revenue water as percent by cost of operating system: 3.2% Real Losses v	valued at Variable Production Cost
Г	Apparent Losses per service connection per day: 11.54 gallons/connection	tion/day
	Real Losses per service connection per day: 27.53 gallons/connection	tion/day
	Real Losses per length of main per day*: N/A	
	Real Losses per service connection per day per psi pressure: 0.25 gallons/connection	ction/day/psi
	From Above, Real Losses = Current Annual Real Losses (CARL): 1,964.35 acre-feet/year	
	Infrastructure Leakage Index (ILI) [CARL/UARL]: 1.13	
* This performance indicator applies f	or systems with a low service connection density of less than 32 service connections/mile of pipeline	



### AWWA Free Water Audit Software: <u>User Comments</u>

WAS v5.0 American Water Works Association. Copyright © 2014, All Rights Reserved.

Use this worksheet to add comments or notes to explain how an input value was calculated, or to document the sources of the information used.

General Comment:	See attached documentation
Audit Item	Comment
Volume from own sources:	
Vol. from own sources: Master meter error adjustment:	
Water imported:	
Water imported: master meter error adjustment:	
Water exported:	
Water exported: master meter error adjustment:	
Billed metered:	
Billed unmetered:	
Unbilled metered:	

Audit Item	Comment
Unbilled unmetered:	
Unauthorized consumption:	
Customer metering inaccuracies:	
Systematic data handling errors:	
Length of mains:	
Number of active AND inactive service connections:	
Average length of customer service line:	
Average operating pressure:	
Total annual cost of operating water system:	
Customer retail unit cost (applied to <u>Apparent Losses):</u>	
Variable production cost (applied to <u>Real Losses):</u>	

Audit Item Comment	
--------------------	--

Audit Item Comment	
--------------------	--

Audit Item	Comment
Audit Item	Comment
------------	---------

Audit Item	Comment

|--|

Audit Item	Comment

Audit Item Comment	
--------------------	--

|--|

		Audit Item	Comment
--	--	------------	---------

Audit Item	
------------	--

Audit Item	Comment
	Audit Item

Audit Item Comment	
--------------------	--

Audit Item	Comment

Audit Item	Comment

斧		AM	/WA Free Wa	ter Audit Software: <u>Wate</u>	er Balance Americ	WAS v5.0 an Water Works Association.
		Wa	tter Audit Report for: Reporting Year: Data Validity Score:	Marin Municipal Water District (21100 2019 62	002) 1/2019 - 12/2019	
		Water Exported 0.000			Billed Water Exported	Revenue Water 0.000
Own Sources			Authorized	Billed Authorized Consumption	Billed Metered Consumption (water exported is removed) 22,716.800 Billed Unmetered Consumption	Revenue Water
(Adjusted for known errors)			Consumption 22,730.193	Unbilled Authorized Consumption	0.000 Unbilled Metered Consumption 5.087	Non-Revenue Water (NRW)
19,986.117				13.393	Unbilled Unmetered Consumption 8.306	
	System Input 25,517.873	Water Supplied 25,517.873		Apparent Losses 823.325	Unauthorized Consumption 63.795 Customer Metering Inaccuracies 702.739	2,801.073
			Water Losses		Systematic Data Handling Errors 56.792	
Water Imported 5,531.756			2,787.679	Real Losses 1,964.354	Leakage on Transmission and/or Distribution Mains Not broken down Leakage and Overflows at Utility's Storage Tanks Not broken down	
					Leakage on Service Connections Not broken down	



### AWWA Free Water Audit Software: Grading Matrix

#### WAS 5.0

	The grading parison of the company and the company and the company and all improvements and palicy as is highlighted is valley. Audit company is likely to be improved by an inditions there is the company and the company an										
Grading >>>	n/a										10
Grading	IIVa	<u> </u>	2		4	WATER SUPPLI	ED State		<u>ه</u>	9	
Volume from own sources:	Select this grading only if the water utility purchases/imports all of its water resources (i.e. has no sources of its own)	Less than 25% of water production sources are metered, remaining sources are estimated. No regular meter accuracy testing or electronic calibration conducted.	25% - 50% of treated water production sources are metered; other sources estimated. No regular meter accuracy testing or electronic calibration conducted.	Conditions between 2 and 4	50% - 75% of treated water production sources are metered, other sources setimated. Occasional meter accuracy testing or electronic calibration conducted.	Conditions between 4 and 6	At least 75% of treated water production sources are metered, <u>or</u> a least 90% of the source flow is derived from metered sources. Meter accuracy testing and/or electronic conducted annualy. Less than 25% of tested meters are found outside of +/- 6% accuracy.	t Conditions between 6 and 8	100% of treated water production sources are metered, meter accuracy lesting and electronic aditoration of related instrumentation is conducted annually, less than 10% of meters are found outside of +/- 6% accuracy	Conditions between 8 and 10	100% of treated water production sources are metered, meter accuracy testing and electronic calibration of related instrumentation is conducted semi-annually, with less than 10% found outside of +/- 3% accuracy. Procedures are reviewed by a third party knowledgeable in the M36 methodology.
Improvements to attain higher data grading for "Volume from own Sources" component:		to qualify for 2: Organize and launch efforts to collect data for determining volume from own sources	to qualify for 4: Locate all water production sources field, launch meter accuracy testing begin to install meters on unmeter sources and replace any obsolete	s on maps and in the g for existing meters, ad water production /defective meters.	to qualify for 6 Formalize annual meter accuracy meters; specify the frequency of installation of meters on unmeter sources and complete replacement of meters.	testing for all source testing. Complete ed water production of all obsolete/defective	<u>to qualify for 8:</u> Conduct annual meter accuracy test related instrumentation on all mete regular basis. Complete project to in defective existing, meters so that en population is metered. Repair or repi +/- 6% accuracy	ing and calibration of r installations on a nstall new, or replace tire production meter ace meters outside of r.	to qualify for 11 Maintain annual meter accuracy te related instrumentation for all meter replace meters outside of +/- 3% acc meter technology; pilot one or mo innovative meters intempt to fu accuracy.	<u>):</u> sting and calibration of installations. Repair or suracy. Investigate new re replacements with rther improve meter	to maintain 10: Standardize meter accuracy test frequency to semi-annual, or more frequent, for all meters. Repair or replace meters outside of +/- 3% accuracy. Continually investigate/pilol improving metering technology.
Volume from own sources master meter and supply error adjustment:	Select n/a only if the water utility fails to have meters on its sources of supply	Inventory information on meters and paper records of measured volumes exist but are incomplete and/or in a very crude condition; data error cannot be determined	No automatic datalogging of production volumes; daily readings are scribed on paper records without any accountability controls. Flows are not balanced across the water distribution system: tank/storage elevation changes are not employed in calculating the "Volume from own sources" component and archived flow data is adjusted only when grossly evident data error occurs.	Conditions between 2 and 4	Production meter data is logged automatically in electronic format and reviewed at least on a monthly basis with necessary corrections implemented. "Volume from own sources" tabulations include estimate of daily changes in tarks/storage facilities. Meter data is adjusted when gross data errors occur, or occasional meter testing deems this necessary.	Conditions between 4 and 6	Hourly production meter data logged automatically & reviewed on at least a weekly basis. Data is adjusted to correct gross error when mater/instrumentation equipment mafunction is detected; and/or error it confirmed by meter accuracy testing, Tank/storage facility elevation changes are automatically used in calculating a balanced "Volume from own sources" component, and data gaps in the archived data are corrected on at least a weekly basis.	<sup>S</sup> Conditions between 6 and 8	Continuous production meter data is logged automatically & reviewed each business day. Data is adjusted to correct gross error from detected meter/instrumentation equipment malfunction and/or results of meter accuracy testing. Tank/storage facility elevation changes are automatically used in 'Volume from own sources' tabulations and data gaps in the archived data are corrected on a daily basis.	Conditions between 8 and 10	Computerized system (SCADA or similar) automatically balances flows from all sources and storages; results are reviewed each business day. Tigh accountability controls ensure that all data gaps that occur in the archived flow data are quickly detected and corrected. Regular calibrations between SCADA and sources meters ensures minimal data transfer error.
Improvements to attain higher data grading for "Master meter and supply error adjustment" component:		to qualify for 2: Develop a plan to restructure recordkeeping system to capture all flow data; set a procedure to review flow data on a daily basis to detect input errors. Obtain more reliable information about existing meters by conducting field inspections of meters and related instrumentation, and obtaining manufacturer literature.	to qualify for 4: Install automatic datalogging equip meters. Complete installation of lev all tanks/storage facilities and incluu automatic calculation routine in a co Construct a computerized listing archive input volumes, tank/storage import/expot flows in order to deter "Water Supplied" volume for the dis a procedure to review this data on detect gross anomalies and	ment on production el instrumentation at de tank level data in omputerized system. or spreadsheet to volume changes and mine the composite tribution system. Set a monthly basis to d data gaps.	to qualify for 6 Refine computerized data collection hourly production meter data that is weekly basis to detect specific data Use daily net storage change to basis "Water Supplied" volume. No basis errors are implemented on a	and archive to include reviewed at least on a anomalies and gaps. nee flows in calculating ary corrections to data a weekly basis.	to qualify for 8: Ensure that all flow data is collectice least an hourly basis. All data is rev errors corrected each business day, variations are employed in calculati Supplied" component. Adjust prod gross error and inaccuracy conf	I and archived on at iewed and detected Tank/storage levels ng balanced Water uction meter data for irmed by testing.	<u>to quality for 11</u> Link all production and tank/storage data to a Supervisory Control & Dat System, or similar computerized mon and establish automatic flow bala regularly calibrate between SCADA an is reviewed and corrected eac	): facility elevation change a Acquisition (SCADA) Intoring/control system, noing algorithm and d source meters. Data ch business day.	to maintain 10: Monitor meter innovations for development of more accurate and less expensive flowmeters. Continue to replace or repair meters as they perform outside of desired accuracy limits. Stay abreast of new and more accurate water level instruments to better record tank/storage levels and archive the variations in storage volume. Keep current with SCADA and data management systems to ensure that archived data is well- managed and error free.
Water Imported:	Select n/a if the water utility's supply is exclusively from its own water resources (no bulk purchased/ imported water)	Less than 25% of imported water sources are metered, remaining sources are estimated. No regular meter accuracy testing.	25% - 50% of imported water sources are metered; other sources estimated. No regular meter accuracy testing.	Conditions between 2 and 4	50% - 75% of imported water sources are metered, other sources estimated. Occasional meter accuracy testing conducted.	Conditions between 4 and 6	At least 75% of imported water sources are metered, meter accuracy testing and/or electronic calibration of related instrumentation is conducted annually for all meter installations. Less than 25% of tested meters are found outside of +/- 6% accuracy.	f Conditions between 6 and 8	100% of imported water sources are metered, meter accuracy testing and electronic calibration of related instrumentation is conducted annually, less than 10% of meters are found outside of +/- 6% accuracy	Conditions between 8 and 10	100% of imported water sources are metered, meter accuracy testing and electronic calibration of related instrumentation is conducted semi- annually for all meter installations, with less than 10% of accuracy tests found outside of +/- 3% accuracy.

Grading >>>	n/a	1	2	3	4	5	6	7	8	9	10
Improvements to attain higher data grading for "Water Imported Volume" component: (Note: usually the water - "supplier selling the water - "the Exporter" - to the utility being audited is responsible to maintain the metering installation measuring the imported volume. The utility should coordinate carefully with the Exporter to ensure that adequate meter upkeep takes place and an accurate measure of the Water Imported volume is quantified.)		to qualify for 2: Review bulk water purchase agreements with partner suppliers; confirm requirements for use and maintenance of accurate metering, identify needs for new or replacement meters with goal to meter all imported water sources.	<u>To qualify for 4:</u> Locate all imported water sources field, launch meter accuracy testing begin to install meters on unmeter interconnections and replace obsole	<u>To qualify for 4</u> : Locate all imported water sources on maps and in the field, launch meter accuracy testing for existing meters, begin to install meters on unmetered imported water interconnections and replace obsolete/defective meters.		: esting for all imported gular meter accuracy led instrumentation. unmetered imported replacement of neters.	to qualify for 8: Complete project to install new, or repl on all imported water interconnection meter accuracy testing for all importe conduct calibration of related instru annually. Repair or replace meters accuracy.	ace defective, meters s. Maintain annual d water meters a mentation at least outside of +/- 6%	to qualify for 10 Conduct meter accuracy testing for annual basis, along with calitor instrumentation. Repair or replace m accuracy. Investigate new meter ter more replacements with innovative improve meter accu	al meters on a semi- tion of all related eters outside of +/- 3% hnology, pilot one or meters in attempt to racy.	to maintain 10: Standardize meter accuracy test frequency to semi-annual, or more frequent, for all meters. Continue to conduct calibration of related instrumentation on a semi-annual basis. Repair or replace meters outside of +/. 3% accuracy. Continually investigate/pilot improving metering technology.
Water imported master meter and supply error adjustment:	Select n/a if the Imported water supply is unmetered, with Imported water quantities estimated on the billing invoices sen by the Exporter to the purchasing Utility.	Inventory information on imported meters and paper records of measured volumes exist but are incomplete and/or in a very crude condition, data error cannot be determined Written agreement(s) with water Exporter(s) are missing or written in vague language concerning meter management and testing.	No automatic datalogging of imported supply volumes; daily readings are scribed on paper records without any accountability controls to confirm data accuracy and the absence of errors and data gaps in recorded volumes. Written agreement requires meter accuracy testing but is vague on the details of how and who conducts the testing.	Conditions between 2 and 4	Imported supply metered flow data is logged automatically in electronic format and reviewed at least on a monthly basis by the Exporter with necessary corrections implemented. Meter data is adjusted by the Exporter when gross data errors are detected. A coherent data trail exists for this process to protect both the selling and the purchasing Utility. Written agreement exists and clearly states requirements and roles for meter accuracy testing and data management.	Conditions between 4 and 6	Hourly Imported supply metered data is logged automatically & reviewed on at least a weekly basis by the Exporter. Data is adjusted to correct gross error when meter/instrumentation equipment mafunction is detected; and to correct for error confirmed by meter accuracy testing. Any data gaps in the archived data are detected and corrected during the weekly review. A cohera data trail exists for this process to protect both the selling and the purchasing Utility.	Conditions between 6 and 8	Continuous Imported supply metered flow data is logged automatically & reviewed each business day by the Exporter. Data is adjusted to correct gross error from detected meter/instrumentation equipment malfunction and/or results of meter accuracy testing. Any data errors/gaps are detected and corrected on a daily basis. A data trail exists for the process to protect both the selling and the purchasing Utility.	Conditions between 8 and 10	Computerized system (SCADA or similar) automatically records data which is reviewed each business day by the Exporter. Tight accountability controls ensure that all error/data gaps that occur in the archived flow data are quickly detected and corrected. A reliable data trai exists and contract provisions for meter testing and data management are reviewed by the selling and purchasing Utility at least once every five years.
Improvements to attain higher data grading for "Water imported master meter and supply error adjustment component:		to qualify for 2: Develop a plan to restructure recordkeeping system to capture all flow data: set a procedure to review flow data on a daily basis to detect input errors. Obtain more reliable information about existing meters by conducting field inspections of meters and related instrumentation, and obtaining manufactureer literature. Review the written agreement between the selling and purchasing Utility.	to qualify for 4: Install automatic datalogging equip supply meters. Set a procedure to 1 monthy basis to detect gross anom Launch discussions with the Expor terms of the written agreements rega testing and data management, re necessary.	went on Imported eview this data on a alides and data gaps. Iters to jointly review fing meter accuracy vise the terms as	<u>to qualify for 6</u> Refine computerized data collection hourly imported supply metered flov at least on a weekly basis to detect 5 and gaps. Make necessary corre errors on a weekly l	; and archive to include vdata that is reviewed specific data anomalies extions to errors/data basis.	to qualify for 8; Ensure that all Imported supply me collected and archived on at least an h is reviewed and errors/data gaps ar business day.	tered flow data is ourly basis. All data e corrected each	to qualify for 10 Conduct accountability checks to co supply metered data is reviewed and c day by the Exporter. Results of all me data corrections should be available fn Exporter and the purchasing Utility. E a regular review and updating of the c the written agreement between the sel Utility; at least every fiv	firm that all Imported orrected each business ter accuracy tests and or sharing between the stablish a schedule for ontractual language in ling and the purchasing e years.	to maintain 10: Monitor meter innovations for development of more accurate and less expensive flowmeters; work with the Exporter to help identify meter replacement needs. Keep communication lines with Exporters open and maintain productive relations. Keep the written agreement ourrent with clear and explicit language that meets the ongoing needs of all parties.
Water Exported:	Select n/a if the water utility sells no bulk water to neighboring water utilities (no exported water sales)	Less than 25% of exported water sources are metered, remaining sources are estimated. No regular meter accuracy testing.	25% - 50% of exported water sources are metered; other sources estimated. No regular meter accuracy testing.	Conditions between 2 and 4	50% - 75% of exported water sources are metered, other sources estimated. Occasional meter accuracy testing conducted.	Conditions between 4 and 6	At least 75% of exported water sources are metered, meter accuracy testing and/or electronic calibration conducted annualy. Less than 25% of tested meters are found outside of +/- 6% accuracy.	Conditions between 6 and 8	100% of exported water sources are metered, meter accuracy testing and electronic calibration of related instrumentation is conducted annually, less than 10% of meters are found outside of +/- 6% accuracy	Conditions between 8 and 10	100% of exported water sources are metered, meter accuracy testing and electronic calibration of related instrumentation is conducted semi- annually for all meter installations, with less than 10% of accuracy tests found outside of +/- 3% accuracy.
Improvements to attain higher data grading for "Water Exported Volume" component: (Note: usually, if the water utility being audited sells (Exports) water to a neighboring purchasing Utility, it is the responsibility of the utility exporting the water to maintain the metering installation measuring the Exported volume. The utility exporting the water should ensure that adequate meter upkeep takes place and an accurate measure of the Water Exported volume is quantified.)		to qualify for 2: Review bulk water sales agreements with purchasing utilities; confirm requirements for use & upkeep of accurate metering, Identify needs to instal new, or replace defective meters as needed.	<u>To qualify for 4:</u> Locate all exported water sources o launch meter accuracy testing for er to install meters on unmetered interconnections and replace obsol	n maps and in field, disting meters, begin exported water atel/defective meters	<u>to qualify for 6</u> Formalize annual meter accuracy to water meters. Continue installation o exported water interconnections obsolete/defective m	isting for all exported of meters on unmetered and replacement of neters.	to qualify for 8: Complete project to install new, or repla on all exported water interconnection meter accuracy testing for all expor Repair or replace meters outside of	ace defective, meters s. Maintain annual ted water meters. +/- 6% accuracy.	to qualify for 10 Maintain annual meter accuracy testin or replace meters outside of +/-3% new meter technology, pilot one or m innovative meters in attempt to impr	; g for all meters. Repair accuracy. Investigate ore replacements with ove meter accuracy.	to maintain 10: Standardize meter accuracy test frequency to semi-annual, or more frequent, for al meters. Repair or replace meters outside of 4-3% accuracy. Continually investigate/pilot improving metering technology.

Grading >>>	n/a	1	2	3	4	5	6	7	8	9	10
Water exported master meter and supply error adjustment:	Select n/a only if the water utility fails to have meters on its exported supply interconnections.	Inventory information on exported meters and paper records of measured volumes exist but are incomplete and/or in a very crude condition; data error cannot be determined. Written agreement(s) with the utility purchasing the water are missing or written in vague language concerning meter management and testing.	No automatic datalogging of exported supply volumes; daily readings are scribed on paper records without any accountability controls to confirm data accuracy and the absence of errors and data gaps in recorded volumes. Written agreement requires meter accuracy testing but is vague on the details of how and who conducts the testing.	Conditions between 2 and 4	Exported metered flow data is logged automatically in electronic format and reviewed at least on a monthly basis, with necessary corrections implemented. Meter data is adjusted by the utility selling (exporting) the water when gross data errors are detected. A coherent data trail exists for this process to protect both the utility exporting the water and the purchasing Utility. Written agreement exists and clearly states requirements and roles for meter accuracy testing and data management.	Conditions between 4 and 6	Hourly exported supply metered data is logged automatically & reviewed on at least a weekly basis by the utility selling the water. Data is adjusted to correct gross error when meter/instrumentation equipment malfunction is detected; and to correct for error found by meter accuracy testing. Any data gaps in the archived data are detected and corrected during the weekly review. A coheracit data trail exists for this process to protect both the selling (exporting) utility and the purchasing Utility.	Conditions between 6 and 8	Continuous exported supply metered flow data is logged automatically & reviewed each business day by the utility selling (exporting) the water. Data is adjusted to correct gross error from detected meter/instrumentation equipment maffunction and any error confirmed by meter accuracy testing. Any data errors/gaps are detected and corrected on a daily basis. A data trail exists for the process to protect both the selling (exporting) Utility and the purchasing Utility.	Conditions between 8 and 10	Computerized system (SCADA or similar) automatically records data which is reviewed each business day by the utility selling (exporting) the water. Tight accountability controls ensure that all error/data gaps that occur in the archived flow data are quickly detected and corrected. A reliable data trail exists and contract provisions for meter testing and data management are reviewed by the selling Utility and purchasing Utility at least once every five years.
Improvements to attain higher data grading for "Water exported master meter and supply error adjustment" component:		to qualify for 2: Develop a plan to restructure recordkeeping system to capture all flow data; set a procedure to review flow data on a daily basis to detect input errors. Obtain more reliable information about existing meters by conducting field inspections of meters and related instrumentation, and obtaining manufacturer literature. Review the written agreement between the utility selling (exporting) the water and the purchasing Utility.	to qualify for 4: Install automatic datalogging equip supply meters. Set a procedure to r monthy basis to detect gross anom Launch discussions with the purcha review terms of the written agreeme accuracy testing and data managem as necessary.	whent on exported eview this data on a lales and data gaps. sing utilities to jointly nats regarding meter nent; revise the terms	to qualify for 6 Refine computerized data collection hourly exported supply metered flow least on a weekly basis to defect sy and gaps. Make necessary corr errors on a weekly	2 and archive to include data that is reviewed at peolfic data anomalies ections to errors/data basis.	to qualify for 8; Ensure that all exported metered flow archived on at least an hourly basis, and errors/data gaps are corrected o	, data is collected and All data is reviewed aach business day.	to qualify for 11 Conduct accountability checks to co- metered flow data is reviewed and co- day by the utility selling the water. accuracy tests and data corrections sharing between the utility and th Establish a schedule for a regular rev contractual language in the written purchasing utilities; at least e	infirm that all exported profirm that all exported proceed each business Results of all meter should be available for p purchasing Utility. ew and updating of the purchasing Utility. agreements with the very five years.	to maintain 10: Monitor meter innovations for development of more accurate and less expensive flowmeters; work with the purchasing utilities to help identify meter replacement needs. Keep communication lines with the purchasing utilities open and maintain productive relations. Keep the written agreement current with clear and exploit language that meets the ongoing needs of all parties.
					AUTHORIZED CO	ONSUMPTION					
Billed metered:	n/a (not applicable). Select n/a only if the entire customer population is not metered and is billed for water service on a flat or fixed rate basis. In such a case the volume entered must be zero.	Less than 50% of customers with volume-based billings from meter readings; flat or fixed rate billing exists for the majority of the customer population	At least 50% of customers with volume-based billing from meter reads; flat rate billing for others. Manual meter reading is conducted, with less than 50% meter read success rate, remainding accounts' consumption is estimated. Limited meter records, no regular meter testing or replacement. Billing data maintained on paper records, with no auditing.	Conditions between 2 and 4	At least 75% of customers with volume-based, billing from meter reads; flat or fixed rate billing for remaining accounts. Manual meter reading is conducted with at least 50% meter read success rate; consumption for accounts with fallee reads is estimated. Purchase records verify age of customer meters; only very limited meter accuracy testing is conducted. Customer meters are replaced only upon complete failure. Computerized billing records exist, but only sporadic internal auditing conducted.	Conditions between 4 and 6	At least 90% of customers with volume-based billing from meter reads; consumption for remaining accounts is estimated. Manual customer meter reading gives at least 80% customer meter reading success rate; consumption for accounts with failed reads is estimated. Good customer meter records exist, but only limited meter accuracy testing is conducted. Regular replacement is conducted for the oldest meters. Computerized billing records exist with annual auditing of summary statistics conducting by utility personnel.	Conditions between 6 and 8	At least 97% of customers exist with volume-based billing from meter reads. At least 90% customer meter reading success rate: grd least 80% meter Reading (AMR) or Advanced Meter Reading (AMR) or Advanced Meter Reading (AMR) or Advanced Metering Infrastructure (AMI) in one or more pilot areas. Good customer meter records. Regular meter accuracy testing guides replacement of statistically significant number of meters each year. Routine auditing of computerized billing records for global and detailed statistics occure, and is verified by third party at least once every five years.	Conditions between 8 and 10	At least 99% of customers exist with volume-based billing from meter reads. At least 95% customer meter reading success rate; <u>or</u> minimum 80% meter reading success rate, with Automatic Metering Infrastructure (AMI) trials underway. Statistically significant customer meter testing and replacement program in place on a confunous basis. Computerized billing field investigation of representative sample of accounts undertaken annually by utility personnet. Audit is conducted by third party auditors at least once every three years.
Improvements to attain higher data grading for "Billed Metered Consumption" component:	If n/a is selected because the customer meter population is unmetered, consider establishing a new policy to meter the customer population and employ water rates based upon metered volumes.	to qualify for 2: Conduct investigations or trials of customer meters to select appropriate meter models. Budget funding for meter installations. Investigate volume based water rate structures.	to qualify for 4: Purchase and install meters on un Implement policies to improve mete Catalog meter information during r identify age/model of existing mete number of meters for accuracy. In billing system.	metered accounts. r reading success. neter read visits to rs. Test a minimal stall computerized	to qualify for 6 Purchase and install meters on u Eliminate fital fee billing and establish structure based upon measured corr achieve verifable success in rem reading barriers. Expand meter acc regular meter replacement program annual auditing of global billing statis	innetered accounts. nappropriate water rate sumption. Continue to oving manual meter uracy testing. Launch , Launch a program of tics by utility personnel.	to qualify for 8: Purchase and install meters on unm customer meter reading success rat assess cost-effectiveness of Autom (AMR) or Advanced Metering Infrastr for portion or entire system; or otherw improvements in manual meter read 97% or higher. Refine meter accura Set meter replacement goals based results. Implement annual auditing records by utility personnel and imp auditing at least once every	etered accounts. If is less than 97%, atic Meter Reading ucture (ANI) system ise achieve ongoing ing success rate to cy testing program. yon accuracy test of detailed billing alement third party five years.	to qualify for 11 Purchase and install meters on umme Automatic Meter Reading (AMR) o Infrastructure (AMI) system trials if n success rate of at least 99% is not as program. Continue emter life conduct planning and budgeting replacement based upon meter life cumulative flow target. Continue ann auditing by utility personnel and condu least once every three	ered accounts. Launch radvanced Metering nanual meter reading nieved within a five-year yot testing program. for large scale meter cycle analysis using ual detailed billing data tot third party auditing at a years.	to maintain 10: Continue annual internal billing data auditing, and third party auditing at least every three years. Continue customer meter accuracy testing to ensure that accurate customer meter readings are obtained and entered as the basis for volume based billing. Stay abreast of improvements in Automatic Meter (and frastructure (AWI) and Avtenrig Infrastructure (AWI) and buildget for justified upgrades in metering, meter reading and billing data management to maintain very high accuracy in customer metering and billing.

Grading >>>	n/a	1	2	3	4	5	6	7	8	9	10
Billed unmetered:	Select n/a if it is the policy of the water utility to meter all customer connections and it has been confirmed by detailed auditing that all customers do indeed have a water meter; i.e. no intentionally unmetered accounts exist	Water utility policy does <u>not</u> require customer metering; flat or fixed fee billing is employed. No data is collected on customer consumption. The only estimates of customer population consumption available are derived from data estimation methods using average future count multiplied by number of connections, or similar approach.	Water utility policy does <u>not</u> require customer metering; flat or fixed fee billing is employed. Some metered accounts exist in parts of the system (pilot areas or District Metered Areas) with consumption read periodicality or recorded on portable dataloggers over one, three, or seven day periods. Data from these sample meters are used to infer consumption for the total customer population. Site specific estimation methods are used for unusual buildings/water uses.	Conditions between 2 and 4	Water utility policy <u>does</u> require metering and volume based billing in general. However, al beral amount of exemptions and a lack of clearly written and communicated procedures result in up to 20% of billed accounts believed to be unmetered by exemption, or the water utility is in transition to becoming fully metered, and a large number of customers remain unmetered. A rough estimate of the annual consumption for all unmetered accounts is included in the annual vater audit, with no inspection of individual unmetered accounts.	Conditions between 4 and 6	Water utility policy <u>does</u> require metering and volume based billing but established exemptions exist for a portion of accounts such as municipa buildings. As many as 15% of billed accounts are unmetered due to this exemption or meter installation difficulties. Only a group estimate of annual consumption for all unmetered accounts is included in the annual water audit, with no inspection of individual unmetered accounts.	Conditions between 6 and 8	Water utility policy <u>does</u> require metering and volume based billing for all custome accounts. However, less than 5% of billed accounts remain unmetered because meter installation is hindered by unusual circumstances. The goal is to minimize the number of unmetered accounts. Reliable estimates of consumption are obtained for these unmetered accounts via site specific estimation methods.	Conditions between 8 and 10	Water utility policy <u>does</u> require metering and volume based billing for all customer accounts. Less than 2% of billed accounts are unmetered and exist because meter installation is hindered by unusual circumstances. The goal exists to minimize the number of unmetered accounts to the extent that is economical. Reliable estimates accounts via site specific estimation methods.
Improvements to attain higher data grading for "Billed Unmetered Consumption" component:		to qualify for 2: Conduct research and evaluate costbenefit of a new water utility policy to require metering of the customer population; thereby greatly reducing or eliminating unnetered accounts. Conduct pilot metering project by instaling water meters in small sample of customer accounts and periodically reading the meters or datalogging the water consumption over one, three, or seven day periods.	to qualify for 4: Implement a new water utility policy metering. Launch or expand plot include several different meter type data for economic assessment of options. Assess sites with access- means to obtain water consumptic customer meter instal	requiring customer metering sludy to full scale metering difficulties to deministration on volumes. Begin lation.	to qualify for 6 Refine policy and procedures to imp participation for all but solidly exer staff resources to review billing rec unmetered properties. Specify mete requirements to install sufficient mete the number of unmetere	; ove customer metering pit accounts. Assign ords to identify errant ring needs and funding rs to significant reduce d accounts	to qualify for 8: Push to install customer meters or Refine metering policy and procedur accounts, including municipal propert meters. Pina special efforts to addre accounts. Implement procedures consumption estimate for the remain accounts awaiting meter in	a full scale basis, es to ensure that all es, are designated for ss "hard-to-access" to obtain a reliable ing few unmetered stallation.	to qualify for 11 Continue customer meter installation r area, with a goal to minimize unmeter the effort to investigate accounts with devise means to install water meters water consumpti	2: throughout the service red accounts. Sustain access diffuculties, and or otherwise measure on.	to maintain 10: Continue to refine estimation methods for unmetered consumption and explore means to establish metering, for as many billed remaining unmetered accounts as is economically feasible.
Unbilled metered:	select n/a if all billing- exempt consumption is unmetered.	Billing practices exempt certain accounts, such as municipal buildings, but written policies do not unsvilable, a reliable count of unavailable. Meter upkeep and meter reading on these accounts is rare and not considered a priority. Due to poor recordikeeping and lack of auditing, water consumption for all such accounts is purely guesstimated.	Billing practices exempt certain accounts, such as municipal buildings, but only scattered, dated written directives exist to justify this practice. A reliable count of unbilled metered accounts is unavailable. Sporadio meter replacement and meter reading occurs on an as- needed basis. The total annual water consumption for all unbilled, metered accounts is estimated based upon approximating the number of accounts and assigning consumption from actively billed accounts of same meter size.	Conditions between 2 and 4	Dated written procedures permit billing exemption for specific accounts, such as municipal properties, but are unclear regarding certain other types of accounts. Meter reading is given low priority and its sporadic. Consumption is quantitied from meter readings where available. The total number of unbilled, unmetered accounts must be estimated along with consumption volumes.	Conditions between 4 and 6	Written policies regarding billing exemptions exist but adherence in practice is questionable. Metering and meter reading for municipal buildings is reliable but sporadic for other unbilled metered accounts is conducted. Water consumption is quantified directly from meter readings where available, but the majority of the consumption is estimated.	Conditions between 6 and 8	Written policy identifies the types of accounts granted a biling exemption. Customer meter management and meter reading are considered secondary priorities, but meter reading is conducted at least annualju to obtain consumption volumes for the annual water audit. High level auditing of biling records ensures tha a reliable census of such accounts exists.	Conditions between 8 and 10	Clearly written policy identifies the types of accounts given a billing exemption, with emphasis on keeping such accounts to a minimum. Customer meter management and meter reading for these accounts is given proper priority and is reliably conducted. Regular auditing confirms this. Total water consumption for these accounts is taken from reliable readings from accurate meters.
Improvements to attain higher data grading for "Unbilled Metered Consumption" component:		to qualify for 2: Reassess the water utility's policy allowing certain accounts to be granted a billing exemption. Draft an outline of a new written policy for billing exemptions, with clear justification as to why any accounts should be exempt from billing, and with the intention to keep the number of such accounts to a minimum.	to qualify for 4: Review historic written directives an allowing certain accounts to be billing outline of a written policy for billing criteria that grants an exemption, wi this number of accounts to a min increasing the priority of reading r accounts at least ann	Id policy documents ng-exempt. Draft an exemptions, identify ith a goal of keeping imum. Consider meters on unbilled nually.	to qualify for 6 Draft a new written policy regardir based upon consensus criteria alio Assign resources to audit meter rect to obtain census of unbilled metere- include a greater number of these m routes for regular mete	; g billing exemptions wing this occurrence. rofs and billing records 1 accounts. Gradually letered accounts to the reading.	to qualify for 8: Communicate billing exemption po organization and implement procedur account management. Conduct insy confirmed in unbilled metered stat accurate meteres exist and are schedu readings. Gradually increase the r metered accounts that are include reading routes.	: icy throughout the as that ensure proper bections of accounts us and verify that led for routine meter umber of unbilled d in regular meter	to qualify for 11 Ensure that meter management (m meter replacement) and meter readi accounts are accorded the same pri Estabilish ongoing annual audting j water consumption is reliably collect annual water audit p	): eter accuracy testing, ng activities for unbilled only as billed accounts. rocess to ensure that ed and provided to the rocess.	to maintain 10: Reassess the utility's philosophy in allowing any water uses to go "unbiled". It is possible to meter and bil all accounts, even if the fee charged for water consumption is discounted or waived. Metering and billing all accounts ensures that water consumption is tracked and water waste from plumbing leaks is detected and minimized.
Unbilled unmetered:		Extent of unbilled, unmetered consumption is unknown due to unclear policies and poor recordkeeping. Total consumption is quantified based upon a purely subjective estimate.	Clear extent of unbilled, unmetered consumption is unknown, but a number of events are randomly documented each year, confirming existence of such consumption, but without sufficient documentation to quantify an accurate estimate of the annual volume consumed.	Conditions between 2 and 4	Extent of unbilled, unmetered consumption is partially known, and procedures exist to document certain events such as miscelianeous fire hydrant uses. Formulae is used to quantify the consumption from such events (time running multiplied by typical flowrate, multiplied by number of events).	Default value of 1.25% of system input volume is employed	Coherent policies exist for some forms of unbilled, unmetered consumption but others await closer evaluation. Reasonable recordkeeping for the managed uses exists and allows for annual volumes to be quantified by inference, but unsupervised uses are guesstimated.	Conditions between 6 and 8	Clear policies and good record/keeping exist for some uses (ex: water used in periodic testing of unmetered fire connections), but other uses (ex: miscellaneous uses of fire hydrants) have limited oversight. Total consumption is a mix of well quantified use such as from formuliae (time running multibiled by typical flow, multiplied by number of events) or temporary meters, and relatively subjective estimates of less regulated use.	Conditions between 8 and 10	Clear policies exist to identify permitted use of water in unbilled, unmetered fashion, with the intention of minimizing this type of consumption. Good records document each occurrence and consumption is quantified va formulae (time running multiplied by typical flow, multiplied by number of events) or use of temporary meters.

Grading >>>	n/a	1	2	3	4	5	6	7	8	9	10
Improvements to attain higher data grading for "Unbilled Unmetered Consumption" component:		to qualify for 5: Utilize the accepted default value of 1.25% of the volume of water supplied as an expedient means to gain a reasonable quantification of this use. to qualify for 2: Establish a policy regarding what water uses should be allowed to remain as unbilled and unmetered. Consider tracking a small sample of one such use (ex: fire hydrant flushings).	Interpretation       Inter		<u>to qualify for 10</u> Refine written procedures to ensure t ummetered water are overseen by a process managed by water utility pers to determine if some of these uses converted to billed and/or m	: hat all uses of unbilled, structured permitting nonel. Reassess policy have value in being elered status.	to maintain 10: Continue to refine policy and procedures with intention of reducing the number of allowable uses of water in unbilled fashion. Any uses that can feasibly become billed and metered should be converted eventually.				
					APPARENT	LOSSES					
Unauthorized consumption:		Extent of unauthorized consumption is unknown due to unclear policies and poor recordkeeping. Total unauthorized consumption is guesstimated.	Unauthorized consumption is a known occurrence, but its extent is a mystery. There are no requirements to document observed events, but periodic field reports capture some of these occurrences. Total unauthorized consumption is approximated from this limited data.	conditions between 2 and 4	Procedures exist to document some unauthorized consumption such as observed unauthorized fire hydrant openings. Use formulae to quantify this consumption (time running multiplied typical flowrate, multiplied by number of events).	Default value of 0.25% of volume of water supplied is employed	Coherent policies exist for some forms of unauthorized consumption (more than simply fire hydrant misuse) but others await closer evaluation. Reasonable surveillance and recordkeeping exist for occurrences that fail under the policy. Volumes quantified by inference from these records.	Conditions between 6 and 8	Clear policies and good auditable recordkeeping exist for certain events (ex: tampering with water meters), illegal bypasses of customer meters); but other occurrences have limited oversight. Total consumption is a combination of volumes from formulae (time x typical flow) and subjective estimates of unconfirmed consumption.	Conditions between 8 and 10	Clear policies exist to identify all known unauthorized uses of water. Staff and procedures exist to provide enforcement of policies and detect violations. Each occurrence is recorded and quantified via formulae (estimated time running multiplied by typical flow) or similar methods. All records and calculations should exist in a form that can be audited by a third party.
Improvements to attain higher data grading for "Unauthorized Consumption" component:		to qualify for 5: Use accepted default of 0.25% of volume of water suppled. to qualify for 2: Review utility policy regarding what water uses are considered unauthorized, and consider tracking a small sample of one such occurrence (ex: unauthorized fire hydrant openings)	to qualify for 5: Use accepted default of 0.25% of s to qualify for 4: Review utility polcy regarding wh considered unauthortzed, and cons sample of one such occurrence (e hydrant opening:	ystem input volume at water uses are ider tracking a small x: unauthorized fire )	to qualify for 5: Utilize accepted default value of 0.25% of volume of water supplied as an expedient means to gain a reasonable quantification of all such use. This is particularly appropriate for water utilities who are in the early stages of the water auditing process.	to qualify for 6 or greater: Finalize policy updates to clearly identify the types of water consumption that are authorized from those usages that fall outside of this policy and are, therefore, unauthorized. Begin to conduct regular field checks. Proceed if the top- down audit already exists and/or a great volume of such use is suspected.	to quality for 8: Assess water utility policies to ensu occurrences of unauthorized consum and that appropriate penalities are p written procedures for detection and various occurrences of unauthorized o are uncovered.	ure that all known ption are outlawed, rescribed. Create I documentation of consumption as they	to qualify for 10 Refine written procedures and assign occurrences of unauthorized consu- locking devices, monitors and other te detect and thwart unauthorize	: staff to seek out likely mption. Explore new chnologies designed to d consumption.	to maintain 10: Continue to refine policy and procedures to eliminate any loopholes that allow or tacitly encourage unauthorized consumption. Continue to be vigilant in detection, documentation and enforcement efforts.
Customer metering inaccuracies:	select n/a only if the entire customer population is unmetered. In such a case the volume entered must be zero.	Customer meters exist, but with unorganized paper records on meters, no meter accuracy testing or meter replacement program for any size of retail meter. Metering workflow is driven chaotically with no proactive management. Loss volume due to aggregate meter inaccuracy is guesstimated.	Poor recordkeeping and meter oversight is recognized by water utility management who has allotted staff and funding resources to organize improved recordkeeping and start meter accuracy testing. Existing paper records gathered and organized to provide cursory disposition of meter population. Customer meters are tested for accuracy only upon customer request.	Conditions between 2 and 4	Reliable recordkeeping exists; meter information is improving as meters are replaced. Meter accuracy testing is conducted annually for a small number of meters (more than just customer requests, but less than 1% of inventory). A limited number of the oldest meters are replaced each year. Inaccuracy volume is largely an estimate, but refined based upon limited testing data.	Conditions between 4 and 6	A reliable electronic record/keeping system for meters exists. The meter population includes a mix of new high performing meters and dated meters with suspect accuracy. Routine, but limited, meter accuracy testing and meter replacement occur. Inaccuracy volume is quantified using a mix of reliable and less certain data.	Conditions between 6 and 8	Ongoing meter replacement and accuracy testing result in highly accurate customer meter population. Testing is conducted on samples of meters of varying age and accumulated volume of throughput to determine optimum replacement time for various types of meters.	Ongoing meter replacement and accuracy testing result in highly accurate ustomer meter population. Statistically significant number of meters are conducted on samples of meters of varying age and the accumulated volume of throughput to determine optimum replacement time for these meters.	Good records of all active customer meters exist and include as a minimum: meter number, account number/location, type, size and manufacturer. Ongoing meter replacement occurs according to a targeted and justified basis. Regular meter accuracy testing gives a reliable measure of composite inaccuracy volume for the customer meter population. New metering technology is embraced to keep overall accuracy a third party knowledgeable in the M36 methodology.

Grading >>>	n/a	1	2	3	4	5	6	7	8	9	10
Improvements to attain higher data grading for "Customer meter inaccuracy volume" component:	If n/a is selected because the customer meter population is unmetered, consider establishing a new policy to meter the customer population and upon metered volumes.	to qualify for 2: Gather available meter purchase records. Conduct testing on a small number of meters believed to be the most inaccurate. Review staffing needs of the metering group and budget for necessary resources to better organize meter management.	to qualify for 4: Implement a reliable record keeping meter histories, preferably using e typically linked to, or part of, the Cuu or Customer Information System accuracy testing to a larger gr	system for customer lectronic methods tomer Billing System a. Expand meter oup of meters.	<u>to quality for 6</u> Standardize the procedures for mete an electronic information system accuracy testing and meter replacer results.	er recordkeeping within . Accelerate meter nents guided by testing	to qualify for 8: Expand annua meter accuracy les statistically significant number of me Expand meter replacement program I significant number of poor performin	sting to evaluate a ter makes/models. or optace statistically g meters each year.	to qualify for 9: Continue efforts to manage meter population with reliable recordkeeping. Test a statistically significant number of meters each year and analyze test results in an ongoing manner to serve as a basis for a target meter replacement strategy based upon accumulated volume throughput.	to qualify for 10: Continue efforts to manage meter population with reliable recordkeeping, meter testing and replacement. Evaluate new meter or more types in 5-10 customer accounts each year in order to pilot improving metering technology.	to maintain 10: Increase the number of meters tested and replaced as justified by meter accuracy test data. Continually monitor development of new metering Infrastructure (AMI) to grasp opportunities for greater accuracy in metering of water flow and management of customer consumption data.
Systematic Data Handling Errors:	Note: all water utilities incur some amount of this error. Even in water utilities with unmetered customer populations and fixed rate billing, errors occur in annual billing tabulations. Enter a positive value for the volume and select a grading.	Policies and procedures for activation of new customer water billing accounts are vague and lack accountability. Billing data is maintained on paper records which are not well organized. No auditing is conducted to confirm billing data handling efficiency. An unknown number of customers escape routine billing due to lack of billing process oversight.	Policy and procedures for activation of new customer accounts and oversight of billing records exist but need refinement. Billing data is maintained on paper records or insufficiently capable electronic database. Only periodic unstructured audiling work is conducted to confirm billing data handing efficiency. The volume of unbilled water due to billing lapses is a guess.	Conditions between 2 and 4	Policy and procedures for new account activation and oversight of billing operations exist but needs refinement. Computerized billing system exists, but is dated or lacks needed functionality. Periodic. limited internal audits conducted and confirm with approximate accuracy the consumption volumes lost to billing lapses.	Conditions between 4 and 6	Policy and procedures for new account activation and oversight of billing operations is adequate and reviewed periodically. Computerized billing system is in use with basic reporting available. Any effect of billing adjustments on measured consumption volumes is well understood. Internal checks of billing data error conducted annually. Reasonably accurate quantification of consumption volume lost to billing lapses is obtained.	Conditions between 6 and 8	New account activation and billing operations policy and procedures are reviewed at least biannually. Computerized billing system includes an array of reports to confirm billing data and system functionality. Checks are conducted routinely to flag and explain zero consumption flag and explain zero consumption accounts. Annual internal checks conducted with third party audit conducted at least once every five years. Accountability checks flag billing tapess is well quantified and reducing year-by-year.	Conditions between 8 and 10	Sound written policy and procedures exist for new account activation and oversight of customer billing operations. Robust computerized billing system gives high functionality and reporting capabilities which are utilized, analyzed and the results reported each billing cycle. Assessment of policy and data handling errors are conducted internally and audited by third party at least once every three years, ensuring consumption lost to billing lapses is minimized and detected as it occurs.
Improvements to attain higher data grading for "Systematic Data Handling Error volume" component:		to qualify for 2: Draft written policy and p	to qualify for 4: Finalize written policy and proceed new billing acocunts and overal management. Implement a comput system. Conduct initial audit of billin this process.	ires for activation of billing operations rized customer billing rg records as part of	to qualify for 6 Refine new account activation ar procedures and ensure consistenc regarding billing, and minimize or billings. Upgrade or replace custo needed functionality - ensure that bi corrupt the value of consumption v internal annual audit	d billing operations y with the utility policy poptruinity for missed mer billing system for illing adjustments don't olumes. Procedurize process.	to qualify for 8: Formalize regular review of new accou and general billing practices. Enhanc of computerized billing system. Form process to reveal scope of data han periodic third party audit to occur at I years.	int activation process e reporting capability alize regular auditing dling error. Plan for east once every five	to qualify for 10 Close policy/procedure loopholes th accounts to go unbilled, or data ha Ensure that billing system reports ar reported every billing cycle. Ensure party audits are conducted at least o	t allow some customer alling errors to exist. utilized, analyzed and that internal and third nce every three years.	to maintain 10: Stay abreast of customer information management developments and innovations. Monitor developments of Advanced Metering Infrastructure (AMI) and integrate technology to ensure that customer endpoint information is well- monitored and errors/lapses are at an economic minimum.
	-				SYSTEM	DATA	-				
Length of mains:		Poorty assembled and maintained paper as-built records of existing water main installations makes accurate determination of system pipe length impossible. Length of mains is guesstimated.	Paper records in poor or uncertain condition (no annual tracking of installations & abandonments). Poor procedures to ensure that new water mains installed by developers are accurately documented.	Conditions between 2 and 4	Sound written policy and procedures exist for documenting new water main installatons, but gaps in management result in a uncertain degree of error in tabulation of mains length.	Conditions between 4 and 6	Sound written policy and procedures exist for permitting and commissioning new water mains. Highly accurate paper records with regular field validation; relectronic records and asset management system in good condition. Includes system backup.	Conditions between 6 and 8	Sound written policy and procedures exist for permitting and commissioning new water mains. Electronic recordkeping such as a Geographical Information System (GIS) and asset management system are used to store and manage data.	Conditions between 8 and 10	Sound written policy exists for managing water mains extensions and replacements. Geographic Information System (GIS) data and asset management database agree and random field validation proves truth of databases. Records of annual field validation should be available for review.
Improvements to attain higher data grading for "Length of Water Mains" component:		to qualify for 2: Assign personnel to inventory current as-built records and compare with customer billing system records and highway plans in order to verify poorly documented pipelines. Assemble policy documents regarding permitting and documentation of water main installations by the utility and building developers; identify gaps in procedures that result in poor documentation of new water main installations.	to qualify for 4: Complete inventory of paper reco installations for several years prior to policy and procedures for com documenting new water ma	ords of water main o audit year. Review missioning and ni nistallation.	to qualify for 6 Finalize updates/mprovements procedures for permitting/comm installations. Confirm inventory of prior to audit year, correct any e	: o written policy and issioning new main records for five years rrors or omissions.	<u>to qualify for 8</u> : Launch random field checks of limited Convert to electronic database suc Information System (GIS) with backup written policy and proce	number of locations. h as a Geographic as justified. Develop dures.	<u>to qualify for 10</u> Link Geographic Information Syst management databases, conduct fi Record field verification informatic	; m (GIS) and asset bid verification of data. in at least annually.	to maintain 10: Continue with standardization and random field validation to improve the completeness and accuracy of the system.

Grading >>>	n/a	1	2	3	4	5	6	7	8	9	10
Number of active AND inactive service connections:		Vague permitting (of new service connections) policy and poor paper recordkeeping of customer connections/billings result in suspect determination of the number of service connections, which may be 10-15% in error from actual count.	General permitting policy exists but paper records, procedural gaps, and weak oversight result in questionable total for number of connections, which may vary 5-10% of actual count.	Conditions between 2 and 4	Written account activation policy and procedures exist, but with some gaps in performance and oversight. Computerized information management system is being brought online to replace dated paper recordkeeping system. Reasonably accurate tracking of service connection installations & abandonments; but count can be up to 5% in error from actual total.	Conditions between 4 and 6	Written new account activation and overall billing policies and procedures are adequate and reviewed periodically. Computerized information management system is in use with annual installations & abandonments totaled. Very limited field verifications and audits. Error in count of number of service connections is believed to be no more than 3%.	Conditions between 6 and 8	Policies and procedures for new account activation and overall billing operations are written, well-structured and reviewed at least biannually. Well managed computerized information management system exists and routine, periodic field checks and internal system audits are conducted. Counts of connections are no more than 2% in error.	Conditions between 8 and 10	Sound written policy and well managed and audited procedures ensure relable management of service connection population. Computerized information management system, Customer Billing System (GIS) information agree, field validation proves truth of databases. Count of connections recorded as being in error is less than 1% of the entire population.
Improvements to attain higher data grading for "Number of Active and Inactive Service Connections" component:	Note: The number of Service Connections does <u>not</u> include fire hydrant leads/lines connecting the hydrant to the water main	to qualify for 2: Draft new policy and procedures for new account activation and overall billing operations. Research and collect paper records of installations & abandonments for several years prior to audit year.	to qualify for 4: Refine policy and procedures for ne- and overall billing operations. Rese recordkeeping system (Customer Ini Customer Billing System) to impro- format for service conne-	w account activation earch computerized formation System or we documentation ections.	to qualify for 6 Refine procedures to ensure consist activation and overal billing policy to connections or decommission e Improve process to include at total prior to audit ye	: ency with new account establish new service disting connections. for at least five years ar.	to qualify for 8: Formalize regular review of new acc overalt billing operations policies and p random field checks of limited num Develop reports and auditing m computerized information manag	ount activation and procedures. Launch nber of locations. echanisms for gement system.	<u>to qualify for 10</u> Close any procedural loopholes that i undocumented. Link computerized in system with Geographic Informatio formalize field inspection and inform processes. Documentation of new service connections encounters sever balances.	: allow installations to go formation management in System (GIS) and ation system auditing or decommissioned ral levels of checks and	to maintain 10: Continue with standardization and random field validation to improve knowledge of system.
	Note: if customer water	Gradings 1-9 apply if customer pro these cases the average distance b	operties are unmetered, if customer me etween the curb stop or boundary sep	eters exist and are loc parating utility/custom	ated inside the customer building prer er responsibility for service connection means to quantify this value. (See	nises, or if the water util piping, and the typical the "Service Connection	ity owns and is responsible for the entire first point of use (ex: faucet) or the custo on Diagram" worksheet)	e service connection   omer meter must be o	piping from the water main to the custor juantified. Gradings of 1-9 are used to g	ner building. In any of grade the validity of the	Either of two conditions can be met for
Average length of customer service line:	meters are located outside of the customer building next to the curb stop or boundary separating utili/vustomer responsibility, then the auditor should answer "Yes" to the question on the Reporting Worksheet asking about this. If the grading description listed under the Grading of U(a) will be followed, with a value of zero automatically entered at a Grading of 10. See the Service Connection Diagram worksheet for a visual presentation of this distance.	Vague policy exists to define the delineation of water utility ownership and customer ownership of the service connection piping. Curb stops are perceived as the breakpoint but these have not been well-maintained or documented. Most are butried or obscured. Their location varies widely from site-to- site, and estimating this distance is arbitrary due to the unknown location of many curb stops.	Policy requires that the curb stop serves as the delineation point between water utility ownership and customer ownership of the service connection piping. The piping from the vater main to the curb stop to the the property of the water utility; and the piping from the curb stop to the customer building is owneed by the customer building is owneed by the customer. Curb stop locations are not well documented and the average distance is based upon a limited number of locations measured in the field.	Conditions between 2 and 4	Good policy requires that the curb stop serves as the delineation point between water utility ownership and customer ownership of the service connection piping. Curb stops are generally instaled as needed and are reasonably documented. Their location varies widely from site-to- site, and an estimate of this distance is hindered by the availability of paper records of limited accuracy.	Conditions between 4 and 6	Clear written policy exists to define utility/customer responsibility for service connection piping. Accurate, well-maintained paper or basic electronic recordkeeping system exists. Periodic field checks confirm piping lengths for a sample of customer properties.	Conditions between 6 and 8	Clearly worded policy standardizes the location of curb stops and meters, which are inspected upon installation. Accurate and well maintained electronic records exist with periodic field checks to confirm locations of service lines, curb stops and customer meter pits. An accurate number of customer properties from the customer billing system allows for reliable averaging of this length.	Conditions between 8 and 10	a <u>crading of 10</u> : a) Customer water meters exist outside of customer buildings next to the curb stop or boundary separating utility/customer responsibility for service to the quession on the Reporting Working asking about this condition. A value of zero and a Grading of 10 are automatically entered in the Reporting Worksheet. b). Meters exist inside customer buildings, or properties are unmetered. In either case, answer "No" to the Reporting Worksheet question on meter location, and enter a distance determined by the auditor. For a Grading of 10 this value must be a very reliable number from a Geographic information System (GIS) and confirmed by a statistically valid number of field checks.
Improvements to attain higher data grading for "Average Length of Customer Service Line" component:		to qualify for 2: Research and collect paper records of service line installations. Inspect several sites in the field using pipe locators to locate curb stops. Obtain the length of this small sample of connections in this manner.	to qualify for 4: Formalize and communicate po utility/customer responsibilities for s pipping. Assess accuracy of pape inspection of a small sample of servic pipe locators as needed. Resea migration to a computerized inform system to store service conn	i service connection records by field ce connections using rich the potential aution management lection data.	to qualify for 6 Establish coherent procedures to en stop, meter instalation and docur Gain consensus within the water util of a computerized information m.	; sure that policy for curb rentation is followed. ty for the establishment anagement system.	<u>to qualify for 8:</u> Implement an electronic means of rec via a customer information system, cu or Geographic Information System (G process to conduct field checks of a locations.	i ordkeeping, typically stomer billing system, IS). Standardize the limited number of	<u>to qualify for 10</u> Link custorner information manag Geographic Information System (GIS for field verification of	: ement system and ), standardize process f data.	to maintain 10; Continue with standardization and random field validation to improve knowledge of service connection configurations and customer meter locations.
Average operating pressure:		Available records are poorly assembled and maintained paper records of supply pump characteristics and water distribution system operating conditions. Average pressure is guesstimated based upon this information and ground devations from crude topographical maps. Widely varying distribution system pressures due to undulating terrain, high system head loss and weak/erratic pressure controls further compromise the validity of the average pressure calculation.	Limited telemetry monitoring of scattered pumping station and water storage tank sites provides some static pressure data, which is recorded in handwritten logbooks. Pressure data is gathered at individual sites only when low pressure complaints arise. Average pressure is determined by averaging relatively crude data, and is affected by significant variation in ground elevations, system head loss and gaps in pressure controls in the distribution system.	Conditions between 2 and 4	Effective pressure controls separate different pressure zones; moderate pressure variation across the system, occasional open boundary valves are discovered that breech pressure zones. Basic telemetry monitoring of the distribution system logs pressure data electronically. Pressure data gathered by gauges or dataloggers at fire hydrants or buildings when low pressure complaints arise, and during fire flow tests and system flushing. Reliable poographical data exists. Average pressure is calculated using this mix of data.	Conditions between 4 and 6	Reliable pressure controls separate distinct pressure zones; only very occasional open boundary valves are encountered that breach pressure zones. Wel-covered telementry monitoring of the distribution system (not just pumping at source treatment plants or wells) logs extensive pressure data electronically. Pressure gathered by gauges/datalogges at fire hydrants and buildings when low pressure complaints arise, and during fire flow tests and system flushing. Average pressure is determined by using this mix of reliable data.	Conditions between 6 and 8	Well-managed, discrete pressure zones exist with generally predictable pressure fluctuations. A current ful- scale SCADA System or similar realtime monitoring system exists to monitor the water distribution system and collect data, including real time pressure readings at representative sites across the system. The average system pressure is determined from reliable monitoring system data.	Conditions between 8 and 10	Well-managed pressure districts/zones, SCADA System and hydraulic model exist to give very precise pressure data across the water distribution system. Average system pressure is reliably calculated from extensive, reliable, and cross-checked data. Calculations are reported on an annual basis as a minimum.

Grading >>>	n/a	1	2	3	4	5	6	7	8	9	10
Improvements to attain higher data grading for "Average Operating Pressure" component:		to qualify for 2: Employ pressure gauging and/or datalogging equipment to obtain pressure measurements from fire hydrants. Locate accurate topographical maps of service area in order to confirm ground elevations. Research pump data sheets to find pump pressure/flow characteristics	to qualify for 4: Formalize a procedure to u gauging/datalogging equipment to g during various system events suc complaints, or operational testing. G and flow data at different flow regi pressure controls (pressure reduc valves, partially open boundary w property configure pressure zones. data from these efforts available to g average pressure	se pressure jather pressure data h as low pressure ather pump pressure mes. Identify faulty ing valves, altitude alves) and plan to Make all pressure enerate system-wide a.	to qualify for 6 Expand the use of pressure ga equipment to gather scattered p representative set of sites, based up areas. Utilize pump pressure and f supply head entering each press Correct any faulty pressure control valves, altitude valves, partialy ope ensure properly configured pressure pressure dataset from these activité wide average pres	ging/datalogging pressure data at a ion pressure zones or ow data to determine ure zone or district. s (pressure reducing n boundary valves) to zones. Use expanded s to generate system- sure.	to qualify for 8: Install a Supervisory Control and Data System, or similar realtime monitoring system parameters and control oper calibration schedule for instrument accuracy. Obtain accurate topograp pressure data gathered from field extensive, reliable data for press	Acquisition (SCADA) s system, to monitor ations. Set regular tion to insure data incial data and utilize surveys to provide ure averaging.	to qualify for 10 Annually, obtain a system-wide avera the hydraulic model of the distributio calibrated via field measurements in system and confirmed in comparisor data.	; ge pressure value from system that has been the water distribution s with SCADA System	to maintain 10: Continue to refine the hydraulic model of the distribution system and consider linking it with SCADA System for real- time pressure data calibration, and averaging.

Grading >>>	n/a	1	2	3	4	5	6	7	8	9	10
					COST D	ATA					
Total annual cost of operating water system:		Incomplete paper records and lack of financial accounting documentation on many operating functions makes calculation of water system operating costs a pure guesstimate	Reasonably maintained, but incomplete, paper or electronic accounting provides data to estimate the major portion of water system operating costs.	Conditions between 2 and 4	Electronic, industry-standard cost accounting system in place. However, gaps in data are known to exist, periodic internal reviews are conducted but not a structured financial audit.	Conditions between 4 and 6	Reliable electronic, industry-standard cost accounting system in place, with all pertinent water system operating costs tracked. Data audited periodically by utility personnel, but not a Certified Public Accountant (CPA).	Conditions between 6 and 8	Reliable electronic, industry-standard cost accounting system in place, with all pertinent water system operating costs tracked. Data audited at least annually by utility personnel, and at least once every three years by third- party CPA.	Conditions between 8 and 10	Reliable electronic, industry-standard cost accounting system in place, with all pertinent water system operating costs tracked. Data audited annually by utility personnel and annually also by third-party CPA.
Improvements to attain higher data grading for "Total Annual Cost of Operating the Water System" component:		to qualify for 2: Gather available records, institute new financial accounting procedures to regularly collect and audit basic cost data of most important operations functions.	<u>to qualify for 4:</u> Implement an electronic cost acc structured according to accounting utilities	counting system, standards for water	to qualify for 6: Establish process for periodic interna operating costs; identify cost dat procedures for tracking these o	I audit of water system a gaps and institute utstanding costs.	to qualify for 8: Standardize the process to conduct r on an annual basis. Arrange for CP records at least once every t	outine financial audit A audit of financial hree years.	to qualify for 10 Standardize the process to conduct audit by a CPA on an an	: a third-party financial nual basis.	to maintain 10: Maintain program, stay abreast of expenses subject to erratic cost changes and long-term cost trend, and budget/track costs proactively
Customer retail unit cost (applied to Apparent Losses):	Customer population unmetered, and/or only a fixed fee is charged for consumption.	Antiquated, cumbersome water rate structure is used, with periodic historic amendments that were poorly documented and implemented; resulting in classes of customers being billed inconsistent charges. The actual composite billing rate likely differs significantly from the published water rate structure, but a lack of auditing leaves the degree of error indeterminate.	Dated, cumbersome water rate structure, not always employed consistently in actual billing operations. The actual composite published water rate structure, and a reasonably accurate estimate of the degree of error is determined, allowing a composite billing rate to be quantified.	Conditions between 2 and 4	Straight-forward water rate structure in use, but not updated in several years. Billing operations reliably employ the rate structure. The composite billing rate is derived from a single customer class such as residential customer accounts, neglecting the effect of different rates from varying customer classes.	Conditions between 4 and 6	Clearly written, up-to-date water rate structure is in force and is applied reliably in billing operations. Composite customer rate is determined using a weighted average residential rate using volumes of water in each rate block.	Conditions between 6 and 8	Effective water rate structure is in force and is applied reliably in billing operations. Composite customer rate is determined using a weighted average composite consumption rate, which includes residential, commercial, industrial, institutional (CII), and any other distinct customer classes within the water rate structure.	Conditions between 8 and 10	Current, effective water rate structure is in force and applied reliably in billing operations. The rate structure and calculations of composite rate - which includes residential, commercial, industrial, institutional (CII), and other distinct customer classes - are reviewed by a third party knowledgeable in the M36 methodology at least once every five years.
Improvements to attain higher data grading for "Customer Retail Unit Cost" component:		to qualify for 2: Formalize the process to implement water rates, including a secure documentation procedure. Create a current, formal water rate document and gain approval from all stakeholders.	to qualify for 4: Review the water rate structure and needed. Assess billing operations the billing operations incorporate the es structure.	update/formalize as o ensure that actual tablished water rate	to qualify for 6: Evaluate volume of water used in each usage block by residential users. Multiply volumes by full rate structure.	Launch effort to fully meter the customer population and charge rates based upon water volumes	<u>to qualify for 8:</u> Evaluate volume of water used in eac classifications of users. Multiply vo structure.	h usage block by all lumes by full rate	<u>to qualify for 10</u> Conduct a periodic third-party audit usage block by all classifications of u by full rate structu	; of water used in each sers. Multiply volumes re.	to maintain 10: Keep water rate structure current in addressing the water utility's revenue needs. Update the calculation of the customer unit rate as new rate components, customer classes, or other components are modified.
Variable production cost (applied to Real Losses):	Note: if the water utility purchases/imports its entire water supply, then enter the unit purchase cost of the bulk water supply in the Reporting Worksheet with a grading of 10	Incomplete paper records and lack of documentation on primary operating functions (electric power and treatment costs most importantly) makes calculation of variable production costs a pure guesstimate	Reasonably maintained, but incomplete, paper or electronic accounting provides data to roughly estimate the basic operations costs (pumping power costs and treatment costs) and calculate a unit variable production cost.	Conditions between 2 and 4	Electronic, industry-standard cost accounting system in place. Electric power and treatment costs are reliably tracked and allow accurate weighted calculation of unit variable production costs based on these two inputs and water imported purchase costs (if applicable). All costs are audited internally on a periodic basis.	Conditions between 4 and 6	Reliable electronic, industry-standard cost accounting system in place, with all pertinent water system operating costs tracked. Pertinent additional costs beyond power, treatment and water imported purchase costs (if applicable) such as liability, residuals management, wear and tear on equipment, impending expansion of supply, are included in the unit variable production cost, as applicable. The data is audited at least annually by utility personnel.	Conditions between 6 and 8	Reliable electronic, industry-standard cost accounting system in place, with all pertinent primary and secondary variable production and water imported purchase (if applicable) costs tracked. The data is audited at least annually by utility personnel, and at least once every three years by a third-party knowledgeable in the M36 methodology.	Conditions between 8 and 10	Either of two conditions can be met to obtain a grading of 10: 1) Third party CPA audit of all pertinent primary and secondary variable production and water imported purchase (I applicable) costs on an annual basis. or: 2) Water supply is entirely purchased as bulk water imported, and the unit purchase cost - including <u>all</u> applicable marginal supply costs - serves as the variable production cost if <u>all</u> applicable marginal supply costs are not included in this figure, a grade of 10 should <u>not</u> be selected.
Improvements to attain higher data grading for "Variable Production Cost" component:		to <u>qualify for 2</u> : Gather available records, institute new procedures to regularly collect and audit basic cost data and most important operations functions.	<u>to qualify for 4</u> ; Implement an electronic cost acc structured according to accounting utilities	ounting system, standards for water	to qualify for 6: Formalize process for regular interna costs. Assess whether additional or management, equipment wear, imp expansion) should be included to representative variable pro	al audits of production ssts (liability, residuals ending infrastructure o calculate a more duction cost.	to qualify for 8: Formalize the accounting process to components (power, treatment) as v components (licibility, residuals manage to conduct audits by a knowledgable once every three year	include direct cost vell as indirect cost ement, etc.) Arrange third-party at least rs.	to qualify for 10 Standardize the process to conduct audit by a CPA on an an	a third-party financial nual basis.	to maintain 10: Maintain program, stay abreast of expenses subject to erratic cost changes and budget/track costs proactively



## AWWA Free Water Audit Software: Customer Service Line Diagrams

WAS v5.0

American Water Works Association. Copyright © 2014, All Rights Reserved.

# Average Length of Customer Service Line

The three figures shown on this worksheet display the assignment of the Average Length of Customer Service Line, Lp, for the three most common piping configurations.

### Figure 1 shows the

configuration of the water meter outside of the customer building next to the curb stop valve. In this configuration Lp = 0 since the distance between the curb stop and the customer metering point is essentially zero.

### Figure 2 shows the

configuration of the customer water meter located inside the customer building, where Lp is the distance from the curb stop to the water meter.

### Figure 3 shows the

configuration of an unmetered customer building , where Lp is the distance from the curb stop to the first point of customer water consumption, or, more simply, the building line.

In any water system the Lp will vary notably in a community of different structures, therefore the average Lp value is used and this should be approximated or calculated if a sample of service line measurements has been gathered.



Fauc

Typical house connection: Unmetered

Not to scale

Click for more information

Connection to Water Main

Figure 3

Cur

合	AWWA Free Water Audit Software: WAS v5.0 Definitions Copyright © 2014, All Rights Reserved.
Item Name	Description
Annarent	= unauthorized consumption + customer metering inaccuracies + systematic data handling errors Apparent Losses include all types of inaccuracies associated with customer metering (worn meters as well as improperly sized meters or wrong type of meter
Find	for the water usage profile) as well as systematic data handling errors (meter reading, billing, archiving and reporting), plus unauthorized consumption (theft or illegal use). NOTE: Over-estimation of Apparent Losses results in under-estimation of Real Losses. Under-estimation of Apparent Losses results in over-estimation of Real Losses.
	= billed water exported + billed metered + billed unmetered + unbilled metered + unbilled unmetered consumption
	The volume of metered and/or unmetered water taken by registered customers, the water utility's own uses, and uses of others who are implicitly or explicitly authorized to do so by the water utility; for residential, commercial, industrial and public-minded purposes.
AUTHORIZED	Typical retail customers' consumption is tabulated usually from established customer accounts as billed metered consumption, or - for unmetered customers - billed unmetered consumption. These types of consumption, along with billed water exported, provide revenue potential for the water utility. <b>Be certain to</b> <b>tabulate the water exported volume as a separate component and do not "double-count" it by including in the billed metered consumption</b> <b>component as well as the water exported component.</b>
Find	Unbilled authorized consumption occurs typically in non-account uses, including water for fire fighting and training, flushing of water mains and sewers, street cleaning, watering of municipal gardens, public fountains, or similar public-minded uses. Occasionally these uses may be metered and billed (or charged a flat fee), but usually they are unmetered and unbilled. In the latter case, the water auditor may use a default value to estimate this quantity, or implement procedures for the reliable quantification of these uses. This starts with documenting usage events as they occur and estimating the amount of water used in each event. (See Unbilled unmetered consumption)
View Service Connection Diagram	This is the average length of customer service line, Lp, that is owned and maintained by the customer; from the point of ownership transfer to the customer water meter, or building line (if unmetered). The quantity is one of the data inputs for the calculation of Unavoidable Annual Real Losses (UARL), which serves as the denominator of the performance indicator: Infrastructure Leakage Index (ILI). The value of Lp is multiplied by the number of customer service connections to obtain a total length of customer owned piping in the system. The purpose of this parameter is to account for the unmetered service line infrastructure that is the responsibility of the customer for arranging repairs of leaks that occur on their lines. In many cases leak repairs arranged by the water utility on utility-maintained piping. Leaks run longer - and lose more water - on customer-owned service piping, than utility owned piping.
Average length of customer service line	If the customer water meter exists near the ownership transfer point (usually the curb stop located between the water main and the customer premises) this distance is zero because the meter and transfer point are the same. This is the often encountered configuration of customer water meters located in an underground meter box or "pit" outside of the customer's building. The Free Water Audit Software asks a "Yes/No" question about the meter at this location. If the auditor selects "Yes" then this distance is set to zero and the data grading score for this component is set to 10.
Find	If water meters are typically located inside the customer premise/building, or properties are unmetered, it is up to the water auditor to estimate a system-wide average Lp length based upon the various customer land parcel sizes and building locations in the service area. Lp will be a shorter length in areas of high density housing, and a longer length in areas of low density housing and varied commercial and industrial buildings. General parcel demographics should be employed to obtain a composite average Lp length for the entire system.
	Refer to the "Service Connection Diagram" worksheet for a depiction of the service line/metering configurations that typically exist in water utilities. This worksheet gives guidance on the determination of the Average Length, Lp, for each configuration.
Average operating pressure Find	This is the average pressure in the distribution system that is the subject of the water audit. Many water utilities have a calibrated hydraulic model of their water distribution system. For these utilities, the hydraulic model can be utilized to obtain a very accurate quantity of average pressure. In the absence of a hydraulic model, the average pressure may be approximated by obtaining readings of static water pressure from a representative sample of fire hydrants or other system access points evenly located across the system. A weighted average of the pressure can be assembled; but be sure to take into account the elevation of the fire hydrants, which typically exist several feet higher than the level of buried water pipelines. If the water utility is compiling the water audit for the first time, the average pressure can be approximated, but with a low data grading. In subsequent years of auditing, effort should be made to improve the accuracy of the average pressure quantity. This will then qualify the value for a higher data grading.
Billed Authorized Consumption	All consumption that is billed and authorized by the utility. This may include both metered and unmetered consumption. See "Authorized Consumption" for more information.
Billed metered consumption Find	All metered consumption which is billed to retail customers, including all groups of customers such as domestic, commercial, industrial or institutional. It does NOT include water supplied to neighboring utilities (water exported) which is metered and billed. Be sure to subtract any consumption for exported water sales that may be included in these billing roles. Water supplied as exports to neighboring water utilities should be included only in the Water Exported component. The metered consumption data can be taken directly from billing records for the water audit period. The accuracy of yearly metered consumption data can be refined by including an adjustment to account for customer meter reading lag time since not all customer meters are read on the same day of the meter reading period. However additional analysis is necessary to determine the lag time adjustment value, which may or may not be significant.
Billed unmetered consumption Find	All billed consumption which is calculated based on estimates or norms from water usage sites that have been determined <u>by utility policy</u> to be left unmetered. This is typically a very small component in systems that maintain a policy to meter their customer population. However, this quantity can be the key consumption component in utilities that have not adopted a universal metering policy. This component should NOT include any water that is supplied to neighboring utilities (water exported) which is unmetered but billed. Water supplied as exports to neighboring water utilities should be included only in the Water Exported component.

Item Name	Description
Customer metering inaccuracies Find	Apparent water losses caused by the collective under-registration of customer water meters. Many customer water meters gradually wear as large cumulative volumes of water are passed through them over time. This causes the meters to under-register the flow of water. This occurrence is common with smaller residential meters of sizes 5/8-inch and 3/4 inch after they have registered very large cumulative volumes of water, which generally occurs only after periods of years. For meters sized 1-inch and larger - typical of multi-unit residential, commercial and industrial accounts - meter under-registration can occur from wear or from the improper application of the meter; i.e. installing the wrong type of meter or the wrong size of meter, for the flow pattern (profile) of the consumer. For instance, many larger meters have reduced accuracy at low flows. If an oversized meter is installed, most of the time the routine flow will occur in the low flow range of the meter, and a significant portion of it may not be registered. It is important to properly select and install all meters, but particularly large customer meters, size 1-inch and larger. The auditor has two options for entering data for this component of the audit. The auditor can enter a percentage under-registration (typically an estimated value), this will apply the selected percentage to the two categories of metered consumption to determine the volume of water not recorded due to customer meter inaccuracy. Note that this percentage is a composite average inaccuracy for <u>all</u> customer meters in the entire meter population. The percentage will be multiplied by the sum of the volumes in the Billed Metered and Unbilled Metered components. Alternatively, if the auditor has substantial data from meter testing activities, he or she can calculate their own loss volumes, and this volume may be entered directly. Note that a value of zero will be accepted but an alert will appear asking if the customer population is unmetered. Since all metered systems have some degre
Customer retail unit cost Find	The Customer Retail Unit Cost represents the charge that customers pay for water service. This unit cost is applied routinely to the components of Apparent Loss, since these losses represent water reaching customers but not (fully) paid for. Since most water utilities have a rate structure that includes a variety of different costs based upon class of customer, a weighted average of individual costs and number of customer accounts in each class can be calculated to determine a single composite cost that should be entered into this cell. Finally, the weighted average cost should also include additional charges for sewer, storm water or biosolids processing, <u>but only if</u> these charges are based upon the volume of potable water consumed. For water utilities in regions with limited water resources and a questionable ability to meet the drinking water demands in the future, the Customer Retail Unit Cost might also be applied to value the Real Losses; instead of applying the Variable Production Cost to Real Losses. In this way, it is assumed that every unit volume of leakage reduced by leakage management activities will be sold to a customer. Note: the Free Water Audit Software allows the user to select the units that are charged to customers (either \$/1,000 gallons, \$/hundred cubic feet, or \$/1,000 litres) and automatically converts these units to the units that appear in the "WATER SUPPLIED" box. The monetary units are United States dollars, \$.
Infrastructure Leakage Index (ILI) Find	The ratio of the Current Annual Real Losses (Real Losses) to the Unavoidable Annual Real Losses (UARL). The ILI is a highly effective performance indicator for comparing (benchmarking) the performance of utilities in operational management of real losses.
Length of mains	Length of all pipelines (except service connections) in the system starting from the point of system input metering (for example at the outlet of the treatment plant). It is also recommended to include in this measure the total length of fire hydrant lead pipe. Hydrant lead pipe is the pipe branching from the water main to the fire hydrant. Fire hydrant leads are typically of a sufficiently large size that is more representative of a pipeline than a service connection. The average length of hydrant leads across the entire system can be assumed if not known, and multiplied by the number of fire hydrants in the system, which can also be assumed if not known. This value can then be added to the total pipeline length. Total length of mains can therefore be calculated as: Length of Mains, miles = (total pipeline length, miles) + [ {(average fire hydrant lead length, ft) x (number of fire hydrants)} / 5,280 ft/mile ] or Length of Mains, kilometres = (total pipeline length, kilometres) + [ {(average fire hydrant lead length, metres) x (number of fire hydrants)} / 1,000 metres/kilometre ]
NON-REVENUE WATER Find	= Apparent Losses + Real Losses + Unbilled Metered Consumption + Unbilled Unmetered Consumption. This is water which does not provide revenue potential to the utility.
Number of <u>active</u> <u>AND inactive</u> service connections Find	Number of customer service connections, extending from the water main to supply water to a customer. Please note that this includes the actual number of distinct piping connections, including fire connections, whether active or inactive. This may differ substantially from the number of customers (or number of accounts). Note: this number does not include the pipeline leads to fire hydrants - the total length of piping supplying fire hyrants should be included in the "Length of mains" parameter.
Real Losses Find	Physical water losses from the pressurized system (water mains and customer service connections) and the utility's storage tanks, up to the point of customer consumption. In metered systems this is the customer meter, in unmetered situations this is the first point of consumption (stop tap/tap) within the property. The annual volume lost through all types of leaks, breaks and overflows depends on frequencies, flow rates, and average duration of individual leaks, breaks and overflows.
Revenue Water	Those components of System Input Volume that are billed and have the potential to produce revenue.
Service Connection Density Find	=number of customer service connections / length of mains

Item Name	Description			
	Apparent losses caused by accounting omissions, errant computer programming, gaps in policy, procedure, and permitting/activation of new accounts; and any type of data lapse that results in under-stated customer water consumption in summary billing reports.			
	Systematic Data Handling Errors result in a direct loss of revenue potential. Water utilities can find "lost" revenue by keying on this component.			
	Utilities typically measure water consumption registered by water meters at customer premises. The meter should be read routinely (ex: monthly) and the data transferred to the Customer Billing System, which generates and sends a bill to the customer. <u>Data Transfer Errors</u> result in the consumption value being less than the actual consumption, creating an apparent loss. Such error might occur from illegible and mis-recorded hand-written readings compiled by meter readers, inputting an incorrect meter register unit conversion factor in the automatic meter reading equipment, or a variety of similar errors.			
Systematic data handling errors	Apparent losses also occur from <u>Data Analysis Errors</u> in the archival and data reporting processes of the Customer Billing System. Inaccurate estimates used for accounts that fail to produce a meter reading are a common source of error. Billing adjustments may award customers a rightful monetary credit, but do so by creating a negative value of consumption, thus under-stating the actual consumption. Account activation lapses may allow new buildings to use water for months without meter readings and billing. Poor permitting and construction inspection practices can result in a new building lacking a billing account, a water meter and meter reading; i.e., the customer is unknown to the utility's billing system.			
Find	Close auditing of the permitting, metering, meter reading, billing and reporting processes of the water consumption data trail can uncover data management gaps that create volumes of systematic data handling error. Utilities should routinely analyze customer billing records to detect data anomalies and quantify these losses. For example, a billing account that registers zero consumption for two or more billing cycles should be checked to explain why usage has seemingly halted. Given the revenue loss impacts of these losses, water utilities are well-justified in providing continuous oversight and timely correction of data transfer errors & data handling errors.			
	If the water auditor has not yet gathered detailed data or assessment of systematic data handling error, it is recommended that the auditor apply the default value of 0.25% of the the Billed Authorized Consumption volume. However, if the auditor <u>has</u> investigated the billing system and its controls, and <u>has</u> well validated data that indicates the volume from systematic data handling error is substantially higher or lower than that generated by the default value, then the auditor should enter a quantity that was derived from the utility investigations and select an appropriate grading. <u>Note:</u> negative values are not allowed for this audit component. If the auditor enters zero for this component then a grading of 1 will be automatically assigned.			
Total annual cost of operating the water system       These costs include those for operations, maintenance and any annually incurred costs for long-term upkeep of the drinking water supply an system. It should include the costs of day-to-day upkeep and long-term financing such as repayment of capital bonds for infrastructure expa improvement. Typical costs include employee salaries and benefits, materials, equipment, insurance, fees, administrative costs and all othe sustain the drinking water supply. Depending upon water utility accounting procedures or regulatory agency requirements, it may be approp depreciation in the total of this cost. This cost should not include any costs to operate wastewater, biosolids or other systems outside of drinking				
Unauthorized consumption Find	Includes water illegally withdrawn from fire hydrants, illegal connections, bypasses to customer consumption meters, or tampering with metering or meter reading equipment; as well as any other ways to receive water while thwarting the water utility's ability to collect revenue for the water. Unauthorized consumption results in uncaptured revenue and creates an error that understates customer consumption. In most water utilities this volume is low and, if the water auditor has not yet gathered detailed data for these loss occurrences, it is recommended that the auditor apply a default value of 0.25% of the volume of water supplied. However, if the auditor has investigated unauthorized occurrences, and has well validated data that indicates the volume from unauthorized consumption is substantially higher or lower than that generated by the default value, then the auditor should enter a quantity that was derived from the utility investigations. Note that a value of zero will not be accepted since all water utilities have some volume of unauthorized consumption occurring in their system. Note: if the auditor selects the default value for unauthorized consumption, a data grading of 5 is automatically assigned, but not displayed on the Reporting Worksheet.			
	UARL (gallons)=(5.41Lm + 0.15Nc + 7.5Lc) xP,			
	or UARL (litres)=(18.0Lm + 0.8Nc + 25.0Lc) xP			
Unavoidable Annual Real Losses (UARL) Find	where: Lm = length of mains (miles or kilometres) Nc = number of customer service connections Lp = the average distance of customer service connection piping (feet or metres) (see the Worksheet "Service Connection Diagram" for guidance on deterring the value of Lp) Lc = total length of customer service connection piping (miles or km) Lc = Nc X Lp (miles or kilometres) P = Pressure (psi or metres) The UARL is a theoretical reference value representing the technical low limit of leakage that could be achieved if all of today's best technology could be successfully applied. It is a key variable in the calculation of the Infrastructure Leakage Index (ILI). Striving to reduce system leakage to a level close to the UARL is usually not needed unless the water supply is unusually expensive, scarce or both. NOTE: The UARL calculation has not yet been proven as fully valid for very small, or low pressure water distribution systems. If, in <u>allons:</u> (Lm x 20) + Nc < 3000 or P < 35psi in <u>litres:</u> (Lm x 20) + Nc < 3000 or P < 25m then the calculated UARL value may not be valid. The software does not display a value of UARL or ILI if either of these conditions is true.			

Item Name	Description					
Unbilled Authorized Consumption	All consumption that is unbilled, but still authorized by the utility. This includes Unbilled Metered Consumption + Unbilled Unmetered Consumption. See "Authorized Consumption" for more information. For Unbilled Unmetered Consumption, the Free Water Audit Software provides the auditor the option to select a default value if they have not audited unmetered activities in detail. The default calculates a volume that is 1.25% of the Water Supplied volume. If the auditor has carefully audited the various unbilled, unmetered, authorized uses of water, and has established reliable estimates of this collective volume, then he or she may enter the volume directly for this component, and not use the default value.					
Unbilled metered consumption Find	Metered consumption which is authorized by the water utility, but, for any reason, is <u>deemed by utility policy</u> to be unbilled. This might for example include metered water consumed by the utility itself in treatment or distribution operations, or metered water provided to civic institutions free of charge. It does <u>not</u> include water supplied to neighboring utilities (water exported) which may be metered but not billed.					
Unbilled unmetered consumption Find	Any kind of Authorized Consumption which is neither billed or metered. This component typically includes water used in activities such as fire fighting, flushing of water mains and sewers, street cleaning, fire flow tests conducted by the water utility, etc. In most water utilities it is a small component which is very often substantially overestimated. It does NOT include water supplied to neighboring utilities (water exported) which is unmetered and unbilled – an unlikely case. This component has many sub-components of water use which are often tedious to identify and quantify. Because of this, and the fact that it is usually a small portion of the water supplied, it is recommended that the auditor apply the default value, which is 1.25% of the Water Supplied volume. Select the default percentage to enter this value. If the water utility has carefully audited the unbilled, unmetered activities occurring in the system, and has well validated data that gives a value substantially higher or lower than the default volume, then the auditor should enter their own volume. However the default approach is recommended for most water utilities. Note that a value of zero is not permitted, since all water utilities have some volume of water in this component occurring in their system.					
Units and Conversions	The user may develop an audit based on one of three unit selections: 1) Million Gallons (US) 2) Megalitres (Thousand Cubic Metres) 3) Acre-feet Once this selection has been made in the instructions sheet, all calculations are made on the basis of the chosen units. Should the user wish to make additional conversions, a unit converter is provided below (use drop down menus to select units from the yellow unit boxes): Enter Units: Convert From Converts to 1 Million Gallons (US) = 3.06888329 Acre-feet (conversion factor = 3.06888328973723)					
Use of Option Buttons	To use the default percent value choose this button To enter a value choose this button and enter the value in the cell to the right Pent: Value: 1.25% • • • • • • • • • • • • • • • • • • •					
Variable production cost (applied to Real Losses) Find	The cost to produce and supply the next unit of water (e.g., \$/million gallons). This cost is determined by calculating the summed unit costs for ground and surface water treatment and all power used for pumping from the source to the customer. It may also include other miscellaneous unit costs that apply to the production of drinking water. It should also include the unit cost of bulk water purchased as an import if applicable. It is common to apply this unit cost to the volume of Real Losses. However, if water resources are strained and the ability to meet future drinking water demands is in question, then the water auditor can be justified in applying the Customer Retail Rate to the Real Loss volume, rather than applying the Variable Production Cost. The Free Water Audit Software applies the Variable Production costs to Real Losses by default. However, the auditor has the option on the Reporting Worksheet to select the Customer Retail Cost as the basis for the Real Loss cost evaluation if the auditor determines that this is warranted.					
Volume from own sources Find	The volume of water withdrawn (abstracted) from water resources (rivers, lakes, streams, wells, etc) controlled by the water utility, and then treated for potable water distribution. Most water audits are compiled for utility retail water distribution systems, so this volume should reflect the amount of <u>treated</u> drinking water that entered the distribution system. Often the volume of water measured at the effluent of the treatment works is slightly less than the volume measured at the raw water source, since some of the water is used in the treatment process. Thus, it is useful if flows are metered at the effluent of the treatment works. If metering exists only at the raw water source, an adjustment for water used in the treatment process should be included to account for water consumed in treatment operations such as filter backwashing, basin flushing and cleaning, etc. If the audit is conducted for a wholesale water agency that sells untreated water, then this quantity reflects the measure of the raw water, typically metered at the source.					

Item Name	Description
Volume from own sources: Master meter and supply error adjustment Find	An estimate or measure of the degree of inaccuracy that exists in the master (production) meters measuring the annual Volume from own Sources, and any error in the data trail that exists to collect, store and report the summary production data. This adjustment is a weighted average number that represents the collective error for all master meters for all days of the audit year and any errors identified in the data trail. Meter error can occur in different ways. A meter or meters may be inaccurate by under-registering flow (did not capture all the flow), or by over-registering flow (overstated the actual flow). Data error can occur due to data gaps caused by temporary outages of the meter or related instrumentation. All water utilities encounter some degree of inaccuracy in master meters and data errors in archival systems are common; thus a value of zero should <u>not</u> be entered. Enter a negative percentage or value for metered data under-registration; or, enter a positive percentage or value for metered data over-registration.
Water exported	The Water Exported volume is the bulk water conveyed and sold by the water utility to neighboring water systems that exists outside of their service area. Typically this water is metered at the custody transfer point of interconnection between the two water utilities. Usually the meter(s) are owned by the water utility that is selling the water: i.e. the exporter. If the water utility who is compiling the annual water audit sells bulk water in this manner, they are an exporter of water. Note: The Water Exported volume is sold to wholesale customers who are typically charged a wholesale rate that is different than retail rates charged to the retail customers existing within the service area. Many state regulatory agencies require that the Water Exported volume be reported to them as a quantity separate and distinct from the retail customer billed consumption. For these reasons - and others - the Water Exported volume is always quantified separately from Billed Authorized Consumption in the standard water audit. <b>Be certain not to "double-count" this quantity by including it in both the Water Exported box and the Billed Metered Consumption box of the water audit Reporting Worksheet. This volume should be included only in the Water Exported box.</b>
Water exported: Master meter and supply error adjustment Find	An estimate or measure of the volume in which the Water Exported volume is incorrect. This adjustment is a weighted average that represents the collective error for all of the metered and archived exported flow for all days of the audit year. Meter error can occur in different ways. A meter may be inaccurate by under-registering flow (did not capture all the flow), or by over-registering flow (overstated the actual flow). Error in the metered, archived data can also occur due to data gaps caused by temporary outages of the meter or related instrumentation. All water utilities encounter some degree of error in their metered data, particularly if meters are aged and infrequently tested. Occasional errors also occur in the archived data. Thus, a value of zero should <u>not</u> be entered. Enter a negative percentage or value for metered data under-registration; or enter a positive percentage or value for metered and under-registration; or enter a positive percentage or value for the testing can be used to help quantify the meter error adjustment. Corrections to data gaps or other errors found in the archived data should also be included as a portion of this meter error adjustment.
Water imported Find	The Water Imported volume is the bulk water purchased to become part of the Water Supplied volume. Typically this is water purchased from a neighboring water utility or regional water authority, and is metered at the custody transfer point of interconnection between the two water utilities. Usually the meter(s) are owned by the water supplier selling the water to the utility conducting the water audit. The water supplier selling the bulk water usually charges the receiving utility based upon a wholesale water rate.
Water imported: Master meter and supply error adjustment Find	An estimate or measure of the volume in which the Water Imported volume is incorrect. This adjustment is a weighted average that represents the collective error for all of the metered and archived imported flow for all days of the audit year. Meter error can occur in different ways. A meter may be inaccurate by under-registering flow (did not capture all the flow), or by over-registering flow (overstated the actual flow). Error in the metered, archived data can also occur due to data gaps caused by temporary outages of the meter or related instrumentation. All water utilities encounter some level of meter inaccuracy, particularly if meters are aged and infrequently tested. Occasional errors also occur in the archived metered data. Thus, a value of zero should <u>not</u> be entered. Enter a negative percentage or value for metered data under-registration; or, enter a positive percentage or value for metered data over-registration. If regular meter accuracy testing is conducted on the meter(s) - which is usually conducted by the water utility selling the water - then the results of this testing can be used to help quantify the meter error adjustment.
WATER LOSSES	= apparent losses + real losses Water Losses are the difference between Water Supplied and Authorized Consumption. Water losses can be considered as a total volume for the whole system, or for partial systems such as transmission systems, pressure zones or district metered areas (DMA); if one of these configurations are the basis of the water audit.

AWWA Free Water Audit Software: <u>Determining Water Loss Standing</u>						
Water Audit Report for:Marin Municipal Water District (2110002)Reporting Year:2019Data Validity Score:62						
Water Loss Control Planning Guide						
Functional Focus Area	Level I (0-25)	Water A	Audit Data Validity Level Level III (51-70)	/ Score Level IV (71-90)	Level V (91-100)	
Audit Data Collection	Launch auditing and loss control team; address production metering deficiencies	Analyze business process for customer metering and billing functions and water supply operations. Identify data gaps.	Establish/revise policies and procedures for data collection	Refine data collection practices and establish as routine business process	Annual water audit is a reliable gauge of year-to-year water efficiency standing	
Short-term loss control	Research information on leak detection programs. Begin flowcharting analysis of customer billing system	Conduct loss assessment investigations on a sample portion of the system: customer meter testing, leak survey, unauthorized consumption, etc.	Establish ongoing mechanisms for customer meter accuracy testing, active leakage control and infrastructure monitoring	Refine, enhance or expand ongoing programs based upon economic justification	Stay abreast of improvements in metering, meter reading, billing, leakage management and infrastructure rehabilitation	
Long-term loss control		Begin to assess long-term needs requiring large expenditure: customer meter replacement, water main replacement program, new customer billing system or Automatic Meter Reading (AMR) system.	Begin to assemble economic business case for long-term needs based upon improved data becoming available through the water audit process.	Conduct detailed planning, budgeting and launch of comprehensive improvements for metering, billing or infrastructure management	Continue incremental improvements in short-term and long-term loss control interventions	
Target-setting			Establish long-term apparent and real loss reduction goals (+10 year horizon)	Establish mid-range (5 year horizon) apparent and real loss reduction goals	Evaluate and refine loss control goals on a yearly basis	
Benchmarking			Preliminary Comparisons - can begin to rely upon the Infrastructure Leakage Index (ILI) for performance comparisons for real losses (see below table)	Performance Benchmarking - ILI is meaningful in comparing real loss standing	Identify Best Practices/ Best in class - the ILI is very reliable as a real loss performance indicator for best in class service	
	For validity scores of 5	) or below, the shaded blocks s	hould not be focus areas until l	better data validity is achieved.		

Once data have been entered into the Reporting Worksheet, the performance indicators are automatically calculated. How does a water utility operator know how well his or her system is performing? The AWWA Water Loss Control Committee provided the following table to assist water utilities is gauging an approximate Infrastructure Leakage Index (ILI) that is appropriate for their water system and local conditions. The lower the amount of leakage and real losses that exist in the system, then the lower the ILI value will be.

<u>Note:</u> this table offers an approximate guideline for leakage reduction target-setting. The best means of setting such targets include performing an economic assessment of various loss control methods. However, this table is useful if such an assessment is not possible.

General Guidelines for Setting a Target ILI (without doing a full economic analysis of leakage control options)					
Target ILI Range	Financial Considerations	Operational Considerations	Water Resources Considerations		
1.0 - 3.0	Water resources are costly to develop or purchase; ability to increase revenues via water rates is greatly limited because of regulation or low ratepayer affordability.	Operating with system leakage above this level would require expansion of existing infrastructure and/or additional water resources to meet the demand.	Available resources are greatly limited and are very difficult and/or environmentally unsound to develop.		
>3.0 -5.0	Water resources can be developed or purchased at reasonable expense; periodic water rate increases can be feasibly imposed and are tolerated by the customer population.	Existing water supply infrastructure capability is sufficient to meet long-term demand as long as reasonable leakage management controls are in place.	Water resources are believed to be sufficient to meet long-term needs, but demand management interventions (leakage management, water conservation) are included in the long-term		
>5.0 - 8.0	Cost to purchase or obtain/treat water is low, as are rates charged to customers.	Superior reliability, capacity and integrity of the water supply infrastructure make it relatively immune to supply shortages.	Water resources are plentiful, reliable, and easily extracted.		
Greater than 8.0	Greater than 8.0 Although operational and financial considerations may allow a long-term ILI greater than 8.0, such a level of leakage is not an effective utilization of water as a resource. Setting a target level greater than 8.0 - other than as an incremental goal to a smaller long-term target - is discouraged.				
Less than 1.0	Less than 1.0 If the calculated Infrastructure Leakage Index (ILI) value for your system is 1.0 or less, two possibilities exist. a) you are maintaining your leakage at low levels in a class with the top worldwide performers in leakage control. b) A portion of your data may be flawed, causing your losses to be greatly understated. This is likely if you calculate a low ILI value but do not employ extensive leakage control practices in your operations. In such cases it is beneficial to validate the data by performing field measurements to confirm the accuracy of production and customer meters, or to identify any other potential sources of error in the data.				

		AWW <u>Examples</u>	/A Free W of Compl	/ater Audit S eted and Va	oftware: <u>lidated Audits</u>		American Water W Copyright © 2014, All	WAS vs Vorks Associa Rights Rese
Example Aduit 21       Market Free Water Aduit Software: Reporting Version:       Description:       Description: <thdescription:< td=""><td>Example 1a: Million Gallon</td><td>s: <b>Example 1b</b>: Million Gallo Performance Indicators</td><td>ins: s</td><td>(</td><td>Example 2a: Megalitr Reporting Workshe</td><td>res: et</td><td>Example 2b: Megalitres: Reporting Worksheet</td><td></td></thdescription:<>	Example 1a: Million Gallon	s: <b>Example 1b</b> : Million Gallo Performance Indicators	ins: s	(	Example 2a: Megalitr Reporting Workshe	res: et	Example 2b: Megalitres: Reporting Worksheet	
	Example A	udit 1a:	/A Free W	ater Audit So	oftware:		W/ American Water Work	AS v5.0
Image: Second	Click to access definition	Water Audit Report for: City	y of Asheville	(01-11-010)	1		Copynght © 2014, All Re	ghts Reserved
Advance to extended at: MLICO ALLORS (US) PET VEAC         In state the careful ording of the problem grade why:       Mater Medican Careful or Careful ording of the problem grade why:       Mater Medican Careful ording of the problem grade why:       Mater Medican Careful ording of the problem grade why:       Mater Medican Careful ording of the problem grade why:       Mater Medican Careful ording of the problem grade why:       Mater Medican Careful ording of the problem grade why:       Mater Medican Careful ording of the problem grade why:       Mater Medican Careful ording of the problem grade why:       Mater Medican Careful ording of the problem grade why:       Mater Medican Careful ording of the problem grade why:       Mater Medican Careful ording of the problem grade why:       Mater Medican Careful ording of the problem grade why:       Mater Medican Careful ording of the problem grade why:       Mater Medican Careful ording of the problem grade why:       Mater Medican Careful ording of the problem ording ording ording of the problem ording or	Click to add a comment     Click to add a comment     Please enter data in the white cells bek     the input data by grading each compon	Reporting Year: ow. Where available, metered values should ent (n/a or 1-10) using the drop-down list to	2013 d be used; if me the left of the ir	7/2012 - 6/2013 tered values are unan uput cell. Hover the m	vailable please estimate a val	ue. Indicate your conduction of the	onfidence in the accuracy of grades	f
To select the correct data gadage bore: But during the or access data for the lightest data data data during the order of an difference of the select during the order of an difference of the select during the order of an difference of the select during the order of an difference of the select during the order of an difference of the select during the order of an difference of the select during the order of an difference of the select during the order of an difference of the select during the order of the order of the sel		All volumes t	to be entered	as: MILLION GAL	LONS (US) PER YEAR			
WATER SUPPLIED       Volume from our sources:       Image: 1       Image: 1       Volume from our sources:       Image: 1       Image: 1       Volume from our sources:       Image: 1       Image: 1<	To select the the u	correct data grading for each input, det tility meets or exceeds <u>all</u> criteria for the	ermine the hig at grade and a	hest grade where Il grades below it.		Master Meter E	rror Adjustments	
Verter insported:       10       0.000       Norw       Enter registive for or value for under registration         VATER SUPPLIED:       7,067.430       Norw       Enter registive for or value for under constration         AUTHORIZED CONSUMPTON       Billed unnetered:       10       17,750, Norw       Enter registive for or value for under constration         AUTHORIZED CONSUMPTON       Billed unnetered:       10       17,750, Norw       Enter registive for or value for under constration         AUTHORIZED CONSUMPTON       Billed unnetered:       10       17,750, Norw       Enter registive for or value for under constration         AUTHORIZED CONSUMPTON:       0.000, Norw       4,967,797, Morv       Value       Value       10,750, Norv         MATER LOSSES (Water Supplied - Authorized Consumption)       2,099,433, Norv       Value	WATER SUPPLIED	Volume from own sources: +	2 7	Enter grading	MG/Yr	> Pcnt:	Value:	MG/Yr
Water exponte:       10       10       0.000       MOY       Enter register (%) or value for under-registration         AUTHOREED CONSUMPTION       Billed metered:       0       17       7.007.430       MOY       Enter positive % or value for under-registration         AUTHOREED CONSUMPTION       Billed metered:       0       17       7.707.150       MOY       Enter positive % or value for under-registration         Unbilled functioned is a provide in the reacommended default value       115.750       MOY       Enter positive % or value for under-registration         WATER LOSSES (Water Supplied - Authorized Consumption)       2.090.633       MOY       Enter positive %       Value:         MATER LOSSES (Water Supplied - Authorized Consumption)       2.090.633       MOY       Enter positive %       Value:       Value: <td></td> <td>Water imported:</td> <td>7 n/a</td> <td>0.000</td> <td>MG/Yr + ?</td> <td></td> <td><ul> <li>265.450</li> <li>0</li> </ul></td> <td>MG/Yr</td>		Water imported:	7 n/a	0.000	MG/Yr + ?		<ul> <li>265.450</li> <li>0</li> </ul>	MG/Yr
WATER SUPPLIED:     7,067.430     Monv     Enter polite %, or value for under registration       AUTHORIZED CONSUMPTION     Biled meteret:     0     0     0.000     Monv       Biled unneteret:     0     0     0.000     Monv     Unit of on value for our engistration       Unitide unneteret:     0     0     0.000     Monv     Unit of value for our engistration       Unitide unneteret:     0     0     0.000     Monv       WATER LOSSES (Water Supplied - Authorized Consumption:     0     157.790     Monv       Default option selected for synthetic data handing errors:     1     115.000     Monv       Default option selected for synthetic data handing errors:     1     115.500     Monv       Default option selected for synthetic data handing errors:     1     157.800     Monv       Real Losses (Current Annual Real Losses - Aparent Losses:     1     1958.708     Monv       Water Losses - United towneterol     1     1.958.708     Monv       Valet Losses - United towneterol     1     1.958.708     Monv       Valet Losses - Water		Water exported: +	? n/a	0.000	MG/Yr + ?		• •	MG/Yr
AUTHORIZED CONSUMPTION Bild immetered: AUTHORIZED CONSUMPTION Bild immetered: AUTHORIZED CONSUMPTION Bild immetered: AUTHORIZED CONSUMPTION: AUTHORIZE		WATER SUPPLIED:		7.067.430	MG/Yr	Enter negative 9	% or value for under-regis 6 or value for over-registra	stration
NUMERCED CONSUMPTION       Billed metered:				.,,		Enter positive /		_
Biled unnetered: 10 10 10 000 WOY. Unbiled unnetered: 10 10 10 000 WOY. WATER LOSSES (Water Supplied - Authorized Consumption) 2,099.633 WOY. Default option selected for unauthorized consumption - a grading of 5 is applied but not displayed Customer metering inaccuracies: 10 1 11 220 WOY. Default option selected for systematic data handing errors - a grading of 5 is applied but not displayed Customer metering inaccuracies: 10 1 11 220 WOY. Default option selected for systematic data handing errors - a grading of 5 is applied but not displayed Apparent Losses: 10 11 1200 WOY. Beal Losses (Current Annual Real Losses or CARL) Real Losses (Current Annual Real Losses or CARL) Real Losses - Water Losses : 10 956.789 WOY. WATER LOSSES (Current Annual Real Losses or CARL) Real Losses - Unbiled Matered - Unbiled Matered - Unbiled Matered - Water Losses : 20,098.633 WOY. Water Losses - Barbier data handing errors - a grading of 5 is applied but not displayed MATER LOSSES (Current Annual Real Losses or CARL) Real Losses - Unbiled Matered Matered - Unbiled Matered - Unbiled Mater	AUTHORIZED CONSUMPTION	Billed metered:	2 8	4,782,250	MG/Yr		for help using option	
Ubbiled metered is 1 2 1 1 1200 Morr Ubbiled metered is greater than the recommended default value AUTHORIZED CONSUMPTION: A 4967.797 Morr WATER LOSSES (Water Supplied - Authorized Consumption) Apparent Losses Unauthorized consumption 2 2099.633 Morr Default option selected for unauthorized consumption - a grading of 5 is applied but not displayed Customer metering inaccuracies: 9 2 1 113200 Morr Default option selected for unauthorized consumption - a grading of 5 is applied but not displayed Authorized consumption: 9 2 1 113200 Morr Default option selected for unauthorized consumption - a grading of 5 is applied but not displayed Apparent Losses : 9 1 10556.7869 Morr Default option selected for Systematic data handing errors - a grading of 5 is applied but not displayed Apparent Losses : 9 1958.7869 Morr Matter Losses : 0 2 7 5 1558.7869 Morr Matter Losses = CARD Real Losses ar CARD NON-REVENUE WATER NON-REVENUE WATER NON-REVENUE WATER Non-Revenue Water Losses : 9 2 7 552.859 Morr Matter Losses = Valent Obtions enclosed at the outbook of property line? Average length of calconer service line has been applied but not displayed Average length of calconer service line has been applied at the calconer service line has been applied Average length of apparent Losses: 9 2 1 1253 pair Matter Losses Protection service line has been applied Average length of apparent Losses: 9 2 1 10 1533 pair Matter Losses Protection service line has been applied Average length of apparent Losses: 9 2 1 10 1533 pair Matter Losses Protection cost (appl		Billed unmetered:	? n/a	0.000	MG/Yr		buttons below	
Unblied Unimetered is greater than the recommended default value           AUTHORIZED CONSUMPTION:         Image: Consumption:		Unbilled metered: +	2 7	27.757	MG/Yr	Pont:	Value:	NO NO
AUTHORIZED CONSUMPTION:     4,967.797     No/Yr     Use bulloos to select state and apparent Losses Unauthorized consumption     2,099.633     No/Yr     Use bulloos to select state apparent Losses Unauthorized consumption     7,069     No/Yr       Default option selected for unauthorized consumption - a grading of 5 is applied but not displayed Customer meeting inaccuracies     7,011200     MO/Yr       Default option selected for unauthorized consumption - a grading of 5 is applied but not displayed Customer meeting inaccuracies:     1,958.709     MO/Yr       Default option selected for Systematic data handing errors - a grading of 5 is applied but not displayed Apparent Losses:     1,958.709     MO/Yr       Real Losses (Current Annual Real Losses or CARL) Real Losses = Water Losses - Apparent Losses:     2,099.633     MO/Yr       Water Losses = Videl under dominetroid     2,285.100     MO/Yr       Water Losses = Unable dimeterid     2,099.633     MO/Yr       Water Losses = Unable dimeterid     2,099.633     MO/Yr       Water Losse = Unable dimeterid     2,097.633     1,236.5     Inide		Unbilled Unmetered volume entered	is greater than	the recommended	default value		A 1157.790	MO/TI
WATER LOSSES (Water Supplied - Authorized Consumption)       2,099.633       MG/Yr         Apparent Losses       Unauthorized consumption:       17,669       MG/Yr         Default option selected for unauthorized consumption:       17,669       MG/Yr         Default option selected for unauthorized consumption:       17,669       MG/Yr         Systemaic data handling errors:       11,920       MG/Yr         Default option selected for systematic data handling errors:       11,958, MG/Yr         Real Losses (Current Annual Real Losses of CARL)       Real Losses - Apparent Losses:       149,582,769         Real Losses - Unabled Metered       NOV.REVENUE WATER       2,096,33       MG/Yr         NON-REVENUE WATER       2,205,80       MG/Yr       MG/Yr         Non-Revenue Water       2,205,80       MG/Yr       MG/Yr         Water Losses - Unabled Metered       Unauthorized connection density:       2,205,80       MG/Yr         Water Losses - Unabled Metered       Unauthorized connection density:       2,205,80       MG/Yr         Water Losses - Unabled Metered       Length of mains:       2,205,80       MG/Yr         Water Losses - Unabled Metered       Length of mains:       2,205,80       MG/Yr         Water Losses - Unable Metered       Length of mains:       2,07       55,260       <		AUTHORIZED CONSUMPTION:		4,967.797	MG/Yr		Use buttons to select percentage of water supplied	
Abademic Losses Unauthorized consumption: I T 17,868 MG/Y Default option selected for unauthorized consumption = a grading of 5 is applied but not displayed Customer metric for unauthorized consumption = a grading of 5 is applied but not displayed Apparent Losses: I 19,558,769 MG/Y Real Losses = Water Losses - Apparent Losses: I 1,958,769 MG/Y Watter Losses = Water Losses - Apparent Losses: I 1,958,769 MG/Y Watter Losses = Water Losses - Apparent Losses: I 1,958,769 MG/Y Water Losses = Water Losses - Apparent Losses: I 1,958,769 MG/Y Water Losses = Water Losses - Apparent Losses: I 1,958,769 MG/Y Water Losses = Water Losses - Apparent Losses: I 1,958,769 MG/Y Water Losses - Unbied Memory - Unbied Memory - Monitor displayed Apparent Losses: I 1,958,769 MG/Y Water Losses - Unbied Memory - Mathematice duration of the Mater Authory - Mater Losses - Apparent Losses: I 1,958,769 MG/Y Water Losses - Unbied Memory - Unbied Memory - Mater Losses - Apparent Losses: I 1,958,769 MG/Y Water Losses - Unbied Memory - Unbied Memory - Mater Losses - CARL) Real Losse - CARL Length of mains: I I I 1,228,5 m Mas Number of active AND inactive serve ince Anone anne Service connection density: I 1,228,5 m Mas Number of active AND inactive serve ince Anone anne Are cusbmer metris typically located at the curbitop or property Ine? Average length of customer service line has been set to zero and a data grading score of 10 has been applied Average operating pressure: I I I I I I I I I I I I I I I I I I I	WATER LOSSES (Water Supplied	d - Authorized Consumption)		2,099.633	MG/Yr	-	OR value	
Default option selected for unauthorized consumption - a grading of 5 is applied but not displayed Customer metering inaccuracies. 7 1 111:20, MG/Yr Systematic data handing errors - a grading of 5 is applied but not displayed Apparent Losses : 1 408.444 MG/Y Real Losses = Water Losses - Apparent Losses: 1 1,958.789 MG/Yr WATER LOSSES: 2,099.633 MG/Yr Water Losses - Water Losses - Apparent Losses: 1 1,958.789 MG/Yr Water Losses - Water Losses - Apparent Losses: 2 1,958.780 MG/Yr Water Losses - Unbilde Utametered SYSTEM DATA Length of mains: 1 4 1,225 mites Number of active AND inactive service connection density Average length of customer service into a ser	Apparent Losses	Unauthorized consumption: +	2	17 669	MG/Yr	Pcnt:	▼ Value:	MG/Yr
Customer metering inaccuracies:       Image: Control of Systematic data handling errors - a grading of Sis applied but not displayed Apparent Losses:       Image: Control of Systematic data handling errors - a grading of Sis applied but not displayed Apparent Losses:       Image: Control of Systematic data handling errors - a grading of Sis applied but not displayed Apparent Losses:       Image: Control of Systematic data handling errors - a grading of Sis applied but not displayed Apparent Losses:       Image: Control of Sis applied but not displayed Apparent Losses:       Image: Control of Sis applied but not displayed Apparent Losses:       Image: Control of Sis applied but not displayed Apparent Losses:       Image: Control of Sis applied but not displayed Apparent Losses:       Image: Control of Sis applied but not displayed Apparent Losses:       Image: Control of Sis applied but not displayed Apparent Losses:       Image: Control of Sis applied but not displayed Apparent Losses:       Image: Control of Sis applied but not displayed Apparent Losses:       Image: Control of Sis applied but not displayed Apparent Losses:       Image: Control of Sis applied but not displayed Apparent Losses:       Image: Control of Sis applied but not displayed Apparent Losses:       Image: Control of Sis applied but not displayed Apparent Losses:       Image: Control of Sis applied but not displayed Apparent Losses:       Image: Control of Sis applied but not displayed Apparent Losses:       Image: Control of Sis applied Bit not Control of Sis applied Bit not Sis applied	Default optic	on selected for unauthorized consum	ption - a grad	ing of 5 is applied	but not displayed	0.2070		
Systematic data handling errors: 10 10 5 1 11,666 MG/Y Default option selected for Systematic data handling errors: a grading of 5 is applied but not displayed Apparent Losses: 1 10,558,769 MG/Y Real Losses = Water Losses or CARLI Real Losses = Water Losses or CARLI Real Losses = Water Losses: 2 1,958,769 MG/Y WATER LOSSES: 2,009,633 MG/Y NON-REVENUE WATER NON-REVENUE WATER NON-REVENUE WATER NON-REVENUE WATER: 2 2,285,180 MG/Y WATER LOSSES: 2,009,633 MG/Y NON-REVENUE WATER NON-REVENUE WATER: 2 2,285,180 MG/Y WATER LOSSES: 2,009,633 MG/Y NON-REVENUE WATER Non-Revenue and the second of		Customer metering inaccuracies:	2 7	111.220	MG/Yr	2.26%	• •	MG/Yr
Default option selected for Systematic data handling errors - a grading of 5 is applied but not displayed Apparent Losses: 12 140.844 WAY Real Losses (Current Annual Real Losses or CARL) Real Losses = Water Losses: 2 1,956.789 MG/Yr WATER LOSSES: 2,099.633 MG/Yr NON-REVENUE WATER 0.0.4.4.4.4.4.4.4.4.4.4.4.4.4.4.4.4.4.4		Systematic data handling errors:	? 5	11.956	MG/Yr	0.25%	• •	MG/Yr
Real Losses - Water Losses - Apparent Losses:       2       1,958.788       MG/Yr         WATER LOSSES:       2,099.633       MG/Yr         NUN-REVENUE WATER:       2       2,285.180       MG/Yr         Vater Losses - Unbilde Metered + Unbilde Unmetered       2       2,285.180       MG/Yr         SYSTEM DATA       Length of mains:       2       2       1,236.5       m/es         Number of active AND inactive service connections:       2       2       55.260       conn.mile main         Are customer meters typically located at the curtostop or property Ine?       Yes       (length of service line, brown the utity)         Average length of customer service line:       2       4       145.3       psi         COST DATA       Total annual cost of operating water system:       2       1       53.260       Streer         Customer retail unit cost (applied to Apparent Losses):       2       1       53.223       Stri00 cubic fiest (ccf)       Streer         Customer retail unit cost (applied to Real Losses):       2       1       53.232       \$MMinon gailous:       Use Customer Retail Unit Cost to wake real losses         WATER AUDIT DATA VALIDITY SCORE:       *** YOUR SCORE IS: 72 out of 100 ***       A weighted scale for the components of consumption and water loss is included in the calculation of the Water Audit Data Validit	Default o	ption selected for Systematic data ha Apparent Losses:	andling errors	s - a grading of 5 i 140.844	s applied but not displaye MG/Yr	ed		
WATER LOSSES:       2,099.633       MGYr         NON-REVENUE WATER       2,285.180       MGYr         Water Losses + Unbilled Ummetered       2,285.180       MGYr         Water Losses + Unbilled Ummetered       1,236.5       miles         SYSTEM DATA       Length of mains:       2       1         Number of active AND inactive service onnections:       2       7       55.256         Service connection density:       2       45       conn.mile main         Are customer meters typically located at the curbstop or property line?       Yees       (length of service line, beyond the property boundery, that is the responsibility of the utity)         Average length of customer service line has been set to zero and a data grading score of 10 has been applied       Average operating pressure:       2       10       \$33,630.676       \$Ywar         Cost DATA       Total annual cost of operating water system:       2       10       \$33,630.676       \$Ywar         Customer retail unit cost (applied to Apparent Losses):       2       10       \$33,630.676       \$Ywar         Customer retail unit cost (applied to Real Losses):       2       10       \$33,630.676       \$Ywar         Customer retail unit cost (applied to Real Losses):       2       10       \$33,630.676       \$Ywar         Average length of	Real Losses (Current Annual Rea Real Losses = V	I <u>Losses or CARL)</u> Vater Losses - Apparent Losses:	2	1,958.789	MG/Yr			
NON-REVENUE WATER       NON-REVENUE WATER:       Z.285.180       MG/Vr         = Water Losses + Unbilled Metered + Unbilled Unmetered       Europh of mains:       Q.4       1.226.5       miles         SYSTEM DATA       Length of mains:       Q.4       1.528.5       connumber of active AND inactive service connections:       Q.7       555.265         Service connection density:       Q.7       45       connumber main         Are customer meters typically located at the curbstop or property line?		WATER LOSSES:		2,099.633	MG/Yr			
NON-REVENUE WATER:       Image: Control of the served in the serve served in the served	NON-REVENUE WATER							-
Were real unit oct (applied to Real Losses):  Water Aubling Data Validity Score  Water Aubling Data Validity Score  Water Aubling Data Validity Score  Water Stream Content of the main o		NON-REVENUE WATER:	2	2,285.180	MG/Yr			
Length of mains:	= Water Losses + Unbilled Metered + Ur	ibilied Unmetered						-
Number of active AND inactive service connection density:       1 <td>STOTEMBAIA</td> <td>l ength of mains: +</td> <td>2 4</td> <td>1 236 5</td> <td>miles</td> <td></td> <td></td> <td></td>	STOTEMBAIA	l ength of mains: +	2 4	1 236 5	miles			
Service connection density: 2 45 conn/mile main Are customer meters typically located at the curbstop or property line? Average length of customer service line: 1 2	Number of active	AND inactive service connections:	2 7	55,256	in itos			
Are customer meters typically located at the curbstop or property line? Average length of customer service line: 2 2 Ves (length of service line, beyond the property boundary, that is the responsibility of the utility) Average length of customer service line has been set to zero and a data grading score of 10 has been applied Average operating pressure: 2 2 4 145.3 psi COST DATA Total annual cost of operating water system: 2 2 4 145.3 psi COST DATA Total annual cost of operating water system: 2 2 10 \$33,630,676 \$?/Year Customer retail unit cost (applied to Apparent Losses): 2 2 10 \$33,630,676 \$?/Year Customer retail unit cost (applied to Real Losses): 2 2 6 \$335.94 \$?/Million gallons Use Customer Retail Unit Cost to value real losses WATER AUDIT DATA VALIDITY SCORE: Prover SCORE IS: 72 out of 100 *** A weighted scale for the components of consumption and water loss is included in the calculation of the Water Audit Data Validity Score PRORITY AREAS FOR ATTENTION: Based on the information provided, audit accuracy can be improved by addressing the following components: 1: Volume from own sources 2: Variable production cost (applied to Real Losses): 3: Unauthorized consumption		Service connection density:	2	45	conn./mile main			
Average length of customer service line: 2	Are customer meters typically locat	ed at the curbstop or property line?		Yes	(longth of convice lin	bound the pror	oct:	
Average length of customer service line has been set to zero and a data grading score of 10 has been applied Average operating pressure: • • • • • • • • • • • • • • • • • • •	Avera	age length of customer service line;	2		boundary, that is the	e responsibility of t	he utility)	
COST DATA         Total annual cost of operating water system:         4       2       10       \$33,630,676       \$/Year         Customer retail unit cost (applied to Apparent Losses):       4       2       10       \$33,630,676       \$/Year         Variable production cost (applied to Real Losses):       4       2       10       \$33,630,676       \$/Year         Variable production cost (applied to Real Losses):       4       2       10       \$33,630,676       \$/Year         WATER AUDIT DATA VALIDITY SCORE:         WATER AUDIT DATA VALIDITY SCORE:         Mater Score IS: 72 out of 100 ***         A weighted scale for the components of consumption and water loss is included in the calculation of the Water Audit Data Validity Score         PRIORITY AREAS FOR ATTENTION:         Based on the information provided, audit accuracy can be improved by addressing the following components:         1: Volume from own sources       2         2: Variable production cost (applied to Real Losses)       3         3: Unauthorized consumption       4	Average length of	customer service line has been set to Average operating pressure:	o zero and a	data grading scor 145.3	e of 10 has been applied psi			
Total annual cost of operating water system: * ? 10 \$33,630,676 \$/Year   Customer retail unit cost (applied to Apparent Losses): * ? 10 \$33,220 \$/100 cubic feet (ccf)   Variable production cost (applied to Real Losses): * ? ? ? ?   WATER AUDIT DATA VALIDITY SCORE:   *** YOUR SCORE IS: 72 out of 100 ***   A weighted scale for the components of consumption and water loss is included in the calculation of the Water Audit Data Validity Score   PRIORITY AREAS FOR ATTENTION:   Based on the information provided, audit accuracy can be improved by addressing the following components:   1: Volume from own sources   2: Variable production cost (applied to Real Losses)	COST DATA							
Variable production cost (applied to Apparent Losses):	Total ann	ual cost of operating water system:	2 10	\$33,630,676	\$/Year			
WATER AUDIT DATA VALIDITY SCORE: *** YOUR SCORE IS: 72 out of 100 *** A weighted scale for the components of consumption and water loss is included in the calculation of the Water Audit Data Validity Score PRIORITY AREAS FOR ATTENTION: Based on the information provided, audit accuracy can be improved by addressing the following components: 1: Volume from own sources 2: Variable production cost (applied to Real Losses) 3: Unauthorized consumption	Customer retail unit Variable produ	ction cost (applied to Apparent Losses): +	7 6	\$3.22 \$335.94	\$/100 cubic feet (ccf) \$/Million gallons Use Cu	ustomer Retail Unit Co	ost to value real losses	
YOUR SCORE IS: 72 out of 100 ***      A weighted scale for the components of consumption and water loss is included in the calculation of the Water Audit Data Validity Score      PRIORITY AREAS FOR ATTENTION: Based on the information provided, audit accuracy can be improved by addressing the following components:      Volume from own sources      Variable production cost (applied to Real Losses)      Unauthorized consumption	WATER AUDIT DATA VALIDITY SCO	RE:						
A weighted scale for the components of consumption and water loss is included in the calculation of the Water Audit Data Validity Score  PRIORITY AREAS FOR ATTENTION: Based on the information provided, audit accuracy can be improved by addressing the following components:  1: Volume from own sources  2: Variable production cost (applied to Real Losses)  3: Unauthorized consumption		*** YC	OUR SCORE I	S: 72 out of 100 **	*			
Based on the information provided, audit accuracy can be improved by addressing the following components:  1: Volume from own sources  2: Variable production cost (applied to Real Losses)  3: Unauthorized consumption	A we	ighted scale for the components of consumpti	on and water los	s is included in the cal	culation of the Water Audit Data	Validity Score		
1: Volume from own sources 2: Variable production cost (applied to Real Losses) 3: Unauthorized consumption	Based on the information provided, audit	accuracy can be improved by addressing the	following compo	onents:				
2: Variable production cost (applied to Real Losses) 3: Unauthorized consumption	1: Volume from own sources							
3: Unauthorized consumption	2: Variable production cost (applie	d to Real Losses)						
	3: Unauthorized consumption							

<u>Exam</u>	ple Audit 1b:	AWWA Free Wa System Attributes an	ater Audit Softw d Performance I	are: Indicators	WAS v5.0 American Water Works Association Copyright © 2014, All Rights Reserved
	Water A	Audit Report for: City of Asheville Reporting Year: 2013 7/ *** YOUR WATER AUDIT DATA V	(01-11-010) /2012 - 6/2013 /ALIDITY SCORE IS: 72	2 out of 100 ***	
System Attributes:		A	Apparent Losses:	140.844	MG/Yr
		+	Real Losses:	1,958.789	MG/Yr
			Water Losses:	2,099.633	MG/Yr
		Unavoidable Annual Real	Losses (UARL):	794.34	MG/Yr
		Annual cost of A	Apparent Losses:	\$606,265	
		Annual cost	of Real Losses:	\$658,036	Valued at Variable Production Cost
Performance Indicato	<u>rs:</u>				Return to Reporting Worksheet to change this assumpiton
	Non-reve	nue water as percent by volume of	Water Supplied:	32.3%	
	Financiai: - Non-rev	enue water as percent by cost of o	perating system:	3.9%	Real Losses valued at Variable Production Cost
	Г	Apparent Losses per service con	nection per day:	6.98	gallons/connection/day
	Start I	Real Losses per service con	nection per day:	97.12	gallons/connection/day
Operational Efficie	ncy:	Real Losses per length o	of main per day*:	N/A	
	_ Real Los	ses per service connection per day	per psi pressure:	0.67	gallons/connection/day/psi
	From Above,	Real Losses = Current Annual Real	Losses (CARL):	1,958.79	million gallons/year
		Infrastructure Leakage Index (ILI	) [CARL/UARL]:	2.47	
* This performance ind	cator applies for systems with a	a low service connection density of I	less than 32 service con	nections/mile of pipeli	ne

2

Example Audit 2a:	AWWA Fre <u>Rep</u>	e Water Audit S orting Workshe	oftware: et		V American Water Wo Copyright © 2014, All	VAS v5.0 orks Association Rights Reserved
Click to access definition  Click to add a comment  Report	eport for: The City of ting Year: 2013	Calgary 1/2013 - 12/2013				
Please enter data in the white cells below. Where available, meters the input data by grading each component (n/a or 1-10) using the $\sigma$	ed values should be used; drop-down list to the left of	; if metered values are una f the input cell. Hover the r	vailable please estimate a vanouse over the cell to obtain	alue. Indicate your confider a description of the grades	ice in the accuracy	of
All volum	es to be entered as: N	MEGALITRES (THOUS	AND CUBIC METRES) PE	RYEAR		_
To select the correct data grading for e the utility meets or exceeds <u>al</u>	each input, determine the criteria for that grade a	and all grades below it.		Master Meter Error A	djustments	
WATER SUPPLIED	<	Enter grading	in column 'E' and 'J'	> Pcnt	Value:	
Volume from own	sources: + 7 7	174,324.000	ML/Yr +	7 1.00% 🖲 🔿	>	ML/Yr
Water Water	exported: + 7 n/a	8.190.131	ML/Yr + 1	7 1.00%		ML/Yr ML/Yr
		101 100 070		Enter negative % or v	alue for under-reg	gistration
WATER SU	JPPLIED:	164,488.979	ML/Yr	Enter positive % or va	alue for over-regis	tration
AUTHORIZED CONSUMPTION		105 111 000	111.04-		Click here:	
Billed u	nmetered: + ? 8	3.503.386	ML/Yr		buttons below	1
Unbilled	metered: + ? 7	166.157	ML/Yr	Pcnt	Value:	_
Unbilled u	nmetered: 🛨 🙎 6	1,444.000	ML/Yr	00	1,444.000	ML/Yr
AUTHORIZED CONSU	IMPTION: ?	130,224.811	ML/Yr	1	Use buttons to sele percentage of wate supplied	ct Pr
	2.5.2		and a second	-	OR value	
WATER LOSSES (Water Supplied - Authorized Consump	otion)	34,264.168	ML/Yr			
Apparent Losses		111 000	MINE	Pent	Value:	MI Mr
Default option selected for unauthor	rized consumption - a	411.222	hut not displayed	0.25%	<u></u>	ML/TI
Customer metering inac		1 265 429	MI/Yr	1.00%		ML/Yr
Systematic data handli	ng errors: + ?	312.778	ML/Yr	0.25%	>	ML/Yr
Default option selected for Syste	ematic data handling e	errors - a grading of 5	is applied but not display	yed		
Apparent	t Losses: 🧧	1,989.429	ML/Yr			
Real Losses (Current Annual Real Losses or CARL) Real Losses = Water Losses - Apparent	t Losses: 2	32.274.739	ML/Yr			
WATER	00050	24 264 469				
	LOSSES:	34.204.100	MI /Yr			
	LOSSES:	34,204.108	ML/Yr			
NON-REVENUE WATER	WATER: 2	35.874.325	ML/Yr			-
NON-REVENUE WATER NON-REVENUE = Water Losses + Unbilled Metered + Unbilled Unmetered	WATER: 2	34,204.108	ML/Yr ML/Yr			_
NON-REVENUE WATER = Water Losses + Unbilled Metered + Unbilled Unmetered SYSTEM DATA	WATER: 2	35,874.325	MLYr MLYr			_
NON-REVENUE WATER = Water Losses + Unbilled Metered + Unbilled Unmetered SYSTEM DATA Length	of mains: + ? 8	34,204.100 35,874.325 4,945.0	ML/Yr ML/Yr kilometers			-
NON-REVENUE WATER = Water Losses + Unbilled Metered + Unbilled Unmetered SYSTEM DATA Length Number of active AND inactive service con- Service connection	of mains: + 2 8 nections: + 7 8	34,204.106 35,874.325 4,945.0 312,075	ML/Yr ML/Yr kilometers			_
NON-REVENUE WATER NON-REVENUE = Water Losses + Unbilled Metered + Unbilled Unmetered SYSTEM DATA Length Number of <u>active AND inactive</u> service con Service connection	of mains: + ? 8 nnections: + ? 8 n density: ?	34,204.108 35,874.325 4,945.0 312,075 63	ML/Yr ML/Yr kilometers conn./km main			-
NON-REVENUE WATER = Water Losses + Unbilled Metered + Unbilled Unmetered SYSTEM DATA Length Number of <u>active AND inactive</u> service con Service connection Are customer meters typically located at the curbstop or prop	WATER: 2 of mains: + 7 8 nnections: + 7 8 n density: 7 perty line?	34,204.108 35,874.325 4,945.0 312,075 63 No	ML/Yr ML/Yr kilometers conn./km main (length of service)	line, <u>beyond</u> the property		_
NON-REVENUE WATER = Water Losses + Unbilled Metered + Unbilled Unmetered SYSTEM DATA Length Number of <u>active AND inactive</u> service con Service connection Are customer meters typically located at the curbstop or prop <u>Average</u> length of customer service s	WATER: 2 of mains: + 7 8 nnections: + 7 8 n density: 7 overty line? rvvice line: + 7 8	34,204.100 35,874.325 4,945.0 312,075 63 No 12.0	ML/Yr ML/Yr kilometers conn./km m ain metres (length of service   boundary, that is t	line, <u>beyond</u> the property he responsibility of the utili	y)	_
NON-REVENUE WATER = Water Losses + Unbilled Metered + Unbilled Unmetered SYSTEM DATA Length Number of <u>active AND inactive</u> service con Service connection Are customer meters typically located at the curbstop or prop <u>Average</u> length of customer se Average operating	WATER:     2       of mains:     +     ?       annections:     +     ?       an density:     ?       perty line?       prvice line:     +     ?       pressure:     +     ?	34,204.100 35,874.325 4,945.0 312,075 63 No 12.0 50.8	ML/Yr ML/Yr kilometers conn./km m ain metres (length of service) metres (head)	line, <u>beyond</u> the property he responsibility of the utili	y)	_
NON-REVENUE WATER = Water Losses + Unbilled Metered + Unbilled Unmetered SYSTEM DATA Length Number of <u>active AND inactive</u> service con Service connection Are customer meters typically located at the curbstop or prop <u>Average</u> length of customer service	of mains: + ? 8 nnections: + ? 8 n density: ? prvice line: + ? 8 pressure: + ? 8	34,204.108 35,874.325 4,945.0 312,075 63 No 12.0 50.8	ML/Yr ML/Yr kilometers conn./km main metres (length of service) boundary, that is t metres (head)	line, <u>beyond</u> the property he responsibility of the utili	y)	_
NON-REVENUE WATER NON-REVENUE = Water Losses + Unbilled Metered + Unbilled Unmetered SYSTEM DATA Length Number of <u>active AND inactive</u> service con Service connection Are customer meters typically located at the curbstop or prop <u>Average</u> length of customer service Average operating COST DATA	of mains: + ? 8 nections: + ? 8 n density: ? perty line? provice line: + ? 8 pressure: + ? 8	34,204.108 35,874.325 4,945.0 312,075 63 No 12.0 50.8	ML/Yr ML/Yr kilometers conn./km main metres (length of service   boundary, that is t metres (head)	line, <u>beyond</u> the property he responsibility of the utili	y)	_
NON-REVENUE WATER NON-REVENUE = Water Losses + Unbilled Metered + Unbilled Unmetered SYSTEM DATA Length Number of <u>active AND inactive</u> service con Service connectio Are customer meters typically located at the curbstop or prop <u>Average</u> length of customer se Average operating COST DATA Total annual cost of operating wate	of mains: + ? 8 nections: + ? 8 n density: ? prvice line: + ? 8 pressure: + ? 8 pressure: + ? 8	34,204.108 35,874.325 4,945.0 312,075 63 No 12.0 50.8 \$169,973,759	ML/Yr ML/Yr kilometers conn./km main metres (length of service   boundary, that is t metres (head) \$/Year	line, <u>beyond</u> the property he responsibility of the utili	y)	_
NON-REVENUE WATER = Water Losses + Unbilled Metered + Unbilled Unmetered SYSTEM DATA Length Number of <u>active AND inactive</u> service con Service connection Are customer meters typically located at the curbstop or prop <u>Average</u> length of customer see Average operating COST DATA Total annual cost of operating water Customer retail unit cost (applied to Apparen	WATER:       2         of mains:       #       2         nections:       #       7         n density:       7         prety line?       *         pressure:       #       7         appressure:       #       7         appression:       #       7 <t< td=""><td>34,204.108 35,874.325 4,945.0 312,075 63 No 12.0 50.8 \$169,973,759 \$2,35</td><td>ML/Yr ML/Yr kilometers conn./km main metres (length of service) metres (boundary, that is t metres (head) \$/Year \$/1000 litres</td><td>line, <u>beyond</u> the property he responsibility of the utili</td><td>y)</td><td>_</td></t<>	34,204.108 35,874.325 4,945.0 312,075 63 No 12.0 50.8 \$169,973,759 \$2,35	ML/Yr ML/Yr kilometers conn./km main metres (length of service) metres (boundary, that is t metres (head) \$/Year \$/1000 litres	line, <u>beyond</u> the property he responsibility of the utili	y)	_
NON-REVENUE WATER           = Water Losses + Unbilled Metered + Unbilled Unmetered           SYSTEM DATA           Length           Number of active AND inactive service con Service connection           Are customer meters typically located at the curbstop or prop Average length of customer set           Average operating           COST DATA           Total annual cost of operating wate           Customer retail unit cost (applied to Apparen Variable production cost (applied to Real	USSES:         WATER:       2         of mains:       +       2         nections:       +       7         n density:       7         prety line?       *         procenty line?       *         pressure:       +       7         8       *       7         9       *       7         4       7       9         1       Losses):       +       7	34,204,108 35,874.325 4,945.0 312,075 63 No 12.0 50.8 \$169,973,759 \$2,35 \$73,54	ML/Yr ML/Yr kilometers conn./km m ain metres (length of service   metres boundary, that is t metres (head) \$/Year \$/1000 litres \$/Megalitre ☑ Use [	line, <u>beyond</u> the property he responsibility of the utilit Customer Retail Unit Cost to v	y) alue real losses	_
NON-REVENUE WATER           = Water Losses + Unbilled Metered + Unbilled Unmetered           SYSTEM DATA           Length           Number of active AND inactive service con Service connection           Are customer meters typically located at the curbstop or prop Average length of customer se           Average operating           COST DATA           Total annual cost of operating wate Customer retail unit cost (applied to Apparen Variable production cost (applied to Real           WATER AUDIT DATA VALIDITY SCORE:	VATER: 2 of mains: + 7 8 nnections: + 7 8 n density: 7 perty line? pressure: + 7 8 pressure: + 7 8 pressure: + 7 8 er system: + 7 9 t Losses): + 7 9 il Losses): + 7 9	34,204.100 35,874.325 4,945.0 312,075 63 No 12.0 50.8 \$169,973,759 \$2,35 \$73,54	ML/Yr ML/Yr kilometers conn./km main metres (length of service   boundary, that is t metres (head) \$/Year \$/Megalitre Use (	line, <u>beyond</u> the property he responsibility of the utili Customer Retail Unit Cost to v	y) alue real losses	_
NON-REVENUE WATER         = Water Losses + Unbilled Metered + Unbilled Unmetered         SYSTEM DATA         Length         Number of active AND inactive service con Service connection         Are customer meters typically located at the curbstop or prop Average length of customer se         Average operating         COST DATA         Total annual cost of operating wate Customer retail unit cost (applied to Apparen Variable production cost (applied to Real         WATER AUDIT DATA VALIDITY SCORE:	VATER: 2 of mains: + 7 8 nnections: + 7 8 n density: 7 perty line? proise line: + 7 8 pressure: + 7 8 pressure: + 7 8 er system: + 7 8 t Losses): + 7 9 i Losses): + 7 9 *** YOUR SCC	34,204.100 35,874.325 4,945.0 312,075 63 No 12.0 50.8 \$169,973,759 \$2.35 \$73.54 ORE IS: 72 out of 100 **	ML/Yr ML/Yr kilometers conn./km main metres (length of service) boundary, that is t metres (head) \$/Year \$/1000 litres \$/Megalitre Use (	line, <u>beyond</u> the property he responsibility of the utili Customer Retail Unit Cost to v	y) alue real losses	_
NON-REVENUE WATER         = Water Losses + Unbilled Metered + Unbilled Unmetered         SYSTEM DATA         Length         Number of active AND inactive service con Service connection         Are customer meters typically located at the curbstop or prop Average length of customer se         Average operating         COST DATA         Total annual cost of operating wate Customer retail unit cost (applied to Apparen Variable production cost (applied to Real         WATER AUDIT DATA VALIDITY SCORE:         A weighted scale for the compone	WATER: 2 of mains: + 7 8 nnections: + 7 8 n density: 7 perty line? pressure: + 7 8 pressure: + 7 8 pressure: + 7 8 t Losses): + 7 9 t Losses): + 7	34,204,100 35,874.325 4,945.0 312,075 63 No 12.0 50.8 \$169,973,759 \$2.35 \$73.54 DRE IS: 72 out of 100 **	ML/Yr ML/Yr kilometers conn./km main metres (length of service) boundary, that is t metres (head) \$/Year \$/1000 litres \$/Megalitre Use t kulation of the Water Audit Da	line, <u>beyond</u> the property he responsibility of the utili Customer Retail Unit Cost to v	y) alue real losses	_
NON-REVENUE WATER         = Water Losses + Unbilled Metered + Unbilled Unmetered         SYSTEM DATA         Length         Number of active AND inactive service con Service connection         Are customer meters typically located at the curbstop or prop Average length of customer se         Average operating         COST DATA         Total annual cost of operating wate Customer retail unit cost (applied to Apparen Variable production cost (applied to Real         WATER AUDIT DATA VALIDITY SCORE:         A weighted scale for the compone         PRIORITY AREAS FOR ATTENTION*	WATER:       2         of mains:       +       ?       8         nnections:       +       ?       8         n density:       ?       ?       9         pressure:       +       ?       8         pressure:       +       ?       8         er system:       +       ?       9         t Losses):       +       ?       9         wit Losses):       +       ?       9         nt Losses):       +       ?       9         min consumption and wa       *** YOUR SCC       ***	34,204,100 35,874.325 4,945.0 312,075 63 No 12.0 50.8 \$169,973,759 \$2.35 \$73.54 DRE IS: 72 out of 100 ** ter loss is included in the ca	ML/Yr ML/Yr kilometers conn./km main metres boundary, that is t metres (head) \$/Year \$/1000 litres \$/Megalitre ☑ Use to boundary Use to \$/Megalitre ☑ Use to boundary Use to \$/Megalitre ☑ Use to \$//	line, <u>beyond</u> the property he responsibility of the utili Customer Retail Unit Cost to v	y) alue real losses	
NON-REVENUE WATER       NON-REVENUE         = Water Losses + Unbilled Metered + Unbilled Unmetered       SYSTEM DATA         SySTEM DATA       Length         Number of active AND inactive service conservice connection       Service connection         Are customer meters typically located at the curbstop or proprese dynamic active applied of customer service correse       Average length of customer service conservice         COST DATA       Total annual cost of operating wate Customer retail unit cost (applied to Apparen Variable production cost (applied to Read         WATER AUDIT DATA VALIDITY SCORE:       A weighted scale for the compone         PRIORITY AREAS FOR ATTENTION:       Bread on the information resting and and a scale action to be information resting and the scale on the scale on the information resting and the scale on the scale on the information resting and the scale on the information resting and the scale on t	USSES:         WATER:       2         of mains:       +       7       8         nnections:       +       7       8         prestyline?       *       7       8         pressure:       +       7       8         er system:       +       7       8         er system:       +       7       9         I Losses):       +       7       9         water system:       +       7	34,204,100 35,874.325 4,945.0 312,075 63 No 12.0 50.8 \$169,973,759 \$2.35 \$73.54 DRE IS: 72 out of 100 * ter loss is included in the ca	ML/Yr ML/Yr Kilometers conn./km main metres boundary, that is t metres (head) \$/Year \$/1000 litres \$/Megalitre Use t t kulation of the Water Audit Data	line, <u>beyond</u> the property he responsibility of the utili Customer Retail Unit Cost to v	y)	
NON-REVENUE WATER       NON-REVENUE         = Water Losses + Unbilled Metered + Unbilled Unmetered       SYSTEM DATA         SySTEM DATA       Length         Number of active AND inactive service conservice connection       Service connection         Are customer meters typically located at the curbstop or proprese dynamic active and the curbstop or proprese dynamic active service conservice connection         Are customer meters typically located at the curbstop or proprese dynamic active act	USSES:         WATER:       2         of mains:       +       7       8         nnections:       +       7       8         presty line?       *       7       8         pressure:       +       7       8         er system:       +       7       8         er system:       +       7       9         I Losses):       +       7       9         with cosses):       +       7       9         with cosses):       +       7       9         I Losses):       +       7       9         with cosses):       +       7       9         with cosses       +       7       9	34,204,100 35,874.325 4,945.0 312,075 63 No 12.0 50.8 \$169,973,759 \$2.35 \$73.54 DRE IS: 72 out of 100 * ter loss is included in the ca	ML/Yr ML/Yr kilometers conn./km main metres boundary, that is t metres (head) \$/Year \$/1000 litres \$/Megalitre ☑ Use t kulation of the Water Audit Data	line, <u>beyond</u> the property he responsibility of the utili Customer Retail Unit Cost to v	y)	
NON-REVENUE WATER       NON-REVENUE         = Water Losses + Unbilled Metered + Unbilled Unmetered       SYSTEM DATA         Length       Number of active AND inactive service conservice connection         Are customer meters typically located at the curbstop or proprative average length of customer service average length of customer service correlation         COST DATA       Average length of customer service conservice conservice conservice connection         Are customer meters typically located at the curbstop or proprative descenter of customer service conservice conservi	USSES:         WATER:       2         of mains:       +       2         nections:       +       7         nections:       +       7         pressure:       +       7         pressure:       +       7         ar system:       +       7         ar system: <td< td=""><td>34,204,100 35,874.325 4,945.0 312,075 63 No 12.0 50.8 \$169,973,759 \$2.35 \$73.54 DRE IS: 72 out of 100 * ter loss is included in the ca</td><td>ML/Yr ML/Yr Kilometers conn./km main metres (length of service   boundary, that is t metres (head) \$/Year \$/1000 litres \$/Megalitre Use d t culation of the Water Audit Data</td><td>line, <u>beyond</u> the property he responsibility of the utili Customer Retail Unit Cost to v</td><td>y)</td><td></td></td<>	34,204,100 35,874.325 4,945.0 312,075 63 No 12.0 50.8 \$169,973,759 \$2.35 \$73.54 DRE IS: 72 out of 100 * ter loss is included in the ca	ML/Yr ML/Yr Kilometers conn./km main metres (length of service   boundary, that is t metres (head) \$/Year \$/1000 litres \$/Megalitre Use d t culation of the Water Audit Data	line, <u>beyond</u> the property he responsibility of the utili Customer Retail Unit Cost to v	y)	
NON-REVENUE WATER       NON-REVENUE         = Water Losses + Unbilled Metered + Unbilled Unmetered       SYSTEM DATA         Length       Number of active AND inactive service conservice connection         Are customer meters typically located at the curbstop or proprative average length of customer service correspondences         Average operating         COST DATA         Total annual cost of operating wate         Customer retail unit cost (applied to Real         WATER AUDIT DATA VALIDITY SCORE:         A weighted scale for the compone         PRIORITY AREAS FOR ATTENTION:         Based on the information provided, audit accuracy can be improved b         1: Volume from own sources         2: Billed metered	USSES:         WATER:       2         of mains:       +       2         nections:       +       7       8         n density:       7       7         perty line?       +       7       8         pressure:       +       7       8         er system:       +       7       9         t Losses):       +       7       9         I Losses):       +       7       9         www.y addressing the following of the followi	34,204,108 35,874.325 4,945.0 312,075 63 No 12.0 50.8 \$169,973,759 \$2,35 \$73,54 ORE IS: 72 out of 100 ** ter loss is included in the cal components:	ML/Yr ML/Yr kilometers conn./km main metres (length of service   boundary, that is t metres (head) \$/Year \$/1000 litres \$/Megailtre Vuse   culation of the Water Audit Data	line, <u>beyond</u> the property he responsibility of the utili Customer Retail Unit Cost to v	y)	
NON-REVENUE WATER       NON-REVENUE         = Water Losses + Unbilled Metered + Unbilled Unmetered       SYSTEM DATA         Length       Number of active AND inactive service conservice connection         Are customer meters typically located at the curbstop or propraverage length of customer service operating         COST DATA         COST DATA         Total annual cost of operating wate         Customer retail unit cost (applied to Real         WATER AUDIT DATA VALIDITY SCORE:         A weighted scale for the compone         PRIORITY AREAS FOR ATTENTION:         Based on the information provided, audit accuracy can be improved be         1: Volume from own sources         2: Billed metered         3: Customer metering inaccuracies	USSES:         WATER:       2         of mains:       +       ?         nections:       +       ?         n density:       ?       8         n density:       ?       8         presty line?       *       ?       8         pressure:       +       ?       8         er system:       +       ?       9         t Losses):       +       ?       9         it Losses):       +       ?       9         with cosses):       +       ?       9         with cosses       +	34,204,100 35,874.325 4,945.0 312,075 63 No 12.0 50.8 \$169,973,759 \$2,35 \$73,54 DRE IS: 72 out of 100 ** ter loss is included in the car components:	ML/Yr ML/Yr kilometers conn./km main metres (length of service   boundary, that is t metres (head) \$/Year \$/1000 litres \$/Megailtre Use d texture text	line, <u>beyond</u> the property he responsibility of the utili Customer Retail Unit Cost to v	y)	

合	Example Audit 2	AWWA Free V System Attributes a	Vater Audit Softwa and Performance Ir	are: <u>ndicators</u>	WAS v5.0 American Water Works Association. Copyright © 2014, All Rights Reserved.
		Water Audit Report for: The City of Ca Reporting Year: 2013	lgary 1/2013 - 12/2013		
System Att	tributes:	*** YOUR WATER AUDIT DATA	A VALIDITY SCORE IS: 72	out of 100 ***	
			Apparent Losses:	1,989.429	ML/Yr
		+	Real Losses:	32,274.739	ML/Yr
		= _	Water Losses:	34,264.168	MLYr
		Unavoidable Annual R	eal Losses (UARL):	8,015.57	ML/Yr
		Annual cost of	of Apparent Losses:	\$4,675,159	
		Annual o	ost of Real Losses:	\$75,845,637	Valued at Customer Retail Unit Cost
Performan	ce Indicators:				Return to Reporting Worksheet to change this assumption
	ſ	Non-revenue water as percent by volume	of Water Supplied:	21.8%	
	Financial: -{	_ Non-revenue water as percent by cost o	f operating system :	49.6%	Real Losses valued at Customer Retail Unit Cost
	Г	Apparent Losses per service	connection per day:	17.47	litres/connection/day
		Real Losses per service	connection per day:	283.34	litres/connection/day
Operatio	onal Efficiency:	Real Losses per lengt	h of main per day*:	N/A	
	Real Lo	osses per service connection per day per me	ter (head) pressure:	5.58	litres/connection/day/m
	F	rom Above, Real Losses = Current Annual R	eal Losses (CARL):	32,274.74	ML/year
		7 Infrastructure Leakage Index	(ILI) [CARL/UARL]:	4.03	
* This perfo	ormance indicator applies for sy	stems with a low service connection density	of less than 20 service conn	nections/kilometre of	pipeline

	www.awwa.org	AWWA Free Water Audit Software: <u>Acknowledgements</u>	WAS v5.0 American Water Works Association. Copyright © 2014, All Rights Reserved.			
AW\	WA Water Audit Software Version	5.0 Developed by the Water Loss Control Com Association August, 2014	nmittee of the American Water Works			
This so currei	ftware is intended to serve as a basic to nt edition of the AWWA M36 Publication "botto	ol to compile a preliminary, or "top-down", water audit. , Water Audits and Loss Control Programs, for detailed m-up", water audit using the same water audit methodo	It is recommended that users also refer to the guidance on compiling a comprehensive, or plogy.			
DEVELOF	PED BY: Andrew Chastain-Howley, PG*, Will J. Jernigan, P.E. Cavanaug George Kunkel, P.E. Philadelpl Alain Lalonde, P.Eng. Master M Ralph Y. McCord, P.E. Louisvil David A. Sayers Delaware Rive Brian M. Skeens, P.E. CH2M H Reinhard Sturm Water Systems John H. Van Arsdel M.E. Simps	MCSM. Black & Veatch gh & Associates, P.A. nia Water Department feter Canada Inc. le Water Company er Basin Commission IILL s Optimization, Inc. son Company, Inc.				
REFEREN	ICES: - Alegre, H., Hirner, W., Baptis Best Practice' Series, 2000. I - Kunkel, G. et al, 2003. Wate Control. Journal AWWA, 95:8 - AWWA Water Audits and Lo - Service Connection Diagram	sta, J. and Parena, R. Performance Indicators for Water SBN 1 900222 272 r Loss Control Committee Report: Applying Worldwide 65 ss Control Programs, M36 Publication, 3 <sup>rd</sup> Edition, 200 s courtesy of Ronnie McKenzie, WRP Pty Ltd.	r Supply Services. IWA Publishing 'Manual of Best Management Practices in Water Loss 19			
VERSION HISTORY:						
------------------	------------------	--------------------------	--	--	--	--
Version:	Release Date:	Number of Worksheets:	Key Features and Developments			
v1	2005/ 2006	5	The AWWA Water Audit Software was piloted in 2005 (v1.0 beta). The early versions (1.x) of the software restricted data entry to units of Million Gallons per year. For each entry into the audit, users identified whether the input was measured or estimated.			
v2	2006	5	The most significant enhancement in v2 of the software was to allow the user to choose the volumetric units to be used in the audit, Million Gallons or Thousand Cubic Metres (megalitres) per year. Two financial performance indicators were added to provide feedback to the user on the cost of Real and Apparent losses.			
v3	2007	7	In v3, the option to report volumetric units in acre-feet was added. Another new feature in v3 was the inclusion of default values for two water audit components (unbilled unmetered and unauthorized consumption). v3 also included two examples of completed audits in units of million gallons and Megalitres. Several checks were added into v3 to provide instant feedback to the user on common data entry problems, in order to help the user complete an accurate water audit.			
v4 - v4.2	2010	10	v4 (and versions 4.x) of the software included a new approach to data grading. The simple "estimated" or "measured" approach was replaced with a more granular scale (typically 1-10) that reflected descriptions of utility practices and served to describe the confidence and accuracy of the input data. Each input value had a corresponding scale fully described in the Grading Matrix tab. The Grading Matrix also showed the actions required to move to a higher grading score. Grading descriptions were available on the Reporting Worksheet via a pop-up box next to each water audit input. A water audit data validity score is generated (max = 100) and priority areas for attention (to improve audit accuracy) are identified, once a user completes the requied data grading. A service connection diagram was also added to help users understand the impact of customer service line configurations on water losses and how this information should be entered into the water audit software. An acknoweldgements section was also added. Minor bug fixes resulted in the release of versions 4.1 and 4.2. A French language version was also made available for v4.2.			
ν5	2014	12	In v5, changes were made to the way Water Supplied information is entered into software, with each major component having a corresponding Master Meter Error Adjustment entry (and data grading requirement). This required changes to the data validity score calculation; v5 of the software uses a weighting system that is, in part, proportional to the volume of input components. The Grading Matrix was updated to reflect the new audit inputs and also to include clarifications and additions to the scale descriptions. The appearance of the software was updated in v5 to make the software more user-friendly and several new features were added to provide more feedback to the user. Notably, a dashboard tab has been added to provide more visual feedback on the water audit results and associated costs of Non-Revenue Water. A comments sheet was added to allow the user to track notes, comments and to cite sources used.			

Appendices 2020 Urban Water Management Plan Marin Municipal Water District



# Appendix L

# Resolution XXX on UWMP and WSCP 2020 Update

Pending Appendix

Appendices 2020 Urban Water Management Plan Marin Municipal Water District



# Appendix M

# **Plan Submittal Documentation**

Pending Appendix

# 1972-2022 Honoring Park Lands for the People



National Park Service

Q

**E**MENU

Golden Gate

National Recreation Area California

# Parks For The People

Experience a park so rich it supports 19 distinct ecosystems with over 2,000 plant and animal species. Go for a hike, enjoy a vista, have a picnic or learn about the centuries of overlapping history from California's indigenous cultures, Spanish colonialism, the Mexican Republic, US military expansion and the growth of San Francisco. All of this and more awaits you, so get out and find your park.



## 50 Years of Golden Gate

Golden Gate National Recreation Area was established on October 27, 1972

2/14/23, 3:06 PM



Plan Like A Park Ranger

Have the best trip you can with these TOP 10 tips from our Rangers. Golden Gate National Recreation Area (U.S. National Park Service)



## International Site of Conscience

At Golden Gate visitors can engage in a history and culture as varied and diverse as they are.



# A UNESCO Biosphere Reserve

A highly diverse complex of terrestrial, coastal, and marine environments.



# Find Your "Virtual" Golden Gate

Find ways to connect with GGNRA from a distance through digital opportunities and activities to do in your home or neighborhood.



**Passport Stamp** 

Visit the parks in person or virtually, and print out a passport stamp(s) for your book.



Visitor Accessibility

Find out what accessibility resources are available and how to access them.



New NPS App!

Access from the palm of your hand: download and experience all 420+ National Parks.



Most Dog-Friendly National Park

Learn more about where you can enjoy the park with your furry friends.

Was this page helpful?						
Yes						
No						
An official form of the United States government. Provided by Touchpoints						

#### CONTACT INFO

#### Mailing Address:

Golden Gate National Recreation Area Building 201, Fort Mason San Francisco , CA 94123-0022

#### Phone:

415 561-4700

STAY CONNECTED

The HARP software can provide population-level risk estimates as cancer burden or as the number of persons exposed to a selected (user-identified) cancer risk level at block level centroids.

#### 8.2.9.2 Population Estimates for Noncancer Health Impacts

A noncancer chronic, 8-hour, and acute population estimate of the number of people exposed to acute, 8-hour, and chronic HQs or HIs exceeding 0.5 or 1.0, in increments of 1.0, should also be presented. For example, a facility with a maximum chronic HI of 4.0 would present the number of people exposed to a chronic HI of 0.5, 1.0, 2.0, 3.0, and 4.0. The isopleths used in this determination should be drawn using the smallest feasible grid size. The same methods that are described in Chapter 4 and Section 8.2.9 (for the population exposure estimate for cancer risk) should be used in the chronic, 8-hour and acute population estimates. Population estimates for acute, 8-hour, and chronic health impacts should be presented separately.

#### 8.2.9.3 Factors That Can Impact Population Risk – Cumulative Impacts

Although the Hot Spots program is designed to address the impacts of single facilities and not aggregate or cumulative impacts, there are a number of known factors that influence the susceptibility of the exposed population and thus may influence population risk. Socioeconomic status influences access to health care, nutrition, and outcome after cancer diagnosis. Community unemployment can affect exposure and residency time near a facility. Factors that affect the vulnerability of the population are discussed in the report *Cumulative Impacts: Building a Scientific Foundation* (OEHHA, 2010). Information on many of these factors is relatively easy to obtain at the census tract level. The OEHHA recommends that these types of factors be considered by the risk manager, along with the quantitative measures of population risk. OEHHA is in the process of developing guidance on quantification of the impact of these factors.

### 8.2.10 Cancer Risk Evaluation of Short Term Projects

The local air pollution control districts sometimes use the risk assessment guidelines for the Hot Spots program in permitting decisions for short-term projects such as construction or waste site remediation. Frequently, the issue of how to address cancer risks from short-term projects arises.

Cancer potency factors are based on animal lifetime studies or worker studies where there is long-term exposure to the carcinogenic agent. There is considerable uncertainty in trying to evaluate the cancer risk from projects that will only last a small fraction of a lifetime. There are some studies indicating that dose rate changes the potency of a given dose of a carcinogenic chemical. In others words, a dose delivered over a short time period may have a different potency than the same dose delivered over a lifetime. The OEHHA's evaluation of the impact of early-in-life exposure has reduced some of the uncertainty in evaluating the cancer risk to the general population for shorter-term exposures, as it helps account for susceptibility to carcinogens by age at exposure (OEHHA, 2009).

Due to the uncertainty in assessing cancer risk from very short-term exposures, we do not recommend assessing cancer risk for projects lasting less than two months at the MEIR. We recommend that exposure from projects longer than 2 months but less than 6 months be assumed to last 6 months (e.g., a 2-month project would be evaluated as if it lasted 6 months). Exposure from projects lasting more than 6 months should be evaluated for the duration of the project. In all cases, for assessing risk to residential receptors, the exposure should be assumed to start in the third trimester to allow for the use of the ASFs (OEHHA, 2009). Thus, for example, if the District is evaluating a proposed 5-year mitigation project at a hazardous waste site, the cancer risks for the residents would be calculated based on exposures starting in the third trimester through the first five years of life.

For the MEIW, we recommend using the same minimum exposure requirements used for the residential receptor (i.e., no evaluation for projects less than 2 months; projects longer than 2 months but less than 6 months are assumed to last 6 months; projects longer than 6 months would be evaluated for the duration of the project). Although the off-site worker scenario assumes that the workers are 16 years of age or older with an Age-Sensitivity Factor of 1, another risk management consideration for short-term project cancer assessment is whether there are women of child bearing age at the worksite and whether the MEIW receptor has a daycare center. In this case, the Districts may wish to treat the off-site MEIW in the same way as the residential scenario to account for the higher susceptibility during the third trimester of pregnancy, and for higher susceptibility of infants and children.

Finally, the risk manager may want to consider a lower cancer risk threshold for risk management for very short-term projects. Typical District guidelines for evaluating risk management of Hot Spots facilities range around a cancer risk of 1 per 100,000 exposed persons as a trigger for risk management. Permitting thresholds also vary for each District. There is valid scientific concern that the rate of exposure may influence the risk – in other words, a higher exposure to a carcinogen over a short period of time may be a greater risk than the same total exposure spread over a much longer time period. In addition, it is inappropriate from a public health perspective to allow a lifetime acceptable risk to accrue in a short period of time (e.g., a very high exposure to a carcinogen over a short period of time resulting in a  $1 \times 10^{-5}$  cancer risk). Thus, consideration should be given for very short term projects to using a lower cancer risk trigger for permitting decisions.

	COF our future		
Explore Sc	enarios X	Q Search location	×
II.       Scenario Region       California Coast		GREENBRAE Larkspur Ferry Terminal	
Scenario Topic Flooding Sce	- 🥐	Vood High School Piper Park Doheny Dr	
0 cm Sea Level Rise	None Storm Frequency	Nordstrom       Corte Ma         REI       Corte Ma         Marsh S       Marsh S	
500 cm 300 cm	100 year	Corte Madera Amy's Drive Thrue Flood Depth	
250 cm 200 cm 175 cm 150 cm 125 cm	20 year	CHAPMAN No Data 0 cm (0 ft) 250 cm (8 2 ft)	
100 cm 75 cm 50 cm 25 cm	Annual	Damino Alto Dpen Space Preserve OCOF	
0 cm Sea Level Rise	None Storm Frequency	CoSMoS	